AN INVESTIGATION INTO CONSTRUCTION MANAGEMENT PRACTICES INFLUENCING LABOUR PRODUCTIVITY IN MULTI-STOREY BUILDING PROJECTS

Argaw Tarekegn Gurmu

BSc. Construction Technology and Management, MSc. Civil Engineering (Construction Technology and Management Major)

A thesis submitted in total fulfilment of the requirements for the degree of Doctor of Philosophy

December 2017

Melbourne School of Design Faculty of Architecture, Building and Planning The University of Melbourne

ABSTRACT

Productivity improvement in construction projects is essential for the economic growth of a country, for increasing the profit margin of contractors and for reducing the project delays and the related consequences such as liquidated damages. Previous studies indicate that management related problems are negatively influencing labour productivity in construction projects. However, understanding the management practices which are suitable for improving productivity in specific project type in a certain location, planning the appropriate practices, monitoring the implementation of the planned practices, and assessing whether the implemented practices are associated with high or low productivity can help to increase productivity. Previous studies identified the best practices for increasing productivity in infrastructure and industrial construction projects in North America. However, it is possible that none of these practices is suitable to enhance productivity in multi-storey building construction projects in Australia as the management practices and their effectiveness would depend on the context such as differences in the resources supply chain within the local market, local regulatory requirements and project types. Furthermore, there is little or no research conducted on what these practices might be and which of them are the most important in the context of multi-storey building projects in Victoria, state Australia. Also, no research has been conducted on how the management practices that could enhance productivity in multi-storey building projects are measured, planned, monitored and evaluated. Additionally, no tool has been developed to predict a productivity value or a probability of exceeding a certain productivity value when the levels of planning of the management practices are known.

The purpose of this research is to identify the management practices that have the potential to improve productivity in multi-storey building construction projects, refine and validate scoring tools for measuring, planning, controlling and evaluating the practices in the context of building projects, and develop a tool for predicting the probability of exceeding a baseline productivity factor when the levels of planning of the management practices are determined. The research adopted a two-phase exploratory sequential mixed methods design. During Phase-I, in-depth interviews were conducted with 19 experts who have been involved in the delivery of multi-storey building projects. The qualitative data was analysed, construction management practices that have the potential to improve productivity were identified, and the unweighted

scoring tools for the practices were prepared. In Phase-II, quantitative data was collected from 39 principal contractors on 39 projects using questionnaires. During stage I of the quantitative data analysis, the data collected from the 39 principal contractors' experts regarding the relative importance of the practices in enhancing the productivity of multi-storey building projects were analysed to prioritise the practices identified in Phase I, and on that basis, the weighted scoring tools were prepared. During stage II of the quantitative data analysis, the data collected from the 39 multi-storey building projects were analysed to prioritise the practices identified in Phase I, and on that basis, the weighted scoring tools were prepared. During stage II of the quantitative data analysis, the data collected from the 39 multi-storey building projects were analysed to validate the scoring tools and to develop as well as validate the logistic regression model for predicting the probability of exceeding baseline productivity factor using a sigmoid graph when the scores of the practices are known.

Overall, 47 construction management practices that have the potential to improve productivity in multi-storey building projects in the context of Victoria state, Australia have been identified and prioritised. The practices 'well-defined scope of works,' 'safety and health policy,' and 'safety and health plan' are found to be the three most important practices. The findings of the correlation analyses revealed that all the 47 construction management practices are positively associated with productivity. The output of linear regression analysis also indicated that there is a positive and statistically significant relationship between productivity and the management practices. The findings of Friedman test showed that the practices 'well-defined scope of work,' 'safety and health policy,' 'safety and health plan,' 'hazard analysis,' 'long-lead materials identification,' 'safe work method statement' and 'toolbox safety meetings' are equally important to improve productivity in multi-storey building projects. The study revealed that high levels of project delays. The use of practices also varied according to the project costs.

Long-lead materials identification, procurement plans for materials, and materials delivery schedule are found to be the three most important construction materials management practices. Construction equipment maintenance, construction equipment procurement plan, and construction equipment productivity analysis are identified as the three construction equipment management practices. Traffic control plan, project start-up plan, machinery positioning strategy, project completion plan, and dynamic site layout plan are found to be the top five practices which are related to construction

methods. Well-defined scope of work, short interval plan, buildability review, construction work packages, and regulatory requirement are found to be the five most important pre-construction phase management practices. Clear delegation of responsibility, stability of organizational structure, and crew composition are found to be the three most important human resource management practices. Finally, safety and health policy, safety and health plan, and hazard analysis are the three most important safety and health practices.

The scoring tools which can be used to measure, plan, monitor and evaluate the management practices have been refined and validated. Thus, the scoring tools for construction materials management practices, construction equipment management practices, pre-construction phase management practices, management practices related to construction methods, human resource management practices, and safety and health practices are prepared and validated. The baseline scores against which the users of the scoring tools assess the adequacy of their practices have been determined and validated. The baseline scores can be used as benchmarks to evaluate if the planned or implemented management practices on a certain multi-storey building project would be associated with high or low productivity. Hence, the baseline scores for the overall management practices, construction materials management practices, management practices, management practices, and safety and health practices, and safety and health practices are set.

Logistic regression models that can be used for predicting the probability of exceeding baseline productivity factors are built and validated. The associated sigmoid graphs have been developed to determine the corresponding probabilities when the scores of the construction management practices are known. Thus, the logistic regression models for predicting the probability of exceeding baseline productivity factors based on the aggregated scores of the construction management practices as well as based on the scores of construction materials management practices, construction equipment management practices, preconstruction phase management practices, management practices, and safety and health practices are built and validated. Additionally, linear regression models which are used to predict the productivity factors of multi-storey building

projects based on the scores of the construction management practices are developed and validated.

The study has practical implications. Contractors involved in the construction of multistorey buildings in Victoria state, Australia can implement the identified practices to enhance productivity in their projects. They can also measure their management practices and evaluate whether their practices are adequate or not. Furthermore, the contractors can use the probability-based predictive model to assess the risk of low productivity for specific levels of implementations of the construction management practices. During the project planning phase, the management practices scoring tools can be used to plan appropriate practices which can be implemented to potentially increase productivity. Moreover, the productivity predicting tools can be used to estimate the productivity of a certain multi-storey building project based on the levels of planning of the management practices. By using the estimated productivity, the likelihood of occurrence of project delays can be predicted, and corrective actions can be taken prior to commencing the construction of the project. During the construction phase, the scoring tools can be used to control the implementation of the planned practices. Finally, during the project closure phase, the scoring tools can be used to evaluate whether the implemented practices are associated with the predicted productivity. If not, lessons can be learnt, and suitable practices can be planned for future projects.

This research contributes to the body of knowledge in construction management by identifying and prioritising the management practices which are suitable to enhance labour productivity in multi-storey building construction projects. The findings of the study can facilitate international comparisons as well as sectoral comparison and provide useful information for future researchers and companies seeking construction works in Australia. This study also contributes to the growth accounting framework (neoclassical economic growth theory) by investigating the under-researched areas of the drivers of labour productivity. The technological advancement and capital deepening are the two main factors which were explained by the theory. However, the theory did not provide an in-depth explanation of the management practices. This research increases the understanding of the importance of the management practices for increasing labour productivity in the context of multi-storey building projects.

iv

DECLARATION

This is to certify that:

- the thesis comprises only my original work towards the PhD except where indicated in the Preface,
- due acknowledgement has been made in the text to all other material used,
- the thesis is fewer than 100 000 words in length, exclusive of tables, maps, bibliographies and appendices.

Argaw Tarekegn Gurmu December 2017

v

LIST OF PUBLICATIONS

The following papers have been drawn from this PhD thesis and published in peerreviewed journals and conferences. Parts of these papers appear in different chapters of this thesis.

Journal Articles

- Gurmu, A. T., and Aibinu, A. A. (2017), "Investigation of Construction Equipment Management Practices for Improving Labour Productivity in Multi-Storey Building Construction Projects", *Journal of Construction Engineering and Management*, Vol. 143 No. 10, pp. 0401708.
- Gurmu, A. T., and Aibinu, A. A. (20xx), "Survey of Management Practices Enhancing Labour Productivity in Multi-Storey Building Construction Projects", *International Journal of Productivity and Performance Management*. (Accepted, please refer Appendix-9B for acceptance letter).
- Gurmu, A. T., Aibinu, A. A. and Chan, T. K. (2016), "A study of best management practices for enhancing productivity in building projects: construction methods perspectives", *Construction Economics and Building*, Vol. 16 No. 3, pp. 1-19. (Please refer Appendix-9C).

Conference Papers

- Gurmu, A. T., Aibinu, A. A. and Chan, T. K. (2016), "An investigation into construction methods influencing productivity of building projects", *in Systems engineering test and evaluation conference (SETE)*, 2016, Engineers Australia, pp. 17.
- Gurmu, A. T., Aibinu, A. A. and Chan, T. K. (2016), "Construction Management Practices Influencing Productivity in Building Projects", *in Proceedings of the* 5th World Construction Symposium, 2016, University of Moratuwa, Sri Lanka, pp. 122-132.

vi

ACKNOWLEDGEMENT

First and foremost, glory and praise to Almighty God for giving me strength during times of difficulties and allowing me to accomplish this goal. Next, I owe my deepest gratitude to my supervisor Dr Ajibade Aibinu for his valuable comments, encouragements and motivation to do my PhD. This thesis would have been impossible without his continuous support. Dr Ajibade's kindness, generosity, patience, enthusiasm and immense knowledge helped me to become an independent researcher and finish my research.

I also would like to express my sincere gratitude to Dr Toong-Khuan Chan and Dr Robert Crawford for their constructive comments and supports. I would like to use this opportunity to convey my appreciation to The University of Melbourne for offering me the scholarship to conduct my research and to participate in the conferences which help me to improve my work. Special thanks go to the ABP Graduate Research and Industry Office staff members Jane Trewin and Ceira Barr for their continuous advice and encouragement.

I am indebted to general managers, construction directors, construction managers, projects managers and other construction professionals in the building industry of Victoria State Australia, for their participation in both phases of this research. I would like to extend my sincere appreciation to Mr Grant O'Donnell, Mr Alex Lucantonio, Mr Anthony Garvey, Mr Andrew Walter, Mr Tom Craig, Mr Michael Jamieson, Mr Bernie Nolan and other project managers and project coordinators for their support during the data collection phases of the research.

Finally, I would like to thank my family for their unconditional support, encouragement, love and patience. Special thanks go to my wife Selamawit Yifru, my son Nabeke Argaw and my daughter Loza Argaw who spent sleepless nights and provided me continuous support.

TABLE OF CONTENTS

ABSTRACT	Ι
DECLARATION	V
LIST OF PUBLICATIONS	VI
ACKNOWLEDGEMENT	VII
TABLE OF CONTENTS	VIII
LIST OF FIGURES	XII
LIST OF TABLES	XV

CHAPTER ONE	APTER ONE 1	
1 INTRODUCTION	1	
1.1 INTRODUCTION	1	
1.2 RESEARCH BACKGROUND	1	
1.3 STATEMENT OF THE PROBLEM	3	
1.4 RESEARCH QUESTIONS	4	
1.5 RESEARCH OBJECTIVES	5	
1.6 RATIONALE FOR THE RESEARCH	5	
1.6.1 LACK OF TECHNIQUES OF INTEGRATING PRODUCTIVITY AND THE LEVE	EL OF IMPLEMENTATION	
OF THE MANAGEMENT PRACTICES	5	
1.6.2 LACK OF METHOD OF MEASURING, PLANNING, MONITORING AND EVA	LUATING THE	
MANAGEMENT PRACTICES IN BUILDING CONSTRUCTION PROJECTS	6	
1.6.3 DEARTH OF RESEARCH IDENTIFYING MANAGEMENT PRACTICES HAVIN	NG POTENTIAL TO	
ENHANCE PRODUCTIVITY IN BUILDING PROJECTS	6	
1.7 RESEARCH DESIGN AND METHODOLOGY	7	
1.8 SIGNIFICANCE OF THE RESEARCH	8	
1.9 THESIS STRUCTURE	10	
1.10 SUMMARY	11	

CH	APTER TWO	13
2	LITERATURE REVIEW	13
2.1	INTRODUCTION	13
2.2	CHARACTERISTICS OF THE VICTORIAN CONSTRUCTION INDUSTRY	13
2.3	SYSTEMATIC REVIEW OF LITERATURE	19
2.4	OVERVIEW OF AUSTRALIAN CONSTRUCTION INDUSTRY PRODUCTIVITY STUDIES	24
2.5	FACTORS NEGATIVELY INFLUENCING CONSTRUCTION PROJECTS' PRODUCTIVITY	27
2.6	MANAGEMENT PRACTICES ENHANCING PRODUCTIVITY IN INDUSTRIAL AND INFRASTRUCTUR	Е
	CONSTRUCTION PROJECTS	43
2.7	SUMMARY	49

СН	APTER THREE	51
3	CONCEPTUAL FRAMEWORK	51
3.1	INTRODUCTION	51
3.2	GROWTH ACCOUNTING FRAMEWORK	51
3.3	CONCEPTUAL FRAMEWORK OF CONSTRUCTION MATERIALS MANAGEMENT PRACTICES AND	
	PRODUCTIVITY	53
3.4	CONCEPTUAL FRAMEWORK OF CONSTRUCTION EQUIPMENT MANAGEMENT PRACTICES AND	
	PRODUCTIVITY	58
3.5	CONCEPTUAL FRAMEWORK OF MANAGEMENT PRACTICES RELATED TO CONSTRUCTION	
	METHODS AND PRODUCTIVITY	61
3.6	CONCEPTUAL FRAMEWORK OF PRE-CONSTRUCTION PHASE MANAGEMENT PRACTICES AND	
	PRODUCTIVITY	65
3.7	CONCEPTUAL FRAMEWORK OF HUMAN RESOURCE MANAGEMENT PRACTICES AND	
	PRODUCTIVITY	70
3.8	CONCEPTUAL FRAMEWORK OF SAFETY AND HEALTH PRACTICES AND PRODUCTIVITY	74
		viii

3.9 SUMMARY

CHAPTER FOUR 80 4 **RESEARCH DESIGN AND METHODOLOGY** 80 4.1 INTRODUCTION 80 4.2 RESEARCH DESIGN 80 4.3 RESEARCH APPROACH 82 4.3.1 PHILOSOPHICAL ASSUMPTIONS 82 4.3.2 MIXED METHODS RESEARCH APPROACH 83 4.4 RESEARCH STRATEGY AND ACTIVITIES 84 4.4.1 QUALITATIVE DATA COLLECTION - PHASE I 84 4.4.2 QUANTITATIVE DATA COLLECTION - PHASE II 86 4.5 DATA ANALYSIS STRATEGY 89 89 4.5.1 QUALITATIVE DATA ANALYSIS - PHASE I 4.5.2 METHODS OF COMPUTING RELATIVE IMPORTANCE INDEXES AND PRODUCTIVITY FACTORS 91 92 4.5.3 TECHNIQUES OF PREPARING THE SCORING TOOLS 4.5.4 CORRELATION ANALYSIS 96 4.5.5 LINEAR REGRESSION ANALYSIS 99 100 4.5.6 LOGISTIC REGRESSION ANALYSIS 4.6 SUMMARY 102

СН	APTER FIVE	103
5	QUALITATIVE DATA ANALYSIS AND FINDINGS	103
5.1	INTRODUCTION	103
5.2	PROFILES OF THE INTERVIEW PARTICIPANTS	103
5.3	IDENTIFICATION OF CONSTRUCTION MATERIALS MANAGEMENT PRACTICES INFLUENCING	
	PRODUCTIVITY IN MULTI-STOREY BUILDING PROJECTS	104
5.4	IDENTIFICATION OF CONSTRUCTION EQUIPMENT MANAGEMENT PRACTICES INFLUENCING	
	PRODUCTIVITY IN MULTI-STOREY BUILDING PROJECTS	110
5.5	IDENTIFICATION OF MANAGEMENT PRACTICES RELATED TO CONSTRUCTION METHODS	113
5.6	IDENTIFICATION OF PRECONSTRUCTION PHASE MANAGEMENT PRACTICES INFLUENCING	
	PRODUCTIVITY IN MULTI-STOREY BUILDING PROJECTS	119
5.7	IDENTIFICATION OF HUMAN RESOURCE MANAGEMENT PRACTICES INFLUENCING	
	PRODUCTIVITY IN MULTI-STOREY BUILDING PROJECTS	126
5.8	IDENTIFICATION OF SAFETY AND HEALTH PRACTICES INFLUENCING PRODUCTIVITY IN MUL	TI-
	STOREY BUILDING PROJECTS	133
5.9	SUMMARY	137

5.9	SUMMARY	

CH	CHAPTER SIX	
6	PRELIMINARY QUANTITATIVE DATA ANALYSIS	139
6 .1	INTRODUCTION	139
6.2	PROFILES OF THE COMPANIES AND THE RESPONDENTS	139
6.3	CHARACTERISTICS OF THE PROJECTS AND COMPUTATION OF PRODUCTIVITY FACTORS	141
6.4	DESCRIPTIVE ANALYSIS OF THE CONSTRUCTION MANAGEMENT PRACTICES	143
6.5	SUMMARY	152

СНАРТ	'ER SEVEN	154
7 QU	JANTITATIVE DATA ANALYSIS AND FINDINGS	154
7.1 IN	TRODUCTION	154
7.2 RE	LATIVE IMPORTANCE OF THE CONSTRUCTION MANAGEMENT PRACTICES	154
7.3 Pr	EPARING AND VALIDATING THE WEIGHTED SCORING TOOLS OF THE CONSTRUCTION	
MA	ANAGEMENT PRACTICES	165
7.3.1	PREPARING THE WEIGHTED SCORING TOOLS	165
7.3.2	VALIDATION OF THE WEIGHTED SCORING TOOLS	169
7.3.3	RELIABILITY TEST FOR THE WEIGHTED SCORING TOOLS	179
		ix

78

7.4 Res	SULTS OF THE CORRELATION ANALYSES	180
7.4.1	RELATIONSHIP BETWEEN PRODUCTIVITY AND THE CONSTRUCTION MANAGEMENT PRACTI	CES180
7.4.2	RELATIONSHIPS AMONG THE CONSTRUCTION MANAGEMENT PRACTICES	182
7.4.3	RELATIONSHIP AMONG CONSTRUCTION MANAGEMENT PRACTICES. COMPANY PROFILES A	ND
,	PROJECT CHARACTERISTICS	190
7.5 SUN	ЛМАКУ	198
CHAPT	ER EIGHT	200
8 RE	GRESSION MODELS OF CONSTRUCTION MANAGEMENT PRACTICES AND	
- PR	ODUCTIVITY	200
8.1 INT	RODUCTION	200
8.2 BUI	LDING THE REGRESSION MODELS	200
821	LOGISTIC REGRESSION MODELS OF THE CONSTRUCTION MANAGEMENT PRACTICES AND	
0	PRODUCTIVITY	200
822	LINEAR REGRESSION MODELS OF THE CONSTRUCTION MANAGEMENT PRACTICES AND	-00
0.2.2		224
83 VA	LIDATING THE BEODESSION MODELS	224
831	VALIDATION OF THE LOCISTIC RECRESSION MODEL OF THE CONSTRUCTION MANAGEMEN	4 37
0.5.1	DRACTICES AND DRODUCTIVITY	220
022	FRACTICES AND FRODUCTIVITY	239
8.3.2	VALIDATION OF LINEAK KEGKESSION MODELS OF CONSTRUCTION MANAGEMENT PRACTI	CES 240
0 4 G	AND PRODUCTIVITY	249
8.4 SUN	AMARY	254

CHAPT	ER NINE	255
9 DIS	CUSSION OF FINDINGS	255
9.1 INT	RODUCTION	255
9.2 Rei	LATIVE IMPORTANCE OF THE CONSTRUCTION MANAGEMENT PRACTICES	255
9.3 Rei	LATIONSHIPS AMONG THE CONSTRUCTION MANAGEMENT PRACTICES	275
9.3.1	ASSOCIATIONS AMONG ALL CONSTRUCTION MANAGEMENT PRACTICES	275
9.3.2	INTER-RELATIONSHIPS AMONG THE PRACTICES WITHIN THEIR CATEGORIES	279
9.4 Api	PLICATIONS OF THE SCORING TOOLS AND PREDICTION MODELS	285
9.4.1	THE USE OF THE CONSTRUCTION MANAGEMENT PRACTICES SCORING TOOLS AND TH	IE LOGISTIC
	REGRESSION MODELS DURING PROJECT PLANNING PHASE	285
9.4.2	THE APPLICATION OF THE CONSTRUCTION MANAGEMENT PRACTICES SCORING TOOL	LS AND THE
	LINEAR REGRESSION MODELS DURING PROJECT PLANNING PHASE	297
9.4.3	THE USE OF THE SCORING TOOLS DURING CONSTRUCTION PHASE	298
9.4.4	THE APPLICATION OF THE SCORING TOOLS DURING THE PROJECT CLOSURE PHASE	299
9.5 SUN	IMARY	300

9.5 SUMMARY

CHAPTER TEN	302
10 SUMMARY AND CONCLUSION	302
10.1 INTRODUCTION	302
10.2 SUMMARY OF FINDINGS	302
10.2.1 THE PRIORITIZED CONSTRUCTION MANAGEMENT PRACTICES INFLUENCING PRODUCTIVITY	IN
MULTI-STOREY BUILDING PROJECTS	302
10.2.2 THE SCORING TOOLS OF THE CONSTRUCTION MANAGEMENT PRACTICES	306
10.2.3 THE PROBABILITY-BASED REGRESSION MODELS, THE SIGMOID GRAPHS, AND THE LINEAR	
REGRESSION MODEL	315
10.3 IMPLICATIONS OF THE STUDY	322
10.3.1 PRACTICAL IMPLICATIONS	323
10.3.2 Theoretical Implications	329
10.4 STUDY LIMITATIONS AND FUTURE RESEARCH DIRECTIONS	330
REFERENCES	332
APPENDIX 1: QUESTIONNAIRE	347

х

APPENDIX 2: RAW DATA FOR DESCRIPTIVE DATA ANALYSIS	359
APPENDIX 3: SCORES OF THE BUILDING PROJECTS	365
APPENDIX 4: CORRELATION ANALYSIS RESULTS	371
APPENDIX 5: RAW DATA FOR LOGISTIC REGRESSION ANALYSES	375
APPENDIX 6: RAW DATA FOR COMPANY PROFILES	389
APPENDIX 7: INTERVIEW RESULTS	390
APPENDIX 8: LIST OF JOURNAL ARTICLES CONTAINING THE KEYWORDS	
'PRODUCTIVITY' AND 'CONSTRUCTION'	420
APPENDIX 9: ACCEPTANCE LETTERS AND PUBLISHED PAPER	453

LIST OF FIGURES

FIGURE 1-1	SHARES OF GVA OF VARIOUS INDUSTRIES IN AUSTRALIA	2
FIGURE 1-2	THE CONTRIBUTION OF THE RESEARCH TO A THEORY	9
FIGURE 2-1	FLOW CHART FOR THE SYSTEMATIC REVIEW OF LITERATURE	22
FIGURE 2-2	CLASSIFICATION OF RESEARCH ARTICLES BASED ON COUNTRIES	23
FIGURE 2-3	CLASSIFICATION OF RESEARCH ARTICLES BASED ON AUTHORS	24
FIGURE 3-1	RELATIONSHIP BETWEEN CAPITAL PER WORKER AND OUTPUT PER WORKER	52
FIGURE 3-2	THE EFFECT OF TECHNICAL PROGRESS ON OUTPUT PER WORKER	52
FIGURE 3-3	PRELIMINARY CONCEPTUAL FRAMEWORK OF CONSTRUCTION MATERIALS MANAGEMENT	
	PRACTICES AND PRODUCTIVITY	54
FIGURE 3-4	THE MANUFACTURING PRICE INDEXES OF SELECTED BUILDING MATERIALS	55
FIGURE 3-5	PRELIMINARY CONCEPTUAL FRAMEWORK OF CONSTRUCTION EQUIPMENT MANAGEMENT	
	PRACTICES AND PRODUCTIVITY	58
FIGURE 3-6	PRELIMINARY CONCEPTUAL FRAMEWORK OF MANAGEMENT PRACTICES RELATED TO	
	CONSTRUCTION METHODS AND PRODUCTIVITY	61
FIGURE 3-7	PRELIMINARY CONCEPTUAL FRAMEWORK OF PRECONSTRUCTION PHASE MANAGEMENT	
	PRACTICES AND PRODUCTIVITY	65
FIGURE 3-8	PRELIMINARY CONCEPTUAL FRAMEWORK OF HUMAN RESOURCE MANAGEMENT PRACTICES	
	AND PRODUCTIVITY	71
FIGURE 3-9	PRELIMINARY CONCEPTUAL FRAMEWORK OF SAFETY AND HEALTH PRACTICES AND	
	PRODUCTIVITY	75
FIGURE 3-1	0 THE CONCEPTUAL FRAMEWORK OF THE MANAGEMENT PRACTICES AND PRODUCTIVITY	79
FIGURE 4-1	FLOWCHART OF THE RESEARCH PROCESS	81
FIGURE 4-2	RESEARCH STRATEGY AND ACTIVITIES	85
FIGURE 4-3	FLOW CHART FOR SCORING TOOLS PREPARATION AND VALIDATION	94
FIGURE 4-4	DIAGRAM FOR CORRELATION ANALYSES OF CONSTRUCTION MATERIALS MANAGEMENT	
	PRACTICES	96
FIGURE 4-5	DIAGRAM FOR CORRELATION ANALYSES OF CONSTRUCTION EQUIPMENT MANAGEMENT	
	PRACTICES	96
FIGURE 4-6	DIAGRAM FOR CORRELATION ANALYSES OF PRECONSTRUCTION PHASE MANAGEMENT	~-
	PRACTICES	97
FIGURE 4-7	DIAGRAM FOR CORRELATION ANALYSES OF HUMAN RESOURCE MANAGEMENT PRACTICES	97
FIGURE 4-8	DIAGRAM FOR CORRELATION ANALYSES OF THE MANAGEMENT PRACTICES RELATED TO	00
F	CONSTRUCTION METHODS	98
FIGURE 4-9	DIAGRAM FOR CORRELATION ANALYSES OF SAFETY AND HEALTH PRACTICES	98
FIGURE 4-1	U KEGRESSION ANALYSIS BETWEEN PRODUCTIVITY AND CONSTRUCTION MANAGEMENT	00
Freeman 4 1	PRACTICES.	99
FIGURE 4-1	1 FLOW CHART FOR LOGISTIC REGRESSION MODEL BUILDING AND VALIDATION	01
FIGURE 5-1	DISTRIBUTION OF THE INTERVIEW PARTICIPANTS DASED ON AREAS OF EXPERITSE	03
FIGURE 5-2	DISTRIBUTION OF INTERVIEWEES DASED ON EXPERIENCE	04
FIGURE 3-3	V ALIDATED CONCEPTUAL FRAMEWORK OF CONSTRUCTION MATERIALS MANAGEMENT DRACTICES AND DRODUCTIVITY 1	00
ELCUDE 5 4	PRACTICES AND PRODUCTIVITY	09
FIGURE 3-4	VALIDATED CONCEPTUAL FRAMEWORK OF CONSTRUCTION EQUIPMENT MANAGEMENT DRACTICES AND DRODUCTIVITY 1	12
FIGURE 5 5	FRACTICES AND FRODUCTIVITY	12
FIGURE 3-3	VALIDATED CONCEPTUAL FRAMEWORK OF IVIANAGEMENT FRACTICES RELATED TO	10
FIGURE 5 6	UNSTRUCTION METHODS	19
FIGURE 3-0	V ALIDATED CONCEPTUAL FRAMEWORK OF FRECONSTRUCTION FHASE MANAGEMENT DD ACTICES AND DDODUCTIVITY 1	26
FIGURE 5.7	I KACHCES AND I RODUCTIVITI	
FIGURE 3-7	VALIDATED CONCEPTUAL FRAMEWORK OF HUMAN RESOURCE MANAGEMENT FRACTICES AN DRODUCTIVITY 1 1	27
FIGURE 5-8	VALIDATED CONCEDTIAL FRAMEWORK OF SAFETY AND HEALTU DRACTICES AND	52
1 IOUKE J-0	VALIDATED CONCETTUAL TRAVIE WORK OF DATELT AND HEALTH FRACTICES AND PDODUCTIVITY 1	37
FIGURE 6 1	$\frac{1}{1} PDOEILES OF THE OUESTIONNAIDE SUBVEY DADTICIDANTS 1$	30
FIGURE 6.7	I ROTLES OF THE QUESTIONNAIRE SURVET I ARTICIPANTS	39 40
FIGURE 6-2	DISTRIBUTIONS OF CONTRACTORS BY ANNIHAL TUDNOVED 1	<u>40</u>
FIGUPE 6.4	DISTRIBUTIONS OF THE CONTRACTORS BY VEADS OF EVDEDIENCE 1	<u>4</u> 1
FIGURE 6-5	PERCENTAGE DISTRIBUTIONS OF THE COSTS OF THE RUILDING PROJECTS 1	41
I IGURE 0-J	TERCENTINGE DISTRIBUTIONS OF THE COSTS OF THE DUILDING TROJECTS	т 1

xii

FIGURE 6-6 PERCENTAGE DISTRIBUTIONS OF THE PROJECT DELAYS	142
FIGURE 6-7 PERCENTAGE DISTRIBUTIONS OF THE PRODUCTIVITY FACTORS OF THE PROJECTS	142
FIGURE 6-8 BOX-PLOTS OF THE CONSTRUCTION MANAGEMENT PRACTICES	145
FIGURE 7-1 BOX-PLOTS OF THE PFS OF THE PROJECTS BASED ON TOTAL CONSTRUCTION MANAGEM PRACTICES SCORES	IENT 171
FIGURE 7-2 BOX-PLOTS OF THE PFS OF THE PROJECTS BASED ON THE CONSTRUCTION MATERIALS MANAGEMENT PRACTICES SCORES	173
FIGURE 7-3 BOX-PLOTS OF THE PFS OF PROJECTS WITH LOW AND HIGH CONSTRUCTION EQUIPMENT MANAGEMENT PRACTICE SCORES	174
FIGURE 7-4 BOX-PLOT OF PROJECTS WITH LOW AND HIGH SCORES OF THE MANAGEMENT PRACTICE RELATED TO CONSTRUCTION METHODS	ES175
FIGURE 7-5 BOX-PLOTS OF THE PFS OF THE PROJECTS WITH HIGH AND LOW PRECONSTRUCTION PHA MANAGEMENT PRACTICES SCORES	ASE 177
FIGURE 7-6 BOX-PLOTS OF THE PFS OF THE PROJECTS WITH HIGH AND LOW HUMAN RESOURCE	178
FIGURE 7-7 BOX-PLOTS OF THE PFS OF THE PROJECTS WITH HIGH AND LOW SAFETY AND HEALTH	170
PRACTICES SCORES	1/9
FIGURE 8-1 SCATTER PLOT OF PRODUCTIVITY FACTORS AND CONSTRUCTION MANAGEMENT PRACT	1CES 225
FIGURE 8-2 OUTPUTS OF THE TESTS CONDUCTED TO CONFIRM THE ASSUMPTIONS OF THE LINEAR REGRESSION MODEL OF PRODUCTIVITY AND THE OVERALL CONSTRUCTION MANAGEM PD + OTHERS	IENT
PRACTICES	227
FIGURE 8-3 OUTPUTS OF THE TESTS CONDUCTED TO CHECK THE ASSUMPTIONS OF THE LINEAR	
PRACTICES.	
FIGURE 8-4 OUTPUTS OF THE TESTS CONDUCTED TO ANALYSE THE ASSUMPTIONS OF THE LINEAR	
REGRESSION MODEL OF PRODUCTIVITY AND CONSTRUCTION EQUIPMENT MANAGEMEN PRACTICES	NT 231
FIGURE 8-5 OUTPUTS OF THE TESTS CONDUCTED TO VERIFY THE ASSUMPTIONS OF THE LINEAR	
REGRESSION MODEL OF PRODUCTIVITY AND THE MANAGEMENT PRACTICES RELATED	ГО 222
EXCURST & A QUENTION METHODS	233
FIGURE 6-0 OUTPUTS OF THE TESTS CONDUCTED TO CONFIRM THE ASSUMPTIONS OF THE LINEAR	
REGRESSION MODEL OF PRODUCTIVITY AND THE PRE-CONSTRUCTION PHASE MANAGE	MENI
FIGURE 9.7 OUTDUTS OF THE TESTS CONDUCTED TO VEDIEV THE ASSUMPTIONS OF THE INFAD	233
REGRESSION MODEL OF PRODUCTED TO VERIFY THE ASSUMPTIONS OF THE LINEAR	
DDACTICES	227
FIGURE & & OUTDUITS OF THE TESTS CONDUCTED TO CONFIDM THE ASSUMPTIONS OF THE LINEAR	
PICORE 6-6 OUTPUTS OF THE TESTS CONDUCTED TO CONFIRM THE ASSUMPTIONS OF THE LINEAR RECRESSION MODEL OF PRODUCTIVITY AND THE SAFETY AND HEAT TH PRACTICES	230
FIGURE 8-9 ROC CURVE FOR VALIDATING THE LOGISTIC REGRESSION MODEL OF THE OVERALL	240
CONSTRUCTION MANAGEMENT PRACTICES AND PRODUCTIVITY	240
FIGURE 8-10 SIGMOID GRAPH OF THE OVERALL CONSTRUCTION MANAGEMENT PRACTICES	240
FIGURE 8-11 KOC CURVE FOR VALIDATING THE LOGISTIC REGRESSION MODEL OF THE CONSTRUCT	TION
MATERIALS MANAGEMENT PRACTICES AND PRODUCTIVITY	241
FIGURE 8-12 PROBABILITY GRAPH OF THE CONSTRUCTION MATERIALS MANAGEMENT PRACTICES FIGURE 8-13 ROC CURVE FOR VALIDATING THE CONSTRUCTION FOURPMENT MANAGEMENT PRACT	242 FICES
AND PRODUCTIVITY MODEL	243
FIGURE 8-14 PROBABILITY GRAPH OF THE CONSTRUCTION FOURPMENT MANAGEMENT PRACTICES	243
FIGURE 8-15 ROC CURVE FOR VALIDATING MANAGEMENT PRACTICES RELATED TO CONSTRUCTION	N
METHODS AND PRODUCTIVITY MODEL	
FIGURE 8-16 PROBABILITY PLOT OF MANAGEMENT PRACTICES RELATED TO CONSTRUCTION METHO	ODS244
FIGURE 8-17 ROC CURVE FOR VALIDATING THE PRE-CONSTRUCTION PHASE MANAGEMENT PRACT	ICES 245
FIGURE 8-18 PROBABILITY PLOT OF THE PRECONSTRUCTION PHASE MANAGEMENT PRACTICES	·····275 246
FIGURE 8-19 ROC CURVE FOR VALIDATING HRM PDACTICES AND DDODUCTIVITY MODEL	240 216
FIGURE 8-20 PROBABILITY PLOT OF THE HUMAN RESOURCE MANAGEMENT PRACTICES	240 247
FIGURE 8-21 ROC CURVE FOR VALIDATING THE SAFETY AND HEAT TH PRACTICES AND PRODUCTIV	247 ITV
MODEL	248
FIGURE 8-22 PROBABILITY PLOT OF SAFETY AND HEALTH PRACTICES	
	X111

List of Figures

FIGURE 9-1 FRAMEWORK OF FACTORS NEGATIVELY INFLUENCING PRODUCTIVITY AND CONSTRUCTION	N
MATERIALS MANAGEMENT PRACTICES	.259
FIGURE 9-2 FRAMEWORK OF FACTORS NEGATIVELY INFLUENCING PRODUCTIVITY AND CONSTRUCTION	N
EQUIPMENT MANAGEMENT PRACTICES	.261
FIGURE 9-3 FRAMEWORK OF THE MANAGEMENT PRACTICES RELATED TO CONSTRUCTION METHODS A	ND
FACTORS NEGATIVELY INFLUENCING PRODUCTIVITY	.265
FIGURE 9-4 FRAMEWORK OF THE PRE-CONSTRUCTION PHASE MANAGEMENT PRACTICES AND FACTORS	S
NEGATIVELY INFLUENCING PRODUCTIVITY	.268
FIGURE 9-5 FRAMEWORK OF THE HUMAN RESOURCE MANAGEMENT PRACTICES AND FACTORS	
NEGATIVELY INFLUENCING PRODUCTIVITY	.270
FIGURE 9-6 FRAMEWORK OF SAFETY AND HEALTH PRACTICES AND FACTORS NEGATIVELY INFLUENCE	NG
CONSTRUCTION PROJECTS' PRODUCTIVITY	.274
FIGURE 9-7 ILLUSTRATION OF THE APPLICATION OF THE PROBABILITY GRAPH OF THE OVERALL	
CONSTRUCTION MANAGEMENT PRACTICES	.289
FIGURE 9-8 APPLICATION OF THE SIGMOID GRAPH OF THE MATERIALS MANAGEMENT PRACTICES	.293
FIGURE 9-9 THE USE OF THE SIGMOID GRAPH OF THE CONSTRUCTION EQUIPMENT MANAGEMENT	
PRACTICES	.294
FIGURE 9-10 APPLICATION OF THE PROBABILITY GRAPH OF MANAGEMENT PRACTICES RELATED TO	
CONSTRUCTION METHODS	.295
FIGURE 9-11 THE USE OF THE PROBABILITY PLOT OF THE PRECONSTRUCTION PHASE MANAGEMENT	
PRACTICES	.295
FIGURE 9-12 APPLICATION THE SIGMOID GRAPH OF THE HRM PRACTICES	.296
FIGURE 9-13 APPLICATION OF THE PROBABILITY GRAPH OF SAFETY AND HEALTH PRACTICES	.297
FIGURE 10-1 PROBABILITY GRAPH PLOTTED USING THE LOGISTIC REGRESSION MODEL OF THE OVERA	LL
CONSTRUCTION MANAGEMENT PRACTICES AND PRODUCTIVITY	.316
FIGURE 10-2 PROBABILITY GRAPH PLOTTED USING THE LOGISTIC REGRESSION MODEL OF THE	
CONSTRUCTION MATERIALS MANAGEMENT PRACTICES AND PRODUCTIVITY	.317
FIGURE 10-3 PROBABILITY GRAPH PLOTTED USING THE LOGISTIC REGRESSION MODEL OF THE	
CONSTRUCTION EQUIPMENT MANAGEMENT PRACTICES AND PRODUCTIVITY	.318
FIGURE 10-4 PROBABILITY GRAPH PLOTTED USING THE LOGISTIC REGRESSION MODEL OF THE	
MANAGEMENT PRACTICES RELATED TO CONSTRUCTION METHODS AND PRODUCTIVITY	.319
FIGURE 10-5 PROBABILITY GRAPH PLOTTED USING THE LOGISTIC REGRESSION MODEL OF THE PRE-	
CONSTRUCTION PHASE MANAGEMENT PRACTICES AND PRODUCTIVITY	.320
FIGURE 10-6 PROBABILITY GRAPH PLOTTED USING THE LOGISTIC REGRESSION MODEL OF THE HUMAN	1
RESOURCE MANAGEMENT PRACTICES AND PRODUCTIVITY	.320
FIGURE 10-7 PROBABILITY GRAPH PLOTTED USING THE LOGISTIC REGRESSION MODEL OF THE SAFETY	ľ
AND HEALTH PRACTICES AND PRODUCTIVITY	.321

xiv

LIST OF TABLES

TABLE 2-1 NUMBER OF CONSTRUCTION FIRMS OPERATING IN VICTORIA IN 2014/2015	14
TABLE 2-2 THE MINIMUM WAGES OF CONSTRUCTION WORKERS IN 2015/16	15
TABLE 2-3 SUMMARY OF THE NUMBER OF JOURNAL ARTICLES.	21
TABLE 2-4 SUMMARY OF THE SYSTEMATIC REVIEW OF LITERATURE	23
TABLE 2-5 ELEMENTS, CATEGORIES AND SECTIONS OF THE BEST PRACTICES FOR INDUSTRIAL PROJECTS	s 44
TABLE 2-6 THE LEVELS AND WEIGHTS OF ON-SITE MATERIALS TRACKING TECHNOLOGIES IN THE	
CONTEXT OF INDUSTRIAL PROJECTS	45
TABLE 2-7 ELEMENTS, CATEGORIES AND SECTIONS OF THE BEST PRACTICES FOR INFRASTRUCTURE PROJECTS	46
TABLE 2-8 THE LEVELS AND WEIGHTS OF ON-SITE MATERIALS TRACKING TECHNOLOGIES IN THE	
CONTEXT OF INFRASTRUCTURE PROJECTS	47
TABLE 4-1 PROFILES OF THE INTERVIEW PARTICIPANTS	86
TABLE 4-2 SAMPLE OF THE VALIDATED SURVEY INSTRUMENT	88
TABLE 4-3 PROFILES OF THE QUESTIONNAIRE SURVEY PARTICIPANTS	89
TABLE 4-4 TEMPLATE FOR THE QUALITATIVE DATA ANALYSIS	90
TABLE 4-5 SAMPLE SCORING TOOL FOR THE PRACTICE PROCUREMENT PLAN FOR CONSTRUCTION	
Equipment	95
TABLE 5-1 SUMMARY OF THE INTERVIEW RESULTS OF CONSTRUCTION MATERIALS MANAGEMENT	105
TADLE 5.2 SUMMADV OF INTERVIEW DESULTS OF CONSTRUCTION FOURIMENT AND TOOLS MANAGEME	IUJ
PRACTICES	111
TABLE 5-3 SUMMARY OF THE INTERVIEW RESULTS OF MANAGEMENT PRACTICES RELATED TO CONSTRUCTION METHODS	114
TABLE 5-4 SUMMARY OF THE INTERVIEW RESULTS OF PRECONSTRUCTION PHASE MANAGEMENT	
PRACTICES	120
TABLE 5-5 SUMMARY OF THE INTERVIEW RESULTS OF HUMAN RESOURCE MANAGEMENT PRACTICES	127
TABLE 5-5 SUMMARY OF THE INTERVIEW RESULTS OF SAFETY AND HEALTH PRACTICES	134
TABLE 5 0 DOMINERT OF THE RATER RESOLUTION OF THE PROJECTS	143
TABLE 0 THROBOUTINE TREFORS OF THE PROJECTION MANAGEMENT PRACTICES	144
TABLE 0.2 DESCRIPTION OF THE RESPONSES OF THE CONSTRUCTION MANAGEMENT PRACTICES ON A	
Scale of 1 to 5	146
TABLE 7-1 RELATIVE IMPORTANCE INDEXES OF THE CONSTRUCTION MANAGEMENT PRACTICES	155
TABLE 7-2 NORMALITY TEST RESULTS OF THE CONSTRUCTION MANAGEMENT PRACTICES	157
TABLE 7-3 FRIEDMAN TEST RESULTS OF THE CONSTRUCTION MANAGEMENT PRACTICES	158
TABLE 7-4 OUTPUTS OF FRIEDMAN AND WILCOXON TESTS - CONSTRUCTION MATERIALS MANAGEMENT	т
PRACTICES	159
TABLE 7-5 OUTPUTS OF FRIEDMAN AND WILCOXON TESTS - MANAGEMENT PRACTICES RELATED TO	10)
Construction Methods	161
TABLE 7-6 OUTPUTS OF FRIEDMAN TEST - PRE-CONSTRUCTION PHASE MANAGEMENT PRACTICES	163
TABLE 7-7 OUTPUTS OF FRIEDMAN AND WILCOXON TESTS - HUMAN RESOURCE MANAGEMENT PRACTI	ICES
	164
TABLE 7-8 OUTPUTS OF FRIEDMAN TEST - SAFETY AND HEALTH PRACTICES	165
TABLE 7-9 DISTRIBUTION OF THE WEIGHTS OF THE CONSTRUCTION MANAGEMENT PRACTICES	167
TABLE 7 9 DISTRIBUTION OF THE VERSITIE OF THE CONSTRUCTION FOUR MAIN NORMENT PRACTICES	168
TABLE 7 10 SCORES OF THE CONSTRUCTION MANAGEMENT PRACTICES	169
TABLE 7-17 OUTPUTS OF THE NORMALITY AND T-TESTS FOR PROJECTS GROUPED BASED ON THE TOTAL	ſ
CONSTRUCTION MANAGEMENT PRACTICES SCORES	170
TABLE 7-13 OUTPUTS OF THE NORMALITY AND T-TESTS FOR PROJECTS GROUPED BASED ON MATERIAL	170
MANAGEMENT PRACTICES SCORES	172
TABLE 7-14 OLITPLITS OF THE NORMALITY TEST FOR PROJECTS GROUDED RASED ON THE SCORES OF	114
CONSTRUCTION FOILIPMENT MANAGEMENT PRACTICES	173
TARIE 7-15 OUTPUTS OF THE NORMALITY TEST OF PROJECTS GROUDED RASED ON THE SCORES OF	115
MANAGEMENT PRACTICES RELATED TO CONSTRUCTION METHODS	175
TARIE 7-16 OUTPUTS OF THE NORMALITY AND T-TESTS OF PROJECTS GROUDED RASED ON PRE-	175
CONSTRUCTION PHASE MANAGEMENT PRACTICES SCORES	176
CONSTRUCTION THEORY INCOMPANY INVITED DECRES	110

TABLE 7-17 SPEARMAN CORRELATION COEFFICIENTS OF PRODUCTIVITY AND CONSTRUCTION MANAGEMENT PRACTICES	181
TADLE 7. 18 SPEADMANI CORDEL ATION COEFFICIENTS OF THE SIX CATECODIES OF THE CONSTRUCTION	.101
TABLE /-18 SPEARMAN CORRELATION COEFFICIENTS OF THE SIX CATEGORIES OF THE CONSTRUCTION	105
MANAGEMENT PRACTICES	.103
TABLE 7-19 SPEARMAN S CORRELATION COEFFICIENTS OF THE CONSTRUCTION MATERIALS	100
MANAGEMENT PRACTICES	.186
TABLE 7-20 SPEARMAN'S CORRELATION COEFFICIENTS OF THE MANAGEMENT PRACTICES RELATED TO)
CONSTRUCTION METHODS	.187
TABLE 7-21 SPEARMAN'S CORRELATION COEFFICIENTS OF PRE-CONSTRUCTION PHASE MANAGEMENT	100
PRACTICES	.188
TABLE 7-22 SPEARMAN CORRELATION COEFFICIENTS OF HUMAN RESOURCE MANAGEMENT PRACTICE	S
	.188
TABLE 7-23 SPEARMAN CORRELATION COEFFICIENTS OF SAFETY AND HEALTH PRACTICES	.189
TABLE 7-24 SPEARMAN CORRELATION COEFFICIENTS OF PROJECT CHARACTERISTICS, COMPANY	
PROFILES, AND CONSTRUCTION MANAGEMENT PRACTICES	.190
TABLE 7-25 SPEARMAN CORRELATION COEFFICIENTS OF COMPANY PROFILES, PROJECT	
CHARACTERISTICS, AND CONSTRUCTION MATERIALS MANAGEMENT PRACTICES	.191
TABLE 7-26 SPEARMAN CORRELATION COEFFICIENTS OF COMPANY PROFILES, PROJECT	
CHARACTERISTICS, AND CONSTRUCTION EQUIPMENT MANAGEMENT PRACTICES	.193
TABLE 7-27 SPEARMAN CORRELATION COEFFICIENTS OF COMPANY PROFILES, PROJECT CHARACTERIS	TICS
AND MANAGEMENT PRACTICES RELATED TO CONSTRUCTION METHODS	.194
TABLE 7-28 SPEARMAN CORRELATION COEFFICIENTS OF COMPANY PROFILES. PROJECT	
CHARACTERISTICS. AND PRE-CONSTRUCTION PHASE MANAGEMENT PRACTICES	.195
TABLE 7-29 SPEARMAN CORRELATION COEFFICIENTS OF COMPANY PROFILES. PROJECT	
CHARACTERISTICS AND HRM PRACTICES	196
TABLE 7-30 SPEARMAN CORRELATION COEFFICIENTS OF COMPANY PROFILES PROJECT CHARACTERIS	TICS
AND SAFETY AND HEAT TH PRACTICES	197
TABLE 8-1 SELECTION OF THE LOGISTIC REGRESSION MODEL OF THE OVERALL CONSTRUCTION	.177
MANAGEMENT PRACTICES AND PRODUCTIVITY	202
TABLE 8-2 OUTPUTS OF THE LOGISTIC REGRESSION ANALYSIS OF THE OVERALL CONSTRUCTION	.202
MANAGEMENT PRACTICES	204
TABLE 8-3 SELECTION OF THE LOGISTIC REGRESSION MODEL OF THE CONSTRUCTION MATERIALS	.204
MANAGEMENT PRACTICES AND PRODUCTIVITY	206
TADI E A Outduits of the Logistic Regression Analysis of the Construction Material s	.200
MANAGEMENT PRACTICES	207
TADI E & 5 SEI ECTION OF THE LOCISTIC RECRESSION MODEL OF THE CONSTRUCTION FOURDMENT	.207
MANA CEMENT DEACTICES AND DEODUCTIVITY	200
TADLE 8.6 OUTDUTS OF THE LOCISTIC DECRESSION ANALYSIS OF THE CONSTRUCTION FOURDMENT	.209
MANA CEMENT DE ACTICES	210
TADLE 9.7 SELECTION OF THE LOCISTIC DECRESSION MODEL OF THE MANAGEMENT DRACTICES DELA	.210
TABLE 6-/ SELECTION OF THE LOGISTIC REGRESSION MODEL OF THE MANAGEMENT PRACTICES RELA TO CONSTRUCTION METHODS AND PRODUCTIVITY	1ED 212
TABLE 9 OUTPUTS OF THE LOCISTIC DECREGION ANALYSIS OF MANA CENTRUT DE ACTICES DELATED	.212
TABLE 8-0 OUTPUTS OF THE LOGISTIC REGRESSION ANALYSIS OF MANAGEMENT PRACTICES RELATED	212
TABLE & O SERVICION OF THE LOCIETY OF DECREGATION MODEL OF THE DECONSTRUCTION DUALSE	.213
TABLE 8-9 SELECTION OF THE LOGISTIC REGRESSION MODEL OF THE PRE-CONSTRUCTION PHASE	215
MANAGEMENT PRACTICES AND PRODUCTIVITY	.215
TABLE 8-10 OUTPUTS OF THE LOGISTIC REGRESSION ANALYSIS OF THE PRE-CONSTRUCTION PHASE	017
MANAGEMENT PRACTICES	.217
TABLE 8-11 SELECTION OF THE LOGISTIC REGRESSION MODEL OF THE HUMAN RESOURCE MANAGEME	ENT
PRACTICES AND PRODUCTIVITY	.219
TABLE 8-12 OUTPUTS OF THE LOGISTIC REGRESSION ANALYSIS OF THE HUMAN RESOURCE MANAGEM	ENT
PRACTICES	.220
TABLE 8-13 SELECTION OF THE LOGISTIC REGRESSION MODEL OF THE SAFETY AND HEALTH PRACTICE	ES
AND PRODUCTIVITY	.222
TABLE 8-14 OUTPUTS OF THE LOGISTIC REGRESSION ANALYSIS OF SAFETY AND HEALTH PRACTICES	.223
TABLE 8-15 ALTERNATIVE LINEAR REGRESSION MODELS OF PRODUCTIVITY AND CONSTRUCTION	
Management Practices	.225
TABLE 8-16 OUTPUTS OF THE LINEAR REGRESSION ANALYSIS OF THE CONSTRUCTION MANAGEMENT	
PRACTICES	.226

List of Tables

TABLE 8-17 ALTERNATIVE LINEAR REGRESSION MODELS OF CONSTRUCTION MATERIALS MANAGEME PRACTICES AND PRODUCTIVITY	NT 228
TABLE 8-18 OUTPUTS OF THE LINEAR REGRESSION ANALYSIS OF THE CONSTRUCTION MATERIALS	0
MANAGEMENT PRACTICES	228
TABLE 8-19 ALTERNATIVE LINEAR REGRESSION MODELS OF CONSTRUCTION EQUIPMENT MANAGEME	NT
PRACTICES AND PRODUCTIVITY	230
TABLE 8-20 OUTPUTS OF THE LINEAR REGRESSION ANALYSIS OF THE CONSTRUCTION EQUIPMENT	
MANAGEMENT PRACTICES	230
TABLE 8-21 OUTPUTS OF THE LINEAR REGRESSION ANALYSIS OF THE MANAGEMENT PRACTICES RELA	TED
TO CONSTRUCTION METHODS	
TABLE 8-22 OUTPUTS OF THE LINEAR REGRESSION ANALYSIS OF THE MANAGEMENT PRACTICES RELA	TED
TO CONSTRUCTION METHODS	232
TABLE 8-23 OUTPUTS OF THE LINEAR REGRESSION ANALYSIS OF THE PRE-CONSTRUCTION PHASE	
MANAGEMENT PRACTICES	234
TABLE 8-24 OUTPUTS OF THE LINEAR REGRESSION ANALYSIS OF THE PRE-CONSTRUCTION PHASE	
MANAGEMENT PRACTICES	234
TABLE 8-25 OUTPUTS OF THE LINEAR REGRESSION ANALYSIS OF THE HUMAN RESOURCE MANAGEMEN	NT
PRACTICES	
TABLE 8-26 OUTPUTS OF THE LINEAR REGRESSION ANALYSIS OF THE HUMAN RESOURCE MANAGEME	NT
PRACTICES	
TABLE 8-27 OUTPUTS OF THE LINEAR REGRESSION ANALYSIS OF THE SAFETY AND HEALTH PRACTICES	s238
TABLE 8-28 OUTPUTS OF THE LINEAR REGRESSION ANALYSIS OF THE SAFETY AND HEALTH PRACTICES	s238
TABLE 8-29 PREDICTED PROBABILITIES USING CONSTRUCTION MANAGEMENT PRACTICES AND	
PRODUCTIVITY MODEL	240
TABLE 8-30 PREDICTED PROBABILITIES USING CONSTRUCTION MATERIALS MANAGEMENT PRACTICES	3
Model	241
TABLE 8-31 PREDICTIVE ACCURACY OF THE LOGISTIC REGRESSION MODEL OF THE CONSTRUCTION	
EQUIPMENT MANAGEMENT PRACTICES AND PRODUCTIVITY	
TABLE 8-32 PREDICTED PROBABILITIES USING MANAGEMENT PRACTICES RELATED TO CONSTRUCTION	N
METHODS AND PRODUCTIVITY MODEL	244
TABLE 8-33 PREDICTIVE ACCURACY OF THE PRE-CONSTRUCTION PHASE MANAGEMENT PRACTICES AN	ND
PRODUCTIVITY MODEL	245
TABLE 8-34 PREDICTED PROBABILITIES USING HRM PRACTICES MODEL	247
TABLE 8-35 PREDICTED PROBABILITIES USING SAFETY AND HEALTH PRACTICES AND PRODUCTIVITY	
Model	248
TABLE 8-36 VALIDATION OF THE LINEAR REGRESSION MODEL OF PRODUCTIVITY AND THE CONSTRUC	TION
Management Practices	249
TABLE 8-37 VALIDATION OF THE LINEAR REGRESSION MODEL OF PRODUCTIVITY AND THE CONSTRUC	TION
MATERIALS MANAGEMENT PRACTICES	250
TABLE 8-38 VALIDATION OF THE LINEAR REGRESSION MODEL OF PRODUCTIVITY AND THE CONSTRUC	TION
EQUIPMENT MANAGEMENT PRACTICES	251
TABLE 8-39 VALIDATION OF THE LINEAR REGRESSION MODEL OF PRODUCTIVITY AND THE MANAGEM	ENT
PRACTICES RELATED TO CONSTRUCTION METHODS	251
TABLE 8-40 VALIDATION OF THE LINEAR REGRESSION MODEL OF PRODUCTIVITY AND THE PRE-	
CONSTRUCTION PHASE MANAGEMENT PRACTICES	252
TABLE 8-41 VALIDATION OF THE LINEAR REGRESSION MODEL OF PRODUCTIVITY AND THE HUMAN	
RESOURCE MANAGEMENT PRACTICES	253
TABLE 8-42 VALIDATION OF THE LINEAR REGRESSION MODEL OF PRODUCTIVITY AND THE SAFETY AN	D
HEALTH PRACTICES	253
TABLE 9-1 RELATIVE IMPORTANCE OF THE OVERALL CONSTRUCTION MANAGEMENT PRACTICES	256
TABLE 9-2 THE RELATIVE IMPORTANCE OF MATERIALS MANAGEMENT PRACTICES IN INFRASTRUCTUR	ЧE,
INDUSTRIAL AND MULTI-STOREY BUILDING CONSTRUCTION PROJECTS	257
TABLE 9-3 RELATIVE IMPORTANCE OF THE CONSTRUCTION EQUIPMENT MANAGEMENT PRACTICES IN	
INDUSTRIAL, INFRASTRUCTURE, AND MULTI-STOREY BUILDING CONSTRUCTION PROJECTS .	260
TABLE 9-4 COMPARISON OF THE MANAGEMENT PRACTICES RELATED TO CONSTRUCTION METHODS IN	[
INDUSTRIAL, INFRASTRUCTURE, AND MULTI-STOREY BUILDING CONSTRUCTION PROJECTS .	263
TABLE 9-5 COMPARISON OF THE PRECONSTRUCTION PHASE MANAGEMENT PRACTICES IN INDUSTRIAL	,
INFRASTRUCTURE, AND MULTI-STOREY BUILDING CONSTRUCTION PROJECTS	267

List of Tables

TABLE 9-6 COMPARISON OF HUMAN RESOURCE MANAGEMENT PRACTICES IN INDUSTRIAL,	
INFRASTRUCTURE, AND MULTI-STOREY BUILDING CONSTRUCTION PROJECTS	.269
TABLE 9-7 COMPARISON OF SAFETY AND HEALTH PRACTICES IN INDUSTRIAL, INFRASTRUCTURE, AND	
MULTI-STOREY BUILDING PROJECTS	.273
TABLE 9-8 ILLUSTRATION OF THE USE OF THE SCORING TOOLS OF THE CONSTRUCTION MANAGEMENT	
PRACTICES	.287
TABLE 9-9 PLANNING THE MANAGEMENT PRACTICES BASED ON A PREDETERMINED PROBABILITY VAL	LUE
	.290
TABLE 9-10 THE USE THE SCORING TOOL TO MONITOR THE CONSTRUCTION MANAGEMENT PRACTICES	5299
TABLE 10-1 THE PRIORITIZED CONSTRUCTION MANAGEMENT PRACTICES	.303
TABLE 10-2 THE SCORING TOOLS OF THE CONSTRUCTION MANAGEMENT PRACTICES	.307
TABLE 10-3 THE USER'S GUIDE SHEET	.325
TABLE 10-4 THE SCORING SHEET OF THE USER-FRIENDLY TOOL	.326
TABLE 10-5 THE PROJECT INFORMATION AND MANAGEMENT PRACTICES FOR AN EXAMPLE PROJECT	.327
TABLE 10-6 PRODUCTIVITY REPORTS FOR THE EXAMPLE PROJECT	.328
TABLE 10-7 THE MANAGEMENT PRACTICES SCORES OF THE EXAMPLE PROJECT	.328

xviii

CHAPTER ONE

1 INTRODUCTION

1.1 Introduction

This chapter presents an introduction to the thesis and consists of ten sections. The background of the research is described in Section 1.2. The research problems are stated in Section 1.3, and the research questions are presented in Section 1.4. The objectives of the research are described in section 1.5. The knowledge gap is explained in Section 1.6. In Section 1.7, the research design and methodology is highlighted. The significance of the study is summarised in section 1.8. The structure of the thesis is indicated in section 1.9. Finally, the summary of the chapter is presented in Section 1.10.

1.2 Research Background

The construction industry makes a significant contribution to the economy of a country. In Australia, the industry comprised 8.29% of the annual gross value added (GVA) in the year 2015 making it a significant contributor to the economy (Australian Bureau of Statistics, 2016a). The proportions of the contributions of different industries to Australia's annual GVA from 2010 to 2015 is indicated in Figure 1-1. Accordingly, in the years 2011, 2012 and 2013 the construction industry was the second major contributor, and in the years 2010, 2014 and 2015 the industry was the third largest contributor.

Despite its significance, the productivity growth of the industry is low. According to the Australian Productivity Commission (2015) report, in the year 2013-14, the labour productivity growth for the total economy was 1.4% whereas the labour productivity growth in the construction industry was -1.0%. The report also indicated that multifactor productivity in construction industry declined but only marginally. Thus, the labour productivity growth rate was low when compared to multi-factor productivity. Multifactor productivity is measured as output per unit of labour input in 'hours worked' (Productivity Commission, 2013). In most construction productivity studies conducted at the project level, the term productivity and labour productivity are used interchangeably. El-Gohary and Aziz (2014) explained that in construction,

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

productivity usually refers to labour productivity. In this research, the term productivity refers to the labour productivity at the construction project level.



Figure 1-1 Shares of GVA of Various Industries in Australia (Plotted using data obtained from Australian Bureau of Statistics (2016a))

Enhancing the productivity of construction projects is essential for increasing the profit margins of contractors, for reducing the possibility of the construction project delays, and for economic growth of a country. The occurrence of the project delays could lead to liquidated damage or the financial penalty due to the late delivery of the construction projects by the contractors. According to the Australian Standards' general conditions of contract AS 2124-1992 Clause 35.6, if a contractor fails to complete a project by the practical completion date, then the contractor shall be indebted to the project contract (Standards Australia, 2000). The rate could be some percentage of the project cost, for instance, 0.1% of project cost per day, or a lump sum amount such as \$AUD1000 per day. The improvement in project level productivity can also increase the industry level productivity which in turn leads to the economic growth of Australia. The study conducted by PWC (2013) indicated that increasing Australian construction industry productivity by 1% could result in the national benefit of 1.252 billion Australian Dollars (AUD).

1.3 Statement of the Problem

To date, there is no systematic and established tool for planning, monitoring and evaluating the management practices that have the potential to improve productivity in multi-storey building construction projects. During project planning phase, contractors involved in the construction of multi-storey building projects can use the tool for planning appropriate practices that should be implemented to potentially enhance productivity in their projects. During the construction phase, they can use the tool for monitoring whether the planned practices are adequately implemented or not. If not, corrective actions will be taken to reduce the loss of productivity related to the management of the projects. Moreover, during the project completion phase, the contractors can use the tool to evaluate whether the implemented practices are associated with higher or lower productivity. If the productivity is low, then lessons can be learned from the past project, and suitable practices will be planned for a new project.

In multi-storey building construction projects in Victoria State, Australia, understanding what the management practices are, planning the appropriate practices, monitoring the implementation of the planned practices, and assessing whether the implemented practices are associated with high or low productivity can help to increase productivity. Previous researchers in Australian context reported the loss of productivity due to management-related problems. Hughes and Thorpe (2014) identified rework, lack of materials and overcrowding as some of the major productivity deterrents. However, knowing the management practices that could enhance productivity, and being able to plan and control them could reduce the negative effects of the loss of productivity such as project delay, liquidated damage, and reduction in profit margins of contractors. For example, the practice 'preparation of dynamic site layout' can reduce overcrowding and contractors can prepare site layout plans that are suitable for different phases of construction such as excavation, structure, and site works. Similarly, rework due to changes in design can be reduced by adopting and controlling the implementation of the essential practices such as 'buildability review.' In Australian construction projects, Love (2001) estimated that the mean direct and indirect costs associated with rework are 6.4% and 5.6% of contract value respectively. The loss of productivity associated with lack of materials can also be minimised by implementing practices such as 'long-lead

materials identification.' If the long-lead materials are not identified during preconstruction phase, the productivity of a project can be reduced due to shortage of materials. In the context of Australia, as most construction materials are imported from overseas, missing the implementation of the practice 'long-lead materials identification' can lead to project delay and additional costs associated with the delay. The procurement and delivery of the critical materials could take longer time and projects can be delayed. Thus, identifying and prioritising the management practices, and preparing a method to measure, plan, monitor and evaluate the productivity of the projects.

Additionally, during project planning phase, means of assessing the probability of exceeding baseline productivity based on a measured value of the construction management practices is not developed. The implementation of adequate or higher levels of the practices can help to increase the probability of achieving higher levels of productivity whereas the implementation of inadequate or lower level of the practices decreases the probability of attaining higher levels of productivity. Thus, developing a tool which can be used to estimate this probability is essential. Knowing the probability of achieving higher productivity levels can help construction project managers to plan for the implementation of appropriate practices on a proposed multi-storey building project. For instance, if the chance of having a productivity that exceeds the baseline productivity is low, then the project managers can make the decision to increase the proposed level of implementation of the practices to increase the chances of having an associated higher level of productivity.

1.4 Research Questions

To address the problems described in the previous section, this research is designed to answer the following questions.

- RQ1: What are the management practices that have the potential to improve productivity in multi-storey building construction projects in the context of Victoria State, Australia?
- RQ2: Which management practices are the most important to enhance productivity in multi-storey building construction projects?

- RQ3: How can the management practices that have the potential to improve productivity in multi-storey building construction projects be measured, planned, and controlled?
- RQ4: What tool can be used for predicting the probability of exceeding baseline productivity factor based on a score of the management practices?

1.5 Research Objectives

The main objectives of this research are:

- To identify and prioritise the management practices that have the potential to enhance productivity in multi-storey building construction projects in the context of Victoria State, Australia.
- To refine and validate scoring tools for measuring, planning, monitoring and evaluating the management practices that could improve productivity in multistorey building construction projects.
- To build and validate a tool that can be used to predict the probability of exceeding baseline productivity factor based on a score of the management practices.

1.6 Rationale for the Research

1.6.1 Lack of Techniques of Integrating Productivity and the Level of Implementation of the Management Practices

Although previous studies have developed tools for measuring construction management practices for industrial and infrastructure projects (CII, 2013a, CII, 2013b), the existing tools do not integrate probability function when assessing productivity levels based on the known level of planning or implementation of the practices. Thus, users of the existing scoring tool have no means of assessing or forecasting the probability of exceeding a baseline productivity factor based on the implementation of the construction management practices. For instance, the baseline score for the management practices that have the potential to enhance productivity in industrial projects in North America is found to be 65% or 1300 points out of maximum 2000 points (Caldas et al., 2014). However, the users of this tool can only assess whether their scores are lower or higher than the baseline score. They cannot assess whether their scores would lead to high or low productivity as compared to baseline

5

productivity. The method which can be used to compute the probability of exceeding baseline productivity factor is not developed by previous researchers.

1.6.2 Lack of Method of Measuring, Planning, Monitoring and Evaluating the Management Practices in Building Construction Projects

Previous researchers have developed two different techniques to measure management practices in the context of industrial and infrastructure construction projects (CII, 2013a, CII, 2013b, Nasir, 2013). However, no tool has been developed to measure, plan, monitor and evaluate the management practices for building projects. The existing measurement tools cannot be applied to building projects in Australia because the practices for enhancing productivity in multi-storey building projects could be different and the relative weight assigned to each practice would be different as well. The findings of the previous studies confirmed that the management practices scoring tools vary based on the type of projects, and they cannot be used interchangeably (Nasir et al., 2015, Caldas et al., 2014, CII, 2013a, CII, 2013b).

1.6.3 Dearth of Research Identifying Management Practices Having Potential to Enhance Productivity in Building Projects

Previous studies identified the best practices that could improve productivity in infrastructure and industrial construction projects in North America (Caldas et al., 2014, Nasir et al., 2015). However, it is possible that none of these practices is appropriate for improving productivity in multi-storey building construction projects. Also, management practices and their effectiveness would depend on the context such as differences in the resources supply chain, local regulatory requirements and project type (Bloom and Van Reenen, 2010). Thus, management practices for improving productivity would vary across project types and from country to country. It may also vary across different Australian construction markets. For instance, the use of control system for tool delay is found to be one of the management practices that could enhance productivity in industrial projects, but the practice is not applicable to infrastructure projects (Caldas et al., 2014). Similarly, construction equipment utility requirement has been identified as the practice that could improve productivity in infrastructure projects but not for industrial projects (Nasir et al., 2015). Furthermore, while traffic control plan, project completion plan, and communications, coordination and agreements are identified as the best practices for enhancing productivity in infrastructure construction 6

projects, they are not included in the list of best practices for industrial construction projects in North America (Caldas et al., 2014, Nasir et al., 2015). Moreover, the priority given to the management practices that are suitable for both project types is different.

Thus, projects are not homogeneous when considering management practices that have the potential to enhance their productivity. This study departs and extends existing research in the area by focusing on multi-storey building projects in Victoria State, which is the second largest construction market in Australia (Australian Bureau of Statistics, 2016b). There is little, or no research has been conducted on what these practices might be and which of them are the most important in the context of multistorey building projects in Australia. Therefore, an investigation of management practices which are suitable for improving productivity in multi-storey building construction projects is essential.

1.7 Research Design and Methodology

This study adopted exploratory sequential mixed-methods research design involving a combination of qualitative and quantitative data in two phases. The mixed-methods research design can be classified into three categories: convergent parallel, explanatory sequential and exploratory sequential (Creswell, 2013). In convergent parallel mixed methods, the researcher collects both qualitative and quantitative data, analyses them separately, and then compares the results to see if the findings confirm or disconfirm each other. Explanatory sequential mixed-methods involves a two-phase project in which the researcher collects quantitative data in the first phase, analyses the results, and then uses the results to plan or build on to the second qualitative phase. In exploratory sequential mixed-methods, the researcher first begins by exploring with qualitative data and then uses the findings in a second quantitative phase. The qualitative phase may be used to build an instrument that best fits the sample under study, to identify appropriate instruments to use in the follow-up quantitative phase or to specify variables that need to go into a follow-up quantitative study (Creswell, 2013). In this research, the exploratory sequential mixed-methods is used because construction management practices identified by previous studies might not be applicable to the Victorian construction industry and there could be other practices that are specific to the local industry. Thus, exploratory study is conducted first by collecting and analysing

qualitative data obtained from interviews during the first phase. On the basis of the qualitative data, appropriate management practices are identified, and their levels of implementations are refined to suit the multi-storey building construction projects. To investigate the relationship between productivity and the practices, assign the weights to the management practices, validate the management practices scoring tools in the context of the multi-storey building projects, and develop as well as validate the productivity predicting tools, the quantitative data is collected and analysed during Phase II of the research.

To address part of objective 1 which is "to identify the management practices that have the potential to enhance productivity in multi-storey building construction projects in the context of Victoria State, Australia," data relating to the management practices were collected in Phase 1 using interviews. Nineteen professionals who have experience in delivering multi-storey building projects in Victoria were interviewed. They have a five to forty years work experience and have been working as general manager, construction manager, project manager, project coordinator, project engineer, site engineer, contract administrator, supervisor and cost manager. Finally, the qualitative data analysis reached a saturation point, and management practices that have the potential to improve productivity in multi-storey construction building projects were identified. During Phase I, part of objective 2 or "to refine the scoring tools" was also achieved. The implementation levels of the identified practices were refined, and unweighted scoring tools were prepared. The outputs of the first phase were used as an input to the second phase.

The quantitative data were collected and analysed to address part of objective 1 or " to prioritize the practices," part of objective 2 or " to validate the scoring tools" and objective 3 of the study which is "to build and validate a tool that can be used to predict the probability of exceeding baseline productivity factor based on a score of the management practices." The quantitative data were collected using interview questionnaire survey whereby the survey was self-administered. Respondents were asked the questions face-to-face and requested to write their answers (Fowler Jr and Cosenza, 2009). The approach can increase the response rate, and in this study, it allowed the researcher to clarify issues during the interview survey. The data were collected from principal contractors' experts and multi-storey building projects which

were completed between 2011 and 2016 in the state of Victoria, Australia. The first part of the survey data was analysed using relative importance index technique, the construction management practices were prioritised, the weight proportions for the levels of implementation of the practices were computed, and weighted scoring tools were prepared. The second part of the survey data was analysed, and the scoring tools were validated. Correlation analysis was conducted, and the relationship between productivity and the practices was investigated. Linear and logistic regression analyses were carried out, and the tools which can be used for predicting productivity factors and the probability of exceeding baseline productivity factors based on the score of the management practices were developed and validated.

1.8 Significance of the Research

This study contributes to the neoclassical economic growth theory by investigating the under-researched aspect of the theory. As shown in Figure 1-2, technological advancement, capital deepening and management practices are the drivers of labour productivity. The first two factors are explained in-depth by the theory. The growth accounting framework, which is based on the neoclassical economic theory, proposes that management practices could improve labour productivity. However, the theory does not explain the management practices in detail. For instance, which practices, in which context, could enhance labour productivity is not studied. Thus, this research helps to increase the understanding of the types of practices that can be implemented to increase labour productivity in the context of multi-storey building construction projects.



Figure 1-2 The Contribution of the Research to a Theory

This study also contributes to the body of knowledge in construction management by identifying and prioritising the management practices that are suitable to enhance

9

productivity in multi-storey building projects. The findings of the research can be used as international and sectoral comparisons and provide useful information for future researchers and companies seeking construction works in Australia. Contractors involved in the construction of multi-storey buildings in Victoria State, Australia can implement the identified practices to enhance productivity in their projects. As a result of the improvement in the projects' productivity, the productivity of Australian construction industry will increase which in turn can lead to the growth in the economy of the country. During the project planning phase, the management practices scoring tools can be used to plan appropriate practices which can be implemented to potentially increase productivity. Moreover, the productivity predicting tools can be used to estimate the productivity of a certain multi-storey building project based on the levels of planning of the management practices. By using the estimated productivity, the likelihood of occurrence of project delays can be predicted, and corrective actions can be taken prior to commencing the construction of the project.

The finding of the study is applicable to multi-storey building construction projects in Victoria State, Australia. It can also be adopted by contractors involved in building construction projects in other countries. However, validation is required as management practices might vary from country to country.

1.9 Thesis Structure

This thesis comprises of ten chapters. In Chapter 1, the background of the research, statement of the problem, research questions, objectives of the research, rationale for the research, research design and methodology, significance of the study and thesis structure are presented. In the second chapter, a systematic review of literature is conducted to identify the knowledge gap, the main features of the Victorian construction industry (the context of the study) and factors influencing the productivity of construction projects in Australia as well as in the international contexts are reviewed. Chapter 3 provides the conceptual frameworks of the study. In this chapter, the neoclassical economic growth theory (growth accounting framework) is reviewed, and the preliminary conceptual frameworks of productivity and the construction management practices are developed. The fourth chapter presents the research design, the research approach, and methods of data collection and analysis during both phases of the study. In Chapter 5, the analysis of the qualitative data collected during the first

10

phase of the research answers RQ1. The context-specific management practices that have the potential to improve productivity in multi-storey building construction projects are identified, the implementation levels of the identified practices are refined to suit Victorian construction industry, and on that basis, unweighted scoring tools are prepared.

In the sixth chapter, the preliminary quantitative data has been analysed. The background of respondents is described, productivity factors of the projects are calculated, and the descriptive data analysis of the management practices is carried out. In Chapter 7, the analysis of the data collected during Phase II of the study answers RQ2 and RQ3. The quantitative data is analysed, the weights of the management practices are computed, the weighted scoring tools are prepared and validated, and the relationship between productivity and the management practices is investigated. Chapter 8 provides an answer to RQ4. The linear and logistic regression analyses are conducted, and the tools which can be used for predicting productivity based on a score of the management practices are built and validated. Chapter 9 presents the discussions of the findings of the study. Chapter 10 concludes the thesis by summarising the key findings of the study. The theoretical and practical implications of the research are also presented in Chapter 10. Following the list of references, supplementary information such as questionnaire and raw data used in the quantitative and qualitative data analysis are presented in the Appendix part of the thesis.

1.10 Summary

The purpose of this study is to identify and prioritise the management practices that have the potential to increase productivity in multi-storey building projects in Victoria State, Australia. Furthermore, the study is designed to refine tools which can be used for measuring the practices and predicting the productivity of a multi-storey building project based on a score of the practices. Identification and measurement of the practices are essential for the improvement of productivity. Contractors involved in multi-storey building construction projects can use the scoring tools to assess the adequacy of their practices and take corrective actions to potentially enhance productivity. The scoring tools can be used during planning, construction and completion phases of a project to plan, monitor, and evaluate the management practices respectively. The productivity and probability prediction tools can also be used during

the project planning phase to help contractors' project managers to plan appropriate practices. This chapter presented the research background, statement of the problem, research questions, research objective, rationale for the study, a summary of research design and methodology, the significance of the research, and the thesis structure. The review of the relevant literature is presented in the next chapter.

CHAPTER TWO

2 LITERATURE REVIEW

2.1 Introduction

This chapter presents the comprehensive review of the literature to support the argument described in Chapter 1. It specifically describes the fields of knowledge that are related to the rationale part of this research. The chapter starts by describing the context of the study. The features of the construction industry of the Victoria State, Australia are explained and a multi-storey building project is defined in Section 2.2. A systematic review of the literature has been conducted in Section 2.3, and the knowledge gap is identified. Following this section, the existing literature is categorised into three sections (Section 2.4, 2.5 and 2.6) and reviewed accordingly. Section 2.4 explains an overview of Australian construction industry productivity studies; Section 2.5 describes the factors negatively influencing construction projects' productivity and Section 2.6 addresses management practices that have the potential to enhance productivity in industrial and infrastructure projects. Lastly, the summary of the chapter is presented in Section 2.7.

2.2 Characteristics of the Victorian Construction Industry

Construction projects' productivity is influenced by factors such as government regulations, the project delivery models adopted in the local construction industry, resource supply chain, the availability and strength of unions, and other issues which are peculiar to a certain location. Consequently, the management practices of the construction projects vary from country to country. This section describes the context of the study.

Definition of a Multi-Storey Building Project and Classification of Construction Projects

In Australia, construction projects can be broadly classified into two: engineering and building. This research focuses on productivity improvement in building construction projects because a significant level of activity occurs in this sector. In June 2016, the values of works done in the Australian building and engineering sectors were 54% and 46% respectively (Australian Bureau of Statistics, 2016b). Building construction projects can be classified based on the number of storeys. According to Australian

13

Building Regulations (2006), a multi-storey building has a rise in storeys of more than three. This research investigates productivity improvement in multi-storey building construction projects as their construction processes are more complex due to the involvement of complex engineering services. The South Australian Department for Communities and Social Inclusion (2016) apartment design guideline states that multi-storey buildings have more complex design and construction systems. For the sake of brevity, the term 'building projects' is used to refer to multi-storey building construction projects.

The Market Structure

The construction sector in Victoria State, Australia is dominated by a few large contractors that engage numerous small companies. As indicated in Table 2-1, at the end of 2014/15 financial year, the proportion of companies involved in building construction were 0.06%, 0.78% and 99.15% for firms employing over 200 workers, 20-199 workers, and 0-19 workers respectively (Australian Bureau of Statistics, 2016c).

		Number of Companies Operating at end of 2014/15 financial year			
State	Sector	Non- employing	1-19 Employees	20-199 Employees	200+ Employees
Victoria	House Construction	7455	4806	63	6
	Other Residential Building Construction	4393	1072	21	3
	Non-Residential Building Construction	2508	1164	85	5
	Total	14356	7042	169	14
	Percentage	66.52%	32.63%	0.78%	0.06%

Table 2-1 Number of Construction Firms Operating in Victoria in 2014/2015 (adapted from Australian Bureau of Statistics (2016c))

The Regulatory Bodies

The Australian Fair Work Commission stipulates the minimum wages, ordinary working hours, overtime rates, and other issues related to employment conditions in building construction projects (Fair Work Commission, 2016). The 2015/16 minimum wage is 18.17 Australian Dollars (AUD), and additional allowances are provided based on the experience of the tradesperson. The 2015/16 ordinary working hours is 38 hours per week, and it is between 7:00 am and 6:00 pm Monday to Friday. According to the Commission, all the time worked beyond the ordinary working hours must be paid at the rate of one and half time for the first two hours and double time thereafter and a minimum of three hours is considered for the payment purpose. Similarly, the Commission specifies that all the time worked on Sundays should be paid at double

rate. Moreover, the Commission sets the minimum wages based on the skill level of the employees and prepares the "Building and Construction General On-site Award 2010" in which the minimum wages which correspond to the skill levels of the general construction workers (CW) and engineering construction workers (ECW) are specified (Table 2-2). As indicated in the Table, the minimum 2015/16 hourly wage for CW level 1a is AUD 18.61 and the minimum hourly wage for CW level 8 is AUD 23.69.

Skill Level	Minimum weekly wage	Minimum hourly wage	
Level 8 (CW/ECW 8)	900.40	23.69	
Level 7 (CW/ECW 7)	879.20	23.14	
Level 6 (CW/ECW 6)	854.70	22.49	
Level 5 (CW/ECW 5)	832.50	21.91	
Level 4 (CW/ECW 4)	807.80	21.26	
Level 3 (CW/ECW 3)	783.30	20.61	
CW/ECW 1 (level d)	745.50	19.62	
CW/ECW 1 (level c)	731.80	19.26	
CW/ECW 1 (level b)	721.60	18.99	
CW/ECW1 (level a)	707.00	18.61	

Table 2-2 The Minimum Wages of Construction Workers in 2015/16 (adapted from Fair Work

The skill required for each level is also specified in the Building and Construction General On-site Award 2010. Accordingly, the skill level CW1 (level a) is assigned to workers commencing activities in the construction industry. Their levels become CW1 (level b) and CW1 (level c) after working 3 and 12 months in the industry respectively. However, for CW1 (Level d), the workers are required to have certificates, or they should have construction skills assessment results which are equivalent to the required competency standards in accordance with the Recognition of Prior Learning (RPL) principles or other requirements stipulated in the building and construction award (Fair Work Commission, 2016). According to the Fair Work Commission, Recognition of Prior Learning (RPL) is 'the formal recognition of skills attained through on-the-job experience and/or training and may include formal qualifications such as overseas qualifications, which have up until now been unrecognised.'

The Environment Protection Authority of Victoria (EPA) has prepared guidelines that help contractors to comply with Section 48A (unreasonable noise from residential premises) of the Environment Protection Act 1970. The authority stipulates that the normal working hours for commercial building construction sites and large-scale developments (at least four storeys) in non-residential zone is from 7:00 am to 6:00 pm

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

on weekdays and from 7:00 am to 1:00 pm on Saturday; and for large residential developments in residential areas the normal working hours is from 7:00 am to 6:00 pm during weekdays and from 9:00 am to 1:00 pm on Saturday (EPA, 2016). According to the Authority, failure to control the noise during construction period is an offense against the Environment Protection Act 1970 and a person who contravenes Section 48A of the Act is liable to a penalty of not more than 120 penalty units, and in the case of continuing offence to a daily penalty of not more than 30 penalty units for each day during which the offence continues after conviction (Environment Protection Act, 2010).

In Victoria State, depending on the type of construction activities, the Construction Management Plan (CMP) should be prepared by the contractors and approved by the pertinent city councils prior to the commencement of a project. For instance, according to the guidelines provided by the city of Melbourne, the Construction Management Plan (CMP) is prepared to ensure that the construction sites do not adversely affect the safety, health, amenity, traffic and the surrounding environment (City of Melbourne, 2005). The city council requires the contractors to address the six elements of the CMP: (1) Public Safety, Amenity and Site Security; (2) Operating Hours, Noise and Vibration Controls; (3) Air and Dust Management; (4) Stormwater and Sediment Control; (5) Waste and Materials Reuse; and (6) Traffic Management. According to the city council, the approved CMP is a contract between the contractor and the city, and if there is noncompliance, the city can withdraw Security Deposit which is provided in relation to a CMP and rectify the remedy (City of Melbourne, 2005). The city also stipulates that the following permits should be sought for specific activities: permit to erect a gantry, overhead protective awning over the road or footpath; permit for a vehicle crossing (permanent or temporary); permit to occupy space on road or footpath; permit to erect a hoarding; permit for a road opening; permit for a road closure; permit for a construction zone; permit to use a mobile crane; permit for rubbish skips and builder's bins; permit for works and temporary structures in City of Melbourne parks and gardens; permit to work outside prescribed hours; permit for excavation and/or protection works; and approval for a legal point of discharge or a temporary point of discharge (City of Melbourne, 2005).

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

In Victoria, occupancy permits are required prior to occupation of a new building. According to Victorian Building Authority (VBA), an occupancy permit is a document that signifies that a building surveyor is satisfied that the completed building construction work is appropriate for occupation. Building construction works for new units or apartments always require an occupancy permit, and it is an offence to use a new home that does not have occupancy permit (VBA, 2016).

The Union

The Victorian construction industry is characterised by the presence of Construction, Forestry, Mining, and Energy Union (CFMEU). The union prepares a calendar for construction sites that typically comprises of 36 working hours per week, rostered days off (RDO) and other events in a financial year (CFMEU, 2016a). Moreover, in most construction projects, the union signs a contractual agreement known as Enterprise Bargaining Agreement (EBA) in which the minimum payments and other working conditions are agreed between a contractor and the union. Under certain circumstances, the union has also the authority to suspend the construction works. The health and safety representatives can issue the Provisional Improvement Notice (PIN) to the contractor if there are unsafe working conditions, or they can suspend the construction work if there is an immediate threat to the workers (CFMEU, 2016b). According to Worksafe Victoria (2008), the Provisional Improvement Notice is a written direction requiring a person (contractor) to remedy the breach (or likely breach) of Workplace Health and Safety (WHS) Act or Regulation. The health and safety representatives (HSRs) can issue a PIN if they believe, on reasonable grounds, that a provision of the Occupational Health and Safety Act 2004 (OHS Act) or the Occupational Health and Safety Regulations 2007 is being contravened; or has been contravened, and it is likely that the contravention will continue or be repeated (Worksafe Victoria, 2008a).

The Project Delivery Models

In Australia, the project delivery models which are predominantly used to deliver nonresidential building and civil (road and bridge) projects can be broadly grouped into: (1) Traditional (Construct Only), (2) Design and Construct including common variants such as Design, Construct and Maintain (DCM); Design, Construct and Operate (DCO); and Design, Construct, Maintain and Operate (DCMO), (3) Managed (Managing Contractor, Construction Management), (4) Direct Managed, (5) Relationships (Alliance), and (6)
Public Private Partnership (Austroads, 2014). According to Austroads (2014), the Construction Management delivery model is used for the construction of buildings, and in this model, the project owner engages the designer and trade contractors directly. A construction manager that usually receives a time-based fee or percentage of the cost of the works is employed to act as client's agent and manages the delivery of the construction works. The Managing Contractor delivery model is used for the construction of large complex buildings, and it involves a principal contractor being engaged as a managing contractor that manages the development of the design, coordinates production of construction documentation, enters into contracts and manages the delivery of the works on behalf of the project owner (Austroads, 2014). The managing contractor delivery model typically includes a two-stage contractual arrangement. In Stage 1, based on a preliminary project brief which is prepared by the project owner, the design is developed collaboratively, and the contractor prepares a guaranteed construction sum (GCS) or other variants such as lump sum. If the contractor's offer in Stage 1 is accepted, then Stage 2 involves documentation, construction, and commissioning of the project. The work packages to be subcontracted are competitively tendered by the managing contractor in an open book type of arrangement, and only the actual cost of construction is paid up to the agreed GCS cap (Austroads, 2014).

The Construction Materials Supply

Most building materials in Australia are imported from overseas. Wheeldon (2012) reported that the import of building materials is increasing due to the rising investment costs in the manufacturing sector in Australia. The Housing Industry Association (HIA) recommended the establishment of Australian building product accreditation scheme as the usage of the building materials which are manufactured offshore is increasing in Australia and some of these materials did not comply with Australian standards (HIA, 2012). The Prefabricated construction system is increasing at about 5% per year (PrefabAUS, 2014). Hampson and Brandon (2004) indicated the increment of off-site manufacturing practices in Australian construction industry.

The High-risk Construction Tasks

The Victorian Occupational Health and Safety Regulation 2007 classified the construction industry under the hazardous industries and provided the list of high-risk

construction works. Accordingly, some of these works include any construction activity where there is a risk of a person falling more than 2 metres; demolition works; trench excavation if the excavated depth is more than 1.5 metres; and an activity which involves any movement of the powered mobile plant. The regulation stipulates that any person must not perform high-risk construction work unless a Safe Work Method Statement (SWMS) is prepared for the particular work. The work should also be carried out in accordance with the statement. A Safe Work Method Statement is a document that specifies the high-risk construction activities, the hazards arising from these activities, and the measures to be taken to control the risks (Safe Work Australia, 2014b).

Safe Work Australia stipulates the hierarchy of risk controlling techniques from highest to the lowest level (Safe Work Australia, 2011). Accordingly, Level 1 or eliminating the hazard, for instance, includes avoiding the risk of fall from a height by working on the ground. It is the most effective control measure. Level 2 control measures are used when it is not suitable to use Level 1 techniques. They comprise of: (i) substituting the hazard with something safer, for example, replacing solvent-based paints with water-based solvents (ii) isolating the hazard from people, for instance, install guard rails around exposed edges and holes in floors, and (iii) using engineering controls which involves some mechanical devices, for example, the use of trolleys or hoists to move heavy loads. Level 3 control measures, which are the least effective in minimising risks, rely on human behaviour and supervision. These actions include (i) the use of administrative controls such as developing the procedures on how to operate machinery safely and using signs to warn people, and (ii) the use personal protective equipment (PPE) such as hard hats and gloves (Safe Work Australia, 2011). The next section presents the systematic review of the literature.

2.3 Systematic Review of Literature

This research used Scopus as a search engine to identify the journals which frequently publish construction productivity related articles. Yi and Chan (2014) conducted the critical review of labour productivity research in construction journals from 1983 to 2011 and found that Journal of Construction Engineering and Management (JCEM), Construction Management and Economics (CME), Engineering, Construction and Architectural Management (ECAM), Journal of Management in Engineering (JME),

International Journal of Project Management (IJPM), Automation in Construction (AIC), Building Research and Information (BRI), Building and Environment (BAE), Canadian Journal of Civil Engineering (CJCE), and Journal of Computing in Civil Engineering (JCCE) frequently publish articles which are related to construction labour productivity. The authors used the search engine, Scopus, to identify the journals and articles.

Naoum (2016) conducted the literature review of construction labour productivity studies from 1970 to 2014 and identified journals which publish articles related to construction labour productivity. Accordingly, Journal of Engineering and Management (JCEM), Journal of Management in Engineering (JME), Construction Management and Economics (CME), Engineering, Construction and Architectural Management (ECAM), International Journal of Project Management (IJPM), International Journal of Project Management (IJPPM), Journal of Computing in Civil Engineering (JCCE), Construction Innovation (CI) and Journal of Built Environment (BE) are found to be the journals which frequently publish construction labour productivity related articles. The findings of Naoum (2016) and Yi and Chan (2014) regarding the identification of the journals that published the most construction productivity-related articles are similar.

This research used the following 12 journals as starting point to search the articles related to construction labour productivity which were published between 1983 and 2016. The journals include (1) Journal of Construction Engineering and Management (JCEM), (2) Construction Management and Economics (CME), (3) Engineering, Construction and Architectural Management (ECAM), (4) Journal of Management in Engineering (JME), (5) International Journal of Project Management (IJPM), (6) Automation in Construction (AIC), (7) Building Research and Information (BRI), (8) Building and Environment (BAE), (9) Canadian Journal of Civil Engineering (CJCE), (10) Journal of Computing in Civil Engineering (JCCE), (11) Journal of Productivity and Performance Management (IJPPM) and (12) Journal of Construction Innovation (JCI). Accordingly, 511 articles that have the keywords "productivity" and "construction" in their titles, abstracts and keywords were identified during Stage I of the review. The summary of the results of the literature review during Stage I is

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

presented in Table 2-3, and the list of articles including the names of the authors as well as the years of publications are included in Appendix-8.

No.	Journals	No of articles
1	Journal of Construction Engineering and Management (JCEM)	192
2	Construction Management and Economics (CME)	83
3	Engineering, Construction and Architectural Management (ECAM)	31
4	Journal of Management in Engineering (JME)	28
5	International Journal of Project Management (IJPM)	24
6	Automation in Construction (AIC)	58
7	Building Research and Information (BRI)	16
8	Building and Environment (BAE)	14
9	Canadian Journal of Civil Engineering (CJCE)	37
10	Journal of Computing in Civil Engineering (JCCE)	20
11	Journal of Productivity and Performance Management (IJPPM)	5
12	Construction Innovation (CI)	3
	Total	511

Table 2-3 Summary of the Number of Journal Articles

After reviewing the references sections of the 511 articles, more journals and journal articles, which are relevant to this study, were identified (Figure 2-1). Accordingly, Australasian Journal of Construction Economics and Building (AJCEB), Journal of Infrastructure Systems (JIS), Structural Survey (SS) and Journal of Civil Engineering and Management (JCiEM) are included in the list of journals during Stage II of the searching process. Finally, 558 articles which consist of the keywords 'productivity' and 'construction' were identified.

The procedure followed in the identification of the relevant papers is shown in Figure 2-1. The contents of the 558 articles were reviewed, and the papers which investigate factors influencing productivity in construction projects were included in the final list of articles. Papers which do not address the factors which either negatively or positively affect productivity in construction projects were excluded from the final list (Figure 2-1). Finally, a total of 102 articles were included in the final list, and the detailed review was conducted to identify the knowledge gap.





Figure 2-1 Flow Chart for the Systematic Review of Literature

The final articles were then analysed based on project type, country and the focus of the paper, and the results are presented in Table 2-4. As summarised in the table, in both Australian and international contexts, none of the researchers has identified the management practices that have the potential to enhance productivity in building construction projects and developed a tool to measure the practices. Furthermore, no research has been conducted on how to develop a tool which can be used to predict the probability of exceeding baseline productivity based on a score of the management practices that could enhance productivity in industrial and infrastructure projects and developed management practices scoring tools for these project types (Caldas et al., 2014, Nasir et al., 2015).

I. Research Identifying Factors Influencing Construction Projects' Productivity							
Country	Australia		International				
No of Papers 5		5	95				
II. Research Identifying Management Practices That Have the Potential to Enhance Productivity in Construction Projects and Developing Tools to Measure the Practices							
Country	Australia International						
Project Type	Building	Infrastructure	Industrial	Building	Infrastructure	Industrial	
No of Papers	-	-	-	-	1	1	
III. Research Developing Tool for Predicting Probability of Exceeding Baseline Productivity Based on a							
Score of Management Practices							
Country	Australia International						
Project Type	Building	Infrastructure	Industrial	Building	Infrastructure	Industrial	
No of Papers	-	-	-	-	-	_	

Table 2-4 Summary of the Systematic Review of Literature

The finding of the literature review shows that most previous studies focused on the investigation of factors which deter construction projects' productivity, and these factors are described in Section 2.5. Studies conducted in the Australian context and addressing construction productivity issues are summarised in Section 2.4.

Figure 2-2 indicates the number of articles focusing on the factors influencing productivity in construction projects based on countries or contexts. Figure 2-3 shows the numbers of research papers according to their authors. In the figure, the authors who published more than three articles are presented.



Figure 2-2 Classification of Research Articles Based on Countries



Figure 2-3 Classification of Research Articles Based on Authors

2.4 Overview of Australian Construction Industry Productivity Studies

Most previous studies, in the context of Australia, conducted at construction projects level focus on identifying the factors which hamper construction labour productivity (Chalker and Loosemore, 2016, Loosemore, 2014, Hughes and Thorpe, 2014). However, the previous researchers do not identify and prioritise the construction management practices that have the potential to enhance productivity in multi-storey building construction projects. Doloi (2008) identified three factors which positively influence productivity in building projects. These include project planning, incentives, and job satisfaction. However, materials management, equipment management, safety and health practices and other practices which could influence productivity were not investigated. Moreover, no research has been conducted on how to develop a technique to measure, plan, monitor, and evaluate the practices in the context of building projects in Australia.

Previous studies indicated that management-related problems are negatively influencing construction projects' productivity in Australia. The Business Council of Australia (BCA) found that the main contributing factors for low construction productivity in Australia include the inefficiency in scheduling projects with multiple tasks, unexpected leaving of skilled staff, lack of critical machinery on site when it is needed, and lack of flexibility in the workplace (BCA, 2012). Loosemore (2014) investigated the determinants of Australian construction productivity and found that low quality of relationships with principal contractors; lack of opportunity for early involvement in design; poor or not transparent tender practices; poor administration and document

control, and design management; lack of project management and supervisory skills, particularly in planning, scheduling and coordination; and industrial relations (IR). Hughes and Thorpe (2014) found that rework, incompetent supervisors, and incomplete drawing are the three most critical problems that reduce construction projects' productivity in the Australian context.

Construction productivity studies conducted at the construction industry level in Australia can be summarized under: (1) the significance of improving the productivity of Australian construction industry to the national economy (Walker, 1995, Industry Commission, 1997, PWC, 2013), (2) comparison of Australian construction industry productivity with other developed countries' construction industry productivity (Toner et al., 2001, Loosemore and McGeorge, 2002), and (3) estimation of the total factor productivity (Li and Liu, 2010), and (4) drivers of Australian construction industry productivity (Chancellor, 2015). However, these studies did not investigate the management practices influencing productivity at construction project level.

Previous studies have addressed the importance of increasing construction industry productivity to Australia's economy. Walker (1995) described that improving construction industry productivity is essential for promoting national competitiveness, for enhancing living standards and for achieving a satisfactory rate of growth. The Industry Commission (1997) report also indicated that productivity enhancement is the key ingredient in promoting sustainable economic growth and in improving the living standards of Australians. According to the report, the influence of productivity growth on living standards can come over long periods of time, and small improvements in productivity growth will make a substantial difference for future generations. The report showed that increase in productivity of about half percent could raise average real incomes by an additional 30 percent over the subsequent 40 to 50 years. Moreover, the Commission estimated that productivity growth has accounted for about two-thirds of the improvement in the average real incomes of Australians over the past three decades prior to 1996. Accordingly, the real Gross Domestic Product (GDP) increased from 145 billion Australian Dollars (AUD) in 1964/65 to AUD 430 billion in 1995/96; and the productivity growth had accounted for about AUD 140 billion from the total of AUD 285 billion increase (Industry Commission, 1997). Therefore, improvement in the

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

productivity of multi-storey building construction projects is significant for the economic growth of the country.

A study conducted by PWC (2013) indicated that increasing productivity is fundamental to the Australia's economy. According to the research, over long periods of time, small differences in rates of productivity growth can compound like interest in a bank account and make an enormous difference to a society's prosperity. Moreover, the study showed that higher construction industry productivity implies lower construction costs, higher savings in production costs, and consequently an increase Australia's GDP. The study also concluded that the performance of Australian construction industry in terms of productivity is low, and the research estimated that 1% increase in the productivity of Australian construction industry would result in the national benefit of about AUD1.252 billion. Therefore, increasing the productivity of multi-storey building construction projects can also lower the construction costs, increase profit margins of the contractors, and contributes to the national economy. Moreover, the owners of the projects can be benefited if the productivity of the projects is improved as the contractors' offer could be lower.

Previous researchers made international comparisons of Australian construction industry productivity. However, there are opposite views about the status of the productivity performance of Australian construction industry when compared to other developed countries. Some of the previous researchers argue that Australia's construction industry productivity is high while others claim that it is low. A study conducted by Loosemore and McGeorge (2002) indicated that, in value added per hour, Australia's construction industry productivity is on a par with Japan's and Germany's construction industry productivity and performing slightly better than France and the UK, but lagging behind the US, Canada, and Singapore. Similarly, a report by Young et al. (2008) indicated that Australian construction industry productivity is less than USA's construction industry productivity. According to the report, the proportion of the Australia's construction productivity was 95.9%, 77% and 51.9% of the USA's construction industry productivity in the year 1990, 2000 and 2005 respectively. Likewise, the study conducted by the Business Council of Australia (BCA) indicated that, in Australia, it takes 1.3 hours to execute work that would take 1.0 hour in the USA (BCA, 2012). The study concluded that Australian construction projects have lower

productivity, and they are 40% more costly than similar projects in the USA. However, Best (2012) argued that construction projects in Australia might not be costly if purchasing power parities are utilised instead of the exchange rates when comparing construction projects in the USA and Australia. Although there is no consensus among the previous researchers regarding the status of Australia's productivity, it is desirable to investigate the mechanisms of increasing the productivity of multi-storey building projects as the increment contributes to the economic growth of a country.

The drivers of Australian construction productivity at industry level were identified by previous researchers. Chancellor (2015) found that wage growth, research and development expenditure are the drivers of productivity at the national level. The study also found that increasing the number of apprentices and expenditure on research and development are the main drivers of construction industry productivity in Victoria State, Australia. However, the study did not investigate the productivity drivers at the construction project level. In next section, the review of the factors which deter productivity of construction projects is presented.

2.5 Factors Negatively Influencing Construction Projects' Productivity

Based on the review of previous studies, factors which negatively influence construction projects' productivity can be broadly categorised into construction materials, construction equipment and tools, preconstruction activities, construction methods, human resource management, and safety and health. This section presents the review these factors.

Productivity Problems Related to Construction Materials Management

Previous researchers estimated the adverse effect of the factors related to construction materials on construction projects' productivity. Thomas et al. (1989) conducted a case study of a five-story commercial building and estimated that 18% of the time overruns is attributed to the poor materials management practices. Similarly, Thomas and Sanvido (2000) investigated the effect of inefficient materials management on productivity using three building projects as case studies and found the schedule slippage between 50% and 129%. The factors related to construction materials and reducing construction projects' productivity include the lack of construction materials, delay in delivering materials, poor quality of materials, and improper handling of materials.

Lack of construction materials has been identified by previous researchers as the most critical problem that affects productivity in construction projects. The lack of construction materials was found to be the among the top three factors which reduce construction projects' productivity (Olomolaiye et al., 1987). Zakeri et al. (1996) found the shortage of materials as the critical factor deterring productivity of construction projects. Kaming et al. (1997) identified that the lack of construction materials, which is caused by on-site transportation difficulty and excessive paperwork, is one of the most critical productivity problems. Liberda et al. (2003) identified non-availability of construction materials as the third management related problems that are influencing construction productivity. The shortage of materials due to liquidity problems of the contractors and lack of coordination between the project site and contractor's head office is identified among the major productivity problems (Makulsawatudom et al., 2004). Abdul Kadir et al. (2005) identified material shortage caused by the inaccessibility of items and excessive travel to get them as the most critical factor causing low productivity in residential building construction projects. In the UK, the blue-collar workers mentioned the unavailability of materials and their components among the factors that could potentially reduce their productivity (Chan and Kaka, 2007). According to the survey conducted by Dai et al. (2009b), crafts in the United States responded that the lack of construction materials reduced their productivity. Rivas et al. (2011) concluded that shortage of materials is the most significant impediments to construction productivity. The shortage of materials is also cited as productivity problem in the construction industry of Kuwait (Jarkas and Bitar, 2012). The unavailability of materials is also found to be one of the main factors which hinder the increment of construction projects' productivity (El-Gohary and Aziz, 2014). In Australia, Hughes and Thorpe (2014) identified the shortage of construction materials among the top five productivity problems. The findings of the above studies indicate that material unavailability is one of the major productivity problems in both developed and developing countries. Although the previous studies identified the productivity problems, the materials management practices that could enhance the productivity of the multi-storey building construction projects were not identified and prioritised. Problem identification alone might not improve productivity unless the problem is integrated with the respective solutions (practices).

28

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

Delay in delivering construction materials is found to be another factor which reduces the productivity of construction projects. Work stoppage due to delay in delivery of materials to a construction site was perceived as the most significant factor in Singapore construction industry (Lim and Alum, 1995). Thomas and Sanvido (2000) found that unsequenced deliveries of construction materials caused unnecessary crane movements and rework. Late delivery of materials and coordination problem with suppliers were identified as the major determinants (Abdul Kadir et al., 2005). Similarly, poor procurement plan and delayed payments to the suppliers or sub-contractors led to the delayed delivery of materials and loss of productivity in construction projects (Ghoddousi and Hosseini, 2012). Delay in ordering materials was mentioned among the barriers to increasing the productivity performance (El-Gohary and Aziz, 2014). Thus, the materials management practices that should be implemented to counteract the loss of productivity due to the delay in delivering the materials needs to be identified in the context of multi-storey building construction projects.

The use of poor quality construction materials which leads to rework is also among the significant factors which negatively influence construction projects' productivity. Dai et al. (2009b) found that poor materials quality is among the main problems influencing the productivity of construction projects. Ghoddousi and Hosseini (2012) identified that rework due to the failure of construction materials or construction works to pass tests were among the major impediments to productivity in construction projects. Furthermore, the authors stated that errors in fabrication, which should be rectified by rework, had an adverse impact on the progress of the work. Although quality related issues were mentioned by previous researchers, which materials management practices are suitable to reduce quality related problems were not identified.

Improper handling of construction materials, which results in breakage and loss of the materials, is also mentioned by previous researchers as productivity problems. Thomas et al. (1989) identified multiple handling of materials, improper sorting or marking materials, and errors in fabrication as adverse conditions for productivity. Improper allocation of materials and lack of storage facilities were considered as a major productivity problem in urban high-rise construction sites in Singapore (Kaming et al., 1997). Moreover, Dai et al. (2009b) found that difficulty in tracking materials on site as one of the factors that reduce construction productivity. Rivas et al. (2011) concluded

that locating on-site materials storage areas far from the working areas had reduced the productivity of construction projects. The problems associated with the ease of handling materials is also identified as the third critical determinants of construction productivity (El-Gohary and Aziz, 2014). In the context of Victoria State, Australia, the issue of materials handling could be one of the productivity problems in the construction of multi-storey building projects which needs the implementation of appropriate materials management practices.

Productivity Problems Related to Construction Equipment Management

Construction equipment and tools related problems were identified among the critical factors that negatively influence the productivity of construction projects. The major productivity problems which are frequently cited by previous studies include lack of equipment, tools, and consumables on construction sites; and breakdown of equipment and tools.

Lack of appropriate construction equipment and tools has a significant influence on the production rates of the crews. Olomolaiye et al. (1987) identified lack of proper tools as the third main hindrance to craftsperson productivity. The finding of the study indicated that up to 3.5 hours per person per week could be lost in steel fixing activity due to unavailability of steel bending machine. The authors concluded that poor quality of tools, improper maintenance, and outright shortages were the main causes of the unavailability of tools. Moreover, Makulsawatudom et al. (2004) identified lack of tools and equipment as the fourth deterrent for productivity increment in Thailand. The main reasons cited by the authors were the insufficient supply of tools and equipment; lack of maintenance of tools and equipment which ultimately lead to the use of old and inefficient machinery and its spare part; and error in estimation of the capacity of the equipment that resulted in the purchase of inefficient machines. Furthermore, Dai et al. (2009a) found that unavailability of the crane, forklifts, man lifts (hoists), consumables and tools; and misplacement of tools were the critical factors reducing the productivity of craftsmen. Some of the previous studies indicated that lack of tools and equipment is caused by various factors such as shortage of funds, concurrent construction of various projects leading to an increase in the demand for equipment, failure to report broken down tools and equipment; and lack of organized, safe and secure storages in construction sites (Hughes and Thorpe, 2014, Ghoddousi and Hosseini, 2012). The

construction of multi-storey building projects in Victoria could also face the problem of unavailability of construction equipment and tool; the tradesmen could spend much time in waiting for the delivery of the required materials due to the unavailability of the equipment such as forklift, crane or hoists; and, in some circumstances, tools could also be misplaced and difficult to find them. Thus, investigation of good construction equipment and tools management practices is essential to alleviate the problems in Victorian context.

Breakdown of tools and equipment is another major productivity problem that reduces productivity in construction projects. Olomolaiye et al. (1987) explained that, on top of poor initial quality, handling problems and lack of maintenance such not oiling at appropriate intervals and replacing worn parts are the main cause of the breakdown of the machinery. Similarly, Alinaitwe et al. (2007) found that the frequent breakdown of equipment such as water pumps and vibrators due to poor maintenance and lack of service affect the productivity of construction projects in Uganda. Moreover, various researchers have concluded that equipment breakdown is the critical factor which reduces productivity (Kaming et al., 1997, Makulsawatudom et al., 2004, Ghoddousi and Hosseini, 2012, Hughes and Thorpe, 2014). Breakage of construction equipment and tools could also be one of the productivity problems in the construction of multistorey building projects in Victoria, and investigation of the most suitable construction equipment management practices is significant for counteracting the adverse effect of these problems.

Productivity Problems during Pre-Construction Phase

Previous researchers identified the factors which have negative impacts on construction projects' productivity and related to pre-construction phase activities. These factors include inadequate planning, disruption of utilities, lack of contracting strategies, frequent changes, the complexity of designs, incomplete project information, and delay in permits.

Inadequate or lack of construction plan is one of the major factors which are frequently mentioned by previous researchers as impediments to construction projects' productivity. Zakeri et al. (1996) explained that the lack of proper work plan by site managers could lead to project delay since priorities might not be given to the execution of activities on the critical path. Durdyev and Mbachu (2011) found the inadequacy of

planning as one of the constraints to on-site construction productivity in New Zealand. Ghoddousi and Hosseini (2012) identified lack of construction planning, and improper planning and sequencing of the tasks as the most critical factors that have adverse effects on the construction projects' productivity in Iran. The absence of construction planning as an impediment to construction projects' productivity was also identified by El-Gohary and Aziz (2014). In the construction of building projects in Victoria State, Australia, if the construction tasks are not clearly planned and sequenced to the lowest possible execution level or a daily activity, then there could be project delay as the crews might not fully understand what activities to execute and when they should be started and completed.

Disruption of utilities such as power and water is also among the critical factors which negatively influence productivity. Lim and Alum (1995) found that stoppage of utilities has an adverse effect on productivity. The authors explained that the main causes of the disruption of power were the damage to the underground cables by excavators, and the reason for water interruption was the bursting of water pipes during excavation works. Alinaitwe et al. (2007) also identified disruption of power and water services as factors that negatively affect construction productivity. Moreover, disruption of power and water services are considered as the main factors which negatively influence the productivity of construction projects in Iran (Ghoddousi and Hosseini, 2012). In the context of Victoria, Australia, the lack of water can delay activities that need water during their placement. For instance, if a project is using in-situ concrete, then the lack of water affects the production of concrete, and the concrete crew will be idle. Consequently, there could be the loss of productivity. Power disruption could also have a similar effect on activities that need power in their production and fixing such as welding of door and windows, trusses, and other related tasks.

The use of inappropriate contract types is also mentioned as one of the impediments to construction productivity. Lim and Alum (1995) described that work stoppages due to the insolvency of subcontractors were the main causes of the loss of the construction projects' productivity. Durdyev and Mbachu (2011) identified the type of procurement adopted as the productivity constraints in New Zealand. In the context of multi-storey building projects in Victoria, the lack of investigating and adopting suitable project delivery or contracting strategies could lead to project delays, cost overruns, and

disputes. For instance, if the contract is awarded to a subcontractor without examining its technical and managerial capacity, the consequence could be the loss of productivity due to the lack of the capability of the subcontractors. Besides the contract types, insufficient considerations of the delivery methods such as 'design and construct' or 'construct only' during the planning stage could affect productivity. For example, if the adopted project delivery model is 'construct' only, then the subcontractor has to wait for a designer who has the authority to make amendments to the design.

Another productivity problem that is frequently cited by previous researchers is the changes in design, drawing, specification and scope of works. Rojas and Aramvareekul (2003) identified that change in the scope of works is one of the most critical factors that negatively influence productivity. Alinaitwe et al. (2007) found that change in design is one of the main problems influencing construction productivity in Uganda. In New Zealand, Durdyev and Mbachu (2011) mentioned the frequency of design changes as one of the constraints of productivity. Ghoddousi and Hosseini (2012) concluded that rework due to changes in design, drawings or specifications is the most critical factor that affects productivity. El-Gohary and Aziz (2014) found that work interruptions due to design changes are some of the productivity problems in the construction projects in Egypt. Other researchers also found that change order due to errors in design affects workers morale, causes rework, disrupts the sequence of activities and ultimately decreases construction productivity (Zakeri et al., 1996, Makulsawatudom et al., 2004, Abdul Kadir et al., 2005, Jarkas and Bitar, 2012). The issue of the incomplete document that is subject to numerous revisions and changes could be one of the productivity problems in multi-storey building construction projects in Victoria, Australia. These frequent changes can cause rework, idleness, and dispute between the principal contractor and subcontractors.

The complexity of designs and provision of less emphasis on the constructability issues can also negatively influence productivity. Alinaitwe et al. (2007) found that the complexity of designs hampers the productivity performance of Ugandan construction projects. Enshassi et al. (2009) mentioned the effect of project complexity as the major problems in construction projects in Gaza. Durdyev and Mbachu (2011) also identified project complexity as an internal constraint to on-site productivity in New Zealand. Jarkas and Bitar (2012) indicated that the complex nature of the design could result in

rework and loss of productive working hours. El-Gohary and Aziz (2014) found that the lack of integration between design and construction or lack of consideration for constructability issues is one of the most critical factors which negatively influences productivity in construction projects, and the study concluded that there is a serious lack of cohesion between the design and construction teams in Egyptian construction industry. If a design of multi-storey building project in Victoria is complex, it could take a longer time to understand and implement the design, and productivity could be affected. Furthermore, if the design of the multi-storey building is not buildable by using the existing construction methods, the project's productivity performance can also be affected.

Provision of incomplete information such as work order, specification and drawings to crew members can also negatively influence the productivity of construction projects. Olomolaiye et al. (1987) found that instruction delay reduced the productive time of the operatives as they wait for work order from the supervisor. An incomplete design was considered to be the second most significant factor in Thailand, and the main reason mentioned by the authors was the inability of the client to provide sufficient time and budget for the designer to complete the design (Makulsawatudom et al., 2004). According to Alinaitwe et al. (2007), poor communication due to inaccurate instructions and drawings highly influenced productivity as the level of literacy of construction operatives were low. The study revealed that most communications were verbal which could not be as accurate as the written ones. Similarly, El-Gohary and Aziz (2014) identified lack of clarity of instructions and information exchange in construction projects in Egypt. Furthermore, slow in responding to questions regarding drawings, errors in drawings and availability of drawings were mentioned as productivity problems in US construction projects (Dai et al., 2009b, Dai et al., 2009a). Added to this, Rivas et al. (2011) revealed that lack of construction experiences of the designers is one of the reasons for poor drawings, and for the wastages of time in the interpretation of design during the construction phase. By the same token, delay in responding to request for information (RFI) is among communication gaps leading to low productivity (Jarkas and Bitar, 2012). Finally, the Australian construction projects' productivity is also negatively affected by the lack of complete design (Hughes and Thorpe, 2014). The study explained that the reason for the incomplete design was insufficient details

provided by designers, inadequate examination of completed drawings and inadequate time provided to draftsperson. Thus, in the context of Australia, an investigation of the management practices which have positive impacts on productivity and counteract the adverse effect of the lack of complete project information on productivity is essential.

Delays in getting permits as well as stoppage of works due to the breach of regulatory requirements are some of the causes of the loss of construction projects' productivity. Lim and Alum (1995) explained that work stoppage orders due to infringement of government regulations affected productivity in the construction projects in Singapore. Moreover, Dai et al. (2009a) found that waiting for the work permits such as hot work permit and confined space permit delayed the productivity of construction projects. Other researchers also revealed the influence of regulatory requirement on productivity (Alinaitwe et al., 2007, Chan and Kaka, 2007, Durdyev and Mbachu, 2011). As there are various regulatory requirements in the building and construction industry of Victoria, failure to comply with these requirements could also lead to loss of productivity in building construction projects.

Productivity Problems Related to Human Resource Management

Previous studies have identified the problems that hamper construction projects' productivity from the human resource perspective. These factors include lack of skilled supervisors and craftsmen; absenteeism; lack of motivation and employee turnover; inadequate allocation of workers and frequent change of employees; and unfairness in payment.

The skill deficiency of operatives and supervisors has a negative impact on the productivity of construction projects. Olomolaiye et al. (1987) found that supervisors' incompetence was one of the contributing factors for low productivity of operatives. The study revealed that a high proportion of inspection delays could have been avoided if the supervisors had provided clear instruction and understood the project details. Makulsawatudom et al. (2004) ranked the presence of incompetent supervisors as the third critical factor which reduces construction projects productivity. The study suggested that the incompetent supervisors might be responsible for the defective works, and for inappropriate application of tools and equipment. El-Gohary and Aziz (2014) explained that the most common characteristics of unskilled operatives are the presence of low and faulty outputs coupled with unreasonably high inputs. Similar

studies indicated the adverse impact of the lack of skill and experience of construction workers on productivity (Liberda et al., 2003, Alinaitwe et al., 2007, Chan and Kaka, 2007, Dai et al., 2009b, Ghoddousi and Hosseini, 2012, Jarkas and Bitar, 2012, Hughes and Thorpe, 2014). In the construction of multi-storey building projects in Victoria, the shortage of skill and experience of the workforce can affect productivity as well as the quality of the works. If the operatives do not have the required level of skills, then they might perform poor quality works which can be subjected to rejection by supervisors. As a result of this, contractors can do reworks to rectify the defects, and the productivity of the projects is affected.

Absenteeism is also often mentioned by previous researchers as one of the factors related to human resource and negatively influencing the productivity of construction projects. Lim and Alum (1995) mentioned that absenteeism is the fourth critical factor curtailing productivity of construction projects in Singapore. Similarly, Makulsawatudom et al. (2004) ranked absenteeism as the fifth factor reducing productivity in the construction projects in Thailand. Rivas et al. (2011) identified that the main causes of absenteeism are sickness, fatigue, alcoholism, personal problems and time for personal activities. Furthermore, many studies mentioned the effects of absenteeism on construction projects' productivity (Olomolaiye et al., 1987, Kaming et al., 1997, Durdyev and Mbachu, 2011, Hughes and Thorpe, 2014). In the multi-storey building construction projects in Victoria, absenteeism which could be unavoidable due to ill health or deliberate absence to search for more lucrative jobs could hamper productivity. Thus, the progress of the building works could be delayed due to unavailability of operatives on the scheduled date. It could also lead to the lag of an overall project completion date if the delay occurs on the critical path.

The lack of motivation by workers to work with full potential is also one of the factors which have negative impacts on productivity. Liberda et al. (2003) identified that the lack of motivation is the second critical factor, under human resource category, which impinged productivity of construction projects in Canada. Durdyev and Mbachu (2011) identified the level of motivation or commitment of the workforce as one of the factors influencing productivity in New Zealand. Jarkas and Bitar (2012) found that reduced motivation of labour is ranked first among the labour related factors which hamper construction productivity. The study indicated that if operatives are motivated, they are

more enthusiastic and initiative; they work harder and respond faster to instructions; they feel satisfied and responsible; and they are more productive than demotivated workers. Similarly, if construction employees in the multi-storey building projects in Victoria are not motivated, they could exert minimal effort which can reduce the potential to get maximum possible productivity.

Employee turnover is the other factor which negatively influences productivity in construction projects. Kaming et al. (1997) revealed that the principal causes of turnover in the construction projects in Indonesia are the lack of adequate work volume, better pay from other projects, distance from home to the site, lack of better working environment and the presence of less challenging works. Similarly, Rivas et al. (2011) found that high turnover occurs due to the expectation of better salary by employees and the company do not provide them; personal problems and preference for working close to home; lack of incentives; short-term contracts; and supervisor's mistreatment. Other studies also investigated and revealed the negative impact of turnover on construction productivity (Makulsawatudom et al., 2004, Alinaitwe et al., 2007, Chan and Kaka, 2007, Durdyev and Mbachu, 2011). By the same token, if the rate of employee turnover in multi-storey building projects in Victoria is high, then there could be a disruption to construction progress as the employees leave with their knowledge and skill that they have acquired from the project.

The frequent change of staff and inappropriate allocation of workers in their respective crews are mentioned as some of the determinants of construction projects' productivity. Olomolaiye et al. (1987) found that changing the crew members were one of the productivity problems in Nigerian construction projects. Zakeri et al. (1996) identified changing crew size as one of the impediments to construction productivity in Iran. Abdul Kadir et al. (2005) revealed that an allocation of an inadequate number of workers is one of the major causes of delay in Malaysian construction projects. Alinaitwe et al. (2007) found that the change of Foremen and other crew members affect the productivity of building projects in Uganda. Dai et al. (2009b) concluded that pulling people off a task before it is completed has an adverse impact on productivity. Other similar studies indicated the impact of the change in both management staff and labour on productivity (Kaming et al., 1997, Liberda et al., 2003, Ghoddousi and Hosseini, 2012, Jarkas and Bitar, 2012). In the context of Victoria, Australia, there

could also be a frequent change in the multi-storey building construction projects' employees which can negatively influence the productivity of the projects. Thus, investigation of the most suitable human resource management is significant to reduce the loss of productivity due to the frequent changes of the workers.

The inconsistency in payment schemes is another human resource related factor which negatively influences the productivity of construction projects. Abdul Kadir et al. (2005) found that workers strike due to unpaid work are some of the determinants of construction projects' productivity in Malaysia. Likewise, Dai et al. (2009a) identified the uses of different pay scales and per diem rates for the same job on the same project have negative effects on the performance of the construction workers. El-Gohary and Aziz (2014) mentioned that the payment scheme for labourers such as paying on a daily wage basis or paying a lump sum for a certain task has an impact on productivity. The issue of untimely payment of wages to the construction workers in Victoria could also reduce the productivity of building projects, and implementation of the most appropriate human resource management is essential to enhance productivity.

Productivity Problems Related to Construction Methods

Based on the findings of the previous studies, the factors which reduce construction projects productivity and related to construction methods include improper sequencing of works, excessive shift works or overtime, lack of adequate instruction and supervision, poor site layout, and lack of management commitment to investigate new technologies.

Improper sequencing of the works of subcontractors is found to be one of the factors reducing productivity in construction projects. Rojas and Aramvareekul (2003) found that the lack of subcontractor integration is one of the main productivity determinants in construction projects. Chan and Kaka (2007) identified that lack of sequencing and interference is one of the productivity problems in the UK construction projects. Similarly, Dai et al. (2009b) concluded that out of sequence work assignments is one of the main reasons for low productivity performance in the US construction projects. Other researchers such as Motwani et al. (1995), Liberda et al. (2003), Alinaitwe et al. (2007), Rivas et al. (2011), Jarkas and Bitar (2012) and El-Gohary and Aziz (2014) also investigated the impact of poor sequencing of works on productivity. As the construction of multi-storey building project in Victoria involves many subcontractors,

inappropriate sequencing of the works of the subcontractors could result in loss of productivity. The crews of the subcontractors could interfere which leads to a slow progress of the project.

The lack of suitable working schedule strategies is another factor identified as productivity problem. Jarkas and Bitar (2012) ranked working overtime as the sixth critical factor influencing productivity in Kuwait. The study suggested that the operatives could lose their strength and moral due to the prolonged working hours; as a result, there could be low productivity and rework due to poor workmanship. Similarly, Hughes and Thorpe (2014) identified work overload as the fourth critical factor influencing the productivity of construction projects in Australia. The study suggested that working continuously for seven days per week has more negative impact on productivity than working overtime during the normal working days. Other studies also confirmed the effect of long working hours on productivity (Liberda et al., 2003, Ghoddousi and Hosseini, 2012, El-Gohary and Aziz, 2014). Therefore, the use of excessive shift length, unscheduled breaks, and working overtime could be practised in Victorian multi-storey building construction projects, and investigation of the appropriate management practice is significant to reduce the loss of productivity. The use of excessive shift length, for instance, can increase workload to the workers and cause physical tiredness which in turn leads to the reduced efficiency.

Inadequate instruction and supervision are some of the critical problems which reduce productivity in construction projects. Zakeri et al. (1996) concluded that repeating construction activities due to poor instruction and supervision, and delays in the inspection are the most critical factors that hinder construction productivity in Iran. Kaming et al. (1997) found that rework due to poor instruction is the second most significant problem that reduces the craftsmen's productivity in Indonesia. Liberda et al. (2003) identified supervision problems as one of the factors which negatively affect productivity in Alberta, Canada. Alinaitwe et al. (2007) identified inspection delay as one of the factors which decrease productivity in Ugandan building projects. In the US, Dai et al. (2009b) identified the inadequate instruction by supervisors as a critical productivity problem. Rivas et al. (2011) found that waiting for inspections, lack of preplanning of quality control and not notifying the quality controlling procedure in advance are the major problems of construction projects productivity in Chile.

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

Ghoddousi and Hosseini (2012) mentioned the stoppage of works due to inspection delays in construction projects in Iran. The inspection delay by site management is also concluded as a major impediment to productivity in Kuwait (Jarkas and Bitar, 2012). In Victorian multi-storey building construction projects, the lack of adequate inspection by contractors' supervisors could lead to rework and overall project delay. Furthermore, the productivity of the contractors could be hampered due to the lack of appropriate instruction.

Poor job site layout is another factor that negatively influences construction projects' productivity. Kaming et al. (1997) concluded that overcrowding on a construction site is among the major factors that reduce construction projects' productivity. Liberda et al. (2003) identified that congested work areas, poor site layout, and insufficient parking and other facilities on site are the major factors which reduce construction projects' productivity in Canada. Chan and Kaka (2007) mentioned that congestion of construction site is also one of the main factors influencing construction productivity in the UK. Alinaitwe et al. (2007) found that poor access and overcrowding on the site are the main factors reducing the productivity of construction projects in Uganda. According to the survey conducted by Dai et al. (2009a), long travel time due to the size of a project and the location of storage space being far from the working areas are some of the factors which reduce crafts' productivity in the US construction projects. Rivas et al. (2011) identified that overcrowded working area due to lack of coordination among supervisors and allocation of an excessive number of workers is the main productivity reducing factor in construction projects in Chile. In New Zealand construction projects, poor site layout is identified as the major productivity problem (Durdyev and Mbachu, 2011). Ghoddousi and Hosseini (2012) found that poor job site layout and overcrowding are the factors reducing the productivity of construction projects in Iran. Jarkas and Bitar (2012) identified unsuitability of storage location, site access, confinement of working space and poor site layout as the main factors which negatively influence construction projects' productivity in Kuwait. In the Australian context, poor site layout and overcrowding are also found to be some of the problems influencing construction projects' productivity (Hughes and Thorpe, 2014). Therefore, investigation of the techniques which reduce the loss of productivity of the Victorian multi-storey building construction projects due to poor site layout is significant.

The lack of management commitment to assess and implement new technologies is also identified as a factor which influences the productivity of construction projects. Chan and Kaka (2007) identified that the lack of investigation of prefabrication and standardisation as the productivity determinants in the UK construction projects. Durdyev and Mbachu (2011) found that technology and process related issues such as inadequacy of method of construction, resistance to accepting new technologies, insufficiency of technology, lack of awareness or training in new technologies, inadequate IT infrastructure and its application in the construction industry are the factors reducing construction projects' productivity in New Zealand. Ghoddousi and Hosseini (2012) found that utilising the traditional construction methods instead of modern technology has reduced productivity in construction projects in Iran. El-Gohary and Aziz (2014) identified poor construction technology as one of the factors negatively influencing construction projects' productivity in Egypt. In multi-storey building projects in Victoria, the lack of implementation of appropriate technology during planning and construction of the projects might also influence productivity.

Productivity Problems Related to Safety and Health

Previous researchers have identified factors which are related to safety and health and hamper construction projects' productivity. These factors include the occurrence of accidents in construction projects, unsafe work conditions, lack of cleanliness of construction sites, alcohol and substance abuse, and health condition of the employees.

Accidents in construction projects are some of the main factors that can reduce the productivity of the construction projects. Lim and Alum (1995) found that the occurrence of accidents that resulted in a death of an employee led to the issuance of stop-work orders by the government authorities, and productivity was negatively affected. Enshassi et al. (2007) identified three types of accidents which influence productivity in the construction projects. The first one is the accident that results in the death of an injured worker and the stoppage of work for several days. The second one is an accident that causes an injured labourer to be hospitalised for at least 24 hours and decreases the productivity of a crew in which the affected employ belongs. The third one is a minor accident that results from objects such as nails and steel wires and has a minor effect on the productivity of a crew. Kazaz and Ulubeyli (2007) described that occupational injuries at construction sites have an adverse effect on the reputation of a

company, decrease productivity and result in a huge cost. Other researchers also mentioned that accidents in construction sites are the major deterrents to productivity (Makulsawatudom et al., 2004, Abdul Kadir et al., 2005, Alinaitwe et al., 2007, Jarkas and Bitar, 2012, Hughes and Thorpe, 2014). Therefore, to minimise the occurrence of accidents and reduce the negative impacts of the accidents on the productivity of multistorey building projects in Victoria, investigation of appropriate safety and health practices is essential.

Unsafe work conditions such as noise, dust, radiation, poor ventilation, low lighting, limited access, unavailability of personal protective equipment and other work-related conditions have adverse effects on construction projects' productivity. Liberda et al. (2003) identified that noise, dust and radiation are some of the causes for the decrement of construction projects' productivity in Canada. Enshassi et al. (2007) found that insufficient lighting, inadequate ventilation and working at height are the three critical factors influencing construction projects productivity. Ghoddousi and Hosseini (2012) described that high level of dust and noise, low lighting, and ventilation affect the productivity of operatives in construction projects in Iran. El-Gohary and Aziz (2014) mentioned that working at heights is one of the problems affecting productivity in Egypt's construction projects. Since the working conditions of most construction projects in Victoria State, Australia, might also face productivity problems due to unsafe working conditions.

Lack of cleanliness of construction sites, alcohol and substance abuse, as well as the health conditions of employees, can significantly affect productivity in construction projects. Lim and Alum (1995) indicated that alcoholism and health issues such as mosquito breeding due to poor housekeeping are some of the factors that influence productivity. Alinaitwe et al. (2007) identified alcoholism, drug abuse, and poor health of workers such as sickness and general weakness as the problems impinging productivity in building construction projects in Uganda. Ghoddousi and Hosseini (2012) also mentioned that poor housekeeping is one of the productivity constraints in construction projects in Iran. Other researchers such as Chan and Kaka (2007) and Durdyev and Mbachu (2011) also mentioned the issue of the health of the workforces on productivity. The construction workers involved in multi-storey building

construction projects in Victoria, Australia, might use alcohol and drugs which could affect their productivity. Thus, the practices of how these issues are controlled or minimised worth investigation. In next section, review of the management practices that have the potential to enhance productivity in industrial and infrastructure projects are described.

2.6 Management Practices Enhancing Productivity in Industrial and Infrastructure Construction Projects

Project Level Versus Activity Level Management Practices

Construction management practices that could improve productivity at project level can also enhance productivity at an activity level. However, practices which could improve the productivity of an activity might not necessarily increase the productivity of a construction project. For instance, locating cranes in most appropriate location for the construction of many work packages can improve the productivity of a project or group of activities such as concrete work, façade, and structural steel works. However, positioning a crane in the most suitable position for only one activity could enhance the productivity of that activity, for instance, façade work, but might not improve the productivity of other groups of activities or a project. Similarly, procurement of a crane by considering the maximum weight to be lifted in a single work package can affect the productivity of other work packages as these work packages might have the heaviest materials which the procured crane could not lift. Thus, focusing on activity level management practices might not increase the overall project productivity.Therefore, this research focuses on productivity improvement at the project level.

Previous researchers have identified project level management practices that have the potential to improve productivity in industrial and infrastructure construction projects and developed scoring tools for these project types in North America (Nasir et al., 2015, Caldas et al., 2014, CII, 2013b, CII, 2013a, Nasir, 2013).

Management Practices Enhancing Productivity in Industrial Construction Projects

The management practices that could improve productivity in industrial projects in North America have been identified and organised into elements, sections, and categories. There are 53 elements which are grouped into 18 Sections and 6 Categories (Table 2-5). The six Categories have been defined based on the experience of the panel of experts (Caldas et al., 2014).

Table 2-5 Elements,	, Categories	and Sections	of the	Best Pra	actices for	r Industrial Pr	ojects
	(adapted from	m Caldas et al.	(2014)	and CII	(2013a))		

Categories	Sections	Elements	Weights
		A1. Project Team Material Status Database	36
		A2. On-site Material Tracking Technology	45
I Materials	A. Material Management	A3. Material Delivery Schedule	59
Management	Systems	A4. Procurement Plan for Materials and Equipment	56
	B. Receipt and Inspection	B1. Material Inspection Process	63
	of Materials	B2. Material Inspection Team B2. Doct receipt Processing and Maintenance	48
		A1 Site Tool and Consumplies Management Strategy	30
	A Site Tool Management	A1. She Tool and Consumatics Management Strategy A2 Tool Tracking Systems	29
	A. Site 1001 Management	A3 On-site Tool Maintenance	29
II. Equipment		A4. Control System for Tool Delays	25
Logistics	B. Machinery Availability	B1. Construction Machinery Productivity Analysis	67
	,	B2. Equipment Maintenance	106
	A. Short Interval Planning	A1. Short Interval Planning	74
		B1. Well Defined Scope of Work	19
		B2. Utilization of Software to Assist in Generating Work	11
		packages D2 Draiget Model Deguirements	12
III. Craft	B. Workface Planning	B5. Project Model Requirements	12
Information		B4. Dedicated Planner	15
Systems		B5. Identify Required Permitting	13
		B6. Engineering Work Packages (EWP)	14
		B7. Construction Work Packages (CWP)	15
		B8. Installation Work Packages (IWP)	15
	C. Constructability Review	C1. Design Readiness for Construction	67
	A Turining and	C2. PPMOF Evaluation	40
	A. Iraining and Development	A1. Trades recifical fraining A2. Career Development	33
IV Human	Development		
IV. Human	B Behaviour	B1. Nonfinancial Incentive Programs	24
Management	D. Denuviour	B2. Financial Incentive Programs B3. Social Activities	28
Management		B5. Social Activities	10
	C. Organizational Structure	C1. Maintain Stability of Organizational Structure	32
		C2. Clear Delegation of Responsibility	36
	D. Employment	D1. Retention Plan for Experienced Personnel D2 Exit Interviews	40
	A Sequence and	A1. Integrated Schedule	47
	Scheduling of Work	A2. Work Schedule Strategies	30
	C	A3. Schedule Execution and Management	46
V. Construction	B. Start-up.	B1. Planning for Start-up	36
V. Construction Mathada	Commissioning, and	B2. Testing Procedures	28
Wiethous	Turnover Plan	B3. System Turnover Procedure	34
	C. New Technology	C1. New Equipment Investigation	20
	Investigation	C2. New Information System Investigation	16
	-	C3. New Materials Technologies Investigation	18
	D. Site Layout Plan	D1. Dynamic Site Layout Plan	38
	5	D2. Site Security Plan	29
		D3. Equipment Positioning Strategy	36
		A1. Zero Accident Techniques	38
	A. Job Safety	A2. Task Safety Analysis	36
VI. Environment		A3. Inclutication of Potential Hazards	54 21
Safety, and		A5. System Test Hazards Planning	24
Health	B Substance Abuse	,	88
	Programs	B1. Substance Abuse Programs	
	C. Safety Training and	C1. OSHA Compliance Training	62
	Orientation	C2. Toolbox Safety Meeting	81
		Total Weight	2000
		č	

44

The 18 sections are further organised into 6 categories: (1) Materials Management, (2) Equipment Logistics, (3) Craft Information Systems, (4) Human Resource Management, (5) Construction Methods, and (6) Environment, Safety, and Health (Caldas et al., 2014). Each practice is assigned a weight based on the relative importance of the practice in improving the productivity of industrial construction projects in North America. The total weight is 2000 points (Table 2-5).

Scoring Tool for Management Practices Enhancing Productivity in Industrial Construction Projects

The scoring tools for the 53 practices have been developed by the Construction Industry Institute (CII, 2013a). The tool has been validated by collecting productivity and management practices data from 33 industrial projects in North America. The scoring tool for one of the practices, On-site Materials Tracking Technologies, is indicated in Table 2-6. Accordingly, if a contractor plans to implement Level 2 or "the materials are assigned a laydown and storage area and the information is recorded," then the score for this practice will be 12. Using similar procedure, other practices are scored, and a total score for an industrial project in North America is computed. For the sake of brevity, the scoring tools for the other 52 practices are not presented here.

Table 2-6 The Levels and Weights of On-site Materials Tracking Technologies in the Context
of Industrial Projects (adapted from CII (2013a))

	Practice: On-site Materials Tracking Technologies	Weights
Level 0	On-site materials tracking technology is not applicable.	0
Level 1	No tracking is done on site beyond receivables	1
Level 2	Materials are assigned a laydown and storage area and the information is recorded.	12
Level 3	Continuation of level 2, plus the location information is kept updated in a software system, and well-defined processes for establishing lay down area grids, developing pick lists, flagging, warehouse organization (if applicable), and other necessary functions.	23
Level 4	Continuation of level 3, plus the system is supported by tracking software and also supplemented by barcode, GPS, or RFID systems for automated location tracking.	34
Level 5	Continuation of level 4, plus the tracking system is completely automated and integrated with other project processes.	45

Management Practices Enhancing Productivity in Infrastructure Construction Projects Previous researchers identified 61 management practices that have the potential to improve productivity in infrastructure projects (Nasir et al., 2015). The 61 practices are grouped into 20 sections, and the sections are organised into six categories: (1) Materials Management, (2) Construction Machinery and Equipment Logistics, (3) Execution Approach, (4) Human Resources Management, (5) Construction Methods, and (6) Health And Safety (Table 2-7).

Categories	Sections	Elements	Weight
		A1. Procurement Procedures and Plans for Materials and Equipment	35
	A. Procurement Strategy	A2. Long-Lead/Critical Equipment & Materials Identification	49
I. Materials		A3. Procurement Team	19
Management	B. Materials Management	B1. Project Team Materials Status Database B2. On-Site Material Tracking Technology	30
	Systems	B3. Materials Delivery Schedule	20
	C. Receipt and Inspection of	C1. Materials Inspection Process	9
	Materials	C2. Materials Inspection Team	6
	A Construction Machinery	C3. Post Receipt Preservation & Maintenance	5
	and Equipment	A1. Procurement Procedures & Plans for Construction Machinery A2. Construction Machinery Productivity Analyses	57 59
II Construction	Availability	A3. Construction Machinery and Equipment Maintenance	59
Machinery	B Tools and Equipment	B1 Site Tools and Equipment Management Strategy	36
and	Management Best	B2 Tools & Equipment Tracking	49
Equipment	Practices	B3. On-site Tools Maintenance	9
Logistics		B4 Construction Machinery & Equipment Utility Requirements	9
		A1 Short Interval Planning	22
		A2. Well defined scope of work	30
	A. Planning	A3. Use of Software	9
		A4. Dedicated Planner	37
III. Execution		A5. Construction Work Packages (CWP)	18
Approach	B. Constructability Reviews	B1. Design readiness for construction	41
		B2. Utility Alignment & Adjustments	39
		B3. Contract Types/Strategies	38
		B4. Model Requirements/3D Visualization	8
	C. Acquisition Strategy	C1. Right of Way, Land, and Utilities Acquisition Strategy	9 13
		C3. Utility Agreements	24
		D1. Environmental Requirements	51
	4 D1 .	D2. Regulatory Requirements/Permitting Requirements	44
IV. Human	A. Planning	A1. Crews Composition/Crew Formation A2. Skills Assessment and Evaluation	54 50
Resources	B. Training and	B1. Employees / Trades Technical Training	49
Management	Development	B2. Career development	35
	C. Behaviour	C1. Nonfinancial Incentive Programs	32 40
		C3. Social Activities	48
	D. Organizational Structure	D1. Maintain Stability of Organization Structure	7
	E Employment	D2. Clear Delegation of Responsibility E1. Patention Plan for Experienced Personnel	21
	E. Employment	E2. Exit Interview	33
	A. Project Schedule Control	A1. Integrated Schedule	35
		A2. Work Schedule Strategies	10
V. Construction	D. Site Levent Dlen	A3. Schedule Execution and Management	52
Methods	B. She Layout Flair	B2. Traffic Control Plan	28
		B3. Site security plan	8
		B4. Machinery and Equipment positioning strategy	19
	C. Design/Construction Plan	C1. Communications. Coordination & Agreements	50
	and Approach	C2. Project start-up plan	18
		C3. Project Completion Plan	30
		C4. Innovations & New Technologies	60
		C5. Housekeeping	42
	A. Job Site Safety	A1. Formal Health and Safety Policy	64
		A2. meaning and Safety Plans/Zero Accident Techniques	64 56
		A4. Hazards Analysis	57
VI. Health and		A5. Hazards Planning	60
Safety	B. Substance Abuse Program	B1. Drugs and Alcohol Testing Program	13
	C. Health and Safety	C1. Health and Safety Training Programs	60
	Training and Orientation	C2. Toolbox Safety Meetings	45
		I otal Weight	2000

Table 2-7 Elements, Categories and Sections of the Best Practices for Infrastructure Projects (adapted from Nasir et al. (2015), CII (2013b))

The weight for each practice was assigned based on the relative significance of the practices in improving productivity in infrastructure projects. The total maximum score for a certain infrastructure project is 2000 (Table 2-7). In this research context, the term management practices or practices is used to refer to the elements.

Scoring Tool for Management Practices Enhancing Productivity in Infrastructure Construction Projects

The scoring tools for the 61 practices that have the potential to improve productivity in infrastructure projects have been developed by Construction Industry Institute (CII, 2013b). The tool has been validated by collecting and analysing productivity and the management practices data from 31 infrastructure projects in North America. In Table 2-8, the scoring tool for the practice, On-site Materials Tracking Technologies, is indicated for comparison with the scoring tool of the same practice in the context of industrial projects. For the sake of brevity, the scoring tools for the other 60 practices are not presented here.

Table 2-8 The Levels and Weights of On-site Materials Tracking Technologies in the Context of Infrastructure Projects (adapted from CII (2013b))

	Practice: On-site Materials Tracking Technologies	Weights
Level 0	On-site materials tracking technology is not applicable.	0
Level 1	No tracking is done on site beyond receivables	1
Level 2	Materials are assigned a laydown and storage area and the information is recorded manually on paper.	3
Level 3	Continuation of level 2, plus the location information is kept updated in a software system, and well-defined processes for establishing lay down area grids, developing pick lists, flagging, warehouse organization (if applicable) are established and followed.	5
Level 4	Continuation of level 3, plus the system is supported by tracking software and also supplemented by barcode, GPS, or RFID systems for automated location tracking.	7
Level 5	Continuation of level 4, plus the tracking system is completely automated and integrated with other project processes.	8

Comparison of Management Practices and Scoring Tools for Industrial and Infrastructure Construction Projects

The findings of the previous studies indicate that some of the practices which are suitable for enhancing productivity in industrial projects are not suitable for infrastructure projects. Under materials management category, the practices Long-Lead/Critical Equipment and Materials Identification, and Procurement Team are not identified as the practices which could enhance productivity in industrial projects. Under Equipment Logistics category, the practice Control System for Tool Delays is not 47

considered as good practice for infrastructure projects; and Procurement Procedures and Plans for Construction Machinery, and Construction Machinery and Equipment Utility Requirements are found to be not suitable for industrial projects. Under execution approach/ craft information systems category, the practices Engineering Work Packages (EWP); Installation Work Packages (IWP); and Prefabrication, Preassembly, Modularization and Offsite Fabrication (PPMOF) Evaluation are considered as practices that could not enhance productivity in infrastructure projects. The practices Utility Alignment and Adjustments; Contract Types/Strategies; Right of Way, Land, and Utilities Acquisition Strategy; Contracts and Agreements with Agencies; Utility Agreements; and Environmental Requirements are not suitable for industrial projects.

The practices Crews Composition/Crew Formation, and Skills Assessment and Evaluation are some of the human resource management practices which are not included in the list of practices that could increase productivity in industrial projects. In the construction methods category, the practices Testing Procedures, and System Turnover Procedure are not suitable for enhancing productivity in infrastructure projects. Traffic Control Plan; Communications, Coordination, and Agreements are found to be the practices which are not appropriate to improve productivity in industrial projects. The practice Formal Health and Safety Policy is found to be a practice which is not included in the list good practices for industrial projects in the Safety category. The findings of the previous studies suggest that projects are not homogeneous when considering the practices which could enhance their productivity. Thus, multi-storey building construction projects would have different practices that could enhance their productivity.

The relative importance or weight of the practices which are common to both project types are also different. In the materials management category, Project Team Materials Status Database is one of the most important practices for improving productivity in infrastructure projects whereas the same practice is found to be least important for enhancing productivity in industrial projects. Similarly, Short Interval Planning is found to the most critical practice for industrial projects under Craft Information System category, but the practice is not given more weight in improving productivity in infrastructure projects. Clear Delegation of Responsibility is the least important human resource management practice for enhancing productivity in infrastructure projects.

whereas the practice is one of the most important practices for industrial projects. Under construction methods category, Integrated Schedule is found to be the most critical practice in industrial projects whereas Innovations and New Technologies is the most important practice in the context of infrastructure projects. The practice Substance Abuse Programs is the most critical safety and health practice for industrial projects whereas the practice is the least essential for infrastructure projects. Thus, the findings of the previous studies show that management practices are not equally important in improving the productivity of construction projects. Hence, the relative importance of the practices in enhancing productivity in multi-storey building projects in the context of Victoria State, Australia would be different.

Due to the difference in the weights of the practices, the scoring tools of the management practices are not interchangeable. As indicated in Table 2-6 and Table 2-8 above, although the verbal description of the levels of implementation of the practice 'On-site Materials Tracking Technologies' is the same in both project types, different scoring tools have been used for measuring the practice in the context of industrial and infrastructure projects. The weight assigned to each level is different in the two project types. Thus, if a contractor involved in industrial construction plans to implement level 2 or "the materials are assigned a lay down and storage area and the information is recorded," then the score will be 12 points. However, if a contractor involved in the construction of infrastructure project plan to implement the same level in the industrial construction project, the score will be 3 points. This indicates that although the scoring tools are validated for industrial and infrastructure construction projects in similar environment, North America, the tools are different. Therefore, the existing tools cannot be directly used for measuring management practices in multi-storey building projects in Victora as the type of practices as well as their relative weights would be different in the context of Australia.

2.7 Summary

The loss of productivity in construction projects can lead to project delays, liquidated damage, additional cost and reduction in the profit margins of contractors. Previous researchers, both in Australia and international contexts, identified the factors which negatively influence construction projects productivity. These productivity deterrents can be classified into problems related to construction materials, equipment and human

resource management; problems related to construction methods, and safety and health; and pre-construction phase productivity problems. However, as most of these factors are related to the management of the construction projects, they can be minimised by adopting practices which are suitable for a specific project type (building, industrial and infrastructure) in a certain environment or country. Moreover, a tool which is used to measure, plan, monitor, and evaluate the practices is essential to improve productivity.

A systematic review of the literature has been conducted to identify the knowledge gap. Accordingly, previous studies did not identify the management practices that have the potential to enhance productivity in the context of multi-storey building projects, prepare scoring tools for measuring these practices, and develop tools for predicting the probability of exceeding baseline productivity based on a score of the practices. The Construction Industry Institute (CII) has developed two scoring tools for infrastructure and industrial projects in North America. Analysis of the existing tools indicates that the scoring tools are not interchangeable. Some practices which are suitable to improve productivity in industrial projects are not important for infrastructure and vice versa. The relative importance of the practices which are common in industrial and infrastructure projects is also different. Thus, the practices for improving productivity in building projects would be different, and the relative weights of the practices would also be different; consequently, new scoring tools should be prepared and validated. The conceptual frameworks of productivity and the management practices in the context of multi-storey building projects in Victoria are presented in the next chapter. Conceptual Framework

CHAPTER THREE

3 CONCEPTUAL FRAMEWORK

3.1 Introduction

This chapter provides the conceptual framework for the study. First, the neoclassical economic growth theory (the growth accounting framework) is described in Section 3.2. Following this section, the preliminary conceptual frameworks for the study have been developed based on the context of the research and the review of relevant literature (Section 3.3 to Section 3.8). The preliminary conceptual frameworks of productivity and construction materials management practices, construction equipment management practices, pre-construction phase management practices, management practices related to construction methods, human resource management practices, and safety and health practices are described in Section 3.3, Section 3.4, Section 3.5, Section 3.6, Section 3.7 and Section 3.8 respectively. Finally, the summary of the chapter is presented in Section 3.9.

3.2 Growth Accounting Framework

This research investigates the under-researched areas (the management practices aspects) of the growth accounting framework. The framework states that capital deepening, technological progress and management practices can increase the labour productivity. The growth accounting framework originates from the neoclassical theory of economic growth formulated by Solow (1956). The theory explains that the increase in labour force, capital accumulation, and technological progress lead to the rise in output. According to the theory, the rise in the labour force and capital will result in a temporary increment in the output because of the diminishing returns to capital and labour. Diminishing returns refers to the decrease in the rate of growth of output when one of the factors of production (for instance, capital) increases while the amount of another factor (labour) stays constant and vice versa. For example, in the context of construction, consider a certain building project having an insufficient number of concrete workers and sufficient equipment for the concrete production and placing. If more workers are employed, the in-situ concrete production will increase at the beginning of the process. However, after all the required concrete crew members are deployed, adding more workers might not increase the rate of the concrete production

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

Conceptual Framework

and placement. Similarly, the supply of additional concrete equipment might not increase the production and placement of concrete as the required machinery is already provided to the crew.

Figure 3-1 and Figure 3-2 are plotted to explain the Solow economic growth model. As shown in Figure 3-1, at the beginning of the production process, increasing the capital per worker (k) from k_1 to k_2 leads to an increment in the output per worker (y) from y_1 to y_2 and the rate of growth of the output per worker is high or the slope of the production function is steep. However, due to the diminishing returns to capital, the increment of capital per worker will not lead to a continuous increment of output per labour (refer point k_3 , k_4 , y_3 , y_4 in Figure 3-1). The rate of growth of output per worker after y_3 is very low or insignificant.



Figure 3-1 Relationship between Capital per Worker and Output per Worker (adapted from Solow (1956))

In Figure 3-2, the effect of technological progress on output per worker (y) is presented. Due to the technical change in the production process, the production graph shifts from the original position 1 to new position 2. The output per worker (y) also increases from point y_4 to point y_5 without a change in capital per worker (k₄).





An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

52

Conceptual Framework

Based on the neoclassical economic growth theory, the growth accounting framework has been developed. The framework states the importance of management practices besides the technological progress in improving labour productivity (Organisation for Economic Co-operation and Development, 2001, Productivity Commission, 2013, Australian Bureau of Statistics, 2015a). The growth accounting framework describes that the labour productivity growth can be achieved through capital deepening and technical progress which includes management practices (Australian Bureau of Statistics, 2015a). Capital deepening or capital intensity refers to changes in capital to labour ratio and increased capital deepening implies that each labour has more capital to work with to produce certain output (Productivity Commission, 2016). According to Australian Bureau of Statistics (2015b), besides the technological advancement, the technical progress which could increase labour productivity include changes in the management practices, economies of scale and scope, and reallocation of capital and labour. Productivity Commission (2013) also indicated that apart from technological progress and innovations, the increment of labour productivity could occur due to changes in the management practices. However, which types of management practices, in which sector or context, are not investigated. Thus, this research explores the practices in the context of multi-storey building construction projects.

The construction management practices that have the potential to enhance labour productivity in construction projects could be categorized into (1) construction materials management practices, (2) construction equipment and tools management practices, (3) management practices related to construction methods, (4) pre-construction phase management practices, (5) human resource management practices, and (6) safety and health practices (CII, 2013a, CII, 2013b). The preliminary conceptual frameworks for the six categories of the construction management practices are presented in the subsequent sections.

3.3 Conceptual Framework of Construction Materials Management Practices and Productivity

The preliminary conceptual framework of productivity and the construction materials management practices is presented in Figure 3-3. The suitability of the practices shown in the hypothetical framework for improving the productivity of multi-storey building
projects will be investigated during Phase I of this research and the validated conceptual framework will be developed.

Procurement Plans for Construction Materials

Procurement plan for construction materials could be one of the materials management practices that could enhance productivity in multi-storey building projects. Arditi (1985) mentioned the importance of materials procurement plan as the potential area for productivity improvement in the US construction projects. Arditi and Mochtar (1996) concluded that to achieve maximum project efficiency, the timely procurement of materials is crucial. According to the survey conducted by the authors, improving the procurement practices were given higher priority to enhance construction projects' productivity in Indonesia. Nasir (2013) found that materials procurement plan is one of the best materials management practices for improving productivity in infrastructure projects. Caldas et al. (2014) confirmed that materials procurement plan can enhance productivity in industrial projects. In the context of multi-storey building projects in Victoria, preparing the procurement plans for both critical (long-lead items) and non-critical materials could improve productivity by reducing the project delay due to the unavailability of building materials.





Long-lead Materials Identification

Identification of long-lead (critical) materials could also be one of the practices that have the potential to increase productivity in multi-storey building projects. Abdul

Kadir et al. (2005) recommended that to enhance productivity the project management team should plan ahead to ensure that the critical materials are identified, procured and available on site every time. Jergeas (2009) also suggested that to improve productivity in oil and gas capital projects in Alberta, Canada the project team should ensure the availability of critical materials. The author concluded that look-ahead procurement and logistics plans for these materials are essential for enhancing productivity (Jergeas, 2009). Long-lead materials identification is also considered as the best materials management practice for improving productivity in infrastructure projects in North America (Nasir, 2013). Identification of the long-lead materials and preparation of procurement plan for these materials might also improve the productivity of multistorey building construction projects in the context of Victoria State, Australia since most of the critical materials are imported from overseas. If these materials are not identified early, then the project could be delayed. According to Wheeldon (2012), most building materials are purchased from abroad due to the rising costs of manufacturing in Australia. Figure 3-4 shows the manufacturing price indexes of architectural aluminium, paint and coatings, ceramics, and glass products from 2002 to 2016. Accordingly, the prices of the selected building materials are increasing. The proposition that long-lead materials identification is one of the potential areas for the enhancement of the Victorian multi-storey building construction projects' productivity will be verified.



Figure 3-4 The Manufacturing Price Indexes of Selected Building Materials (Plotted using data obtained from Australian Bureau of Statistics (2017))

Materials Procurement Team, Materials Status Database, and Post-Receipt Preservation

Assigning the materials procurement team, controlling the status of materials on site, and maintenance of the received materials could also be considered as some of the materials management practices that could enhance productivity in building projects. Jergeas (2009) recommended that dedicated follow-up of the materials procurement activities is essential to improve productivity since the practice can reduce the workers' idleness. Nasir (2013) concluded that forming materials procurement team; keeping track of the quantity of the utilised materials and identifying the remaining materials; and preservation of the received materials on construction sites are the practices that have the potential to improve productivity in infrastructure projects. Furthermore, Caldas et al. (2014) identified that materials status database and preservation of construction materials are the best practices for increasing productivity in industrial construction projects in North America. In the context of Victorian multi-storey building construction projects, as the majority of the building materials are imported from overseas, the assignment of materials procurement team could be essential to follow up the procurement process. Furthermore, the preservation and maintenance of the construction materials on multi-storey building construction sites in Victoria could also be essential. If the materials are not maintained and damaged on site, it could take a long time to procure and replace the damaged materials. As a result, the productivity of the project could be affected.

On-site Materials Tracking Technology

The use of materials tracking technology on multi-storey building construction sites could also be one of the potential areas where productivity gain could be obtained. Grau et al. (2009) assessed the impact of materials tracking technologies on crafts' productivity and quantified the impact of automating, identifying and localising engineered components on productivity in industrial projects. The results of the study indicated that materials tracking technologies can significantly improve productivity. Nasir et al. (2010) confirmed that the implementation of on-site material tracking technology on industrial projects such as refineries and power plants can improve productivity. Similarly, the implementation of materials tracking technology has been confirmed as one of the best practices that enhance productivity in infrastructure and

industrial projects (Nasir, 2013, Caldas et al., 2014). For multi-storey building projects in Victoria, the use of materials tracking technologies could also help contractors to identify where a particular material or its components are located on building sites. Consequently, the time spent by the workers to search the materials can be reduced, and productivity could be improved.

Materials Delivery Schedule

The preparation of materials delivery schedule could also be one of the materials management practices that could improve productivity in multi-storey building projects. Arditi (1985) mentioned that materials delivery plan and schedule are among the potential areas for productivity improvement in construction projects in the US. Bell and Stukhart (1987) identified materials planning as one of the most critical materials management systems (MMS). Nasir (2013) confirmed that materials delivery schedule is the best materials management practices for improving productivity in infrastructure projects. Caldas et al. (2014) also confirmed that materials delivery schedule is the best practice for enhancing productivity in industrial projects. Likewise, El-Gohary and Aziz (2014) suggested the importance of materials delivery plan in increasing the construction projects' productivity in Egypt. In the context of multi-storey building projects in Victoria, materials delivery schedule could also be an important practice to improve productivity. If the materials delivery schedule is prepared, and the materials are supplied to a site as per the schedule, disruptions to other schedules such as crane usage schedules can be minimised. For instance, in the absence of materials delivery schedule, many materials might be delivered to a building site at the same time; consequently, some materials might not be lifted to the required working level due to the disruption of the crane schedule. Furthermore, the project site can be overcrowded, and productivity could be influenced.

Materials Inspection Process and Materials Inspection Team

Developing materials inspection process and formation of materials inspection team for a certain multi-storey building project could improve its productivity. Makulsawatudom et al. (2004) suggested that careful inspection of construction materials could be one of the potential areas where productivity can be gained in the construction projects in Thailand. Arditi (1985) mentioned the significance of checking the product availability and standardisation as a potential for increasing construction projects' productivity.

Similarly, materials standardisation and checking the product availability by the designer are identified as factors which have positive impacts on construction projects' productivity in Indonesia (Arditi and Mochtar, 1996). El-Gohary and Aziz (2014) also recommended the preparation of painstaking documentations for materials specifications to reduce the adverse effects of factors affecting productivity. The preparation of materials inspection strategy and the formation of inspection teams for multi-storey building construction projects in Victoria might also improve productivity. Inspecting the critical materials during the manufacturing process as well as before they are supplied to a building site can reduce the project delays due to the rejection of non-conforming materials during the construction phase.

3.4 Conceptual Framework of Construction Equipment Management Practices and Productivity

Based on a review of previous studies, a preliminary conceptual framework of the construction equipment and tools management practices and productivity in multistorey building projects is developed (Figure 3-5). The framework will be modified based on the findings of Phase I of the study in which the relevant independent variables (construction equipment and tools management practices that improve productivity in multi-storey building projects in Victoria State, Australia) are identified.





Construction Equipment Procurement Plan

Construction equipment procurement plan could be one of the construction equipment management practices that could enhance the productivity of multi-storey building construction projects. According to Prasertrungruang and Hadikusumo (2007), good equipment acquisition strategy is one of the success factors in construction projects.

Stewart (2002b) recommended the preparation of equipment procurement plan to reduce productivity loss due to the shortage of equipment on construction projects. Stewart (2002a) suggested that when there are many projects in the industry and/or when there is high workload on a certain construction project, the construction equipment should be leased for a longer time to reduce the loss of productivity. Procurement procedures and plans are also found to be the best practices that have the potential to improve productivity in industrial projects (Caldas et al., 2014). Thus, construction equipment procurement planning might also be one of the practices that could increase productivity in multi-storey building projects in the context of Victoria as the practice might help to reduce stoppage of building construction works due to shortage of construction equipment in the state.

Construction Equipment Maintenance

Preventive maintenance of construction equipment could also be one of the equipment management practices that could enhance the productivity of multi-storey building construction projects. Wireman (2005) described that equipment maintenance has the potential to increase the profitability of contractors by increasing the availability of equipment and by reducing the number of unexpected breakdowns or service interruptions. According to the author, preventive maintenance activities such as lubrication program, routine inspections, and adjustments should be conducted to avoid potential problems that can be corrected before they happen. Arditi (1985) concluded that machinery maintenance is one of the possible areas where productivity gain could be obtained in the US construction projects. Similarly, Arditi and Mochtar (1996) identified maintenance of construction equipment as one of the potential areas where productivity improvement can be attained in the construction projects in Indonesia. For multi-storey building construction projects in Victoria, equipment such as cranes and hoists might need preventive maintenance as their stoppage could influence activities on a critical path whose delay might affect an overall projects' progress. Thus, implementation of this practice could improve productivity.

Construction Equipment Productivity Analysis

Analysing the performance of construction equipment could be another equipment management practice that could increase the productivity of multi-storey building construction projects. Crespo Márquez and Sánchez Herguedas (2004) described that

good record keeping during the operation of the equipment helps contractors to analyse the performance of the machinery and to take corrective actions. Nasir et al. (2015) found that construction equipment productivity analysis is one of the best practices that could improve productivity in infrastructure projects. Analysing the performances of various equipment and choosing the most suitable construction machinery for a certain multi-storey building project in Victoria could help contractors to reduce unnecessary cost and increase productivity.

Tools Tracking Technologies

The use of tracking technologies such as bar codes and radio frequency identification (RFID) could also have important applications in tools management. The technology can be used to identify the location of tools, the date when the item is checked out, and the person who took the tool. Morse (1990) recommended the tracking technology to reduce theft and to track the location of the tools. Goodrum et al. (2006) explained the significance of tools availability for improving productivity and developed a tool tracking and inventory system which is capable of storing operation and maintenance data using RFID tags. As the construction of multi-storey building projects might involve numerous tools, tracking technology could be necessary to enhance their productivity.

Tools Management Strategy and Tools Maintenance

Tools management strategy and tools maintenance could be other tools management practices that could enhance productivity in multi-storey building construction projects. Morse (1990) found that authorising the tool manager to purchase some tools is one of the efficient tool management systems. Similarly, tools and consumables management strategy are recommended as the potential area for productivity improvement in industrial projects (Caldas et al., 2014). Finally, conducting onsite tool maintenance is found to be one of the best practices for improving productivity in infrastructure projects (Nasir, 2013). The author suggested that pre-maintenance activities such as identification of the tools which need maintenance, planning the date when maintenance will be conducted, and assigning qualified personnel are essential practices. Efficient tools management systems could also be important for multi-storey building projects in Victoria as the damage or loss of the tools might influence the productivity of the projects.

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

3.5 Conceptual Framework of Management Practices Related to Construction Methods and Productivity

Project management methods are defined as a system of practices, techniques, procedures, and rules used by those who work in the discipline (PMI, 2013). In construction projects, the techniques of integration of different schedules; schedule controlling methods; mechanisms used in the preparation of site layout; project start-up and completion procedures; and investigation of suitable technologies could be some of the methods that could enhance productivity in multi-storey building construction projects. The preliminary conceptual framework of the management practices related to construction methods and productivity is shown in Figure 3-6. The framework will be validated before conducting Phase II of this research.



Figure 3-6 Preliminary Conceptual Framework of Management Practices Related to Construction Methods and Productivity

Integrated Schedule

The use of integrated schedule during construction period could be considered as a good practice to increase productivity in multi-storey building construction projects. Arditi and Mochtar (1996) suggested integration of management functions is one of the areas where productivity gain could be obtained. The study indicated that combination of cost and schedule control is one of the critical success factors in completing the project on time. Caldas et al. (2014) confirmed that integrated schedule is one of the best practices

for improving productivity in industrial projects. The integration of work schedule, materials procurement and delivery schedule, budget and other information required for the execution of multi-storey building projects in Victoria could be essential for enhancing their productivity. If all the required information is found on the same page or interconnected, then the project team can easily track and control the progress of the construction of a multi-storey building project.

Work Schedule Strategies

The practices of adopting different working hours strategies could also be another potential area for productivity improvement in multi-storey building projects. Hanna et al. (2008) found that shift schedule is effective as compared to overtime and overmanning in reducing the project duration. The authors opined that the use of shift schedule reduces physical fatigue and site congestion that could occur if overtime and overmanning practices are implemented. Scheduled overtime is also suggested as a practice that has a positive impact on productivity as it builds the morale of employees (CII, 2013a). According to the Environmental Protection Authority of Victoria (EPA), the normal working hours for most projects should be between 7:00 am and 6:00 pm on weekdays (EPA, 2016). Contractors in Victoria are also required to work as per the calendar prepared by the Construction, Forestry, Mining and Energy Union (CFMEU) which is typically 36 hours per week (CFMEU, 2016a). Given the working hours restrictions set by authorities such as Environmental Protection Authority and the CFMEU, adopting various working hours strategies such as scheduled overtime for critical and nearly critical tasks could be essential practices for delivering multi-storey building projects within the contract time.

Schedule Control

The use of appropriate project schedule controlling techniques for different activities is another area where multi-storey building projects' productivity gain could be obtained. The controlling techniques include the methods used to measure the work progress, to analyse the data, to report the status and to take corrective actions. Some of the techniques of measurement include units completed, incremental milestone, start/finish, and manager or supervisor judgment (Neil and Knudsen, 1990, Attalla, 1997). According to the authors, the units completed is recommended for tasks that can be measured easily such as concrete, steel, and excavation; the incremental milestone

method is suitable for an activity with different sub-activities, for example, three coats of cement sand plastering, that should be handled in sequence and each sub-activity is allocated certain percentage upon their completion; the start/finish technique is suitable for tasks that do not have clearly defined intermediate milestone and/or the time required for their execution is difficult to estimate, for instance, testing lift; and subjective judgment of the supervisor is appropriate technique when the task is minor in nature such as installing architectural trim. Some of the analysis techniques include the progress curve or S-Curve and time variance analysis (Attalla, 1997). In the context of multi-storey building projects in Victoria, delay in the progress of an activity on a critical path could affect the progress of other activities and lead to overall project delay. Thus, implementation of the suitable schedule controlling techniques might be an important practice to monitor the execution of the activities as per their schedules and to reduce delays.

Dynamic Site Layout Plan

The implementation of the practice dynamic site layout could also enhance productivity in multi-storey building projects. Dynamic site layout refers to a sequence of layouts each of which is used for a discrete time interval or for a certain project phase, and together covering the entire duration of a construction project (Tommelein and Zouein, 1993). Previous researchers developed a dynamic site layout models and indicated the significance of implementing the techniques to enhance productivity (Tommelein et al., 1992, Tommelein and Zouein, 1993, Elbeltagi et al., 2004). As most multi-storey building construction projects in the central business district of Melbourne, Australia, have restricted working spaces, the use of dynamic site layout plans could help contractors in reducing congestion on sites. By using the dynamic site layout plan some parts of a building, which is under construction, can be used as a store, office and other temporary facility allowing early start of the external works.

Machinery Positioning Strategy, Site Security Plan, and Traffic Control Plan

Construction machinery positioning strategy might also be an important practice that has the potential to enhance productivity in multi-storey building projects. Choi and Harris (1991) proposed a mathematical model for determining the most suitable tower crane location for building projects. Similarly, Zhang et al. (1999) developed a computer model to optimise the location of a group of tower cranes. Safe Work

Australia prepared a code of practice to assist contractors in preparing good traffic management plans by providing information about traffic signs, the distance between pedestrians and vehicles, and vehicles movement (Safe Work Australia, 2014c). Investigation of the most appropriate location for a crane could also be significant to increase productivity in multi-storey building projects in Victoria. If the crane is positioned wrongly, its relocation cost could be high, and productivity could be lost. Thus, developing a strategy to position the crane using different models might be an essential practice. The integration of the traffic control plan and site security plan with the site layout plan could also be significant to enhance the productivity of the multistorey building projects. Since most construction materials in Australia are imported from overseas, any loss or damage to them could incur the loss of productivity due to unavailability of the materials locally. Thus, site security plans might be an important practice to reduce theft and loss of materials. Traffic control plans could also be an essential practice as there are various local traffic regulations such as Road Safety (Traffic Management) Regulations 2009 that could suspend the construction of multistorey projects due to noncompliance with the regulation.

Project Start-up and Project Completion Plans, and New Technologies

Project start-up plan and project completion plan might also be some of the management practices that have the potential to enhance productivity in multi-storey building projects. Fangel (1984) recommended that to reduce delays related to project start-up, the project team should prepare a project management manual that contains the details of pre-commencement activities. The author also suggested project start-up meetings prior to commencing a construction project. Kerzner (2010) proposed project kick-off meetings as one of the best practices that should be included in the project start-up process. Nasir (2013) found that project start-up plan, project completion plan, and innovations and new technologies are some of the best practices to enhance productivity in infrastructure projects. In the context of multi-storey building projects in Victoria, since various subcontractors could finish their works at different times, the principal contractor. Thus, completion plans could be a noteworthy practice to facilitate the handover of a multi-storey building project to the client. New technologies might also be essential in multi-storey building construction projects. For instance, to detect

clashes among engineering services, the software could be used to develop 3D models which help to visualise the services lines.

3.6 Conceptual Framework of Pre-Construction Phase Management Practices and Productivity

The preliminary conceptual framework of productivity and preconstruction phase management practices is presented in Figure 3-7. The suitability of the independent variables (the preconstruction phase management practices) to improve the productivity of multi-storey building projects will be confirmed during Phase I of this research, and the framework will also be validated.



Figure 3-7 Preliminary Conceptual Framework of Preconstruction Phase Management Practices and Productivity

Short Interval Plan

The preparation of short interval plan could be one of the construction management practices that could enhance productivity in multi-storey building projects. According to Hamzeh et al. (2008), look ahead planning (short interval planning) denotes the production planning having a time frame which usually spans between two to six weeks in which activities are broken down into the level of operations. Hadavi and Krizek (1993) concluded that the implementation of short-interval goals which consists of a day to day activities of the craftsmen can lead to higher construction projects' productivity. Waly et al. (1999) described that the short interval plans can increase productivity by reducing the nonproductive hours due to waiting for the resources. The

authors have also developed a model that is used to generate the short interval schedules (SIS). Tucker (1986) recommended that planning is the most important practice to improve construction productivity. Doloi (2008) found that increasing the level of preplanning and programming is the most critical factor to increase the productivity of construction projects. Huang et al. (2009) mentioned planning as an important management practice that could positively affect productivity. For multi-storey building construction projects in Victoria, the preparation of the short interval plan could also enhance their productivity since the plan can help the contractors' project managers to clearly identify the day to day activities including the resources required for the execution of the activities. The construction materials delivery schedule and crane utilisation schedule can also be prepared based on the short interval construction work schedule, and the loss of productivity due to waiting for the construction materials could be minimised.

Dedicated Planner

The assignment of a dedicated planner for a certain multi-storey building project could also be one of the practices that could enhance its productivity. COAA (2013b) recommended the assignment of a dedicated planner for the increment of on-site productivity since the dedicated planner can prepare detailed installation work plans which are subjected to minor revisions during the implementation phase. CII (2013b) identified the employment of committed planners as the potential areas for improving the productivity of infrastructure projects. The preparation of master work schedule and short interval work schedules for the construction of multi-storey building projects in Victoria by dedicated planners could also be essential to increase productivity. The planners might have sufficient data and experience that enable them to prepare schedules which need minor revisions.

Construction Work Packages

The preparation of manageable construction work packages (CWP) could be one of the preconstruction phase management practices that have the potential to improve the productivity of multi-storey building projects. According to the Construction Owners Association of Alberta (COAA), construction work package is defined as 'an executable construction deliverable that defines in detail a specific scope of work and includes a budget and schedule that can be compared with actual performance' (COAA, 2013a).

The Construction Industry Institute (CII) identified that the preparation of manageable construction work packages is one of the best practices for enhancing productivity in industrial and infrastructure projects (CII, 2013a, CII, 2013b). Nasir (2013) also verified that the implementation of the practice construction work package can enhance productivity in infrastructure projects in North America. In Australia, since the 'managing contractor' is the commonly practised delivery model in multi-storey building projects (Austroads, 2014), the principal contractors need to prepare a manageable and complete work package by considering the capacity of the local subcontractors. Thus, the implementation of the practice 'construction work packages' could be essential to enhance the productivity of the multi-storey building projects.

Well-Defined Scope of Works

Well-defined scope of works could also be one of the practices that have the potential to increase the productivity of multi-storey building projects. Defining the scope refers to the development of a common understanding as to what to include or exclude from a certain work package (Turbit, 2005). PMBoK (2000) recommended that the project scope needs to include all the necessary works and only the essential works to complete the project successfully. Jergeas (2009) found that clearly defining the scope of works in oil and gas projects in Canada is a good practice to enhance the performance of the projects. Liao (2008) concluded that industrial projects with better scope definition also have better construction productivity. For multi-storey building construction projects in Victoria, defining the scope of works could improve productivity since the practice can reduce the interference among the craftspersons or subcontractors. If the scope of works is well-defined, the conflict between the crews (subcontractors) that execute successive tasks can be minimised, and those who carried out the defective works or missed the execution of a certain activity can easily be identified and corrective actions can be taken accordingly.

Contrat Types

Selection of appropriate construction contract types could also be one of the preconstruction phase management practices that could enhance productivity in multistorey building projects. Contract types such as a lump sum, unit price, and cost-plus denote the agreement of how the subcontractor is paid for the work it has executed (Gordon, 1994). Jergeas (2009) indicated that the contractual strategy adopted for a

certain construction project is one of the key factors for successful completion of the project. The study suggested that the use of lump sum contracts as good practice for productivity. Arditi and Mochtar (1996) found that construction management contracting (CM) is the preferred delivery method to increase productivity in Indonesian construction projects. Lavender (2014) suggested that 'construction management' is the delivery method that is preferred for completing projects early, and the method allows the involvement of a client. The choice of suitable contract types by the contractors operating in Victoria could assist them to enhance the productivity of their multi-storey building construction projects. For instance, if the main contract (the contract between the client and principal contractor) is lump sum type, the principal contractor could also choose the lump sum subcontract type to share the risks (cost overruns and project delays) with the respective subcontractors. Furthermore, if the main contract allows the principal contractor to carry out design revisions or minor amendments, including the statements which authorise the subcontractors to make minor design modifications in the subcontract agreements could be essential to reduce the loss of productivity related to the designs.

Use of Software in Planning Workpackages and Model Development

The use of software in preparing the construction work packages and the development of 3D models could be some of the potential areas where productivity of the multistorey building projects could be increased. Kang et al. (2008) and O'Connor and Yang (2004) concluded that there is a strong correlation between schedule performance and the use of IT. Rojas and Aramvareekul (2003) recommended the enhanced use of IT systems in construction projects as one of the potential areas for productivity improvement. Arditi and Mochtar (1996) identified the application of computer in scheduling as the practice which could increase productivity. Thomas et al. (2001) concluded that as the implementation of three-dimensional (3D) modelling using computer-aided design (CAD) increases, the schedule performances of the projects also increase. Caldas et al. (2014) identified the 'utilisation of software in generating work packages' and 'project model development' as the best practices that improve productivity in industrial projects. For multi-storey building construction projects in Victoria, the use of software in planning can help to reduce the human errors in breaking down of the works and in defining the activities as well as their sequences

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

appropriately. Thus, productivity could be enhanced as the interference between the subcontractors or construction crews decreases due to the clarity of the sequences of the activities. Furthermore, the use of 3D models in multi-storey building projects, which involve the intertwining of different engineering services such as mechanical, electrical and planning, could help the project team to reduce the overlapping of the services and to improve on-site productivity.

Buildability Review

Conducting buildability review before the implementation of the building designs could also be important practice to enhance the productivity of the multi-storey building projects. Poh and Chen (1998) confirmed the positive relationship between productivity and buildability. The authors investigated the buildability of 37 building projects in Singapore and found that projects with higher buildability reviews also have higher productivity. El-Gohary and Aziz (2014) suggested that increasing the buildability level of design is the first step in improving productivity in Construction projects in Egypt. Jergeas (2009) mentioned that conducting the buildability reviews on a regular basis throughout the project phases can increase the project performance. Therefore, the implementation of the practice buildability review in multi-storey building projects in Victoria could also improve the projects' productivity. The involvement of construction team in the design process could help to reduce the loss of productivity due to the design problems.

Utilities Alignment and Regulatory Requirement

Prior identification and adjustment of the utility lines such as water, electricity, gas and telecommunication, and the identification of the regulatory requirements can be some of the practices that could increase productivity in multi-storey building projects. Nasir (2013) and CII (2013b) confirmed that the adjustment of utility lines and the identification of the regulatory requirements are some of the potential areas for productivity improvement in infrastructure projects. In the context of Victoria, since contractors are required to obtain various permits, the prior identification of the regulatory requirements could help them to reduce the loss of productivity associated with the suspension of the projects. For instance, the contractors working in the City of Melbourne should get their construction management plan (CMP) approved. Also, they need to get many permits such as a permit to erect a gantry, construction zone permit,

working outside prescribed hours permit and others (City of Melbourne, 2005). Thus, early identification of the type of permits and getting approvals from the concerned authority could help to reduce the chance of stopping the construction activities due to the non-compliance with the local regulatory requirements. Similarly, as the damage to the public utilities during excavation works could result in the suspension of the construction of projects, identification and protection of the existing utility lines before commencing the construction works could be an essential practice.

3.7 Conceptual Framework of Human Resource Management Practices and Productivity

In Figure 3-8, the hypothetical conceptual framework of productivity and the human resource management (HRM) practices is presented. The suitability of the proposed HRM practices in the context of multi-storey building projects in Victoria will be verified after collecting and analysing the qualitative data.

Financial Incentive Programs

The amount and timeliness of remuneration, as well as the incentive schemes, could be one of the human resource management practices that could enhance productivity in multi-storey building projects. Maloney (1983) identified that sharing the profit obtained from a project is one of the effective programs for construction productivity improvement. Kazaz and Ulubeyli (2007) concluded that incentive payments, the adequacy of the workers' payment as compared to others who are working on similar projects and making payment on time are the most important factors that increase workers' motivation. Dai et al. (2009a) mentioned the importance of equal pay in the same geographical area as one of the factors influencing productivity. Furthermore, Jergeas (2009) investigated the importance of sharing the incentive or bonus among the tradesmen, foremen and management staff for increasing project performance. Therefore, motivating the construction workers involved in multi-storey building construction projects in Victoria by providing financial incentives could improve the projects' productivity since the motivated workers could exert high levels of efforts which help them to achieve a higher level of productivity.

CHAPTER 3

Conceptual Framework



Figure 3-8 Preliminary Conceptual Framework of Human Resource Management Practices and Productivity

Non-financial Incentive Programs

Besides the financial incentives, provision of non-financial incentives to construction workers could also be an essential practice that could enhance the productivity of multistorey building construction projects. Luthans (2000) concluded that recognition is one of the most effective leadership tools that improve the performance of employees. Graham and Unruh (1990) cited in Luthans (2000), suggested the mechanisms for operationalization of the non-financial reward systems include congratulating an employee for a job well done; writing a personal note and publicly recognising an employee for good performance; and conducting morale-building meetings to celebrate successes. Fagbenle et al. (2004) compared the performances of construction projects that have implemented non-financial incentives with projects that have not implemented the incentive schemes and found that the productive time of bricklayers who received non-financial incentive is improved significantly. For multi-storey building construction workers in Victoria, providing non-financial incentives such as job recognition and appreciation by the contractors could also play a vital role to build the morale of employees and to achieve better performance.

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

Career Development Plan and Social Activities

Organising social activities and preparing the career development plans for construction workers can be some of the human resource management practices that could enhance productivity in multi-storey building construction projects. Hewage et al. (2011) concluded that the construction productivity cannot be improved by not only working hard and fast but also by developing the social skills of workers. Kazaz and Ulubeyli (2007) identified social activity as one of the socio-psychological factors that influence construction projects' productivity in Turkey. Tabassi et al. (2012) described that a well-organized human resource development program is an essential strategy for construction firms as human capital play a significant role in the success of an organisation. In the context of multi-storey building construction projects as these events can help to increase the team spirit among the crew members. Similarly, the career development plans could motivate employees and enhance their performance.

Employees Training

Employees training is also one of the human resource management practices that could increase productivity in multi-storey building construction projects. Wang et al. (2010) found that training craftsmen can increase productivity by 5%, decrease absenteeism by 2.5% and reduce turnover by 10%. Allmon et al. (2000) suggested that training employees to be multi-skilled workers can enhance productivity in construction projects. Rojas and Aramvareekul (2003) found that activity-specific training is the second major productivity driver in the manpower category. The authors recommended that before commencing execution of any task, formal education and training or sitespecific training should be provided to the construction workers. Jergeas (2009) recommended that training the low-level supervisors and site staff as well as investing in apprentice training programs are essential practices that should be implemented to enhance project performance. By the same token, various studies identified employees training as an essential practice for productivity (Arditi, 1985, Arditi and Mochtar, 1996, Hewage et al., 2011, Tabassi et al., 2012). Therefore, training construction workers involved in multi-storey building construction projects in Victoria can also be one of the practices that could enhance the productivity of the projects. The training can help the employees to execute good quality works which might not be rejected by their

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

supervisors. Consequently, reworks can be minimised and the time spent to rectify the defects can be saved.

Crew Composition, Organisational Structure and Clear Delegation of Responsibilities

Crew formation, maintaining the stability of the organisational structure and clearly specifying the duties and responsibilities of construction workers can also be some of the human resource management practices that could increase productivity in multistorey building construction projects. Maloney (1983) mentioned that team building practice is one of the approaches which enhance the forces acting to increase the productivity of labour. Liberda et al. (2003) also identified crew composition and team spirit as the most significant practices under human and management category. Caldas et al. (2014) identified the stability of organisational structure as the best practice that has a potential to increase the productivity of industrial projects. Likewise, CII (2013b) recommended that to keep the organisational structure stable, changes in key personnel should be avoided; plans on how to incorporate any unusual staff changes have to be prepared; and clauses which prohibit or reduce the replacement of key personnel should be included in the contract document. Nasir (2013) identified clarifying the delegation of the responsibilities of personnel as one of the best practices for enhancing productivity in infrastructure projects. In the context of Victorian multi-storey building construction projects, the formation of construction crews that have different skill levels, clarification of the employees' roles and maintaining the stability of the projects' organisational structure could also help to enhance the projects' productivity. These practices can assist the project managers to have crews having necessary skills and experience. The practices could also help workers to know the persons who are responsible for resolving specific problems that hamper their performances. Additionally, the knowledge and skills obtained from the particular project could be retained.

Exit Interviews and Skill Assessment

Conducting exit interviews and skill assessment could be some of the practices that could enhance productivity in multi-storey building projects. Kazaz and Ulubeyli (2007) described that conducting regular interviews and understanding the problems of the construction workers are some of the socio-psychological factors that influence the productivity of construction projects. According to CII (2013a) and CII (2013b), exit

interviews, and skill assessment and evaluation are the best human resource practices for increasing productivity in infrastructure and industrial projects. Dai et al. (2009a) mentioned the importance of conducting construction workers' performance reviews and taking corrective actions to enhance the productivity of construction projects. Moreover, Nasir (2013) found that doing skill assessment and evaluation by checking employees' previous experience and certificates before commencing the execution of any task can help the project manager to form better crews. In multi-storey building construction projects in Victoria, conducting exit interviews could assist contractors to get information regarding the problems that hamper the productivity of their projects. Employees who quit their jobs could provide reasons which demotivate them. Based on the identified problems, the contractors can take corrective actions which help to enhance the productivity of the projects.

Retention Plan for Experienced Personnel

Retention plan for experienced personnel can also be an important practice to increase the productivity of multi-storey building projects. Employee retention is a process of encouraging employees to stay with the organisation for maximum possible time (Narang, 2013). According to Hong et al. (2012), effective human resource management practices such as employee empowerment, training and development, appraisal system and compensation are the principal factors for employee retention. Kakar et al. (2015) confirmed that employee retention is positively associated with employee empowerment, training and development; and compensation. In multi-storey building construction projects in Victoria, the practice 'retention of experienced employees' could be important as it helps to retain the knowledge and skills gained during the construction of the specific projects. The knowledge, in turn, helps contractors to be more competitive and productive.

3.8 Conceptual Framework of Safety and Health Practices and Productivity

The preliminary conceptual framework of productivity and safety and health practices is shown in Figure 3-9. Eight independent variables (safety and health practices) which will be verified during Phase I of this research are proposed based the review of the literature.

Safety and Health Policy

Preparation of safety and health policy for multi-storey building construction projects could be important practice to enhance their productivity. According to Worksafe Tasmania, a safety and health policy is a written statement which shows the commitment of a construction company's management and workers to reduce the risks to the safety and health of construction workers, visitors, and others who might be affected during the construction process (Worksafe Tasmania, 2016). Sawacha et al. (1999) found that organisational policy on safety is the most dominant factor that influences the safety performance of construction projects in the UK. The authors recommended that contractors employing five or more personnel must have a safety policy. Jergeas (2009) concluded that ensuring safety program at the project site is crucial in providing a safe work environment. Furthermore, the preparation of formal health and safety policy at a project level is identified as a good safety practice that could increase productivity in construction projects (CII, 2013b). Therefore, preparing safety and health policies for multi-storey building construction projects in Victoria could help to reduce the likelihood of the occurrence of accidents which negatively affect the productivity of the projects.



Figure 3-9 Preliminary Conceptual Framework of Safety and Health Practices and Productivity *Safety and Health Plan*

Preparation of safety and health plan to reduce the number of accidents on multi-storey building construction projects could also be one of the safety and health practices that could enhance their productivity. According to Safe Work Australia (2011), a

workplace health and safety (WHS) plan is 'a written plan that sets out the arrangements for managing the site health and safety matters.' The plan includes project-specific safety and health rules and procedures to be followed in receiving, assessing, monitoring and reviewing the safe work method statements (Safe Work Australia, 2011). CII (2013a) found that health and safety planning techniques such as pre-task planning for safety, safety orientation and training, safety incentive programs, alcohol and substance abuse programs, and accident investigations are the best practices that have the potential to improve productivity in industrial projects. Nasir (2013) and CII (2013b) confirmed that zero accident techniques are among the best safety and health practices which could increase productivity in infrastructure projects. In the context of multi-storey building construction projects in Victoria, preparation of project-specific safety and health plan which outlines procedures and techniques to be used in managing safety at the construction sites could help contractors to reduce the occurrence of accidents and to increase the productivity of the workers.

Safe Work Method Statement and Hazard Analysis

The preparation of the safe work method statement (SWMS) for high-risk construction activities and conducting hazard analyses for activities with no SWMS could also be some of the practices that could improve the productivity of multi-storey building construction projects. Hazard is a 'situation or thing that has a potential to harm a person', and hazard analysis involves identification of the hazards, assessing the risks and devising the controlling mechanisms (Safe Work Australia, 2011). Accordingly, the use of ladders, incorrectly erected equipment, unguarded holes, falling objects, welding fumes, the presence of asbestos, and hazardous manual tasks are some examples of hazards. Caldas et al. (2014) found that task safety analysis is the practice that has the potential to enhance productivity in industrial projects. Nasir et al. (2015) confirmed that hazard analysis is one of the safety and health practices that can improve productivity in infrastructure projects. For multi-storey building projects in Victoria, analysis of the hazard associated with both high-risk and low-risk construction activities might lead to enhanced productivity since the analysis could minimise the likelihood of the occurrence of accidents on the construction sites. Furthermore, the occupational health and safety regulations in Australia oblige contractors to prepare the Safe Work Method Statement (SWMS). For instance, Work and Health Safety Regulations 2011

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

Part 6.3 Article 299 and 300 stipulate the obligations of contractors to prepare SWMS. The regulation also specifies the associated penalty of up to 30,000 AUD for the noncompliance. Thus, preparation of SWMS could help contractors to reduce the suspension of their projects and to avoid the penalty.

Safety and Health Training, and Safety Meetings

Providing safety and health training and conducting toolbox safety meetings could be some of the essential practices that have the potential to increase the productivity of the multi-storey building construction projects. Jergeas (2009) concluded that safety training can play a significant role to improve productivity in oil and gas projects in Canada. Dai et al. (2009b) also mentioned the impact of health and safety training on productivity.Toolbox safety meeting is found to be one of the best practices that can positively influence productivity in industrial and infrastructure construction projects (CII, 2013b, CII, 2013a). In the context of Victoria, to ensure the safety of construction workers, WorkSafe Victoria obliges contractors to make sure that the safety training is provided to all employees involved in construction projects (Worksafe Victoria, 2016). Thus, to comply with the regulatory requirements and to reduce the rate of accidents in multi-storey building construction projects, providing safety training as well as conducting regular safety meetings could be essential.

Housekeeping and Alcohol Testing Program

Regular housekeeping, and alcohol and substance abuse testing programs can also be some of the safety and health practices that could enhance productivity in multi-storey building projects. According to CII (2013a), housekeeping involves scheduling a suitable time either weekly or bi-weekly to ensure that the workface is organised and all materials, tools, and equipment are properly stored, cleaned up and disposed of the site. Jergeas (2009) found that neat and clean work environment is essential to ensure safety and to increase the productivity of construction projects. Hinze and Wilson (2000) confirmed that conducting alcohol and substance abuse programs is one of the mechanisms to achieve zero accidents in construction projects. The study indicated that implementing good safety practices improves safety performances which are measured in terms of lost time injury rates. According to Standards Australia (1990), lost time injury is any occurrence which results in a permanent disability and/ or time lost from work by one day or a shift or more as well as a fatality. The lost time injury frequency

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

rates (LTIFR) are the number of occurrences of injury in a specified period for one million hours worked (Standards Australia, 1990). In other words, the LTIFR is computed by dividing the number of occurrences in a given period, say a year, by the number of hours worked in that specific period. Since the ratio could be small, the output is multiplied by one million (Standards Australia, 1990). In the construction of multi-storey building projects in Victoria, the implementation of the practice housekeeping could help to keep the construction sites clear and reduce the rate of accidents which negatively influence the productivity of the projects. Furthermore, alcohol and drugs testing programs can assist to minimise the construction workers' injuries. If employees execute construction activities after consuming alcohols and drugs, the rate of accidents on project sites could increase.

3.9 Summary

Six preliminary conceptual frameworks of the construction management practices that have the potential to enhance productivity in multi-storey building construction projects have been developed based on the growth accounting framework and findings of the previous studies. The conceptual frameworks will be validated after conducting Phase I of this research. The materials management practices' framework relates nine independent variables (the materials management practices) to the productivity of multistorey building projects. The conceptual framework of the construction equipment practices links six independent variables (the equipment and tools management practices) and one dependent variable (productivity).

The framework for management practices related to construction methods relates ten independent variables (the management practices related to construction methods) and one dependent variable (productivity). Similarly, ten independent variables (pre-construction phase management practices) and one dependent variable (productivity) are linked in the conceptual framework of the pre-construction phase management practices. The human resource management framework relates eleven independent variables (the human resource management practices) and one dependent variable (productivity). Finally, eight safety and health practices (the independent variables) are related to the productivity (dependent variable) of multi-storey building construction projects using the preliminary conceptual framework of safety and health practices. Overall, 54 potential management practices that could enhance productivity in multi-

storey building projects in Victoria State, Australia, are related to productivity (Figure 3-10).



The next chapter presents the research design and methodology adopted to achieve the objectives of this study.

CHAPTER FOUR

4 RESEARCH DESIGN AND METHODOLOGY

4.1 Introduction

This chapter presents the research design and methodology adopted to achieve the objectives of the study. Section 4.2 describes the research design and presents the flow chart of the research process. Section 4.3 discusses the research approach which includes the philosophical assumptions of the study and the mixed methods research approach. In Section 4.4, the research strategy and activities, as well as the associated inputs and outputs, are described. The two phases of the research and the corresponding research questions and objectives are also explained in Section 4.4. Section 4.5 discusses the data analysis strategy used in Phase I and Phase II of the study. The qualitative data analysis methods, the units of analysis and units of observation for the second phase of the research, and the descriptive and inferential statistical analysis techniques such as correlation, linear and logistic regression analyses are described in Section 4.5. Finally, the summary of the chapter is presented in Section 4.6.

4.2 Research Design

The research design is a systematic and a logical plan prepared for directing a research study, and it stipulates the objectives of the study, the methodology, and techniques to be adopted for achieving the objectives (Krishnaswami and Satyaprasad, 2010). Accordingly, it forms a blueprint for the measurement, collection, and analysis of data. It is the plan, structure, and strategy of the investigation conceived so as to obtain answers to research questions. This study uses exploratory sequential mixed methods research design which involves the combination of qualitative and quantitative data in two phases. Amaratunga et al. (2002) explained the importance of mixed methods research design in built environment studies as it focuses on the strengths of both qualitative and quantitative methods. The schematic diagram of the research activities that are designed to answer the research questions is indicated in Figure 4-1.

This research commenced with a review of the literature to provide information regarding the context of the study and to identify the knowledge gap. After the preliminary literature review, the research problem was stated. Following the problem identification, an extensive literature review was conducted to identify the construction management

practices that have the potential to improve productivity in construction projects, and preliminary conceptual framework was developed. The first phase of the research was then begun by conducting face-to-face in-depth interviews with professionals who have experience in delivering multi-storey building construction projects in Victoria State, Australia. The qualitative data analysis reached a saturation point, the management practices that are suitable to the local context were identified, the instrument used to measure the practices was verified, and the preliminary conceptual framework was validated during Phase I.



Based on the findings of Phase I, Phase II of the research was conducted. During this stage, quantitative data were collected using interview questionnaire survey whereby the survey was self-administered; the quantitative data were analysed; the management practices were prioritised; the scoring tools and logistic regression models were developed; and the probability graphs were plotted. Detailed descriptions of the methods of collecting and analysing data for Phase I and Phase II are presented in Section 4.4 and 4.5.

4.3 Research Approach

4.3.1 Philosophical Assumptions

This research uses the mixed methods research approach which is based on pragmatism worldview. There are four research paradigms that may be used in any research: postpositivism, constructivism, transformative, and pragmatism (Creswell, 2013). Postpositivists reflect a deterministic philosophy about research in which causes probably determine effects or outcomes. The problems studied by post-positivists reflect issues that need to identify and assess the causes that influence the outcomes. Social constructivists hold the assumption that individuals seek understanding of the world in which they live and work. Individuals develop subjective meanings of their experiences, meanings directed toward certain objects. Transformative worldview holds that research inquiry needs to be intertwined with politics and a political change agenda to confront social oppression and social justice at whatever levels it occurs (Mertens, 2010).

Pragmatism as a philosophy arises out of actions, situations, and consequences. In this paradigm, there is a concern with applications that work in certain situation and solutions to problems. Thus, instead of focusing on methods, researchers emphasise the research problem and use all approaches available to understand it (Creswell, 2013). Talisse and Aikin (2008) described that pragmatists are driven by the desire to achieve their ends; and they have little interest in idealisation, nit-picking argument, abstraction, or theory of any sort. Pragmatism worldview is considered most suitable for this research because the study focuses on the problems leading to the loss of productivity in multi-storey building construction projects, and it explores appropriate solutions or practices that could improve the productivity. Management practices vary according to project types and contexts and the solutions in one context might not be effective in

another. Thus, this study explored context-specific construction management practices as some of the solutions to the productivity problems.

There are three major research approaches that could be used in any research: qualitative, quantitative, and mixed methods (Creswell, 2013). The quantitative research approach is more suitable for the positivism paradigm, and qualitative research approach is adopted for the constructivism paradigm (Rose et al., 2014). The research approach suitable for pragmatism worldview is mixed methods research (Creswell, 2013). Accordingly, this research used mixed methods research approach.

4.3.2 Mixed Methods Research Approach

Mixed research designs have been classified into three categories: convergent parallel, explanatory sequential and exploratory sequential (Creswell, 2013). In convergent parallel mixed methods, the researcher collects both qualitative and quantitative data, analyses them separately, and then compares the results to see if the findings confirm or disconfirm each other. Explanatory sequential mixed methods involve a two-phase project in which the researcher collects the quantitative data in the first phase, analyses the results, and then uses the results to plan or build on to the second qualitative phase.

In exploratory sequential mixed methods, the researcher first begins by exploring with qualitative data and then uses the findings in a second quantitative phase. The qualitative phase may be used to build an instrument that best fits the sample under study, to identify appropriate instruments to be used in the follow-up quantitative phase or to specify variables that need to go into a follow-up quantitative study (Creswell, 2013). In this research, exploratory sequential mixed methods is used because construction management practices identified by previous studies might not be applicable to the Victorian construction industry, and there could be other practices that are specific to the local industry. Thus, exploratory study was conducted first by collecting and analysing qualitative data obtained from interviews during the first phase. The preliminary conceptual framework developed based on literature review was modified on the basis of the qualitative data. Furthermore, the instrument used to measure construction management practices was verified. To investigate the relationship between productivity and the practices, to prioritise the practices, and to develop the scoring tools and probability-based predictive models, quantitative data were collected and analysed during Phase II of the research.

4.4 Research Strategy and Activities

The two phases of the data collection and analysis, as well as the detailed descriptions of the activities, are illustrated in Figure 4-2. The research questions, research objectives, inputs, activities, and outputs of Phase I and Phase II of the study are shown in the figure.

4.4.1 Qualitative Data Collection - Phase I

Phase I of the research addressed the first research question, which states "RQ1: what are the management practices that have the potential to improve productivity in multi-storey building construction projects in the context of Victoria State, Australia?" To answer this research question, part of Objective 1 or "to identify the management practices that have the potential to enhance productivity in multi-storey building construction projects in the context of Victoria State, Australia" is devised (Objective 1a in Figure 4-2). During Phase I, part of the third research question or "RQ3: how can the management practices that have the potential to improve productivity in multi-storey building construction projects be measured, planned and controlled?" is also addressed. Accordingly, part of the second objective or "to refine the scoring tools for measuring, planning, monitoring and evaluating the management practices that could improve productivity in multi-storey building construction projects" is formulated (Objective 2a in Figure 4-2).

The qualitative data is typically collected through observing the behaviour or interviewing the participants (Creswell, 2013). However, observation is not suitable to collect the data regarding the management practices since they cannot be observed. Hence, interview was used as a method to collect the qualitative data in the context of this research. Nineteen professionals who have experience in delivering multi-storey building projects in Victoria were interviewed. They have a five to forty years work experience and have been working as general manager, construction manager, project manager, project coordinator, project engineer, site engineer, contract administrator, supervisor and cost manager (Table 4-1).

CHAPTER 4

Research Design and Methodology



Figure 4-2 Research Strategy and Activities

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

Experts	Number	Experience (years)
General Manager	1	29
Construction Manager	1	26
Project Managers	6	10-40
Project Coordinators	2	6,8
Contract Administrators	2	5,10
Project Engineer	1	12
Project Planner	1	10
Site Engineers	2	7
Supervisors	2	16,25
Cost Manager	1	6
Total	19	

Table 4-1 Profiles of the Interview Participants

The experts were selected based on their experience in working for subcontractors and principal contractors that are involved in the construction of multi-storey building projects in Victoria State, Australia. A snowballing technique was used to select the participants for the interviews. In this technique, some experienced building construction experts were contacted first, and the researcher asked them to nominate other experienced professionals who can participate in the interviews. The semistructured interview questions comprise lists of construction management practices that have the potential to improve productivity in construction projects. For each proposed practice, the questions include: Does this practice exist? How is it practised by local contractors? Does this practice improve productivity in multi-storey building construction projects? What other practices enhance productivity in multi-storey building construction projects? How do you measure productivity in multi-storey building projects? Each interview lasted for an average of 1.5 hours. The interviews were conducted until the data analysis reached a saturation point. Saturation refers to the point where similar reasons for accepting or rejecting a particular practice were given by the participants.

4.4.2 Quantitative Data Collection - Phase II

Phase II addressed the second research question, part of the third research question and the fourth research question. The second research question states "RQ2: which management practices are the most important to enhance productivity in multi-storey building construction projects?" To answer this research question, part of Objective 1 or "to prioritise the management practices that have the potential to enhance productivity in multi-storey building construction projects in the context of Victoria State, Australia" is prepared (Objective 1b in Figure 4-2). Part of Objective 2 or "to validate the scoring tools

for measuring, planning, monitoring and evaluating the management practices that could improve productivity in multi-storey building construction projects (Objective 2b in Figure 4-2)" is developed to answer the third research question. The fourth research question which is "RQ4: what tool can be used for predicting the probability of exceeding baseline productivity factor based on a score of the management practices?" is answered by formulating Objective 3 or "to build and validate a tool that can be used to predict the probability of exceeding baseline productivity factor baseline produc

Survey data can be collected using mail, telephone, the internet, personal interview or group administration, and the survey can be self-administered, or an interviewer asks the questions and records the answers (Fowler Jr and Cosenza, 2009). In this research, the quantitative data were collected using interview questionnaire survey whereby the survey was self-administered. The respondents were asked the questions face-to-face and requested to write their answers. The approach can increase the response rate, and in this study it allowed the researcher to clarify issues during the interview survey.

The questionnaire consists of two parts (Appendix-1). In the first part, the respondents were asked to rate the relative importance of the construction management practices identified during Phase I. A response scale of 1 to 5 was used to achieve optimum reliability and validity (Jamieson, 2004, Lozano et al., 2008). In the rating, 1=not important; 2=slightly important, 3=somewhat important, 4=very important, and 5=extremely important practice to improve the productivity of multi-storey building projects. In the second part, the respondents were asked to provide information on a specific multi-storey building project which they have completed within last five years (2011-2016). The information includes project cost and project time among other things. The respondents were also asked to choose the level of implementation of the practices on the multi-storey building projects they nominated (Appendix 1 C). The levels of implementation of the practices were measured on the basis of a survey instrument developed as part of Phase I of the study (unweighted scoring tools in Figure 4-2). In preparing the measuring instrument, an existing instrument developed by Construction Industry Institute (CII) was used as a starting point (CII, 2013b, CII, 2013a). On that basis, a new tool was developed and validated through interviews with local experts during Phase I of the study. For instance, the survey tool (unweighted scoring tool) for the practice

'construction equipment procurement plan' is shown in Table 4-3, and the respondents were asked to choose the level of implementation of the specific practice. The validated survey tools (unweighted scoring tools) for all practices are indicated in the Appendix 1C.

Table 4-2 Sample of the Validated Survey Instrument (adapted from (CII, 2013b))

Please check ($$) one box per element in space provided in front of the level of implementation of the management practice. Please do not leave any element unchecked.			
() Procurement Plan for Construction Equipment			
Level A	A procurement plan for construction equipment is not applicable for this building project.		
Level B	Construction equipment procurement plan is not prepared for this building project.		
Level C	Construction equipment procurement strategies such as hiring or leasing are prepared for this building project.		
Level D	Continuation of Level C and there is a procedure for identifying reputation of potential equipment suppliers		
Level E	Continuation of Level D, plus plan identifies necessary equipment and onsite resources to support delivery.		
Level F	Continuation of Level E and construction equipment schedule is integrated with other project schedules.		

Before choosing the sample multi-storey building projects for Phase II of this research, a list and addresses of contractors were obtained from the registry of Victorian Department of Treasury and Finance (Department of Treasury and Finance, 2015). The Department maintains a list of contractors that are pre-qualified to provide different construction activities. They have been assessed against a range of management, technical and financial pre-qualification criteria to ensure they meet the high standards required of government suppliers. According to the Department's requirement, firms applying for pre-qualification must demonstrate that they have the management capability, qualifications, expertise, experience, technical, and financial capacity to deliver construction projects successfully. All the 39 principal contractors on the list having experience in constructing multi-storey building projects were selected. Multi-storey building projects that have been delivered by these contractors were identified from the companies' websites. The questionnaires were then distributed to the contractors. The persons directly responsible for the construction of the multi-storey building projects by these companies were contacted and were the respondents. They were construction directors, operation managers, construction managers, project managers, project coordinators, and site managers (Table 4-3). The letters explaining the objectives of the research and the benefits of participating in the research were sent to the respondents via email and in person. The researchers then sent follow-up emails and phone calls requesting the participation and setting suitable time and places for the interviews. The results of the study can be generalised because the sample

size of 39 represents all potential principal contractors that have the capability to execute multi-storey building projects. Moreover, the suitability of the survey data was checked using tests such as Events per Variable (EPV) and acceptable results were found (see logistic regression section for EPV). Thus, the sample size is adequate to enable a valid conclusion to be drawn.

Respondents	Number	Years of Experience
Construction Director	1	20
Operation Manager	1	21
Construction Managers	4	20-43
Project Managers	18	8-41
Site Managers	2	14, 30
Project Coordinators	4	7-10
Contract Administrators	7	6-27
Project Engineers	2	7,8
Total	39	

Table 4-3 Profiles of the Questionnaire Survey Participants

4.5 Data Analysis Strategy

This section provides the methods used in analysing the qualitative data in the first subsection (4.5.1) and techniques adopted in analysing the quantitative data in the subsequent subsections.

4.5.1 Qualitative Data Analysis - Phase I

The analysis of the qualitative data is conducted in three concurrent steps: data reduction, data display, and conclusion (Rose et al., 2015). Data reduction is a form of analysis that sharpens, sorts, focuses, discards, and organises data in a way that final conclusions can be drawn (Miles and Huberman, 1994). Writing summaries, coding, and making clusters are common methods used in data reduction process. The latter two methods are more suitable when the research is entirely inductive in nature. In this study, writing summary was used as a technique to reduce the transcribed interviews. The reduced data can be displayed using matrices, graphs, charts and networks. In this research, matrix technique was used as it is suitable to display the summaries of the responses in matrix boxes. Once the data is displayed, the conclusion is drawn by either noticing the patterns of similarities and differences between categories and/or processes, clustering, making contrasts and comparisons and noting relations between concepts (Rose et al., 2015). During the analysis, the audiotaped interviews were first transcribed and a matrix was prepared in
Excel spreadsheet to match the responses of an expert and the construction management practices (Table 4-4). A summary of each interview result was written in a matrix box, and a conclusion was drawn for each practice. Similar iterative procedures were used for all the interview results. The similarity between the successive summaries was observed to find saturation point. After analysing the outcome of the fifteenth interview, similar explanations for the construction management practices that have the potential to improve productivity in multi-storey building projects were observed. Although the saturation point was reached at the fifteenth interviewee, more interviews were conducted until the nineteenth participant for the sake of validating the saturation point.

Table 4-4 presents the summary of the transcribed text for the practice 'Construction Machinery Positioning Strategy' to illustrate how the summarised data was displayed. The summaries of the interview results for all the management practices are presented in the Qualitative Data Analysis and Findings chapter (Chapter 5).

	Respondents (R)							
Management	R ₁₈		R19	Overall				
Practices	Summary	Conclusion	Summary	Conclusion	Conclusion			
Construction Machinery Positioning Strategy	"We sit down and work out a site layout details. It is interrelated with many things such as traffic control plan, access points, the way materials could easily get into the site, the size or the footprint of the job, crane types and number requirements, the location of the cranes and other issues. For example, we locate cranes for the maximum flexibility; we place them to get as much coverage as we can. Thus, integrating and developing a strategy for positioning construction machinery is critical for productivity."	suitable	"For instance, one of the critical equipment on our site is a crane, and its location is planned by considering the weight to be loaded, the street, distance of placement and the nearby buildings. The crane needs to reach the street; it needs to reach the heaviest lift; it needs to reach the entire site. There might be another building that is taller than the building under construction, and the crane should not hit that building. All these factors are taken into account when positioning a crane and it is good practice to have a strategy for positioning construction equipment."	suitable	suitable			

Table 4-4 Template for the Qualitative Data Analysis

Finally, the output of the first phase was used as input to the second phase which comprised quantitative data collection and analysis. The construction management practices that were described as suitable to improve productivity in multi-storey building projects by the participants were included in the list for the industry-wide survey. Furthermore, the instrument used to measure the identified practices was verified.

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

4.5.2 Methods of Computing Relative Importance Indexes and Productivity Factors

Part one of the questionnaire was analysed to assign weights to the practices identified using the interviews (Phase I) and to rank them accordingly. Relative Importance Index (RII) and mean value can be used for ranking purpose. According to Lam et al. (2007), both methods produce similar rankings, but RII method is used to derive relative indices within the range of 0-1 which makes the relative comparisons of different variables easy. Holt (1997) explained that many researchers in construction management prefer RII as the relative comparison of variables whose indices less than or equal to one is easier to perceive. Doloi (2012) stated that the mean and standard deviations are not reliable statistics for assessing the overall ranking of the attributes, and the author used the relative importance weights as input for factor and regression analyses. This research adopted RII technique as it is suitable for ranking purpose, and it is recommended for inferential statistical analyses. The following equation (Equation 4-1) was used for RII computation (Enshassi et al., 2007, El-Gohary and Aziz, 2014). Moreover, Friedman's and Wilcoxon's tests were conducted to check the significance of the difference in the weights assigned to the practices. Finally, part of objective 1 or 'to prioritise the practices' was achieved.

$$RII = \frac{5(n_5) + 4(n_4) + 3(n_3) + 2(n_2) + n_1}{5(n_5 + n_4 + n_3 + n_2 + n_1)}$$
 (Equation 4-1)

The number of respondents who selected 1 for 'not important,' 2 for 'slightly important,' 3 for 'somewhat important,' 4 for 'very important,' and 5 for 'extremely important' practice to improve productivity in multi-storey building projects are shown as n_1 , n_2 , n_3 , n_4 , and n_5 respectively.

The second part of the questionnaire was analysed to compute the productivity factors of the 39 multi-storey building projects nominated by the 39 participating contractors. The construction projects' productivity can be expressed in either absolute or relative term. In absolute term, the units of measure of both output and input are shown in the productivity value. For example, productivity can be measured in tonnes of steel erected per work-hour or meters of pipe installed per work-hour or the project cost per hours worked. The relative measure or Productivity Factor (PF) refers to the ratio of the actual to planned productivity (Equation 4-4). The actual productivity is computed using project value as output and actual time as input (Equation 4-2). The planned

productivity is calculated using project value as output and planned time as input (Equation 4-3).

Actual productivity
$$(AP) = \frac{Output}{Actual Completion Date-Project Start Date}$$
 (Equation 4-2)
Planned productivity $(PP) = \frac{Output}{Planned Completion Date-Project Start Date}$ (Equation 4-3)
Productivity Factor $(PF) = \frac{Actual Productivity (AP)}{Planned Productivity (PP)}$ (Equation 4-4)

According to Alby (1994), Productivity Factor (PF) is a useful measure to compare the productivity of different construction projects. McDonald and Zack (2004) explained that measuring productivity using productivity factor is more significant in substantiating claims for the lost labour productivity. Nasir (2013) concluded that the use of productivity factor as a metric to measure productivity is more beneficial than adopting the absolute measure. According to the study, PF can be useful to measure the productivity of any project and to make a comparison with another project. Moreover, the weighted average of projects' productivity can be computed since PF is independent of the units of measurement. Caldas et al. (2014) validated the use of PF to measure productivity in industrial projects. In this research, PF is adopted as it is suitable to measure and compare the productivity of different multi-storey building construction projects.

4.5.3 Techniques of Preparing the Scoring Tools

Figure 4-3 shows the procedures used in preparing the 'validated and weighted scoring tools' which can be used for measuring, planning, monitoring and evaluating the management practices in the context of multi-storey building projects. First, the qualitative data was analysed, and practices which are suitable to improve productivity in the context of multi-storey building projects in Victoria were identified. Then, the levels of implementation for the identified practices were prepared based on the existing tools developed by the Construction Industry Institute (CII) for industrial and infrastructure projects. The practices which are not implemented by contractors in Victoria were deleted, and terminologies which are not suitable for building construction projects were excluded. Consequently, the 'unweighted scoring tools' were prepared and validated by conducting interviews with local experts during Phase I of the study.

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

The validated and unweighted scoring tools were used to collect the data regarding the implementation levels of the identified practices from the multi-storey building projects (please refer Part II of the questionnaire in Appendix-1 for the survey instrument). The data concerning the relative importance of the practices were collected from contractors' experts who are responsible for the delivery of the nominated multi-storey building projects (please refer Part I of the questionnaire in Appendix-1).

During Stage I of the quantitative data analysis, the weight computed for each practice was used in the preparation of the 'weighted scoring tool' (Figure 4-3). The weights were proportionally distributed among the five levels of the practices. For instance, the weight for the practice 'construction equipment procurement plan' is 0.69, and the proportions are Level A=0; Level B= $1/5 \times (0.69) = 0.14$; Level C= $2/5 \times (0.69) = 0.28$; Level D= $3/5 \times (0.69) = 0.41$; Level E= $4/5 \times (0.69) = 0.55$; and Level F= 0.69 (Table 4-5). Similarly, the weight proportions for other practices were computed and the 'weighted scoring tools' were developed. The tools are presented in Summary and Conclusion chapter (Chapter 10). For the sake of brevity, the term 'scoring tools' refers to the 'weighted scoring tools.' During Stage II of the quantitative data analysis, the data collected from the multi-storey building construction projects were transformed into equivalent scores using the weighted scoring tools. The analysis was then conducted to validate the weighted scoring tools.



Figure 4-3 Flow Chart for Scoring Tools Preparation and Validation

-		
	Planning / Implementation levels	Weights
Level A	A procurement plan for construction equipment is not applicable for this building project.	0
Level B	Construction equipment procurement plan is not prepared for this building project.	0.14
Level C	Construction equipment procurement strategies such as hiring or leasing are prepared for this building project.	0.28
Level D	Continuation of Level C and there is a procedure for identifying reputation of potential equipment suppliers	0.41
Level E	Continuation of Level D, plus plan identifies necessary equipment and onsite resources to support delivery.	0.55
Level F	Continuation of Level E and construction equipment schedule is integrated with other project schedules.	0.69

Table 4-5 Samp	ole Scoring	Tool for th	ne Practice	Procurement	Plan for	Construction	Equip	oment
1	0						1 1	

During validation of the scoring tools, the mean, mode, and median management practices' scores were computed to choose a baseline score for grouping the projects. If the data is not highly skewed, mean is the best measure of central tendency; otherwise, the median is the recommended value (Lurd and Lurd, 2013). According to Kim (2013), if Z-score for skewness (skewness/standard error) is less than 1.96, the distribution could be considered as normal. Projects with scores less than a central value (baseline) were classified under Group-1(low-score) and those having scores greater than the baseline were classified under Group-2 (high-score). The data that satisfied the assumptions of parametric tests such as normality were analysed using ANOVA, and the data that did not satisfy the parametric tests' assumption were analysed using Mann-Whitney U-test (non-parametric test). Finally, the difference between the two groups was investigated, and the scoring tools were validated (Chapter 7).

The internal consistency of the scoring tools was checked by running reliability test using SPSS-24. The test is essential to ensure the reliability of the dataset before conducting subsequent analyses such as correlation, linear and logistic regression analyses. Cronbach alpha (α) is the most popular reliability statistic which determines the consistency of items in a survey instrument and the recommended acceptable minimum value of the Cronbach alpha (α) is 0.70 (Santos, 1999). The results of the reliability and validity tests are presented in Chapter 7. Finally, Objective 2 of the study which is 'to refine and validate scoring tools for measuring, planning, monitoring and evaluating the management practices that could improve productivity in multi-storey building construction projects' is achieved.

4.5.4 Correlation Analysis

Correlation analyses were conducted to investigate the association between productivity and the construction management practices. After checking the assumptions of the parametric tests such as normality, the correlation analyses between productivity and the practices identified in Phase I and prioritised in Phase II of the study were conducted. Moreover, the interrelationships among the practices were investigated. For the sake of brevity, the diagram representing the interrelationship among all the practices is not shown here. The results of the analyses are presented in Chapter 7.

Figure 4-4 shows the correlation analysis among the practices categorised under 'Construction Materials Management Practice' category.



Figure 4-4 Diagram for Correlation Analyses of Construction Materials Management Practices Figure 4-5 indicates the correlation analysis conducted among the practices categorised



Figure 4-5 Diagram for Correlation Analyses of Construction Equipment Management Practices 96

In Figure 4-6, the correlation analyses for preconstruction phase management practices are shown, and the results of the analyses are presented in the quantitative data analysis and findings chapter.



Figure 4-6 Diagram for Correlation Analyses of Preconstruction Phase Management Practices Figure 4-7 shows the correlation analyses conducted between the practices grouped under the 'Human Resource Management Practices' category.



Figure 4-7 Diagram for Correlation Analyses of Human Resource Management Practices

97

In Figure 4-8, the correlation analyses of management practices related to construction methods are presented.



Figure 4-8 Diagram for Correlation Analyses of the Management Practices Related to Construction Methods

The correlation analyses for safety and health practices are indicated in Figure 4-9.



Figure 4-9 Diagram for Correlation Analyses of Safety and Health Practices

Correlation analysis was also conducted to investigate the relationship between construction management practices, project cost, project delay and company profiles (company size, company experience, and annual turnover). In the analysis, the construction companies were categorised based on the number of employees (less than 19, 20-199, and greater than 200) to determine their sizes. The classification system is

adopted from the Australian Bureau of Statistics (ABS). For instance, according to ABS classification, at the end of 2015 financial year, the proportions of companies involved in building construction were 0.06%, 0.78% and 99.15% for firms employing over 200 workers, 20-199 workers, and less than 19 workers respectively (Australian Bureau of Statistics, 2016c).

4.5.5 Linear Regression Analysis

To investigate the proportion of variance in the productivity explained by the construction management practices, bivariate linear regression analysis was conducted. According to Vittinghoff and McCulloch (2007), both observational and experimental data can be analysed by using the regression analysis, and the analysis tells how the independent variables might affect the dependent variable. The author explained that if the independent variables cannot be controlled, the data is observational. On the other hand, if the factors being studied can be controlled, the data is experimental. In this research context, since the management practices of the construction projects cannot be controlled by the researcher, the data is observational.



Figure 4-10 Regression Analysis between Productivity and Construction Management Practices

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

After testing the assumptions of the linear regression analysis such as linearity by using SPSS-24, the linear regression analyses were conducted to investigate the relationship between productivity and the six categories of the practices: construction materials management practices, construction equipment management practices, preconstruction phase management practices, human resource management practices, management practices related to construction methods, and safety and health (Figure 4-10). The linear regression analysis between productivity (the dependent variable) and the aggregated construction management practices (the independent variable) was also conducted. The results of the analyses are presented in Chapter 7.

4.5.6 Logistic Regression Analysis

Logistic regression analyses were conducted to address Objective 3 of the study which is 'to build and validate a tool that can be used to predict the probability of exceeding baseline productivity factor based on a score of the management practices.' Before running the analysis, the suitability of the sample for the logistic regression analysis was checked by computing the number of events per variable (EPV). The rule of thumb for sample size for logistic regression analysis states that EPV should be greater than 10 (Peduzzi et al., 1996). Some authors argue that the minimum of 10 events per predictor is conservative (Vittinghoff and McCulloch, 2007). In this research context, the number of positive events refers to the number of projects whose productivity factors exceed the baseline productivity factor (PF); the predictor is the construction management practices (one variable); and the minimum EPV of 10 is used to develop the logistic regression models. The results of all the computed EPVs are presented in Chapter 8.

The three measures of central tendency (mean, mode and median) of the projects' productivity factors and Z-score for the skewness of the productivity factors data (skewness/standard error) were computed, and the central-PF value was fixed (Lurd and Lurd, 2013, Kim, 2013). The Events Per Variables (EPVs) were calculated using (central-PF value), (central-PF value) + 5% × (central-PF value), (central-PF value) - 5% × (central-PF value), (central-PF value) + 10% × (central-PF value), and (central-PF value) - 10% × (central-PF value) as baseline productivity factors.

After fixing the cut-off productivity factors, the datasets were divided into model building and validation. Random numbers using Microsoft Excel was assigned to each data point so that the data can be randomly distributed. Based on the random numbers, 100

the data was sorted in ascending order before it was split into model building and validation dataset. In regression analysis, random sample splitting and K-fold cross validation techniques can be used to divide the data into model building and validation datasets (Steyerberg et al., 2001). In this research, 3-fold and 4-fold cross-validation techniques were used as they are less biased than sample splitting method (Abou-Assaleh et al., 2004). Alternative models using the 3-fold and 4-fold cross validation techniques were developed and compared before choosing the final models (Figure 4-11). The criteria for selection include the overall prediction accuracy and statistical significance of the variable in the model. The analyses and selection of the models are described in Chapter 8. Linear regression models were also developed using the cross-validation technique. Figure 4-11 indicates the flow chart for the process used in logistic regression model development and validation.



101

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

To validate the logistic regression models, the probabilities of validation data sets were computed; productivity factors were predicted as binary values (1,0) and compared with the actual productivity factors; Receiver Operating Characteristic (ROC) curves were drawn and compared with the reference lines; and area under the ROC curves (AOC) were computed. According to Hanley and McNeil (1982), the AUC is interpreted as acceptable if the graph is close to the left top boundary, and AUC of 0.5 and less implies no accuracy and AUC=1 indicates perfect accuracy. To check the reliability of the logistic regression models bootstrapping using 1000 samples were conducted. Willems and Van Aelst (2005) explained that the use of 1000 or more samples provides accurate output in bootstrapping. Finally, sigmoid graphs were drawn by computing the probabilities of exceeding baseline productivity factor by using the validated model's equation.

4.6 Summary

This chapter presented the research design and methodology used in the study. The overall research process has been described using a flowchart. The philosophical assumptions of the study were explained, and the suitability of mixed methods research approach for this research has been addressed. The types of mixed methods research design have been reviewed and the 'exploratory sequential mixed methods' has been found to be suitable for this research. In the exploratory sequential mixed methods, the qualitative data is collected and analysed first, and the output of this stage is used as an input to the second stage. On the basis of the findings of the first phase, the quantitative data is collected and analysed during the second phase.

The qualitative data collection and analysis techniques such as the use of semistructured interviews, the deployment of matrix box to display the summaries of the interviews results, and the techniques of determining the saturation point and arriving at a conclusion have been described. The quantitative data collection and analysis techniques which include the use of interview questionnaire survey in which the survey was self-administered; techniques of computing productivity factors and relative importance indexes; methods of refining and validating the scoring tools; correlation and linear regression analyses; and probability-based models building and validation techniques have been explained. In next chapter, the qualitative data analysis and findings are presented.

CHAPTER FIVE

5 QUALITATIVE DATA ANALYSIS AND FINDINGS

5.1 Introduction

In this chapter, the analyses and findings of Phase I of the study are presented. The profiles of the interview participants are described in Section 5.2. Following this section, the analysis and findings of the construction management practices are presented in six subsequent sections (from Section 5.3 to Section 5.8). Section 5.3 explains the analysis and findings of the practices categorised under 'construction materials management practices.' Section 5.4 describes the analysis and findings of the practices organised under 'construction equipment management practices.' The analyses and findings of the management practices related to construction methods, preconstruction phase management practices, human resource management practices, and safety and health practices are presented in Section 5.6, Section 5.7, and Section 5.8 respectively. Finally, Section 5.9 provides the summary of the chapter.

5.2 **Profiles of the Interview Participants**

The percentage distribution of the interview participants based on the areas of expertise is shown in Figure 5-1. Accordingly, 32% of the interviewees are project managers. Contract administrators, project coordinators, site engineers, and site supervisors each comprise 11% of the participants; and general manager, construction manager, project planner, project planner, and cost manager each consists of 5% of the total interviewees.



Figure 5-1 Distribution of the Interview Participants Based on Areas of Expertise

Figure 5-2 presents the percentage distribution of the interview participants based on their experience in building construction. In the analysis, the participants' experience is grouped into 0 to 4 years, 5 to 9 years, 10 to 14 years, 15 to 19 years and greater than 20 years (Figure 5-2). Accordingly, about 68% of the participants have more than ten years, 32% of them have experience between 5 to 9 years, and no participant has experience less than five years.



■ 0 to 4 ■ 5 to 9 ■ 10 to 14 ■ 15 to 19 ■ 20+

Figure 5-2 Distribution of Interviewees Based on Experience

5.3 Identification of Construction Materials Management Practices Influencing Productivity in Multi-Storey Building Projects

Six construction materials management practices that have the potential to improve productivity in multi-storey building projects are identified. These are 'Procurement Plans for Materials,' 'Long-Lead Materials Identification,' 'Materials Status Database,' 'Materials Delivery Schedule,' 'Material Inspection Process,' and 'Materials Inspection Team.' However, the practices 'Procurement Team,' 'On-site Materials Tracking Technology,' and 'Post Receipt Preservation and Maintenance' which are included in the preliminary conceptual framework of the construction materials management practices are found to be not significant to improve the productivity of multi-storey building projects in the context of Victoria State, Australia. The summary of the interview responses is indicated in Table 5-1.

Table 5-1 Summary of the Interview Results of Co	onstruction Materials Management Practices
--	--

Construction Materials Management Practices	Summary of the Interview Responses	Conclusion
Procurement Plans for Materials	"Materials procurement plan is right on the front end. The potential suppliers are identified; the required materials are estimated upfront; the tracking sheets and charts are prepared; different stages of the procurement are represented using different colours on the chart, and any changes made during the procurement process are tracked."	suitable
Long-Lead Materials Identification	"Usually the long lead times are façades, tiles, and services such as lifts, generators, and boilers. All these key primary elements have long lead time. These items are identified earlier and tabulated. The approximate lead times for these materials are also estimated; for example, item = tiles, source = Italy, duration = 8 weeks. Most items imported from overseas are categorised as long lead items."	suitable
Materials Status Database	"The contractor's project manager get information regarding the work progress from supervisors or subcontractors and prepare the material status report which comprises of the percentage of the materials utilised and the estimates of the quantities of materials required to complete the remaining works. The supervisors then monitor the work on a regular basis; they identify the remaining items and request the procurement of these materials if any."	suitable
Materials Delivery Schedule	"The materials delivery schedule is prepared based on the construction work program; the principal contractor's supervisors discuss the materials delivery dates with respective subcontractors/suppliers and prepare the schedule. The schedules are written on a whiteboard, and the materials delivery is tracked every day. Some materials such as façade are tracked during the manufacturing process; when they are on site; and after their installation."	suitable
Material Inspection Process	"The inspection is made upfront, the consultants are engaged in the inspection, and sometimes materials are inspected during the manufacturing stage. Inspection is made for various items such as reinforcement, formwork, glass and concrete test. The materials should be inspected before being installed. The subcontractors or suppliers provide the quality assurance documents for all items they bring to a site and the principal contractor's project management team conducts quality assurance checks."	suitable
Materials Inspection Team	"The project coordinators and specialist consultants are involved in the inspection process. For instance, façade expert and project coordinator or manager inspect the façade, and structural engineers take part in the inspection of reinforcement bars. Other consultants are also involved depending on the type of the materials or works."	suitable
Procurement Team for Materials	" A project coordinator or a project engineer or a contract administrator can oversee the procurement process. Forming the materials procurement team is not essential for productivity."	not suitable
On-site Materials Tracking Technology	"The materials are not stored on site; they are delivered to the site when required; thus, tracking technologies are not significant for productivity. When materials are brought to a building construction site, they are placed in a location that is very close to where they are installed."	not suitable
Post-Receipt Preservation and Maintenance	"Most building materials are not stored on sites due to the shortage of the storage areas, and the practice 'post-receipt preservation and maintenance' is not relevant for productivity improvement in multi- storey building construction projects."	not suitable

105

The implementation of the practice 'materials procurement plan' has a positive impact on the productivity of multi-storey building construction projects. The interview participants described that the building industry structure in Victoria, Australia influences the procurement strategy. They explained that most principal contractors do not directly involve in purchasing the construction materials, but they manage the procurement process of the materials. The respondents described that the materials procurement program, which is based on the main construction program, is developed by the principal contractor and provided to the subcontractors or materials suppliers. Some interviewees described that the contractors collect information regarding the location of the manufacturers (overseas / local), the time it takes to manufacture the materials, and the duration of the delivery of the materials to prepare the materials procurement plan.

The respondents explained that extra time or buffer time is allowed while preparing the procurement plan as it might take a longer time to manufacture the materials. According to the interviewees, the principal contractors monitor the procurement process to ensure the manufacturing and the delivery of the materials. The principal contractors request subcontractors or material suppliers to issue letters which confirm the manufacture of the materials; they can also request the bill of lading to make sure that materials are shipped. Some interviewees described that verifications can sometimes be made by travelling to the country where the materials are manufactured. The participants suggested that the continuous monitoring of the procurement process by conducting a weekly meeting and reviewing the available materials on site is essential to reduce the shortage of materials and to enhance productivity.

Identification of long-lead items or critical materials is found to be one of the materials management practices that have the potential to enhance productivity in multi-storey building projects. The interviewees described that identification of the long-lead items and preparation of the package leading program, a program that is prepared to suit the long lead items, prior to tendering is essential to enhance productivity. Some participants proposed that requesting the tenderers to submit the quotations for the materials including the source and lead times is essential to identify the critical materials easily. For instance, besides the price of the tiles, tenderers can be requested to provide additional information such as the tiles will be imported from Italy, and it will

take eight weeks to deliver to a project site. Other interviewees explained that most building materials that are imported from abroad are categorised as long lead items. Some of these materials include tiles, light fittings, façade and lift.

The practice of preparing the material status database has been found to have a positive impact on the productivity of multi-storey building projects. The respondents explained that the principal contractor's project manager gets a report from subcontractors regarding the progress of their works, reviews the work progress report and prepares the materials status report for each work package. Some interviewees suggested that including information such as 'the shop drawing is prepared,' 'materials are in production,' 'materials are ready for shipping,' 'materials are shipped,' 'ETA (Estimated Time of Arrival) of the materials,' and 'materials' installation date' in the material status database is essential. Other respondents described that, in some projects, it is entirely the responsibility of subcontractor or materials supplier to quantify and order the right materials that are sufficient for the job. They explained that some principal contractors provide the necessary documents to the subcontractors and transfer the risks associated with procurement of materials to them. Some interviewees stated that certain principal contractors carry out continuous monitoring of the materials and assist the subcontractors or suppliers to deliver the right quantity and quality of materials.

Preparation of construction materials delivery schedule is found to be one of the construction materials management practices that could improve productivity in multistorey building projects. The interview participants described that the construction materials delivery schedule is commonly prepared on a weekly basis, and it is done in consultation with site supervisors. The interviewees explained that the schedule helps to avoid the delivery of various materials at the same time. The suppliers are allowed to access a building site at designated times and use cranes or lifts accordingly. They also described that the delivery schedule is shown on board at a building site where subcontractors or materials suppliers can easily get materials related information. The schedule is updated to accommodate for delays in delivery due to, for instance, inclement weather and other unforeseen conditions. Some respondents explained that the schedule is displayed on a whiteboard in a meeting room so that the delay in the knows which materials are delivered on a specific date. They stated that the delay in the

schedule can also be shown on the board. The interviewees suggested that careful planning of the materials delivery schedule has paramount significance to reduce delays, particularly, in projects located in the central business districts (CBD) areas since delivering some materials to these projects might not be allowed during the normal working hours.

Developing a procedure for materials inspection is found to be an essential practice to enhance productivity in multi-storey building projects. Some respondents described that the best practice for the materials inspection starts from the fabrication phase. The respondents recommended that some imported materials should be inspected while they are overseas. The inspection should also be conducted after they arrive in Australia; when they are delivered to a site; and after they are installed. According to the interviews, inspection during fabrication helps to ensure that the materials are manufactured as per the required standards or quality level. Some interviewees explained that subcontractors or material suppliers provide the quality assurance documents for all items they are supplying, and the principal contractor's project team conducts quality control checks to make sure that the materials are procured according to the specifications provided in the contract document. Other respondents described that the materials samples are kept on site until the project is completed, and the quality of the supplied materials are checked against the samples. They also stated that there are technical data or product data signed off by the specialist consultant against which quality checks are done, and the consultants make scheduled site visits to identify defects in the installed materials.

The formation of materials inspection team is found to be an important practice to enhance productivity in multi-storey building projects. The respondents explained that the project coordinators, supervisors or persons in charge of a particular trade and a project manager conduct inspections. The project coordinators track the procurement process and conduct on-site and off-site inspections with the assistance of the specialist consultants. For instance, some interviewees described that for façade materials, façade experts are members of the inspection team. Other respondents explained that preparation of an Inspection and Test Plan (ITP), which is a program for inspecting and testing of materials and works, is a good practice to assure quality and enhance productivity. According to the interviewees, the inspection checklist for each item is

prepared, and the site manager or other assigned staff conduct the quality assurance as per the checklist.

Finally, based on the findings of the interviews, the nine independent variables in the preliminary conceptual framework of the construction materials management are reduced to six independent variables, and the validated framework is shown in Figure 5-3.



Figure 5-3 Validated Conceptual Framework of Construction Materials Management Practices and Productivity

The materials management practices such as 'procurement team for materials,' 'on-site materials tracking technology,' and 'post-receipt preservation and maintenance' are found to be not suitable to enhance productivity in multi-storey building projects. The interviewees described that forming the construction materials procurement teams is not essential for enhancing productivity. They explained that in most multi-storey building projects in Victoria, the projects are broken down into many packages, and supply and install contracting type is used. The interviewees described that a project coordinator or a project engineer or a contract administrator can oversee the procurement process.

On-site materials tracking technology is also found to be not important practice for improving productivity in multi-storey building construction projects. The interview participants described that as most building materials are not stored on project sites due to the shortage of storage spaces, the materials tracking technology is not required. They explained that all materials are not brought and stored on a building site. The materials are delivered when they are needed and placed close to their installation areas. Some interviewees added that to reduce the damages of the materials, they are not stored on building sites. The respondents suggested, to minimise the materials' damage on sites,

the delivery of the materials one day before their scheduled installation date. According to the interviewees, if materials are delivered on the date close to their scheduled installation date and placed near their installation areas, the loss of productivity due to the mishandling of the materials can be reduced.

The practice post-receipt preservation and maintenance is also not considered as a good practice for improving multi-storey building construction projects' productivity. The interview participants explained that implementation of this practice is recommended when there are sufficient materials storage areas on project sites; however, most multi-storey building projects in the city of Melbourne do not have sufficient storage spaces, and the practice is not essential for increasing productivity.

5.4 Identification of Construction Equipment Management Practices Influencing Productivity in Multi-Storey Building Projects

Construction equipment procurement plan, construction equipment productivity analysis and construction equipment maintenance are found to be the three construction equipment management practices that have the potential to improve productivity in multi-storey building projects (Table 5-2).

The interview results reveal that preparing plan for either leasing or hiring of construction equipment has a positive impact on productivity. Some participants explained that cranes and hoists are the most frequently used equipment in multi-storey building construction projects. According to the interviewees, the lack of proper procurement procedures can negatively influence the productivity of the projects. Other interviewees explained that subcontractor involved in structure works provide equipment such as a crane, and the principal contractor negotiates on how other subcontractors use the equipment. Moreover, principal contractors can identify any equipment required for the execution of a specific task and include in the subcontract agreement so that a particular subcontractor is responsible for hiring it.

According to the interviews, the productivity of a crane, for instance, is analysed by considering its speed in lifting materials and the source of power. For example, electric crane does not make noise, and contractors can start their works early in the morning without disturbing neighbours. However, if it is a diesel crane, they could not start early as the crane's engine is noisy. The principal contractor should start work as per the

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

permission provided to them by the city council. Cranes that have fast speed and capable of lifting maximum weight to be lifted at a particular building project are preferred. For a mobile crane, weather conditions are also considered. Contractors need to look at weather forecasts before bringing the equipment to the sites or signing an agreement with the crane hiring company.

Table 5-2 Summary of Interview Results of Construction Equipment and Tools Management Practices

Construction Equipment	Summary of the Interview Responses	Conclusion
Construction Equipment Procurement Plan	"In high-rise building projects, a crane is the major equipment that influences productivity. Sometimes there are many construction projects in Melbourne city and cranes or other construction equipment might not be available for a particular job. Thus, procurement of the equipment should be planned ahead of the commencement of a project to reduce delay. Thus, the practice ' equipment procurement plan' is vital to improve productivity."	suitable
Construction Equipment Productivity Analysis	"The practice is essential. For instance, a low-speed hoist costs less money, but it takes longer time. On the other hand, a high-speed hoist lifts materials quickly and saves construction time. For high- rise building, it is recommended to use two high-speed hoists side by side to increase productivity. The time the crane takes to place concrete panels or other items is recorded to analyse its productivity."	suitable
Construction Equipment Maintenance	"Contractors should ensure that equipment is calibrated, inspected and maintained. Induction should be provided when new equipment is delivered to a building site. In some situations, if there is no maintenance record, the equipment might not be allowed to be used on a particular project. Therefore, equipment maintenance is necessary for productivity improvement."	suitable
Tools Management Strategy	"Most of the tools used in multi-storey building construction projects are not stored on site due to a shortage of spaces. Thus the implementation of the practice 'tools management strategy' is not essential for productivity enhancement."	not suitable
Tools Tracking Technologies	"The practice is not significant to increase the productivity of multi- storey building construction projects as most tools are not stored on sites."	not suitable
Tools Maintenance	"The life span of the tool is monitored, and corrective actions can be taken, but the practice is not essential for enhancing productivity."	not suitable

In analysing the hoists' productivity, the number of stories of a building and the floor area is considered. The interview participants explained that for a high-rise building, greater than thirty stories, it is recommended to use two high-speed hoists side by side than using two low-speed hoists. The cost of a slow speed hoist is less than that of a high-speed, but it takes longer time and productivity is affected if the slow-speed is deployed. The time required to transport materials from ground to the desired working level is recorded and analysed to choose the most suitable hoist. Some respondents

added that for small jobs, up to three storeys, the choice of the speed does not make any difference in productivity. However, for multi-storey buildings, the selection of appropriate hoist can make a major difference.

Construction equipment maintenance is found to be one of the practices for improving productivity in multi-storey building projects. According to some participants, crane, for instance, is commonly serviced every two weeks, and the hoist is serviced once a week. When subcontractors are responsible for hiring construction equipment, they are required to provide evidence regarding the maintenance of the equipment to the principal contractor. Other respondents added that inspecting and maintaining the construction equipment or crane, in particular, is fundamental because lifting building materials is a risky task. If materials fall while lifting due to the usage of unsafe cranes, the contractor is responsible for damages to nearby properties or persons. The respondents also described the regulatory requirement that must be fulfilled regarding the frequency of maintenance and what items should be inspected regularly.

Based on the findings of the interviews on construction equipment and tools management practices, the six independent variables in the preliminary conceptual framework of construction equipment management practices are reduced to three independent variables, and the validated conceptual framework is shown in Figure 5-4.



Figure 5-4 Validated Conceptual Framework of Construction Equipment Management Practices and Productivity

Tools management strategy, tools tracking technologies, and tool maintenance are found to be the practices which are not significant to enhance productivity in multistorey building projects. The respondents explained that the majority of contractors involved in the delivery of multi-storey building projects in Victoria State, Australia are managing contractors, and they do not purchase and store the tools on building sites. Thus, the practice 'tools management strategy' is not relevant to improve productivity in multi-storey building projects.

The use of tools tracking technology is found to be a practice which is not important to enhance productivity in multi-storey building projects. The interviewees described that due to the limited number of tools owned by a particular contractor, on-site tracking technologies such as the use of tags or barcodes is not significant. Other interviewees added that because of the shortage of on-site storage spaces, most tools are not stored on sites, and the use of tools tracking technologies is not required. The respondents suggested that it is preferable to include the cost of tools in the labour cost than purchasing and managing the tools. The respondents explained that a tradesperson who executes a particular task can bring the required tools and the tools' cost can be included in the hourly labour cost. The participants of the interviews explained that the implementation of the practice 'tools maintenance' has less relevance in improving productivity in multi-storey building projects. They described that maintenance of machinery such as forklifts is more relevant for enhancing productivity than the maintenance of tools.

5.5 Identification of Management Practices Related to Construction Methods

The summary of the interview results (Table 5-3) shows ten management practices that have the potential to enhance productivity in multi-storey building construction projects, and the practices are related to construction methods. These include integrated schedule, work schedule strategies, schedule control, dynamic site layout plan, traffic control plan, site security plan, machinery positioning strategy, project start-up plan, project completion plan, and innovations and new technologies.

A scheduling system that integrates work, materials procurement and delivery, machinery, budget, and other resources has a positive impact on the productivity of multi-storey building projects. For example, if work schedule is integrated with materials and machinery schedules then the project teams are aware of when a particular activity is executed, what type of machinery should be hired, how much material and manpower should be deployed, and track the delivery of the materials. The research participants described that the best performing contractors link all the schedules together and update the schedules on a weekly or biweekly basis.

Table	5-3	Summary	of	the	Interview	Results	of	Management	Practices	Related	to
		Constructio	on N	Meth	ods						

Management Practices Related to Construction Methods	Summary of the Interview Responses	Conclusion
Integrated Schedule	"Linking various programs is essential. For instance, most contractors try to link materials procurement, lead time, work status, main work program and other issues such as the schedule for FFE (Fittings, Fixture, and Equipment) together."	suitable
Work Schedule Strategies	"There is an agreement which dictates the normal working hours per week. However, depending on the type of work (critical or near critical activities) the principal contractor negotiates with subcontractors to schedule flexible working hours. For instance, for the sake of safety, jumping of crane might not be made during normal working hours. Thus, there should be a negotiation between contractors on the date and time when the activity is done."	suitable
Schedule Control	"Principal contractors prepare target program which is usually shorter than the contract program and monitors the progress of the works executed by subcontractors accordingly. Some supervisors work in close collaboration with subcontractors, control the schedules and report to the principal contractor's project manager. Regular weekly or biweekly meetings are conducted to resolve issues which hamper the progress."	suitable
Dynamic Site Layout Plan	"The site plan is constantly changing for the sake of productivity. The gates in and out, for instance, might be altered throughout the construction period. They could be changed after completion of excavation or during superstructure and landscaping works. The site layout plan should also consider the location of temporary facilities at various stages of the project."	suitable
Traffic Control Plan	"Traffic control plan is done at different stages of construction. For example, there are traffic control plans during excavation, pouring concrete and site works. The traffic control plan is usually done by a certified traffic management company. The principal contractor employs trained traffic controllers or subcontractors are responsible for controlling the traffic during the execution of the subcontracted works."	suitable
Site Security Plan	"Site security depends on the type and the stage of the building construction projects. If long lead materials which are imported from overseas are stolen or lost towards the end of a project, there could be a project delay."	suitable
Machinery Positioning Strategy	"The location of the critical machine such as tower cranes is planned by considering the weight to be loaded, distance from the street and the nearby buildings or other property. Thus, developing a strategy for positioning cranes is critical for productivity."	suitable
Project Start-up Plan	"Prior to the start of a building construction project, the project team conducts start-up meeting that helps them to go through the list of items that trigger their memory."	suitable
Project Completion Plan	"Good project completion plan should start on day one. This could be done by reviewing the project plans and listing out all the things that need to be completed to hand over the project in time. For example, the dates when to obtain certificates of the completed tasks and other issues should be listed on spreadsheets."	suitable
Innovations and New Technologies	"There is no specific department responsible for investigating new technologies and innovations in most construction companies. However, in some situations, a project manager has the authority to implement suitable technology for a particular project."	suitable

Developing suitable working hours strategies is found to be another important management practice that has the potential to improve productivity in multi-storey building projects. There are working hour restrictions imposed by city councils, and contractors are required to develop a strategy to reduce project delays. Most principal contractors in Victoria, use the calendar prepared by Construction, Forestry, Mining and Energy Union (CFMEU) which is typically 36 working hours per week. They schedule to work either four, five or six days a week and sign contracts with their subcontractors accordingly. Some respondents described that they schedule to work only six days per week and consider Sundays and Rostered Days Off (RDOs) as non-working days. Other respondents explained that on Saturdays, they reduce the working hours by half (50%). Some interviewees described that they prefer to work five days per week because they believe that people are not productive on Saturday, Sunday and RDOs. The participants added that it is expensive to work on Sundays and after hours as the rates provided in the Enterprise Bargaining Agreement (EBA) or in the Fair Work Commission's awards is high. Other interviewees described that the working hours are influenced by Construction, Forestry, Mining and Energy Union (CFMEU) and contractors are not flexible to develop different strategies such as working after hours, on Saturdays and shifts. However, for critical works that cannot be executed during the normal working hours, for example, for crane jumping, Saturdays or after hours can be scheduled.

The practice 'schedule control' is found to be one of the management practices that are related to construction methods and positively influencing productivity in multi-storey building projects. The interviewees explained that the principal contractors prepare a target program which is shorter in duration than the main contract program, and the subcontractors sign the agreement based on the target program. The respondents described that based on the target program, the subcontractors plan the number and the types of workers and machinery to be deployed to meet the target program. After the commencement of the construction works, the principal contractor's project management team monitor the progress of the works executed by subcontractors; and prepare a report which comprises of the percentage completed, the remaining duration and the forecasted completion dates. The project manager also conducts a weekly or biweekly meeting to discuss issues related to the project status and other matters which influence the productivity of a multi-storey building project.

Dynamic site layout plan is found to be an essential practice that could improve productivity in multi-storey building projects. Some interviewees described that the site plan is constantly changing for the sake of productivity, and it is prepared by considering different phases of a building construction. They suggested that the gates should be planned to allow easy movement of trucks and workers, and the location of the gates could be changed to suit different activities in the project. For instance, the location of the gates during excavation, superstructure, and landscaping or site works could be different. The respondents proposed two gates (gate-in and gate-out) during the excavation works. If there is only one gate, it is not suitable for trucks to get into and out of the site and productivity can be reduced. Some interview participants described that although the practice of adopting dynamic site layout is important, it should be planned ahead to be effective. They explained that experienced contractors include their site logistics plan in a tender document. The changes in a plan that is based on the stages of construction are clearly shown in the tender document so that subcontractors are aware of the future changes in the site layout. Moreover, the respondents described that integration of the traffic control and the site layout plans has a paramount importance to reduce the loss of productivity.

Traffic control plan is found to be one of the management practices that could enhance productivity in multi-storey building projects. The interviewees explained that traffic control plans that are suitable for different stages of the building construction are prepared. The respondents suggested that traffic control plans should be prepared during excavation, pouring of concrete, delivering precast panels, and completion of the project (execution of the site works). The interviewees explained that the traffic control plan is usually done by a certified traffic management company and approved by the relevant city councils. Some of the interview participants described that the principal contractor can employ qualified traffic controllers or subcontractors are responsible for assigning their traffic controllers.

Site security plan is found to be an important practice that could influence the productivity of multi-storey building construction projects. The respondents described that the level of the site security depends on the phases of the project and the value of the properties installed or stored inside the building under construction. They explained that as the project gets towards completion, there might be an impact on the facility or

some items could be stolen, and the project completion date will be delayed. As a result, the level of security is higher towards the end of the project. Some interviewees explained that certain contractors have developed procedures to build the hoardings, install the lights on the hoardings, assign after hour security guards, and place security cameras and localised- standard lamps on a site. Other respondents explained that if the site is not safe and secure, the productivity of employees decreases as they are not motivated to work in such environment. Moreover, if some of the items are stolen from the site, the contractor has to purchase and replace that item. Consequently, there can be the loss of productivity due to waiting for the replacement of the stolen or damaged materials.

The construction machinery positioning strategy is found to be an essential practice that influences productivity in multi-storey building projects. According to the interviewees, the critical machine for these projects is a tower crane, and its location on a construction site is planned by considering the weight to be loaded, the distance from the street from which the materials are lifted, the area of a building, and the distance from the existing buildings. The respondents explained that the crane needs to reach the street; it should have the capacity to lift the heaviest materials; it should cover the entire area; and it should not hit nearby buildings or other properties. In some circumstances, more than one crane can be used on a particular project if the above conditions are not satisfied by a single crane. Other respondents described that before the commencement of a building construction, the project manager and crane experts compute the footprint of the building; identify the number of cranes required for the given building area; and determine the appropriate location of the equipment or cranes. The respondents suggested that careful analysis of the position of the construction equipment using either 2D drawings or 3D model is an important practice to conduct the construction works smoothly. Besides improving productivity, the construction equipment location analysis saves cost. For instance, the number of cranes can be reduced from two to one after conducting the analysis. The interviewees described that project delays could occur due to the improper positioning of the construction equipment. They explained that, in some sites, the precast panels were too heavy to be lifted by a crane and the design of the panels was changed to suit the capacity of the existing crane. This resulted in the loss of productivity of the project due to design changes.

The preparation of the project start-up plan is essential for the smooth commencement of the construction of multi-storey building projects. The respondents explained that if the project start-up plan is done early, then a contractor can commence the construction of a building on time, and the chance of occurrence of initial project delay can be reduced. Prior to starting the project, the contractor's project team needs to conduct start-up meeting that helps them to go through the list of items that should be completed before the commencement date. The start-up plan assists the project team in identifying which tasks are completed and what activities are not executed yet. The interviewees explained that some contractors prepare a checklist which consists of many questions such as Are permits obtained? Are materials procured? Is earthwork machinery hired? Thus, the project start-up plan helps the project team to fine-tune what they are going to execute.

The practice 'project completion plan' influences the productivity of multi-storey building projects. According to the interviewees, the implementation of an effective project completion plan needs to start immediately after the commencement of the construction works. Respondents suggested that items which need certificates during the project handover phase should be sorted out early, and a project manager needs to track them starting from the project commencement date. For example, the concrete test certificates should be collected and documented immediately after the completion of the task. The respondents explained that if the relevant certificates are not obtained, the project handover process can be delayed. The project completion plan also states how temporary works and equipment such as crane are going to be dismantled. Some respondents described that, in some projects, the project completion plans. According to the interviewees, the completed building project might not be used unless the building surveyor or the relevant authority provides a certificate of occupancy. The project completion plan needs to address what should be fulfilled to obtain the certificate.

The use of new technologies and innovations in managing construction projects has a positive impact on the productivity of multi-storey building projects. The interviewees described that in some building construction projects, supervisors have their own tablet computers and use them for receiving and sending site information. These supervisors rarely use pen and papers to communicate with the project manager. Training is

provided to the supervisors regarding the utilisation of new technologies. Some participants explained that projectors, whiteboard, automatic photocopy, dual monitors, wireless internet, and other facilities are provided to the project sites. The respondents described that most project managers of the principal contractors use commercially available software as a communication tool with subcontractors and/or supervisors. For instance, the project manager can send the pictures of the defects of the executed works including the defects' specific locations as well as the rectification order to the concerned subcontractors. Some interviewees described that contractors use a drawing review system in which drawings are reviewed electronically rather than manually.

Finally, based on the findings of the interviews, ten management practices related to construction methods are identified, and the validated conceptual framework is presented in Figure 5-5.





5.6 Identification of Preconstruction Phase Management Practices Influencing Productivity in Multi-Storey Building Projects

The interview results revealed ten preconstruction management practices that have positive impacts on the productivity of multi-storey building construction projects in the context of Victoria State, Australia. These include Well-Defined Scope of Works, Short Interval Plan, Buildability Review, Construction Work Packages, Regulatory Requirement, Contract Types, Use of Software in Planning Work Packages, Utilities

119

Alignment, Dedicated Planner, and Model Development. In Table 5-4, the summary of

the interview responses is presented.

Table 5-4 Summary of the Interview Results of Preconstruction Phase Management Practices

Preconstruction Phase Management Practices	Summary of the Interview Responses	Conclusion
Well-Defined Scope of Works	"The scope of works is prepared before tendering phase, and there is a section called scope of works in the tender document for instance, façade work package has its scope of works, the structure work has its scope of works, and finishing and fitting work packages also have their scope of works. A well-defined scope of works is linked with drawings and specification, and it can be prepared based on the previous scope of works developed for similar projects. A model scope of works which can be used to prepare project specific scope of works should be prepared by contractors."	suitable
Short Interval Plan	"The short interval plan is commonly prepared on a bi-weekly basis. In the plan, the daily tasks and the required resources are indicated. The short interval plan is usually prepared after conducting project team meeting. The Project manager, contract administrator, project engineer and Foremen or supervisors are involved in the meeting. Based on the size of the project, the construction manager, sometimes, involves in the preparation of the plan."	t suitable
Buildability Review	"In some projects, there is a logistic planner or a person who reviews the design before its implementation. Depending on the scope of a project, besides the design manager, other professionals are also involved in the buildability review. Design reviews need to be regularly conducted. Getting inputs from subcontractors is also essential."	e suitable
Construction Work Packages	"A building project can be broken down into approximately 20 to 25 work packages. Tile, façade, structure, lift and landscape are some of the examples of the work packages. The construction work packages are prepared to suit the capacity of local subcontractors. Each work package has work program, drawings, specifications, budget, the scope of works, and site logistics."	e suitable f
Regulatory Requirement	"Various permits should be obtained from local authorities These include traffic management plan, construction management plan, connection to services permit, planning permit, building permit, demolitions permit, asset protection permit, and occupancy permit among others."	g suitable
Contract Types	"Predominantly, lump sum or one figure contract is used Sometimes when the scope of work is not clearly defined or small, the principal contractors employ daily labourers to execute the activity. Generally, work packages having costs exceeding AUD 50,000 are classified as major works and more detailed contract documents are required."	suitable
Use of Software in Planning Work Packages	"Various software is used in planning the work packages. This includes Microsoft Excel, MS Project, Primavera, Job Pack and other proprietary software."	s I suitable
Utilities Alignment	"Before commencing the construction works, the 'Dial Before You Dig' referral system is used to identify the location of the existing services. Contractors should get approval from water gas and electric power authorities before connecting to these utilities. The project owners should also contact the respective authorities."	e suitable
Dedicated Planner	"Assigning the dedicated planner at a company and/or a project level is essential. The planner has relevant experience and records such as productivity data and activity durations which help him/her prepare good project programs."	t I suitable
Model Development	"3D-model development is essential as it facilitates communication of the project information. The model is vital for the construction of complex structural steel works. It is also significant to identify the clashes among different services such as mechanical, electrical and plumbing."	s suitable

The practice well-defined scope of works is found to be one of the practices that have the potential to enhance productivity in multi-storey building projects. Some of the interview participants described that if uncertainties in the scope of works are reduced, or if the scope is defined well, the productivity of a building project can increase since contractors are certain on what they do and when they do. They explained that defining the scope of works involves clarifying in words what a contractor needs to do to deliver a work package or a project. For instance, the scope of works for a subcontractor involved in façade works could state the responsibilities of the subcontractor such as the provision of all necessary allowances for coping and flashing, the use of the subcontractor's own means of access, the safety requirements, and the completion of the works in accordance with the program provided in the appendix to the contract. The respondents described that a well-defined scope of works is linked with drawings and specification, and it can be prepared based on the previous scope of works which was developed for similar projects.

The preparation of a short interval plan is also found to be one of the practices that have the potential to improve productivity in multi-storey building projects. The interviewees described that the plan shows the daily activities together with the resource requirements. Some respondents explained that the short period look-ahead programs (one week or two weeks' programs) help to focus on specific tasks. They described that short interval planning is beneficial as it indicates what contractors should do at a certain point in time. For instance, if the current tasks are excavation and pouring of foundations' concrete, then the project team can focus on planning and deploying of the resources which are required for the execution of these tasks. Some respondents explained that contractors usually prepare the master program and detailed short interval plans. They described that if the main program, for instance, provides nine days for excavation and earthworks, then the short interval plan provides details such as three days for bulk excavation, two days for cart away of waste soil, two days to compact the soil and two days as buffer time. Other interviewees stated that some contractors prepare the main works program that is shorter than the contract program or a project completion program which is agreed between the client and the contractor. According to the interviewees, a weekly or bi-weekly look ahead programs are prepared based on the milestones dates provided in the main works program. The interview participants

suggested that the short interval plan should not be prepared using the top-down approach in which a project manager or planner dictates the durations and completion dates. They recommended that the plan should be done by involving other project team members such as site manager, supervisors or forepersons, project coordinator/project engineer and contract administrator.

Buildability review is found to be the essential practice to enhance productivity in multi-storey building projects. Some interviewees suggested procedures that could facilitate the review of the drawings before being issued to the site. They described that when a consultant issues drawings, the document controller receives and hangs the drawings on red sticks; and the project coordinators, design manager, project manager, and services managers review the drawings on the sticks and send back to the consultant for revision. The respondents recommended that drawing should be reviewed by the project team for compliance with the design brief, for any scope changes which have a major impact on cost, and for the purpose of constructability. Some interviewees explained that it could be difficult to get a design manager who knows the detailed construction processes of the building projects. They explained that most local design managers are architects who understand the details on paper, but they might not necessarily understand the construction process. According to the interviewees, in some projects or activities, how much time it will take to install can be as important as the actual cost of the material, and conducting design review can save cost and improve productivity. The respondents explained that the involvement of contractor in the design process is essential. Accordingly, awarding a contract when about 80% of the design is completed and involving the contractor to complete the remaining 20% of the design is a good practice for productivity. The participants described that this practice provides an opportunity for the contractor to provide comments regarding the constructability issues. The interviewees suggested that for the sake of productivity and smooth workflow, it is preferable to novate the original designer than to engage a new designer.

The practice of preparing manageable 'construction work packages' is found to be one of the factors which positively influence the productivity of multi-storey building projects. The interviewees explained that the building project's work packages are prepared by considering the skill and the financial capacity of the local subcontractors. Some respondents described that the principal contractors do not want to break the

project into very small work packages that do not worth the subcontractors to tender. Other participants suggested that, during the subcontractor selection, besides the document review, an interview should also be conducted to ensure that the subcontractor has the required capacity and experience. The respondents described that certain contractors have the policy of 'gauging before engaging,' and every subcontractor is interviewed before awarding a contract. The interview participants also explained that in high-rise building construction projects, there are approximately between 20 and 30 work packages, and structural concrete work is one of the major work packages. The interviewees described that for the sake of productivity, it is preferable not to break down the structure package into smaller work packages. Some interviewees stated that it is not recommended to subcontract the formwork, the reinforcement bar work, and concreting activities separately. Accordingly, it is desirable to choose one subcontractor that is responsible for placing reinforcement bars and formworks, for supplying and pumping concrete, and for providing all the necessary machinery. The interviewees explained that, for a principal contractor, it is preferable to manage one subcontractor than managing three or four subcontractors that are involved in the concrete construction. Some interviewees described that the practice of awarding a concrete work package to a single subcontractor is essential not only for enhancing productivity but also for getting workers with multi-skills. For instance, workers doing formwork can also pour concrete, and those pouring concrete can do the reinforcement work.

Identification of the permitting requirements is found to be one of the preconstruction phase management practices that could increase productivity in multi-storey building projects. The interviewees described that various permits should be approved before commencing the construction of a building project. They described that some permits have predetermined lead times. For instance, it could take between 6 to 12 weeks to get the construction management plan (CMP) approved by the city councils. This lead time should be considered during the preparation of the work program. According to the respondents, unless the CMP is approved, the construction activities cannot be commenced. In the CMP, contractors are required to address how they plan to manage issues such as public safety, amenity and site security, operating hours, noise and vibration controls, air and dust, storm water and sediment, waste and materials re-use,

and traffic. The interviewees also explained that the approvals of other permits such as working on Saturdays and Sundays, working after hours, connection to the existing services, demolition, building permit, occupancy permit, traffic management plan and asset protection permits are required during the construction of multi-storey buildings in Victoria. The respondents described that some permits could take a longer time, and the productivity of a project can be influenced if the approval time is not considered while preparing the construction work program. Some interviewees described that a building permit can be approved in a shorter time than CMP and traffic management plan.

The contracting strategy is another preconstruction phase management practice that influences productivity in multi-storey building projects. The interviewees suggested that the use of back-to-back subcontracting approach is important to increase productivity. In this approach, if the head contract, for instance, is lump sum type, then the subcontract should also be lump sum type. The conditions in the head contract or the contract between the client and principal contractor are stated in the subcontract too. Thus, the subcontractors can get the chance to involve in the design, and the variations during the construction phase can be reduced. The respondents suggested that adopting lump sum contract type and involving contractors in the design of a multi-storey building can minimise the project risks and enhance productivity. Moreover, the interviewees recommended the use of appropriate contract forms; for instance, the work packages which cost over Australian Dollar (AUD) 50,000 are usually classified as the major works, and they need more detailed contract documents. However, work packages with a value less than AUD 50,000 can be considered as minor works.

The use of software in planning is found to be an essential practice that could enhance productivity in multi-storey building projects. The participants of the interviews described that preparation of the construction work schedules and the work packages using software can improve the productivity. Different software can be used to break down a project into manageable tasks. By using the software, work schedules which show materials and manpower resources, budget, milestone dates, and the durations can be prepared easily. According to the interviewees, most contractors use Microsoft Project as a planning tool, and some contractors use Primavera. Excel spreadsheet is also used for short interval planning. Some respondents explained that certain contractors have their own proprietary software that is used for planning purpose.

The practice 'utility alignment' is considered to be an important preconstruction phase management practice that has the potential to improve productivity in multi-storey building construction projects. Some interview participants described that in the construction of multi-storey buildings in Victoria, utility alignment is becoming a significant issue because of the presence of restrictions to connect the new utility lines to the existing networks. The interviewees explained that the infrastructure might not be available at the required location and some of the existing sewers do not have the capacity to provide services to the new buildings. Some respondents explained that although it is the responsibility of the client to settle the issue of external utilities, the contractor has to understand where the existing utility lines are and identify the best route to connect the new line to existing one. Other respondents explained that on most building projects, the 'Dial Before You Dig' referral system is used to identify the location of the services before commencing earthworks.

The assignment of a dedicated planner has a positive influence on the productivity of multi-storey building projects. The respondents explained that the complexity of a building project determines whether to assign the dedicated planner or not. Some respondents explained that when the project is small (less than three storeys), there is no need to have the dedicated planner as the project manager can do the work of the planner. Other respondents suggested that for multi-storey building projects, the assignment of the dedicated planner at either project or company level is considered as a good practice for enhancing productivity. The respondents explained that the dedicated planners could have rich experience and productivity data, and they can prepare better project programs.

The practice 'model development' is found to be one of the preconstruction phase management practices that have the potential to improve productivity in multi-storey building projects. The interview participants described that in building construction, the use of 3D models is essential to detect the clashes among building services such as mechanical, electrical and plumbing. The models can also be used in structural steel works. A structural steel subcontractor can use the 3D models to manufacture and install the structural elements. Some interviewees explained that the models are essential to facilitate communication. The 3D models help to visualise the design of an item. Some participants suggested that clients or project owners should make decisions
to invest in the use of 3D models. They explained that the investment could worth because it can reduce risks, particularly, in complex building projects. The use of the models has paramount significance in the installation of mechanical, electrical, and plumbing services. Thus, the contractors can easily see every service lines and avoid overlapping of the services; consequently, claims for additional time and cost due to the complexity of the design is reduced.

Finally, ten preconstruction phase management practices are identified using the interviews, and the validated conceptual framework of the pre-construction phased management practices is shown in Figure 5-6.



Figure 5-6 Validated Conceptual Framework of Preconstruction Phase Management Practices and Productivity

5.7 Identification of Human Resource Management Practices Influencing Productivity in Multi-Storey Building Projects

Eleven human resource management practices that have the potential to enhance productivity in multi-storey building projects have been identified. These comprise of Clear Delegation of Responsibility, Stability of Organisational Structure, Crew Composition, Retention Plan for Experienced Personnel, Employees Training, Skill Assessment and Evaluation, Career Development, Financial Incentive Programs, Social Activities, Non-Financial Incentive Programs, and Exit Interviews. In Table 5-5, the summary of the findings is presented.

Human Resource Management Practices	Summary of the Interview Results	Conclusion
Clear Delegation of Responsibility	"The roles and responsibilities of the staff are included in the employment contract. The line manager and project manager prepare the job descriptions, and meetings are conducted so that the staff can ask questions regarding their duties and responsibilities."	suitable
Stability of Organisational Structure	"Depending on the phases of the construction project, the organisational structure could change; however, the key personnel should not leave the site. Towards the end of the project, the number of the workforce on the site is reduced. The project manager or contract administrator could be sufficient to manage the project."	suitable
Crew Composition	"The principal contractors recruit qualified professionals to form management crew, and subcontractors are responsible for forming their crews depending on the types of the work packages. Project manager, contract administrator, site manager, project coordinators and supervisors are among the crew members of most principal contractors operating in Victoria."	suitable
Retention Plan for Experienced Personnel	"Employees can be retained by promoting them. Most companies promote their personnel and increase the salaries of the employees. The majority of contractors prefer to promote the employees internally rather than recruiting externally."	suitable
Employees Training	"The construction companies provide various training to their workers. These include first aid, health and safety, the operation of cranes, working on scaffolding, techniques of using new machinery, erecting precast concrete panels, supervision, negotiation, and mental resilience."	suitable
Skill Assessment and Evaluation	"Most principal contractors assess the competency of their subcontractors or employees. The subcontractors should have the proper training to do a specific task. For example, workers doing the scaffolding should be licensed; workers involved in dogging and rigging should also have licenses; and all construction workers need to have general construction induction cards."	suitable
Career Development	"Companies have their own culture for the career paths. Some contractors conduct an annual review to discuss issues regarding the career development of their employees. Others held meetings every six or three months to address the issues of employee development."	suitable
Financial Incentive Programs	"Financial incentives are provided to motivate employees. Pay for performance scheme is practised in some construction companies. In some projects, the bonus is given to the workers if the projects are completed less than the targeted budgets."	suitable
Social Activities	"Social activities are organised every month or two months, and sometimes on public holidays. Some of the activities include involving in marathon running, dinner, Christmas party, and team exercises such as soccer game. Most contractors allocate budget for these activities."	suitable
Non-Financial Incentive Programs	"For instance, allowing the worker to do a variety of works can be considered as a non-financial incentive. Some contractors post the names of the employees who perform well. There are also incentives systems which are not directly paid for the employees. These could be providing health care to the workers."	suitable
Exit Interviews	"When an employee is quitting, the project manager or other responsible personnel conduct an interview to understand why the employee leaves the project and to know what employment conditions to improve."	suitable

Table 5-5 Summar	v of the Interview	Results of Human	Resource Management	Practices
ruore e e buillinu		results of fighting	resource management	1 14001000

127

Clear delegation of responsibility is found to be one of the human resource management practices that could enhance productivity in multi-storey building projects. The interviewees described that the responsibility of an employee should be clarified before commencing the construction works. However, as a project progresses, employee's role could change, and the changes can be notified through project team meeting. Some participants explained that the roles and responsibilities of each employee are emailed to all members of the project team. Other respondents suggested that besides sending the written roles and responsibilities to the respective employees, conducting a meeting to clarify any issue related to employment is also vital. During the meeting, employees get the chance to ask questions regarding the job descriptions before the start of the construction works. The interviewees described that, in some situations, it might be difficult to describe the roles and responsibilities of the employees in written form; as a result, there could be a gap in delegating the duties. However, they suggested that the gap can be minimised by forming a team that comprises of employees who worked together on previous similar projects.

Maintaining the stability of the organisational structure of a project is one of the factors that positively influence the productivity of multi-storey building projects. The interviewees suggested that an employee who is a member of the project team should not leave the job if the stability of the organisational structure is to be maintained. The employees who are working on the project from its starting date have knowledge about the status of the job, the assumptions made to estimate the project cost, and other project specific information. Some respondents recommended that as experienced employees' knowledge is gold for construction projects, these workers should not leave the project. The interviewees explained that in most building projects, the project team is formed by a dedicated person at the head office who assesses employees' experience and choose the new project's team members. Some respondents explained that towards the end of a building project, the number of employees decreases. When the project is substantially completed, the company may decide to have a project manager or contract administrator and a few labourers.

The practice 'crew composition' is found to be an important human resource management practice that has the potential to improve productivity in multi-storey building projects. According to the interviews, the principal contractor's project team,

in most projects, comprises of project manager, contract administrator, project coordinators, site manager and supervisors. Some principal contractors organise the project team that consists of project manager, site engineer, general Forman and Forman. The respondents described that construction manager and design managers can also be the members of the project team, but their offices might not necessarily be on the building construction sites. The interviewees explained that subcontractors are responsible for forming crews for the trades of works they have subcontracted, and the principal contractor has no direct control over their crew compositions. However, in some circumstances, the principal contractors can request subcontractors to have a Lead Foreman within their crews and to employ experienced tradesperson and labourers. Some respondents described that, in some circumstances, the principal contractors mention the names of key personnel in the subcontract agreements to oblige subcontractors to include skilled employees in their crews.

Retention plan for experienced personnel is found to be an essential human resource management practice that can be implemented to improve productivity in multi-storey building projects. The interviewees described that the experienced personnel in construction are highly valued as they have knowledge about the construction methods, building Acts, regulations, codes and standards, and they should be retained in the company. The respondents explained that there are many regulatory changes in the building industry which could affect productivity, and knowledge about these issues is crucial. Some interviewees explained that when experienced employees who have been working on construction sites are retired, contractors assign them in the head office to retain the knowledge. The other strategy which is used by contractors to retain the key personnel is through promotion. They try to promote internally rather than recruiting externally so that they can lower the cost of recruitment and retain the knowledge which helps them to be competitive.

Providing training to the construction workers is considered to be an essential practice to improve productivity in multi-storey building projects. The interview participants suggested that employees should be trained since the local building regulations could regularly be amended, and workers need to be aware of the changes. Some respondents stated that training such as first aid, crane operation, working on scaffolding, and Occupational Health and Safety (OH&S) are commonly provided to the construction

workers. Other respondents described that some contractors offer training to improve negotiation, leadership and communication skills of their employees. According to the interviewees, the training requirement varies depending on the type of the trades of works. For example, there is a regulatory requirement which obliges tradesperson such as electricians and plumbers to complete their apprenticeship before being fully certified. The absence of regulatory requirement could result in a bad workmanship; the plumbing system might not work properly; and the users of the completed project could be electrocuted because of improper installation of electric systems. For other trades such as plastering, workers can be trained on the job.

Skill assessment and evaluation is found to be an important practice that could enhance productivity in multi-storey building projects. Some interviewees suggested that before the commencement of any construction activity, a project manager or staff who is in charge of organising the site crews should evaluate the skills of the employees and understand where their experiences fit and what their strengths and weaknesses are. Moreover, the mechanisms to improve the workers' skill should also be investigated. The respondents explained that when the works are subcontracted, the project manager has the responsibility to ensure that each subcontractor has the required competency to do a specific task. This can be done by requesting the recommendation letters from the previous employer and by investigating the evidence such as pictures of the previously executed works. The interviewees suggested that employees doing a particular task should take proper training, and they should have certificates. For instance, project managers need to check if the workers have a certificate for dogging and rigging before starting the jobs related to crane or lifting works. Some respondents explained that to understand the skill level of the subcontractors' tradesperson, the supervisors of the principal contractors monitor the works of each worker and identify any defects. If there are defects, the site supervisors report to the project manager, and the manager of the subcontracting firm is notified to make corrective action.

Career development plan is another human resource management practice which influences productivity in multi-storey building projects. Some respondents described that career development is about having a vision for the workers and discussing the career paths which the employees would prefer. They explained that in most building projects, a formal meeting is held every six months or a year to discuss the issues

related to career development. Some interviewees suggested that it is preferable to promote an existing employee to a new position than employing a new worker because the new employee might take a longer time to adapt to the company culture. Thus, there could be a loss of productivity until the new employee learns the company procedures and develops effective communication with the existing project team members. Besides enhancing productivity, career development plan can also save costs associated with the recruitment of new employees.

The practice 'financial incentive programs' is found to be one of the essential human resource management practices that could improve productivity in multi-storey building projects. The interview participants described that financial incentives are provided to motivate employees, and the incentives vary based on the employees' roles. Some interviewees explained that certain contractors promote their employees and increase payment depending on the level of responsibility of an employee. In some circumstances, the bonus is given to workers if the actual cost of a project is less than the targeted budget. Moreover, an increase in payment based on the performance of the employee or 'pay for performance scheme' is practised by some construction companies. Other interviewees suggested that financial incentives are more important for subcontractors than principal contractors. They explained that if subcontractors offer good incentives, the chance of getting skilled labourer increases, and the more skilled the employees, the better chance of winning the tender is.

Organising social activities as means of motivating project team is found to be one of the factors which positively influence productivity in multi-storey building construction projects. Some respondents described that in order to keep the project team fresh and excited, social activities every month or two months or on public holidays are organised by contractors. Some of the activities include involving in marathon running, bowling, dinner, Christmas party, and soccer games. The respondents explained that some contractors allocate budget for every project and plan the event dates.

Non-financial incentive programs are among the human resource management practices that could enhance productivity in multi-storey building projects. Some interviewees explained that non-financial benefits are very important for principal contractors in particular, and allowing employees to do a variety of works or enrichment of the job is one of the effective non-financial incentive systems. Other respondents described that

contractors post names and photos of the well-performing employees on the notice board at head office and project sites. They also announce their names on a team meeting and media functions. The interviewees stated that there are other non-financial incentive schemes such as health care services which the employees do not pay for.

The practice 'exit interviews' is found to be an essential practice that positively influences productivity in multi-storey building construction projects. The interviewees described that conducting an exit interview helps a contractor to understand why an employee leaves and what the firm should do to change the employee's mind. Other issues which the contractor needs to improve could also be identified via exit interviews. Some respondents suggested that the interview should be conducted at both project site and head office levels. Other interviewees explained that exit interviews and written feedbacks to a contractor are vital provided that the contractor takes corrective actions. According to the respondents, in some circumstances, some contractors disregard all the comments given by the employee during the exit interview. They consider that the employee was not good and the comments are not relevant.

Finally, eleven human resource management practices are identified, and the validated conceptual framework of human resource management practices is shown in Figure 5-7.



Figure 5-7 Validated Conceptual Framework of Human Resource Management Practices and Productivity

132

5.8 Identification of Safety and Health Practices Influencing Productivity in Multi-Storey Building Projects

'Safety and health policy,' 'safety and health plan,' 'hazard analysis,' 'safe work method statement,' 'toolbox safety meetings,' 'housekeeping,' and 'safety and health training' are found to be the seven important safety and health practices that have the potential to enhance productivity in multi-storey building projects. However, the practice 'drugs and alcohol testing program' is considered to be not vital to improve productivity in multi-storey building projects in the context of Victoria State, Australia. Summary of the interview results of safety and health practices is presented in Table 5-6.

The preparation of the safety and health policy for a certain multi-storey building project is found to be an essential practice that could improve productivity. According to the interviewees, most contractors have safety and health policy at company and project levels. The respondents explained that some construction companies have established the minimum requirements that help them to manage the risks at their projects and use the term Global Minimum Requirements (GMRs) for the minimum safety standards. Other participants explained that some contractors have an accredited safety and health management systems. The interviewees explained that the project specific safety policies are usually displayed at places where most employees can see. Moreover, the construction workers are informed during site induction phase, and the copies of the safety and health policies are provided to subcontractors.

Safety and health plan is found to be one of the safety and health practices that have the potential to improve productivity in multi-storey building projects. Some interviewees explained that contractors plan to reduce the number of accidents on building construction sites as it is difficult to avoid accidents completely. The respondents explained that certain contractors have developed their own methodology to reduce the accidents rates by focusing on engineering controlling techniques than on the administrative controlling techniques. They explained that the administrative controlling technique, loads are calculated, and appropriate materials are recommended to ensure the safety of workers. According to the interviews, in some building projects, there are rewards for employees who achieve the zero accident plan.

Other respondents described that, in some projects, contractors plan not to exceed a specified value of the lost-time injury frequency rate (LTIFR). The lost-time injury frequency rate (LTIFR) of the previous projects is computed and used as a benchmark to prepare the target LTIFR for new projects.

Table 5-6 Summary	of the Interview	Results of Safety	and Health Practices
-------------------	------------------	-------------------	----------------------

Safety and Health Practices	Summary of the Interview Responses	Conclusion
Safety and Health Policy	"Most contractors have safety and health policy at company and project levels. Some contractors have accredited safety policy. The project specific policies are clearly displayed on sites, and workers are informed during the induction process. Some principal contractors also provide copies of their safety policies to their subcontractors."	suitable
Safety and Health Plan	"The maximum number of lost-time injuries are planned for new building project based on the number of accidents occurred in the past projects. The lost-time injury frequency rate (LTIFR) is computed for the previous projects and used as a benchmark to prepare the target LTIFR for new projects."	suitable
Hazards Analysis	"Workers give answers to the safety-related questions which are distributed every month or two months to identify the potential hazards. There are also workshops to identify hazards which are not considered during task safety analysis."	suitable
Safe Work Method Statement	"In some projects, subcontractors prepare a Safe Work Method Statement (SWMS) for each task, and the principal contractors review and approve the SWMS. Safety manager monitors the progress of subcontractors on a daily basis and checks if the works are executed as per the approved SWMS. Safety audits are also conducted regularly, for instance, every six months."	suitable
Toolbox Safety Meetings	"The practice of conducting safety meetings varies from project to project. On some sites, there is a regular weekly or fortnightly toolbox safety meeting. On other projects, the meeting is conducted when high-risk works are executed."	suitable
Housekeeping	"The construction sites need to be clean; otherwise, accidents could occur and productivity will be lost. In most projects, two big bins are provided by a principal contractor and subcontractors place the waste materials in the bins. The subcontractors are required to have the green bins or medium size bins. The principal contractor needs to make sure that the statement 'a subcontractor should clean its working area' is included in the scope of works of each subcontractor."	suitable
Safety and Health Training	"Every worker has to take a minimum of general construction induction training and has to have a white card or red card or other certificates. Graduates and cadets should take about six months training regarding safety issues while working on different sites. Training on the use of new equipment and removal of asbestos is also conducted."	suitable
Drugs and Alcohol Testing Program	"Drugs and alcohol testing programs are not common in the building industry. The issue of alcohol and drugs are discussed during the induction phase, but there is no formal testing program. Generally, on most building construction jobs, drugs and alcohol testing is not conducted since it is not included in the employment agreement."	not suitable

134

Hazard analysis is another safety and health practice that positively influences productivity in multi-storey building construction projects. The interview participants suggested that the potential hazards that are not identified during the preparation of the safe work method statement (SWMS) should be analysed separately. The interviewees described that conducting hazard analysis is also essential for 'works of minor nature' or works which the safety regulations do not oblige contractors to prepare SWMS. The respondents explained that to identify the potential hazards, the construction workers give answers to the safety-related questions which are distributed every month or two months. Workshops are also organised to identify hazards.

The practice 'Safe Work Method Statement (SWMS)' is also found to be one of the safety and health practices that could enhance productivity in multi-storey building projects. The interviewees explained that SWMS is prepared by conducting safety analysis for each task classified under high-risk construction works. To prepare the SWMS, the activities involved and the procedures for execution of the task are identified, potential hazards are listed, and mechanisms for minimising or eliminating the risks are proposed. Finally, the person who monitors the compliance with the SWMS is assigned. Some interviewees described that, in some projects, SWMS is prepared by subcontractors and approved by the principal contractor. Other respondents explained that there are safety representatives at project sites and head office who oversee the Occupational Health and Safety. The safety audit is conducted every six months or year in the presence of a safety representative from the head office.

Toolbox safety meeting is found to be one of the safety and health practices that positively influence productivity in multi-storey building construction projects. Some interview participants recommended that there should be regular on-site meetings to discuss various issues related to safety. The meetings help the project team to share safety information and to solve problems jointly. Some interviewees explained that in building construction projects in Victoria, toolbox safety meetings are usually conducted every week or two weeks. Other respondents described that on some building projects, tool box meeting is not regularly scheduled. It is conducted when executing risky tasks that require workers to discuss the installation methods of the tasks. For example, lifting of heavy precast panels requires safety meeting. The participants suggested that besides regular toolbox meetings, the daily brief or pre-start induction is

also essential. The daily orientation helps the workers to know what is happening on that particular date.

The practice 'housekeeping' is found to be one of the important safety and health practices that have the potential to improve productivity in multi-storey building projects. The interviewees explained that the lack of cleanliness of project sites can result in accidents and loss of productivity. The respondents described that, in most building projects in Victoria, two large bins are provided by principal contractors, and subcontractors put the waste materials in the bins. In other projects, subcontractors are entirely responsible for managing their wastes. However, the responsibilities of the subcontractors need to be specified in the scope of works or in the subcontract agreement. Some interviewees described that the principal contractor's supervisor can stop the subcontractor's personnel from working on a particular site if the housekeeping is not good. Other respondents explained that to keep the site clean and safe, most construction materials are not stored on a building site.

The practice 'safety and health training' is considered to be one of the safety and health practices that positively influence productivity in multi-storey building projects. The interviewees described that safety training and inductions need to be conducted. Since regulations might change, the latest information must be provided to the construction workers. Some interview participants explained that every worker should have a minimum of white card or red card or other evidence which shows the completion of the general construction induction training. Moreover, the construction workers should be given project specific information prior to commencing the construction activities. The interviewees described that some principal contractors require their employees to have Certificate 3 in Work Health and Safety. First aid, asbestos removal, and new equipment operation are some of the training provided to the building construction workers. Some respondents suggested that graduates and cadets should be trained for about six months regarding the safety and health issues.

Finally, seven safety and health practices are found to have the potential to improve productivity in multi-storey building projects, and the validated frame of safety and health practices shown in Figure 5-8.

CHAPTER 5



Figure 5-8 Validated Conceptual Framework of Safety and Health Practices and Productivity

The practice 'drugs and alcohol testing program' is found to be not suitable to improve productivity in multi-storey building projects in the context of Victoria State, Australia. Some interview participants explained that drug and alcohol testing is not conducted in most building construction projects in Victoria, but the issue is getting attention since workers come to a building site after drinking alcohol, particularly on Saturdays. Other respondents described that the use of alcohol and drugs is addressed during the induction phase, but there is no formal testing program. According to the interview, the safety representatives can figure out the drunken workers and prohibit them from commencing the construction works. Other interviewees described that drugs and alcohol testing is not conducted in most building construction projects since it is not included in the employment agreement.

5.9 Summary

After analysing the qualitative data, part of Objective 1 or "to identify construction management practices that have the potential to enhance productivity in multi-storey building construction projects in the context of Victoria State, Australia" is achieved. Moreover, part of Objective 2 or "to refine the scoring tools for measuring, planning, monitoring and evaluating the management practices that could improve productivity in multi-storey building construction projects" is met. Forty-seven context specific construction management practices have been identified, and the validated conceptual frameworks of the practices have been developed. The construction management practices, three construction equipment management practices, ten management practices related to

construction methods, ten preconstruction phase management practices, eleven human resource management practices, and seven safety and health practices. Finally, the 47 independent variables (the construction management practices) indicated in the validated conceptual frameworks (Figure 5-3, Figure 5-4, Figure 5-5, Figure 5-6, Figure 5-7, and Figure 5-8) are used as input to the second phase of the research which comprises of quantitative data collection and analysis (please refer Part I of the Questionnaire in Appendix-1 for the list of the practices). Furthermore, the levels of implementation of the 47 practices were refined from the existing tools based on the findings of the interviews, and the unweighted scoring tools for the practices in the context of multi-storey building projects were prepared. The unweighted scoring tools were validated during Phase I of the research and used as an instrument to collect quantitative data (please refer Part II-C of the Questionnaire in Appendix-1 for the validated instruments or unweighted scoring tools). The preliminary quantitative data analysis is presented in the next chapter.

CHAPTER SIX

6 PRELIMINARY QUANTITATIVE DATA ANALYSIS

6.1 Introduction

This chapter presents the preliminary analysis of the quantitative data. The background of the respondents involved in Phase II of the research and the profiles of the companies are described in Section 6.2. The productivity factors of the projects are computed, and the results are shown in Section 6.3. In Section 6.4, the descriptive analyses of the construction management practices' data are conducted. Thus, the descriptive statistics of the construction materials management practices, the construction equipment management practices, the management practices related to construction methods, the preconstruction phase management practices, the human resource management practices, and the safety and health practices are presented in Section 6.4. Finally, the summary of the chapter is provided in Section 6.5.

6.2 Profiles of the Companies and the Respondents

The profiles of the questionnaire survey participants are indicated in Figure 6-1. Accordingly, the majority of the participants (46%) are project managers with an average of 13 years of experience in building construction.



Figure 6-1 Profiles of the Questionnaire Survey Participants

About 15% of the respondents are contract administrators having an average of 12 years of work experience;13% of the survey participants are project coordinators with an average of nine years of experience; 10% of the participants are construction managers having an average of 29 years of experience; projects engineers with an average of eight

years of experience comprises 5% of the total respondents; site managers having an average of 22 years of experience also comprises 5% of the total respondents; 3% of the participants are operation managers having 21 years of experience; and 3% of the total respondents consist of construction directors with 20 years of experience (Figure 6-1).

Figure 6-2 presents the distributions of the contractors by the company sizes. Accordingly, 59% of them have employees greater than 200, and 41% of them have between 20 and 199 employees. This research will explore if larger companies have better management practices than smaller companies.



Figure 6-2 Distributions of Contractors by Company Sizes

The annual turnover of the contractors in the year 2014/15 is indicated in Figure 6-3. Accordingly, 38% of the contractors have the annual turnover between 100 and 500 million Australian Dollars (AUD), 29% of them have the annual turnover between 501 and 1000 million AUD, 13% of the contractors have between 1001 and 2000 million AUD, and 21% of them have the annual turnover exceeding 2000 million AUD. This study will investigate whether companies with high annual turnover also implement higher levels of the construction management practices.



Figure 6-3 Distributions of Contractors by Annual Turnover

The experience of the contractors in the construction industry is presented in Figure 6-4. Accordingly, 29% of the contractors have experience between 10 to 20 years; 21% of them have between 21 and 40 years; 32% of the construction companies have between

140

41 and 60 years; about 13% of them have between 61 and 80 years; and 5% of them have greater than 80 years of experience. This research will explore if the implementation level of the management practices is associated with the experience of the contractors.



Figure 6-4 Distributions of the Contractors by Years of Experience

6.3 Characteristics of the Projects and Computation of Productivity Factors

Before calculating the projects' productivity factors, the percentage distributions of their costs are computed, and the results are presented in Figure 6-5. Accordingly, 36% of the projects have project costs between 8 and 19 million Australian Dollar (AUD), 21% of them have project costs between 20 to 39 million AUD, 18% of them have project costs between 40 and 100 million AUD,13% of them have project costs ranging from 101 to 150 million AUD, 8% of them have project costs between 151 and 200 million AUD and 5% of them have project costs greater than 200 million AUD.



Figure 6-5 Percentage Distributions of the Costs of the Building Projects

In Figure 6-6, the proportions of the building projects which incurred project delays are shown. Accordingly, about 54% of the projects are completed on and before the planned completion date; 21% of them incurred project delays ranging from 1day to 50 days; 10% of them have project delays between 51 and 100 days; about 5% of them incurred project delays ranging from 101 to 149 days; and 10% of them have project delays exceeding 150 days.



Figure 6-6 Percentage Distributions of the Project Delays

The productivity factors (PFs) of the projects are computed using the project durations, and project costs and the results are shown in Table 6.1. The minimum PF value is 0.59, the maximum PF value is 1.14, and the average PF value is found to be 0.95. Based on their PFs, the projects are grouped and analysed. The proportions of projects with PFs ranging from 0.5 to 0.70, 0.71 to 0.90, 0.91 to 1.10, and exceeding 1.10 are presented in Figure 6-7. Accordingly, 8% of the projects have PFs ranging from 0.50 to 0.70, 28% of the projects have PFs ranging from 0.50 to 0.70, 28% of the projects have PFs between 0.71 and 0.90, 54% of the projects have productivity factors ranging from 0.91 to 1.10, and 10% of the projects have PFs exceeding 1.10.



An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

Project Code	Project Start Date	Planned completion	Actual completion	Project Cost (\$AUD	planned input	actual input	Actual productivity	Planned Productivitv	Productivity Factor (PF)
	(a)	date (b)	date (d)	Million) (f)	g=(b-a)	h=(d-a)	i = (f)/(h)	j=(f)/(g)	(i)/(j)
R1	1-Mar-13	1-Jun-14	1-Jun-14	30	457	457	0.0656	0.0656	1.00
R2	1-Jun-11	1-Mar-14	1-Aug-14	100	1004	1157	0.0864	0.0996	0.87
R3	29-Jan-15	16-Oct-15	12-Oct-15	106	260	256	0.4141	0.4077	1.02
R4	1-Aug-14	10-Feb-16	1-Feb-16	42.5	558	549	0.0774	0.0762	1.02
R5	13-Jan-14	4-Dec-14	2-Dec-14	28	325	323	0.0867	0.0862	1.01
R6	1-Feb-12	1-Jun-15	1-Mar-16	110	1216	1490	0.0738	0.0905	0.82
R7	1-Oct-14	1-Dec-15	1-Apr-16	50	426	548	0.0912	0.1174	0.78
R8	1-May-14	1-Nov-15	1-Oct-15	160	549	518	0.3089	0.2914	1.06
R9	29-May-12	30-Aug-14	30-Aug-14	150	823	823	0.1823	0.1823	1.00
R10	1-May-14	1-Oct-15	1-Aug-15	32	518	457	0.0700	0.0618	1.13
R11	1-Jan-10	29-Oct-10	23-May-11	17	301	507	0.0335	0.0565	0.59
R12	1-Sep-14	31-Dec-14	1-Feb-15	20	121	153	0.1307	0.1653	0.79
R13	1-Jun-14	1-Sep-15	1-Aug-15	240	457	426	0.5634	0.5252	1.07
R14	13-Jan-14	2-Dec-14	12-Feb-15	25	323	395	0.0633	0.0774	0.82
R15	19-Oct-12	17-Dec-13	14-Feb-14	17	424	483	0.0352	0.0401	0.88
R16	1-Jan-12	1-Oct-13	1-Oct-13	15	639	639	0.0235	0.0235	1.00
R17	1-Oct-13	1-Apr-15	1-Jan-16	175	547	822	0.2129	0.3199	0.67
R18	1-Apr-12	1-Sep-13	1-Sep-13	14	518	518	0.0270	0.0270	1.00
R19	1-Apr-15	1-May-16	31-Mar-16	15	396	365	0.0411	0.0379	1.08
R20	27-May-14	27-Nov-14	22-Dec-14	13.6	184	209	0.0651	0.0739	0.88
R21	28-Apr-15	3-Jan-16	13-Jan-16	10.6	250	260	0.0408	0.0424	0.96
R22	1-Jun-13	1-May-15	1-May-15	20	699	699	0.0286	0.0286	1.00
R23	1-Jul-09	1-Feb-12	1-Jun-12	200	945	1066	0.1876	0.2116	0.89
R24	1-May-13	1-May-14	1-Jun-14	14.5	365	396	0.0366	0.0397	0.92
R25	1-Aug-12	1-Jun-14	1-Jul-14	18.5	669	699	0.0265	0.0277	0.96
R26	28-May-15	23-Dec-15	26-Feb-16	10.8	209	274	0.0394	0.0517	0.76
R27	1-Mar-14	1-May-16	1-Mar-16	105	792	731	0.1436	0.1326	1.08
R28	1-Jan-15	15-Aug-15	11-Jan-16	12	226	375	0.0320	0.0531	0.60
R29	1-Apr-13	1-Apr-14	1-May-14	30	365	395	0.0759	0.0822	0.92
R30	1-Jan-14	1-Aug-15	1-Jun-15	70	577	516	0.1357	0.1213	1.12
R31	7-Mar-11	24-Feb-12	30-Mar-12	8.2	354	389	0.0211	0.0232	0.91
R32	1-Feb-12	1-Jun-13	1-Apr-13	80	486	425	0.1882	0.1646	1.14
R33	30-May-15	24-Mar-16	27-Apr-16	13.3	299	333	0.0399	0.0445	0.90
R34	6-Sep-14	16-May-16	1-Apr-16	197	618	573	0.3438	0.3188	1.08
R35	1-Jul-12	1-Jan-14	1-Dec-13	15	549	518	0.0290	0.0273	1.06
R36	1-Jun-14	25-Dec-14	25-Dec-14	28	207	207	0.1353	0.1353	1.00
R37	1-Oct-13	1-Dec-14	1-Nov-14	59	426	396	0.1490	0.1385	1.08
R38	1-Apr-14	1-Mar-16	1-Mar-16	115	700	700	0.1643	0.1643	1.00
R39	4-Dec-12	30-Oct-14	28-Aug-14	40	695	632	0.0633	0.0576	1.10

 Table 6-1 Productivity Factors of the Projects

6.4 Descriptive Analysis of the Construction Management Practices

The descriptive data analysis of the 47 construction management practices revealed that the practice well-defined scope of works has the highest mean value of 4.79, and the practice model development has the lowest mean value of 2.69 (Table 6-2). The descriptive statistics of all the management practices are shown in Table 6-2, and the raw data used in the computation is presented in Appendix-2.

Table 6-2 Descriptive	Statistics of the	Construction	Management	Practices
1			U	

	N7	D	10	17	М	ean	6(1 D	V
Construction Management Practices	Ν	Range	Min	Max	Statistic	Std. Error	Std. Dev	Var.
Construction Materials Management Practices								
(1) Procurement Plans for Materials	39	4	1	5	4.23	0.13	0.84	0.71
(2) Long-Lead Materials Identification	39	3	2	5	4.59	0.11	0.68	0.45
(3) Materials Status Database	39	4	1	5	3.31	0.16	0.98	0.96
(4) Materials Delivery Schedule	39	3	2	5	3.74	0.15	0.94	0.88
(5) Material Inspection Process	39	4	1	5	3.49	0.15	0.94	0.89
(6) Materials Inspection Team	39	3	1	4	2.85	0.15	0.96	0.92
Construction Equipment Management Practices								
(1) Procurement Plan for Construction Equipment	39	4	1	5	3.46	0.16	0.97	0.94
(2) Construction Equipment Productivity Analysis	39	4	1	5	3.03	0.16	0.99	0.97
(3) Construction Equipment Maintenance	39	4	1	5	3.95	0.18	1.12	1.26
Management Practices Related to Construction Methods								
(1) Integrated Schedules	39	4	1	5	3.28	0.14	0.89	0.79
(2) Work Schedule Strategies	39	3	2	5	3.67	0.10	0.66	0.44
(3) Schedule Control	39	3	2	5	3.87	0.12	0.77	0.59
(4) Dynamic Site Layout Plan	39	2	3	5	4.08	0.11	0.70	0.49
(5) Traffic Control Plan	39	2	3	5	4.26	0.11	0.72	0.51
(6) Site Security Plan	39	3	2	5	3.67	0.13	0.81	0.65
(7) Machinery Positioning Strategy	39	3	2	5	4.08	0.14	0.93	0.86
(8) Project Start -up Plan	39	3	2	5	4.26	0.12	0.75	0.56
(9) Project Completion Plan	39	4	1	5	4.13	0.15	0.95	0.9
(10) Innovations and New Technologies	39	4	1	5	3.18	0.14	0.89	0.78
Preconstruction Phase Management Practices			-	-				
(1) Short Interval Plan	39	3	2	5	4 38	0.11	0.71	0.51
(2) Well-Defined Scope of Work	39	3	2	5	4 79	0.09	0.57	0.33
(3) Use of Software in Planning Work Packages	39	3	2	5	3.82	0.15	0.94	0.89
(4) Dedicated Planner	39	4	1	5	3.05	0.18	1 10	1.21
(5) Construction Work Packages	39	2	3	5	4 28	0.11	0.69	0.47
(6) Buildability Review	39	2	3	5	4 38	0.09	0.59	0.35
(7) Utilities Alignment	39	3	2	5	3 79	0.12	0.77	0.59
(8) Contract Types	39	3	2	5	3.87	0.12	0.80	0.64
(9) Model Development	39	4	1	5	2.69	0.17	1.03	1.06
(10) Regulatory Requirement	30	3	2	5	4.26	0.14	0.85	0.72
Human Desember Management Desetions	39	J	2	5	4.20	0.14	0.85	0.72
(1) Graw Composition	20	4	1	5	2.95	0.16	0.00	0.08
(1) Crew Composition (2) Skill Assassment and Evoluation	39	4	1	5	3.65	0.10	1.00	1.01
(2) Skill Assessment and Evaluation (3) Employees Training	39	4	1	5	3.09	0.16	0.00	0.00
(d) Career Development	39	4	1	5	3.74	0.10	0.99	0.99
(4) Carcer Development	39	4	1	5	2.04	0.15	1.02	1.04
(6) Financial Incentive Programs	39	4	1	5	3.20	0.10	1.02	1.04
(b) Financial incentive Programs	39	4	1	5	2.30	0.17	0.97	0.76
(1) Social Activities	39	2	2	5	3.30	0.14	0.87	0.76
(b) Stability of Organizational Structure	39	3	2	5	4.08	0.12	0.74	0.55
(9) Clear Delegation of Responsibility	39	3	2	5	4.18	0.13	0.79	0.65
(10) Retention Plan for Experienced Personnel	39	3	2	5	3.82	0.13	0.82	0.68
(11) Exit Interviews	39	4	I	5	2.87	0.17	1.03	1.06
Safety and Health Practices			-					
(1) House Keeping	39	2	3	5	4.44	0.09	0.60	0.358
(2) Safety and Health Policy	39	2	3	5	4.74	0.08	0.50	0.248
(3) Safety and Health Plan	39	2	3	5	4.62	0.10	0.63	0.401
(4) Safe Work Method Statement	39	2	3	5	4.56	0.11	0.68	0.463
(5) Hazards Analysis	39	2	3	5	4.62	0.11	0.67	0.453
(6) Safety and Health Training	39	2	3	5	4.44	0.13	0.79	0.621
(7) Toolbox Safety Meetings	39	2	3	5	4.54	0.11	0.68	0.466

The median values of the 47 construction management practices are computed, and the distributions of the data are indicated in Figure 6-8. Box plots are used to display the results since they are suitable to compare the medians of the practices. The practices long-lead materials identification, well-defined scope of works, safety and health policy,

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

safety and health plan, safe work method statement, hazards analysis, safety and health training, and toolbox safety meetings have the highest median values of 5.



Figure 6-8 Box-plots of the Construction Management Practices

The percentage distributions of the respondents who selected 1 for not important, 2 for slightly important, 3 for somewhat important, 4 for very important, and 5 for extremely important practice are shown in Table 6-3. Accordingly, most respondents (85%) perceived the implementation of the practice well-defined scope of works as 'extremely important' (a response scale of 5) to improve productivity in multi-storey building construction projects. Overall, the majority of the respondents (36%) rated that the 47 construction management practices are very important (a response scale of 4) to improve productivity in multi-storey building verifies the finding obtained in Phase I of this research. The qualitative data analyses in Phase I revealed that the 47 construction management practices have the potential to improve productivity in multi-storey building projects in the context of Victoria State, Australia.

Table 6-3	Distribution of th	e Responses	of the (Construction	Management	Practices	on a	Scale
	of 1 to 5							

	Not	Slightly	Somewhat	Very	Extremely
Construction Management Practices	Important	important	important	important	important
	(1)	(2)	(3)	(4)	(5)
Construction Materials Management Practices (CMMP)					
Procurement Plans for Materials	3%	0%	10%	46%	41%
Long-Lead Materials Identification	0%	3%	3%	28%	67%
Materials Status Database	5%	15%	28%	46%	5%
Materials Delivery Schedule	0%	10%	28%	38%	23%
Material Inspection Process	3%	8%	44%	31%	15%
Materials Inspection Team	10%	23%	38%	28%	0%
Average of CMMP	4%	10%	25%	36%	25%
Construction Equipment Management Practices (CEMP)					
Procurement Plan for Construction Equipment	3%	13%	33%	38%	13%
Construction Equipment Productivity Analysis	8%	15%	51%	18%	8%
Construction Equipment Maintenance	3%	10%	18%	28%	41%
Average of CEMP	5%	13%	34%	28%	21%
Management Practices Related to Construction Methods (MPRCM)					
Integrated Schedules	5%	10%	38%	44%	3%
Work Schedule Strategies	0%	3%	36%	54%	8%
Schedule Control	0%	3%	28%	51%	18%
Dynamic Site Layout Plan	0%	0%	21%	51%	28%
Traffic Control Plan	0%	0%	15%	44%	41%
Site Security Plan	0%	5%	38%	41%	15%
Machinery Positioning Strategy	0%	10%	8%	44%	38%
Project Start - up Plan	0%	3%	10%	46%	41%
Project Completion Plan	3%	3%	15%	38%	41%
Innovations and New Technologies	3%	18%	44%	31%	5%
Average of MPRCM	1%	6%	2.5%	44%	24%
Pre-Construction Phase Management Practices (PCPMP)				/ •	
Short Interval Plan	0%	3%	5%	44%	49%
Well-Defined Scope of Work	0%	3%	0%	13%	85%
Use of Software in Planning Work Packages	0%	8%	31%	33%	28%
Dedicated Planner	10%	15%	1/%	21%	10%
Construction Work Packages	1070	0%	13%	2170 //6%	/10/0
Buildshility Review	0%	0%	5%	51%	41%
Utilities Alionment	0%	3%	33%	/6%	18%
Contract Types	0%	50/	230/	-1070 510/	210/
Model Davelopment	120/	200/	2370 410/	120/	21/0 50/
Perulatory Dequirement	1370	2070 50/-	41/0	280/	J70 16%
Average of PCPMP	20/		210/0	36%	4070
Human Decourse Management Directions (HDMD)	270	//0	21/0	5070	5570
Crew Composition	20/	Q0/-	180/	160/-	26%
Skill Assessment and Evaluation	970 80/	30/	15%	4070 62%	130/
Skii Assessment and Evaluation	070 50/	570	1370	5.40/	1370
Employees Training	50/	50/	18%	5.4%	1870
Van Einengiel Incentive Decomputer	J 70	1.00/	2070	260/	070
Financial Incentive Programs	870	10%	38%	30%0 410/	870
Financial Incentive Programs	8%0	10%	51%	41%	10%
Social Activities	0%	15%	44%	51%	10%
	0%	3%	15%	54%	28%
Clear Delegation of Responsibility	0%	5%	15%	44%	38%
Retention Plan for Experienced Personnel	0%	5%	28%	46%	21%
Exit Interviews	8%	31%	33%	25%	5%
Average of HRMP	4%	9%	26%	45%	1/%
Satety and Health Practices (SHP)	00/	00/	FO (1001	4007
Housekeeping	0%	0%	5%	46%	49%
Satety and Health Policy	0%	0%	3%	21%	11%
Satety and Health Plan	0%	0%	8%	23%	69%
Sate Work Method Statement	0%	0%	10%	23%	67%
Hazards Analysis	0%	0%	10%	18%	72%
Safety and Health Training	0%	0%	18%	21%	62%
Toolbox Safety Meetings	0%	0%	10%	26%	64%
Average of SHP	0%	0%	9%	25%	66%
Average of the Construction Management Practices (CMP)	3%	7%	23%	36%	31%

146

Descriptive Statistics of the Construction Materials Management Practices

As shown in Table 6-2 above, under the construction materials management category, the practice long-lead materials identification has a mean of 4.59 which is the maximum value, and the practice materials inspection team has the mean of 2.85 which is the minimum value. Moreover, 'long-lead materials identification' has the smallest standard deviation (SD=0.68) and smallest standard error (0.11) which indicates that there is low variability among the responses to this practice. The practice 'materials status database' has the highest standard deviation of 0.98 which shows that the perception of the respondents regarding the importance of this practice to enhance productivity in multi-storey building projects is different.

The median values of the six construction materials management practices are computed, and the distributions of the data are indicated in Figure 6-8 above. Accordingly, long-lead materials identification has the highest median value of 5; materials status database and materials delivery schedule have both median values of 4, and materials inspection process and materials inspection team have both the lowest median values of 3.

The percentage distributions of the respondents regarding the importance of the six construction materials management practices are shown in Table 6-3 above. Accordingly, in this category, most respondents (67%) rated the long-lead materials identification as 'extremely important' practice for enhancing productivity in multi-storey building projects, and none of them responded the practice as 'not important.' About 46% of the participants responded that 'materials status database' and 'procurement plans for materials' are very important (response scale of 4) practices to improve productivity in multi-storey building projects. Forty-four percent of the respondents rated the practice materials inspection process as somewhat important (response scale of 3). None of the participants responded materials inspection team as extremely important. Overall, most of the participants (36%) rated the six construction materials management practices are important (Table 6-3). This finding confirms the finding of the first phase of the study which also revealed that the six construction materials management practices are important to improve productivity in multi-storey building projects.

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

Descriptive Statistics of the Construction Equipment Management Practices

In the construction equipment management practices category, the practice construction equipment maintenance has the highest mean value of 3.95 and construction equipment productivity analysis has the lowest mean value of 3.03 (Table 6-2). Procurement plan for construction equipment has a mean value of 3.46. The practice construction equipment maintenance has the highest standard deviation (1.12) which indicates the high variabilities among the responses to this practice. The practice procurement plan for construction equipment has the lowest variance (0.94) which shows low variability among the data.

The distribution of the construction equipment management practices data is shown using the box plots indicated in Figure 6-8 above. Accordingly, the practice construction equipment productivity analysis has the lowest median value of 3, and the practices 'procurement plan for construction equipment' and 'construction equipment maintenance' have both median values of 4.

The percentage distributions of responses (on a scale of 1 to 5) of the construction equipment management practices are presented in Table 6-3 above. Accordingly, in the construction equipment management practices category, the majority of the respondents (41%) rated construction equipment maintenance is 'extremely important' (response scale of 5) to improve productivity in multi-storey building construction projects. Thirty-eight percent of the participants rated the practice procurement plan for construction equipment as 'very important' (response scale of 4). About 51% of the participants responded that construction equipment productivity analysis is 'somewhat important' (response scale of 3) for improving productivity. As shown in Table 6-3, the overall average percentage distribution for 'not important,' 'slightly important,' 'somewhat important,' 'very important' and 'extremely important' are 4%, 13%, 34%, 28%, and 21% respectively. Accordingly, the majority of the respondents (34%) rated the three construction equipment management practices as 'somewhat important' (response scale of 3) to improve productivity in building projects.

Descriptive Statistics of the Management Practices Related to Construction Methods

The highest mean value of 4.26 is obtained for practices 'traffic control plan' and 'project start-up plan' in the category of management practices related to construction methods (Table 6-2). The practice 'innovations and new technologies' has the lowest

mean value of 3.18. The practice 'project completion plan' has the highest variance of 0.90 which indicates the high variability of the data, and the practice 'work schedule strategies' has the lowest variance of 0.44 which implies the variability among the perception of the respondents regarding the importance of this practice is low.

The median values of the ten management practices related to construction methods are computed, and the results are shown in Figure 6-8. Accordingly, the median values of Work Schedule Strategies, Schedule Control, Dynamic Site Layout Plan, Traffic Control Plan, Site Security Plan, Machinery Positioning Strategy, Project Start-up Plan and Project Completion Plan are equal to 4. 'Integrated Schedules' and 'Innovations and New Technologies' have both minimum median values of 3.

The percentage distribution of the responses (on a scale of 1 to 5) of the ten management practices related to construction methods is presented in Table 6-3 above. Accordingly, about 44% of the respondents rated the practice integrated schedule as 'very important,' and 3% of them responded as 'extremely important.' For the practice work schedule strategies, 54% of the participants responded that the practice is very important, and none of them rated the practice as 'not important' (response scale of 1). All respondents did not rate schedule control, dynamic site layout plan, traffic control plan, site security plan, machinery positioning strategy, and project start-up plan as 'not important' to improve the productivity of building projects. Overall, most respondents (44%) rated that the management practices related to construction methods are very important to improve productivity in multi-storey building projects. This result confirms the findings of Phase I of the study regarding the suitability of the management practices related to construction projects. The qualitative data analysis in the previous chapter revealed that the ten practices related to construction methods are important for improving productivity in building projects.

Descriptive Statistics of the Pre-Construction Phase Management Practices

The practice well-defined scope of works has the highest mean value of 4.79, and the practice model development has the lowest mean value of 2.69 in the pre-construction phase management practices category (Table 6-2). The practice dedicated planner has a highest standard deviation of 1.10 which indicates that respondents have different views about the importance of this practice. 'Well-defined scope of works' has the lowest

standard deviation of 0.57 which shows that the participants have similar views on the importance of adequately defining the scope of works to improve productivity.

The median values and the distribution of the ten preconstruction phase management practices' data are plotted using box and whisker diagram (Figure 6-8 above). Accordingly, the practice well-defined scope of works has the highest median value of 5. Dedicated planner and model development have the lowest median values of 3. As shown in the figure, the practice dedicated planner has the longest box plot indicating that the range of the responses to this practice is high. Short Interval Plan, Use of Software in Planning Work Packages, Construction Work Packages, Buildability Review, Utilities Alignment, Contract Types, and Regulatory Requirement have equal median values of 4.

The percentage distribution of the responses for each preconstruction phase management practice is shown in Table 6-3 above. Accordingly, the survey participants rated well-defined scope of work (85%), short-interval plan (49%), regulatory requirement (46%), and buildability review (44%) as extremely important practice to improve productivity in multi-storey building construction projects. The practice model development has been rated as 'extremely important' by a few respondents (5%). The practices 'dedicated planner' and 'model development' are perceived as not important practices for improving productivity in building projects by 10% and 13% of the respondents respectively (Table 6-3). Overall, the majority of the respondents (36%) perceive the preconstruction phase management practices as very important (response scale of 4). This finding verifies the finding of Phase I of the research in which the ten preconstruction phase management practices are found to be essential to improve productivity in multi-storey building projects.

Descriptive Statistics of the Human Resource Management Practices

In the human resource management category, the practice clear delegation of responsibility has the highest mean value of 4.17 with a standard error of 0.13 (Table 6-2 above). Exit interview has the lowest mean value of 2.87 with a standard error of 0.17. The practice 'financial incentive programs' has the highest standard deviation of 1.06 which indicates that respondents have different perception regarding the importance of this practice in improving productivity in building projects. On the other hand, the

practice stability of organisational structure has the lowest standard deviation of 0.55 which implies that there is less dispersion of the data from the mean value.

The distribution of the eleven human resource management practices' data is presented using boxplots indicated in Figure 6-8 above. Accordingly, the practices non-financial incentive programs, social activities, and exit interviews have the lowest median values of 3. Crew Composition, Skill Assessment and Evaluation, Employees Training, Career Development, Financial Incentive Programs, Stability of Organisational Structure, Clear Delegation of Responsibility, and Retention Plan for Experienced Personnel have equal median values of 4.

The percentage distributions of the responses (on a scale of 1 to 5) of the human resource management practices concerning their importance in improving the productivity of multi-storey building projects are provided in Table 6-3 above. Accordingly, 46% of the respondents rated the practice crew composition as very important (response scale of 4). Similarly, the participants perceive skill assessment (62%), employees training (54%), career development (54%), stability of organizational structure (54%), clear delegation of responsibility (44%), and retention plan for experienced personnel (46%) as 'very important' (response scale of 4). Overall, most respondents (45%) rated that human resource management practices are very important to improve productivity in building projects (Table 6-3). This finding confirms the finding of Phase I of this research which revealed that the eleven human resource management practices have the potential to improve productivity in multi-storey building construction projects.

Descriptive Statistics of the Safety and Health Practices

As shown in Table 6-2 above, all the seven safety and health practices have the minimum values of 3, the maximum values of 5 and range of 2. The practice 'safety and health policy' has the highest mean value of 4.74 with a minimum standard error of 0.08. 'Safety and health training' and 'housekeeping' have both mean values of 4.44 which is the minimum value. Safety and health training has the highest standard deviation of 0.79 which implies that the data is more dispersed. The practice 'safety and health policy' has the lowest standard deviation of 0.60 which indicates that the level of dispersion of the data is low.

The medians of the seven safety and health practices are shown in Figure 6-8 above. Accordingly, the practice housekeeping has the lowest median value of 4. Safety and Health Policy, Safety and Health Plan, Safe Work Method Statement, Hazards Analysis, Safety and Health Training, and Toolbox Safety Meetings have equal median values of 5.

In Table 6-3 above, the percentage distributions of the relative importance of the seven safety and health practice on a scale of 1 to 5 are provided. Accordingly, none of the respondents rated the safety and health practices as 'not important' (response scale of 0) and 'slightly important' (response scale of 1). The proportions of respondents who rated the practices housekeeping, safety and health policy, safety and health plan, safe work method statement, hazard analysis, safety and health training, and toolbox safety meetings as 'extremely important' are 49%, 77%, 69%, 67%,72%, 62%, and 64% respectively. Overall, the majority of the respondents (66%) rated that safety and health practices are 'extremely important' (response scale of 5) to improve productivity in multi-storey building construction projects. This finding verifies the finding of Phase I of the research in which the seven safety and health practices have been identified as essential practices that have the potential to improve productivity in multi-storey building projects.

6.5 Summary

This chapter described the background of the questionnaire survey participants. Moreover, the profiles of the contractors that have nominated the 39 multi-storey building projects are explained. Based on the characteristics of the construction companies, various propositions which will be confirmed in the next chapter were made. In Chapter 7, the detailed quantitative data analysis will be carried out to answer the following questions related to the profiles of the companies: Do contractors with higher turnover implement higher levels of the construction management practices? Do contractors having higher experience implement higher levels of the construction management practices?

The productivity factors (PFs) of the building projects which are used as inputs to the analysis conducted in next chapter were computed, and the projects were analysed based on their PFs. The findings of the preliminary quantitative data analysis confirmed

that the management practices identified in Phase I of the research are essential for improving productivity in multi-storey building projects. The construction materials management practices, the management practices related to construction methods, the preconstruction phase management practices and the human resource management practices were perceived by the majority of the respondents as 'very important' (response scale of 4). The safety and health practices were rated as extremely important (response scale of 5) practices to improve productivity in multi-storey building construction projects. The equipment management practices have been considered as somewhat important (response scale of 3). The next chapter presents the detailed quantitative data analysis and findings.

CHAPTER SEVEN

7 QUANTITATIVE DATA ANALYSIS AND FINDINGS

7.1 Introduction

In this chapter, the quantitative data is analysed to achieve part of Objective 1 or 'to prioritize the construction management practices' and part of Objective 2 of the study which is 'to validate scoring tools for measuring, planning, monitoring and evaluating the management practices that could improve productivity in multi-storey building construction projects.' In Section 7.2, the relative importance indexes of the management practices are computed, and the practices are ranked accordingly. In Section 7.3, the scoring tools for the management practices are weighted and validated. The reliability tests for the tools are also carried out in Section 7.3. The correlation analyses between productivity and the management practices are conducted, and the findings are presented in Section 7.4. Moreover, the associations among the company profiles, project characteristics, and the management practices are investigated in Section 7.4. Finally, the chapter is summarised in Section 7.5.

7.2 Relative Importance of the Construction Management Practices

To address part of Objective 1 or 'to prioritise the management practices,' the relative importance indexes (RII) of the practices are computed using Equation 7-1.

$$RII = \frac{5(n_5) + 4(n_4) + 3(n_3) + 2(n_2) + n_1}{5(n_5 + n_4 + n_3 + n_2 + n_1)}$$
 Equation 7-1

In the equation, n_5 represents the number of respondents who rated 5 (extremely important); n_4 stands for the number of respondents who rated 4 (very important); n_3 denotes the number of respondents who rated 3 (somewhat important); n_2 represents the number of respondents who rated 2 (slightly important); and n_1 stands for the number of respondents who rated 1 (not important). For instance, the RII of the practice Short Interval Plan is computed to be 0.88 (refer the computation below). Using similar technique, the weights for other 46 management practices are calculated, and the results are shown in Table 7-1. Based on their weights, the practices are then ranked.

Relative Importance Index (RII) of 'Short Interval Plan' $=\frac{5 \times 19 + 4 \times 17 + 3 \times 2 + 2 \times 1 + 1 \times 0}{5 \times (19 + 17 + 2 + 1 + 0)} = 0.88$ The practice well-defined scope of works with RII of 0.96 is found to be the most important practice. Safety and health policy (RII=0.95) is ranked second, safety and

CHAPTER 7

Quantitative Data Analysis and Findings

health plan (RII=0.92) and hazard analysis (RII=0.92) are both ranked third. All the safety and health practices are ranked within the top ten which implies that, in the context of Victoria, the safety and health practices are more important in improving productivity in multi-storey building construction projects than other practices.

Table 7-1 Relative Importance Indexes of the Construction Management Prac

Construction Management Practices	RII	Overall Rank	Construction Management Practices		Rank within Category
Well-Defined Scope of Work	0.96	1	Construction Materials Management Practices (CMMP)		
Safety and Health Policy	0.95	2	Long-Lead Materials Identification	0.92	1
Safety and Health Plan	0.92	3	Procurement Plans for Materials	0.85	2
Hazards Analysis	0.92	3	Materials Delivery Schedule	0.75	3
Long-Lead Materials Identification	0.92	5	Material Inspection Process	0.70	4
Safe Work Method Statement	0.91	6	Materials Status Database	0.66	5
Toolbox Safety Meetings	0.91	7	Materials Inspection Team	0.57	6
Housekeeping	0.89	8	Average weight of CMMP	0.74	
Safety and Health Training	0.89	8	Construction Equipment Management Practices (CEMP)	0.70	
Short Interval Plan	0.88	10	Construction Equipment Maintenance	0.79	1
Buildability Review	0.88	10	Procurement Plan for Construction Equipment	0.61	2
Construction Work Packages	0.86	12	Construction Equipment Productivity Analysis	0.79	3
Project Start -up Plan	0.85	13	Average weight of CEMP Management Practices Related to Construction Methods (MPRCM)	0.73	
Regulatory Requirement	0.85	13	Traffic Control Plan	0.851	1
Procurement Plans for Materials	0.85	15	Project start-up plan	0.851	1
Clear Delegation of Responsibility	0.84	17	Machinery Positioning Strategy	0.826	3
Project Completion Plan	0.83	18	Project Completion Plan	0.820	3
Machinery Positioning Strategy	0.82	19	Dynamic Site Layout Plan	0.815	5
Dynamic Site Layout Plan	0.82	20	Schedule Control	0.015	6
Stability of Organizational Structure	0.82	20	Work Schedule Strategies	0.733	7
Construction Equipment Maintenance	0.82	20	Site Security Plan	0.733	7
Contract Types	0.77	22	Integrated Schedule	0.755	9
Schadula Control	0.77	23	Innevations and New Technologies	0.636	10
Craw Composition	0.77	24	Avarage weight of MPPCM	0.030	10
	0.77	24	Average weight of MI KCM	0.77	
Use of Software in Planning Work Packages	0.76	26	Pre-Construction Phase Management Practices (PCPMP)		
Retention Plan for Experienced Personnel	0.76	26	Well-Defined Scope of Work	0.959	1
Utilities Alignment	0.76	28	Short Interval Plan	0.877	2
Materials Delivery Schedule	0.75	29	Buildability Review	0.877	2
Employees Training	0.75	29	Construction Work Packages	0.856	4
Skill Assessment and Evaluation	0.74	31	Regulatory Requirement	0.851	5
Work Schedule Strategies	0.73	32	Contract Types	0.774	6
Site Security Plan	0.73	32	Use of Software in Planning Work Packages	0.764	7
Career Development	0.71	34	Utilities Alignment	0.759	8
Material Inspection Process	0.7	35	Dedicated Planner	0.61	9
Procurement Plan for Construction Equipment	0.69	36	Model Development	0.538	10
Financial Incentive Programs	0.67	37	Average weight of PCPMP	0.79	
Social Activities	0.67	37	Human Resource Management Practices (HKMP)	0.026	1
Materials Status Database	0.66	39	Clear Delegation of Responsibility	0.836	1
Integrated Schedules	0.66	40	Stability of Organizational Structure	0.815	2
Non-Financial Incentive Programs	0.65	41	Crew Composition	0.769	3
Innovations and New Technologies	0.64	42	Retention Plan for Experienced Personnel	0.740	4
Dedicated Planner	0.61	43	Employees framing	0.749	3
Construction Equipment Productivity Analysis	0.61	44	Skill Assessment and Evaluation	0.738	0
Exit Interviews	0.57	43	Career Development	0.708	/
Madel Development	0.57	40	Financial incentive Programs	0.072	8
Model Development	0.54	4/	Social Activities	0.672	8
			Non-Financial incentive Programs	0.651	10
			Exit Interviews	0.374	11
			Average weight of HKMP	0.72	
			Safety and Health Practices (SHP)	0.040	1
			Safety and Health Plan	0.949	2
			Salety and fieldff Filli	0.923	2
			Safe Work Method Statement	0.923	4
			Toolbox Safety Meetings	0.913	+
			Housekeeping	0.908	6
			Safety and Health Training	0.887	6
			Average weight of SHP	0.007	0
			nverage weight 0j 511	0.71	

155

The average weights of the practices categorised under safety and health practices, preconstruction phase management practices, management practices related to construction methods, construction materials management practices, human resource management practices, and construction equipment management practices are 0.91, 0.79, 0.77, 0.74, 0.72, and 0.70 respectively (Table 7-1). Accordingly, the Safety and Health practices are ranked first, pre-construction phase management practices are ranked second, management practices related to construction methods are ranked third, construction materials management practices are ranked fourth, human resource management practices are ranked fifth, and construction equipment management practices are ranked sixth. The finding indicates that the implementation of Safety and Health practices is more important than the implementation of construction equipment management practices in enhancing productivity in multi-storey building projects.

Normality test is conducted to decide whether to use parametric or non-parametric tests in making statistical commentary regarding the significance of the differences among the weights of the 47 construction management practices. If assumptions of the parametric test (for instance, normality) are fulfilled, 'repeated measures ANOVA' will be used, or else the Friedman test which is non-parametric will be adopted. In the normality test, the null hypotheses of the Shapiro-Wilk test which states 'the data is normally distributed' will be rejected if p-values are less than 5%. Accordingly, the normality test is conducted using SPSS 24, and all the *p-values* of the Shapiro-Wilk test are less than 0.05 indicating that the data is not normally distributed (Table 7-2). Thus, the non-parametric test or Friedman test is used to analyse the differences among the weights of the practices.

The Friedman test for the 47 management practices is conducted and a *p-value* less than 0.001 is obtained (Table 7-3). The finding implies that there are significant differences among the weights of the practices. Iterative Friedman Test is carried out to identify the practices with similar weights, and a *p-value* of 0.16 is obtained for the top seven management practices (Table 7-3). This indicates that well-defined scope of work, safety and health policy, safety and health plan, hazard analysis, long-lead materials identification, safe work method statement and toolbox safety meetings are equally important to improve productivity in multi-storey building projects.

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

		p-Wilk	
Construction Management Practices	Statistic	df	Sig.
Construction Materials Management Practices			
Procurement Plans for Materials	0.754	39	< 0.001
Long-Lead Materials Identification	0.632	39	< 0.001
Materials Status Database	0.863	39	< 0.001
Materials Delivery Schedule	0.875	39	< 0.001
Material Inspection Process	0.886	39	0.001
Materials Inspection Team	0.864	39	< 0.001
Construction Equipment Management Practices			
Construction Equipment Maintenance	0.834	39	0.002
Procurement Plan for Construction Equipment	0.900	39	< 0.001
Construction Equipment Productivity Analysis	0.887	39	0.001
Management Practices Related to Construction Methods			
Integrated Schedule	0.840	39	< 0.001
Work Schedule Strategies	0.811	39	< 0.001
Schedule Control	0.849	39	< 0.001
Dynamic Site Layout Plan	0.805	39	< 0.001
Traffic Control Plan	0.788	39	< 0.001
Site Security Plan	0.861	39	< 0.001
Machinery and Equipment Positioning Strategy	0.792	39	< 0.001
Project start-up plan	0.793	39	< 0.001
Project Completion Plan	0.803	39	< 0.001
Innovations and New Technologies	0.895	39	0.002
Pre-construction Phase Management Practices			
Short Interval Plan	0.743	39	< 0.001
Well-Defined Scope of Work	0.409	39	< 0.001
Use of Software in Planning Work Packages	0.866	39	< 0.001
Dedicated Planner	0.907	39	0.003
Construction Work Packages	0.782	39	< 0.001
Buildability Review	0.738	39	< 0.001
Utilities Alignment	0.848	39	< 0.001
Contract Types	0.851	39	< 0.001
Model Development	0.906	39	0.003
Regulatory Requirement	0.781	39	< 0.001
Human Resource Management Practices			
Crew Composition	0.854	39	< 0.001
Skill Assessment and Evaluation	0.748	39	< 0.001
Employees Training	0.818	39	< 0.001
Career Development	0.814	39	< 0.001
Non-Financial Incentive Programs	0.886	39	0.001
Financial Incentive Programs	0.882	39	0.001
Social Activities	0.874	39	< 0.001
Stability of Organizational Structure	0.822	39	< 0.001
Clear Delegation of Responsibility	0.818	39	< 0.001
Retention Plan for Experienced Personnel	0.863	39	< 0.001
Exit Interviews	0.913	39	0.005
Safety and Health Practices			
Housekeeping	0.733	39	< 0.001
Safety and Health Policy	0.553	39	< 0.001
Safety and Health Plan	0.633	39	< 0.001
Safe Work Method Statement	0.654	39	< 0.001
Hazards Analysis	0.606	39	< 0.001
Safety and Health Training	0.687	39	< 0.001
Toolbox Safety Meetings	0.675	39	< 0.001

Т	able	: 7-	21	No	rma	litv	γT	est	R	esu	lts	of	th	e (Construe	ctic	on I	M	anage	ment	Pr	acti	ices	
						-													0					

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

47 Construction Management Practices	N	Mean Rank	χ2 approximation	df	p-value
Integrated Schedule	30	15 31	11		
Work Schedule Strategies	30	19.51			
Schedule Control	30	23.06			
Dynamic Site Layout Plan	30	25.00			
Traffic Control Plan	30	20.42			
Site Security Den	20	20.05			
Machinery Positioning Strategy	39	20.40			
Project stort up plan	20	20.01			
Project Statt-up plan	20	20.97			
Innevations and New Technologies	20	12.00			
Dra surgement Diang for Motorials	20	20.01			
Procurement Plans for Materials	39	28.81			
Long-Lead Materials Identification	39	34.83			
Materials Status Database	39	15.54			
Materials Delivery Schedule	39	21.96			
Materials nspection Process	39	18.21			
Materials InspectionTeam	39	10.78			
Procurement Plans for Construction Machinery	39	17.65			
Construction Machinery Productivity Analyses	39	11.92			
Construction Machinery Maintenance	39	25.23			
Short Interval Planning	39	31.45			
Well Defined Scope of Work	39	37.22			
Use of Software in Planning	39	22.46			
Dedicated Planner	39	13.51			
Construction Work Packages	39	29.68	648.17	46	< 0.001
Buildability Review	39	30.51			
Utilities Alignment	39	21.82			
Contract types	39	23.74			
Model Requirements	39	9.49			
Regulatory Requirements	39	29.79			
Crew Composition	39	23.42			
Skill Assessment	39	21.49			
Employees Training	39	22.44			
Career Development	39	18.73			
Non-Financial Incentive Programs	39	15.59			
Financial Incentive Programs	39	17.68			
Social Activities	39	16.32			
Stability of Organizational Structure	30	26.44			
Clear Delegation of Pesponsibility	30	20.44			
Decigation of Responsibility	39	27.70			
Retention Plan for Experienced Personnel	39	22.49			
Exit Interviews	39	11.32			
House Keeping	39	32.08			
Health and Safety Policy	39	36.56			
Health and Safety Plans	39	34.73			
Task Safety Analysis	39	34.01			
Hazards Analysis	39	35.00			
Health and Safety Training	39	32.18			
Toolbox Safety Meetings	39	33.87			
Top 7 Construction Management Practices	Ν	Mean Rank	χ2 approximation	df	p-value
Long-Lead Materials Identification	39	3.97	- *		
Well Defined Scope of Work	39	4.45			
Health and Safety Policy	39	4.29			
Health and Safety Plans	39	3.91	9.27	6	0.16
Task Safety Analysis	39	3.74	2.21	0	0.10
Hazards Analysis	30	3.96			
Toolbox Safaty Meetings	30	3.67			
TOODOX Salety Meetings	37	5.07			

Table 7-3 Friedman Test Results of the Construction Management Practices

The Friedman and Wilcoxon Test results as well as the relative importance of the practices within their respective categories are described in the following subsections.

158

Relative Importance of the Construction Materials Management Practices

Long-materials identification is found to be the most critical materials management practice that could increase productivity in multi-storey building construction projects (Table 7-1). Procurement plans for materials, materials delivery schedule, materials inspection process, and materials status database are ranked second, third, fourth and fifth respectively. Materials inspection team is ranked last in the construction materials management category.

The outputs of the Friedman test for the construction materials management practices are shown in Table 7-4, and the p-value is found to be statistically significant at 5% level of significance (p < 0.001 < 0.05). Thus, the null hypothesis which states 'the weights of the six construction materials management practices are equal' is rejected. Therefore, there is a significant difference between the weights assigned to the six practices. However, as the Friedman test is not suitable to identify where the difference lies, posthoc analysis or Wilcoxon test is conducted. Wilcoxon test is chosen since it is the most suitable test after analysing the data using the Friedman test (Pereira et al., 2015).

Table 7-4 Outputs of Friedman and Wilcoxon Tests - Construction Materials Management Practices

		1 Iuotiee					
	_						
Construction Materials Management Practices	Ν	Mean Rank	χ^2 approximation	df	p-value	_	
Procurement Plans for Materials	39	4.5	99.886	5	< 0.001		
Long-Lead Materials Identification	39	5.19					
Materials Status Database	39	2.67					
Materials Delivery Schedule	39	3.63					
Material Inspection Process	39	3.06					
Materials Inspection Team	39	1.95					
		Wilcoxon Te	est			-	
Construction Metanials Management Descriptions	P-values						
Construction Materials Management Practices	PPM		LLM	MSD	MDS	MIP	MIT
Procurement Plans for Materials		_	0.004	<0.001	0.001	<0.001	< 0.001
Long-Lead Materials Identification		_	_	< 0.001	< 0.001	< 0.001	< 0.001
Materials Status Database		_	_	_	0.001	0.235	0.014
Materials Delivery Schedule		_	_	_	_	0.059	<0.001
Material Inspection Process		_	_	_	_	_	0.001
Materials Inspection Team		_	_	_	_	_	_

Note: PPM=Procurement Plans for Materials, LLM= Long-Lead Materials Identification, MSD= Materials Status Database, MDS= Materials Delivery Schedule, MIP= Material Inspection Process, MIT= Materials Inspection Team.

The results of the Wilcoxon test indicate that the null hypothesis which states 'the weights of materials inspection process and materials status database' is the same is accepted as the p-value exceeds 0.05 (p=0.235>0.05). Similarly, the null hypothesis for materials inspection process and materials delivery schedule is not rejected since 159

p=0.059>0.05. This implies that there are no significant differences between the weights of materials inspection process and materials status database. However, there are significant differences among the thirteen combinations that are indicated in bold in Table 7-4. For instance, there is a significant difference between the top three materials management practices (long-lead materials identification, procurement plans for materials and materials delivery schedule).

Relative Importance of the Construction Equipment Management Practices

As shown in Table 7-1, in the construction equipment management practices category, construction equipment maintenance is ranked first, procurement plan for construction equipment is ranked second, and construction equipment productivity analysis is ranked third. The result of Friedman test for the construction equipment management practices shows p < 0.001, df = 2, and Chi-square=15.57. The mean ranks for procurement plan for construction equipment, construction equipment productivity analysis, and construction equipment maintenance are 2, 1.63 and 2.37 respectively. Based on the p-value, the null hypothesis for the test which states 'there are no differences among the weights of the three construction equipment management practices' is rejected and the alternative hypothesis which states 'there are significant differences among the weights' is accepted. However, since Friedman test does not indicate which variable is significantly different from another, Wilcoxon test is carried out. Pair-wise comparisons using Wilcoxon's test indicate p=0.007 < 0.05 for construction equipment procurement plan and construction equipment productivity analysis; p=0.031<0.05 for construction equipment procurement plan and construction equipment maintenance; and p=0.001<0.05 for construction equipment productivity analysis and construction equipment maintenance. All the three possible combinations are statistically significant which shows that the weights of the practices are different from each other.

Relative Importance of the Management Practices Related to Construction Methods

The practice traffic control plan and project start-up plan have both equal weight of 0.851 and ranked first in the category of management practices related to construction methods (Table 7-1). Machinery positioning strategy, project completion plan, and dynamic site layout plan are ranked from third to fifth. The practice 'innovations and new technologies' is assigned a weight of 0.636 and ranked last. To check if there are statistically significant differences among the weights of the ten management practices

related to construction methods, Friedman test is conducted, and the results are presented in Table 7-5. For instance, to investigate if there is a significant difference between the weights 0.851 and 0.826 (project start-up plan and machinery positioning strategy), the Friedman test is carried out and a *p*-value less than 0.0001 is obtained. Since the p-value is much less than the cut-off *p*-value of 0.05, the null hypothesis which states 'the weights of all the ten management practices related to construction methods are equal' is rejected. Thus, there are statistically significant differences among the weights assigned to the practices.

Table 7-5 Outputs of Friedman and Wilcoxon Tests - Management Practices Related to Construction Methods

F	riedman T	Fest								
Management Practices Related to Construction Methods		Mean Rank	χ ² approximation	df	p-value	-				
Integrated Schedule	39	3.72								
Work Schedule Strategies	39	4.76								
Schedule Control	39	5.49								
Dynamic Site Layout Plan	39	6.24								
Traffic Control Plan	39	6.76								
Site Security Plan	39	4.86	74.73	9	< 0.0001					
Machinery and Equipment Positioning Strategy	39	6.29								
Project start-up plan	39	6.82								
Project Completion Plan	39	6.59								
Innovations and New Technologies	39	3.47								
			Wilcoxon T	est						
Management Practices Related to			F	-values						
Construction Methods	IS	WSS	SC	DSLP	TCP	SSP	MPS	PSP	PCP	INT
Integrated Schedule	_	0.013	0.001	<0.001	< 0.001	0.04	< 0.001	< 0.001	< 0.001	0.557
Work Schedule Strategies	-	-	0.046	0.014	< 0.001	1	0.028	0.002	0.013	0.003
Schedule Control	_	-	-	0.242	0.023	0.158	0.252	0.036	0.14	< 0.001
Dynamic Site Layout Plan	-	-	-	-	0.159	0.009	0.95	0.162	0.947	< 0.001
Traffic Control Plan	-	-	-	-	-	0.001	0.233	0.917	0.414	< 0.001
Site Security Plan	-	-	-	-	-	-	0.023	0.002	0.025	0.004
Machinery and Equipment Positioning Strategy	_	-	-	-	-	-	-	0.339	0.836	< 0.001
Project start-up plan	-	-	-	-	-	-	-	-	0.446	< 0.001
Project Completion Plan	_	_	-	-	-	-	_	_	-	< 0.001
Innovations and New Technologies	_	-	-	-	-	-	-	-	-	-
Note: IS=Integrated Schedule, WSS= Work Schedule Strat	egies, SC=So	chedule Contro	ol, DSLP= Dynamic S	ite Lay out	Plan, TCP=Tr	affic Control I	Plan, SSP=Site	Security Plan,	MPS=Mach	inerv

Note: IS=Integrated Schedule, WSS= Work Schedule Strategies, SC=Schedule Control, DSLP= Dynamic Site Layout Plan, TCP=Traffic Control Plan, SSP=Site Security Plan, MPS=Machine Positioning Strategy, PSP=Project Start-up Plan, PCP= Project Completion Plan, INT=Innovation and new Technologies.

By using Wilcoxon test, pair-wise analyses between the ten management practices related to construction methods are conducted, and the results are shown in Table 7-5. Accordingly, 45 possible combinations are analysed, and twenty-nine of them are found to be statistically significant. For instance, the p-value between work schedule strategies and dynamic site layout plan is 0.01 indicating that the difference between them is significant. On the other hand, 15 combinations of the practices are found to statistically insignificant at the 5% level of significance. From the 15 combinations, 10 of them represent the combinations of the top 5 practices (refer *p-values* indicated in bold in Table 7-5). The *p-value* between traffic control plan and project start-up plan is 0.92 >
0.05 which shows that there is no significant difference between the weights of these two practices. Similarly, the *p*-value between project completion plan and dynamic site layout plan is 0.95 > 0.05 indicating that there is no major difference between the weights of these practices (0.826 and 0.815). After analysing the ten combinations, it is found that there are no significant differences among the weights of the top five management practices related to construction methods. Therefore, traffic control plan, project start-up plan, machinery positioning strategy, project completion plan and dynamic site layout plan are equally important to improve productivity in multi-storey building projects.

Relative Importance of the Pre-Construction Phase Management Practices

In the pre-construction phase category, the practice well-defined scope of work with RII of 0.96 is ranked first. Short interval planning and buildability review with RII of 0.88 are both ranked second (Table 7-1). Dedicated planner and model development with the relative importance indexes of 0.61 and 0.54 are ranked ninth and tenth respectively.

In Table 7-6, the results of Friedman and Wilcoxon Tests for the pre-construction phase management practices are presented. The Friedman's *p-value* (<0.001) indicates that there are differences among the weights of the pre-construction phase management practices. Out of 45 possible combinations in the Wilcoxon Test (Table 7-6), nine of them have *p-values* greater than 0.05 (statistically insignificant), and 36 of them have *p-values* less than 0.05 (statistically significant).

Short interval plan and buildability review have a *p*-value of 1 which indicates that there is no difference between their weights. There is also no significant difference between the weights of short interval plan and construction work packages as the p=0.49>0.05. Short interval plan and regulatory requirement have a *p*-value of 0.34>0.05, and the difference between their weights is insignificant. No significant difference between the weights of buildability review and construction work packages is found (p=0.4>0.05). Similarly, the weights of buildability review and regulatory requirement are similar as p=0.43>0.05. Construction work packages and regulatory requirement also have a *p*-value of 0.84>0.05. Based on these findings, it is concluded that there is no significant difference between the practices ranked from 2 to 5 in preconstruction phase category. Thus, short interval plan, buildability review, construction work packages, and

regulatory requirement are equally important to improve productivity in multi-storey building projects.

ri	leaman res	śl.								
Pre-Construction Phase Management Practices	Ν	Mean Rank	χ2 approximation	df	p-value					
Short Interval Plan	39	6.83								
Well-Defined Scope of Work	39	8.13								
Use of Software in Planning Work Packages	39	4.94								
Dedicated Planner	39	3.13								
Construction Work Packages	39	6.51								
Buildability Review	39	6.67	151.324	9	< 0.0001					
Utilities Alignment	39	4.74								
Contract Types	39	5.21								
Model Development	39	2.37								
Regulatory Requirement	39	6.47								
			Wilcoxon Test							
Pro Construction Phase Management Practices			P-1	values						
Tre-Construction Trase Management Tractices	SIP	WSW	US	DP	CWP	BR	UA	CT	MD	RR
Short Interval Plan	-	0.008	0.008	< 0.001	0.488	1	0.001	0.003	< 0.001	0.343
Well-Defined Scope of Work	-	-	< 0.001	< 0.001	0.002	0.002	< 0.001	< 0.001	< 0.001	0.003
Use of Software in Planning Work Packages	-	_	-	0.003	0.014	0.002	0.954	0.824	< 0.001	0.023
Dedicated Planner	-	_	-	_	< 0.001	< 0.001	< 0.001	< 0.001	0.022	< 0.001
Construction Work Packages	-	_	-	_	_	0.403	0.007	0.006	< 0.001	0.837
Buildability Review	-	_	-	_	_	_	< 0.001	0.003	< 0.001	0.433
Utilities Alignment	-	_	-	_	_	_	_	0.628	< 0.001	0.002
Contract Types	-	_	_	_	_	_	_	_	< 0.001	0.027
Model Development	-	_	-	_	-	-	_	_	_	< 0.001
Regulatory Requirement	_	_	_			_	-	_	_	-
Note: SIP= Short Interval Plan, WSW=Well-defined Scope of Wo UA= Utilities Alignment, CT= Contract Types, MD= Model Do	ork, US= Use evelopment, F	of Software in R=Regulatory	Planning Work Package	s, DP= Dec	licated Planner,	CWP= Cons	struction Wor	k Package, Bl	R= Buildabilit	ty Review,

 Table 7-6 Outputs of Friedman Test - Pre-Construction Phase Management Practices

As shown in Table 7-6, there are no significant differences among contract types and use of software in planning work packages (p=0.82>0.05); contract types and utilities alignment (p=0.63>0.05); and use of software in planning work packages and utilities alignment (p=0.95>0.05). Thus, contract types, use of software in work packaging and utilities alignment have equal importance in improving productivity in multi-storey building construction projects. In other words, the practices ranked from 6 to 8 have similar weights.

Relative Importance of the Human Resource Management Practices

Clear delegation of responsibility with RII of 0.84 is ranked first in the human resource management (HRM) practices category. Stability of organisational structure having RII of 0.82 is ranked second (Table 7-1). Financial incentive programs and social activities have equal weights (0.67), and both are ranked eighth. Non-financial incentive programs (0.65) and exit interviews (0.57) are ranked tenth and eleventh respectively. The result of Friedman test for HRM practices is indicated in Table 7-7 and a *p-value* less than 0.001 is found. The finding shows that there are significant differences among the 11 human resource management practices.

	Table 7-7 Ou	tputs of Friedman	and Wilcoxon	Tests - Hu	man Resource	Management Practice
--	--------------	-------------------	--------------	------------	--------------	---------------------

	Friedman	Test									
Human Resource Management Practices	Mean Rank	Ν	χ2 approximation	df	p-value						
Crew Composition	6.85										
Skill Assessment and Evaluation	6.38										
Employees Training	6.69										
Career Development	5.71							SoA SOS CDR RPEP 0.018 0.318 0.066 0.846 0.101 0.076 0.025 0.51 0.377 0.073 0.054 0.689 0.246 0.003 0.001 0.095 0.567 0.001 0 0.011 0 0.011 0.047 0 0.011 0.011 0 0.011 0.028			
Non-Financial Incentive Programs	4.87						So.4 SOS CDR RPEP 0.018 0.318 0.066 0.846 0.101 0.076 0.025 0.51 0.037 0.073 0.054 0.689 0.246 0.005 0.003 0.095 0.567 0.001 0 0.017 0.975 0.001 0 0.047 _ 0 0 0.011 0.457 0.103 0.028				
Financial Incentive Programs	5.26	39	71.163	10	< 0.0001						
Social Activities	4.94										
Stability of Organizational Structure	7.54										
Clear Delegation of Responsibility	7.74										
Retention Plan for Experienced Personnel	6.45										
Exit Interviews	3.58										
			Wilcoxon Test	t							
Uuman Pasauraa Managamant Practicas			P-v	alues							
Human Resource Management Practices	CC	SA	ET	CD	NFIP	FIP	SoA	SOS	CDR	RPEP	EI
Crew Composition	-	0.299	0.388	0.117	0.003	0.046	0.018	0.318	0.066	0.846	0
Skill Assessment and Evaluation	-	-	0.593	0.291	0.018	0.116	0.101	0.076	0.025	0.51	0
Employees Training	-	-	-	0.106	0.038	0.1	0.037	0.073	0.054	0.689	0
Career Development	-	-	-	-	0.106	0.375	0.246	0.005	0.003	0.095	0.001
Non-Financial Incentive Programs	-	-	-	-	-	0.627	0.567	0.001	0	0.017	0.067
Financial Incentive Programs	-	-	-	-	-	-	0.975	0.001	0.001	0.047	0.041
Social Activities	_	_	-	_	_	_	_	0	0	0.011	0.016
Stability of Organizational Structure	-	-	_	_	-	-	-	-	0.457	0.103	0
Clear Delegation of Responsibility	_	_	-	_	-	_	_	_	_	0.028	0
Retention Plan for Experienced Personnel	-	-	_	_	-	-	-	-	-	-	0
Exit Interviews	_	_	-	_	_	_	_	_	_	-	-
Notes: CC= Crew Composition, SA=Skill Assessment, I Activities, SOS= Stability of Organizational Structure, C	ET = Employee DR= Clear Del	Training, Cl egation of R	D = Career Development, esponsibility, RPEP=Ret	NFIP= No ention Plan	n-Financial Ince for Experienced	entive Progra	ms, FIP= Fin EI= Exit Inter	ancial Incentiviviews.	ve Programs,	SoA= Social	

The results of the Wilcoxon test are shown in Table 7-7. Fifty-five possible pair-wise comparisons are analysed, and 28 combinations have a p-value less than 0.05 and 27 pairs have p-values greater than 0.05 (p-values shown in bold in Table 7-7). If the pvalue is less than 0.05, the null hypothesis of the Wilcoxon test which states 'the weights of the HRM practices are equal' is rejected. For instance, there is no significant difference between the weights of clear delegation of responsibility and stability of organisational structure (p=0.457>0.05). Clear delegation of responsibility and crew composition have a *p*-value of 0.066>0.05 which indicates that there is no significant difference between the two practices' weights. The p-value for the stability of the organisational structure and crew composition is 0.318 which shows that there is no significant difference between their weights. Similarly, there are 27 combinations which have no statistically significant differences (Table 7-7). On the other hand, a significant difference between the weights of 'clear delegation of responsibility' and 'retention plan for experienced personnel' (p=0.028<0.05) is found, and there are 28 combinations which have significant differences (p-values indicated in bold in Table 7-7).

Based on the findings of the Wilcoxon Test, it is concluded that there are no significant differences among the weights of the top three human resource management practices. Thus, clear delegation of responsibility, the stability of the organisational structure and

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

crew composition are equally important to improve productivity in multi-storey building construction projects.

Relative Importance of Safety and Health Practices

The practice safety and health policy has the maximum weight of 0.95 and ranked first in the safety and health category. Safety and health plan and hazard analysis have equal weights of 0.92, and both ranked second. Safe work method statement and toolbox safety meetings with weights of 0.913 and 0.908 are ranked fourth and fifth respectively. Housekeeping and safety and health training have both equal weights of 0.887 and ranked sixth.

The results of Friedman test shows are presented in Table 7-8, and a *p-value* (0.08) which is greater than 0.05 is obtained. This implies that the null hypothesis of the test or 'there are no significant differences among the weights of the safety and health practices' is accepted. Thus, all the safety and health practices are equally important to improve productivity in multi-storey building construction projects.

Safety at	nd Health Practices	Mean Rank	N	χ2 approximation	df	p-value
Housekeeping		3.67	39	11.468	10	0.075
Safety and Health Policy	I	4.51				
Safety and Health Plan		4.15				
Safe Work Method State	ement	3.95				
Hazards Analysis		4.19				
Safety and Health Traini	ing	3.65				
Toolbox Safety Meeting	S	3.87				

Table 7-8 Outputs of Friedman Test - Safety and Health Practices

7.3 Preparing and Validating the Weighted Scoring Tools of the Construction Management Practices

7.3.1 Preparing the Weighted Scoring Tools

To prepare the weighted scoring tools, the weight of each practice computed in Section 7.2 is proportionally distributed among the five levels of implementation a particular practice. For instance, the weight proportions for construction equipment management practices are computed as follows. Procurement plan for construction equipment's weight is 0.69, and the weight proportions are Level A= 0, Level B = $1/5 \times (0.69) = 0.14$, Level C = $2/5 \times (0.69) = 0.28$, Level D = $3/5 \times (0.69) = 0.41$, Level E= $4/5 \times (0.69) = 0.55$, and Level F = 0.69. The weight of the practice construction equipment

productivity analysis is 0.61, and the weight proportions are Level A= 0, Level B = $1/5 \times (0.61) = 0.12$, Level C = $2/5 \times (0.61) = 0.24$, Level D = $3/5 \times (0.61) = 0.37$, Level E = $4/5 \times (0.61) = 0.49$, and Level F= 0.61. For the practice construction equipment maintenance, its weight is 0.79 and the weight proportions are Level A=0, Level B = $1/5 \times (0.79) = 0.16$, Level C = $2/5 \times (0.79) = 0.32$, Level D = $3/5 \times (0.79) = 0.47$; Level E= $4/5 \times (0.79) = 0.63$, and Level F= 0.79. Using similar techniques, the weight proportions of all the 47 construction management practices are computed and the results are presented in Table 7-9.

By using the weight proportions indicated in Table 7-9, the weighted scoring tools (the 'scoring tools') for the construction management practices in the context of multi-storey building projects are prepared. For illustration purpose, the scoring tools for the three management practices categorised under construction equipment management practices are presented in this section (Table 7-10). The scoring tools for all management practices are provided in the Summary and Conclusion Chapter (Chapter 10).

Table 7 0 Distribution of the	Waights of the	Construction	Vana agent ant Dug ationa
Table 7-9 Distribution of the	weignis of the	Construction	vianagement Practices
		001100100000000000000000000000000000000	

				Weight nr	oportions		
Construction Management Practices	Weight	Level A	Level R	Level C	Level D	Level F	Level F
Construction Materials Management Practices		Levern	Leverb	Levere	LeverD	LeverL	Lever1
Procurement Plans for Materials	0.85	0	0.17	0.34	0.51	0.68	0.85
Long-Lead Materials Identification	0.92	0	0.18	0.37	0.55	0.73	0.92
Materials Status Database	0.66	0	0.13	0.26	0.4	0.53	0.66
Materials Delivery Schedule	0.75	0	0.15	0.3	0.45	0.6	0.75
Material Inspection Process	0.7	0	0.14	0.28	0.42	0.56	0.7
Materials Inspection Team	0.57	0	0.11	0.23	0.34	0.46	0.57
Construction Equipment Management Practices	0101	-					
Procurement Plan for Construction Equipment	0.69	0.00	0.14	0.28	0.41	0.55	0.69
Construction Equipment Productivity Analysis	0.61	0.00	0.12	0.24	0.37	0.49	0.61
Construction Equipment Maintenance	0.79	0.00	0.16	0.32	0.47	0.63	0.79
Management Practices Related to Construction Methods		0.00	0.10	0.52	0.17	0.05	0.77
Integrated Schedule	0.66	0	0.13	0.26	0.4	0.53	0.66
Work Schedule Strategies	0.73	0	0.15	0.29	0.44	0.58	0.73
Schedule Control	0.77	0	0.15	0.31	0.46	0.62	0.77
Dynamic Site Lay Out Plan	0.83	0	0.16	0.33	0.49	0.66	0.82
Traffic Control Plan	0.85	0	0.17	0.34	0.51	0.68	0.85
Site Security Plan	0.73	0	0.15	0.29	0.44	0.58	0.73
Machinery Positioning Strategy	0.82	0	0.16	0.33	0.49	0.66	0.82
Project Start-Up Plan	0.85	0	0.17	0.34	0.51	0.68	0.85
Project Completion Plan	0.83	0	0.17	0.33	0.5	0.66	0.83
Innovations and New Technologies	0.64	0	0.13	0.26	0.38	0.51	0.61
Pre-Construction Phase Management Practices		-					
Well-defined Scope of Work	0.96	0	0.19	0.38	0.58	0.77	0.96
Short Interval Plan	0.88	0	0.18	0.35	0.53	0.7	0.88
Buildability Review	0.88	0	0.18	0.35	0.53	0.7	0.88
Construction Work Packages	0.86	0	0.17	0.34	0.52	0.69	0.86
Regulatory Requirement	0.85	0	0.17	0.34	0.51	0.68	0.85
Contract Types	0.77	0	0.15	0.31	0.46	0.62	0.77
Use of Software in planning Work Packages	0.76	0	0.15	0.3	0.46	0.61	0.76
Utilities Alignment	0.76	0	0.15	0.3	0.46	0.61	0.76
Dedicated Planner	0.61	0	0.12	0.24	0.37	0.49	0.61
Model Development	0.54	0	0.11	0.22	0.32	0.43	0.54
Human Resource Management Practices							
Crew Composition	0.77	0	0.15	0.31	0.46	0.62	0.77
Skill Assessment and Evaluation	0.74	0	0.15	0.3	0.44	0.59	0.74
Employees Training	0.75	0	0.15	0.3	0.45	0.6	0.75
Career Development	0.71	0	0.14	0.28	0.43	0.57	0.71
Non-Financial Incentive Programs	0.65	0	0.13	0.26	0.39	0.52	0.65
Financial Incentive Programs	0.67	0	0.13	0.27	0.4	0.54	0.67
Social Activities	0.67	0	0.13	0.27	0.4	0.54	0.67
Stability of Organizational Structure	0.82	0	0.16	0.33	0.49	0.66	0.82
Clear Delegation of Responsibility	0.84	0	0.17	0.34	0.5	0.67	0.84
Retention Plan for Experienced Personnel	0.76	0	0.15	0.3	0.46	0.61	0.76
Exit Interviews	0.57	0	0.11	0.23	0.34	0.46	0.57
Safety and Health Practices							
Housekeeping	0.89	0	0.18	0.36	0.53	0.71	0.89
Safety and Health Policy	0.95	0	0.19	0.38	0.57	0.76	0.95
Safety and Health Plan	0.92	0	0.18	0.37	0.55	0.74	0.92
Safe Work Method Statement	0.91	0	0.18	0.36	0.55	0.73	0.91
Hazards Analysis	0.92	0	0.18	0.37	0.55	0.74	0.92
Safety and Health Training	0.89	0	0.18	0.36	0.53	0.71	0.89
Toolbox Safety Meetings	0.91	0	0.18	0.36	0.54	0.73	0.91

Table 7-10 Scoring Tools for Construction Equipment Management Practices

1. Proci	rement Plan for Construction Equipment	Weights
Level A	A procurement plan for construction equipment is not applicable for this building project.	0
Level B	Construction equipment procurement plan is not prepared for this building project.	0.14
Level C	Construction equipment procurement strategies such as hiring or leasing are prepared for this building project.	0.28
Level D	Continuation of Level C and there is a procedure for identifying reputation of potential equipment suppliers	0.41
Level E	Continuation of Level D, plus the plan identifies necessary equipment and onsite resources to support delivery.	0.55
Level F	Continuation of Level E and construction equipment schedule is integrated with other project schedules.	0.69
2. Const	truction Equipment Productivity Analysis	
Level A	Construction equipment is not used on this building project.	0
Level B	Construction equipment is utilized but requirements and usage are not planned and tracked.	0.12
Level C	Construction equipment requirements are planned and tracked but they are not tied to a schedule. The usage is tracked against budget only.	0.24
Level D	Continuations of Level C, and regular meeting is conducted to assess the requirement of construction equipment.	0.37
Level E	Continuation of Level D, and construction equipment resource curves are drawn and resources are leveled.	0.49
Level F	Continuation of Level E and equipment schedule is adjusted based on the audit report that shows equipment downtime and other information.	0.61
3. Const	truction Equipment Maintenance	
Level A	Construction equipment maintenance is not applicable to this building project.	0
Level B	Construction equipment maintenance is not planned for this building project.	0.16
Level C	Construction equipment is logged in simplified spreadsheet and maintenance is carried out by operator request.	0.32
Level D	Continuation of Level C, and equipment maintenance record is linked to individual construction equipment, and maintenance is centrally scheduled and administered.	0.47
Level E	Continuation of Level D, and a computer program is used to record and administer equipment maintenance information such as required and accomplished maintenance logs and usage logs.	0.63
Level F	Continuation of Level E and maintenance schedule is electronically updated and maintenance due notices are automatically issued to concerned parties.	0.79

To validate the scoring tools and to conduct inferential statistical analyses, the scores of the management practices of the projects are computed based on the respondents' selected level of implementation of each practice on the nominated 39 projects. For instance, for a project coded R1, the respondent ticked Level B for construction equipment procurement plan, Level B for construction productivity analysis, and Level C for construction equipment maintenance, and the equivalent total score of construction equipment management practice for that particular project is 0.14 + 0.12 + 0.32 = 0.58 (Table 7-11). Similarly, based on the respondent's chosen levels of implementations, the project's materials management practices score =1.88 (sum of the scores of 6 practices), management practices related to construction methods' score = 3.09 (sum of 10 practices), preconstruction phase management practices', and safety and health practices' score = 4.94 (sum of 7 practices). Thus, the project's total construction management practices score = 0.58 + 1.88 + 3.09 + 4.18 + 3.30 + 4.94 = 168

17.97 (Table 7-11). By using similar procedures, the 39 projects' management practices scores are computed, and the results are presented in Table 7-11.

	Construction Materials	Construction Equipment	Management Practices	Pre-construction	Human Resource	Safety and	Construction
Projects	Management Practices	Management Practices	Related to Construction	Phase Management	Management	Health	Management Practicess
	Scores	Scores	Methods Scores	Practices	Practices	Practices	Scores
	(A)	(B)	(C)	(D)	(E)	(F)	G=A+B+C+D+E+F
R1	1.88	0.58	3.09	4.18	3.30	4.94	17.97
R2	3.77	1.35	6.77	6.10	6.52	5.31	29.82
R3	2.91	0.83	4.89	4.90	4.47	6.03	24.03
R4	3.14	1.08	5.55	5.36	6.96	5.29	27.38
R5	2.28	0.58	5.36	4.07	4.67	4.55	21.51
R6	1.36	0.89	5.44	5.68	5.73	6.39	25.49
R7	3.49	1.67	6.14	5.69	6.57	6.39	29.95
R8	2.99	1.11	6.16	4.94	5.50	5.49	26.19
R9	2.98	1.23	5.09	5.46	5.45	6.03	26.24
R10	3.97	1.23	5.72	5.31	6.13	5.29	27.65
R11	1.53	0.58	3.86	3.65	3.52	4.01	17.15
R12	2.49	1.41	4.85	4.50	5.06	6.02	24.33
R13	1.84	0.97	4.57	4.40	5.36	6.03	23.17
R14	1.97	0.44	4.31	2.61	4.23	2.73	16.29
R15	2.17	0.71	4.81	5.03	4.50	5.66	22.88
R16	1.79	0.59	4.33	5.15	2.59	5.49	19.94
R17	2.37	0.72	2.90	3.30	2.05	5.12	16.46
R18	2.23	0.00	4.06	4.52	4.30	5.65	20.76
R19	1.45	0.56	3.31	3.00	4.49	5.68	18.49
R20	1.41	0.58	3.96	2.77	3.39	4.39	16.50
R21	2.66	0.58	5.10	3.46	5.32	6.03	23.15
R22	2.44	1.39	5.86	5.33	6.74	5.29	27.05
R23	2.66	1.37	3.61	4.61	5.27	6.21	23.73
R24	1.18	0.58	2.64	3.85	2.48	4.03	14.76
R25	1.08	0.68	1.89	2.80	2.04	4.76	13.25
R26	1.32	0.00	2.78	3.99	3.88	4.39	16.36
R27	1.92	1.55	6.17	5.85	7.19	6.03	28.71
R28	1.66	0.16	2.14	1.19	0.92	4.00	10.07
R29	2.34	1.47	5.03	5.25	5.03	6.21	25.33
R30	3.91	1.49	5.31	5.50	3.81	5.85	25.87
R31	2.77	0.96	3.69	4.39	3.97	5.86	21.64
R32	3.81	1.69	5.58	7.11	5.14	6.03	29.36
R33	1.57	0.42	1.95	1.79	2.64	5.48	13.85
R34	2.97	1.08	5.24	6.01	4.54	5.85	25.69
R35	1.68	0.59	4.48	2.05	4.14	5.29	18.23
R36	2.74	1.39	5.95	4.09	5.59	6.21	25.97
R37	3.92	2.09	6.97	6.72	6.27	6.21	32.18
R38	1.53	0.70	5.58	5.27	4.89	5.12	23.09
R39	3.61	1.15	5.86	6.53	5.74	6.39	29.28

Table 7-11	Scores of the	Construction	Management	Practices
10010 / 11		001100100001011		

Note: $\mathbf{A} = PPM + LLM + MSD + MDS + MIP + MIT$; $\mathbf{B} = PPCE + CEPA + CEM$; $\mathbf{C} = IS + WSS + SC + DSLP + TCP + SSP + MPS + PSP + PCP + INT$; $\mathbf{D} = SIP + WSW + US + DP + CWP + BR + UA + CT + MD + RR$; $\mathbf{E} = CC + SA + ET + CD + NFIP + FIP + SOA + SOS + CDR + RPEP + EI$; $\mathbf{F} = HK + SHP + SMP + SMP$

PPM=Procurement Plan for Materials, ILM= Long-lead Materials Identification, MSD=Materials Status Database, MDS= Materials Delivery Schedule, MIP=Materials Inspection Process MIT =Materials Inspection Team, PPC=Procurement Plans for Construction Equipment, CEPA=Construction Equipment Productivity Analysis, CEM=Construction Equipment Maintenance; IS=Integrated Schedule, WSS= Work Schedule Strategies, SC=Schedule Control, DSLP= Dynamic Site Layout Plan, TCP=Traffic Control Plan, SSP=Site Security Plan, MPS=Matchinery Positioning Strategy, PSP=Project Start-up Plan, PCP= Project Completion Plan, INT=Innovation and new Technologies; SIP= Short Interval Plan, WSW=Well-defined Scope of Work, US= Use of Software in Planning Work Packages, DP= Dedicated Planner, CWP= Construction Work Package, BR= Buildability Review, UA= Utilities Alignment, CT= Contract Types, MD= Model Development, RR=Regulatory Requirement; CC=Crew Composition, SA=Skil Assessment and Evaluation, ET= Employee Training, CD = Carer Development, NFIP= Non-Financial Incentive Programs, FIP= Financial Interviewe; HK=Housekeeping, SHP= Safety and Health Plan, SWMS = Safe Work Method Statement, HA= Hazard Analysis, SHT= Safety and Health Training, and TSM= ToolboxSafety Meeting.

7.3.2 Validation of the Weighted Scoring Tools

To address part of Objective 2 of the study which is 'to validate scoring tools for measuring, planning, monitoring and evaluating the management practices that could improve productivity in multi-storey building construction projects,' the mean, mode,

and median scores were computed to choose a baseline score for grouping the projects into high and low scores. The total construction management practices' median score = 23.09, mean score = 22.75, skewness = -0.386, standard error (*SE*) of skewness = 0.378 and Z=-0.386/0.378 =-1.02 are obtained. If data is not highly skewed, mean is the best measure of central tendency (Lurd and Lurd, 2013). According to Kim (2013), if *Zscore* for skewness is less than 1.96, the distribution could be considered as normal or not highly skewed. Thus, the mean score of 22.75 is used as a baseline for classification since the absolute value of *Z* (1.02) is less than 1.96. Projects with total construction management practices scores less than mean value (22.75) are classified under Group-1(low-score), and those having scores greater than the baseline (22.75) are classified under Group-2 (high-score).

To decide the type of test to be used for analysing the difference between the groups, normality test is conducted, and the results are shown in Table 7-12. The Shapiro-Wilk test result for Group 1 indicates p = 0.37 > 0.05 which implies the data is normally distributed. For group 2, p = 0.31 > 0.05 is found indicating that the data in group 2 is also normally distributed. Levene's test is also conducted to check the homogeneity of variance, and p = 0.382 > 0.05 is found. This shows that the variances are homogeneous and another assumption of the parametric test is satisfied. Thus, t-test can be conducted to investigate the difference between the two groups.

			Normality	Test				
	Descrip	otive				Test of N	Normality	
						Shapi	ro-Wilk	
	Ν	Mean PF	Minimum PF	Maximum PF		Statistic	df	Sig
Group 1	21	1	0.78	1.14	Group 1	0.952	21	0.371
Group 2	21	0.9	0.59	1.08	Group 2	0.942	18	0.313
			t- Tes	t				
		t-test for Equ	uality of Mean	IS				
		t	df	. Sig.(2- Mean	Std. Error	95% Confid of the D	95% Confidence Interval of the Difference	
				tatiea)	Dijjerence	Difference	Lower	Upper
DE	Equal variances assumed	2.221	37	0.033	0.097	0.044	0.008	0.185
PF	Equal variances not assumed	2.176	31.638	0.037	0.097	0.044	0.006	0.187

 Table 7-12 Outputs of the Normality and t-Tests for Projects Grouped Based on the Total

 Construction Management Practices Scores

The results of t-test are shown in Table 7-12, and the *p*-value is found to be 0.03 < 0.05. The finding implies that building projects with higher levels of implementation of construction management practices also have higher productivity. Therefore, the scoring 170

tools are valid. The mean productivity factor (PF) for Group-1 and Group-2 are 0.996 and 0.899 respectively. The box-plots of the two groups are drawn to visually represent the difference between the PFs of the two groups (Figure 7-1).



Figure 7-1 Box-Plots of the PFs of the Projects Based on Total Construction Management Practices Scores

The validity tests for the scoring tools of the six categories of the construction management practices are conducted, and the results are presented in subsequent sections.

Validity Test of the Scoring Tools of the Construction Materials Management Practices To validate the scoring tools for the construction materials management practices, the projects are divided into high score group (Group 1) and low score group (Group 2) based on their construction materials management practices scores. To determine the baseline for classification, the mean score = 2.4, the median score = 2.34, the mode score =1.53, the skewness=0.37, the SE of skewness=0.38 and Z= (skewness)/ (SE of skewness) = 0.98 are obtained. As Z=0.98 is less than 1.96, the mean materials management practices score (2.4) is considered for classification purpose. Accordingly, 20 projects are grouped under Group 1, and 19 projects are grouped under Group 2 based on their construction materials management practices scores.

Normality tests are conducted for the two independent groups, and the results are shown in Table 7-13. The Shapiro-Wilk test's hypotheses are H_o (null): the data in the group is normally distributed, and H_a (alternative): the data is not normally distributed. Accordingly, for Group 1, the null hypothesis is not rejected since the p = 0.096 > 0.05

and the data is normally distributed. The <i>p</i> -value for Group 2 is 0.033 which is less than
0.05. However, $Z = \text{Skewness}/SE = 0.27 < 1.96$, and the data in Group 2 can be
considered as normally distributed. Test of homogeneity (Levene's test) is also
conducted. The hypotheses for Levene's test are Ho (null): the variances of Groupland
Group 2 projects are equal, and H_a (alternative): the variances of Group 1 and Group 2
projects are different. Accordingly, a $p = 0.343 > 0.5$ is obtained. This shows that the
null hypothesis is accepted, and the variances of the data are homogeneous. Therefore,
the parametric test (ANOVA or t-test) could be used to check the difference between the
means of two groups. Since there are only two independent samples (groups),
independent sample t-test is used in the analysis (Table 7-13).

		Ivia	nagement	. Practices	scores			
			Norn	nality Test				
	De	scriptive				Test of N	ormality	
						Shapir	•o-Wilk	
	Ν	Mean PF	Minimum PF	Maximum PF		Statistic	df	Sig
Group 1	20	0.996	0.941	1.051	Group 1	0.919	20	0.096
Group 2	19	0.905	0.83	0.979	Group 2	0.891	19	0.033
	t- Test							
		t-test for Equ	ality of Mean	S				
		t	df	Sig.(2-	Mean Difference	Std. Error	95% Confiden of the Dif	nce Interval ference
		-		iulieu)	Dijjerence	Dijjerence	Lower	Upper
PF	Equal variances assumed	0.922	0.343	2.09	37	0.044	0.091	0.044
PF	Equal variances not assumed			2.07	33.57	0.046	0.091	0.044

Table 7-13 Outputs of the Normality and t-Tests for Projects Grouped Based on Material Management Practices Scores

Since the variances are equal, the results in the first row of the t-test output are considered. Accordingly, the null hypothesis for this t-test or the means of the groups are equal is rejected as p = 0.04 < 0.05. Therefore, there is a statistically significant difference between the Group 1 and Group 2 projects, and the scoring tools for construction materials management practices are valid. The box-plots of the two groups are drawn to visually represent the difference between the PFs (Figure 7-2). The mean PF of Group-1 is approximately 1, and the mean PF of Group-2 is 0.91.



Figure 7-2 Box-Plots of the PFs of the Projects Based on the Construction Materials Management Practices Scores

Validity Test of the Scoring Tools of the Construction Equipment Management Practices

To validate the scoring tools of the construction equipment management practices, the mean construction equipment management practices score = 0.93, the mode score=0.58, the median score=0.89, the skewness=0.157, standard error= 0.378, and Z= (0.157/0.378) = 0.41< 1.96 are computed. The mean score (0.93) is chosen as the baseline for classifying projects since the data can be considered as normal (Z<1.96). The projects are then grouped into Group 1(high score), and Group 2 (low score) based on their construction equipment management practices scores, and statistical analysis is conducted to check the difference between the groups. Normality test is conducted for the two groups, and the results are presented in Table 7-14. Accordingly, both groups' p-values are found to be significant (*p-value* for Group 1= 0.043 < 0.5, *p-value* for Group 2 = 0.027 < 0.5) which implies that the data is not normally distributed. Hence, Mann-Whitney test is used instead of t-test or one-way ANOVA.

		Descr	riptive	<u> </u>	Test of Normality					
				95% CI for	r Mean PF		Shapiro	-Wilk		
	Ν	Mean PF Std. Error		Lower	Upper	Statistic df			Sig	
_					Bound		Statistic	ui	Sig	
Group 1	19	1.009	0.027	0.953	1.066	Group 1	0.897	19	0.043	
Group 2	20	0.896	0.033	0.827	0.964	Group 2	0.89	20	0.027	

 Table 7-14 Outputs of the Normality Test for Projects Grouped Based on the Scores of Construction Equipment Management Practices

The test statistic for Mann-Whitney U test is 98, the asymptotic p-value is 0.01, and the exact *p-value* is 0.009. The mean rank for Group 1 and Group 2 are 24.84 and 14.40 respectively. The conclusion to be drawn from the Mann-Whitney U test is the rejection of the null hypothesis (the two groups are the same) as the *p-value* is less than 0.05.

173

Therefore, there is a significant difference between the productivity factors of the two groups, and the scoring tools of the construction equipment management practices are valid. Box plots are drawn to visualise the difference between Group 1 and Group 2 projects (Figure 7-3). The mean PF for Group 1 and Group 2 are 1.01 and 0.90 respectively.



Figure 7-3 Box-plots of the PFs of Projects with Low and High Construction Equipment Management Practice Scores

Validity Test of the Scoring Tools of the Management Practices Related to Construction Methods

The mean score of the management practices related to construction methods, the median score, the skewness, and standard error for skewness are computed to determine a baseline score which is used to divide projects into high-score and low-score based on their management practices related to construction methods' scores. Accordingly, mean score =4.84, median score=4.93, skewness=-0.32, and standard error for skewness=0.38 are found. Since Z=|-0.32/0.38|=0.84<1.96, the data is considered as normal, and the mean score (4.84) is chosen as a measure of central tendency. The projects are then divided into two groups using the mean score as a baseline. The first group (Group 1 or high score) are projects that have the scores of the management practices related to construction methods greater than the mean score (4.84), and the second group (Group 2 or low score) are projects that have the scores less than the mean value.

The results of the normality test and the descriptive statistics are shown in Table 7-15. There are 21 projects in Group 1 and 18 projects in Group 2, and the mean productivity factor (PF) of Group 1 and Group 2 are 0.88 and 1.01 respectively. The following hypotheses are tested using Shapiro-Wilk analysis H_0 : the PF data is normally distributed and H_a : the data is not normally distributed. Accordingly, for Group 1, the *p*-

¹⁷⁴

value is 0.008 < 0.05 and the null hypothesis is rejected or the PFs data for Group 1 is not normally distributed. For group 2, the *p-value* is 0.108 > 0.05, and the null hypothesis is accepted indicating that the data is normally distributed. However, since both groups are not normally distributed non-parametric tests (Mann-Whitney test) is chosen to check if there is a statistically significant difference between the productivity factors of Group 1 and Group 2 projects and to validate the scoring tools for management practices related to construction methods.

	manage			u to constru		nous					
		Descriptive			Test of Normality						
			95% CI f	ôr Mean PF		Shapi	iro-Wilk				
N Mean PF		Lower Bound	Upper Bound		Statistic	df	Sig				
Group 1	21	1.01	0.96	1.06	Group 1	0.867	21	0.008			
Group 2	18	0.88	0.81	0.96	Group 2	0.916	18	0.108			

 Table 7-15 Outputs of the Normality Test of Projects Grouped Based on the Scores of Management Practices Related to Construction Methods

The result of the Mann-Whitney test shows a *p-value* of 0.003 < 0.05 which indicates that there is a statistically significant difference between the two groups. The finding implies that projects with higher levels of implementation of the management practices related to construction methods also have higher productivity. Therefore, the scoring tools of the management practices related to construction methods are valid. Finally, as shown in Figure 7-4, the PFs of the projects are plotted to visualise the difference between the two groups.



Figure 7-4 Box-Plot of Projects with Low and High Scores of the Management Practices Related to Construction Methods

Validity Test of the Scoring Tools of the Pre-Construction Phase Management Practices To validate the scoring tools of the preconstruction phase management practices, the projects are grouped into high and low scores by using the mean preconstruction phase management practices score of 4.52 since the Z value of the data (1.26) is less than the reference Z-value (1.96). Projects with preconstruction phase management practices scores greater than or equal to 4.52 are grouped under Group 1(high score) and projects with scores less than 4.52 are grouped under Group 2 (low score). To decide the type of test to be used for analysing the difference between the projects grouped based on their preconstruction phase management practices scores, normality test is conducted, and the results are shown in Table 7-16. The Shapiro-Wilk normality test result for Group 1 indicates a p = 0.07 > 0.05 which implies the data is normally distributed. For group 2, a p = 0.043 < 0.05 which also shows the data is normally distributed. Levene's test is also conducted to check the homogeneity of variance and p = 0.362 > 0.05 is found. This indicates that the variances are homogeneous and another assumption of the parametric test is satisfied. Thus, the t-test can be conducted to investigate the difference between the two groups. The results of the t-test are shown in Table 7-16, and the *p*-value is found to be 0.01 < 0.05. Thus, there is a significant difference between the two groups, and the scoring tools for preconstruction phase management practices are valid. Projects with higher scores of the preconstruction phase management practices have also higher productivity factors, and projects with lower scores have lower productivity factors.

				0				
			Normalit	y Test				
	Descrip	otive				Test of N	lormality	
						Shapi	ro-Wilk	
	Ν	Mean PF	Minimum PF	Maximum PF		Statistic	df	Sig
Group 1	18	1.01	0.96	1.07	Group 1	0.903	18	0.065
Group 2	21	0.9	0.83	0.96	Group 2	0.905	21	0.043
			t- Te	est				
		t-test for Equ	uality of Mean	IS				
		t	df	Sig.(2- tailed)	Mean	Std. Error Difference	95% Confidence Interval of the Difference	
				uncu)	Difference	Dijjerence	Lower	Upper
DE	Equal variances assumed	2.704	37	0.01	0.115	0.043	0.029	0.201
L L.	Equal variances not assumed	2.756	36.701	0.009	0.115	0.042	0.03	0.2

Table 7-16 Outputs of the Normality and t-Tests of Projects Grouped Based on Pre-Construction Phase Management Practices Scores

To visualise the difference between the two groups, the box-plots of the productivity factors of the projects grouped based on their preconstruction phase management practices scores are shown in Figure 7-5.



Figure 7-5 Box-plots of the PFs of the Projects with High and Low Preconstruction Phase Management Practices Scores

Validity Test of the Scoring Tools of the Human Resource Management Practices To validate the scoring tools for the human resource management (HRM) practices, the mean and the median HRM practices scores are computed, and the central value is identified. Accordingly, median = 4.67, mean = 4.63, skewness = -0.446, *SE* of skewness = 0.378 and *Z*=-0.446/0.378=-1.18 are found. Since the absolute value of the *Z-value* is less than 1.96, the data can be considered as normal, and the mean value (4.63) is used as starting point to divide the data into higher and lower scores. Mean score = 4.63, mean + (5% × mean score) = 4.86, and mean score – (5% × mean score) = 4.39 are used to determine the baseline.

Mann-Whitney U test is conducted to identify if there is a significant difference between Group-1 and Group-2 projects since the data is not normally distributed. In the first trial, the mean value is used as a baseline and insignificant *p-value* (p = 0.09 >0.05) is obtained. During the second trial, using mean+ (5% × mean score) as a baseline, insignificant result (p = 0.051 > 0.05) is found. In the third trial, using mean score – (5% × mean score) =4.39, significant result (p = 0.016 < 0.05) is obtained. Thus, building projects with scores greater than or equal to 4.39 are grouped under Group-1, and those with scores less than 4.39 are grouped under Group-2. The finding shows that projects with higher productivity factors (PFs) also have higher HRM practice scores, and the scoring tools for HRM practices are valid. The mean PF of Group-1 is approximately

1.0, and the mean PF of Group-2 is 0.88. To visualise the difference between the two groups, box-plots indicated in Figure 7-6 are plotted.



Figure 7-6 Box-plots of the PFs of the Projects with High and Low Human Resource Management Practices Scores

Validity Test of the Scoring Tool of the Safety and Health Practices

The three measures of central tendency or mean, median, mode scores of the safety and health practices are calculated, and the cut-off value for grouping them into high and low scores is determined. The mode=6.03, median=5.65, mean=5.43, Skewness=-1.26, standard error (*SE*) of Skewness = 0.38, and *Z*= (Skewness)/ (*SE*) =-3.33 are found. Since the absolute value of *Z* is greater than 1.96, the data is not considered as normally distributed, and mean is not the suitable measures of central tendency. Therefore, the median value is chosen to group projects into a high score (Group-1) and low score (Group-2). Median score = 5.65, median score + (5% × median score) = 5.93, median score - (5% × median score) = 5.37, and median score - (7.5% × median scores) = 5.23 are compared to determine a baseline.

Mann-Whitney U test is used as the data does not fulfil the requirement of the parametric test. In the first trial, the median value is used as a baseline, and insignificant p-value (p=0.053>0.05) is obtained. During the second trial, using median score + (5% × median score) as a baseline, insignificant result (p=0.258>0.05) is found. In the third trial, using median score – (5% × median score) as a baseline, insignificant result (p=0.258>0.05) is found. In the third trial, using median score – (5% × median score) as a baseline, insignificant p-value (0.05) is found. In the fourth trial, using median score – (7.5% × median score) as a baseline, and significant value (p=0.004<0.05) is obtained. Thus, building projects with scores greater than or equal to 5.23 are grouped under Group-1, and those with scores less than 5.23 are classified under Group-2. The scoring tools for safety and health

practices are valid since there is a significant difference between the productivity factors of the projects with high and low safety and health practices' scores. The mean PF of Group-1 is approximately 1.0, and the mean PF of Group-2 is 0.83. The box-plots of the productivity factors of the two groups are shown in Figure 7-7.



Figure 7-7 Box-plots of the PFs of the Projects with High and Low Safety and Health Practices Scores

7.3.3 Reliability Test for the Weighted Scoring Tools

Before conducting correlation and regression analyses, the internal consistency of the scoring tools of the construction management practices was checked by running the reliability test using SPSS-24. Cronbach alpha (α) is the most popular reliability statistic which determines the consistency of items in a survey instrument and the recommended acceptable minimum value of α is 0.70 (Santos, 1999). In this research, the Cronbach alpha value of 0.95 is found which indicates that the scoring tools for the 47 construction management practices are reliable.

The reliability tests for scoring tools of the six categories of the construction management practices were also conducted. Accordingly, Cronbach alpha values of 0.84, 0.74, 0.75, 0.84, 0.85, and 0.75 were obtained for construction materials management practices, construction equipment management practices, management practices related to construction methods, preconstruction phase management practices, human resource management practices, and safety and health practices respectively. The findings indicate that all the Cronbach alpha values are greater than the minimum acceptable value of 0.70. Hence the scoring tools for the six categories of the construction management practices are reliable.

7.4 Results of the Correlation Analyses

7.4.1 Relationship between Productivity and the Construction Management Practices

To investigate the relationships between productivity and the management practices, Spearman correlation analysis is conducted using SPSS-23 since the construction management practices' data (projects' scores of the construction management practices) is not normally distributed and failed to fulfil the assumptions of the parametric tests. The results are interpreted based on Cohen's effect size in which correlation coefficients of 0.1, 0.3 and 0.5 represent weak (small), moderate (medium) and strong (large) associations respectively (Cohen, 2013). Accordingly, correlation coefficient (r_s) < 0.1 implies very weak correlation; $0.1 \le r_s < 0.3$ shows weak (low) association; $0.3 \le r_s \le$ 0.5 implies medium (moderate) relationship; and $r_s > 0.5$ indicates strong (high) association. In Table 7-17, the results of the correlation analysis between productivity and the management practices are presented.

As shown in Table 7-17, all the construction management practices are positively correlated with productivity which confirms the findings of Phase I of this research. However, the correlation between the level of implementation of the construction management practices and productivity performance might not necessarily be a cause-effect relationship. Construction materials management practices have positive, medium strength and statistically significant ($r_s=0.45$, p=0.005<0.05) association with productivity. This implies that as the level of implementation of the construction materials management practices increases productivity also increases. Among the six construction materials management practices, long-lead materials identification has the highest correlation coefficient ($r_s=0.41$, p=0.009). The finding confirms that long-lead materials identification is the most critical materials management practice to improve productivity in multi-storey building projects (ranked as number 1 in section 7.2 in materials status database ($r_s=0.37$, p=0.02) are moderately and positively associated with productivity. The associations are significant at 5% level of significance.

'Construction equipment management practices' is positively and significantly correlated with productivity ($r_s=0.366$, p=0.02<0.05). This finding confirms the finding obtained in Phase I of the research (Chapter 5) in which the three construction 180

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

equipment management practices are found to have a potential to enhance productivity in multi-storey building projects.

Table 7-17 Spearman Correlation	Coefficients	of Productivity	and Construction	Management
	Due	tiona		

	Flactices													
Co	nstruction Mater	rials	Manage	ment Practices	related	Preco	nstruction H	Phase	Hu	ıman Resour	rce	Saf	ety and Hea	ılth
Ma	nagement Pract	ices	to Co	nstruction Meth	ods	Manag	gement Prac	ctices	Mana	igement Pra	ctices		Practices	
N=39		PF	N=39		PF	N=39		PF	N=39		PF	N=39		PF
MMP	Cor. Coeff. (r_s)	.445**	MPRCM	Cor. Coeff.	.440**	РСРВР	$Coeff. \ (r_{\scriptscriptstyle s})$.455**	HRM	Coeff. (r_s)	.364*	SHP	Coeff. (r _s)	.343*
	Sig. (2-tailed)	0.005		Sig. (2-tailed)	0.004		Sig.	0.004		Sig.	0.023		Sig.	0.032
РРМ	Cor. Coeff. (r_s)	.330*	IS	Cor. Coeff.	.466**		$Coeff. \ (r_s)$.355*		Coeff. (rs)	0.190		$Coeff. \left(r_{s} \right)$	0.247
	Sig. (2-tailed)	0.04		Sig. (2-tailed)	0.003	SIP	Sig.	0.027	CC	Sig.	0.246	HK	Sig.	0.130
LLM	Cor. Coeff. (r_s)	.414**	WSS	Cor. Coeff.	0.277		$Coeff. \ (r_s)$.408**		$Coeff. \left(r_s \right)$	0.208		$Coeff. \left(r_s \right)$.335*
LLIII	Sig. (2-tailed)	0.009		Sig. (2-tailed)	0.087	WSW	Sig.	0.01	SA	Sig.	0.205	SHPo	Sig.	0.037
MSD	Cor. Coeff. (r_s)	.367*	SEM	Cor. Coeff.	0.199		$Coeff. \ (r_s)$.368*		Coeff. (r _s)	0.237		$Coeff. \left(r_{s} \right)$.357*
WISD	Sig. (2-tailed)	0.022	SLIVI	Sig. (2-tailed)	0.224	US	Sig.	0.021	ΕT	Sig.	0.147	HSP	Sig.	0.025
MDS	Cor. Coeff. (r_s)	0.233	DSI P	Cor. Coeff.	0.260		Coeff. (r_s)	0.117		Coeff. (r _s)	0.218		Coeff. (r _s)	.323*
WID5	Sig. (2-tailed)	0.153	DOLI	Sig. (2-tailed)	0.110	DP	Sig.	0.478	CD	Sig.	0.182	SWMS	Sig.	0.045
MIP	Cor. Coeff. (r_s)	0.143	тср	Cor. Coeff.	0.189		$Coeff. \ (r_s)$	0.225		Coeff. (r_s)	0.056		$Coeff. \left(r_s \right)$.409**
	Sig. (2-tailed)	0.386	101	Sig. (2-tailed)	0.249	CWP	Sig.	0.169	NFIP	Sig.	0.736	HA	Sig.	0.01
MIT	Cor. Coeff. (r_s)	0.263		Cor. Coeff.	0.032		$Coeff. \ (r_s)$	0.229		$Coeff. \left(r_s \right)$	0.243		$Coeff. \left(r_s \right)$	0.216
	Sig. (2-tailed)	0.106	SSP	Sig. (2-tailed)	0.849	BR	Sig.	0.162	FIP	Sig.	0.135	HST	Sig.	0.187
Cor	struction Equip	nent		Cor. Coeff.	.318*	UA	Coeff. (r_s)	0.299	SoA	Coeff. (r_s)	.367*		Coeff. (r_s)	0.091
Ma	inagement Pract	ices	MDG	6° (2 (1))	0.040		a.	0.065		а.	0.000	TSM	с.	0.50
CEMP	Cor. Coell. (r_s)	0.300	MP 5	Sig. (2-tailed)	0.049	CT	Sig.	0.005	505	Sig.	0.022		Sig.	0.58
	Sig. (2-tailed)	0.022		Cor. Coeff.	.527	CI	Coeff. (r _s)	.329	505	Coeff. (r _s)	.400**			
PPCE	Cor. Coeff. (r_s)	0.435	PSP	Sig. (2-tailed)	0.001		Sig.	0.041		Sig.	0.003			
	Sig. (2-tailed)	0.006		Cor. Coeff.	.425**	MD	Coeff. (r _s)	0.288	CDR	Coeff. (r _s)	0.277			
СЕРА	Cor. Coeff. (r_s)	0.330*	РСР	Sig. (2-tailed)	0.007		Sig.	0.075		Sig.	0.087			
CLIA	Sig. (2-tailed)	0.04		Cor. Coeff.	0.115	DD	Coeff. (rs)	0.168	RPEP	Coeff. (rs)	0.199			
CEM	Cor. Coeff. (r_s)	0.162	INT	Sig. (2-tailed)	0.485	ĸĸ	Sig.	0.307		Sig.	0.224			
CEM	Sig. (2-tailed)	0.324							EI	Coeff. (r _s)	0.26			
-									1.1	Sig	0.11			

Note: PF= Productivity Factor, MMP= Materials Management Practices, PPM=Procurement Plan for Materials, LLM= Long-lead Materials Identification, MSD=Materials Status Database, MDS= Materials Delivery Schedule, MIP=Materials Inspection Process, MIT =Materials Inspection Team, CEMP= Construction Equipment Management Practices, PPCI=Procurement Plans for Construction Equipment, CEPA=Construction Equipment Productivity Analysis, CEM=Construction Equipment Maintenance, MPRCM= Management Practices, Related to Construction Methods, IS=Integrated Schedule, WSS= Work Schedule Strategies, SC=Schedule Control, DSLP= Dynamic Site Layout Plan, TCP=Traffic Control Plan, SSP=Site Security Plan, MPS=Machinery Positioning Strategy, PSP=Project Start-up Plan, PCP= Project Completion Plan, INT=Innovation and new Technologies, PCPMP= Pre-Construction Phase Management Practices, SIP= Short Interval Plan, WSW=Well-defined Scope of Work, US= Use of Software in Planning Work Packages, DP= Dedicated Planner, CWP= Construction Work Package, BR= Buildability Review, UA= Utilities Alignment, CT= Contract Types, MD= Model Development, RR=Regulatory Requirement, HRM= Human Resource Management Practices, CC= Crew Composition, SA=Skill Assessment and Evaluation, ET= Employee Training, CD= Career Development, NFIP= Non-Financial Incentive Programs, SOA= Social Activities, SOS= Stability of Organizational Structure, CDR= Clear Delegation of Responsibility, RPEP=Retention Plan for Experienced Personnel, EI= Exit Interviews, SHP= Safety and Health Planc, SWMS = Safe Work Method Statement, HA=Hazard Analysis, SM= Safety and Health Policy, SHP= Safety and Health Plan, SWMS = Safe work Method Statement, HA=Hazard Analysis, SM= Safety and Health Training, and TSM= Toolbox Safety Meeting,N= Number of Samples, ** Correlation is significant at 0.01, * Correlation is significant at 0.05.

The management practices related to construction methods are positively associated with productivity, and the relationship is statistically significant (r_s =0.44, p=0.004<0.05). The finding corroborates the previous two findings regarding the management practices related to construction methods. In Phase I of the research, it is found that the ten management practices related to construction methods have the potential to improve productivity (Chapter 5), and in Chapter 6, it was found that the management practices related to construction methods are 'very important' (response scale of 4) to enhance productivity in multi-storey building projects.

The preconstruction phase management practices are positively correlated with productivity, and the association is statistically significant ($r_s=0.46$, p=0.004<0.05).

Among the ten preconstruction phase management practices, the practice well-defined scope of work has the highest correlation coefficient (r_s =0.41, p=0.01). This finding confirms the finding obtained in Section 7.2 in which well-defined scope of work is ranked first.

The human resource management (HRM) practices are positively associated productivity, and the relationship is significant (r_s =0.36, p=0.02<0.05) which indicates that as the level of implementation of the HRM practices increases productivity also increases. Among the eleven HRM practices, the stability of organisational structure has the highest correlation coefficient (r_s =0.47, p=0.003<0.05). The finding confirms that maintaining the stability of the organisational structure is the second most important HRM practices (section 7.2).

Safety and health practices have a positive and statistically significant association with productivity (r_s =0.34, p=0.03<0.05). This finding confirms the findings in previous chapters. In Chapter 5, the qualitative data analysis revealed that the seven safety and health practices have the potential to improve productivity in multi-storey building projects. In Chapter 6, the preliminary quantitative data analysis showed that the safety and health practices are 'extremely important' (response scale of 5) to improve productivity in multi-storey building projects.

7.4.2 Relationships among the Construction Management Practices

Correlation analysis is conducted to investigate the inter-relationship among the 47 construction management practices. For the sake of brevity, the table indicating the correlation coefficients of all the practices is provided in the Appendix-4. The relationships among the six categories of the management practices and the inter-relationships among the practices within their respective categories are explained in the following subsections.

Relationships among Materials Management Practices and the Practices in other Categories

Procurement Plans for Materials is strongly associated with: Procurement Plans for Construction Equipment ($r_s=0.51$, p<0.01), Construction Equipment Productivity Analysis ($r_s=0.59$, p<0.01), and Use of Software in Planning Work Packages ($r_s=0.66$, p<0.01). Long -Lead Materials Identification has a strong association with Procurement

Plans for Construction Equipment ($r_s=0.53$, p<0.01). Materials Status Database has a strong correlation with: Procurement Plans for Construction Equipment ($r_s=0.61$, p<0.01), Construction Equipment Productivity Analysis ($r_s=0.52$, p<0.01), and Contract Types ($r_s=0.55$, p<0.01). Materials Delivery Schedule is strongly correlated with Procurement Plans for Construction Equipment ($r_s=0.53$, p<0.01). Material Inspection Process is strongly correlated with Contract Types ($r_s=0.55$, p<0.01).

Relationships among Equipment Management Practices and the Practices in other Categories

Procurement Plans for Construction Equipment has a strong correlation with: Short Interval Plan ($r_s=0.65$, p<0.01), Contract Types ($r_s=0.58$, p<0.01), Regulatory Requirement ($r_s=0.57$, p<0.01), Stability of Organizational Structure ($r_s=0.57$, p<0.01), Safety and Health Policy ($r_s=0.67$, p<0.01), and Safety and Health Training ($r_s=0.63$, p<0.01). Construction Equipment Productivity Analysis has a strong association with: Short Interval Plan ($r_s=0.63$, p<0.01), Stability of Organization Structure ($r_s=0.54$, p<0.01), and Safety and Health Policy ($r_s=0.64$, p<0.01). Construction Equipment Maintenance is strongly associated with: Safety and Health Policy ($r_s=0.50$, p<0.01), and Hazards Analysis ($r_s=0.50$, p<0.01).

Relationships among Management Practices to Construction Methods and the Practices in other Categories

Integrated Schedule has strong association with: Utilities Alignment ($r_s=0.52$, p<0.01), Contract Types ($r_s=0.56$, p<0.01), and Hazards Analysis ($r_s=0.57$, p<0.001). Work Schedule Strategies has strong association with: Procurement Plans for Materials ($r_s=0.56$, p<0.01), Contract Types ($r_s=0.50$, p<0.01), and Skill Assessment and evaluation ($r_s=0.53$, p<0.01). Schedule Control has strong association with: Buildability Review ($r_s=0.67$, p<0.01), and Contract Types ($r_s=0.71$, p<0.01). Dynamic Site Layout Plan has strong correlation with: Safe Work Method Statement ($r_s=0.50$, p<0.01), and Procurement Plans for Construction ($r_s=0.51$, p<0.01). Traffic Control Plan has strong associations with: Short Interval Plan ($r_s=0.54$, p<0.01), Use of Software in Planning Work Packages ($r_s=0.51$, p<0.01), Construction Work Packages ($r_s=0.50$, p<0.01), Regulatory Requirement ($r_s=0.53$, p<0.01), Clear Delegation of Responsibility ($r_s=0.51$, p<0.01), Safe Work Method Statement ($r_s=0.55$, p<0.01), and Hazards Analysis ($r_s=0.54$, p<0.01). Site Security Plan has a strong correlation with: Short Interval Plan

($r_s=0.52$, p<0.01), and Construction Work Packages ($r_s=0.53$, p<0.01). Machinery Positioning Strategy is strongly correlated with: Utilities Alignment ($r_s=0.51$, p<0.01), and Financial Incentive Programs ($r_s=0.57$, p<0.01). Project start-up plan is strongly associated with: Construction Work Packages ($r_s=0.51$, p<0.01), Buildability Review ($r_s=0.60$, p<0.01), Contract Types ($r_s=0.68$, p<0.01), Social Activities ($r_s=0.64$, p<0.01), Exit interviews ($r_s=0.50$, p<0.01), and Hazards Analysis ($r_s=0.52$, p<0.01). Project Completion Plan is strongly related with: Long-Lead Materials Identification ($r_s=0.54$, p<0.01), Materials Inspection Team ($r_s=0.54$, p<0.01), Procurement Plans for Construction Equipment ($r_s=0.53$, p<0.01), Short Interval Plan ($r_s=0.57$, p<0.01), Contract Types ($r_s=0.61$, p<0.01), Regulatory Requirement ($r_s=0.52$, p<0.01), Crew Composition ($r_s=0.51$, p<0.01), Employees Training ($r_s=0.53$, p<0.01), Career Development ($r_s=0.55$, p<0.01), Exit interviews ($r_s=0.51$, p<0.01) and Hazards Analysis ($r_s=0.57$, p<0.01). Innovations and New Technologies has a strong association with: Contract Types ($r_s=0.54$, p<0.01) and Exit interviews ($r_s=0.53$, p<0.01).

Relationships among Pre-Construction Phase Management Practices and the Practices in other Categories

Short Interval Plan has a strong association with: Crew Composition ($r_s=0.61$, p<0.01), Skill Assessment ($r_s=0.59$, p<0.01), and Safety and Health Training ($r_s=0.59$, p<0.01). Construction Work Packages is strongly correlated with: Crew Composition ($r_s=0.55$, p<0.01), and Skill Assessment ($r_s=0.52$, p<0.01). Utilities Alignment has a strong association with: Skill Assessment ($r_s=0.55$, p<0.01), Career Development ($r_s=0.51$, p<0.01), and Financial Incentive Programs ($r_s=0.53$, p<0.01). Contract Types and Exit Interviews are strongly related ($r_s=0.57$, p<0.01). Regulatory Requirement is strongly correlated with Skill Assessment ($r_s=0.51$, p<0.01).

Relationships among Human Management Practices and the Practices in other Categories

Skill Assessment and Safety and Health Training have a strong relationship (r_s =0.51, p<0.01). Financial Incentive Programs has a strong correlation with Safety and Health Training (r_s =0.58, p<0.01). Social Activities and Safety and Health Plan are associated strongly (r_s =0.55, p<0.01). Stability of Organization Structure is strongly correlated with: Safety and Health Policy (r_s =0.50, p<0.01), and Safety and Health Plan (r_s =0.53,

p < 0.01). The practice exit interview has a very strong association with safety and health plan (r_s=0.85, p<0.01).

Relationships among Safety and Health Practices and the Practices in other Categories Health and Safety Policy has a strong association with Procurement Plan for Construction Equipment ($r_s=0.66$, p<0.01). The relationship between Safe Work Method Statement and Traffic Control Plan is strong ($r_s=0.55$, p<0.01). Health and Safety Training and Procurement Plans for Construction Equipment are strongly correlated ($r_s=0.63$, p<0.01).

Summary of the Relationships among the Six Categories of the Management Practices

The summary of the relationships among the six categories of the management practices is presented in Table 7-18. The finding indicates that all the six categories of the construction management practices are positively and strongly correlated, and the relationships are statistically significant at 1% level of significance. However, the correlation coefficient between 'management practices related to construction methods' and 'preconstruction phase management practices' is the highest (0.88); and the Spearman correlation coefficient between 'safety and health practices' and 'construction materials management practices' is the lowest (0.50).

	e						
		MRCM	CMMP	CEMP	PPMP	HRMP	SHP
MRCM	Coeff. (r _s)	1.000					
	Sig.						
CMMP	Coeff. (r_s)	.664**	1.000				
	Sig.	.000					
CEMP	Coeff. (r _s)	.643**	.704**	1.000			
	Sig.	.000	.000				
PPMP	Coeff. (r _s)	.878**	.655**	.763**	1.000		
	Sig.	.000	.000	.000			
HRMP	Coeff. (r _s)	.701**	.584**	.682**	.701**	1.000	
	Sig.	.000	.000	.000	.000		
SHP	Coeff. (r _s)	.507**	.498**	.660**	.598**	.586**	1.000
	Sig.	.001	.001	.000	.000	.000	

 Table 7-18 Spearman Correlation Coefficients of the Six Categories of the Construction

 Management Practices

Legend: MRCM=Management Practices Related to Construction Methods, CMMP= Construction Materials Management Practices, CEMP = Construction Equipment Management Practices, PPMP= Preconstruction Phase Management Practices, HRMP = Human Resource Management Practices, SHP= Safety and Health Practices, ** Correlation is significant at 0.01.

The inter-relationships among the practices within their respective categories are described in the following subsections.

Inter-relationships among Construction Materials Management Practices

In Table 7-19, the results of the correlation analysis of the materials management practices are presented. Accordingly, all the six construction materials management practices are positively correlated. Procurement plan for materials is strongly and positively associated with: long-lead material identification ($r_s=0.64$, p<0.001), materials status database ($r_s=0.53$, p=0.001), and materials inspection team ($r_s=0.61$, p<0.001). The relationships are statistically significant at 1% level of significance. Long-lead materials identification is strongly and significantly correlated with materials status database ($r_s=0.57$, p<0.001). Materials status database is strongly associated with materials inspection team ($r_s=0.64$, p<0.001). Materials inspection process is strongly and significantly correlated with materials inspection team ($r_s=0.64$, p<0.001). Materials inspection process is strongly and significantly correlated with materials inspection team ($r_s=0.64$, p<0.001).

	Practices											
N=39		PPM	LLM	MSD	MDS	MIP	MIT					
	Sig. (2-tailed)											
	Cor. Coeff. (r _s)	1.000										
PPM	Sig. (2-tailed)											
	Cor. Coeff. (r _s)	.644**	1.000									
LLM	Sig. (2-tailed)	.000										
	Cor. Coeff. (r _s)	.534**	.565**	1.000								
MSD	Sig. (2-tailed)	.000	.000									
	Cor. Coeff. (r _s)	.336*	.423**	.345*	1.000							
MDS	Sig. (2-tailed)	.037	.007	.032								
	Cor. Coeff. (r _s)	.366*	.201	.437**	.027	1.000						
MIP	Sig. (2-tailed)	.022	.219	.005	.868							
	Cor. Coeff. (r _s)	.605**	.496**	.644**	.399*	.690**	1.00					
MIT	Sig. (2-tailed)	.000	.001	.000	.012	.000	-					

Table 7-19 Spearman's Correlation Coefficients of the Construction Materials Management

Note: PPM=Procurement Plan for Materials, LLM= Long-lead Materials Identification, MSD=Materials Status Database, MDS= Materials Delivery Schedule, MIP=Materials Inspection Process MIT =Materials Inspection Team, N= Number of Samples, ** Correlation is significant at 0.01, * Correlation is significant at 0.05.

Inter-relationships among Construction Equipment Management Practices

The correlation analysis revealed that all the three construction equipment management practices are interrelated. Procurement plan for construction equipment is strongly correlated with construction equipment productivity analysis (r_s =0.772, p<0.001). The relationship is statistically significant at 1% level of significance. The practices procurement plan for construction equipment and construction equipment maintenance have statistically significant and medium strength association (r_s =0.424, p=0.01<0.05). Construction equipment productivity analysis is moderately correlated with construction equipment maintenance (r_s =0.419, p=0.008<0.05).

Inter-relationships among Management Practices Related to Construction Methods

All the ten management practices related to construction methods are positively correlated with each other (Table 7-20). Accordingly, integrated schedule and schedule control (r_s =0.569, p<0.001); work schedule strategy and project completion plan (r_s =0.514, p=0.001); dynamic site layout plan and machinery positioning strategy (r_s =0.604, p<0.001); and traffic control plan and site security plan (r_s =0.734, p<0.001) have strong and statistically significant relationships. The findings indicate that projects that have implemented a higher level of one management practice that is related to construction methods also implement higher levels of other practices that are related to construction methods.

Table	7-20	Spearman's	Correlation	Coefficients	of the	Management	Practices	Related	to
		Construction	Methods			-			
		IS	WSS		TCP	SSP MPS	DSD D	CP INT	7

		IS	WSS	SC	DSLP	TCP	SSP	MPS	PSP	PCP	INT
	Cor. Coeff.	1.00									
IS	Sig. (2-tailed)	-									
10	Cor. Coeff.	.491**	1.00								
WSS	Sig. (2-tailed)	.001	-								
	Cor. Coeff.	.569**	.424**	1.00							
SEM	Sig. (2-tailed)	.000	.007	-							
	Cor. Coeff.	.476**	.431**	.452**	1.00						
DSLP	Sig. (2-tailed)	.002	.006	.004	-						
ТСР	Cor. Coeff.	.227	.240	.175	.381*	1.00					
	Sig. (2-tailed)	.165	.141	.288	.017	-					
	Cor. Coeff.	.211	.343*	.193	.263	.734**	1.00				
SSP	Sig. (2-tailed)	.197	.033	.239	.106	.000	-				
	Cor. Coeff.	.423**	.279	.362*	.604**	.454**	.395*	1.00			
MPS	Sig. (2-tailed)	.007	.086	.023	.000	.004	.013	-			
	Cor. Coeff.	.468**	.243	.214	.240	.452**	.286	.264	1.00		
PSP	Sig. (2-tailed)	.003	.137	.192	.142	.004	.077	.105	-		
	Cor. Coeff.	.367*	.514**	.258	.250	.426**	.487**	.417**	.418**	1.00	
РСР	Sig. (2-tailed)	.022	.001	.113	.124	.007	.002	.008	.008	-	
	Cor. Coeff.	.201	.225	.106	.275	.256	.247	.380*	.268	.473**	1.00
INT	Sig. (2-tailed)	.220	.168	.521	.091	.115	.130	.017	.099	.002	-

Note: IS=Integrated Schedule, WSS= Work Schedule Strategies, SC=Schedule Control, DSLP= Dynamic Site Layout Plan, TCP=Traffic Control Plan, SSP=Site Security Plan, MPS=Machinery Positioning Strategy, PSP=Project Start-up Plan, PCP= Project Completion Plan, INT=Innovation and new Technologies. ** Correlation is significant at 0.01; * Correlation is significant at 0.05

Inter-relationships among Pre-Construction Phase Management Practices

In Table 7-21, the results of the correlation analysis of the pre-construction phase management practices are presented. Accordingly, all the ten preconstruction phase management practices are inter-related. However, the relationships between: short interval plan and construction work package ($r_s=0.59$, p<0.001); short interval plan and regulatory requirement ($r_s=0.52$, p=0.001); short interval plan and use of software in planning work packages ($r_s=0.51$, p=0.001); well-defined scope of work and contract types ($r_s=0.51$, p=0.001); construction work packages and buildability review ($r_s=0.51$, $r_s=0.51$, $r_s=0.51$, p=0.001); more packages and buildability review ($r_s=0.51$, $r_s=0.51$

p=0.001); and buildability review and contract types ($r_s=0.54$, p<0.001) are strong and statistically significant.

N=39		SIP	WSW	US	DP	CWP	BR	UA	CT	MD	RR
	Coeff. (r _s)	1.000									
SIP	Sig.										
	Coeff. (r_s)	.317*	1.000								
WSW	Sig.	.049									
	Coeff. (r_s)	.505**	.352*	1.000							
US	Sig.	.001	.028								
	Coeff. (r _s)	.424**	.051	.433**	1.000						
DP	Sig.	.007	.756	.006							
	Coeff. (r _s)	.591**	.416**	.410**	.113	1.000					
CWP	Sig.	.000	.009	.010	.495						
	Coeff. (r _s)	.468**	.381*	.415**	.091	.501**	1.00				
BR	Sig.	.003	.017	.009	.580	.001	-				
UA	Coeff. (r _s)	.458**	.243	.340*	.130	.227	.401*	1.00			
	Sig.	.003	.136	.034	.431	.164	.012	-			
СТ	Coeff. (r _s)	.384*	.510**	.329*	.129	.399*	.542**	.417**	1.00		
	Sig.	.016	.001	.041	.433	.012	.000	.008	-		
MD	Coeff. (r _s)	.106	.292	.097	.186	.124	.008	.023	.058	1.00	
	Sig.	.522	.072	.556	.257	.451	.960	.892	.728	-	
RR	Coeff. (r _s)	.517**	.148	.448**	.388*	.359*	.372*	.483**	.364*	.102	1.00
	Sig.	.001	.368	.004	.015	.025	.020	.002	.023	.537	-

Table 7-21 Spearman's Correlation Coefficients of Pre-Construction Phase Management Practices

Legend: SIP= Short Interval Plan, WSW=Well-defined Scope of Work, US= Use of Software in Planning Work Packages, DP= Dedicated Planner, CWP= Construction Work Package, BR= Buildability Review, UA= Utilities Alignment, CT= Contract Types, MD= Model Development, RR=Regulatory Requirement

Inter-relationships among Human Resource Management Practices

Table 7-22 presents the results of the correlation analyses of the human resource management practices. Accordingly, all the eleven HRM practices are interrelated. The relationships between crew composition and skill assessment and evaluation (r_s =0.63, p<0.001); crew composition and employees training (r_s =0.59, p<0.001); crew composition and financial incentive programs (r_s =0.63, p=0.001); skill assessment and evaluation and financial incentive program (r_s =0.53, p=0.001); career development and financial incentive program (r_s =0.53, p=0.001); career development and social activities (r_s =0.53, p=0.001); career development and exit interview (r_s =0.52, p=0.001); Non-financial incentive programs and financial incentive programs (r_s =0.57, p<0.001); stability of organizational structure and retention plan for experienced personnel (r_s =0.57, p<0.001) are strong and statistically significant.

Table 7-22 Spearman Correlation Coefficients of Human Resource Management Practices

		CC	SA	ET	CD	NFIP	FIP	SoA	SOS	CDR	RPEP	EI
	Coeff. (r _s)	1.000										
CC	Sig.											
	Coeff. (r _s)	.630**	1.000									
SA	Sig.	.000										
	Coeff. (r _s)	.587**	.399*	1.000								
ET	Sig.	.000	.012									
	Coeff. (r _s)	.433**	.487**	.273	1.000							
CD	Sig.	.006	.002	.093								
	Coeff. (r _s)	.371*	.320*	.134	.338*	1.000						
NFIP	Sig.	.020	.047	.414	.035							
	Coeff. (r _s)	.506**	.525**	.307	.566**	.573**	1.000					
FIP	Sig.	.001	.001	.057	.000	.000						
SoA	Coeff. (r _s)	.290	.276	.167	.528**	.486**	.493**	1.000				
	Sig.	.073	.089	.310	.001	.002	.001					
SOS	Coeff. (r _s)	.314	.352*	.420**	.275	.140	.263	.369*	1.000			
	Sig.	.051	.028	.008	.090	.394	.106	.021				
CDR	Coeff. (r_s)	.459**	.347*	.214	.334*	.018	.174	.048	.491**	1.000		
	Sig.	.003	.031	.190	.038	.912	.290	.770	.001			
RPEP	Coeff. (r _s)	.246	.136	.242	.323*	.208	.308	.334*	.569**	.424**	1.000	
	Sig.	.131	.410	.138	.045	.205	.056	.038	.000	.007		
EI	Coeff. (r _s)	.332*	.145	.301	.522**	.384*	.376*	.430**	.476**	.431**	.452**	1.000
	Sig.	.039	.380	.063	.001	.016	.018	.006	.002	.006	.004	

Legend: CC= Crew Composition, SA=Skill Assessment and Evaluation, ET= Employee Training, CD = Career Development, NFIP= Non-Financial Incentive Programs, FIP= Financial Incentive Programs, SoA= Social Activities, SOS= Stability of Organizational Structure, CDR= Clear Delegation of Responsibility, RPEP=Retention Plan for Experienced Personnel, EI= Exit Interviews, ** Correlation is significant at 0.01, * Correlation is significant at 0.05.

Inter-relationships among Safety and Health Practices

In Table 7-23, the results of the correlation analysis of the safety and health practices are presented. Accordingly, all the safety and health practices are positively correlated. The relationships between 'safety and health policy' and 'safety and health plan' (r_s =0.57, p<0.001); and 'safety and health policy' and 'safety and health training' (r_s =0.66, p<0.001) are strong and statistically significant at 1% level of significance.

		HK	SHPo	SHP	SWMS	HA	SHT	TSM
НК	Coeff. (r _s) Sig.	1.000						
	Coeff. (rs)	.463**	1.000					
SHPo	Sig.	.003						
	Coeff. (rs)	.335*	.571**	1.000				
HSP	Sig.	.037	.000					
	Coeff. (r _s)	.362*	.250	.221	1.000			
SWMS	Sig.	.023	.124	.176				
	Coeff. (rs)	.314	.448**	.144	.400*	1.000		
HA	Sig.	.052	.004	.383	.012			
	Coeff. (r _s)	.296	.664**	.423**	.086	.398*	1.000	
HST	Sig.	.067	.000	.007	.603	.012		
TSM	Coeff. (rs)	.004	.300	.235	.200	.016	.400*	1.000
	Sig.	.981	.064	.150	.222	.925	.012	

Table 7-23 S	pearman Correl	lation Co	efficients o	of Safety and	Health Practices

Note: HK= Housekeeping, SHPo = Safety and Health Policy, SHP= Safety and Health Plan, SWMS = Safe Work Method Statement, HA= Hazard Analysis, SHT= Safety and Health Training, and TSM= Toolbox Safety Meeting, ** Correlation is significant at 0.01, * Correlation is significant at 0.05.

7.4.3 Relationship among Construction Management Practices, Company Profiles and Project Characteristics

Correlation analysis is conducted to investigate the relationships among the construction management practices, project delay, project cost, and company profiles. The results of the analysis are indicated in Table 7-24.

Table 7-24 Spearman Correlation Coefficients of Project Characteristics, Company Profiles, and Construction Management Practices

			e					
		PF	Project Cost	Project Delay	Annual Turnover	Company Experience	Company Size	СМР
Project Cost	Coeff. Sig.	.401* .011	1.000					
Project Delay	Coeff. Sig.	940** .000	281 .084	1.000				
Annual	Coeff.	.118	.356	041	1.000			
Turnover	Sig.	.575	.081	.847				
Company	Coeff.	.141	.547**	022	.062	1.000		
Experience	Sig.	.399	.000	.898	.769			
Company Size	Coeff.	.177	.593**	122	.565**	.536**	1.000	
	Sig.	.280	.000	.460	.003	.001		
CMP	Coeff.	.490**	.615**	422**	.122	.307	.291	1.000
	Sig.	.002	.000	.007	.562	.061	.072	

Legend: PF= Productivity Factor, CMP=Construction Management Practices, ** Correlation is significant at 0.01, * Correlation is significant at 0.05.

As shown in Table 7-24, project cost and the construction management practices are positively and strongly correlated, and the relationship is statistically significant ($r_s=0.62$, p<0.001). The finding suggests that the increase in the building project costs due to the increment of the scope of works could be one of the reasons for the variations in the implementation levels of the construction management practices.

Project delay and construction management practices are negatively correlated (r_s =-0.42, p=0.01) which indicates that higher levels of implementation of the practices are associated with low project delays. Project delay and productivity have a very strong negative correlation coefficient (r_s =-0.94, p<0.001) which implies that increasing projects' productivity can reduce project delays.

Project costs and productivity have a medium and significant association ($r_s=0.40$, p=0.01). This finding indicates that projects with higher costs also have higher productivity. Productivity is positively but weakly associated with company experience ($r_s=0.12$, p=0.58), annual turnover ($r_s=0.14$, p=0.40), and company size ($r_s=0.18$, p=0.28). Moreover, the relationships are not statistically significant.

Company size and the construction management practices have medium strength association, but the relationship is not significant ($r_s=0.31$, p=0.06>0.05). Annual turnover and company experience have weak and statistically insignificant relationships with the construction management practices (Table 7-24).

The relationships among the six categories of the construction management practices, company profiles, and project characteristics are explained in the following subsections.

Relationships among Construction Materials Management Practices, Company Profiles, and Project Characteristics

The relationships among project delay, project cost, company experience, annual turnover, company size and the construction materials management practices are analysed, and the results are presented in Table 7-25. Accordingly, all the six construction materials management practices are negatively correlated with project delay. This indicates that as the levels of implementation of the materials management practices increase, the possibility of delaying the completion time of the building projects decreases. Long-lead materials identification has a medium and statistically significant correlated with project delay (r_s =-0.30, p=0.06). Materials inspection team (r_s =-0.22, p=0.18) and material inspection process (r_s =-0.16, p=0.32) are weakly correlated with project delay.

 Table 7-25 Spearman Correlation Coefficients of Company Profiles, Project Characteristics, and Construction Materials Management Practices

		MMP	PPM	LLM	MSD	MDS	MIP	MIT
	Corr. Coeff.	-0.375	-0.277	-0.353	-0.303	-0.129	-0.163	-0.217
Project Delay	Sig.(2-tailed)	0.019	0.088	0.027	0.061	0.434	0.322	0.184
Annual Turnover	Corr. Coeff.	0.056	-0.002	0.097	-0.028	0.019	0.099	-0.026
a	Sig. (2-tailed)	0.790	0.992	0.644	0.895	0.929	0.637	0.900
Company Experience	Corr. Coeff.	0.286	0.514**	0.413**	0.175	0.096	-0.138	0.103
	Sig. (2-tailed)	0.082	0.001	0.010	0.294	0.567	0.409	0.538
Company Size	Corr. Coeff.	0.227	0.340*	0.139	0.059	0.049	0.039	0.093
	Sig. (2-tailed)	0.165	0.034	0.399	0.721	0.767	0.812	0.574
Project Cost	Corr. Coeff. Sig. (2-tailed)	0.474 ^{**} 0.002	0.635 * <0.001	0.296 0.067	0.332* 0.039	0.149 0.367	0.147 0.371	0.355* 0.027

Legend: MMP=Material Management Practice (overall score), PPM=Procurement Plan for Materials, LLM=Long Lead Materials Identification, MSD=Material Status Database, MDS=Materials Delivery Schedule, MIP=Materials Inspection Process, MIT= Materials Inspection Team.

There is a very weak and insignificant correlation between annual turnover and construction materials management practices ($r_s=0.06$, p=0.79>0.05). Company

experience is strongly associated with procurement plans for materials ($r_s=0.51$, p=0.001). Company size is significantly and moderately related to procurement plan for materials ($r_s=0.34$, p=0.03). Positive, moderate, and significant association ($r_s=0.47$, p=0.002) between construction materials management practices and the project cost is found. This implies that the level of implementation of the construction materials management practices increases with project cost. Among the six construction materials management practices, the practice procurement plans for materials is strongly associated with project cost ($r_s=0.64$, p<0.001).

Relationships among Construction Equipment Management Practices, Company Profiles, and Project Characteristics

Correlation analysis is carried out to investigate the relationship among construction equipment management practices, project delay, project cost, annual turnover, company experience and company size (Table 7-26). The project cost is positively correlated with construction equipment management practices (r_s =0.59, p<0.001) which implies that the level of implementation of the construction equipment management practices increases with project cost.

Project delay is negatively correlated with construction equipment management practices (r_s =-0.30, p=0.07) which indicates that the possibility of occurrence of the project delay decreases as the levels of implementation of the equipment management practices increase. Annual turnover (r_s =0.19, p=0.37) and company experience (r_s =0.28, p=0.09) are weakly correlated with construction equipment management practices. Company size is significantly and moderately associated with construction equipment management practices (r_s =0.40, p=0.01<0.05) which implies that larger companies implement higher levels of construction equipment management practices.

		CEMP	PPCE	CEPA	CEM
	Corr. Coeff.	0.585**	0.453**	0.589**	0.418**
Project Cost	Sig. (2-tailed)	< 0.001	0.004	< 0.001	0.008
	Corr. Coeff.	-0.298	-0.358*	-0.227	-0.151
Project Delay	Sig. (2-tailed)	0.065	0.025	0.165	0.358
	Corr. Coeff.	0.188	0.116	0.337	0.006
Annual Turnover	Sig. (2-tailed)	0.368	0.580	0.100	0.977
	Corr. Coeff.	0.276	0.284	0.274	0.266
Company Experience	Sig. (2-tailed)	0.093	0.084	0.096	0.107
	Corr. Coeff.	0.400*	0.251	0.394*	0.379^{*}
Company Size	Sig. (2-tailed)	0.012	0.123	0.013	0.017

Table 7-26 Spearman Correlation	Coefficients of Company	Profiles,	Project	Characteristics,	and
Construction Equipme	ent Management Practices				

Legend: CEMP= Construction Equipment Management Practices, PPCE=Procurement Plans for Construction Equipment, CEPA=Construction Equipment Productivity Analysis, CEM=Construction Equipment Maintenance, ** Correlation is significant at 0.01, * Correlation is significant at 0.05.

Relationships among Management Practices Related to Construction Methods, Company Profiles, and Project Characteristics

Correlation analysis is also conducted to check if the implementation levels of the management practices related to construction methods vary based on project cost, annual turnover of the companies, the experience of the companies and company sizes. The results of the analyses are shown in Table 7-27. Accordingly, all the ten management practices related to construction methods are negatively correlated with building projects' delays. This finding indicates that project with a low level of implementation of the practices or low scores is at high risk of incurring project delay. In other words, high levels of implementation of the management practices related to construction methods are associated with low project delay. Among the ten practices, project start-up plan (r_s =-0.524, p=0.001) is strongly related to project delay; and integrated schedule (r_s =-0.470, p=0.003), project completion plan (r_s =-0.366, p=0.022), and machinery positioning strategy (r_s =-0.349, p=0.030) are moderately and significantly correlated with project delays. Work schedule strategies (r_s =-0.204, p=0.213), dynamic site layout plan ($r_s=-0.262$, p=0.107), schedule execution and management (r_s =-0.195, p=0.234), and traffic control plan (r_s =-0.177, p=0.280) are weakly associated with project delay. Site security plan (r_s =-0.011, p=0.949), and innovations and new technologies (r_s =-0.079, p=0.635) have a very weak association with project delay.

193

 Table 7-27 Spearman Correlation Coefficients of Company Profiles, Project Characteristics and Management Practices Related to Construction Methods.

		IS	WSS	SC	DSLP	TCP	SSP	MPS	PSP	PCP	INT	TS
Project	Corr.Coeff.	-0.470	-0.204	-0.195	-0.262	-0.177	-0.011	-0.349	-0.524	-0.366	-0.079	-0.425
Delay	Sig.(2-tailed)	0.003	0.213	0.234	0.107	0.280	0.949	0.030	0.001	0.022	0.635	0.007
Annual Turnover	Corr.Coeff.	0.263	0.346	-0.384	0.39	0.011	-0.141	-0.064	0.087	-0.059	0.111	0.071
	Sig.(2-tailed)	0.204	0.090	0.058	0.054	0.957	0.502	0.762	0.680	0.780	0.597	0.735
Company	Corr.Coeff.	0.274	0.181	0.270	0.240	0.231	0.302	-0.065	.331*	0.150	.327*	0.313
Experience	Sig.(2-tailed)	0.096	0.277	0.101	0.147	0.162	0.065	0.700	0.042	0.370	0.045	0.056
	Corr.Coeff.	0.381*	0.284	0.289	0.250	0.001	0.053	-0.015	0.102	0.129	0.274	0.274
Company Size	Sig.(2-tailed)	0.017	0.080	0.075	0.124	0.998	0.750	0.930	0.536	0.436	0.092	0.092
Project	Corr.Coeff.	0.417**	0.236	0.227	0.414**	0.361*	0.35^{*}	0.284	0.376^{*}	0.319*	0.368*	0.497**
Cost	Sig.(2-tailed)	0.008	0.148	0.164	0.009	0.024	0.029	0.079	0.018	0.047	0.021	0.001

Legend: IS=Integrated Schedule, WSS= Work Schedule Strategies, SC=Schedule Control, DSLP= Dynamic Site Layout Plan, TCP=Traffic Control Plan, SSP=Site Security Plan, MPS=Machinery Positioning Strategy, PSP=Project Start-up Plan, PCP= Project Completion Plan, INT=Innovation and new Technologies, TS= Total Score of the best practices

The management practices related to construction methods are strongly and positively correlated with project cost (r_s =0.50, p=0.001). The finding suggests that one of the main reasons for variations in the levels of implementation of the management practices related to construction methods could be the change in project costs. As a project cost increases due to increment in the scope of works, the level of implementation of the management practices related to construction methods also increases.

Company's annual turnover is not associated with the management practices related to construction methods ($r_s=0.071$, p=0.735). Company experience is moderately correlated with management practices related construction methods ($r_s=0.313$, p=0.056). However, the association is not statistically significant. Company size is also moderately and insignificantly correlated ($r_s=0.274$, p=0.092) with the management practices related to construction methods. The finding suggests that larger and experienced companies may not implement higher levels of the practices if the scope of a project is small. Among the ten practices, integrated schedule ($r_s=0.381$, p=0.017) is significantly associated with company size indicating that larger companies integrate various project schedules than smaller companies.

Relationships among Preconstruction Phase Management Practices, Company Profiles, and Project Characteristics

The results of the correlation analysis among the pre-construction phase management practices, project delay, project cost, and company profiles are presented in Table 7-28. Accordingly, negative association (r_s =-0.37, p=0.02<0.05) between the preconstruction

phase management practices and project delay is found which indicates that implementation of the preconstruction phase management practices could reduce project delays. Among the ten pre-construction phase management practices, the practice 'well-defined scope of work' has the highest correlation coefficient (r_s =-0.43, p=0.007<0.05) which is statistically significant, and dedicated planner has the lowest association (r_s =-0.02, p=0.91) with the project delay. Contract types (r_s =-0.33, p=0.04) and use of software in planning work packages (r_s =-0.32, p=0.049) have medium associations with project delay; model development (r_s =-0.32, p=0.05), short interval plan (r_s =-0.24, p=0.14), utilities alignment (r_s =-0.22, p=0.18), construction work packages (r_s =-0.13, p=0.42), and buildability review (r_s =-0.18, p=0.28) are weakly correlated with project delay.

 Table 7-28 Spearman Correlation Coefficients of Company Profiles, Project Characteristics, and Pre-Construction Phase Management Practices

		PPMP	SIP	WSW	US	DP	CWP	BR	UA	CT	MD	RR
Project Delay	Coeff.	37*	24	43**	32*	02	13	18	22	33*	26	07
	Sig.	.02	.14	.007	.05	.91	.42	.28	.18	.04	.11	.68
Project Cost	Coeff.	.59**	.57**	.19	.48**	.54**	.36*	.28	.38*	.38*	.23	.26
	Sig.	.000	.000	.24	.002	.000	.02	.09	.02	.02	.16	.12
Annual Turnover	Coeff.	.09	.00	.27	22	.12	.15	08	.07	.43*	16	16
	Sig.	.66	.99	.19	.29	.58	.47	.70	.74	.03	.43	.44
Company Experience	Coeff.	.28	.36*	.05	.18	.48**	08	05	.25	.01	07	.41*
	Sig.	.09	.03	.76	.28	.003	.64	.76	.13	.95	.66	.01
Company Size	Coeff.	.19	.37*	02	.03	.42**	05	08	.21	.16	.11	.14
	Sig.	.24	.02	.91	.85	.01	.77	.64	.19	.33	.53	.38

Legend: PPMP= Preconstruction Phase Management Practices, SIP= Short Interval Plan, WSW=Well-defined Scope of Work, US= Use of Software in Planning Work Packages, DP= Dedicated Planner, CWP= Construction Work Package, BR= Buildability Review, UA= Utilities Alignment, CT= Contract Types, MD= Model Development, RR=Regulatory Requirement, *significant at 5%, ** significant at 1%.

Project cost is significantly and strongly correlated with preconstruction phase management practices ($r_s=0.59$, p<0.001) which indicates that as the project cost increases the implementation levels of the practices also increase. Short interval planning ($r_s=0.57$, p<0.001) and dedicated planner ($r_s=0.54$, p<0.001) have statistically significant and strong correlation with project costs. Use of software in planning work packages ($r_s=0.48$, p=0.002), construction work packages ($r_s=0.36$, p=0.02), utilities alignment ($r_s=0.38$, p=0.02), and contract types ($r_s=0.38$, p=0.02) have medium associations with project cost (Table 7-28). Buildability review ($r_s=0.38$, p=0.09), regulatory requirement ($r_s=0.19$, p=0.24) have weak relationships with project cost.

Annual turnover has a very weak correlation with the pre-construction phase management practices (r_s =0.09, p=0.66) which implies the level of implementation of the preconstruction phase management practices is not related to company's turnover. The relationship between company experience and pre-construction phase management practices is weak (r_s =0.28, p=0.09). Similarly, company size is weakly correlated with pre-construction phase management practices (r_s =0.19, p=0.24).

Relationships among Human Resource Management Practices, Company Profiles, and Project Characteristics

In Table 7-29, the results of correlation analyses of project delay, project cost, company profiles and the human resource management (HRM) practices are presented. Among the eleven HRM practices, project delay is strongly associated with the stability of organisational structure (r_s = -0.47, p=0.003) and social activities (r_s =-0.41, p=0.01).

 Table 7-29 Spearman Correlation Coefficients of Company Profiles, Project Characteristics, and HRM Practices

		HRM	CC	SA	ET	CD	NFIP	FIP	SoA	SOS	CDR	RPEP	EI
Project Delay	Coeff.	303	082	102	205	140	077	182	407*	470**	204	195	262
	Sig.	.061	.621	.535	.210	.396	.640	.267	.010	.003	.213	.234	.107
Project Cost	Coeff.	.502**	.293	.405*	.189	.314	.352*	.379*	.404*	.417**	.236	.227	.414**
	Sig.	.001	.070	.011	.248	.051	.028	.017	.011	.008	.148	.164	.009
Annual	Coeff.	0.10	-0.06	0.10	-0.14	0.33	0.02	-0.12	0.02	0.26	0.35	-0.38	0.39
Turnover	Sig.	0.65	0.79	0.65	0.51	0.11	0.92	0.56	0.93	0.20	0.09	0.06	0.05
Company	Coeff.	0.32	0.14	0.28	-0.02	0.09	0.32^{*}	0.26	0.05	0.27	0.18	0.27	0.24
Experience	Sig.	0.05	0.40	0.09	0.90	0.60	0.05	0.11	0.77	0.10	0.28	0.10	0.15
Company	Coeff.	.32*	0.04	0.18	0.01	0.23	0.28	0.28	0.20	.381*	0.28	0.29	0.25
Size	Sig.	0.05	0.83	0.28	0.98	0.17	0.08	0.09	0.23	0.02	0.08	0.08	0.12

Legend: HRM= Best Human Resource Management Practices, CC= Crew Composition, SA=Skill Assessment and Evaluation, ET= Employee Training, CD = Career Development, NFIP= Non-Financial Incentive Programs, FIP= Financial Incentive Programs, SoA= Social Activities, SOS= Stability of Organizational Structure, CDR= Clear Delegation of Responsibility, RPEP=Retention Plan for Experienced Personnel, EI= Exit Interviews, ** Correlation is significant at 0.01, * Correlation is significant at 0.05.

Project cost and HRM practices are strongly correlated ($r_s=0.51$, p=0.001) which implies that projects having larger scope have higher levels of implementation of the HRM practices, and among the 11 practices the 'stability of organisational structure' has the highest correlation coefficient ($r_s=0.42$, p=0.008).

The annual turnover of construction companies is very weakly correlated with HRM practices (r_s =0.1, p=0.65). Company experience is moderately associated with HRM practices (r_s =0.32, p=0.05), but the relationship is not statistically significant. Among the 11 HRM practices, non-financial incentive program has the highest correlation coefficient (r_s =0.32, p=0.047), and employee training (r_s =-0.02, p=0.90) has no 196

relationship with company experience. Company size and HRM practices have significant and moderate strength (r_s =0.32, p=0.04) relationship. This implies that the implementation of the HRM practices is more important for larger contractors than for smaller contractors. Among the eleven HRM practices, the stability of organisational structure has a stronger association with company size (Table 7-29).

Relationships among Safety and Health Practices, Company Profiles, and Project Characteristics

The Spearman correlation coefficients of project delay, project cost, and safety and health practices are presented in Table 7-30. The association between project cost and safety and health practices is positive, and the relationship is significant (r_s =0.46, p=0.003) which implies that the levels of implementation of safety and health practices increase when project costs increase.

 Table 7-30 Spearman Correlation Coefficients of Company Profiles, Project Characteristics and Safety and Health Practices

			-						
		SHP	HK	SHPo	SHP	SWMS	HA	SHT	TSM
	Coeff.	.458**	.304	.482**	.284	.149	.304	.279	.347*
Project Cost	Sig.	.003	.060	.002	.080	.365	.060	.086	.030
	Coeff.	243	212	219	286	273	342*	175	.134
Project Delay	Sig.	.135	.196	.181	.077	.092	.033	.286	.416
Annual Turnover	Coeff.	.185	.275	.242	.341	005	090	136	.078
	Sig.	.376	.184	.244	.095	.982	.667	.518	.712
Company Experience	Coeff.	.329*	.206	.361*	.157	.274	.067	.154	.104
	Sig.	.043	.215	.026	.346	.096	.691	.357	.535
Company Size	Coeff.	.368*	.235	.433**	.147	.068	.198	.407*	.248
	Sig.	.021	.149	.006	.372	.681	.227	.010	.127

Note: SHP= Safety and Health Practices, HK= Housekeeping, SHPo = Safety and Health Policy, SHP= Safety and Health Plan, TSA = Safe Work Method Statement, HA= Hazard Analysis, SHT= Safety and Health Training, and TSM= Toolbox Safety Meeting, ** Correlation is significant at 0.01, * Correlation is significant at 0.05.

Project delay is negatively correlated with safety and health practices (r_s =-0.24, p=0.14). However, the degree of association is weak as well as insignificant. The result indicates that increasing the level of implementation of safety and health practices might not reduce the possibility of occurrence of project delay.

Annual turnover and safety and health practices are positively correlated, but the relationship is not significant ($r_s=0.19$, p=0.38). Company experience and the safety and health practices are positively associated, and the relationship is statistically significant ($r_s=0.33$, p=0.04). The finding shows that experienced companies have better safety and health practices. Among the seven safety and health practices, safety and health policy
Quantitative Data Analysis and Findings

has the strongest association with company experience ($r_s=0.36$, p=0.03). This implies that the more experienced the contractors are, the better their safety policies are.

Company size is also positively associated with safety and health practices ($r_s=0.37$, p=0.02) which indicates that larger construction companies have higher levels of implementation of the safety and health practices. Among the seven practices, safety and health policy has the highest correlation coefficient ($r_s=0.43$, p=0.006). The relationship is also statistically significant at 1% level of significance. Safety and health training has a medium strength as well as statistically significant association with company size ($r_s=0.41$, p=0.01). The findings indicate that larger construction firms also have higher levels of implementation of 'safety and health policy' and 'safety and health training.'

7.5 Summary

After analysing the quantitative data, part of Objective 1 or 'to prioritize the construction management practices that have the potential to enhance productivity in multi-storey building construction projects in the context of Victoria State, Australia," and part of Objective 2 of the study which is 'to validate the scoring tools for measuring, planning, monitoring and evaluating the management practices that could improve productivity in multi-storey building construction projects' are achieved.

In this chapter, the relative importance indexes of the 47 construction management practices were computed and the practice well-defined scope of works has been ranked first. All the safety and health practices are ranked in the top ten. The weights of the practices were used to develop the scoring tools. The validity and reliability tests for the scoring tools of the construction management practices were carried out, and acceptable results were found. Furthermore, the validity and reliability of the scoring tools for the six categories of the construction management practices were checked, and the tools were found to be valid and reliable. This implies that the six scoring tools can be separately used to measure, plan and control the practices. For instance, a project's materials management practices score can be computed in isolation, and its relationship with productivity can be investigated. The baseline score to assess a project's construction management practices is found to be 22.75. Thus, a project with overall construction management practices score exceeding 22.75 can be considered as a high-score and the associated productivity could also be high. When projects are assessed 198

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

Quantitative Data Analysis and Findings

based on the scores of the six categories of the management practices, projects with scores exceeding: 2.4 for construction materials management practices, 0.93 for construction equipment management practices, 4.84 for management practices related to construction methods, 4.52 for preconstruction phase management practice, 4.39 for human resource management practices, and 5.23 for safety and health practices can be considered as high score projects and their productivity would also be high.

Correlation analyses between 47 construction management practices and productivity were conducted, and the findings revealed that all the practices are positively associated with productivity in multi-storey building projects. The correlation analyses among the construction management practices show that the practices are positively correlated. The relationships among company profiles, project characteristics and the construction management practices were also investigated, and the findings indicate that project cost is positively and strongly associated with the practices. Project delay and construction management practices are negatively correlated. The associations among the aggregated construction management practices (47 practices) and company size, annual turnover and company experience are statistically insignificant. However, company size is positively associated with 'construction equipment management practices,' 'human resource management practices,' and 'safety and health practices', and the relationships are statistically significant. Company experience and the safety and health practices are positively correlated, and the association is statistically significant. In next chapter, the productivity prediction models based on the levels of implementations of the management practices are developed and validated.

CHAPTER EIGHT

8 REGRESSION MODELS OF CONSTRUCTION MANAGEMENT PRACTICES AND PRODUCTIVITY

8.1 Introduction

This chapter presents the analysis conducted to achieve the third objective of the thesis which is 'to build and validate a tool that can be used to predict the probability of exceeding baseline productivity factor based on a score of the management practices.' In Section 8.2, the logistic regression models which relate the overall or the 47 construction management practices and productivity are developed. Moreover, the logistic regression models relating each of the six categories of the construction management practices and productivity are built. The linear regression models of productivity and the construction management practices are also presented in Section 8.2. In Section 8.3, the logistic and linear regression models are validated. Finally, the summary of the chapter is provided in Section 8.4.

8.2 Building the Regression Models

Two approaches have been used to develop the regression models of the construction management practices and productivity. In the first approach, logistic regression analyses are conducted to build logistic regression models which are used to predict the probability of exceeding the baseline productivity factor. In this approach, based on a project's score of the construction management practices, the productivity factor of the particular project is estimated with respect to the baseline productivity factor. In the second approach, linear regression analysis is carried out to develop a linear regression model for predicting the productivity factor of a project based on the project's score of the construction management practices.

8.2.1 Logistic Regression Models of the Construction Management Practices and Productivity

Before conducting the logistic regression analysis, the 39 data points are divided into two: model building datasets and validation dataset. Three-fold and 4-fold crossvalidation techniques were used as they are less biased than sample splitting method (Abou-Assaleh et al., 2004). To determine the baseline productivity factor (PF) or the

cut-off PF which is used to transform projects' productivity factor into binary equivalent, the 3 measures of central tendency (mean, mode and median) as well as the skewness of the PF data were computed using SPSS 24. Accordingly, the mean PF = 0.95, mode PF = 1.00, median PF = 1.00, skewness = -0.893, standard error = 0.378, and Z = |(-0.893/0.378)| = 2.36 were found. The mean value is the best measure of central tendency if the data is not highly skewed, and the median is the preferred value if the data is skewed (Lurd and Lurd, 2013). According to Kim (2013), if *Z*-score for skewness is greater than 1.96, the distribution of the sample is not considered as normal. Thus, the median value of 1.00 is used as a cut-off PF value since the *Z value* = 2.36 is greater than the allowable *Z value* of 1.96.

The rule of thumb for the sample size of the logistic regression analysis states that the number of Events Per Variable should be greater than 10 (Peduzzi et al., 1996). Some authors argue that the minimum of 10 events per predictor is conservative (Bowerman and Murphree, 2014). In this research, the number of positive events refers to the number of projects with productivity factors greater than the baseline PF, and there is one predictor (construction management practices). The Events Per Variables (EPV's) are computed using median PF = 1.00, median PF + (5% × median PF) =1.05, median PF - (15% × median PF) = 0.95, median PF - (10% × median PF) = 0.90, and median PF - (15% × median PF) = 0.85 as baseline productivity factors. Accordingly, using median PF as a baseline value, EPV=21>10; using (median PF) + (5% × median PF) as a cut-off value, EPV=12>10; using (median PF) - (5% × median PF) as a baseline, EPV=23>10; using (median PF) - (10% × median PF) as a cut-off value, EPV=26>10; using (median PF) as a baseline EPV=31>10 are obtained. Therefore, the data is suitable to run the logistic regression analysis.

As shown in Table 8-1, 15 alternative models using 3-fold cross validation and 20 alternative models using 4-fold cross validation technique were developed, and the best model of the 47 construction management practices and productivity is selected. The raw data used in the analyses are indicated in Appendix-5A7. In the 3-fold cross validation, among the 15 models, four models with better predictive accuracy and statistically significant coefficients are chosen. The first model has the predictive accuracy of 88.5% and *p*-value of 0.03, and it is coded A. The second model with predictive accuracy of 84.6% and *p*-value of 0.01 is coded B. The third model having

the predictive accuracy of 84.6% and *p-value* of 0.03 is coded C. The fourth model with predictive accuracy of 73.1% and *p-value* of 0.01 is coded D.

Table 8-1 Selection of the Logistic Regression Model of the Overall Construction Management Practices and Productivity

		Alte	rnative Models Using	3-fold Cross Valid	lation	
			i i	Predictive Accurac	y and Significance of the M	Nodels
			Trial-1		Trial-2	Trial-3
			Validation Dataset ¹¹	Model Bu	uilding Dataset 12	Model Building Dataset 13
	Baseline Producti	ivity Factors (PFs)	Model Building Datase	et ²¹ Validation	n Dataset 22	Model Building Dataset 23
			Model Building Datase	et ³¹ Model Bu	uilding Dataset 32	Validation Dataset 33
	Cut-off PF1=0.85	Predictive Accuracy (%)	84.63 (B)		80.8	76.9
		Significance of B	0.03		0.25	0.36
	Cut-off PF ₂ =0.90	Predictive Accuracy (%)	84.62 (C)		69.2	61.5
	-	Significance of B	0.01		0.48	0.35
	Cut-off PF3=0.95	Predictive Accuracy	73.14 (D)		65.4	69.2
	2	Significance of β	0.01		0.22	0.19
	Cut-off PF ₄ =1.00	Predictive Accuracy (%)	69.2		65.4	65.4
		Significance of β	0.01		0.07	0.05
	Cut-off PE=1 05	Predictive Accuracy	88.51 (A)		76.9	73.1
		Significance of ß	0.03		0.15	0.06
		Alte	rnative Models Using	4-fold Cross Valid	lation	
			Predictive 2	Accuracy and Signi	ficance of the Models	
		Trial-4	!	Trial-5	Trial-6	Trial-7
		Validation Dataset14	Model	Building Dataset15	Model Building Dataset ¹⁶	Model Building Dataset ¹⁷
Baseline I	Productivity Factors (PFs)	Model Building Datas Model Building Datas	et ²⁴ Valida det ³⁴ Model	tion Dataset ²⁵ Building Dataset ³⁵	Model Building Dataset ²⁶ Validation Dataset ³⁰	Model Building Dataset ²⁷ Model Building Dataset ³⁷
		Model Building Datas	et ⁴⁴ Model	Building Dataset45	Model Building Dataset46	Validation Dataset47
Cut-off PF1=0.85	Predictive Accuracy (%)	82.85 (E)	8	9.7	75.9	80
111 0.05	Significance of β	0.04	0	.12	0.14	0.4
Cut-off	Predictive Accuracy (%)	75.97 (G)	7	9.3	69	70
FF2-0.90	Significance of β	0.01	0	.21	0.12	0.4
Cut-off	Predictive Accuracy (%)	69	7	2.4	65.5	66.7
PF3-0.93	Significance of β	0.02	0	.12	0.06	0.21
Cut-off	Predictive Accuracy (%)	65.5		69	65.5	66.7
rr4-1.00	Significance of β	0.02	0	.04	0.02	0.06
Cut-off	Predictive Accuracy (%)	79.36 (F)	6	5.5	82.8	66.7
PF5=1.05	Significance of β	0.04	0	.06	0.05	0.06
	C 1D 1 C	1 11 2 0	11 111 11 11	21 21 12 22 22	12 22 122 1 6	1 1 1 12

Note: A, B, C and D are codes for good models in 3-fold cross validation; 11, 21, 31, 12, 22, 32, 13, 23, and 33 are codes for the datasets each with 13 elements in 3-fold cross validation; E, F, and G are codes for good models in 4-fold cross-validation; 14, 24, 34, 44, 15, 25, 35, 45, 16, 26, 36, 46, 17, 27, 37, 47 are codes for the groups of data each with 10 data points in 4-fold cross validation.

Comparison of the Selected Models												
Model's code	Overall accuracy	Model AUC	Sig. of Model's AUC	Coefficient (β)	Sig. of β	Constant (β_o)	Sig of β_o	Cut-off PF	Validation AUC	Model Building dataset codes		
А	88.51%	0.95	< 0.01	0.28	0.03	-7.24	0.02	1.05	0.54	21+31		
С	84.62%	0.8	0.02	0.41	0.01	-7.65	0.02	0.90	0.73	21+31		
В	84.63%	0.88	0.03	0.34	0.03	-4.87	0.08	0.85	0.5	21+31		
E	82.85%	0.87	0.04	0.23	0.04	-3.49	0.13	0.85	0.56	24+34+44		
F	79.36%	0.86	0.01	0.26	0.04	-7.01	0.03	1.05	0.64	24+34+44		
G	75.97%	0.75	0.04	0.28	0.01	-5.52	0.03	0.90	0.81	24+34+44		
D	73.14%	0.78	0.02	0.30	0.01	-6.02	0.02	1.00	0.73	21+31		

In the 4-fold cross validation, among 20 alternative models, 3 models with better predictive accuracy and statistically significant coefficients are selected. The model having a predictive accuracy of 82.8% and *p-value* of 0.04 is coded E; the model with the predictive accuracy of 79.3% and *p-value* of 0.04 is coded F; and the model having a predictive accuracy of 75.9% and *p-value* of 0.01 is coded G. The seven models (A, B,

C, D, E, F, and G) are then compared among themselves to choose the best logistic regression model (Table 8-1). The criteria for selection include predictive accuracy, Area Under the Curve (AUC) and significance of the coefficients.

Model-A has the highest predictive accuracy (88.5%) but its validation dataset's AUC (0.54) is close to the rejection value (0.50), and it is not chosen as the best model (Table 8-1). Model-B also has a low AUC (0.50) of the validation dataset, and it cannot be the best model. Model-E has insignificant constant (p=0.13>0.05). Model-F (79.36%), Model-G (75.97%), and Model-D (73.14%) have low predictive accuracy as compared to Model-C (Table 8-1). Thus, Model-C is selected as the best model since it has a better predictive accuracy (84.6%); its variable's coefficient is significant (p=0.01<0.05); the model's constant is significant (p=0.02<0.05); and its AUC is acceptable (0.80>0.5). Therefore, the data sets coded 21 and 31 were used in model building; dataset coded 11 was used as a validation dataset (refer Appendix-5A7 for the raw data); and PF=0.90 was used as a baseline PF. The results of the logistic regression analysis of the best model (Model-C) are presented in Table 8-2.

As indicated in Table 8-2, in Block-0 of the logistic regression analysis, the true negative percentage is 0.0%, the true positive percentage is 100.0%, and overall model accuracy is 69.2%. These imply that all projects with productivity factor (PF) greater than 0.90 are correctly predicted, and no project with PF less than 0.90 is correctly predicted. The variable in the equation is insignificant (p=0.06>0.05) and the variable not in the equation is significant (p<0.001) which implies that the model in Block-1 is better than the model in Block-0. In Block-1, the Chi-square (12.53) of the Omnibus test is statistically significant at 5% level of significance (p<0.001) which shows that the model in Block-1 is better than the null model (model in Block-0). The significance value of Chi-square (5.21) of Hosmer and Lemeshow test of goodness is greater than the acceptable *p*-value (p=0.63>0.05) and the null hypothesis of the test which states 'the construction management practices and productivity model is good in fitting the data' is accepted. Therefore, the model in Block-1 is good.

Table 8-2 Outputs of the Logistic Regression Analysis of the Overall Construction Management Practices

	managemen	it i iuci	lees						
			Block	-0					
		Clas	sificatio	n Table					
			I	Predicted					
			P	°F		Percentage (Correct		
	Observed		0	1					
a a	PF	0	0	8		0			
Step 0		1	0	18		100			
	Overall Percentage					69.2			
The cut	t value = 0.50								
		Variab	les in the	e Equatio	п				
Sten ()		В	S . E	Wald	df	Sig	Exp(B)		
Step 0	Constant	0.811	0.425	3.642	1	0.056	2.25		
		Variable.	s not in t	he Equai	ion				
				Score	df	Sig.			
Step 0	Con Mgt Practices			10.55	1	0.001			
	Overall Statistics			10.55	1	0.001			
				Block-	1				
			Omnibu	s Test for	Model	Coefficients			
		Chi-s	quare			df			Sig.
	Step	12.	525			1			0
Step 1	Block	12.	525			1			0
	Model	12.	525			1			0
			Hos	mer and	Lemesh	ow Test			
Step 1	Chi-square		C	lf -			Sig	•	
	5.21			/	·· 77	11	0.63	4	
				Classifica	ition 1a	Die	diatad		
				D	Ē	Pie	uicteu		
	Observed			0	ľ	1	Percer	ntage Correct	
	00301700	0		6		2		75	
Step 1	PF	1		2		16		88.9	
	Overa	ll Percenta	ge					84.6	
The cu	it value = 0.50		0						
			Va	riables in	the Eq	uation			
		В	S	. E	Wald		df	Sig.	Exp(B)
Stop 1	Con Mgt. Practices	0.407	0.1	158	6.601		1	0.01	1.502
Sup I	Constant	-7.654	3.2	218	5.657		1	0.017	0
		В	ootstraj	oping					
					Boot	tstrap			
		R	Riac	SE	Sia	95% Conf	idence Interval		
			Dias	SE	ыg.	Lower	Upper		
Stop 1	Con Mgt. Practices	0.407	1.535	8.302	0.005	005 0.225 21.136			
Step 1	Constant	-7 654	-26 78	140.98	0.007	-382 19	-3 737		

The overall prediction accuracy of the model in Block-1 is 84.6% which is better than the null model's predictive accuracy (69.2%).The true negative prediction accuracy of the model is 75.0%, and the true positive prediction accuracy is 88.9% (Table 8-2). Wald Chi-square (6.60) for the variable in the equation is statistically significant at 5% level of significance (p=0.01<0.05). Similarly, the Wald Chi-square (5.66) for the constant is statistically significant (p=0.02<0.05).The Exp(B) for the variable 'construction management practices' is 1.5 which indicates that for one unit increase in 204

the score of the construction management practices (the variable), the odd of exceeding the productivity factor of 0.90 increases by a factor of 1.5.

To test the reliability of the logistic regression model of the overall or the 47 construction management practices and productivity, bootstrapping using 1000 samples are conducted using SPSS-24, and the results are shown in Table 8-2. Accordingly, the coefficient (0.41) and constant (-7.65) in the bootstrapping are found to be similar to the model's coefficient and constant. The *p*-values of both the variable (*p*=0.005) and constant (*p*=0.007) in bootstrapping are statistically significant. Therefore, the construction management practices and productivity model is reliable.

By using the coefficients indicated in Table 8-2, the model's equation is constructed as follows.

$$Log(Odds)=Logit(P_i)=Ln\left(\frac{P_i}{1-P_i}\right)=0.41CMP_i-7.65$$

Where Pi=probability of exceeding PF=0.90 of an i^{th} project, and CMP_i = the overall score of the construction management practices for an i^{th} project. To compute the probabilities easily, the above equation is simplified as indicated in Equation 8-1.

$$P_i = \frac{e^{0.41CMP_i - 7.65}}{1 + e^{0.41CMP_i - 7.65}}$$
 Equation 8-1

The results of the logistic regression analysis of the six categories of the construction management practices are explained in the following subsections.

Logistic Regression Model of Construction Materials Management Practices and Productivity

Fifteen alternative logistic regression models of the six construction materials management practices and productivity are developed using 3-fold cross-validation technique, and the results are presented in Table 8-3. Accordingly, the model that has the highest predictive accuracy of 88.5% and statistically significant coefficient of p=0.03<0.05 is chosen and coded H (Table 8-3). Among the 20 alternative models developed using 4-fold cross-validation technique, two models with the highest predictive accuracy are selected. The first model (coded I) has the predictive accuracy of 89.7% and a *p*-value of 0.03 < 0.05; and the second model (coded J) has the predictive accuracy of 82.8%, and the coefficient of the independent variable (β) is statistically significant at 5% level of significance (p=0.02 < 0.05).

Table	8-3	Selection	of	the	Logistic	Regression	Model	of	the	Construction	Materials	Management
		Practices	and	l Pro	ductivity							

Alternative Models Using 3-fold Cross Validation													
			0	Predictive A	Accuracy	and Significance of	he Models						
			Tria	l-1		Trial-2	Trial-	3					
			Validation Data	aset ¹¹	Model B	Building Dataset 12	Model Building	g Dataset 13					
	Baseline Producti	vity Factors (PFs)	Model Building	g Dataset ²¹	Validatio	on Dataset ²²	Model Building	g Dataset ²³					
		,	Model Building	g Dataset ³¹	Model B	Building Dataset 32	Validation Data	iset 33					
Cut	off DE -0.95	Predictive Accuracy (%)	80.8			80.8	76.9						
Cut-	011 FF1=0.85	Significance of B	0.19			0.34	0.24						
C (CDE 0.00	Predictive Accuracy (%)	73.1			65.4	65.4						
Cut-	$0 \Pi PF_2 = 0.90$	Significance of β	0.04			0.57	0.46						
a .	2005 0.05	Predictive Accuracy (%)	57.7			69.2	57.7						
Cut-	off PF ₃ =0.95	Significance of β	0.07			0.35	0.21						
Cut	off DE -1 00	Predictive Accuracy (%)	57.7			61.5	61.5						
Cut-	011 PF ₄ -1.00	Significance of β	0.09			0.16	0.08						
C (CDF 1.05	Predictive Accuracy (%)	88.5 (H)			80.8	73.1						
Cut-	оп PF ₅ =1.05	Significance of β	0.03			0.16	0.04						
		Alternative	Models Using 4	-fold Cross V	Validatio	n							
			Predictive	Accuracy and	d Signific	ance of the Models							
		Trial-4		Trial	-5	Trial-6	Trial-	7					
		Vali	dation Dataset ¹⁴	Model Build	ing	Model Building	Model Building	g Dataset ¹⁷					
Baseline Pro	oductivity Factors	Model Bu	uilding Dataset ²⁴	Validation D	ataset25	Model Building	Model Building	g Dataset ²⁷					
	(PFs)	Model Bu	uilding Dataset ³⁴	Model Build	ing	Validation Dataset ³⁶	Model Building	g Dataset ³⁷					
		Model Bu	uilding Dataset44	Model Build	ing	Model Building	Validation Data	iset ⁴⁷					
Cut-off Pred	lictive Accuracy (%)		79.3	86.2		72.4	80						
PF1=0.85 Sign	ificance of β		0.08	0.58		0.24	0.24						
Cut-off Pred	lictive Accuracy (%)		72.4	75.9		55.2	70						
PF2=0.90 Sign	ificance of β		0.03	0.62		0.27	0.48						
Cut-off Pred	lictive Accuracy (%)		62.1	65.5		65.5	63.3						
PF3=0.95 Sign	ificance of β		0.04	0.33		0.12	0.29						
Cut-off Pred	lictive Accuracy (%)		58.6	55.2		65.5	60						
PF4=1.00 Sign	ificance of β		0.06	0.15		0.06	0.12						
Cut-off Pred	Cut-off Predictive Accuracy (%) 82.83 (J) 69 89.71 (I) 66.7												
PF5=1.05 Significance of β 0.02 0.09 0.03 0.06													
Note: H is code	Note: H is code for a good model in 3-fold cross validation; 11, 21, 31, 12, 22, 32, 13, 23, and 33 are codes for the datasets each with 13 elements in												
3-fold cross val	3-told cross validation; I, and J are codes for good models in 4-fold cross-validation; 14, 24, 34, 44, 15, 25, 35, 45, 16, 26, 36, 46, 17, 27, 37, 47 are												
codes for the g	roups of data each wi	th 10 data points in 4-fold	d cross validation	n.									

Comparison of the Selected Models												
Model's code	Overall accuracy	Model AUC	Sig. of Model's AUC	Coefficient (β)	Sig. of β	Constant (β_o)	Sig of β_o	Cut-off PF	Model Building dataset codes			
Ι	89.70%	0.84	0.03	1.81	0.03	-6.46	0.01	1.05	16+26+46			
Н	88.50%	0.91	< 0.01	1.58	0.03	-4.56	0.01	1.05	21+31			
J	82.80%	0.95	0.04	1.52	0.02	-4.62	0.01	1.05	24+34+44			

The 3 selected models (H, I and J) are compared based on the statistical significance of their coefficients, their AUCs, and their predictive accuracy (Table 8-3). All the 3 models have AUC > 0.5; their AUCs are statistically significant; the variables' coefficients are statistically significant at 5% level of significance; and the constants of the 3 models are also statistically significant. However, the model coded I is selected as it has the best predictive accuracy (89.7%). The data sets coded 16, 26 and 46 are used in model building, the data set coded 46 is used to validate the model, and PF=1.05 is used as a cut-off productivity factor (refer Appendix-5A1 for the raw data).

The results of the logistic regression analysis of the selected model (Model-I) are shown in Table 8-4. In Block-0, the true negative percentage of 100.0%, the true positive percentage of 0.0%, and overall model accuracy of 82.8% are found. These imply that

all projects with productivity factors (PFs) less than 1.05 are correctly predicted, and no project with PF greater than 1.05 is correctly predicted. The variable in the equation (-1.57) is significant (p=0.001<0.05) and the variable not in the equation (7.03) is also significant (p=0.01<0.05).

Table	8-4	Outputs	of	the	Logistic	Regression	Analysis	of	the	Construction	Materials
		Managem	ent	Prac	tices						

		Block-0							
	Class	sification	Table						
			I	Predicted					
			P	PF					
	Observed		0	1	Perce	entage C	orrect		
~ ^	PF	0	24	0		100			
Step 0		1	5	0		0			
TTI .	Overall Percentage					82.8			
The cut	value = 0.50		7						
	variable	es in ine E R	s F	Wald	df	Sig	Evn(B)		
Step 0	Constant	_1 569	0.492	10.182	1	0.001	0.208		
	Variables	not in the	e Eauatio	10.102 m	1	0.001	0.200		
	,		<u> </u>	Score	df	S	ig.		
Step 0	Con. Mate. Management Practices			7.032	1	0.	008		
	Overall Statistics			7.032	1	0.	008		
			Block-1						
		Omnibus	Test for N	Iodel Co	efficients				
		Chi-s	quare			df			Sig.
a. 1	Step	7.	342			1			0.007
Step I			342			1			0.007
	Model	/	542 	1	Tart	I			0.007
	Chi-square	1105m		ff	Test			Sig	
Step 1	7.125			7				0.416	
		C	lassificat	ion Table	2				
						Pred	icted		
				Р	F		-Percenta	ore Correct	
	Observed		0		1		1 ereenta		
Step 1	PF	0	2	23		1		95.8	
-		1		2	-)		60 80.7	
The cu	Overall Percentage							89.7	
The eu		Vario	ables in t	he Equat	ion				
		В	S.	. E	Wald	(₫f	Sig.	Exp(B)
Stop 1	Con. Mat. Management Practices	1.808	0.8	806	5.037		1	0.025	6.099
Sup I	Constant	-6.461	2.4	484	6.764		1	0.009	0.002
	Bo	ootstrapp	ing						
			Boot	strap					
		В	Bias	SE	Sig.				
Step 1	Con. Mat. Management Practices	1.808	25.397	106.28	0.021				
Step 1	Constant	-6.461	-91.58	390.47	0.016				

In Block-1, the Chi-square (7.34) of the Omnibus test is statistically significant at 5% level of significance (p=0.01<0.05) which shows that the model is better than the null model (Block-0). The Hosmer and Lemeshow test shows p=0.42>0.05 which implies 207

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

the model in Block-1 is good (Table 8-4). The overall prediction accuracy of the model is 89.7%, and the value is greater than the null model's prediction accuracy (82.8%). The true negative prediction accuracy of the model is 95.8%, and the true positive prediction accuracy is 60.0%. Wald Chi-square (5.04) for the variable in the equation is statistically significant at 5% level of significance (p=0.03<0.05). Similarly, the Wald Chi-square (6.76) for the constant is statistically significant (p=0.01<0.05). The Exp(B) for the variable (construction materials management practices) is 6.10 which indicates that for one unit increase in the score of the construction materials management practices, the odd of exceeding the productivity factor of 1.05 increases by a factor of 6.10. In Table 8-4, the results of bootstrapping of the construction materials management practices data by using 1000 samples are presented. Accordingly, the pvalues of both the variable (p=0.021<0.05) and the constant (p=0.016<0.05) are found to be statistically significant. The coefficient and the constant obtained after bootstrapping is similar to the model's coefficient and constant. Therefore, the logistic regression model of the construction materials management practices and productivity is reliable.

Finally, by using the variable's coefficient and the constant indicated in Table 8-4, the equation for the construction materials management practices and productivity model is developed as shown in Equation 8-2.

$$Log(Odds) = Logit(P_i) = Ln\left(\frac{P_i}{1 \cdot P_i}\right) = 1.81MMP_i - 6.46$$

In the equation, P_i is the probability of exceeding PF=1.05 for an ith project, and MMP_i is the score of the construction materials management practices for an ith project. The above equation can be simplified as follows:

$$P_i = \frac{e^{1.81MMP_i - 6.46}}{1 + e^{1.81MMP_i - 6.46}}$$
 Equation 8-2

Logistic Regression Model of Construction Equipment Management Practices and Productivity

The results of the 35 alternative logistic regression models of the construction equipment management practices and productivity are presented in Table 8-5. Accordingly, the model developed using model building datasets coded 21 and 31 and PF=0.90 as a baseline productivity factor is chosen among the 15 alternative models built using 3-fold cross-validation technique (refer Appendix-5 A3 for the datasets). The

predictive accuracy of the model is 76.9%, and coefficient of the variable (β) is statistically significant (*p*=0.04<0.05). Among the 20 models developed using 4-fold cross-validation technique, only two construction equipment management practices and productivity models are statistically significant; however, they have lower prediction accuracy (62.1% and 60%) as compared to prediction accuracy of the model chosen in 3-fold cross-validation. Therefore, the model with a predictive accuracy of 76.91% and a p-value of 0.04 is chosen as the best model, and the results of the logistic regression analysis are presented in Table 8-6.

 Table 8-5 Selection of the Logistic Regression Model of the Construction Equipment

 Management Practices and Productivity

	Alternative Models Using 3-fold Cross Validation										
			Predictive Accuracy an	d Significance of the Models							
			Trial-1	Trial-2	Trial-3						
			Validation Dataset ¹¹	Model Building Dataset 12	Model Building Dataset 13						
			Model Building Dataset ²¹	Validation Dataset ²²	Model Building Dataset ²³						
Baseline P	Productivity Factors (PFs)		Model Building Dataset ³¹	Model Building Dataset 32	Validation Dataset ³³						
Cut-off	Predictive Accuracy (%)		80.8	80.8	76.9						
PF1=0.85	Significance of β		0.12	0.46	0.38						
Cut-off	Predictive Accuracy (%)		76.91	65.4	65.4						
PF2=0.90	Significance of β		0.04	0.79	0.51						
Cut-off	Predictive Accuracy (%)		73.1	61.5	57.7						
PF ₃ =0.95	Significance of β		0.13	0.57	0.55						
Cut-off	Predictive Accuracy (%)		61.5	65.4	61.5						
PF4=1.00	Significance of β		0.09	0.26	0.41						
Cut-off	Predictive Accuracy (%)		61.5	76.9	65.4						
PF5=1.05	Significance of β		0.03	0.19	0.04						
		Alternative Mo	dels Using 4-fold Cross	Validation							
		Predictive	Accuracy and Significant	ce of the Models							
		Trial-4	Trial-5	Trial-6	Trial-7						
		Validation Dataset14	Model Building Dataset ¹⁵	Model Building Dataset ¹⁶	Model Building Dataset ¹⁷						
		Model Building Dataset ²⁴	Validation Dataset ²⁵	Model Building Dataset ²⁶	Model Building Dataset ²⁷						
		Model Building Dataset ³⁴	Model Building Dataset ³⁵	Validation Dataset ³⁶	Model Building Dataset ³⁷						
Baseline P	Productivity Factors (PFs)	Model Building Dataset44	Model Building Dataset ⁴⁵	Model Building Dataset ⁴⁶	Validation Dataset47						
Cut-off	Predictive Accuracy (%)	79.3	86.2	72.4	80						
PF1=0.85	Significance of B	0.12	0.3	0.38	0.45						
Cut-off	Predictive Accuracy (%)	75.9	75.9	65.5	70						
PF2=0.90	Significance of β	0.08	0.23	0.5	0.63						
Cut-off	Predictive Accuracy (%)	69	69	62.1	60						
PF3=0.95	Significance of β	0.21	0.27	0.37	0.7						
Cut-off	Predictive Accuracy (%)	58.6	69	65.5	56.7						
PF4=1.00	Significance of β	0.14	0.1	0.21	0.56						
Cut-off	Predictive Accuracy (%)	62.1	58.6	82.8	60						
PF5=1.05	Significance of B	0.04	0.05	0.07	0.04						
Note: 11-2	1 31 12 22 32 13 23 at	nd 33 are codes for the data	asets each with 13 element	s in 3-fold cross validation: 14	24 34 44 15 25 35 45						

16, 26, 36, 46, 17, 27, 37, 47 are codes for the groups of data each with 10 data points in 4-fold cross validation.

As shown in Table 8-6, the results of Block-0 (model without independent variable) indicate the true negative percentage of 0%, the true positive proportion of 100% and overall accuracy is 69.20%. The true negative is the number of projects with observed PF less than the cut-off PF = 0.90 and predicted as 0 (less than cut-off PF), and true positive implies projects with PF greater than or equal cut-off PF and predicted as 1. No

project is predicted as 0, but there are eight projects which should have been predicted as 0, and the percentage is 0/8=0% (Table 8-6). However, all projects with PF value of 1 are correctly predicted ((18/18) × 100% = 100%). The overall percentage is computed as (18/26) ×100% = 69.2%.

					Block	x-0					
Management Practices											
Table	8-6	Outputs	of	the	Logistic	Regression	Analysis	of	the	Construction	Equipment

	Blo	ck-0						_		
	Classifica	tion Ta	ıble					_		
]	Predic	ted		_		
			I	PF	Dorog	ntaga C	orraat			
	Observed		0	1	Perce	ntage C	onect	_		
	DE	0	0	8		0		-		
Step 0	PF	1	0	18		100				
	Overall Percentage					69.2				
Cut va	hue = 0.50									
	Variables in	the Ea	uatior	1				-		
		B	S.E	Wald	df	Sig	Exp(B)	-		
Step 0	Constant	0.81	0.43	3 64	1	0.06	2 25	-		
	Variables not	in the F	avati	0n	1	0.00	2.23	_		
	v unubles not	in inc L	iquuti	Score	df	S	iσ	-		
Step ()	Con Eat Management Practices			4.82	1	0	<u>15.</u> 03	-		
Step 0	Overall Statistics			4.02	1	0	03			
	Overall Statistics	D	aalt 1	4.02	1	0.	.03	-		
	Orma	ihua Ta	OCK-1	Madal	Conffe	ionta				
	Omn	Chi a	si jor .	Model	_oejjic	af			Sia	
	Store .	Cni-s	quare					Sig.		
G/ 1	Step	٦.، -	44			1			0.02	
Step 1	Block	5.4	44			I			0.02	
	Model	5.4	44			1			0.02	
	H	losmer	and L	emesho	w Tes	t				
Step 1	Chi-square		(df				Sig.		
Step 1	3.7			6				0.72		
		Clas	ssifica	tion Tal	ble					
						Р	redicted			
				Р	PF		- 1	Percentage Co	orrect	
	Observed			0		1		ere en age e		
Step 1	ÞF	0		3		5		37.5		
Step 1		1		1		17		94.4		
	Overall Percentage							76.9		
Cut va	lue = 0.50									
		Variab	les in	the Equ	iation					
		В	S	. E	Wald	(lf	Sig.	Exp(B)	
Stop 1	Con. Eqt. Management Practices	2.29	1	.15	3.98		1	0.04	9.9	
Step 1	Constant	-0.98	0	.93	1.12		1	0.03	0.38	
	Bootst	rappin	g							
				1	Bootstr	rap		-		
						9:	5%	-		
		В	Bias	S.E	Sig.	Lower	Upper	-		
~	Con. Eqt. Management Practices	2.29	3.83	42.4	0.03	0.26	13.62	-		
Step 1	Constant	-0.98	-2.1	23.91	0.03	-7.14	-1.08			
		2.75								

In Block-1 or the analysis which considers an independent variable, the Chi-square (5.44) of the Omnibus test is statistically significant at 5% level of significance

210

(p=0.02<0.05) indicating that the model in Block-1 is better than the null model (model in Block-0). The *p*-value (0.72) of the Chi-square (3.7) in Hosmer and Lemeshow test of goodness is greater than 0.05 which indicates that the construction equipment management practices and productivity model is acceptable.

The overall prediction accuracy of the model is 76.9%, and it is better than the null model's predictive accuracy. The true negative prediction accuracy of the model is 37.5%, and the true positive prediction accuracy is 94.4%. Wald Chi-square (3.98) for the variable in the equation is statistically significant at 5% level of significance (p=0.04<0.05). The Exp(B) for the variable construction equipment practices is 9.9 which indicates that for a one-unit increase in the score of the construction equipment management practices, the odd of exceeding the productivity factor of 0.90 increases by a factor of 9.9. The results of the reliability test using 1000 samples show the coefficient of 2.29, constant = -0.98, and the *p*-values of the coefficient and the constant = 0.03 < 0.05 (Table 8-6). Thus, the model of construction equipment management practices and productivity is reliable.

By using the variable's coefficient and the constant indicated in Table 8-6, the model's equation is constructed as follows:

$$Log(Odds) = Logit(P_i) = Ln\left(\frac{P_i}{1-P_i}\right) = 2.29CEMP_i-0.98$$

Where P_i =probability of exceeding PF=0.90 for an ith project, and CEMP_i = score of construction equipment management practice for an ith project. For the sake of computing the probabilities easily, the above equation is simplified and represented by Equation 8-3.

$$P_i = \frac{e^{2.29 \text{CEMP}_i \cdot 0.98}}{1 + e^{2.29 \text{CEMP}_i \cdot 0.98}}$$
 Equation 8-3

Logistic Regression Model of Management Practices Related to Construction Methods and Productivity

The results of the logistic regression analysis of productivity and the management practices related to construction methods are presented in Table 8-7. Accordingly, among the 15 models developed using the 3-fold cross-validation technique, 3 of them have higher predictive accuracy and statistically significant coefficients (p=0.01<0.05). The 3 models have a similar predictive accuracy of 80.8% and coded K, L, and M for

further analysis. Model-K is developed by using the model building dataset coded 21 and 31 (Appendix-5 A4), and PF=0.90 as a baseline productivity factor. Model-L is built using similar datasets as Model-K but using PF=0.95 as the cut-off value. Model-M is also developed using similar datasets as Model-L but using PF=1.00 as baseline PF. Using the 4-fold cross validation technique, 3 models are found to have the higher predictive capacity (86.2%, 82.8%, and 80.0%). However, as the coefficients of the models are not statistically significant (p=0.2>0.05, p=0.07>0.05, and p=0.53>0.05 respectively), the models are not selected as good models. The other 17 models in the 4-fold cross-validation technique (Table 8-7). Finally, Model-K is considered as the best model since the ROCs for Model-L and Model-M cannot be drawn, or the models are not suitable for prediction purpose.

 Table 8-7 Selection of the Logistic Regression Model of the Management Practices Related to Construction Methods and Productivity

	Alternative Models Using 3-fold Cross Validation										
			Predictive Accuracy and	Significance of the Model	š						
			Trial-1	Trial-2	Trial-3						
			Validation Dataset ¹¹	Model Building Dataset	Model Building Dataset 13						
			Model Building Dataset ²¹	Validation Dataset ²²	Model Building Dataset 23						
Baseline F	Productivity Factors (PFs)		Model Building Dataset ³¹	Model Building Dataset	Validation Dataset ³³						
Cut-off	Predictive Accuracy (%)		80.8	80.8	76.9						
PF1=0.85	Significance of β		0.037	0.346	0.518						
Cut-off	Predictive Accuracy (%)		80.80 (K)	65.4	65.4						
PF ₂ =0.90	Significance of β		0.014	0.505	0.348						
Cut-off	Predictive Accuracy (%)		80.80 (L)	69.2	69.2						
PF3=0.95	Significance of β		0.011	0.225	0.216						
Cut-off	Predictive Accuracy (%)		80.80(M)	76.9	73.1						
PF4=1.00	Significance of β		0.01	0.038	0.052						
Cut-off	Predictive Accuracy (%)		76.9	76.9	61.5						
PF5=1.05	Significance of β		0.067	0.215	0.208						
	•	Alternative Mode	ls Using 4-fold Cross Val	lidation							
		Predictive Ac	ccuracy and Significance of	of the Models							
		Trial-4	Trial-5	Trial-6	Trial-7						
		Validation Dataset ¹⁴	Model Building Dataset ¹⁵	Model Building Dataset ¹⁶	Model Building Dataset ¹⁷						
		Model Building Dataset ²⁴	Validation Dataset ²⁵	Model Building Dataset ²⁶	Model Building Dataset ²⁷						
		Model Building Dataset ³⁴	Model Building Dataset ³⁵	Validation Dataset ³⁶	Model Building Dataset ³⁷						
Baseline P	Productivity Factors (PFs)	Model Building Dataset ⁴⁴	Model Building Dataset ⁴⁵	Model Building Dataset ⁴⁶	Validation Dataset47						
Cut-off	Predictive Accuracy (%)	79.3	86.2	72.4	80						
PF1=0.85	Significance of β	0.06	0.2	0.16	0.53						
Cut-off	Predictive Accuracy (%)	72.4	75.9	65.5	70						
PF2=0.90	Significance of β	0.02	0.48	0.1	0.37						
Cut-off	Predictive Accuracy (%)	75.9	65.5	69	70						
PF3=0.95	Significance of β	0.01	0.23	0.07	0.15						
Cut-off	Predictive Accuracy (%)	75.9	75.9	75.9	73.3						
PF4=1.00	Significance of β	0.01	0.04	0.01	0.04						
Cut-off	Predictive Accuracy (%)	75.9	69	82.8	60						
PF5=1.05	Significance of β	0.12	0.18	0.07	0.23						

Note: K, L and M are codes of good models in 3-fold cross validation; 11, 21, 31, 12, 22, 32, 13, 23, and 33 are codes for the datasets each with 13 elements in 3-fold cross validation; 14, 24, 34, 44, 15, 25, 35, 45, 16, 26, 36, 46, 17, 27, 37, 47 are codes for the groups of data each with 10 data points in 4-fold cross validation.

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

212

The results of the logistic regression analysis of the selected model (Model-K) are shown in Table 8-8. In Block-0, the true negative percentage = 0.0%, the true positive percentage = 100.0%, and overall model accuracy = 69.2%. These indicate that all projects with productivity factors (PFs) greater than 0.90 are correctly predicted, and no project with PF less than 0.90 is correctly predicted. The variable in the equation (p=0.06>0.05) is insignificant, and the variable not in the equation is statistically significant (p=0.01<0.05). This shows that the model in Block 1 is better than the model in Block 0.

Table	8-8	Outputs	of	the	Logistic	Regression	Analysis	of	Management	Practices	Related	to
		Construc	ctio	n M	fethods							

		Block-0							_	
	Class	ification T	able						-	
				Predicte	d				_	
]	PF	Doro	ontago Cor	root			
	Observed		0	1	Perc	entage Col	lect		_	
	DF	0	0	8			0			
Sten ()	11	1	0	18			100			
Step 0	Overall Percentage					f	59.2			
The cu	t value = 0.50	· 1 E							-	
	Variable	s in the Ed	quation	Wald	15		Q:-	Euro(D)	-	
Step 0	Constant	B 0.911	5. E	2 642	di	1	51g	2 25	-	
	Variables	0.011	Equation Faultion	5.042		1	0.030	2.23	-	
	, unubics	noi in inc	Бушино	Score	df		Sig		-	
Step 0	Mgt. Practices Related to Con. Methods			9.093	1		0.003			
····F	Overall Statistics			9.093	1		0.003			
		E	Block-1						•	
	(Omnibus T	est for N	1odel Co	oefficie	nts				
		Chi-squ	are	df					Sig.	
	Step	10.34		1					0.001	
Step 1	Block	10.34		1					0.001	
	Model	10.34	1.1	1	<i>T</i>				0.001	
	Chi squara	HOSME.	r ana Le	emesnow	Test		Sig			
Step 1	3 626	7					51g. 0.822			
	5.020		assificati	ion Table	e		0.822			
					-		Predicte	d		
					PF	7		_ г) araantaga Carr	aat
	Observed			0		1		- r	reicentage Con	ect
Sten 1	bE	0		5		3			62.5	
Step 1		1		2		16			88.9	
-	Overall Percentage								80.8	
The cu	it value = 0.50	<i>U</i> ·	11 • 7		<i>,</i> •					
		Varia	bles in ti	ne Equai	tion	Wald		46	Sig	Evp(D)
	Mat Practices Related to Con Methods	1 513	0	616		6.033		1	0.014	4 542
Step 1	Constant	-5 791	2	639		4 816		1	0.014	0.003
	Bo	otstrappi	ng	057		1.010		1	0.020	0.005
	-	Bool	strap Ba	ased on I	1000 \$	Samples	_			
		В	Bias	SE		Sig.	_			
Step 1	Mgt. Practices Related to Con. Methods	1.513	4.514	47.685		0.002				
Sup I	Constant	-5.791	-17.98	185.33		0.002				

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

As indicated in Table 8-8, in Block-1, the Chi-square (10.34) of the Omnibus test is statistically significant (p=0.001<0.05), and the *p-value* of the Chi-square (3.63) in Hosmer and Lemeshow test of goodness is greater than 5% (p=0.82>0.05). Thus, the model in Block-1 is acceptable. The overall prediction accuracy (80.8%) of the model in Block-1 is greater than the prediction accuracy (69.2%) of the model in Block-0. The true negative prediction accuracy of the model is 62.5%, and the true positive prediction accuracy is 88.9%. Wald Chi-square (6.03) for the variable in the equation is statistically significant (p=0.01<0.05). Similarly, the Wald Chi-square (4.82) for the constant is statistically significant (p=0.03<0.05). The Exp(B) for the variable 'management practices related to construction methods, the odd of exceeding the productivity factor of 0.90 increases by a factor of 4.82.

The results of bootstrapping of the management practices related to construction methods' data by using 1000 samples are presented in Table 8-8. The p-values of both the variable and the constant are found to be statistically significant (p=0.002<0.05). Therefore, the logistic regression model of the management practices related to construction methods and productivity is reliable. Finally, by using the variable's coefficient and the constant, the model's equation is constructed as follows:

$$Log(Odds) = Logit(P_i) = Ln\left(\frac{P_i}{1-P_i}\right) = 1.51MPRCM_i-5.79$$

 P_i = probability of exceeding PF=0.90 for an ith project, and MPRCM_i = score of management practices related to construction methods for an ith project. The above equation can be simplified as:

$$P_i = \frac{e^{1.51MPRCM_i - 5.79}}{1 + e^{1.51MPRCM_i - 5.79}}$$
 Equation 8-4

Logistic Regression Model of Pre-Construction Phase Management Practices and Productivity

In Table 8-9, 35 alternative pre-construction phase management practices and productivity models are presented. Accordingly, using the 3-fold cross-validation analysis, two models with better overall prediction accuracy and statistically significant coefficients are chosen. The model with a predictive accuracy of 76.9% is coded N, and the model with a predictive accuracy of 73.1% is coded O for comparison with models developed using 4-fold cross-validation technique. Using the 4-fold cross validation

214

technique, four models with better predictive accuracy are selected. The first model has a predictive accuracy of 86.2% and coded P; the second model having the predictive accuracy of 75.9% is coded Q; the third model with the predictive accuracy of 72.4% and a *p*-value of 0.02 is coded R, and the fourth model having the predictive accuracy of 72.4% and a p-value of 0.04 is coded S (Table 8-9).

 Table 8-9 Selection of the Logistic Regression Model of the Pre-Construction Phase Management

 Practices and Productivity

	Alternative Models Using 3-fold Cross Validation									
			Pre	dictive Accu	racy and Significance of th	e Models				
			Trial-1		Trial-2	Trial-3				
			Validation Dataset ¹¹	Model Build	ding Dataset 12	Model Building Dataset 13				
	Baseline Product	tivity Factors (PFs)	Model Building Dataset ²¹	Validation I	Dataset 22	Model Building Dataset 23				
			Model Building Dataset ³¹	Model Build	ding Dataset 32	Validation Dataset 33				
	Cut off DE =0.95	Predictive Accuracy	80.8		80.8	76.9				
	Cut-on Pr ₁ -0.85	Significance of β	0.06		0.27	0.35				
	Cut off DE =0.00	Predictive Accuracy	76.9 (N)		69.2	65.4				
	Cut-on PT ₂ =0.90	Significance of β	0.02		0.46	0.17				
	C (CDE 0.05	Predictive Accuracy	69.2		61.5	69.2				
	Cut-off $PF_3=0.95$	Significance of β	0.04		0.36	0.14				
	C () (CDE 1.00	Predictive Accuracy	73.1 (O)		61.5	73.1				
	Cut-off $PF_4=1.00$	Significance of β	0.03		0.1	0.05				
	Cut - 60 DE -1.05	Predictive Accuracy	84.6		80.8	76.9				
	Cut-on PF ₅ -1.05	Significance of β	0.06		0.16	0.08				
		Altern	ative Models Using 4-fold	l Cross Vali	dation					
			Predictive Accu	racy and Sig	nificance of the Models					
		Trial-4	Tric	al-5	Trial-6	Trial-7				
		Validation Dataset14	Model Buildi	ng Dataset ¹⁵	Model Building Dataset16	Model Building Dataset ¹⁷				
Baseline	Productivity Factors	Model Building Dataset	Validation Da	ntaset25	Model Building Dataset ²⁶	Model Building Dataset ²⁷				
	(PFs)	Model Building Dataset	Model Buildi	ng Dataset ³⁵	Validation Dataset ³⁶	Model Building Dataset ³⁷				
		Model Building Dataset	¹⁴ Model Buildi	ng Dataset ⁴⁵	Model Building Dataset ⁴⁶	Validation Dataset47				
Cut-off	Predictive Accuracy	82.8	89	7	72.4	80				
PF1=0.85	Significance of B	0.12	0	11	0.14	0.45				
Cut-off	Predictive Accuracy	75.9 (Q)	79	.3	69	70				
PF2=0.90	Significance of B	0.03	0.2	21	0.07	0.25				
Cut-off	Predictive Accuracy	65.5	65	.5	65.5	56.7				
PF3=0.95	Significance of B	0.06	0.3	23	0.06	0.21				
Cut-off	Predictive Accuracy	72.4 (S)	75	.9	72.4 (R)	66.7				
PF4=1.00	Significance of B	0.04	0.0	07	0.02	0.09				
Cut-off	Predictive Accuracy	79.3	6	9	86.2 (P)	73.3				
PF5=1.05	Significance of B	0.09	0.	.1	0.03	0.13				
Note: N an	d O are codes for good	models in 3-fold cross val	idation; 11, 21, 31, 12, 22,	32, 13, 23, a	nd 33 are codes for the data	sets each with 13 elements in 3-				
C 1 1			1 6 1 11 46							

fold cross validation (Appendix-5 A4); Q, R, S and T are codes for good models in 4-fold cross validation ; 14, 24, 34, 44, 15, 25, 35, 45, 16, 26, 36, 46, 17, 27, 37, 47 are codes for the groups of data each with 10 data points in 4-fold cross validation (Appendix-5 A4).

Comparison of the Selected Models									
Model's code	Overall accuracy	Sig. of Model's AUC	Coefficient (β)	Sig. of β	Constant (β_o)	Sig of β_o	Cut-off PF	Model's dataset code	Validation dataset code
Р	86.20%	0.13	1.7	0.03	-10.3	0.02	1.05	16+26+46	36
Ν	76.90%	0.2	1.03	0.02	-3.3	0.06	0.9	21+31	11
Q	75.90%	0.09	0.75	0.03	-2.53	0.1	0.9	24+34+44	14
0	73.10%	0.05	0.86	0.03	-3.32	0.05	1	21+31	11
R	72.40%	0.02	1.1	0.02	-5.26	0.02	1	16+26+46	36
S	72.40%	0.08	0.71	0.04	-2.88	0.07	1	24+34+44	14

The six good models (N, O, P, Q, R, and S) which are identified using 3-fold and 4-fold cross validation techniques are compared among each other to choose the best model of the pre-construction phase management practices and productivity. The Receiver Operating Characteristic (ROC) curves for the six models are drawn, and the Area Under the Curves (AUCs) are computed (Table 8-9). Furthermore, the statistical significance of the models' constants and AUCs are considered in the comparisons.

As shown in Table 8-9, Model-P has a better predictive accuracy (86.2%), but its AUC is insignificant (p=0.13>0.05). Thus, it was not chosen as the best model. Model-N, Model-Q, Model-O, and Model-S have insignificant p-values for both AUC and β_0 -values. Hence, Model-R is chosen for further analysis as it has significant values for AUC, β , and β_0 (Table 8-9). Thus, the cut-off PF is 1.00, the datasets coded as 16, 26, and 46 are used in logistic regression analysis and dataset coded 36 is kept for validation purpose (refer Appendix-5 A4 for the raw data). The results of the selected model (Model-R) are provided in Table 8-10.

In Table 8-10, the results of the logistic regression analysis of productivity and preconstruction phase management practices in Block-0 are presented. Accordingly, the true negative percentage is 100%, the true positive percentage is 0.0%, and overall model accuracy is 55.2%. These imply that all projects with productivity factor less than 1.00 are correctly predicted, and no project with PF greater than 1.00 is correctly predicted. The variable in the equation is insignificant (p=0.58>0.05), and the variable not in the equation is significant (p=0.01<0.05) which indicates that the model in Block-1 is better than the model in Block-0.

In Block-1, the Chi-square (9.33) of the Omnibus test is statistically significant at 5% level of significance (p=0.002<0.05) which shows that the model is better than the null model (Block-0). The significance value of the Chi-square (9.78) of Hosmer and Lemeshow test of goodness is greater than rejection percentage of the null hypothesis (p=0.28>0.05), and the null hypothesis which states the model is good in fitting the data is accepted. Thus, the model in Block-1 is good.

The overall prediction accuracy of the model is 72.4% which is better than the null model's predictive accuracy (55.2%). The true negative prediction accuracy of the model is 75.0%, and the true positive prediction accuracy is 69.2%. Wald Chi-square (5.82) for the variable in the equation is statistically significant (p=0.02<0.05). Similarly, the Wald Chi-square (5.77) for the constant is statistically significant (p=0.02<0.05). The Exp(B) for the variable pre-construction phase management practices is 3.0 which indicates that for a one-unit increase in the score of the pre-construction phase management practices, the odd of exceeding the productivity factor of 1.00 increases by a factor of 3.0.

Table 8-10 Outputs of the Logistic Regression Analysis of the Pre-Construction Phase Management Practices

		Bloc	k-0						
	(Classificati	ion Tabl	е				-	
				Predicted	1			-	
			I	PF	-Percent	tage Corr	ect		
	Observed		0	1	1 010 011			-	
a . a	PF	0	16	0		1	100		
Step 0	Querall Percentage	1	13	0		5	0		
The cut	t value = 0.50					5	5.2		
The ea	Vari	ables in ti	he Eaua	tion				-	
~ ~		В	S.E	Wald	df	Sig	Exp(B)	-	
Step 0	Constant	-0.208	0.373	0.309	1	0.578	0.813	-	
	Varial	bles not in	the Eq1	uation				-	
				Score	df	Sig.			
Step 0	Pre-Con. Phase Managemen	nt		7.848	1	0.005			
•	Practices Overall Statistics			7 8/8	1	0.005			
	Overall Statistics		Bloc	/.040 k-1	1	0.005			
		Omnib	us Test	for Mode	l Coeffic	cients			
		Chi-squ	iare	df	55			Sig.	
	Step	9.336		1				0.002	
Step 1	Block	9.336		1				0.002	
	Model Chi square	9.336		1		Sig		0.002	
Step 1	9 78	8				0 281			
		÷	Classij	fication T	Table				
							Predicted		
				P	PF		Percentage Corre	et	
	Observed		0		1		Teleellage Colle		
Step 1	PF	0		12		4		75	
1	Overall Dereentees	1		4		9		69.2 72.4	
Cut val	he = 0.50							/2.4	
eur ru		V	ariables	in the E	quation				
		В	S. E		Wald	df		Sig.	Exp(B)
	Pre-Con. Phase Management							-	
Step 1	Practices	1.1	0.	456	5.817		1	0.016	3.005
	Constant	-5.255	2.	188	5.768		1	0.016	0.005
		Bootstra	apping					_	
					Boo	otstrap	C1 1 1	-	
		р	Diac	SE	C i a	95% C	Unner Interval	-	
	Pre-Con Phase Management	<u> </u>	2 002	5E 60 761	51g.	0 47	0pper 2 772	-	
Step 1	Constant	-5.255	-9.421	274.06	0.003	-12.51	-2.508		

Bootstrapping to rest the reliability of the pre-construction phase management practices and productivity model is conducted, and the results are presented in Table 8-10. Accordingly, the coefficient of the model (1.10) is found to be statistically significant (p=0.005<0.05). Similarly, the constant (-5.26) is also found to be statistically significant (p=0.003<0.05) which implies that the model is reliable. By using the coefficient and the constant, the model's equation is constructed as follows.

217

$$Log(Odds) = Logit(P_i) = Ln\left(\frac{P_i}{1-P_i}\right) = 1.10PCPMP_i-5.26$$

Where P_i = probability of exceeding PF=1.00 of an ith project, and PCPMP_i = a score of pre-construction phase management practices of an ith project. The above equation can be simplified as indicated in Equation 8-5.

$$P_i = \frac{e^{1.10PCPMP_i-5.26}}{1+e^{1.10PCPMP_i-5.26}}$$
 Equation 8-5

Logistic Regression Model of the Human Resource Management Practices and Productivity

The results of the logistic regression analysis of the human resource management practices and productivity using 3-fold and 4-fold cross-validation techniques are presented in Table 8-11. Accordingly, among 15 models in 3-fold cross-validation, five models are chosen based on their predictive accuracy and statistical significance. The model with highest predictive accuracy (code T) is obtained by analysing the datasets coded 21 and coded 31 and using cut-off PF value of 0.85. The second model (code U) is developed using the same datasets as Model-T but using PF=0.90 as the cut-off value. The third model (coded V), fourth model (coded W) and fifth model (coded X) are built using similar datasets as that of Model-T but using cut-off PF values of 0.95, 1.0 and 1.05 respectively.

The 20 models developed using 4-fold cross-validation are also shown in Table 8-11. Among them, the model which is statistically significant and having the better predictive capacity (72.4%) is chosen and coded Y. Finally, six models (coded T, U, V, W, X and Y) are compared to select the best HRM practices and productivity model. To choose the best model among the six models, the ROC is drawn, and the Area under Curve (AUC) for each model is computed using SPSS-24.

Model-T has the highest predictive capacity (84.6%) and AUC (0.96), but its constant is insignificant (p=0.18>0.05). Thus, Model-T cannot be the best model. Model-U has good predictive capacity (80.8%); its coefficient of the independent variable (β) is statistically significant (p=0.02<0.05); its AUC is greater than 0.5; and its p-values of both the variable and constant are statistically significant (Table 8-11). Model-V, Model-W, and Model-X have low prediction accuracy as compared to Model-U's prediction accuracy; as a result, they cannot be best models.

Table 8-11 Selection of the Logistic Regression Model of the Human Resource Management Practices and Productivity

		Altern	native Models Using 3-fold Cross Validation									
			Predictiv	e Accuracy and Significance of the N	Models							
			Trial-1	Trial-2	Trial-3							
			Validation Dataset ¹¹	Model Building Dataset 12	Model Building Dataset 13							
	Baseline Product	ivity Factors (PFs)	Model Building Dataset ²¹	Validation Dataset 22	Model Building Dataset 23							
			Model Building Dataset ³¹	Model Building Dataset 32	Validation Dataset 33							
	Cut off DE =0.95	Predictive Accuracy (%)	84.6 (T)	80.8	76.9							
	Cut-on rr ₁ =0.65	Significance of β	0.04	0.25	0.89							
	Cut - # DE -0.00	Predictive Accuracy (%)	80.8 (U)	65.4	65.4							
	Cut-off PF ₂ =0.90	Significance of β	0.02	0.55	0.87							
	Cut off DE =0.05	Predictive Accuracy (%)	76.9 (V)	61.5	53.8							
	Cut-on 113-0.95	Significance of β	0.02	0.23	0.67							
	Cut-off PE.=1.00	Predictive Accuracy (%)	73.1 (W)	61.5	61.5							
	Cut-01114 1.00	Significance of β	0.03	0.12	0.24							
	Cut-off PE=1.05	Predictive Accuracy (%)	73.1 (X)	76.9	76.9							
	Cut-01115 1.05	Significance of β	0.04	0.25	0.17							
		Altern	ative Models Using 4-fold Cross	Validation								
			Predictive Accuracy a	nd Significance of the Models								
		Trial-4	Trial-5	Trial-6	Trial-7							
		Validation Dataset14	Model Building	Dataset ¹⁵ Model Building Dataset ¹⁶	Model Building Dataset ¹⁷							
Baselin	e Productivity Factors	Model Building Dataset24	Validation Datas	et ²⁵ Model Building Dataset ²⁶	Model Building Dataset27							
Busenn	(PFs)	Model Building Dataset ³⁴	4 Model Building	Dataset ³⁵ Validation Dataset ³⁶	Model Building Dataset37							
	1 9	Model Building Dataset44	4 Model Building	Dataset ⁴⁵ Model Building Dataset ⁴⁶	Validation Dataset47							
Cut-off	Predictive Accuracy (%)	86.2	89.7	72.4	80							
PF1=0.85	Significance of B	0.07	0.09	0.31	0.86							
Cut-off	Predictive Accuracy (%)	72.4 (Y)	79.3	58.6	70							
PF2=0.90	Significance of β	0.03	0.2	0.35	0.83							
Cut-off	Predictive Accuracy (%)	69	69	69	56.7							
PF3=0.95	Significance of β	0.03	0.09	0.17	0.55							
Cut-off	Predictive Accuracy (%)	65.5	69	58.6	70							
PF4=1.00	Significance of β	0.04	0.04	0.08	0.18							
Cut-off	Predictive Accuracy (%)	75.9	58.6	82.8	60							
PF5=1.05	Significance of B	0.06	0.1	0.17	0.13							

Note: T, U, V, W and X are codes for good models developed using 3-fold cross validation; 11, 21, 31, 12, 22, 32, 13, 23, and 33 are codes for the datasets each with 13 elements in 3-fold cross validation (Appendix-5 A5); Y is code for good model in 4-fold cross-validation; 14, 24, 34, 44, 15, 25, 35, 45, 16, 26, 36, 46, 17, 27, 37, 47 are codes for the groups of data each with 10 data points in 4-fold cross validation (Appendix-5 A5).

Comparison of the Selected Models											
Model's code	Overall accuracy	Model AUC	Sig. of Model's AUC	Coefficient	$(\beta) \begin{array}{c} Sig. \ of \\ \beta \end{array}$	Constant (β_o)	Sig of β_o	Cut-off PF	Model Building dataset code	Validation dataset	
Т	84.60%	0.96	0.001	0.99	0.04	-2.657	0.18	0.85	21+31	11(3-fold)	
U	80.80%	0.9	0.001	1.5	0.02	-5.677	0.04	0.9	21+31	11(3-fold)	
V	76.90%	0.84	0.002	1.34	0.02	-5.616	0.03	0.95	21+31	11(3-fold)	
W	73.10%	0.83	0.003	0.92	0.03	-5.06	0.03	1	21+31	11(3-fold)	
Х	73.10%	0.83	0.013	1.12	0.04	-4.745	0.04	1.05	21+31	11(3-fold)	
Y	72.40%	0.66	0.18	0.95	0.03	-3.611	0.07	0.9	24+34+44	14(4-fold)	

Model-Y has insignificant constant and AUC. Therefore, Model-U is selected as the best model. The cut-off productivity factor (PF) of 0.90 and the datasets coded as 21, 31 are used in logistic regression analysis, and the dataset coded 11 is used for validation purpose. The results of the logistic regression analysis of the selected model (Model-U) are provided in Table 8-12.

Table	8-12	Outputs	of	the	Logistic	Regression	Analysis	of	the	Human	Resource
		Manager	men	t Pra	ctices						

			Blo	ck-0						
		(Classifica	tion Tab	le			-		
]	Predicted	1			-		
			Р	٢F	Percent	age Corr	rect			
	Observed		0	1				-		
Step ()	PF	0	0	8		1	0			
Step 0	Overall Percenta	l ge	0	18		6	9.2			
The cur	t value = 0.50					0				
The eu	t value 0.50	17		1 F				-		
		Var D	s E	the Equ	ation df	Sia	$E_{vn}(\mathbf{D})$	-		
Sten ()		Б	5. E	wald	di	Sig	Exp(D)	-		
Step 0	Constant	0.811	0.425	3.642	1	0.056	2.25	_		
		Varia	bles not i	in the Eq	uation			-		
G. 0				Score	df	Sig.				
Step 0	HRM Practices			9.353	1	0.002				
	Overall Statistics	8		9.333 Blo	l ek-1	0.002				
			Omn	ihus Tesi	t for Mo	del Coeff	ficients			
		Chi-sau	are	df	. joi 1100	ici coejj	icientis		Sig	
	Step	11.21		1					0.001	
Step 1	Block	11.21		1					0.001	
_	Model	11.21		1					0.001	
			I	Iosmer a	nd Lem	eshow Te	est			
Step 1	Chi-square	df				Sig.				
~~r	3.278	7				0.858				
				Class	ification	Table	Due die 4e d			
				p	F		Predicted			
	Observed		0	1	1		Percentage Corre	ect		
		0		4		4		50		
Step I	PF	1		1	1	17		94.4		
	Overall Percenta	ge						80.8		
The cu	it value is 0.50									
				Variable	es in the	Equatio	n			
		В	S . E		Wald	df		Sig.		Exp(B)
	HRM Practices	1 504	0.656		5 2 5 6	1		0.022		4.5
Step 1	Constant	-5.677	2.817		4.062	1		0.022		0.003
			Bootst	rapping						
				<u> </u>	Boo	tstrap		-		
						95% C	onfidence Interval	-		
		В	Bias	SE	Sig.	Lower	Upper	-		
Stor 1	HRM Practices	1.504	1.676	18.708	0.006	0.779	5.982	-		
Step 1	Constant	-5.677	-6.702	76.568	0.013	-21.42	-2.374			

As shown in Table 8-12, in Block-0, the true negative percentage is 0%, the true positive percentage is 100%, and overall model accuracy is 69.2%. The variable in the equation is insignificant (p=0.06>0.05) and the variable not in the equation is significant (p=0.02<0.05). The variables not in the equation in Block-0 refer to the independent 220

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

variable or HRM practices. Thus, the model which incorporates the independent variable or the model in Block-1 is more significant.

In Block-1, the Chi-square (11.2) of the Omnibus test is statistically significant at 5% level of significance (p=0.001<0.05) which shows that the model is better than the null model (model in Block-0). The p-value of the Chi-square (3.28) of Hosmer and Lemeshow test of goodness is greater than rejection percentage of the null hypothesis (p=0.86>0.05). Thus, the null hypothesis which states 'the HRM practices and productivity model' is good in fitting the data is accepted.

By using the coefficients presented in the Table 8-12, the equation for the logistic regression model of Productivity and HRM practices is constructed as follows.

$$Log(Odds) = Logit(P_i) = Ln\left(\frac{P_i}{1-P_i}\right) = 1.5HRMP_i-5.68$$

 P_i = probability of exceeding PF=0.9 for an ith project, and HRMP_i = a score of HRM practices for an ith project. The above equation can be simplified as indicated in Equation 8-6.

$$\mathbf{P}_{i} {=} \frac{\mathrm{e}^{\mathrm{1.5HRMP}_{i}{\text{-}5.68}}}{\mathrm{1 {+} e^{\mathrm{1.5HRMP}_{i}{\text{-}5.68}}}} \qquad \text{Equation 8-6}$$

The reliability test results for the HRM practices and productivity model is presented in Table 8-12. The coefficient (1.50) and the constant (-5.68) obtained after bootstrapping using 1000 sample were found to be similar to the model's coefficient (1.50) and constant (-5.68). Moreover, the bootstrapped coefficient and constant are found to be statistically significant (p=0.01<0.05). Therefore, the model is reliable.

Logistic Regression Model of Safety and Health Practices and Productivity

The 35 alternative logistic regression models of safety and health practices and productivity are shown in Table 8-13. Accordingly, using 3-fold cross-validation, two models with the predictive accuracy of 88.5% and 80.8% are selected, and they are coded AB and CD. The p-values of the two models' coefficients are statistically significant. Both models are obtained during the first trial in which the model building datasets coded 21 and 31 were used in model development, and the validation dataset coded 11 was kept for validation purpose (refer Appendix-5 A6 for the raw data and the respective codes). By using the 4-fold cross-validation technique, only one model is found to be statistically significant, and the model was coded EF for further analysis.

Table 8-13 Selection of the Logistic Regression Model of the Safety and Health Practices and Productivity

		Alterna	tive Models Usin	ng 3-fold Cross	Validatio	on		÷		
				Predictive	Accura	y and Signi	ficance of the	Models		
			Tri	ial-1		Trial	2	1	Frial-3	
	Baseline Product	ivity Factors (PFs)	Validation Data Model Building	aset ¹¹ g Dataset ²¹	Model I Validat	Building Dat ion Dataset ² Building Dat	aset ¹²	Model Buil Model Buil	ding Dataset ¹³ ding Dataset ²³	
		Predictive Accuracy (%)	Niodel Duilding		Widder		aser	vanuation		•
	Cut-off PF1=0.85	a: :a	00.2	(AD)		00.0			00.0	
		Significance of β	0	.03		0.47			0.13	
	Cut-off PF2=0.90	Predictive Accuracy (%)	80.8	8 (CD)		65.4			69.2	
		Significance of β	0	.03		0.54			0.24	
	Cut-off PE_=0.95	Predictive Accuracy (%)	7	3.1		65.4			69.2	
	Cut-01113 0.55	Significance of β	0	.08		0.28			0.15	
	Cut-off PE ₄ =1 00	Predictive Accuracy (%)	6	5.4		57.7			65.4	
		Significance of β	0	.14		0.29			0.09	
	Cut-off PF5=1.05	Predictive Accuracy (%)	6	5.4		76.9			57.7	
		Significance of β	0	.07		0.22			0.08	
		Alterna	tive Models Usi	ng 4-fold Cross	Validatio	on	14.11			
			Predic	tive Accuracy an	d Signifi	cance of the	Models			-
		Trial-4		Trial-5	14	Ti	ial-6	1	rial-7	
		Validation Dataset14		Model Building	Dataset'	' Model Bui	ding Dataset	Model Buil	ding Dataset17	
		Model Building Dataset ²⁴		Validation Data:	set ²³	Model Bui	ding Dataset ²⁶	Model Buil	ding Dataset27	
Baselin	e Productivity Factors	Model Building Dataset		Model Building	Dataset	Validation	Dataset 30	Model Buil	ding Dataset"	
	(PFs)	Model Building Dataset**		Model Building	Dataset"	9 Model Bui	ding Dataset ⁴⁰	Validation	Dataset*/	
Cut-off	Predictive Accuracy (%)	86.2		86.2			79.3		83.3	
PF1=0.85	Significance of β	0.05		0.12			0.04		0.13	
Cut-off	Predictive Accuracy (%)	75.9		79.3			52.1		73.3	
PF2=0.90	Significance of B	0.07		0.07			0.19		0.25	
Cut-off	Predictive Accuracy (%)	65.5		72.4			55.5		66.7	
PF3=0.95	Significance of β	0.17		0.08			5.08		0.22	
Cut-0II DE4-1.00	Significance of B	0.26		0.00			0.0		0.14	
Cut off	Predictive Accuracy (%)	69		69.0 (EI	6		82.8		66 7	
Cut-0Π PE5=1.05	Significance of B	0.15		0.04	,) 12		0.07	
115-1.05		0.15		0.04			.12		0.07	
Note: AB, 3-fold cro	, and CD are codes for goo ss validation (Appendix-5 the groups of data and wi	A6); EF is code for good m th 10 data points in 4 fold.	alidation; 11, 21, nodel in 4-fold cr	31, 12, 22, 32, 1 oss-validation ; 1	3, 23, an 4, 24, 34	d 33 are cod , 44, 15, 25,	es for the data 35, 45, 16, 26	sets each with 5, 36, 46, 17,	h 13 elements in 27, 37, 47 are	
coues ioi	the groups of data cach wi	un ro data points in 4-1010 (Comparison of t	he Selected Mor	اماد					
Model's code	Overall accuracy	Model AUC	Sig. of Model's AUC	Variable's β	Sig. of β	Constan t (β _o)	Sig of β_o	Cut-off PF	Model Building dataset code	Validation dataset co
AB	88.50%	0.94	0.04	1.85	0.03	-8.071	0.07	0.85	21+31	11
CD	80.80%	0.83	0.03	1.81	0.03	-8.787	0.04	0.9	21+31	11

69.00% 0.71 0.06 0.04 -9 843 0.04 1.05 15+35+45 Model-AB, Model-CD, and Model-EF are compared to choose the best safety and health and productivity model (Table 8-13). Model-AB has the highest overall predictive accuracy (88.5%), acceptable AUC (0.94>0.5), and statistically significant pvalue (p = 0.03 < 0.05) for the variable (safety and health practices). However, the constant of the Model-AB is not statistically significant (p = 0.07 > 0.05); as a result, it is not selected as the best model. Model-CD has a better overall predictive accuracy (80.8%), acceptable AUC (0.83>0.5), statistically significant p-values for the independent variable (p = 0.03 < 0.05) and constant (p = 0.04 < 0.05). Model-EF has insignificant AUC (p = 0.06 > 0.05) and lowest overall predictive accuracy as compared to Model-CD. Therefore, Model-CD is chosen as the best safety and health practices and productivity model. The results of the analysis for the selected model are presented in Table 8-14.

Table 8-14 Outputs of the Logistic Regression Analysis of Safety and Health Practices

			Block-	0					
		Class	ification	Table				_	
				Predicted	1			_	
			I	PF	-Percent	age Cor	rect		
	Observed		0	1	1 010 011			_	
G. 0	PF	0	0	8			0		
Step 0	O11 D	I	0	18			100		
The cur	Overall Percentage						69.2		
The cu	t value is 0.50	Variable	es in the	Eauatio	1			-	
~ ^		В	S. E	Wald	df	Sig	Exp(B)	_	
Step 0	Constant	0.811	0.425	3.642	1	0.056	2.25	-	
	V	ariables	not in th	e Equati	on			_	
				Score	df	Sig.		_	
Step 0	Safety and Health Practices			7.459	1	0.006			
	Overall Statistics			7.459	1	0.006			
				Block-1	l				
		(Omnibus	Test for	Model (Coefficier	nts		
	_	Chi-squ	are	df				Sig	5.
a	Step	7.966		1				0.0	105
Step 1	Block	7.966		1				0.0	105
	Model	7.900	Hosn	1 1er and 1	omosha	w Test		0.0	05
	Chi-square	df	11050	ier unu i	emesno	Sig			
Step 1	7.192	6				0.303			
			(Classifica	tion Tab	ole			
							Predicted		
				F	Έ		- Percentage Correct		
	Observed		0		1		i electituge contect		
Step 1	PF	0		4		4		50	
-	Overell Deve entered	1		1		[7		94.4	
The cr	overall Percentage							80.8	
1110 00			Var	iahles in	the Eau	ation			
		В	S	. E	Wald		df	Sig.	Exp(B)
	Safety and Health Practices	1.813	0.	817	4.921		1	0.027	6.126
Step 1	Constant	-8 787	4	342	4 096		1	0.043	0
	Constant	0.707 Bo	otstran	ning	4.070		1	0.015	
		D 0	ownap	21115	Ro	otstran		-	
					20	95%	Confidence Interval	-	
		В	Bias	SE	Sig.	Lower	Upper	_	
Stop 1	Safety and Health Practices	1.813	1.813	17.73	0.005	0.709	5.041		
Step 1	Constant	-8.787	-9.62	81.61	0.006	-25.24	-3.047		

As shown in Table 8-14, the results of the logistic regression analysis in Block-0 show the true negative percentage of 0%, the true positive percentage of 100%, and overall model accuracy of 69.2%. The variable in the equation is insignificant (p=0.06>0.05) and the variable not in the equation is significant (p=0.02<0.05) which implies that the model in Block-1 is better. In Block-1, the Chi-square (7.97) of the Omnibus test is statistically significant at 5% level of significance (p=0.005<0.05) which shows that the model in Block-1 is better than the null model (the model in Block-0). Similarly, the p-²²³

value (0.30) of the Chi-square (7.19) of Hosmer and Lemeshow test is greater than 5% which indicates that the model in Block-1 is good.

The overall prediction accuracy of the model in Block-1 is 80.8% which is better than the model in Block-0's prediction accuracy (69.2%). The true negative prediction accuracy of the model is 50.0%, and the true positive prediction accuracy is 94.4%. The Wald Chi-square (4.92) for the variable in the equation is statistically significant at 5% level of significance (p=0.03<0.05). The Wald Chi-square (4.10) for the constant is also statistically significant (p=0.04<0.05). The Exp(B) of the variable safety and health practices is 6.1 which implies that for a unit increase in the score of the safety and health practice, the odd of exceeding the productivity factor of 0.90 increases by a factor of 6.1.

By using the coefficients shown in the Table 8-14, the equation for safety and health practices and productivity model is constructed as follows. In the model, P_i is the probability of exceeding productivity factor of 0.9 for an ith project, SHP_i refers to a score of safety and health practices for an ith project, and Ln= natural logarithm.

$$Log(Odds) = Logit(P_i) = Ln\left(\frac{P_i}{1 - P_i}\right) = 1.81SHP_i - 8.79$$

or
$$P_i = \frac{e^{1.81SHP_i \cdot 8.79}}{1 + e^{1.81SHP_i - 8.79}}$$
Equation 8-7

In Table 8-14, the results of the test to check the reliability of the logistic regression model of the safety and health practices and productivity are presented. Accordingly, the coefficient of the independent variable (1.81) and the constant (-8.79) after bootstrapping using 1000 samples are found to be statistically significant at 1% level of significance. This indicates that the safety and health practices and productivity model is reliable.

8.2.2 Linear Regression Models of the Construction Management Practices and Productivity

To investigate the proportion of variance in the productivity factor explained by the construction management practices and to develop linear regression model of productivity and the construction management practices, bivariate linear regression analysis was conducted. Before running the analysis, the scatter plots of the productivity

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

factors and the overall scores of the construction management practices were analysed (Figure 8-1). The graph indicates that the data can be represented by the linear model.

The 3-fold and 4-fold cross-validation techniques were used, and seven alternative linear regression models of the 47 construction management practices and productivity were developed (Table 8-15). The models are compared based on their p-values and coefficient of determination (R-squared). The raw data and the corresponding codes are shown in Appendix-5 A7. Model-a is selected as the best linear regression model of construction management practices and productivity since it has the highest R^2 and a statistically significant p-value.



Figure 8-1 Scatter Plot of Productivity Factors and Construction Management Practices

 Table 8-15 Alternative Linear Regression Models of Productivity and Construction

 Management Practices

3-Fold Cross Validation											
	Validation Dataset ¹¹	Model Building Dataset ¹²	Model Building Dataset ¹³								
	Model Building Dataset ²¹	Validation Dataset ²²	Model Building Dataset ²³								
	Model Building Dataset ³¹	Model Building Dataset 32	Validation Dataset 33								
	Model-a	Model-b	Mod	lel-c							
R^2	0.479	0.229	0.15								
P-value	<0.001	0.013	0.05								
В	0.02	0.014	0.01								
Во	0.575	0.658	0.744								
		4-Fold Cross Valida	tion								
	Validation Dataset ¹⁴	Model Building Dataset ¹⁵	Model Building Dataset ¹⁶	Model Building Dataset ¹⁷							
	Model Building Dataset ²⁴	Validation Dataset ²⁵	Model Building Dataset ²⁶	Model Building Dataset27							
	Model Building Dataset ³⁴	Model Building Dataset35	Validation Dataset ³⁶	Model Building Dataset37							
	Model Building Dataset ⁴⁴	Model Building Dataset45	Model Building Dataset46	Validation Dataset ⁴⁷							
	Model-e	Model-f	Model-g	Model-h							
R^2	0.414	0.269	0.294	0.132							
P-value	< 0.001	0.004	0.002	0.048							
В	0.017	0.014	0.014	0.009							
Bo	0.561	0.644	0.607	0.751							

The results of the linear regression analysis of the selected model (Model-a) are presented in Table 8-16. Accordingly, $R^2 = 0.479$ and a p-value less than 0.001 are found. The findings imply that there is a significant relationship between productivity and the management practices, and about 48% of the total variance in productivity factor can be explained by the management practices and the remaining variance could be explained by factors such as advance in technology and government regulations among others.

Table 8-16 Outputs of the Linear Regression Analysis of the Construction Management Practices

			Model S	Summary		
R		R Squ	are	Adjusted R Square		Std. Error of the Estimate
0.692		0.479		0.458		0.104
			AN	OVA		
	Sum of Squar	res	df	Mean Square	F	Sig.
Regression	0.214		1	0.241	22.092	< 0.001
Residual	0.262		24	0.011		
Total	0.503		25			
			Coeff	îcients		
	_	Unstandardize	d Coefficients	Standardized Coefficients		
	_	В	Std. Error	Beta	t	Sig
(Constant)		0.575	0.084		6.812	< 0.001
Construction Management Practices		0.02	0.004	0.692	4.7	< 0.001

The residuals were analysed to check the assumptions of the linear regression model of productivity and the overall or the 47 construction management practices (Figure 8-2). The Durbin-Watson statistic of 1.92 is found, and the value is close to 2 indicating that the assumption of independence of errors has been met (Thomas and Sudhakumar, 2014). The histogram in Figure 8-2 shows the residuals are normally distributed. The plot of the standardised residuals and standardised predicted values indicates that the points are randomly distributed, and the assumption of homoscedasticity (equality of variances) is met. The scatter plot of the productivity factor and the construction management practices shows a linear relationship. Therefore, the assumption of linearity, independence of errors, homoscedasticity and normality of the errors are satisfied.





Figure 8-2 Outputs of the Tests Conducted to Confirm the Assumptions of the Linear Regression Model of Productivity and the Overall Construction Management Practices

Finally, by using the coefficients indicated in Table 8-16, the equation for the linear regression model of the overall construction management practices and productivity is constructed as follows:

$PF_i = 0.02CMP_i + 0.58$

Where PF_i refers to the productivity factor of an ith project and CMP_i is the overall score of the 47 construction management practices for an ith project.

Linear regression analyses were also conducted to develop the regression models of the six categories of the construction management practices, and the results are presented in the following subsections.

Linear Regression Model of Construction Materials Management Practices and Productivity

The results of the bivariate linear regression analysis conducted to choose the best construction materials management practices and productivity model are shown in

Table 8-17. Accordingly, Model-l is selected as the best model since it has the highest R-squared value and statistically significant coefficients.

3-Fold Cross Validation							
	Validation Dataset ¹¹	Model Building Dataset ¹²	Model Building Dataset ¹³				
	Model Building Dataset ²¹	Validation Dataset ²²	Model Building Dataset ²³				
	Model Building Dataset ³¹	Model Building Dataset 32	² Validation Dataset ³³				
	Model-i	Model-j	Model-	k			
R^2	0.243	0.114	0.17				
P-value	0.01	0.091	0.04				
B (Cofficient)	0.088	0.059	0.061				
Bo (Constant)	0.749	0.795	0.811				
	4-Fold Cross Validation						
	Validation Dataset ¹⁴	Model Building Dataset ¹⁵	Model Building Dataset ¹⁶	Model Building Dataset ¹⁷			
	Model Building Dataset ²⁴	Validation Dataset ²⁵	Model Building Dataset ²⁶	Model Building Dataset ²⁷			
	Model Building Dataset ³⁴	Model Building Dataset ³⁵	Validation Dataset ³⁶	Model Building Dataset ³⁷			
	Model Building Dataset ⁴⁴	Model Building Dataset ⁴⁵	Model Building Dataset ⁴⁶	Validation Dataset47			
	Model-l	Model-m	Model-n	Model-o			
R^2	0.262	0.12	0.181	0.133			
P-value	0.01	0.07	0.02	0.048			
B (Cofficient)	0.09	0.059	0.072	0.054			
Bo (Constant)	0.741	0.821	0.751	0.836			

 Table 8-17 Alternative Linear Regression Models of Construction Materials Management

 Practices and Productivity

The results of the analysis carried out to develop Model-1 are shown in Table 8-18. R^2 value of 0.262 and *p*-value of 0.01 < 0.05 are found. Thus, the model is significant. The finding indicates that about 26.2% of the variances in the productivity factor can be explained by construction materials management practices and the remaining percentage could be explained by other management practices and technology among others.

Table 8-18 Outputs of the Linear Regression Analysis of the Construction Materials Management Practices

		Mod	lel Summary			
R	R Square Adjusted R Square		Adjusted R Square	Std. Error of the Estimate		ate
0.512	0.2	262	0.235	0.12		
		-	ANOVA			
	Sum of Squares	df	Mean Square	F	Sig.	
Regression	0.139	1	0.139	9.594	0.005	
Residual	0.391	27	0.014			
Total	0.53	28				
		С	oefficients			
	Unstandar	dized Coefficients	Standardized Coefficients	_		
	В	Std. Error	Beta	t	Sig	
(Constant)	0.741	0.073		10.201	< 0.001	
Construction Materials Management Practices	0.09	0.029	0.512	3.097	0.005	

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

The scatter plot of productivity factor and the construction materials management practices, and the results of the residuals analyses are indicated in Figure 8-3. The Durbin-Watson statistic of 2.08 is obtained indicating that the errors are independent. The residuals are approximately normally distributed (Figure 8-3). The plot of the standardised residuals and standardised predicted value shows that the errors are randomly distributed, and the assumption of homoscedasticity is satisfied.



Figure 8-3 Outputs of the Tests Conducted to Check the Assumptions of the Linear Regression Model of Productivity and Construction Materials Management Practices

Based on the outputs presented in Table 8-18, the linear regression model of productivity and construction materials management practices can be represented by the following equation.

$PF_i = 0.09CMMP_i + 0.74$

 PF_i = the productivity factor of an ith project and CMMP_i = a score of the construction materials management practices for an ith project.

Linear Regression Model of Construction Equipment Management Practices and Productivity

In Table 8-19, the results of the linear regression analysis carried out to choose the most appropriate linear regression model of construction equipment management practices and productivity are presented. The model with the highest coefficient of determination (0.234) is chosen (Model-q), and the results of the analysis made to develop the model are shown in Table 8-20.

 Table 8-19 Alternative Linear Regression Models of Construction Equipment Management

 Practices and Productivity

3-Fold Cross Validation							
	Validation Dataset ¹¹	Model Building Dataset 12	Model Building Dataset 13				
	Model Building Dataset ²¹	Validation Dataset ²²	Model Building Dataset ²³				
	Model Building Dataset ³¹	Model Building Dataset 32	Validation Dataset ³³				
	Model-q	Model-r	Моа	lel-s			
R ²	0.234	0.123	0.104				
P-value	0.012	0.08	0.11				
Bo (Constant)	0.844	0.815	0.875				
B (Cofficient)	0.13	0.132	0.085				
4-Fold Cross Validation							
	Validation Dataset ¹⁴	Model Building Dataset15	Model Building Dataset ¹⁶	Model Building Dataset ¹⁷			
	Model Building Dataset ²⁴	Validation Dataset ²⁵	Model Building Dataset ²⁶	Model Building Dataset27			
	Model Building Dataset ³⁴	Model Building Dataset ³⁵	Validation Dataset ³⁶	Model Building Dataset ³⁷			
	Model Building Dataset44	Model Building Dataset45	Model Building Dataset46	Validation Dataset47			
	Model-t	Model-u	Model-v	Model-w			
R ²	0.198	0.183	0.122	0.077			
P-value	0.016	0.02	0.063	0.138			
Bo (Constant)	0.845	0.826	0.825	0.896			
B (Cofficient)	0.121	0.139	0.111	0.073			

As shown in Table 8-20, the R^2 of the selected model is 0.234 which indicates that construction equipment management practices can explain about 23.4% of the variances in productivity factor. The coefficient and constant of the model have *p*-values of 0.012 and < 0.001 respectively indicating that the model is significant.

 Table 8-20 Outputs of the Linear Regression Analysis of the Construction Equipment

 Management Practices

			Model Summary			
R	R Square		Adjusted R Square		Std. Error of the Estimate	
0.692	0.234		0.202		0.127	
			ANOVA			
Su	m of Squares	df	Mean Square	F	Sig.	
Regression	0.117	1	0.117	7.314	0.012	
Residual	0.385	24	0.016			
Total	0.503	25				
			Coefficients			
	Unstandardize	ed Coefficients	Standardized Coefficients			
	В	Std. Error	Beta	t	Sig	
(Constant)	0.844	0.049		17.052	< 0.001	
Construction Equipme Management Practices	nt 0.13	0.048	0.483	2.704	0.012	
					23	

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

The Durbin-Watson statistic is found to be 2.09, and the value is close to 2 indicating that the errors are not auto-correlated. The histogram of the residuals of the construction equipment management practices and productivity model is approximately normally distributed (Figure 8-4). The scatterplot of residuals and predicted values indicates that the variances of the errors are homogeneous. Therefore, the assumptions of the linear regression models are satisfied, and the model is acceptable.



Figure 8-4 Outputs of the Tests Conducted to Analyse the Assumptions of the Linear Regression Model of Productivity and Construction Equipment Management Practices

Based on the results indicated in Table 8-20, the linear regression model of productivity and the construction equipment management practices is constructed as shown in the following equation.

$PF_i = 0.13CEMP_i + 0.84$

In the equation PF_i = the productivity factor of an ith project and $CEMP_i$ = a score of the construction equipment management practices for an ith project.

Linear Regression Model of Management Practices Related to Construction Methods and Productivity

Among the seven alternative linear models developed using 3-fold and 4-fold cross validation techniques, the model coded 'x' is chosen as the most appropriate model as it has the highest R-squared (Table 8-21).

3-Fold Cross Validation							
	Validation Dataset ¹¹	Model Building Dataset 12	Model Building Dataset 13				
	Model Building Dataset ²¹	Validation Dataset ²²	Model Building Dataset ²³				
	Model Building Dataset ³¹	Model Building Dataset 32	Validation Dataset ³³				
	Model-x	Model-y	Мос	lel-z			
R ²	0.408	0.194	0.097				
P-value	<0.001	0.024	0.121				
Bo (Constant)	0.605	0.698	0.799				
B (Cofficient)	0.076	0.049	0.032				
	4-Fold Cross Validation						
	Validation Dataset ¹⁴	Model Building Dataset ¹⁵	Model Building Dataset ¹⁶	Model Building Dataset ¹⁷			
	Model Building Dataset ²⁴	Validation Dataset ²⁵	Model Building Dataset ²⁶	Model Building Dataset ²⁷			
	Model Building Dataset ³⁴	Model Building Dataset ³⁵	Validation Dataset ³⁶	Model Building Dataset ³⁷			
	Model Building Dataset44	Model Building Dataset ⁴⁵	Model Building Dataset ⁴⁶	Validation Dataset47			
	Model-ab	Model-cd	Model-ef	Model-gh			
R^2	0.346	0.175	0.26	0.09			
P-value	0.001	0.024	0.005	0.108			
Bo (Constant)	0.63	0.732	0.657	0.817			
B (Cofficient)	0.069	0.048	0.055	0.03			

 Table 8-21 Outputs of the Linear Regression Analysis of the Management Practices Related to

 Construction Methods

The outputs of the analysis for the selected model (Model-x) is shown in Table 8-22. Accordingly, $R^2 = 0.408$ and p < 0.001 are found. The findings indicate that the model is significant and 40.8% of the variance in PF can be explained by management practices related to construction methods.

			Mode	el Summary		
R R Sq		iare	Adjusted R Square		Std. Error of the Estimate	
0.639		0.408		0.383		0.111
			1	ANOVA		
	Sum of S	quares	df	Mean Square	F	Sig.
Regression	0.20	5	1	0.205	16.533	< 0.001
Residual	0.28	8	24	0.012		
Total	0.50	13	25			
			Са	pefficients		
		Unstandardize	ed Coefficients	Standardized Coefficients		
		В	Std. Error	Beta	t	Sig
(Constant)		0.605	0.09		6.738	< 0.001
Management P Construction M	ractices Related to Methods	0.076	0.019	0.639	4.066	< 0.001

Table 8-22 Outputs of the Linear Regression Analysis of the Management Practices Related to Construction Methods

The analysis of the residuals indicates the Durbin-Watson statistic of 2 which implies that the errors are independent. The histogram of the errors is approximately normally 232

distributed (Figure 8-5). The scatter plot of the standardised residuals and predicted value shows that the assumption of the homoscedasticity is fulfilled since the points in the graph are randomly distributed (Figure 8-5).



Figure 8-5 Outputs of the Tests Conducted to Verify the Assumptions of the Linear Regression Model of Productivity and the Management Practices Related to Construction Methods

The equation for the linear regression model of productivity and the management practices related to construction methods is constructed as follows.

$PF_i = 0.08MPRCM_i + 0.61$

 PF_i is the productivity factor of an i^{th} project, and MPRCM_i represents a score of the management practices related to construction methods for an i^{th} project.

Linear Regression Model of Pre-Construction Phase Management Practices and Productivity

The results of the linear regression analysis of the preconstruction phase management practices conducted using the 3-fold and 4-fold cross-validation techniques are shown in Table 8-23. Model-kl in the 3-fold cross-validation has the highest R-squared, but the model's constant (-2.16) is statistically insignificant (p=0.1>0.05), and it is not chosen
as the best model. Model-ij is selected as the most appropriate model as it has the second highest R-squared (Table 8-23).

		3-Fold Cross Validat	ion		
	Validation Dataset ¹¹	Model Building Dataset ¹²	Model Building Dataset 13		
	Model Building Dataset ²¹	Validation Dataset ²²	Model Building Dataset ²³		
	Model Building Dataset ³¹	Model Building Dataset 32	Validation Dataset 33		
	Model-ij	Model-kl	Mode	el-mn	
R ²	0.342	0.43	0.156		
P-value	0.002	< 0.001	0.046		
Bo (Constant)	0.711	-2.155	0.769		
B (Cofficient)	0.058	1.147	0.039		
		4-Fold Cross Validat	ion		
	Validation Dataset ¹⁴	Model Building Dataset15	Model Building Dataset ¹⁶	Model Building Dataset17	
	Model Building Dataset ²⁴	Validation Dataset ²⁵	Model Building Dataset ²⁶	Model Building Dataset ²⁷	
	Model Building Dataset ³⁴	Model Building Dataset ³⁵	Validation Dataset ³⁶	Model Building Dataset ³⁷	
	Model Building Dataset44	Model Building Dataset45	Model Building Dataset46	Validation Dataset47	
	Model-op	Model-qr	Model-st	Model-uv	
R^2	0.273	0.232	0.305	0.109	
P-value	0.004	0.08	0.002	0.075	
Bo (Constant)	0.725	0.731	0.661	0.812	
B (Cofficient)	0.053	0.052	0.058	0.033	

Table 8-23 Outputs of the Linear Regression Analysis of the Pre-Construction Phase Management Practices

The results of the detailed analysis of the Model-ij are presented in Table 8-24. Accordingly, R^2 value of 0.342 and *p*-value of 0.002 < 0.05 are obtained. The findings show that the regression model is statistically significant and about 34.2% of the variance in productivity factor can be explained by the pre-construction phase management practices. The following equation can represent the model. In the equation, PF_i is the productivity factor of an ith project, and PCPM_i is a score of the pre-construction phase management practices for an ith project.

$PF_i = 0.06PCPM_i + 0.71$

Table 8-24 Outputs of the Linear Regression Analysis of the Pre-Construction Phase Management Practices

				Model Summary			
R		R Squ	lare	Adjusted R Square	S	td. Error of the Estim	ate
0.585		0.342		0.314	C	.117	
				ANOVA			
	Sum o	of Squares	df	Mean Square	F	Sig.	
Regression	0	0.172	1	0.172	12.454	0.002	
Residual	0	0.331	24	0.012			
Total	0	0.503	25				
				Coefficients			
	Ţ	Unstandardize	ed Coefficients	Standardized Coefficients			
		В	Std. Error	Beta	t	Sig	
(Constant)		0.711	0.074		9.598	< 0.001	
Pre-Construction Management Prac	Phase ctices	0.058	0.016	0.585	3.529	0.002	

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

The outputs of the test conducted to verify the assumption of the linear regression model of productivity and the pre-construction phase management practices are indicated in Figure 8-6. Accordingly, the assumptions of linearity of the data, as well as the normality, homoscedasticity and the independence of errors are satisfied (Figure 8-6). The residuals are normally distributed, and the residuals are not related to the predicted values. The Durbin-Watson statistic is found to be 2.1. The value is close to 2 indicating that the errors are not dependent.



Figure 8-6 Outputs of the Tests Conducted to Confirm the Assumptions of the Linear Regression Model of Productivity and the Pre-Construction Phase Management Practices

Linear Regression Model of the Human Resource Management Practices and Productivity

In Table 8-25, the outputs of the analysis carried out to select the most suitable linear regression model of the human resource management practices and productivity are presented. Accordingly, Model-wx is chosen since it has the highest coefficient of

determination and its coefficients are statistically significant (Table 8-25). The results of the analysis conducted to develop Model-wx are shown in Table 8-26.

		3-Fold Cross Validat	ion			
	Validation Dataset ¹¹	Model Building Dataset 12	Model Building Dataset ¹³			
	Model Building Dataset ²¹	Validation Dataset ²²	Model Building Dataset ²³	Iodel Building Dataset ²³		
	Model Building Dataset ³¹	Model Building Dataset 32	Validation Dataset ³³			
	Model-wx	Model-yz	Mod	el-ac		
R^2	0.48	0.187	0.032			
P-value	<0.001	0.027	0.38			
Bo (Constant)	0.642	0.757	0.874			
B (Cofficient)	0.069	0.04	0.018			
		4-Fold Cross Validat	ion			
	Validation Dataset ¹⁴	Model Building Dataset15	Model Building Dataset ¹⁶	Model Building Dataset17		
	Model Building Dataset ²⁴	Validation Dataset ²⁵	Model Building Dataset ²⁶	Model Building Dataset ²⁷		
	Model Building Dataset ³⁴	Model Building Dataset ³⁵	Validation Dataset ³⁶	Model Building Dataset ³⁷		
	Model Building Dataset44	Model Building Dataset45	Model Building Dataset46	Validation Dataset47		
	Model-ad	Model-ae	Model-af	Model-ah		
R^2	0.39	0.232	0.183	0.035		
P-value	< 0.001	0.008	0.021	0.32		
Bo (Constant)	0.662	0.761	0.742	0.877		
B (Cofficient)	0.063	0.044	0.04	0.019		

 Table 8-25 Outputs of the Linear Regression Analysis of the Human Resource Management

 Practices

As indicated in Table 8-26, the R^2 of the HRM practices and productivity model is 0.48; the constant has a *p*-value less than 0.001; and the coefficient also has a p-value less than 0.001. The findings imply that the model is statistically significant and about 48% of the variance in the productivity factor can be explained by human resource management practices. The model's equation can be constructed as follows.

$PF_i = 0.07HRMP_i + 0.64$

Where PF_i = the productivity factor of an ith project and $HRMP_i$ = a score of the human resource management practices for an ith project.

		M	odel Summary							
R	R Sc	luare	Adjusted R Square		Std. Error of the Estimate					
0.693	0.48		0.458		0.104					
	ANOVA									
	Sum of Squares	df	Mean Square	F	Sig.					
Regression	0.241	1	0.241	22.145	< 0.001					
Residual	0.262	24	0.11							
Total	0.503	25								
			Coefficients							
	Unstandardi	zed Coefficients	Standardized Coefficients							
	В	Std. Error	Beta	t	Sig					
(Constant)	0.642	0.071		9.1	< 0.001					
Human Resource Management Practice	es 0.069	0.015	0.693	4.706	< 0.001					
					23					

Table 8-26 Outputs of the Linear Regression Analysis of the Human Resource Management Practices

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

The residuals of the model were analysed to check the assumptions of the linear regression analysis and the results are shown in Figure 8-7. The residuals are normally distributed. The Durbin-Watson statistic (1.75) is approximately close to 2 indicating that the errors are not auto-correlated. In the graph of the standardised residuals and standardised predicted values, the residuals are randomly scattered indicating that the assumptions of the homoscedasticity are satisfied.



Figure 8-7 Outputs of the Tests Conducted to Verify the Assumptions of the Linear Regression Model of Productivity and the Human Resource Management Practices

Linear Regression Model of the Safety and Health Practices and Productivity

Seven alternative linear regression models of safety and health practices and productivity were developed before selecting the final model (Table 8-27). Model-am has the highest R^2 , but the constant of the model is statistically insignificant (*p*=0.07>0.05) and it is not selected as the best model. Thus, the model with the second highest R-squared (Model-ai) is chosen. The outputs of the selected model are shown in Table 8-28. Accordingly, R² value of 0.268, *p-value* of the constant = 0.006 < 0.05, and

the *p*-value of the coefficient = 0.007 < 0.05 are found. Thus, the model is significant. The proportion of variance in productivity factor explained by safety and health practices is 26.8%.

		3-Fold Cross Validat	ion			
	Validation Dataset ¹¹	Model Building Dataset 12	Model Building Dataset 13			
	Model Building Dataset ²¹	Validation Dataset ²²	Model Building Dataset ²³			
	Model Building Dataset ³¹	Model Building Dataset 32	Validation Dataset 33			
	Model-ai	Model-aj	Mode	el-ak		
R ²	0.268	0.158	0.202			
P-value	0.007	0.044	0.021			
Bo (Constant)	0.486	0.508	0.574			
B (Cofficient)	0.088	0.079	0.07			
		4-Fold Cross Validat	ion			
	Validation Dataset ¹⁴	Model Building Dataset ¹⁵	Model Building Dataset ¹⁶	Model Building Dataset ¹⁷		
	Model Building Dataset ²⁴	Validation Dataset ²⁵	Model Building Dataset ²⁶	Model Building Dataset ²⁷		
	Model Building Dataset ³⁴	Model Building Dataset ³⁵	Validation Dataset ³⁶	Model Building Dataset ³⁷		
	Model Building Dataset ⁴⁴	Model Building Dataset45	Model Building Dataset46	Validation Dataset47		
	Model-al	Model-am	Model-an	Model-ao		
R^2	0.187	0.276	0.201	0.17		
P-value	0.019	0.003	0.015	0.024		
Bo (Constant)	0.563	0.362	0.528	0.613		
B (Cofficient)	0.072	0.111	0.073	0.064		

Table 8-27 Outputs of the Linear Regression Analysis of the Safety and Health Practices

Based on the constant and coefficient shown in Table 8-28, the linear regression model of safety and health practices and productivity can be expressed using the following equation. In the equation, PF_i stands for the productivity factor of an ith project, and SHP_i is a score of the safety and health practices for an ith project.

$PF_i = 0.09SHP_i + 0.49$

	Model Summary											
R	R Sc	uare	Adjusted R Square		Std. Error of the Estimate							
0.517	0.268	3	0.237		0.124							
			ANOVA									
Su	im of Squares	df	Mean Square	F	Sig.							
Regression	0.135	1	0.135	8.773	0.007							
Residual	0.368	24	0.015									
Total	0.503	25										
			Coefficients									
	Unstandardiz	ed Coefficients	Standardized Coefficients									
	В	Std. Error	Beta	t	Sig							
(Constant)	0.486	0.162		3.01	0.006							
Safety and Health Practic	ces 0.088	0.03	0.517	2.96	0.007							

The Durbin-Watson statistic is found to be 2.2 which is close to two indicating that the errors are independent. The histogram of the residuals shows that the errors are approximately normally distributed (Figure 8-8). The scatter plot of the regression standardised residuals and regression standardised predicted value indicates that the

errors are homogenous. Hence, the assumptions of the linear regression analysis are met.



Figure 8-8 Outputs of the Tests Conducted to Confirm the Assumptions of the Linear Regression Model of Productivity and the Safety and Health Practices

8.3 Validating the Regression Models

8.3.1 Validation of the Logistic Regression Model of the Construction Management Practices and Productivity

To validate the logistic regression model of the overall construction management practices and productivity, probabilities of exceeding baseline productivity factor of 0.90 was predicted by using the validation dataset coded '11' in Table 8-1 as inputs to the model, the Receiver Operating Characteristic (ROC) curve was plotted, and the Area under the Curve (AUC) was computed using SPSS-24. Figure 8-9 indicates the ROC curve and the diagonal line which is used as a reference line. The curve is above the reference line, and the AUC value of 0.73 which is greater than 0.5 is found. Thus, the

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

model is valid. The minimum acceptable value of the AUC is 0.5, and the graph should be above the diagonal line (Hanley and McNeil, 1982).



Figure 8-9 ROC Curve for Validating the Logistic Regression Model of the Overall Construction Management Practices and Productivity

By using the model's equation (Equation 8-1), probabilities are computed, and the sigmoid graph indicated in Figure 8-10 is drawn so that users can easily read probability that corresponds to an overall score of the construction management practices. Some parts of the data used in constructing the graph are indicated in Table 8-29.

Table 8-29 Predicted Probabilities Using Construction Management Practices and

B_0				-7.65						
B_1				0.41						
CMP Score	0	5	10	15	20	25	27	29	30	31
$B_0+B_1\times CMP$	-7.65	-5.62	-3.58	-1.55	0.49	2.52	3.34	4.15	4.56	4.96
P (%)	0.05	0.36	2.70	17.52	61.62	92.56	96.56	98.45	98.96	99.31





240

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

P (%)

0.16

0.94

5.49

Regression Models of Construction Management Practices and Productivity

The results of the analysis conducted to validate the logistic regression models of the six categories of the construction management practices are explained in the subsequent sections.

Validation of the Logistic Regression Model of the Construction Materials Management Practices and Productivity

During the validation process of the logistic regression model of the construction materials management practices and productivity, probabilities of exceeding baseline productivity factor of 1.05 were predicted by using the validation dataset coded '36' in Table 8-3 as inputs to Equation 8-2 (Materials Management Practices and Productivity Model). The Receiver Operating Characteristic (ROC) of the data was plotted, and its Area under the Curve (AUC) was calculated. As shown in Figure 8-11, the ROC curve is above the diagonal line indicating that the model is valid. The AUC= 0.81 > 0.5 which also indicates that the model is valid.



Figure 8-11 ROC Curve for Validating the Logistic Regression Model of the Construction Materials Management Practices and Productivity

After validation, the model's equation (Equation 8-2) is used to predict the probabilities of exceeding PF=1.05, and the sigmoid graph is plotted as shown in Figure 8-12. In Table 8-30, the data used to draw the probability graph is presented.

1 4010 0 0	• • • • • • • •									••••••			
B_0							-6.416						
B_{I}							1.808						
MMP Sco	ore 0	1	2	3	4	4.5	5	5.5	6	6.5	7	8	
$B_0 + B \times M_1$	MP -6.4	2 -4.65	-2.85	-1.04	0.77	1.68	2.58	3.48	4.39	5.29	6.20	8.00	

Table 8-30 Predicted Probabilities Using Construction Materials Management Practices Model

Note: B_0 = the constant, B= the coefficient of the variable, MMP= Construction Materials Management Practices Score, and P= probability of exceeding PF=1.05.

26.17 68.37 86.48 92.95 97.02 98.77

99.50 99.80 99.97



Figure 8-12 Probability Graph of the Construction Materials Management Practices

Validation of the Logistic Regression Model of Construction Equipment Management Practices and Productivity

To validate the logistic regression model of the construction equipment management practices and productivity, probabilities of validation dataset coded '11' in Table 8-5 were computed; productivity factors were predicted as binary values (1, 0) and compared with the actual productivity factors; Receiver Operating Characteristic (ROC) curve was drawn and compared with the reference line; and Area under the ROC Curve (AUC) was computed. Accordingly, about 85% of the PF's are correctly predicted indicating the strength of the model (Table 8-31).

 Table 8-31 Predictive Accuracy of the Logistic Regression Model of the Construction

 Equipment Management Practices and Productivity

Respondents' Code	Actual PF	B_{0}	B_l	CEMP Score	(B ₀ +B ₁ *CEM P) or L	$P = (e^L)/(1 + e^L)$	Predicted PF	Binary Equivalent for Actual PF	Remark
R33	0.90			0.42	-0.024	0.51	1	1.00	Correct
R18	1.00			0.00	-0.980	0.27	0	1.00	Incorrect
R14	0.82			0.44	0.021	0.48	0	0.00	Correct
R12	0.79			1.41	2.249	0.90	1	0.00	Incorrect
R39	1.10			1.15	1.651	0.83	1	1.00	Correct
R26	0.76			0.00	-0.980	0.27	0	0.00	Correct
R38	1.00	-0.98	2.29	0.70	0.615	0.65	1	1.00	Correct
R30	1.12			1.49	2.437	0.92	1	1.00	Correct
R9	1.14			1.23	1.843	0.86	1	1.00	Correct
R27	1.08			1.55	2.567	0.93	1	1.00	Correct
R37	1.08			2.09	3.800	0.98	1	1.00	Correct
R29	0.92			1.47	2.397	0.92	1	1.00	Correct
R35	1.06			0.59	0.382	0.59	1	1.00	Correct
Percentage Con	rectly pr	adjated							85%

Note: B_0 = the constant, B_1 = the coefficient of the variable, CEMP= Construction Equipment Management Practices Score, and P= probability of exceeding PF=0.90.

ROC curve is drawn, and the curve of the graph is greater than the reference line indicating that the model is good in prediction (Figure 8-13). Furthermore, the area under the curve was computed to be 0.85 which is greater than the minimum acceptable value of 0.5.



Figure 8-13 ROC Curve for Validating the Construction Equipment Management Practices and Productivity Model

After validating the model, the probabilities of exceeding PF=0.90 are predicted and the sigmoid graph is plotted (Figure 8-14).



Figure 8-14 Probability Graph of the Construction Equipment Management Practices Validation of the Logistic Regression Model of the Management Practices Related to Construction Methods and Productivity

The logistic regression model of the management practices related to construction methods and productivity is validated by predicting the probabilities of exceeding baseline productivity factor of 0.90 by using the validation dataset coded '11' in Table 8-7 as input to the model's equation (Equation 8-4). After that, the ROC curve was

plotted, and AUC was computed. Figure 8-15 below indicates the ROC curve and the diagonal line which is used as a baseline. Since the graph is above the reference line and AUC=0.77 > 0.5, the model is valid.



Figure 8-15 ROC Curve for Validating Management Practices Related to Construction Methods and Productivity Model

After validation, the model's equation (Equation 8-4) was used to predict the probabilities of exceeding PF=0.90, and the sigmoid graph was plotted as shown in Figure 8-16. In Table 8-32, the data used to draw the probability graph is presented.

 Table 8-32 Predicted Probabilities Using Management Practices Related to Construction Methods and Productivity Model

B_0							-5.791						
В							1.513						
MPCM Score	0	1	2	3	4	4.5	5	5.5	6	6.5	7	7.5	
$B_0+B \times MPCM$	-5.79	-4.28	-2.77	-1.25	0.26	1.02	1.77	2.53	3.29	4.04	4.80	5.56	
P (%)	0.30	1.37	5.92	22.24	56.49	73.45	85.50	92.63	96.40	98.28	99.18	99.62	

Note: B_0 = the constant, B= the coefficient of the variable, MPCM= Management Practices Related to Construction Methods, and P= probability of exceeding PF=0.90.



Figure 8-16 Probability Plot of Management Practices Related to Construction Methods

Validation of the Logistic Regression Model of Pre-Construction Phase Management Practices and Productivity

To validate the logistic regression model of the pre-construction phase management practices and productivity, the procedures used in validating the logistic regression model of other categories of the construction management practices were followed. However, the validation dataset coded '36' in Table 8-9 is used (refer Appendix-5 for the codes of the datasets). In Table 8-33, the results of the comparison of observed and predicted productivity factors are provided. Accordingly, about 80% of the PF's are correctly predicted indicating the strength of the model. The curve of the graph is greater than the reference line which shows that the model is good in prediction (Figure 8-17). Furthermore, the area under the curve (AOC) is computed to be 0.81, and the AOC value is greater than the minimum acceptable value of 0.5. Therefore, the model is valid.



Figure 8-17 ROC Curve for Validating the Pre-Construction Phase Management Practices and Productivity Model

 Table 8-33 Predictive Accuracy of the Pre-Construction Phase Management Practices

 and Productivity Model

Respondents' Code	Actual PF	B_{0}	B_1	PCMP Score	(B ₀ +B ₁ *PCMP) or L	$P = (e^L)/(1 + e^L)$	Predicted PF	Binary Equivalent for Actual PF	Remark
R30	1.12	-5.255	1.100	5.50	0.790	0.688	1	1.00	Correct
R9	1.14			5.46	0.747	0.679	1	1.00	Incorrect
R27	1.08			5.85	1.179	0.765	1	1.00	Correct
R37	1.08			6.72	2.135	0.894	1	1.00	Correct
R29	0.92			5.25	0.525	0.628	0	0.00	Correct
R35	1.06			2.05	-3.003	0.047	0	1.00	Incorrect
R8	1.06			4.94	0.182	0.545	1	1.00	Correct
R31	0.91			4.39	-0.427	0.395	0	0.00	Correct
R19	1.08			3.00	-1.952	0.124	0	1.00	Incorrect
R5	1.01			4.07	-0.775	0.315	1	1.00	Correct
Percentage Cor	rectly pre	edicted							80%

Note: B_0 = the constant, B_1 = the coefficient of the variable, PCMP= Pre-Construction Phase Management Practices Score, and P= probability of exceeding PF=1.0.

By using Equation 8-5, the probabilities of exceeding PF=1.0 are predicted, and the sigmoid graph is plotted as shown in Figure 8-18.



Figure 8-18 Probability Plot of the Preconstruction Phase Management Practices

Validation of the Logistic Regression Model of the Human Resource Management Practices and Productivity

In the process of validation of the logistic regression model of productivity and HRM practices, the probabilities of exceeding PF= 0.90 for the validation dataset coded '11' in Table 8-11were predicted using the model's equation (Equation 8-6); ROC curve was drawn as shown in Figure 8-19 and interpreted with respect to the reference line; and the AUC was computed. Accordingly, the curve is greater than the reference line indicating that the model is good in prediction. Furthermore, the area under the curve is calculated to be 0.67 which is higher than the minimum acceptable value of 0.5. Thus, the model is valid.



Figure 8-19 ROC Curve for Validating HRM Practices and Productivity Model

246

By using Equation 8-6, probabilities were computed, and the sigmoid graph indicated in Figure 8-20 was drawn so that users can easily read probability that corresponds to a score of HRM practices for a particular building project. The portion of the data used in constructing the graph is indicated in Table 8-34.



Table 8-34	Predicted	Probabilities	Using H	RM Pra	ctices Model
1 4010 0-54	1 I culticu	1 100 domines	O sing II.	1111 I I av	chees mouel

B_0				-5.68						
В				1.50						
HRM Score	0	1	2	3	4	5	6	7	7.5	8
$B_0+B\times HRM$	-5.68	-4.17	-2.67	-1.17	0.34	1.84	3.34	4.85	2.99	3.55
P (%)	0.34	1.52	6.48	23.78	58.39	86.33	96.60	99.22	95.23	97.19

Note: B_0 = the constant, B = the coefficient of the variable, HRM= Human Resource Management Practices Score, and P = probability of exceeding PF=0.90.

Validation of the Logistic Regression Model of Safety and Health Practices and Productivity

To validate the logistic regression model of productivity and safety and health practices, the probabilities of exceeding baseline productivity factor of 0.90 were predicted by using the validation dataset coded '11' in Table 8-13 as input to the model, and the AUC of the data was computed. Figure 8-21 shows the ROC and the diagonal line which is used as a baseline. Accordingly, the graph is above the reference line

indicating that the model is valid. Moreover, the AUC= 0.73 > 0.5 which also shows that the model is valid.



Figure 8-21 ROC Curve for Validating the Safety and Health Practices and Productivity Model

The sigmoid graph (Figure 8-22) is plotted by using Equation 8-7, and the data used in constructing the graph is indicated Table 8-35.

Table 8-35 Predicted Probabilities Using Safety and Health Practices and Productivity Model

B_0				-8.79						
B_I				1.81						
SHP Score	0	1	2	3	4	5	6	6.5	7	7.5
$B_0+B_1\times SHP$	-8.79	-6.97	-5.16	-3.35	-1.54	0.28	2.09	3.00	3.90	4.81
P (%)	0.02	1.52	0.57	8.01	17.73	56.91	89.00	95.25	98.02	99.19

Note: B_0 = the constant, B_1 = the coefficient of the variable, SHP= Safety and Health Practice Score, and P = probability of exceeding PF=0.90.



Figure 8-22 Probability Plot of Safety and Health Practices

248

8.3.2 Validation of Linear Regression Models of Construction Management Practices and Productivity

To validate the linear regression model of the overall construction management practices and productivity, the productivity factors of the validation dataset were predicted using the model's equation ($PF = 0.02 \times$ a score of the construction management practices + 0.58), and the mean squared prediction error (MSPR) was computed and compared with the model building data sets' mean squared error (MSE). If the values of MSPR and MSE are similar, then the model is valid (Thomas and Sudhakumar, 2014, Yan and Su, 2009). The following formula is used to compute MSPR (Thomas and Sudhakumar, 2014).

$$MSPR = \frac{\sum_{i=1}^{n} (PF_i - PF_{pi})^2}{n}$$

Where PF_i is the observed productivity factor of an ith validation data point, PF_{pi} is the predicted productivity factor an ith validation data point, and n refers to the number of validation data points.

 Table 8-36 Validation of the Linear Regression Model of Productivity and the Construction

 Management Practices

				Model I	Building D	ata Sets			V	alidation	dataset		
В	Bo	Project	Project	Actual	Predicted		Error	Project	Project	Actual	Predicted	t	Error
(Constant)	(Coefficient)	Code	Score	PF	PF	Error	Squared	Code	Score	PF	PF	Error	Squared
		R33	12.00	0.90	0.82	0.08	0.006	R7	27.53	0.78	1.13	-0.35	0.12
		R18	18.94	1.00	0.96	0.04	0.002	R34	24.17	1.08	1.06	0.02	0.00
		R14	13.52	0.82	0.85	-0.03	0.001	R11	15.23	0.59	0.88	-0.29	0.08
		R12	21.49	0.79	1.01	-0.22	0.048	R32	27.84	1.14	1.14	0.01	0.00
		R39	26.83	1.10	1.12	-0.02	0.000	R25	11.41	0.96	0.81	0.15	0.02
		R26	15.07	0.76	0.88	-0.12	0.014	R24	14.27	0.92	0.87	0.06	0.00
		R38	20.50	1.00	0.99	0.01	0.000	R2	27.42	0.88	1.13	-0.25	0.06
		R30	22.93	1.12	1.04	0.08	0.006	R16	18.59	1.00	0.95	0.05	0.00
		R9	23.34	1.14	1.05	0.09	0.008	R4	24.62	1.02	1.07	-0.06	0.00
		R27	25.64	1.08	1.09	-0.01	0.000	R13	20.01	1.07	0.98	0.09	0.01
		R37	28.55	1.08	1.15	-0.08	0.006	R3	22.57	1.02	1.03	-0.02	0.00
		R29	22.23	0.92	1.02	-0.10	0.010	R23	21.58	0.89	1.01	-0.13	0.02
0.58	0.02	R35	14.31	1.06	0.87	0.19	0.038	R6	22.69	0.82	1.03	-0.22	0.05
0.58	0.02	R8	23.90	1.06	1.06	0.00	0.000		Mean S	Squared I	Predictio	n Error	0.029
		R31	18.92	0.91	0.96	-0.05	0.002						
		R19	16.36	1.08	0.91	0.18	0.032						
		R5	19.47	1.01	0.97	0.04	0.001						
		R10	24.98	1.13	1.08	0.05	0.003						
		R28	8.39	0.60	0.75	-0.15	0.023						
		R20	14.32	0.88	0.87	0.01	0.000						
		R1	16.35	1.00	0.91	0.09	0.009						
		R15	20.09	0.88	0.98	-0.10	0.011						
		R36	22.08	1.00	1.02	-0.02	0.000						
		R17	14.72	0.67	0.87	-0.21	0.044						
		R22	24.25	1.00	1.06	-0.06	0.004						
		R21	19.50	0.96	0.97	-0.01	0.000						
				Me	an Square	d Error	0.0103						

The results of the analysis conducted to validate the linear regression model of the 47 construction management practices and productivity are presented in Table 8-36.

Accordingly, the values of MSE (0.01) and MSPR (0.029) are both close to zero. Thus, the model is valid.

The results of the analyses carried out to validate the linear regression models of the six categories of the construction management practices are presented in the following subsections.

Validation of the Linear Regression Model of the Construction Management Practices and Productivity

The outputs of the analysis conducted to validate the linear regression model of productivity and construction materials management practices are presented in Table 8-37. The finding indicates that MSE (0.01) and MSPR (0.03) are close to each other. Thus, the model is valid.

 Table 8-37 Validation of the Linear Regression Model of Productivity and the Construction

 Materials Management Practices

				Model E	uilding Da	ta Sets				Validation	dataset		
В	Во	Project	Project	Actual	Predicted		Error	Project	Project	Actual	Predicted		Error
(Constant)	(Coefficient)	Code	Score	PF	PF	Error	Squared	Code	Score	PF	PF	Error	Squared
		R3	2.91	1.02	1.00	0.01	0.00	R7	3.49	0.78	1.05	-0.28	0.08
		R23	2.66	0.89	0.98	-0.09	0.01	R34	2.97	1.08	1.01	0.07	0.01
		R6	1.36	0.82	0.86	-0.05	0.00	R11	1.53	0.59	0.88	-0.28	0.08
		R33	1.57	0.90	0.88	0.02	0.00	R32	3.81	1.14	1.08	0.06	0.00
		R18	2.23	1.00	0.94	0.06	0.00	R25	1.08	0.96	0.84	0.12	0.01
		R14	1.97	0.82	0.92	-0.10	0.01	R24	1.18	0.92	0.85	0.08	0.01
		R12	2.49	0.79	0.96	-0.17	0.03	R2	3.77	0.88	1.08	-0.20	0.04
		R39	3.61	1.10	1.06	0.04	0.00	R16	1.79	1.00	0.90	0.10	0.01
		R26	1.32	0.76	0.86	-0.10	0.01	R4	3.14	1.02	1.02	-0.01	0.00
		R38	1.53	1.00	0.88	0.12	0.01	R13	1.84	1.07	0.91	0.17	0.03
		R30	3.91	1.12	1.09	0.03	0.00		Me	an Square	d Prediction	Error	0.03
		R9	2.98	1.14	1.01	0.13	0.02						
		R27	1.92	1.08	0.91	0.17	0.03						
0.74	0.09	R37	3.92	1.08	1.09	-0.02	0.00						
		R29	2.34	0.92	0.95	-0.03	0.00						
		R35	1.68	1.06	0.89	0.17	0.03						
		R8	2.99	1.06	1.01	0.05	0.00						
		R31	2.77	0.91	0.99	-0.08	0.01						
		R19	1.45	1.08	0.87	0.21	0.05						
		R5	2.28	1.01	0.94	0.06	0.00						
		R10	3.97	1.13	1.10	0.04	0.00						
		R28	1.66	0.60	0.89	-0.29	0.09						
		R20	1.41	0.88	0.87	0.01	0.00						
		R1	1.88	1.00	0.91	0.09	0.01						
		R15	2.17	0.88	0.94	-0.06	0.00						
		R36	2.74	1.00	0.99	0.01	0.00						
		R17	2.37	0.67	0.95	-0.29	0.08						
		R22	2.44	1.00	0.96	0.04	0.00						
		R21	2.66	0.96	0.98	-0.02	0.00						
				Me	an Squared	l Error	0.01						

Validation of the Linear Regression Model of the Construction Equipment Management Practices and Productivity

As shown in Figure 8-38, the MSE (0.01) and MSPR (0.02) are similar. Thus, the linear regression model of productivity and construction equipment management practices is valid.

			I I	Model I	Building Da	ta Sets				Validati	on dataset		
В	Во	Project	Project	Actual	Predicted		Error	Project	Project	Actual	Predicted		Error
(Constant)	(Coefficient)	Code	Score	PF	PF	Error	Squared	Code	Score	PF	PF	Error	Squared
		R33	0.42	0.90	0.89	0.00	0.00	R7	1.67	0.78	1.06	-0.28	0.08
		R18	0.00	1.00	0.84	0.16	0.03	R34	1.08	1.08	0.98	0.10	0.01
		R14	0.44	0.82	0.90	-0.08	0.01	R11	0.58	0.59	0.91	-0.32	0.10
		R12	1.41	0.79	1.02	-0.23	0.05	R32	1.69	1.14	1.06	0.08	0.01
		R39	1.15	1.10	0.99	0.11	0.01	R25	0.68	0.96	0.93	0.03	0.00
		R26	0.00	0.76	0.84	-0.08	0.01	R24	0.58	0.92	0.91	0.01	0.00
		R38	0.70	1.00	0.93	0.07	0.00	R2	1.35	0.88	1.02	-0.14	0.02
		R30	1.49	1.12	1.03	0.08	0.01	R16	0.59	1.00	0.92	0.08	0.01
		R9	1.23	1.14	1.00	0.14	0.02	R4	1.08	1.02	0.98	0.04	0.00
		R27	1.55	1.08	1.04	0.04	0.00	R13	0.97	1.07	0.97	0.11	0.01
		R37	2.09	1.08	1.11	-0.04	0.00	R3	0.83	1.02	0.95	0.07	0.00
		R29	1.47	0.92	1.03	-0.11	0.01	R23	1.37	0.89	1.02	-0.13	0.02
0.84	0.12	R35	0.59	1.06	0.92	0.14	0.02	R6	0.89	0.82	0.96	-0.14	0.02
0.64	0.15	R8	1.11	1.06	0.98	0.08	0.01	Mean	Squared 1	Predictio	on Error Er	ror	0.02
		R31	0.96	0.91	0.96	-0.05	0.00						
		R19	0.56	1.08	0.91	0.17	0.03						
		R5	0.58	1.01	0.91	0.09	0.01						
		R10	1.23	1.13	1.00	0.13	0.02						
		R28	0.16	0.60	0.86	-0.26	0.07						
		R20	0.58	0.88	0.91	-0.03	0.00						
		R1	0.58	1.00	0.91	0.09	0.01						
		R15	0.71	0.88	0.93	-0.05	0.00						
		R36	1.39	1.00	1.02	-0.02	0.00						
		R17	0.72	0.67	0.93	-0.27	0.07						
		R22	1.39	1.00	1.02	-0.02	0.00						
		R21	0.58	0.96	0.91	0.05	0.00						
				Me	an Squared	Error	0.01						

 Table 8-38 Validation of the Linear Regression Model of Productivity and the

 Construction Equipment Management Practices

Validation of the Linear Regression Model of the Management Practices Related to Construction Methods and Productivity

The MSPR (0.03) and MSE (0.02) computed using the linear regression equation of the management practices related to construction methods and productivity are close to each other indicating that the model is valid (Table 8-39).

				Model I	Building Da	ta Sets			V	alidation	dataset		
В	Bo	Project	Project	Actual	Predicted		Error	Project	Project	Actual	Predicted		Error
(Constant)	(Coefficient)	Code	Score	PF	PF	Error	Squared	Code	Score	PF	PF	Error	Squared
		R33	2.55	0.90	0.81	0.08	0.01	R7	6.78	0.78	1.15	-0.37	0.14
		R18	4.98	1.00	1.01	-0.01	0.00	R34	6.32	1.08	1.12	-0.04	0.00
		R14	3.06	0.82	0.85	-0.04	0.00	R11	3.76	0.59	0.91	-0.32	0.10
		R12	4.91	0.79	1.00	-0.21	0.04	R32	6.82	1.14	1.16	-0.01	0.00
		R39	6.46	1.10	1.13	-0.03	0.00	R25	2.19	0.96	0.79	0.17	0.03
		R26	3.78	0.76	0.91	-0.15	0.02	R24	3.96	0.92	0.93	-0.01	0.00
		R38	5.58	1.00	1.06	-0.06	0.00	R2	6.80	0.88	1.15	-0.28	0.08
		R30	4.96	1.12	1.01	0.11	0.01	R16	5.42	1.00	1.04	-0.04	0.00
		R9	4.93	1.14	1.00	0.13	0.02	R4	5.40	1.02	1.04	-0.03	0.00
		R27	5.84	1.08	1.08	0.01	0.00	R13	4.15	1.07	0.94	0.13	0.02
		R37	6.24	1.08	1.11	-0.03	0.00	R3	6.32	1.02	1.12	-0.10	0.01
		R29	4.82	0.92	1.00	-0.07	0.01	R23	4.36	0.89	0.96	-0.07	0.01
0.61	0.08	R35	3.15	1.06	0.86	0.20	0.04	R6	5.69	0.82	1.06	-0.25	0.06
0.01	0.08	R8	6.30	1.06	1.11	-0.05	0.00		Mean	n Squared	Prediction	Error	0.03
		R31	3.88	0.91	0.92	-0.01	0.00						
		R19	3.62	1.08	0.90	0.19	0.03						
		R5	5.77	1.01	1.07	-0.07	0.00						
		R10	5.79	1.13	1.07	0.06	0.00						
		R28	2.14	0.60	0.78	-0.19	0.03						
		R20	4.07	0.88	0.94	-0.06	0.00						
		R1	4.37	1.00	0.96	0.04	0.00						
		R15	4.77	0.88	0.99	-0.11	0.01						
		R36	4.96	1.00	1.01	-0.01	0.00						
		R17	3.60	0.67	0.90	-0.23	0.05						
		R22	5.97	1.00	1.09	-0.09	0.01						
		R21	4.20	0.96	0.95	0.02	0.00						
				Me	an Squared	l Error	0.01						

 Table 8-39 Validation of the Linear Regression Model of Productivity and the Management Practices Related to Construction Methods

251

Validation of the Linear Regression Model of the Pre-Construction Phase Management Practices and Productivity

As shown in Table 8-40, the values of MSE (0.01) and MSPR (0.02) are very close to each other indicating that the pre-construction phase management practices and productivity model is valid.

 Table 8-40 Validation of the Linear Regression Model of Productivity and the Pre-Construction Phase Management Practices

				Model H	Building Da	ta Sets			,	Validation	dataset		
В	Во	Project	Project	Actual	Predicted		Error	Project	Project	Actual	Predicted		Error
(Constant)	(Coefficient)	Code	Score	PF	PF	Error	Squared	Code	Score	PF	PF	Error	Squared
		R33	1.79	0.90	0.82	0.08	0.01	R7	5.69	0.78	1.05	-0.28	0.08
		R18	4.52	1.00	0.98	0.02	0.00	R34	6.01	1.08	1.07	0.01	0.00
		R14	2.61	0.82	0.87	-0.05	0.00	R11	3.65	0.59	0.93	-0.34	0.11
		R12	4.50	0.79	0.98	-0.19	0.04	R32	7.11	1.14	1.14	0.01	0.00
		R39	6.53	1.10	1.10	0.00	0.00	R25	2.80	0.96	0.88	0.08	0.01
		R26	3.99	0.76	0.95	-0.19	0.04	R24	3.85	0.92	0.94	-0.02	0.00
		R38	5.27	1.00	1.03	-0.03	0.00	R2	6.10	0.88	1.08	-0.20	0.04
		R30	5.50	1.12	1.04	0.08	0.01	R16	5.15	1.00	1.02	-0.02	0.00
		R9	5.46	1.14	1.04	0.10	0.01	R4	5.36	1.02	1.03	-0.02	0.00
		R27	5.85	1.08	1.06	0.02	0.00	R13	4.40	1.07	0.97	0.10	0.01
		R37	6.72	1.08	1.11	-0.04	0.00	R3	4.90	1.02	1.00	0.01	0.00
		R29	5.25	0.92	1.03	-0.10	0.01	R23	4.61	0.89	0.99	-0.10	0.01
0.711	0.06	R35	2.05	1.06	0.83	0.23	0.05	R6	5.68	0.82	1.05	-0.24	0.06
0.711	0.00	R8	4.94	1.06	1.01	0.05	0.00		Mea	n Squared	d Prediction	n Error	0.02
		R31	4.39	0.91	0.97	-0.06	0.00						
		R19	3.00	1.08	0.89	0.19	0.04						
		R5	4.07	1.01	0.96	0.05	0.00						
		R10	5.31	1.13	1.03	0.10	0.01						
		R28	1.19	0.60	0.78	-0.19	0.03						
		R20	2.77	0.88	0.88	0.00	0.00						
		R1	4.18	1.00	0.96	0.04	0.00						
		R15	5.03	0.88	1.01	-0.13	0.02						
		R36	4.09	1.00	0.96	0.04	0.00						
		R17	3.30	0.67	0.91	-0.24	0.06						
		R22	5.33	1.00	1.03	-0.03	0.00						
		R21	3.46	0.96	0.92	0.04	0.00						
				Me	an Squared	l Error	0.01						

Validation of the Linear Regression Model of the Human Resource Management Practices and Productivity

The HRM practices and productivity linear regression model is valid since the values of MSE(0.02) and MSPR(0.03) are similar (Table 8-41).

Validation of the Linear Regression Model of the Safety and Health Practices and Productivity

Table 8-42 presents the results of the analysis conducted to validate the linear regression model of the safety and health practices and productivity. Accordingly, the *MSPR* value of 0.02 and *MSE* value of 0.01 are found. The findings show that the model is valid.

	1000		Tunuge	Model F	Building D			I	/alidation	dataset			
	Bo	Project	Project	Actual	Predicted	1	Error	Project	Project	Actual	Predicted		Error
(Constant)	(Coefficient)	Code	Score	PF	PF	Error	Squared	Code	Score	PF	PF	Error	Squared
(Constant)	(coefficient)	R33	2.64	0.90	0.82	0.07	0.01	R7	6 57	0.78	1 10	-0.32	0.10
		R18	4 30	1.00	0.94	0.06	0.00	R34	4 54	1.08	0.96	0.12	0.01
		R14	4 23	0.82	0.94	-0.12	0.00	R11	3 52	0.59	0.89	-0.29	0.09
		R12	5.06	0.79	0.99	-0.20	0.04	R32	5.14	1.14	1.00	0.14	0.02
		R39	5.74	1.10	1.04	0.06	0.00	R25	2.04	0.96	0.78	0.17	0.03
		R26	3.88	0.76	0.91	-0.15	0.02	R24	2.48	0.92	0.81	0.11	0.01
		R38	4.89	1.00	0.98	0.02	0.00	R2	6.52	0.88	1.10	-0.22	0.05
		R30	3.81	1.12	0.91	0.21	0.04	R16	2.59	1.00	0.82	0.18	0.03
		R9	5.45	1.14	1.02	0.12	0.01	R4	6.96	1.02	1.13	-0.11	0.01
		R27	7.19	1.08	1.14	-0.06	0.00	R13	5.36	1.07	1.02	0.06	0.00
		R37	6.27	1.08	1.08	0.00	0.00	R3	4.47	1.02	0.95	0.06	0.00
		R29	5.03	0.92	0.99	-0.07	0.00	R23	5.27	0.89	1.01	-0.12	0.01
0.64	0.07	R35	4.14	1.06	0.93	0.13	0.02	R6	5.73	0.82	1.04	-0.23	0.05
0.04	0.07	R8	5.50	1.06	1.03	0.03	0.00		Mea	n Squared	Prediction	n Error	0.03
		R31	3.97	0.91	0.92	-0.01	0.00						
		R19	4.49	1.08	0.95	0.13	0.02						
		R5	4.67	1.01	0.97	0.04	0.00						
		R10	6.13	1.13	1.07	0.06	0.00						
		R28	0.92	0.60	0.70	-0.11	0.01						
		R20	3.39	0.88	0.88	0.00	0.00						
		R1	3.30	1.00	0.87	0.13	0.02						
		R15	4.50	0.88	0.95	-0.08	0.01						
		R36	5.59	1.00	1.03	-0.03	0.00						
		R17	2.05	0.67	0.78	-0.12	0.01						
		R22	6.74	1.00	1.11	-0.11	0.01						
		R21	5.32	0.96	1.01	-0.05	0.00						
				0.01									

 Table 8-41 Validation of the Linear Regression Model of Productivity and the Human Resource Management Practices

Table 8-42 Validation of the Linear Regression Model of Productivity and the Safety and Health Practices

				Model E	Building Da	ta Sets			V	Validation	dataset		
В	Во	Project	Project	Actual	Predicted		Error	Project	Project	Actual	Predicted		Error
(Constant)	(Coefficient)	Code	Score	PF	PF	Error	Squared	Code	Score	PF	PF	Error	Squared
		R33	5.48	0.90	0.98	-0.08	0.01	R7	6.39	0.78	1.07	-0.29	0.08
		R18	5.65	1.00	1.00	0.00	0.00	R34	5.85	1.08	1.02	0.06	0.00
		R14	2.73	0.82	0.74	0.08	0.01	R11	4.01	0.59	0.85	-0.26	0.07
		R12	6.02	0.79	1.03	-0.24	0.06	R32	6.03	1.14	1.03	0.11	0.01
		R39	6.39	1.10	1.07	0.03	0.00	R25	4.76	0.96	0.92	0.04	0.00
		R26	4.39	0.76	0.89	-0.12	0.02	R24	4.03	0.92	0.85	0.07	0.00
		R38	5.12	1.00	0.95	0.05	0.00	R2	5.31	0.88	0.97	-0.09	0.01
		R30	5.85	1.12	1.02	0.10	0.01	R16	5.49	1.00	0.98	0.02	0.00
		R9	6.03	1.14	1.03	0.11	0.01	R4	5.29	1.02	0.97	0.05	0.00
		R27	6.03	1.08	1.03	0.05	0.00	R13	6.03	1.07	1.03	0.04	0.00
		R37	6.21	1.08	1.05	0.03	0.00	R3	6.03	1.02	1.03	-0.02	0.00
		R29	6.21	0.92	1.05	-0.12	0.02	R23	6.21	0.89	1.05	-0.16	0.03
0.40	0.00	R35	5.29	1.06	0.97	0.09	0.01	R6	6.39	0.82	1.07	-0.25	0.06
0.49	0.09	R8	5.49	1.06	0.98	0.08	0.01		Mea	n Squared	Prediction	n Error	0.02
		R31	5.86	0.91	1.02	-0.11	0.01						
		R19	5.68	1.08	1.00	0.08	0.01						
		R5	4.55	1.01	0.90	0.11	0.01						
		R10	5.29	1.13	0.97	0.17	0.03						
		R28	4.00	0.60	0.85	-0.25	0.06						
		R20	4.39	0.88	0.88	0.00	0.00						
		R1	4.94	1.00	0.93	0.07	0.00						
		R15	5.66	0.88	1.00	-0.12	0.01						
		R36	6.21	1.00	1.05	-0.05	0.00						
		R17	5.12	0.67	0.95	-0.29	0.08						
		R22	5.29	1.00	0.97	0.03	0.00						
		R21	6.03	0.96	1.03	-0.07	0.00						
				Me	an Squared	Error	0.01						

8.4 Summary

In this chapter, the quantitative data has been analysed and Objective 3 of the study or 'to build and validate a tool that can be used to predict the probability of exceeding baseline productivity factor based on a score of the management practices' is achieved. Logistic regression model of the overall or the 47 construction management practices and productivity was developed and validated. Moreover, logistic regression models of productivity and the six categories of the construction management practices were built and validated. The reliability tests for the logistic regression models were conducted and acceptable results were found.

Linear regression model of productivity and the overall construction management practices was also developed and validated. The finding shows that about 48% of the variance in the productivity factor can be explained by the construction management practices and the remaining percentage could be explained by changes in technology and government regulations among others. The linear regression models of the six categories of the construction management practices and productivity have been built and validated. Thus, productivity factor of a project can be estimated based on the scores of the construction materials management practices, construction equipment management practices, management practices related to construction methods, preconstruction phase management practices, human resource management practices and safety and health practices. The next chapter provides the discussion of the findings of the study.

CHAPTER NINE

9 DISCUSSION OF FINDINGS

9.1 Introduction

In this chapter, the findings of the study are discussed. Section 9.2 presents the discussions on the relative importance of the construction management practices. Section 9.3 explains the relationships among the construction management practices. In section 9.4, the uses of the scoring tools and the regression models are demonstrated. The applications of the logistic regression models, the linear regression models, the scoring tools during the project planning, construction and closure phases are described in Section 9.4. Finally, the chapter is summarised in Section 9.5.

9.2 Relative Importance of the Construction Management Practices

In this section, the discussions of the findings obtained to achieve Objective 1 of the study or 'to identify and prioritise the management practices that have the potential to enhance productivity in multi-storey building construction projects in the context of Victoria State, Australia' are presented.

Relative Importance of the Overall Construction Management Practices

Overall, 47 management practices that could improve productivity in multi-storey building construction projects are identified. These include six construction materials management practices, three construction equipment management practices, 10 management practices related to construction methods, 10 preconstruction phase management practices, 11 human resource management practices, and seven safety and health practices. The findings of this study have been compared with previous similar studies conducted in the context of infrastructure and industrial projects in North America (Table 9-1).

As shown in Table 9-1, the construction equipment management practices are found to be less important to improve productivity in multi-storey building projects as compared to industrial and infrastructure projects. For instance, in the context of multi-storey building projects, the practice 'construction equipment productivity analysis is ranked 44th among the 47 practices. However, in industrial projects, the practice is ranked 5th

among 53 practices, and in infrastructure projects, the practice is ranked 6th among 61 practices.

Table 9-1 Relative Im	portance of the Overall	Construction Manag	gement Practices

Puildability Projects (This Stud	dw)	Industrial Designets (CII 2012a)		Infinationation Divisets (CII 2012b)	
Buildability Projects (This Stue	dy) Durch	Industrial Projects (CII, 2013a)	Paula	Intrastructure Projects (CII, 20130)	Paul
Practices	капк	Practices	Kank	Practices	капк
Well-Defined Scope of Work	1	Equipment Maintenance	1	Formal Health and Safety Policy	1
Safety and Health Policy	2	Substance Abuse Programs	2	Health and Safety Plans/Zero Accident Techniques	1
Safety and Health Plan	3	Toolbox Safety Meeting	3	Innovations & New Technologies	3
Hazards Analysis	3	Short Interval Planning	4	Hazards Planning	3
Long-Lead Materials Identification	5	Construction Machinery Productivity Analysis	5	Health and Safety Training Programs	3
Safe Work Method Statement	6	Design Readiness for Construction	5	Construction Machinery Productivity Analyses	6
Toolbox Safety Meetings	7	Material Inspection Process	7	Construction Machinery and Equipment Maintenance	6
House barery meetings	,	OSUA Compliance Training	,	Useranda Australia	0
Housekeeping	8	OSHA Compliance Training	8	Hazards Analysis	8
Safety and Health Training	8	Material Delivery Schedule	9	Task Safety Analysis	9
Short Interval Plan	10	Procurement Plan for Materials and Equipment	10	Crews Composition/Crew Formation	10
Buildability Review	10	Post receipt Preservation and Maintenance	11	Dynamic site layout plan	11
Construction Work Packages	12	Material Inspection Team	12	Environmental Requirements	12
Traffic Control Plan	13	Trades Technical Training	13	Skills Assessment and Evaluation	13
Project Start -un Plan	13	Integrated Schedule	13	Communications Coordination & Agreements	13
	10		15		15
Regulatory Requirement	13	Retention Plan for Experienced Personnel	15	Long-Lead/Critical Equipment & Materials Identification	15
Procurement Plans for Materials	16	Schedule Execution and Management	15	Tools & Equipment Tracking	15
Clear Delegation of Responsibility	17	On-site Material Tracking Technology	17	Employees / Trades Technical Training	15
Project Completion Plan	18	PPMOF Evaluation	18	Social Activities	18
Machinery Positioning Strategy	19	Site Tool and Consumables Management Strategy	19	Toolbox Safety Meetings	19
Duramic Site Lavout Plan	20	Dunomic Site Lavout Plan	19	Pagulatory Requirements/Permitting Requirements	20
Dynamic Sic Layour Fair	20		10	Regulatory Requirements/1 errinting Requirements	20
Stability of Organizational Structure	20	Zero Accident Techniques	19	Housekeeping	21
Construction Equipment Maintenance	22	Project Team Material Status Database	22	Design readiness for construction	22
Contract Types	23	Clear Delegation of Responsibility	22	Financial Incentive Programs	23
Collinact Types	23		22	F Halical incentive i logians	23
Schedule Control	24	Planning for Start-up	22	Utility Alignment & Adjustments	24
Crew Composition	24	Equipment Positioning Strategy	22	Contract Types/Strategies	25
Use of Software in Planning Work	26	Teck Safaty Analysis		Procurement Procedures & Plans for Construction	
Packages	20	Task Salety Analysis	22	Machinery	26
Retention Plan for Experienced					
Perconnel	26	System Turnover Procedure	27	Dedicated Planner	26
	20	11 Continue - Detential Magazita	27	C' T L and Family Monogrammant Stratage	20
Utilities Alignment	28	Identification of Potential Hazards	27	Site Tools and Equipment Management Strategy	28
Materials Delivery Schedule	29	Career Development		Procurement Procedures and Plans for Materials and	
Water and Denvery Sene and	27	Calcol Development	29	Equipment	29
Employees Training	29	Maintain Stability of Organizational Structure	30	Career development	29
Skill Assessment and Evaluation	31	Housekeeping	31	Integrated Schedule	29
Work Schedule Strategies	32	Work Schedule Strategies	32	Evit Interview	32
Cite Committe Dian	22	To al Tra alvina Constanta	22	N - a francial la - antica Dra - mana	22
Site Security Plan	32	Tool Tracking Systems	33	Nonfinancial Incentive Programs	33
Career Development	34	Site Security Plan	33	Retention Plan for Experienced Personnel	34
Material Inspection Process	35	On-site Tool Maintenance	35	Project Team Materials Status Database	35
Procurement Plan for Construction	26	E II C D			
Equipment	36	Financial Incentive Programs	35	well defined scope of work	35
Financial Incentive Programs	37	Testing Procedures	35	Project Completion Plan	35
	27	Central Sectors for Teel Delean	20	Traffe Centrel Dien	20
Social Activities	3/	Control System for Tool Delays	38	Iraffic Control Plan	38
Materials Status Database	39	Nonfinancial Incentive Programs	39	Utility Agreements	39
Integrated Schedules	40	System Test Hazards Planning	39	Short Interval Planning	40
Non-Financial Incentive Programs	41	Exit Interviews	41	Clear Delegation of Responsibility	41
Innovations and New Technologies	42	New Equipment Investigation	42	Materials Delivery Schedule	42
Dedicated Planner	43	Well Defined Scope of Work	43	Procurement Team	43
Construction Equipment Productivity		nen bennen scope of none	1.5		.5
Analysis	44	New Materials Technologies Investigation	44	Machinery and Equipment positioning strategy	43
Exit Interviews	45	Social Activities	45	Construction Work Packages (CWP)	45
Materials Inspection Team	46	New Information System Investigation	45	Project start-un plan	45
Madel Development	40	Dediasted Diaman	43	Contracto & Americante mile America	43
Model Development	4/	Dedicated Planner	4/	Contracts & Agreements with Agencies	4/
		Construction Work Packages (CWP)	47	Drugs and Alcohol Testing Program	47
		Installation Work Packages (IWP)	47	Work Schedule Strategies	49
		Engineering Work Packages (EWP)	50	Materials Inspection Process	50
		Identify Required Permitting	51	On-site Tools Maintenance	50
				Construction Machinery & Equipment Utility	
		Project Model Requirements	52	Requirements	50
		Utilization of Software to Assist in Generating Work	52	Use of Software	50
		packages	53		50
				Right of Way, Land, and Utilities Acquisition Strategy	50
				On-Site Material Tracking Technology	55
				Model Requirements/3D Visualization	55
				Site security plan	55
				Maintain Stability of Organization Structure	58
				Materiak Inspection Team	50
					50
				Schedule Execution and Wranagement	39
				Post Receipt Preservation & Maintenance	61

Similarly, 'construction equipment maintenance' is found to be one of the most important practices that could enhance productivity in industrial and infrastructure projects. Nonetheless, the practice is considered to be less important for enhancing productivity in the context of multi-storey building construction projects. The finding implies that since industrial and infrastructure construction projects (engineering projects) are equipment intensive as compared to multi-storey building construction projects, the implementation of the construction equipment management practices is more important for the engineering projects than for the building projects.

Relative Importance of the Construction Materials Management Practices

After analysing the qualitative data (Phase I), six construction materials management practices that could improve productivity in multi-storey building projects in the context of Victoria, Australia are identified. These are Procurement Plans for Materials, Long-Lead Materials Identification, Materials Status Database, Materials Delivery Schedule, Material Inspection Process, and Materials Inspection Team. The quantitative data analysis (Phase II) also revealed that all the six practices are positively correlated with productivity. Comparisons of the findings of this study with previous similar studies are made in Table 9-2.

As indicated in Table 9-2, all the construction materials management practices that have the potential to improve productivity in infrastructure and industrial construction projects are not suitable to enhance productivity in multi-storey building projects. Moreover, the relative importance of the practices is different.

		This State			
Intrastructure Projects (CII, 2013b)	Industrial Projects (CII, 2013a)	This Study			
Construction Materials Management Practices	Rank	Construction Materials Management Practices	Rank	Construction Materials Management Practices	Rank
Long-Lead/Critical Equipment and Materials Identification	1	Material Inspection Process	1	Long-Lead Materials Identification	1
Procurement Procedures and Plans for Materials and Equipment	2	Material Delivery Schedule	2	Procurement Plans for Materials	2
Project Team Materials Status Database	3	Procurement Plan for Materials and Equipment	3	Materials Delivery Schedule	3
Materials Delivery Schedule	4	Post-receipt Preservation and Maintenance	4	Material Inspection Process	4
Procurement Team	5	Materials Inspection Team	5	Materials Status Database	5
Materials Inspection Process	6	On-Site Material Tracking Technology	6	Materials Inspection Team	6
On-Site Material Tracking Technology	7	Project Team Materials Status Database	7		
Materials Inspection Team	8				
Post Receipt Preservation and Maintenance	9				

 Table 9-2 The Relative Importance of Materials Management Practices in Infrastructure, Industrial and Multi-Storey Building Construction Projects

While the practice 'material inspection process' is given top priority for improving productivity in industrial projects in North America, the practice 'long-lead materials

identification' is found to be the most important materials management practice that could enhance productivity in multi-storey building projects in Victoria. This could be due to the materials supply chain systems in Australia in which most construction materials are imported from abroad. Wheeldon (2012) reported that most construction materials in Australia are imported from overseas due to increasing investment costs in the manufacturing sector. The findings of the qualitative data analysis revealed that tiles and façade items such as glass and aluminium are some of the most commonly imported materials because of the rising cost of these materials locally. According to Australian Bureau of Statistics, the manufacturing prices of architectural aluminium, paint and coatings, ceramic products, and glass products increases at the rates of 2.56%, 0.88%, 1.14%, and 0.71% respectively (Australian Bureau of Statistics, 2017). Thus, if the critical (long-lead) materials are not identified early, and their procurement and delivery schedules are not prepared accordingly, the project could be delayed. Therefore, contractors in Victoria State, Australia provide the highest weight for the practice 'identification of long-lead materials' to reduce the associated project delays.

The practices 'on-site material tracking technology' and 'post-receipt preservation and maintenance' are found to be not suitable for improving productivity in multi-storey building construction projects in the context of Victoria State, Australia. The findings of Phase I of the study show that most multi-storey building projects in Victoria are located in cities where there are shortages of storage spaces, and most construction materials are not stored on the building construction sites. The interview results revealed that materials are brought to the construction sites when they are needed, and they are placed close to their installation area. If more materials are stored on the multi-storey building project sites, the sites could be congested, and the potential hazards increase. The occurrence of hazards, in turn, increases the possibility of accidents which could reduce the productivity of the projects. Therefore, most contractors do not prefer to store materials on-sites. Hence, 'on-site material tracking technology' and 'post receipt preservation and maintenance' are found to be not essential practices for enhancing productivity in multi-storey building projects.

By using the construction materials management practices indicated in Table 9-2 column-3, the framework which can assist contractors' project managers to improve the productivity of the multi-storey building projects is developed (Figure 9-1). The

framework relates the factors which negatively influence productivity and the construction materials management practices. For instance, the shortage of construction materials has been identified as one of the critical factors that can reduce the productivity of the construction projects (Rivas et al., 2011, Jarkas and Bitar, 2012, El-Gohary and Aziz, 2014, Hughes and Thorpe, 2014). However, before the commencement of the projects, identification of the long-lead materials and preparation of the materials procurement plan; and during the construction phase, monitoring the status of the materials could assist the project managers to reduce the loss of productivity due to the shortage of materials (Figure 9-1). Moreover, delay in the delivery of construction materials is also frequently mentioned as one of the critical factors which negatively influence productivity in construction projects (Lim and Alum, 1995, Kaming et al., 1997, Thomas and Sanvido, 2000, Ghoddousi and Hosseini, 2012). On the other hand, listing all the materials required for a building project and preparing a master delivery schedule for all materials; identifying the long lead materials; and preparing separate critical materials' procurement and delivery schedules which are integrated with work schedule can help contractors to reduce the loss of productivity due to the delay in the delivery of materials (Figure 9-1).



Figure 9-1 Framework of Factors Negatively Influencing Productivity and Construction Materials Management Practices

Relative Importance of the Construction Equipment Management Practices

After analysing the qualitative data, it is found that the practices construction equipment procurement plan, construction equipment productivity analysis and construction equipment maintenance are found to be the three construction equipment management practices that could improve productivity in multi-storey building projects. The results of quantitative data analysis also confirmed that the three equipment management

practices are positively associated with productivity, and the relationships are statistically significant.

The findings of this study confirmed that not all construction equipment and tools management that are identified for enhancing productivity in infrastructure and industrial projects are suitable to improve productivity in multi-storey building projects. Moreover, the priorities of the practices are different. The comparisons among the construction equipment management practices for the three project types are shown in Table 9-3.

 Table 9-3 Relative Importance of the Construction Equipment Management Practices in Industrial, Infrastructure, and Multi-Storey Building Construction Projects

Infrastructure Projects (CII, 2013b)		Industrial Projects (CII, 2013a)		This Study	
Construction Equipment and Tools Management Practices	Rank	Construction Equipment and Tools Management Practices	Rank	Construction Equipment Management Practices	Rank
Equipment Maintenance	1	Construction Machinery Productivity Analyses	1	Construction Equipment Maintenance	1
Construction Machinery Productivity Analysis	2	Construction Machinery and Equipment Maintenance	1	Construction Equipment Procurement plan	2
Site Tool and Consumables Management	3	Tools and Equipment Tracking	3	Construction equipment productivity analysis	3
Tool Tracking Systems	4	Procurement Procedures and Plans for Construction	4		
On-site Tool Maintenance	5	Site Tools and Equipment Management Strategy	5		
Control System for Tool Delays	6	Construction Machinery and Equipment Utility Requirements	6		
		On-site Tools Maintenance	6		

The practices 'site tools management strategy,' 'tools tracking systems' and 'on-site tools maintenance' are the three practices that are common to infrastructure and industrial projects (Caldas et al., 2014, Nasir et al., 2015). However, these tools management systems are found to be not suitable for enhancing productivity in multi-storey building projects. The findings of the qualitative data analysis revealed that most tools and construction materials are not stored on sites due to the shortage of the storage areas. The interviewees explained that most contractors prefer to include the cost of tools in the labour cost instead of purchasing the tools by themselves. The tradespersons bring the tools to the project sites when required. Hence, the number of tools stored on sites is minimal.

Construction equipment maintenance is found to be the most important practice that has the potential to enhance productivity in multi-storey building projects. This could be due to the presence of the local regulations which oblige contractors to carry out construction equipment inspection and maintenance. For instance, according to Clause 253 of the Work Safety and Health Regulations 2011, any person conducting a business or undertaking involving the management or control of equipment must ensure that maintenance, inspection, and testing of the equipment is carried out by a competent person. Failure to perform such tasks is an offence against the Regulation and result in a 260

financial penalty of AUD18,000 (Work Health and Safety Regulations, 2011). Construction works could also be suspended due to the contravention of the Regulation.

In the category of construction equipment management practices, 'construction equipment procurement plan' and 'construction equipment productivity analysis' are ranked second and third respectively. The finding of the study suggests that contractors involved in multi-storey building construction in Victoria, Australia focus on selecting and hiring of the major construction equipment such as cranes and hoists to reduce the loss of productivity due to unavailability of the equipment. When the demand for construction equipment is greater than its supply, there could be shortage the machinery in the local building industry. Consequently, productivity could be negatively influenced.

Based on the findings of this study, a framework which integrates the construction equipment management practices and factors negatively influencing productivity in construction projects is developed (Figure 9-2).



Management Practices

In Australian construction industry, Hughes and Thorpe (2014) found that the lack of equipment is one of the construction projects' productivity impediments. In international contexts shortage of equipment is also mentioned as productivity deterrent (Durdyev and Mbachu, 2011, Rivas et al., 2011, Ghoddousi and Hosseini, 2012). However, the implementation of the practice 'construction equipment procurement plan' prior to starting the construction project could mitigate the loss of productivity due to the lack of construction equipment (Figure 9-2). Construction equipment procurement planning could help contractors to ensure the availability of the right

equipment such as hoists with suitable speed and cranes having appropriate capacity to lift the heaviest materials on a particular site. The availability of the right equipment, in turn, facilitates the transportation of construction materials to the desired working levels; consequently, the labour productivity could be enhanced. Thus, the development of the construction equipment acquisition strategy and preparation of the equipment usage schedule on a daily basis could reduce the loss of productivity due to unavailability of the construction equipment. The practice of analysing the productivity of the construction equipment before its procurement is also essential. For instance, the procurement of hoists having low-speed and cranes with low lifting capacity could negatively influence the labour productivity. If the machinery is slow, it takes a longer time to transport the materials from ground to the required working level, and productivity can be reduced due to unavailability of the construction materials at the specific working area.

The breakdown of the construction equipment is another major factor which reduces the productivity of construction projects (Liberda et al., 2003, Makulsawatudom et al., 2004, Dai et al., 2009a, Hughes and Thorpe, 2014). However, the implementation of the practice 'construction equipment maintenance' can reduce the adverse effects of the problem (Figure 9-2). Preparing the maintenance schedule for equipment used in the construction of multi-storey buildings and updating the maintenance records regularly and electronically could help to reduce the breakage of construction machinery. Thus, the loss of productivity due to the breakdown of the construction equipment can be minimised.

Relative Importance of the Management Practices Related to Construction Methods

The analysis of the qualitative data revealed 10 management practices related to construction methods. These include 'integrated schedule,' 'work schedule strategies,' 'schedule control,' 'dynamic site layout plan,' 'traffic control plan,' 'site security plan,' 'machinery positioning strategy,' 'project start-up plan,' 'project completion plan,' and 'innovations and new technologies.' However, the practice 'innovations and new technologies' is weakly correlated with productivity. This finding is similar to the conclusion made by Rojas and Aramvareekul (2003) in which the authors stated that introduction of new technologies may be a necessary, but not a sufficient, condition to improve productivity in construction projects.

The quantitative data analysis indicated that 'traffic control plan,' 'project start-up project plan,' 'machinery positioning strategy,' 'project completion plan' and 'dynamic site layout plan' are among the five most important practices that have the potential to improve productivity in multi-storey building projects (Table 9-4 column-2). Moreover, no discernible differences among these five practices are found which implies that they should be implemented jointly to improve productivity in multi-storey building projects. For instance, the optimum location of construction machinery might not be obtained without considering the surrounding traffic and the layout of a project. Similarly, good project completion plan might not be prepared without developing clear project start-up plan.

The findings of this study confirmed that the practices that are given top priority in other project types are considered as less important to enhance productivity in multistorey building construction projects. The comparison of the practices in industrial, infrastructure and multi-storey building construction projects is presented in Table 9-4.

Management Practices Delated to	Ranks of the Practices						
Construction Mothods	Multi-Storey Building	Infrastructure Projects	Industrial Projects				
Construction Methods	Projects (this study)	(CII, 2013b)	(CII, 2013a)				
Traffic Control Plan	1	7	Not included				
Project start-up plan	1	9	4				
Machinery Positioning Strategy	3	8	4				
Project Completion Plan	4	6	Not included				
Dynamic Site Layout Plan	5	2	3				
Schedule control	6	12	2				
Work Schedule Strategies	7	10	7				
Site Security Plan	7	11	8				
Integrated Schedule	9	5	1				
Innovations and New Technologies	10	1	10				

 Table 9-4 Comparison of the Management Practices Related to Construction Methods in Industrial, Infrastructure, and Multi-Storey Building Construction Projects

As shown in Table 9-4, the practice 'traffic control plan' is found to be the most important practice that could enhance the productivity of multi-storey building projects; however, in infrastructure projects, the practice is not given top priority. In industrial projects, the practice is not included in the list of practices that have the potential to enhance productivity in construction projects. In Victoria State, Australia, regulations oblige contractors that use public roads during the construction periods to prepare traffic management plan. Failure to prepare the plan will result in a financial penalty and the suspension of works if accidents occur. According to Road Safety (Traffic Management) Regulations (2009), any person conducting an activity on a road or road

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

263

related areas must maintain a copy of the traffic management plan on the worksite at all times when the work is undertaken, and it should be available for inspection on request by an authorised person. The absence of the plan is an offence against the Regulation. Therefore, to reduce project delay and related penalty, the contractors operating in Victoria considered traffic control plan as the most important practice.

'Project start-up plan' is found to be among the crucial practices that could enhance productivity in multi-storey building projects. The finding implies that contractors in Victoria, Australia, attempt to reduce the initial project delay by using project start-up plan which could be used as a checklist to remind the project team regarding information that is required to start a project. In the plan, the pre-commencement meeting dates, resource requirements, and any other information which could be forgotten during the commencement date can be included.

'Machinery positioning strategy' is considered as the most important practice for improving productivity in multi-storey building projects. However, the practice is rated as having less importance for enhancing productivity in infrastructure projects (Table 9-4). The finding implies that, in multi-storey building projects, the location of the critical machinery such as tower cranes should be carefully analysed for the sake of enhancing productivity. The construction systems in building projects in Victoria such as the use of heavy precast concrete panels could influence the location of the cranes. If the machine is not positioned in a suitable location, the concrete elements might not be placed as per the scheduled time. The contractor might change the design of the panels to reduce their weights or look for other alternatives to place them. Moreover, since most multi-storey building projects are in a city where there are confined spaces, the practice 'machinery positioning strategy' is found to be one of the important practices.

A framework which can assist contractors' project managers to enhance productivity is shown in Figure 9-3. The framework comprises of the management practices related to construction methods and the factors which negatively influence productivity.

The implementation of the management practices indicated in the framework (Figure 9-3) can help contractors to improve the construction projects' productivity by mitigating the adverse effects of the productivity impediments. Previous researchers identified inadequate sequencing of works as one of the productivity problems (Chan and Kaka, 2007, Dai et al., 2009a, Rivas et al., 2011, El-Gohary and Aziz, 2014). On the other 264

hand, the implementation of the practice 'integrated schedule' can mitigate the problem. The construction schedule which integrates the commencement and completion dates of each activity, the quantity as well as the delivery dates of materials and machinery, schedules of other resources required for the execution of the activities, and the status of the activities is essential to improve productivity. Furthermore, updating the main schedule regularly and integrating the work programs of the subcontractors is significant to reduce the loss of productivity due to the overlapping of the works executed by various subcontractors.



Figure 9-3 Framework of the Management Practices Related to Construction Methods and Factors Negatively Influencing Productivity

The use of excessive shift work can cause the physical tiredness that could result in the loss of productivity in construction projects (Ghoddousi and Hosseini, 2012, Jarkas and Bitar, 2012, Hughes and Thorpe, 2014). This problem can be reduced by developing different working hour strategies such as reducing the daily working hours and negotiating the suitable working hours with workers or their unions. Lack of adequate instruction and supervision is also cited as the most common factor reducing construction projects' productivity (Zakeri et al., 1996, Kaming et al., 1997, Alinaitwe et al., 2007, Rivas et al., 2011, Naoum, 2016). However, monitoring the construction works regularly, taking corrective actions and updating the construction schedules periodically can assist contractors' project managers to reduce the loss of productivity associated with supervision (Figure 9-3).

Previous researchers have confirmed that poor site layout has an adverse impact on construction productivity (Rivas et al., 2011, Ghoddousi and Hosseini, 2012, Hughes and Thorpe, 2014, Naoum, 2016). However, the implementation of dynamic site layout plan which is subjected to changes based on the phases of the building construction project, the integration of the traffic control plan with the site layout plans, and careful investigation of the position of the critical construction equipment such as cranes have the potential to improve productivity by reducing the problems associated with a site layout (Figure 9-3).

Relative Importance of the Pre-Construction Phase Management Practices

Ten preconstruction phase management practices that have the potential to improve productivity in multi-storey building construction projects in the context of Victoria, Australia are identified. These are 'Short Interval Plan,' 'Well-Defined Scope of Work,' 'Use of Software in Planning Work Packages,'' Dedicated Planner,' 'Construction Work Packages,' 'Buildability Review,' 'Utilities Alignment,' 'Contract Types,' 'Model Development,' and 'Regulatory Requirement.' The quantitative data analysis showed that all the 10 preconstruction phase management practices are positively correlated with productivity.

The practice 'Well-Defined Scope of Work' is found to be the most important preconstruction phase management practice (Table 9-5 column-3). However, the implementation of the practice is not given top priority to enhance productivity in industrial and infrastructure projects (Table 9-5 column 1 and 2). The finding implies that scope definition is a fundamental practice in multi-storey building projects. If the scope of a building project is defined well, the more accurate completion time of each activity can be estimated, better work schedules can be prepared, variations during the construction phase could be reduced, disputes due to change orders could be minimised, and productivity can be improved. Moselhi et al. (2005) found that change orders due to not defining the scope of works have a substantial impact on the productivity of construction projects. Therefore, to reduce the loss of productivity associated with change orders, contractors operating in Victoria, Australia give priority for the implementation of the practice well-defined scope of work.

As indicated in Table 9-5, this research confirmed that the preconstruction phase management practices and their relative importance vary from project to project. Thus,

implementing practices which are given more importance in one project type might not improve productivity in another project type. For instance, the implementation of the practice 'environmental requirements' is crucial to enhance productivity in infrastructure construction projects but the practice is not significant for increasing productivity in multi-storey building projects.

 Table 9-5 Comparison of the Preconstruction Phase Management Practices in Industrial, Infrastructure, and Multi-Storey Building Construction Projects

Infrastructure Projects (CII, 2013b)		Industrial Projects (CII, 2013a)		This Study	
Management Practices	Rank	Management Practices	Rank	Management Practices	Rank
Environmental Requirements	1	Short Interval Plan	1	Well-Defined Scope of Works	1
Regulatory Requirements/ Permitting Requirements	2	Design Readiness for Construction	2	Short Interval Plan	2
Design Readiness for Construction	3	PPMOF Evaluation	3	Buildability Review	2
Utility Alignment and Adjustment	4	Well-defined Scope of Work	4	Construction Work Packages	4
Contract Types/Strategies	5	Dedicated Planner	5	Regulatory Requirement	5
Dedicated Planner	6	Construction Work Packages	5	Contract Types	6
Well-defined Scope of Works	7	Installation Work Packages	5	Use of Software in Planning Work Packages	7
Utility Agreements	8	Engineering Work Packages	8	Utilities Alignment	8
Short Interval Planning	9	Identify Required Permitting	9	Dedicated Planner	9
Construction Work Packages	10	Project Model Requirements	10	Model Development	10
Contracts and Agreement with Agencies	11	Utilization of Software to Assist in Generating Work Packages	11		
Right-of-Way, land, and Utilities Acquisition Strategy	12				
Use of Software	12				
Model Requirements/3D Visualization	14				

The finding of this research reveals that there are no significant differences among 'short interval plan,' 'buildability review,' 'construction work packages,' and 'regulatory requirement.' This implies that breaking down the project into manageable work packages alone might not be sufficient to improve productivity, the short interval plan for the work packages should be prepared, the review of the design of the specific work package should be conducted, and any regulatory requirement which affects the execution of the work package should also be identified. This suggests that the four practices should be implemented jointly to improve productivity in multi-storey building projects.

Based on the findings of this research, a framework which assists contractors' project managers to improve productivity by mitigating the adverse effects of the factors which negatively influence productivity is developed (Figure 9-4). Inadequate or lack of construction plan/schedule is one of the major factors that decrease the productivity of construction projects (Zakeri et al., 1996, Durdyev and Mbachu, 2011, Ghoddousi and Hosseini, 2012, El-Gohary and Aziz, 2014). However, this problem can be reduced by preparing detailed day to day plan or weekly work plans (short interval plans). Similarly, disruption of utilities such as power and water is among the critical factors 267

influencing productivity (Lim and Alum, 1995, Alinaitwe et al., 2007, Ghoddousi and Hosseini, 2012). On the other hand, reviewing and listing the existing utilities, marking the location of the utilities before excavation and getting approval for connecting to the existing utilities from the concerned authorities can minimise the loss of productivity associated with the disruption of power and water supply (Figure 9-4).



Influencing Productivity

The lack of integration between design and construction have been identified as major impediments to productivity improvement (El-Gohary and Aziz, 2014). Moreover, the complexity of designs hampers the productivity performance (Alinaitwe et al., 2007, Enshassi et al., 2009, Durdyev and Mbachu, 2011). Nevertheless, conducting buildability review before starting construction project can minimise the loss of productivity associated with the complexity of design (Figure 9-4).

Delays in getting permits and stoppage of works due to a breach of regulatory requirements are mentioned as some of the factors which negatively influence construction projects' productivity (Lim and Alum, 1995, Alinaitwe et al., 2007, Dai et al., 2009a, Durdyev and Mbachu, 2011). However, assessing the types of permits and the timeline to get them approved; integrating the expiry date of the permits in work schedule; identifying the issuing authority; and tracking the acquisition, release, and closeout of each permit could assist contractors to reduce the loss of productivity. For instance, in the city of Melbourne, contractors are required to prepare 'Construction Management Plan' and the plan should be approved before the commencement of any construction activity (City of Melbourne, 2005). In the 'Construction Management

Plan' contractors need to explain how to address issues related to public safety, amenity, and site security; operating hours; noise and vibration controls; air and dust management; storm water and sediment control; waste and materials re-use and traffic management (City of Melbourne, 2005).

Relative Importance of the Human Resource Management Practices

Eleven human resource management (HRM) practices that have the potential to improve productivity in multi-storey building projects are identified during Phase I of the research. These include 'Clear Delegation of Responsibility,' 'Stability of Organizational Structure,' 'Crew Composition,' 'Retention Plan for Experienced Personnel,' 'Employees Training,' 'Skill Assessment and Evaluation,' 'Career Development,' 'Financial Incentive Programs,' 'Social Activities,' 'Non-Financial Incentive Programs,' and 'Exit Interviews.' The findings of the quantitative data analysis (Phase II) confirmed that all the 11 HRM practices are positively associated with productivity.

The practices 'clear delegation of responsibility is found to be the most important HRM practices (Table 9-6 column-3). Naoum (2015) also found that delegation of responsibility and involvement of a site manager are the two main factors that have the greatest impact on the site manager's performance. Nitithamyong and Tan (2007) identified that clear delegation of the decision-making authority is one of the success factors for project management consulting firms in Malaysia. Thus, to enhance productivity, contractors involved in multi-storey building projects in Victoria, provide priority for the implementation of the practice 'clear delegation of responsibility.'

Infrastructure Projects (CII, 2013b)		Industrial Projects (CII, 2013a)		This Study	
HRM Practices	Rank	HRM Practices	Rank	HRM Practices	Rank
Crew Composition/ crew formation	1	Employee/ Trades Training	1	Clear Delegation of Responsibility	1
Skill Assessment and Evaluation	2	Retention Plan for Experienced Personnel	2	Stability of Organizational Structure	2
Employee/Trades Training	3	Clear Delegation of Responsibility	3	Crew Composition	3
Social Activities	4	Career Development	4	Retention Plan for Experienced Personnel	4
Financial Incentive Programs	5	Maintain Stability of Organizational Structure	5	Employees Training	5
Career Development	6	Financial Incentive Programs	6	Skill Assessment and Evaluation	6
Exit Interviews	7	Non-Financial Incentive Programs	7	Career Development	7
Non-Financial Incentive Programs	8	Exit Interviews	8	Financial Incentive Programs	8
Retention Plan for Experienced Personnel	9	Social Activities	9	Social Activities	8
Clear Delegation of Responsibility	10			Non-Financial Incentive Programs	10
Maintain Stability of Organizational Structure	11			Exit Interviews	11

 Table 9-6 Comparison of Human Resource Management Practices in Industrial, Infrastructure, and Multi-Storey Building Construction Projects
As indicated in Table 9-6, the practice the 'stability of organisational structure' is one of the most important HRM practices that have the potential to improve productivity in multi-storey building projects. However, the practice is found to be one of the least important practices for enhancing productivity in infrastructure construction projects. The finding of this study suggests that contractors operating in Victoria focus on maintaining the stability of their projects' organisational structures since the frequent change in personnel could be associated with the loss of productivity due to inadequate supervision or lack of project information. There could be the loss of project information, particularly the verbal communications made with the project stakeholders if the key personnel such as project managers, site managers, and supervisors leave the project. Moreover, the recruitment of experienced construction professionals who have appropriate skills could take longer time, and the productivity of the workers is affected due to the absence of a professional who can issue the instruction.

The findings of the study revealed that there is no significant difference among the weights of the top three HRM practices: 'clear delegation of responsibility,' 'stability of organisational structure' and 'crew composition.' This implies that the three practices should be implemented jointly to improve productivity in multi-storey building projects. Thus, the practice of maintaining the stability of organisational without forming a crew that comprises of skilled personnel might not enhance productivity. Moreover, the duties and responsibilities of each staff need to be clarified.

In Figure 9-5, a framework which integrates the human resource management practices and factors negatively influencing construction projects is provided.



270

Previous researchers mentioned that the presence of incompetent or unskilled supervisors on construction sites contribute to rework and loss of productivity as they do not offer accurate information to the craftsmen (Chan and Kaka, 2007, Dai et al., 2009b, Jarkas and Bitar, 2012, Hughes and Thorpe, 2014). However, the provision of technical training after assessing the skill level of the supervisor could be an essential practice to enhance productivity (Figure 9-5). Similarly, lack of motivation by workers is frequently mentioned as one of the productivity determinants from a human resource perspective (Liberda et al., 2003, Durdyev and Mbachu, 2011, Jarkas and Bitar, 2012). On the other hand, employees could be motivated if contractor prepares the career development plan for its staff. The arrangement of financial and non-financial incentives schemes could also be essential to motivate the employees. Furthermore, the organisation of social activities at both company and project levels is necessary to motivate the employees (Figure 9-5). Loosemore et al. (2003) explained that reward is one of the reasons why the majority of individuals seek employment in the highly commercialised construction industry and those who perform well expect to be rewarded in a similar way. According to the authors, the objective of the monetary or non-monetary reward management strategy is to motivate people to work towards an organisation's goals. Thus, the implementation of the four HRM practices (Figure 9-5): 'career development plan,' 'financial incentive programs,' 'non-financial incentive programs,' and 'social activities,' have the potential to improve productivity by reducing adverse impacts of the lack of motivation of employees such as not working with full potential and absenteeism.

The other factor which negatively influences construction project's productivity from HRM perspectives is employees' turnover (Kaming et al., 1997, Makulsawatudom et al., 2004, Rivas et al., 2011). This problem could be reduced by the implementation of the practices 'career development plan,' 'financial incentive programs,' and 'non-financial incentive programs' (Figure 9-5). If the employees are motivated by using these schemes, the rate of employees' turnover could be reduced, and the loss of construction productivity can be minimised.

Relative Importance of the Safety and Health Practices

Seven safety and health practices that have the potential to improve productivity in multi-storey building construction projects have been identified. These include 'safety

and health policy,' 'safety and health plan,' 'hazard analysis,' 'safe work method statement,' 'toolbox safety meetings,' 'housekeeping,' and 'safety and health training. The quantitative data analysis revealed that all the safety and health practices are positively correlated with productivity.

The practices 'Safety and Health Policy,' 'Safety and Health Plan,' and 'Hazards Analysis' are the top three practices in the safety and health category. However, the Friedman and Wilcoxon tests indicated that the weights of the seven safety and health practices are not significantly different from each other. This implies that all safety and health practices should be implemented jointly to enhance productivity in multi-storey building projects in the context of Victoria State, Australia. This could be due to the presence of several occupational safety and health practices. Failure to comply with the regulations results in financial penalty and suspension of the construction work which could be one of the causes of the loss of construction productivity. For instance, Work and Health Safety Regulations 2011 Part 6.5 Article 316 and 317 explain the duty of a construction workers, and the regulation specifies the associated penalty of 18,000 Australian Dollars (AUD) for the noncompliance.

Furthermore, the presence of a strong Construction, Forestry, Mining, and Energy Union (CFMEU) could enforce contractors to implement safety and health practices. The workers might suspend the execution of the construction activities if a site is not safe. The employees can elect the health and safety representatives (HSR's) who have the power to issue Provisional Improvement Notice (PIN) and to cease work when there are urgent matters that are immediate threats to the health and safety of any person (CFMEU, 2016b). According to article 60 of the Occupational Health and Safety Act 2004, the HSR's can issue the Provisional Improvement Notice (PIN) to a contractor to remedy any contravention of the provisions of the Act. Failure to comply with the PIN can result in 2500 penalty units. Furthermore, Article 74 of the Act authorises the health and safety representatives to direct the suspension of the construction work if the work involves an immediate threat to the health or safety of any employee.

Work Health and Safety Regulations (2011) stipulates that a person conducting a business that involves high-risk construction work must, before high-risk construction

work commences, prepare a 'safe work method statement' for the proposed work. Failure to comply would result in a financial penalty of AUD 30, 000 and suspension of the construction work. Thus, contractors in Victoria State, Australia implement the Safety and Health practices to comply with the regulations and to minimise the penalty associated with the noncompliance of the regulations such as financial penalty and suspension of works which in turn influences the productivity of the multi-storey building construction projects.

The finding of this study revealed that the relative importance of the safety and health practices in improving productivity in the multi-storey building, infrastructure, and industrial construction projects is different. Table 9-7 indicates the comparison of the safety and health practices in the three project types.

Table 9-7 Comparison of Safety and Health Practices in Industrial, Infrastructure, and Multi-Storey Building Projects

Infrastructure Projects (CII, 2013b)		Industrial Projects (CII 201)	Multi-Storey Building Projects (This Study)		
		industrial i Tojects (CII, 2013a)			
Environment, Safety and Health Practices	Rank	Environment, Safety and Health Practices	Rank	Safety and Health Practices	Rank
Formal Safety and Health Policy	1	Substance Abuse Programs	1	Safety and Health Policy	1
Safety and Health Plan /Zero Accident Techniques	1	Toolbox Safety Meetings	2	Safety and Health Plan	2
OSHA Compliance Training	3	OSHA Compliance Training	3	Hazards Analysis	2
Hazards Planning	3	Zero Accident Techniques	4	Safe Work Method Statement	4
Hazard Analysis	5	Task Safety Analysis	5	Toolbox Safety Meetings	5
Task Safety Analysis	6	Identification of Potential Hazards	6	Housekeeping	6
Toolbox Safety Meetings	7	Housekeeping	7	Safety and Health Training	6
		System Test Hazards Planning	8		

While 'substance abuse program' is the most significant practice to improve productivity in industrial projects, the practice is found to be not relevant to enhance productivity in multi-storey building projects (Table 9-7). The finding of the qualitative data analysis revealed that testing construction workers for substance abuse is not common in building construction projects in Victoria, Australia. However, since September 2015 the Australian Government made amendments to the Building Code 2013 which requires contractors to have a comprehensive policy for managing drug and alcohol issues including mandatory drug and alcohol testing on Commonwealth funded projects that meets the financial threshold (Building Code (Fitness for Work/Alcohol and Other Drugs in the Workplace) Amendment Instrument, 2015). The code stipulates that if the value of a building work is at least AUD 5,000,000 and represents 50% of the total construction value or if the Commonwealth's contribution to the project exceeds AUD 10, 000,000, drug and alcohol test should be conducted. The qualitative data of

this thesis was collected prior to the amendment of the Building Code 2013, and the survey participants described that the practice is not significant for building projects. However, after the modification of the Code, the practice might be important.

Based on the findings of this study, the framework which integrates the safety and health practices and the factors negatively influencing productivity is developed (Figure 9-6).



Figure 9-6 Framework of Safety and Health Practices and Factors Negatively Influencing Construction Projects' Productivity

The occurrence of accidents in construction sites is mentioned as some of the main factors negatively influencing productivity in construction projects (Enshassi et al., 2007, Kazaz and Ulubeyli, 2007, Jarkas and Bitar, 2012, Hughes and Thorpe, 2014). However, the loss of productivity due to accidents could be reduced by preparing and monitoring the implementation of safety and health policies such as drug and alcohol policy and UV protection policy on each building construction site; and by conducting a hazard analysis for both 'high-risk construction work' and 'works of minor nature' (Figure 9-6). Safe Work Australia (2014a) states that Safe Work Method Statement (SWMS) which include the hazard analysis for high-risk construction works should be prepared, and attention should also be given to the 'work of minor nature' as they could be hazardous. Therefore, conducting a hazard analysis for both low-risk and high-risk works could be essential to reduce accidents in construction sites. Moreover, conducting general induction and site specific safety training is important to create awareness for employees regarding the possible hazards on the specific project and to reduce the number of accidents. Housekeeping is also noteworthy in reducing accidents. For instance, if sharp or other unwanted materials are cleared from a building site, on a

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

regular basis, the chance of occurring accidents on the specific project could be minimised.

Unsafe work conditions such as noise, dust, radiation, poor ventilation, low lighting, limited access, unavailability of personal protective equipment and other related conditions have negative effects on productivity (Liberda et al., 2003, Enshassi et al., 2007, Ghoddousi and Hosseini, 2012, El-Gohary and Aziz, 2014). However, the implementation of appropriate risk control measures which are proposed while preparing the SWMS's could reduce the loss of productivity due to unsafe work conditions (Figure 9-6). In the SWMS, the list of the work tasks is presented in a logical order, the potential hazards and risks associated with each task are identified, and the control measures for the potential risks are described (Safe Work Australia, 2014). Safety and health training is essential to ensure the compliance with the proposed control measures such as the proper use of the personal protective equipment (PPE). Safety training is also beneficial before using new construction equipment. Safe Work Australia (2013) explained that the risk control measures in construction projects include thorough training regarding the use of PPE and effective supervision to ensure that the workers comply with the safety requirements. Therefore, the implementation of the practices safe work method statement, safety and health training, and housekeeping in multi-storey building construction sites could minimise unsafe work conditions and improve productivity.

9.3 Relationships among the Construction Management Practices

This section presents the discussion of the findings of the correlation analyses conducted to investigate the relationships among the 47 construction management practices (subsection 9.3.1). The findings of the correlation analyses of the practices within their respective categories are discussed in this section (subsection 9.3.2).

9.3.1 Associations among All Construction Management Practices

The findings of this study indicate that the practices categorised under 'construction materials management practices' are strongly correlated with practices categorised under 'construction equipment management practices.' This could be due to the type of construction materials which are used in Victoria in which most materials are mainly prefabricated or preassembled elements which need construction equipment or cranes

for lifting purpose. According to PrefabAUS (2014), the prefabricated construction system in Australia is increasing at about 5% per year. If a crane which has the capacity to lift the heaviest materials is not procured and the crane usage is not scheduled on a daily basis, there could be the loss of productivity due to unavailability of materials since the materials might not be transported to the location close to their installation area. Among the three practices which are categorised under construction equipment management practices, 'Procurement Plans for Construction Equipment' is strongly correlated with the practices categorised under 'construction materials management practices.'

The practices in the 'management practices related to construction methods' category are strongly associated with the practices in 'preconstruction phase management practices' category. The regulatory requirement (the practice categorised under preconstruction phase category) is strongly correlated with traffic control plan (the practice categorised under management practices related to construction methods). This indicates that, in the context of the Victorian construction industry, depending on the location of projects, regulatory bodies oblige contractors to prepare the traffic management plans. For instance, Road Safety (Traffic Management) Regulations (2009) dictates that any person conducting an activity on a road or road related areas must maintain a copy of the traffic management plan on the worksite at all times. The practice machinery positioning strategy (in the management practices related to construction methods category) and utilities alignment (in the pre-construction phase management practices category) are strongly associated. This shows that the contractors need to investigate the presence of the existing utilities such as electricity, water and gas before locating the construction machinery, for instance, the tower cranes. To place the foundations of the tower cranes, excavation works are carried out, and the existing utility lines could be affected. The practices project completion plan (in the management practices related to construction methods category) and buildability review (in the pre-construction phase management practices category) are also strongly correlated which indicates that conducting buildability review is essential for the completion of a building project as per the planned time. If a design is not constructible, the project could incur the loss of productivity as the modification of the design might take a long time and the operatives could wait until the design is completed. However,

the intended completion date could be attained if buildability reviews are carried out. Pheng Low (2001) concluded that building projects with higher buildability scores tend to achieve correspondingly higher productivity levels.

The practices in the category of 'human resource management practices (HRM)' are strongly correlated with the practices categorised under 'preconstruction phase management practices,' 'construction equipment management practices,' and 'management practices related to construction methods.' The practices crew composition (in HRM practices category) and short interval plan (in pre-construction phase management practices category) have a strong correlation which implies that to improve productivity in building projects, the short interval plan should specify the type of crews such concreting and plastering crews as well as the number of crew members that will be deployed on a specific date. If the plan does not contain such information, there could be an overlap of the crews, and the working site might be overcrowded which could lead to the loss of productivity. Enshassi et al. (2007) found that the increment of the workforce on a construction site causes overcrowding of labour and interference among the crews which consequently reduce the labour productivity in construction projects. Thus, contractors in Victoria integrates the practices crew composition and short interval plan to improve productivity in their building projects.

The stability of organisational structure (in HRM practices category) and procurement plan for construction equipment (in construction equipment management practices category) have high correlation coefficient. This implies that if the organisational structure of a project is stable, the chance of missing information regarding the procurement of equipment such as oral communications made with equipment suppliers could be low since a person who is involved in the procurement process might not leave a project or a company. On the other hand, frequent change of the project staff who planned the procurement of equipment could lead to delay in the delivery of the equipment which could be associated with the loss of productivity and cost overrun. Manavazhi and Adhikari (2002) found that delay in the procurement of the construction equipment is one of the causes for cost overrun in the road projects. Therefore, contractors in Victoria maintain the stability of their projects' organisational structure for the ease of monitoring the procurement process of the construction resources and minimising the loss of project information.

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

The HRM practices 'career development' and 'employees training' are strongly associated with the project completion plan (one of the practices in the management practices related to construction methods category). The findings indicate that motivating employees by providing training and preparing career development plan could help to achieve the planned project completion date. If employees are motivated, they might not leave a project and contribute to the successful completion of the project their full potential. Loosemore et al. (2003) explained that employees development program helps to ensure that the employees have the skills required for their current roles, and it can also be a motivating factor since the significant training and development indicate a commitment to people and the recipients are more likely to feel valued.

The practices in 'safety and heath practices' category are positively and strongly correlated with practices categorised under 'construction equipment management practices.' This implies that construction equipment is one of the major hazards that could result in accidents, and an increase in the implementation level of the construction equipment management practices could lead to increased safety and health practices. The practice hazards analysis (in Safety and Health practices category) and construction equipment maintenance (in Construction Equipment Management practices category) are strongly associated which implies that maintaining construction equipment could reduce accident rates in building construction projects. For instance, forklifts related risks such as the death of pedestrians due to being struck by the forklift or being hit by loads as well as the injury or death of operators due to tip over could be reduced by providing necessary maintenance to the equipment. WorkSafe Victoria (2006) states that forklift is one of the most dangerous pieces of equipment found at Victorian workplaces, and maintaining the equipment plays a major role in reducing the risks posed by forklifts. Work Health and Safety Regulations (2011) stipulates that contractors must ensure that the maintenance, inspection, and testing of the equipment (crane) are carried out by a competent person. Failure to comply with the provision of the regulation is an offence and results in a financial penalty or suspension of construction works. Thus, to meet the regulatory requirement and to minimise the potential hazard associated with construction equipment, contractors operating in Victoria maintain their construction equipment.

The practices safe work method statement (in safety and health practices category) and traffic control plan (in management practices related to construction methods category) are strongly related. The finding implies that by preparing effective traffic management plan, the risks associated with the execution of a particular task could be minimised as pedestrians can be separated from hazardous equipment such as forklifts and trucks. According to WorkSafe Victoria (2006), an effective traffic management plan uses a range of devices including pedestrian and forklift exclusion zones, safety zones for truck drivers, safety barriers, containment fences, and reduced speed limiting devices such as smart forklifts and traffic signs. As a result, the plan helps to reduce accidents. The strong relationship between the practices safe work method statement (SWMS) and traffic control plan implies that preparation of safe work method statement is essential to safely set up, maintain and remove the traffic control systems. The traffic control activities can be classified as high-risk construction work that requires SWMS since the activities are carried out adjacent to a road and involve the movement of the powered mobile equipment (for example, forklifts). According to Victorian Occupational Health and Safety Regulations (2007), construction work on or adjacent to roadways or railways used by road or rail traffic, or construction work conducted at workplaces where there is any movement of the powered mobile equipment is considered as high-risk, and the work requires SWMS. Article 5.1.9 of the regulation stipulates that if there is non-compliance with the SWMS, the contractor should stop the work immediately or as soon as it is safe to do so. Thus, to comply with the regulation and to avoid the stoppage of the construction activities due to unavailability of SWMS, contractors in Victoria prepare SWMS for the traffic control related tasks.

9.3.2 Inter-Relationships among the Practices within Their Categories

Relationships among the Construction Materials Management Practices

The findings of this study indicate that all the six construction materials management practices are positively correlated to each other, and the associations have various degrees of strength. This implies that as the implementation level of one construction materials management practice increases, the implementation level of another practice also increases. The quantitative data analysis revealed that 'long-lead materials identification' and 'procurement plan for materials' are strongly correlated. This shows that listing all materials required for the construction of a specific building project and

preparing the overall materials procurement plan can assist the project planner/ manager to identify the most critical or long-lead materials whose delay might result in the loss of productivity. In other words, the practice of 'preparing procurement plan for all materials' could assist the contractor's project team not to miss the most critical materials that are required to deliver a building project on time. In the context of Australia where most materials are imported from overseas (Wheeldon, 2012), the construction of building projects could be delayed due to the late delivery of the longlead materials unless they are identified and ordered early. The practice 'material inspection process' and 'materials inspection team' are also strongly associated. This implies that forming the construction materials' inspection teams, who are experienced, could help to prepare better materials inspection procedures since the team members could have better understanding of the materials' specifications and standards.

Relationships among Construction Equipment Management Practices

The three construction equipment management practices: 'Construction Equipment Maintenance,' 'Construction Equipment Procurement Plan,' and 'Construction Equipment Productivity Analysis' are correlated, and the relationships are statistically significant. The practices 'construction equipment procurement plan' and 'construction equipment productivity analysis' are strongly associated. This indicates that preparation of equipment procurement plan without analysing the productivity of the machinery might not improve productivity in building projects.

Relationships among Management Practices Related to Construction Methods

The findings of this study indicated that the 10 management practices categorised under construction methods are positively correlated. However, the practices 'traffic control plan' and 'site security plan;' 'machinery positioning strategy' and 'dynamic site layout plan;' 'project completion plan' and 'work schedule strategies;' and 'schedule control' and 'integrated schedule' are strongly associated.

The findings imply that site security systems need to be considered while preparing the traffic control plan. For instance, if the traffic control plan is poor, the security cameras, their power cords, as well as the site fences could be hit by the vehicles moving in and out of the building sites. Thus, the traffic control plan and site security plan are interrelated. Similarly, effective site layout plan should consider the positions of cranes, hoist and other construction equipment that is deployed at a certain building

280

construction project. If the two practices ('site layout plan' and 'machinery positioning strategy') are not harmonised, productivity could be affected as the location of the construction materials might be far from the position of the crane and the materials could not be transported efficiently to the required working area or level. This could result in the lack of materials which in turn influences the productivity of the construction workers.

The practices project completion plan and working schedule strategies are also strongly associated which implies that the better the working schedule strategies are, the better the project completion plan will be. The working schedule which is prepared by considering the workers' physical fatigue and by involving supervisors and/or other workers who can suggest appropriate working times could help the contractor's project manager to prepare more feasible project completion plan. The chance of achieving the planned completion dates could be higher as the supervisors who are closely monitoring the activities of the labourers are involved in the development of the most suitable working hour strategy.

The practices 'schedule control' and 'integrated schedule' are also strongly correlated. The finding indicates that if different schedules are integrated together, monitoring the progress of construction works could be easier since the supervisor or the project manager can get all the required information in one document. In the integrated schedule, the main work schedule; current work status; materials delivery status; the materials, machinery, and manpower schedule; and other information such as detailed drawings are found in one place and supervisor can easily control the works using the available information. Thus, the probability of the loss of productivity due to lack of project information could be reduced.

Relationships among Preconstruction Phase Management Practices

The findings of the correlation analysis show that the practice 'construction work packages' is strongly related to 'short interval plan' which indicates that to improve productivity in multi-storey building construction projects, short interval plan should be prepared for each work package. The practice 'regulatory requirement' and 'short interval plan' are also strongly correlated. This shows that the type of permits and the dates when the permits expire need to be incorporated in the short interval plan. The practices 'contract types' and 'well-defined scope of works' are strongly associated

281

which implies that the level of defining the scope of the construction work package depends on the type of contracts. For example, in design and construct delivery system, the project scope might not be developed in detail at the beginning of the construction project. However, in 'construct only' delivery system the scope of works should be defined during the tendering phase to reduce changes and the associated productivity loss during the construction phase.

The use of software in creating work packages is also strongly associated with the short interval plan. The finding implies that the use of software could help not to miss activities within a work package which might be forgotten if the work breakdown is conducted manually. If the activities are not missed, their durations can be estimated, the volume of works can be estimated, and their execution dates and the required resources can be planned. Thus, the use of software assists in preparing better short interval plan. The practices 'buildability review' and 'construction work package' are strongly correlated which indicates that conducting buildability review for each work package could enhance productivity in multi-storey building projects. If the design of the work package is complex and could not be implemented easily, the productivity of a project could be negatively influenced.

Relationships among Human Management Practices

There is a strong association between the practices 'crew composition' and 'skill assessment and evaluation' which implies that unless the skill level of a construction worker is known, it might not be feasible to form a crew which is more productive. For instance, the skill level of the contractor's supervisors should be assessed and evaluated before forming the supervision team. Supervisors with low skill level might not communicate the project information via drawings, specifications or other mechanisms to the operatives; as a result, there could be the inadequacy of information which could be associated with the loss of productivity. Tabassi and Bakar (2009) explained the effects of unskilled construction workers in construction projects in Iran. The authors found that project delay, extra costs, and low quality of construction are the consequences of employing unskilled employees. Similarly, Doloi et al. (2012) found that some of the causes of poor labour productivity in Indian construction projects are the employment of unskilled labour and inefficient management skills of the supervisor on the site. Therefore, integration of the practices 'crew composition' and 'skill

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

assessment and evaluation' is essential to enhance productivity in multi-storey building projects the context of Victoria State, Australia.

The practices 'Skill Assessment and Evaluation' and 'Financial Incentive Programs' are strongly associated which implies that as the skill evaluation criteria for selecting certain employee increases, the corresponding financial incentives could also be high. If the financial incentive scheme is low, the chance of getting an employee with better technical as well as managerial skill can also be low. Loosemore et al. (2003) explained that reward is one of the main reason why the majority of individuals seek employment in the construction industry, and those who perform well or highly skilled expect to be rewarded accordingly. The Australian Fairwork Commission also sets minimum wage according to the skill level of employees in the construction industry. The Commission prepared the 'building and construction general on-site award 2010' which specifies the minimum wages which correspond to the skill levels of general construction workers (Construction Workers (CW) level 1 to CW level 8) and engineering construction workers (Engineering Construction Workers (ECW) level 1 to ECW level 9)(Fair Work Commission, 2016). Accordingly, while the minimum hourly wage for CW level 1a in the year 2015/2016 is AUD 18.61, the minimum hourly wage for CW level 8 is AUD 23.69. Thus, there is a relationship between skill level and financial incentives.

'Career Development Plan' and 'Financial Incentive Programs' are among the human resource management practices which are strongly correlated. The finding implies that construction firms that have career development plans for their employees also arrange financial incentive programs to motivate their workers. The practices 'Non-Financial Incentive Programs' and 'Financial Incentive Programs' are also strongly associated. The finding indicates that financial incentive schemes alone might not motivate employees. Non-financial incentive programs should supplement the financial incentives. Fagbenle et al. (2004) found that both financial and non-financial incentive schemes have a positive impact on the productivity, but non-financial incentive scheme is more important. Loosemore et al. (2003) explained that the balance between the monetary and non-monetary rewards depend on internal and external factors such as the financial state of a company, competition within the labour market, and the policies of its rivals.The authors suggested that during the construction boom, there could be a greater emphasis on a monetary reward if rival employers are offering relatively high

wages. Therefore, there is a relationship between financial and non-financial incentive schemes.

The relationship between 'Stability of Organisational Structure' and 'Retention Plan for Experienced Personnel' is strong and statistically significant. The finding shows that to have stable organisational structure, contractors should plan how to retain experienced project managers and other key staff as they could have better skills. According to Nitithamyong and Tan (2007), project management skills are aquired when the manager work with people of different organisational levels from multidisciplinary fields. The authors emphasised that besides the technical knowledge, the project managers should have better managerial skills. Thus, contractors need to retain experienced or key professionals so that the stability of the organisational structure is maintained and productivity could be improved.

Relationships among the Safety and Health Practices

The quantitative data analysis revealed that all the safety and health practices are positively correlated. However, 'safety and health policy' 'safety and health training' are strongly associated. The finding implies that to improve productivity in multi-storey building projects, safety and health training should be included in the safety and health policy prepared at company and project levels. For instance, the policy could explicitly state that workers who do not complete general induction training and get a certificate of completion or construction induction card should not start working on a particular building project. This practice could help to reduce the rate of accidents in the building projects. In Victoria, the regulatory bodies oblige contractors to accept alternative evidence of completing induction training and contractors need to consider these options in their safety policies. The 'red card' which shows the completion of the previous Victorian Construction Industry Basic Induction Course; the white card or a construction induction card issued by WorkSafe Victoria after the successful completion of induction training by registered training organisation (RTO); and a statement of attainment issued by an RTO until the issuance of construction induction card can be considered as evidence for completion of the safety training (WorkSafe Victoria, 2008b). Thus, safety training and safety policy can be inter-related.

9.4 Applications of the Scoring Tools and Prediction Models

This section provides the discussions of the findings obtained to achieve: Objective 2 or "to refine and validate scoring tools for measuring, planning, monitoring and evaluating the management practices that could improve productivity in multi-storey building construction projects," and Objective 3 or "to build and validate a tool that can be used to predict the probability of exceeding baseline productivity factor based on a score of the management practices." The section demonstrates how the scoring tools of the management practices and the regression models of productivity and the practices are used during the planning stage of a multi-storey building project (subsection 9.4.1). It also presents how the scoring tools are used during the construction and the project closure phases in subsection 9.4.2 and 9.4.3 respectively.

9.4.1 The Use of the Construction Management Practices Scoring Tools and the Logistic Regression Models during Project Planning Phase

The scoring tools of the overall construction management practices and the logistic regression model of productivity and the practices can assist the contractor's project manager to plan appropriate practices that can be implemented on a certain multi-storey building construction project. The project manager can make necessary decisions to improve the productivity of a certain multi-storey building project after assessing the risk of low productivity by using the tools and the model. Suppose a project manager plans to implement the practices indicated in Table 9-8. For the practice procurement plan for construction materials, the manager plans to implement "a procurement plan and schedule exists only for large materials and costly items" which is Level C, and the equivalent score for this practice is 0.34. For the practice long-lead materials identification, he/she plans "a procurement plan and schedule exists for long-lead materials" (Level C and score of 0.37); for materials status database, the manager plans "there is a formal paper based system to track materials status" (Level C and score of 0.26); for the practice materials delivery schedule, "materials delivery is planned early in the project and is integrated with a project schedule" has been planned (Level C and score of 0.30); for materials inspection process, "materials inspection is made for all items delivered to the site, there is a lack of organization of the inspection process and materials are not separated into stages of the receipt process nor does it record the location of the materials, and the materials are not marked for tracking" is chosen

285

(Level D and score of 0.42); and for the practice materials inspection team, the manager plans "materials inspections are performed by project manager or another worker rather than materials inspection team" (Level D and score of 0.34).

By using a similar procedure, based on the chosen levels of implementation of each practice, the equivalent score for other practices are computed, and the total score for the particular project is found to be 15 (Table 9-8). Then, by using the sigmoid graph of the overall management practices (Figure 9-7), the probability of exceeding PF=0.90 is 17%. This implies that the actual productivity could be low and the risk of occurring project delay is high. Thus, the project manager should plan to implement higher levels of the management practices and read the probability from the graph until the desired level of productivity is achieved. Therefore, the model and the scoring tools can assist the project manager in planning suitable management practices that are to be implemented on a certain multi-storey building project.

Table	9-8	Illustration	of the	use of the	Scoring	Tools of the	Construction	Management	Practices
	~ ~	111000000000000000000000000000000000000	· · · · · · · ·			10010 01 010	001101101011		

Construction Management Practices	Level	Verbal Description	Score
Procurement Plans for Materials	C	A procurement plan and schedule exists only for large materials and	0.34
Long Lond Materials Identification	C	costly items.	0.27
Materials Status Database	C	There is a formal paper based system to track materials status.	0.37
Materials Delivery Schedule	С	Materials delivery is planned early in the project and is integrated with a project schedule.	0.30
Material Inspection Process	D	Materials inspection is made for all items delivered to the site. There is a lack of organization of the inspection process and materials are not separated into stages of the receipt process nor does it record the location of the materials. The materials are not marked for tracking.	0.42
Materials Inspection Team	D	Materials inspections are performed by project manager or other worker rather than materials inspection team.	0.34
Construction	Materia	als Management Practices Sub-Total Score	2.03
Equipment	В	Construction equipment procurement plan is not prepared for this building project.	0.14
Construction Equipment Productivity Analysis	А	Construction equipment is not used in this building project.	0.00
Construction Equipment Maintenance	В	Construction equipment maintenance is not planned for this building project.	0.16
Construction E	Equipme	ent Management Practices Sub-Total Score	0.30
Integrated Schedule	D	Developing a schedule with resources present but not linked to earned percent complete progress from associated deliverables per activity.	0.39
Work Schedule Strategies	D	Strategy considers multiple work schedules considering critical and near critical activity sequences.	0.44
Schedule Control	D	Consistent follow-up to update schedules periodically, conduct critical path analysis, and prepare progress narrative as required. There is effective team participation in schedule updates. Quantity reports are regularly performed. Upon request, or as the project requires, may include any of the following: change management analysis, risks assessment analysis, date variance analysis, communication with material suppliers to ensure material will arrive on site as per the plan.	0.46
Dynamic Site Layout Plan	С	The project team examines the project schedule and assesses when Temporary Facilities (TF) will be brought in.	0.33
Traffic Control Plan	С	The project has some traffic control plans, and it is used on a reactive basis.	0.34
Site Security Plan	С	There is control during enty and exit from the site, but does not have any other formal security system throughout the site.	0.29
Machinery Positioning Strategy	С	Heavy rigging and lifting studies are accomplished on all critical lifts including evaluation of the machinery, rigging selection, and crane location.	0.33
Project start-up plan	С	A basic start-up plan has been developed with input from the project participants, but the plan has not been implemented.	0.34
Project Completion Plan	С	The project has a handover procedure that defines the parameters of project completion and delineates the requirements for the handover.	0.33
Innovations and New Technologies	C	The project does not have a formal program for the investigation of innovations and new technologies. Implementation of innovations and new systems will only occur after the industry-wide implementation.	0.25
Management Pract	tices Re	lated to Construction Methods Sub-Total Score	3.51
Short Interval Plan	D	equipment, labour, and project information required to complete each task, and activities in the project schedule are not resource loaded.	0.53
Well-Defined Scope of Work	D	Execution is controlled with a master schedule. The design is completed and buildability review has been performed.	0.58
Use of Software in Planning Work Packages	С	The project uses a software system to generate and track work packages. The system is not integrated. Materials, drawing and other information are entered manually. Percent complete is also entered mually by reviewing the statust of workpackage.	0.31
Dedicated Planner	С	Single planner or multiple personnel who prepare initial project schedule and coordinate materials procurement process are assigned.	0.24

287

Continuation of Table 9-8				
Construction Management Practices	Level	Verbal Description	Score	
Construction Work Packages	D	All information required for execution of each Construction Work Package (CWP) has been addressed. The information in each sections displays no in-denth consideration.	0.51	
Buildability Review	D	The contractor has prepareed a detailed schedule for all phases of construction after conducting detailed buildability review.	0.53	
Utilities Alignment	С	The project has a review process for working in areas surrounding utilities. The process includes list of utilities, locations, and warning signs.	0.30	
Contract Types	D	The project has a review process for the contract types. It considers fee structure and financing options. Risks associated with different contract types are also considered. Cost benefit analysis for different types of contracts are performed.	0.46	
Model Development	С	A 4D Model has been established for the project.	0.22	
Regulatory Requirement	C	Initial investigations based on the type of permits; timeline to get the permit approved; the validity period of the permit; the authority issuing the permit; pre-inspection requirements; and fees have been conducted.	0.34	
Pre-constructi	on Pha	se Management Practices Sub-Total Score	4.02	
Crew Composition	С	Crew formation is addressed after the commencement of the	0.31	
Skills Assessment and Evaluation	С	Skills assessment is addressed after the beginning of the construction this project.	0.30	
Employees Training	С	Employees Training is addressed on the jobsite after the beginning of the project.	0.30	
Career Development	C	The organization does not have a formal career development plan for employees, but the management will discuss future plans with them infromally.	0.28	
Non-Financial Incentive Programs	С	The organization has an informal recognition programs that will recognize employees occasionally, but not in a formal manner.	0.26	
Financial Incentive Programs	С	The organization has an informal incentive program that provides incentives occasionally, but not in a formal manner.	0.27	
Social Activities	С	The organization does not formally plan social activities for the employees, and there is only a yearly organization wide social activity.	0.27	
Stability of Organizational Structure	С	The key professionals are named or defined in the contract.	0.33	
Clear Delegation of Responsibility Retention Plan for Experienced Personnel	C C	There is simple and very formal delegation of responsibility. Each project manager is responsible for retention of his workers.	0.33 0.31	
Exit Interviews	С	There is exit interview for key personnel only.	0.23	
Human Res	ource	Management Practices Sub-Total Score	3.18	
Housekeeping Safety and Health Policy	D	Housekeeping occurs on a bi-weekly basis. There is health and safety policy at company and project levels. The policy is documented and publicized; it deals with the health and safety of workers, and it is periodically updated based on the organizational and industry feedback.	0.53 0.57	
Safety and Health Plan	D	Most but not all accident reduction techniques are utilized on the project.	0.55	
Safe Work Method Statement	D	Safe Work Method Statement (SWMS) is prepared and utilized for most tasks on this project.	0.09	
Hazards Analysis	D	Hazards are identified for most tasks that are not covered in SWMS.	0.10	
Safety and Health Training	D	There is project specific induction regarding personal protective equipment, housekeeping and access to site, and others. Supervisors receive additional induction on safety meetings, first aid and medical treatment processes. Some employees should pass fitness for duty testing prior to starting construction of some tasks.	0.07	
Toolbox Safety Meetings	D	The project conducts a weekly meeting at or near breaks. Meetings reiterate job site safety rules.	0.08	
Health and Safety Practices Sub-Total Total C	Score onstru	ction Management Practices Score	2.00	



Figure 9-7 Illustration of the Application of the Probability Graph of the Overall Construction Management Practices

The project manager can also fix the desired probability of exceeding PF=0.90, read the equivalent score, and plan appropriate practices for a certain multi-storey building project. For instance, for 99% chance of exceeding the baseline PF, the corresponding construction management practices score is 31 (Figure 9-7). After reading the score from the graph, the management practices can be planned as shown in Table 9-9. For example, the implementation level of the practice long-lead materials identification can be chosen as Level F or "a separate procurement plan and schedule exists for long-lead materials. There is an established protocol for identifying reputation of potential vendors for the long-lead materials. The plan identifies necessary equipment and onsite resources to support the delivery of the long-lead materials. The procurement schedule is integrated with a project information system that automatically updates the practices can be planned as shown in Table 9-9 to get the overall score of 31. Thus, the project managers of a multi-storey building can follow similar procedures to plan suitable practices for their projects.

CHAPTER 9

Construction Management Practices	Level	Verbal Description	Score
Procurement Plans for Materials	F	A procurement plan includes all materials and consumables. It identifies necessary equipment and onsite resources to support delivery. There is also an established protocol for identifying reputation of potential vendors. The procurement schedule is integrated with a project information system that automatically updates the procurement schedule as the construction schedule changes.	0.85
Long-Lead Materials Identification	F	A separate procurement plan and schedule exists for long-lead materials. There is an established protocol for identifying reputation of potential vendors for the long-lead materials. The plan identifies necessary equipment and onsite resources to support the delivery of long-lead materials. The procurement schedule is integrated with a project information system that automatically updates the procurement schedule as the construction schedule changes.	0.92
Materials Status Database	Е	An available software is used to track materials, but it is only integrated	0.53
Materials Delivery Schedule	F	Internally with the company's project control systems. Materials delivery is planned early in the project and is integrated with a project schedule. The schedule is automatically updated on receipt of new information as procurement proceeds. The schedule is automatically linked with procurement, materials management, and overall project scheduling systems. Materials delivery planning and management is completely integrated with other automated project processes including automated materials tracking throughout the supply chain.	0.75
Material Inspection Process	E	The materials inspection is made for all items delivered to the site. The inspection is conducted during manufacturing and onsite. Materials inspections are organized immediately upon delivery of materials, the materials are verified if they conform to the standards, and they are arranged for on-site tracking.	0.56
Materials Inspection Team	D	Materials inspections are performed by a project manager or another worker rather than materials inspection team.	0.34
Construct	ion Mai	terials Management Practices Sub-Total Score	3.95
Procurement Plan for Construction Equipment	D	Construction equipment procurement strategies such as hiring or leasing are prepared for this building project. There is a procedure for identifying	0.41
Construction Equipment Productivity Analysis	D	Construction equipment requirements are planned and tracked but they are not tied to a schedule. The usage is tracked against budget only. Regular meeting is conducted to assess the requirement of construction equipment.	0.37
Construction Equipment Maintenance	E	Construction equipment maintenance record is linked to individual construction equipment, and maintenance is centrally scheduled and administered. A computer program is used to record and administer equipment maintenance information such as the required and accomplished maintenance logs and usage logs.	0.63
Constructi	on Equi	pment Management Practices Sub-Total Score	1.41
Integrated Schedule	Е	Developing a schedule with resources present but not linked to earned percent complete progress from associated deliverables per activity. Resources are updated to reflect current work content or quantity adjustments.	0.53
Work Schedule Strategies	Е	Strategy considers multiple work schedules considering critical and near critical activity sequences. It also considers the potential impact on worker fatigue, supervision, safety, and absenteeism.	0.58
Schedule Control	F	The schedule is rigorously updated based on manual input of quantity reports, critical and near critical path is analysed, progress narrative is prepared and there is effective team participation in schedule updates. Quantity reports rigorously done by trained individual(s). Material suppliers routinely contacted to track the status of material delivery dates. Change management analysis, risks assessment analysis, date variance analysis may be conducted as a project requires. There is progress tracking using 3D imaging and other techniques.	0.77
Dynamic Site Layout Plan	F	The project team examines the project schedule and assesses when Temporary Facilities (TF) will be brought in, analyses what sizes of the TFs will be needed prior to the start of the project, and considers the addition and removal of TFs at different stages of the project. The team also analyses the layout of the project including where the different parties will be working and place their TFs in the optimum location in order to limit travel time to and from TFs.	0.82

Continuation of Table 9-9				
Construction Management Practices	Level	Verbal Description	Score	
Traffic Control Plan	F	The project has a traffic control plan and equipment for all times of the day including trained persons for traffic control. There is a trained traffic control supervisor. The project has an approved contingency plan to accommodate unexpected situations and has designed and constructed alternate arrangements for traffic such as detours and flyovers.	0.85	
Site Security Plan	D	out procedure and security guards at every gate. The site has implemented security measures to ensure the preservation of company assets. Protocols have been identified for searching individuals and their personal property. Searching are conducted randomly.	0.44	
Machinery Positioning Strategy	F	Heavy rigging and lifting studies are accomplished on all critical lifts including evaluation of the machinery, rigging selection, and crane location. There is thorough use of 3D modelling/visualization to locate machinery.	0.82	
Project start-up plan	F	A start-up plan that identifies the duties and responsibilities of each stakeholder has been developed. The plan has cost analysis and detailed scheduling components, it is well communicated to all the stakeholders, and the plan will be implemented on the project by incorporating feedbacks from the stakeholders or project participants.	0.85	
Project Completion Plan	F	The project has a formal handover process that defines the necessary documentation, parameters of completion and other issues to assure proper handover of a project. The procedure has been reviewed and agreed by the stakeholders. The plan is approved by project management team and is reviewed for applicability during all phases of the handover process.	0.83	
Innovations and New Technologies	D	The organization has an informal program for the investigation of innovations, and they will investigate the feasibility of the new technologies on a regular basis.	0.38	
Management Pr	actices	Related to Construction Methods Sub-Total Score	6.87	
Short Interval Plan	F	Short interval plan indicates the required materials, tools and equipment, labour, and project information required to complete each task, and activities in the project schedule are resource loaded. The short interval plan considers the effects of congestion. Alternate plans such as modified equipment schedules, shortened shifts, lengthened shifts, or added shifts in case the original plans did not work are prepared.	0.88	
Well-Defined Scope of Work	F	The design is completed and buildability review has been performed; duration for the work package is estimated; material availability, testing and inspection requirements are defined; and budget and quantities are included in the scope of work.	0.96	
Use of Software in Planning Work Packages	F	The project uses a software system to generate and track work packages. The system automatically includes the drawings and material delivery status. It is updated regularly and automatically, and provides current design drawing information, updated status of materials, implementation schedule with durations and quantities, test and inspection status, and percent complete. The work nackage status is undated electronically.	0.76	
Dedicated Planner	Е	Single planner or multiple personnel who prepare initial plan and coordinate with procurement are assigned. The planner initiate and maintain communication regarding the delivery and inspection of materials and continuously update the schedule. There is onsite inspection to track consumption of materials as necessary.	0.48	
Construction Work Packages	F	All information required for execution of each Construction Work Package (CWP) has been addressed. All sections have received in-depth consideration. The construction work package can be performed easily based on the information documented for each CWP.	0.86	
Buildability Review	F	The contractor has prepareed a detailed schedule for all phases of construction after conducting detailed buildability review. The owner is involved in buildability review process. All buidability reviews and schedules are completed before construction.	0.88	
Utilities Alignment	F	The project has a review process for working in areas surrounding utilities. The process includes list of utilities, locations, and warning signs. The necessary city council approvals have been obtained and utility adjustment plan is prepared. Communication with businesses and local community have been made. Arrangements for temporary shut offs and interruptions have been made. The physical markings described to mark the horizontal route of underground facilities related to utilities alignments have been integrated with other project schedules.	0.76	
Contract Types	F	The project has a review process for the contract types. It considers fee structure and financing options. Risks associated with different contract types are also considered. Cost benefit analysis for different types of contracts are performed. The labour market information and reports are also studied and considered before awarding contracts. The local conditions in the project area are studied and monitored regularly. The company has established protocols for selecting designers, material suppliers, and manufacturers.	0.77	

CHAPTER 9

		Continuation of Table 9-9	
Construction Management Practices	Level	Verbal Description	Score
Model Development	D	A 4D Model has been developed for the project. The model is updated manually.	0.32
Regulatory Requirement	F	Initial investigations based on the type of permits, timeline to get the permit approved, the validity period of the permit, the authority issuing the permit, pre- inspection requirements, and fees have been conducted. Permit requirements are linked to schedules. A system is established to track the acquisition and validity period of each permit. The system is automatically updated based on continued updates of schedule and permit expiry dates.	0.85
Pre-const	ruction I	Phase Management Practices Sub-Total Score	7.52
Crew Composition	Е	Crew formation is addressed before the commencement of construction works. Experience and knowledge of employees, job requirements, and location of the project are considered while forming crew. The performance of the crew is assessed after completion of each task and corrective measures are taken accordingly.	0.62
Skills Assessment and Evaluation	D	Skills assessment is addressed before the beginning of construction work; and the previous experience, skill and knowledge of workers are considered during evaluation.	0.44
Employees Training	Е	Training is provided to an employee when he/she begins working for the company and, if needed, extra training will be conducted on the job site. A worker is certified to work in or supervise specific trade.	0.60
Career Development	D	The organization has a formal career development plan for employees, but it only addresses short term career developments.	0.43
Non-Financial Incentive Programs	D	The organization has a formal recognition program that provides recognition on long term basis.	0.39
Financial Incentive Programs	D	The organization has a formal incentive program that provides incentives on long term basis.	0.40
Social Activities	D	The organization formally plans a social activity once or twice a year in which the project managers will attend, along with a yearly organization wide social activity.	0.40
Stability of Organizational Structure	Е	The key professionals are named or defined in the contract. It is also stated that they cannot be changed without notice and prior approval.	0.65
Clear Delegation of Responsibility	Е	The delegation of responsibility is formal, but differing between technical and administrative staff.	0.67
Retention Plan for Experienced Personnel	Е	Incentives such as employee training and others are available. Employee training is a requirement for junior staff. The employer makes available a list of opportunities for the next project.	0.61
Exit Interviews	С	There is exit interview for key personnel only.	0.23
Human	n Resour	ce Management Practices Sub-Total Score	5.44
Housekeeping Safety and Health Policy	F	Major travel paths are organized and clean. Housekeeping occurs weekly. There is health and safety policy at company and project levels. The policy is documented and publicized; it deals with the health and safety of workers, and it is periodically updated based on the organizational and industry feedback. The organization screens project participants for their health and safety programs and chooses those with records of good performance. Health and safety policy is integrated with procurement processes. Money is budgeted for the	0.71
		construction projects to address various health and safety issues.	
Safety and Health Plan	F	very proactive approach towards safety.	0.92
Safe Work Method Statement	F	Safe Work Method Statements (SWMS's) are utilized daily on the project for all tasks and the project team prepares additional SWMS's as the task changes.	0.91
Hazards Analysis	F	Hazards are identified for all tasks (high-risk and low-risk) and incorporated into the project's task specific safety planning process.	0.92
Safety and Health Training	E	There is project specific induction regarding personal protective equipment, housekeeping and access to site, and others. Supervisors receive additional induction on safety meetings, first aid and medical treatment processes. Some employees should pass fitness for duty testing prior to starting construction of some tasks. The orientation addresses zero accidents philosophy and general project safety rules.	0.71
Toolbox Safety Meetings Health and Safety Practices Sub-Total	F Score	The project conducts weekly meetings at a prearranged time, generally at the start of the day. Meetings address current job status, hazards posed by upcoming project activities, and corrective actions taken. It reviews recorded injuries and near misses, and reiterates job site safety rules and expectations. Time is set aside during the meeting for interactive discussion and allows worker feedback. The date of the meeting varies or can be conducted daily.	0.91 6.03
То	tal Cone	truction Management Practices Score	31.22

The logistic regression models of the six categories of the construction management practices and productivity can also be used to estimate the probability of exceeding baseline productivity factors. The discussions on the application of the logistic regression models of the six categories of the construction management practices are made in the following sub-sections.

Application of the Logistic Regression Model of the Construction Materials Management Practices and Productivity

Consider the construction materials management practices score of a project indicated in Table 9-8. The total score of the materials management practices is 2 (Table 9-8), and the equivalent probability of exceeding PF=1.05 is 5% (Figure 9-8). This implies that the actual productivity of a project can be less than the planned productivity, and the project could be delayed. Thus, corrective actions should be taken. The action could be increasing the levels of implementations of the construction materials management practices that are more important to enhance productivity in multi-storey building construction projects.





Application of the Logistic Regression Models of the Construction Equipment Management Practices and Productivity

To illustrate how the logistic regression model of the construction equipment management practices and productivity is used, the score of the example project in Table 9-8 is used. Accordingly, the total score of the construction equipment

management practices for the project is 0.30 (Table 9-8), and the probability of exceeding PF=0.90 is approximately 43% (Figure 9-9). This implies that the actual productivity is less than the planned productivity and the possibility of occurrence of the project delay is high. However, the project manager can make other trials by increasing the level of implementation of the construction equipment management practices to maximise the chance of completing the project early.



Figure 9-9 The Use of the Sigmoid Graph of the Construction Equipment Management Practices

Application of the Logistic Regression Models of the Management Practices Related to Construction Methods and Productivity

As shown in Table 9-8, based on the project manager's planned levels of the management practices related to construction methods, the total score for this category is 3.51. The sigmoid graph which is plotted using the logistic regression model of the management practices related to construction methods can be used to estimate the probability of exceeding PF= 0.90. Accordingly, the probability value is 0.40 (Figure 9-10). This implies that the actual productivity during the construction could be less than planned productivity and the risk level of a project delay will be higher. On the other hand, the project manager can increase the chance of completing a project on or before the contract date by increasing the level of implementation of the practices as shown in Table 9-9 above.



Figure 9-10 Application of the Probability Graph of Management Practices Related to Construction Methods

Application of the Logistic Regression Models of the Preconstruction Phase Management Practices and Productivity

For the project considered in Table 9-8, the total score of the pre-construction phase management practices is 4 and probability of exceeding PF=1.00 is 30% (Figure 9-11) which implies that with this score, there could be a high chance of incurring project delay. Thus, the project manager needs to increase the level of implementation of the preconstruction phase management practices, read the probability from the graph and plan appropriate practices before commencing the construction of the project.



Figure 9-11 The Use of the Probability Plot of the Preconstruction Phase Management Practices

Application of the Logistic Regression Models of the Human Resource Management Practices and Productivity

The score of the human resource management (HRM) practices for the example project in Table 9-8 is 3.18, and the equivalent probability of exceeding PF=0.90 can be read from the probability graph of the HRM practices. Accordingly, the probability value is about 26% (Figure 9-12). This indicates that the actual productivity of the particular project can be less than the planned productivity and the chance of occurrence of the project delay is high.



Figure 9-12 Application the Sigmoid Graph of the HRM Practices

Application of the Logistic Regression Models of the Safety and Health Practices and Productivity

For the example project considered in Table 9-8, the total safety and health practices score is 2.0, and the probability of exceeding PF=0.90 is 0.01% (Figure 9-13) which implies that the chance of occurrence of project delay is high. Thus, the project manager should take corrective actions prior to the commencement of the construction.



Figure 9-13 Application of the Probability Graph of Safety and Health Practices

9.4.2 The Application of the Construction Management Practices Scoring Tools and the Linear Regression Models during Project Planning Phase

The management practices' scoring tools and the linear regression model of productivity and the practices can be used to investigate the proportion of productivity factor explained by the construction management practices. In this research, it is found that the management practices can explain about 48% of the variation in productivity factor and the relationship between productivity and the management practices is statistically significant.

The productivity factor of a certain multi-storey building project can be determined based on a measured value of the management practices by using the linear regression model of productivity and the practices. For instance, the overall score of the project considered in Table 9-8 is 15.03, and the productivity factor of the specific project can be computed using Equation 9-1 (linear regression model of overall management practices).

$$PF_{ex} = 0.02CMP_{ex} + 0.58$$
 Equation 9-1

In the equation, PF_{ex} is the productivity factor of the example project, and CMP_{ex} is the overall score of the construction management practices for the example project or 15.03. By inserting the score of the project in Equation 9-1, the productivity factor of the particular project is found to **0.88**. This implies that the actual productivity of the

project is 0.88 times the planned productivity and the project could be delayed. In other words, the actual completion time (duration) could be (1/0.88 = 1.14) times the planned duration. For example, if the planned completion time of the project is 300 days, then the actual completion time could be $1.14 \times 300 = 342$ days. The expected project delay could be 342 - 300 = 42 days. Thus, the project manager should make another trial, predict productivity factor, estimate the project delay and make the necessary decisions regarding the implementation levels of the management practices. Therefore, the linear regression model and the scoring tool can assist the project manager to plan appropriate management practices for the proposed project.

9.4.3 The Use of the Scoring Tools during Construction Phase

The scoring tools can be used to monitor the implementation of the management practices. Consider the six management practices which are planned by a contractor's project manager in Table 9-10. The project manager planned the practice material inspection process as "the materials inspection is made for all items delivered to the site. The inspection is conducted during manufacturing and onsite. Materials inspections are organised immediately upon delivery of materials, the materials are verified if they conform to the standards, and they are arranged for on-site tracking or level E" (Table 9-10). However, during the construction phase, the manager monitors the implementation of the particular practice and found that the implementation level is "materials inspection is made for all items delivered to the site. There is a lack of organisation of the inspection process and materials are not separated into stages of the receipt process nor does it record the location of the materials and mark the materials for tracking or Level D" (Table 9-10). Thus, since the implemented practice is different from the planned practice corrective action is required for the practice 'materials inspection process' (Table 9-10). If the corrective measures are not taken, the chance of achieving the planned productivity level would be low.

However, for the practice 'traffic control plan,' the manager planned to implement "the project has a traffic control plan and equipment for all times of the day including trained persons for traffic control. There is a trained traffic control supervisor. The project has an approved contingency plan to accommodate unexpected situations and has designed and constructed alternate arrangements for traffic such as detours and flyovers or Level F," and while controlling the implementation of the practice, a similar result was found ²⁹⁸

(Table 9-10). Thus, corrective action is not required for this practice. For other management practices, by using a similar procedure, the project manager can take corrective measures after comparing the planned and implemented practices. Therefore, the scoring tools can be used to control the implementation of management practices.

Construction		Planned Practices		Implemented Practices		
Management Practices	Level	Verbal Description	Level	Verbal Description	Corrective Action	
Material Inspection Process	E	The materials inspection is made for all items delivered to the site. The inspection is conducted during manufacturing and onsite. Materials inspections are organized immediately upon delivery of materials, the materials are verified if they conform to the standards, and they are arranged for on-site tracking.	D	Materials inspection is made for all items delivered to the site. There is a lack of organization of the inspection process and materials are not separated into stages of the receipt process nor does it record the location of the materials and mark the materials for tracking.	Required	
Construction Equipment Maintenance	E	Construction equipment maintenance record is linked to individual construction equipment, and maintenance is centrally scheduled and administered. A computer program is used to record and administer equipment maintenance information such as the required and accomplished maintenance logs and usage logs.	С	Construction equipment is logged in simplified spreadsheet and maintenance is carried out by operator request.	Required	
Traffic Control Plan	F	The project has a traffic control plan and equipment for all times of the day including trained persons for traffic control. There is a trained traffic control supervisor. The project has an approved contingency plan to accommodate unexpected situations and has designed and constructed alternate arrangements for traffic such as detours and flyovers.	F	The project has a traffic control plan and equipment for all times of the day including trained persons for traffic control. There is a trained traffic control supervisor. The project has an approved contingency plan to accommodate unexpected situations and has designed and constructed alternate arrangements for traffic such as detours and flyovers.	Not required	
Short Interval Plan	F	Short interval plan indicates the required materials, tools and equipment, labour, and project information required to complete each task, and activities in the project schedule are resource loaded. The short interval plan considers the effects of congestion. Alternate plans are prepared such as modified equipment schedules, shortened shifts, lengthened shifts, or added shifts in case the original plans did not work.	D	Short interval plan indicates the required materials, tools and equipment, labour, and project information required to complete each task, and activities in the project schedule are not resource loaded.	Required	
Crew Composition	E	Crew formation is addressed before the commencement of construction works. Experience and knowledge of employees, job requirements, and location are considered. The performance of the crew is assessed after completion of each task and corrective measures are taken accordingly.	E	Crew formation is addressed before the commencement of construction works. Experience and knowledge of employees, job requirements, and location are considered. The performance of the crew is assessed after completion of each task and corrective measures are taken accordingly.	Not required	
Housekeeping	Е	Major travel paths are organized and clean. Housekeeping occurs weekly.	D	Housekeeping occurs on a bi-weekly basis.	Required	

T-11. 0 10 TL	TI 41 C	T 1 + - M + + +	Constant Month	D
Table 9-10 The	Use the Scoring	T LOOL TO MODITOR THE	Construction Manag	rement Practices
10010 / 10 110	obe the beorning		Construction munus	

9.4.4 The Application of the Scoring Tools during the Project Closure Phase

The management practices scoring tools can also be used during the project closure phase. The effectiveness of the implemented practices can be evaluated, and lessons can be learned from the completed project. For instance, consider the project manager has planned and implemented the practices indicated in Table 9-9. The overall score is 31.22, and the equivalent productivity factor (PF) of the project using the linear regression model's equation (Equation 9-1) can be computed as $0.02 \times 31.22 + 0.58 =$

1.20. This implies that the actual project duration could be 1/1.20 or 0.83 times the planned duration. For instance, if the planned duration is 400 days, the expected actual duration would be 332 days (estimated at the start of the project). However, after the project is completed (during evaluation of project performance or project closure phase), the actual duration, for instance, is found to be 380 days. This shows that the implemented practices might not be associated with PF=1.20 and higher levels of the practices should be planned for the next project. Thus, lessons can be learnt from the completed project.

9.5 Summary

This chapter provided the discussions of the findings obtained to achieve Objective 1 or "to identify and prioritize the management practices that have the potential to enhance productivity in multi-storey building construction projects in the context of Victoria State, Australia," Objective 2 or "to refine and validate scoring tools for measuring, planning, monitoring and evaluating the management practices that could improve productivity in multi-storey building construction projects," and Objective 3 or "to build and validate a tool that can be used to predict the probability of exceeding baseline productivity factor based on a score of the management practices."

The management practices and their relative importance in improving the productivity of the multi-storey building construction projects in the context of Victoria State, Australia have been discussed. The findings of this study were compared with the results of previous studies, and the possible reasons for the variations were explained. The relationships among the management practices in the six categories (construction materials management practices, construction equipment management practices, management practices related to construction methods, human resource management practices, pre-construction phase management practices, and safety and health practices) were described. The inter-relationships among the practices within their categories were also explained.

The applications of the scoring tools of the management practices and the logistic regression models during the project planning phase were discussed by using an example project. The productivity factor of the example project was predicted using the linear regression model, and the project delay associated with the planned practices was estimated. The application of the scoring tools to monitor the implementation of the $\frac{300}{300}$

planned management practices was explained by providing an example project. Finally, the uses of the scoring tools to evaluate if the implemented practices are associated with the predicted productivity factor were also discussed with the help of an example. The next chapter presents the summary and conclusions of the study.

CHAPTER TEN

10 SUMMARY AND CONCLUSION

10.1 Introduction

This chapter provides the summary of the key findings of the research. It also addresses the contributions of the study, the limitations, and the recommendations for future research. In Section 10.2, the summaries of the findings of the study are described. In this section, the objectives of the study are restated, and the main findings are explained; and the scoring tools of the construction management practices, the probability graphs and the linear regression model of the management practices and productivity are presented. Section 10.3 discusses the theoretical and practical contributions of the research. Finally, the limitations of the study and directions for future studies are described in Section 10.4.

10.2 Summary of Findings

This section summarizes the findings obtained to achieve the three objectives of the study which are (1) to identify and prioritize the management practices that have the potential to enhance productivity in multi-storey building construction projects in the context of Victoria State, Australia, (2) to refine and validate scoring tools for measuring, planning, monitoring and evaluating the management practices that could improve productivity in multi-storey building construction projects, and (3) to build and validate a tool that can be used to predict the probability of exceeding baseline productivity factor based on a score of the management practices.

10.2.1 The Prioritized Construction Management Practices Influencing Productivity in Multi-Storey Building Projects

Overall, 47 construction management practices that have the potential to improve productivity in multi-storey building projects in the context of Victoria State, Australia have been identified and prioritised. These comprise of six construction materials management practices, three construction equipment management practices, 10 management practices related to construction methods, 10 pre-construction phase management practices, 11 human resource management practices, and seven safety and health practices. Safety and Health practices are found to be the most important practices among the six main categories of the construction management practices, and 302

all the seven health and safety practices are ranked with the top ten. The practices welldefined scope of works, safety and health policy, and safety and health plan are found to be the three most important management practices (Table 10-1).

Table 10-1 The Prioritized Construction Management Practices

Construction Management Practices	RII	Overall Rank	Construction Management Practices	RII	Rank within Category
Well-Defined Scope of Work	0.96	1	Construction Materials Management Practices (CMMP)		
Safety and Health Policy	0.95	2	Long-Lead Materials Identification	0.92	1
Safety and Health Plan	0.92	3	Procurement Plans for Materials	0.85	2
Hazards Analysis	0.92	3	Materials Delivery Schedule	0.75	3
Long-Lead Materials Identification	0.92	5	Material Inspection Process	0.70	4
Safe Work Method Statement	0.91	6	Materials Status Database	0.66	5
Toolbox Safety Meetings	0.91	7	Materials Inspection Team	0.57	6
Housekeeping	0.89	8	Average weight of CMMP	0.74	
Sheet Internal Dian	0.89	0	Construction Equipment Maintenance	0.70	1
Short Interval Plan	0.88	10	Construction Equipment Maintenance	0.79	1
Construction Work Packages	0.86	10	Construction Equipment Productivity Analysis	0.01	2
Traffic Control Plan	0.85	13	Average weight of CFMP	0.73	5
Project Start -up Plan	0.85	13	Management Practices Related to Construction Methods (MPRCM)	0.75	
Regulatory Requirement	0.85	13	Traffic Control Plan	0.851	1
Procurement Plans for Materials	0.85	16	Project start-up plan	0.851	1
Clear Delegation of Responsibility	0.84	17	Machinery Positioning Strategy	0.826	3
Project Completion Plan	0.83	18	Project Completion Plan	0.821	4
Machinery Positioning Strategy	0.82	19	Dynamic Site Layout Plan	0.815	5
Dynamic Site Layout Plan	0.82	20	Schedule Control	0.769	6
Stability of Organizational Structure	0.82	20	Work Schedule Strategies	0.733	7
Construction Equipment Maintenance	0.79	22	Site Security Plan	0.733	7
Contract Types	0.77	23	Integrated Schedule	0.656	9
Schedule Control	0.77	24	Innovations and New Technologies	0.636	10
Crew Composition	0.77	24	Average weight of MPRCM	0.77	
Use of Software in Planning Work Packages	0.76	26	Pre-Construction Phase Management Practices (PCPMP)		
Retention Plan for Experienced Personnel	0.76	26	Well-Defined Scope of Work	0.959	1
Utilities Alignment	0.76	28	Short Interval Plan	0.877	2
Materials Delivery Schedule	0.75	29	Buildability Review	0.877	2
Employees Training	0.75	29	Construction Work Packages	0.856	4
Skill Assessment and Evaluation	0.74	31	Regulatory Requirement	0.851	5
Work Schedule Strategies	0.73	32	Contract Types	0.774	6
Site Security Plan	0.73	32	Use of Software in Planning Work Packages	0.764	7
Career Development	0.71	34	Utilities Alignment	0.759	8
Material Inspection Process	0.7	35	Dedicated Planner	0.61	9
Procurement Plan for Construction Equipment	0.69	36	Model Development	0.538	10
Financial Incentive Programs	0.67	37	Average weight of PCPMP	0.79	
Social Activities	0.67	37	Human Resource Management Practices (HRMP)		
Materials Status Database	0.66	39	Clear Delegation of Responsibility	0.836	1
Integrated Schedules	0.66	40	Stability of Organizational Structure	0.815	2
Non-Financial Incentive Programs	0.65	41	Crew Composition	0.769	3
Innovations and New Technologies	0.64	42	Retention Plan for Experienced Personnel	0.764	4
Dedicated Planner	0.61	43	Employees Training	0.749	5
Construction Equipment Productivity Analysis	0.61	44	Skill Assessment and Evaluation	0.738	6
Exit Interviews	0.57	45	Career Development	0.708	7
Materials Inspection Team	0.57	46	Financial Incentive Programs	0.672	8
Model Development	0.54	47	Social Activities	0.672	8
			Non-Financial Incentive Programs	0.651	10
			Exit Interviews	0.574	11
			Average weight of HRMP	0.72	
			Safety and Health Practices (SHP)	0.040	
			Salety and Health Policy	0.949	1
			Salety and Health Plan	0.923	2
			nazaius Ailalysis Sofe Work Method Statement	0.923	2
			Jaie work Method Statement	0.913	4
			Housekeeping	0.908	6
			Safety and Health Training	0.887	6
			Average weight of SHP	0.91	5

The findings of the correlation analyses revealed that all the 47 construction management practices are positively associated with productivity. The output of linear regression analysis also showed that there is a positive and statistically significant relationship between productivity and the management practices. The findings of Friedman test indicate that the practices 'well-defined scope of work,' 'safety and health policy,' 'safety and health plan,' 'hazard analysis,' 'long-lead materials identification,' 'safe work method statement' and 'toolbox safety meetings' are equally important to improve productivity in multi-storey building projects.

Long-lead materials identification, procurement plans for materials and materials delivery schedule are found to be the three most important practices in the construction materials management practices category (Table 10-1). The results of the study revealed that high levels of implementation of the construction materials management practices are associated with low project delays and high project costs. Annual turnover, company experience and company sizes are positively correlated with construction materials management practices, but the associations are not significant.

Construction equipment maintenance, construction equipment procurement plan, and construction equipment productivity analysis are found to be the three construction equipment management practices that positively influence productivity in multi-storey building projects (Table 10-1). The study revealed that high levels of implementation of the construction equipment management practices are associated with high productivity and low project delays. The levels of implementation of the construction equipment management practices and company size. Annual turnover and company experience are positively correlated with construction equipment management practices, but the associations are not statistically significant.

Traffic control plan, project start-up plan, machinery positioning strategy, project completion plan, and dynamic site layout plan are found to be the top five most important practices in the category of management practices related to construction methods (Table 10-1). The findings of Friedman and Wilcoxon tests indicated that there are no discernible differences among the top five practices. Hence, these practices should be implemented jointly to improve productivity in multi-storey building projects. The study revealed that high levels of implementation of the management practices related to construction methods are associated with low project delays and high ₃₀₄

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

productivity. The level of implementation of the management practices related to construction methods increases as project cost increases. Positive correlations among company's annual turnover, company size, company experience and the management practices related to construction methods are found. However, these relationships are not statistically significant.

Among the 10 pre-construction phase management practices, well-defined scope of work, short interval plan, buildability review, construction work packages, and regulatory requirement are found to be the top five most important practices (Table 10-1). The results of Friedman and Wilcoxon tests showed that short interval plan, buildability review, construction work packages, and regulatory requirement are equally important. Thus, these practices need to be implemented jointly to improve productivity in multi-storey building construction projects. The correlation analysis revealed that higher levels of implementation of the pre-construction phase management practices are associated with higher productivity, higher project costs and lower project delays. Annual turnover has a very weak correlation with the pre-construction phase management practices. Company experience and company size are positively correlated with pre-construction phase management practices, but the relationships are not statistically significant.

Clear delegation of responsibility, stability of organizational structure and crew composition are found to be the three most important human resource management (HRM) practices (Table 10-1). The study revealed that there is no significant difference among the three practices, and the author suggested that they should be implemented jointly to enhance productivity in multi-storey building projects. The correlation analysis indicated that project delay is negatively correlated with HRM practices which imply that the increment in the level of implementation of the HRM practices could decrease the occurrence of delays in multi-storey building projects. The relationship between company size and HRM practices is positive and significant. This indicates that the level of implementation of HRM increases with company size. The project cost is strongly associated with HRM practices. Company experience and project turnover are positively correlated with HRM practices, but the relationships are not statistically significant.
Safety and health policy, safety and health plan, and hazard analysis are the three most important practices in the safety and health category (Table 10-1). It has been found that the seven safety and health practices are equally important and the author suggested the joint implementation of the seven safety and health practices to enhance productivity in multi-storey building construction projects in the context of Victoria State, Australia. The results of correlation analysis revealed that the level of implementation of the safety and health practices increases with project costs. Project delay is negatively correlated with safety and health practices, but the degree of association is weak and insignificant. This indicates that increasing the level of implementation of safety and health practices might not reduce the possibility of occurrence of project delay. Company experience and company size are positively correlated with safety and health practices, and the relationships are statistically significant. The finding implies that experienced and large construction companies implement higher levels of safety and health practices. Annual turnover and safety and health practices are positively correlated, but the relationship is not significant.

10.2.2 The Scoring Tools of the Construction Management Practices

The scoring tool which can be used to measure, plan, monitor and evaluate the 47 management practices have been refined and validated. The reliability test for the scoring tools was also conducted and findings show that the tools are reliable. The validated scoring tools are presented in Table 10-2 below. The baseline scores against which the users of the scoring tools assess the adequacy of their practices have been fixed and validated. The baseline scores can be used as a benchmark to assess if the planned or implemented management practices on a certain multi-storey building project would be associated with high or low productivity. Thus, contractors' project managers can use the scoring tools indicated in Table 10-2 to compute their projects' scores and compare with the baseline scores. Accordingly, multi-storey building projects with the overall construction management practices can be considered as less-productive projects, and projects with overall construction management practices can be regarded as high-score projects, and they could have higher productivity.

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

Table	10-2	The	Scoring	Tools	of the	Construction	Management	Practices
							<u> </u>	

	A. Construction Materials Management Practices Scoring Tools	
	A1. Procurement Plans for Materials	Weights
Level A	A procurement plan for materials is not applicable	0.00
Level B	There is no documented procurement plan for materials.	0.17
Level C	A procurement plan and schedule exists only for large materials and costly items.	0.34
Level D	Continuation of Level C, plus the plan includes all materials and consumables. Also, there is an established protocol for identifying reputation of potential vendors.	0.51
Level E	Continuation of Level D, plus the plan identifies necessary equipment and onsite resources to support delivery	0.68
Level F	Continuation of Level E, plus the procurement schedule is integrated with a project information system that automatically updates the procurement schedule as the construction schedule changes.	0.85
	A2. Long lead Materials Identification	
Level A	A separate procurement plan for long-lead materials is not applicable.	0.00
Level B	There is no documented procurement plan for long-lead materials.	0.18
Level C	A procurement plan and schedule exists for long-lead materials.	0.37
Level D	of the long-lead materials.	0.55
Level E	Continuation of Level D, plus the plan identifies necessary equipment and onsite resources to support delivery of long-lead materials.	0.73
Level F	Continuation of Level E, plus the long-lead materials procurement schedule is integrated with a project information system that automatically updates the procurement schedule as the construction schedule changes.	0.92
	A3. Materials Status Database	
Level A	Material status database is not applicable.	0.00
Level B	There is no formal paper-based system used to track materials status.	0.13
Level C	There is a formal paper-based system to track materials status.	0.26
Level D	Available software is used to track materials status, but it is not integrated with company's project control systems or used by other contractors.	0.40
Level E	Commercially available software is used to track materials status but it is only integrated internally with the company's project control systems.	0.53
Level F	Continuation of Level E, plus the software is used by all stakeholders in the materials supply chain system.	0.66
	A4. Materials Delivery Schedule	
Level A	Materials delivery schedule is not applicable.	0.00
Level B	There is no documented materials delivery schedule.	0.15
Level C	Materials delivery is planned early in the project and is integrated with a project schedule.	0.30
Level D	Continuation of Level C, plus the schedule is automatically updated on receipt of new information as the procurement proceeds.	0.45
Level E	Continuation of Level D, plus the schedule is automatically linked with procurement and overall project scheduling systems.	0.60
Level F	Continuation of Level E, plus materials delivery planning and management is completely integrated with other automated project processes including automated materials tracking throughout the supply chain.	0.75
	A5. Material Inspection Process	
Level A	The material inspection process is not applicable.	0.00
Level B	There is no materials inspection process.	0.14
Level C	Materials inspection is made for large items or costly items only.	0.28
Level D	Materials inspection is made for all items delivered to the site. There is a lack of organization of the inspection process and materials are not separated into stages of the receipt process. The locations of	0.42
	materials are not recorded. The materials are not marked for tracking.	
Level E	The inspection is conducted during manufacturing and onsite. Materials inspections are organized immediately upon delivery of materials; the materials are verified if they conform to the standards, specifications, and drawings; and they are arranged for on-site tracking. However, there is no separation of materials into stages of receipt.	0.56
Level F	Continuation of Level E, plus the process includes separation of material into categorical stages of the receipt process (e.g. awaiting inspection, scrap, and/or awaiting for shipment).	0.70

Table 10-2 continued

	A6. Materials Inspection Team	Weights
Level A	Materials inspection team is not applicable.	0.00
Level B	There is no materials inspection team.	0.11
Level C	There is a designated person who conducts materials inspection but no training and qualification of the person is specified.	0.23
Level D	Continuations of Level C, plus inspections are performed by a project manager or another worker rather than a team	0.34
Level E	Continuation of Level D, plus there is an inspection team who can inspect materials.	0.46
	Continuation of Level E, plus the members of the inspection team are experts at inspection processes and	
Level F	procedures. The team can adequately inspect the materials and understand the material specifications and	0.57
	standards. B. Construction Equipment Management Practices Scoring Tools	
	B1. Procurement Plan for Construction Equipment	
Level A	A procurement plan for construction equipment is not applicable for this building project.	0.00
Level B	Construction equipment procurement plan is not prepared for this building project.	0.14
Level C	Construction equipment procurement strategies such as hiring or leasing are prepared for this building project.	0.28
Level D	Continuation of Level C and there is a procedure for identifying reputation of potential equipment suppliers	0.41
Level E	Continuation of Level D, plus the plan identifies necessary equipment and onsite resources to support delivery.	0.55
Level F	Continuation of Level E and construction equipment schedule is integrated with other project schedules.	0.69
	B2. Construction Equipment Productivity Analysis	
Level A	Construction equipment is not used in this building project.	0.00
Level B	Construction equipment is utilized but requirements and usage are not planned and tracked.	0.12
Level C	Construction equipment requirements are planned and tracked but they are not tied to a project schedule. The usage is tracked against budget only.	0.24
Level D	Continuation of Level C and regular meeting is conducted to assess the requirement of construction equipment.	0.37
Level E	Continuation of Level D, and the construction equipment resource curves are drawn and resources are leveled	0.49
Level F	Continuation of Level E and equipment schedule is adjusted based on the audit report that shows equipment downtime and other information	0.61
	B3. Construction Equipment Maintenance	
Level A	Construction equipment maintenance is not applicable for this building project.	0.00
Level B	Construction equipment maintenance is not planned for this building project.	0.16
Level C	Construction equipment is logged in simplified spreadsheet and maintenance is carried out by operator request.	0.32
Level D	Continuation of Level C and equipment maintenance record is linked to individual construction equipment, and maintenance is centrally scheduled and administered.	0.47
Level E	Continuation of Level D and a computer program is used to record and administer equipment maintenance information such as required and accomplished maintenance logs and usage logs.	0.63
Level F	Continuation of Level E and maintenance schedule is electronically updated and maintenance due notices are automatically issued to the concerned parties.	0.79
	C. Management Practices Related to Construction Methods Scoring Tools	
	C1. Integrated Schedule	0.00
Level A	The use of an integrated schedule using critical path method (CPM) is not applicable.	0.00
Level B	The use of an integrated schedule using CPM has not been addressed.	0.13
Level C	Developing a schedule with no resources present and managing schedule status via duration or remaining duration but not linked to earned percent complete progress for associated deliverables per activity.	0.26
Level D	Developing a schedule with resources present but not linked to earned percent complete progress for associated deliverables per activity.	0.40
Level E	Continuation of Level D, plus resources are updated to reflect current work content or quantity adjustments.	0.53
Level F	Continuation of Level E, plus earned progress for the activity is based on measured or assessed work completed per deliverables per activity. Progress measurement is performed in application adapted specifically for each deliverable.	0.66

	Table 10-2 continued	
	C2. Work Schedule Strategies	Weights
Level A	The development of a work schedule strategy is not applicable	0.00
Level B	The development of a work schedule strategy has not been addressed	0.15
Level C	The strategy is based on a single work schedule be it either a straight time such as 36 hours per week	0.29
	schedule, overtime, or other work schedule strategies.	0.29
Level D	Strategy considers multiple work schedules considering critical and near critical activity sequences.	0.44
Level E	Continuation of Level D, plus strategy considers the potential impact on worker fatigue, supervision, safety,	0.58
	and absenteeism.	
Level F	Continuation of Level E, plus each potential strategy's impact is analysed for manpower density and	0.73
	congestion at an area or sub-area level.	
Laval A	The development of a calculul compliance plan is not applicable	0.00
Level A	The development of a schedule compliance plan is not applicable.	0.00
Level B	The development of a schedule compliance plan has not been addressed.	0.15
Level C	prograss parrative as required. There is effective team participation in schedule undates	0.31
	Continuation of Level C, plus quantity reports are regularly performed. Upon request, or as the project	
	requires may include any of the following: change management analysis risks assessment analysis date	
Level D	variance analysis, communication with material suppliers to ensure material will arrive on site as per the	0.46
	nlan	
	The schedule is rigorously undated based on manual input of quantity reports, critical and pear critical path	
	is analysed progress narrative is prepared and there is effective team participation in schedule undates	
Level E	Ouantity reports risorously done by trained individual(s) Material suppliers routinely contacted to track the	0.62
Lever	status of material delivery dates. Change management analysis, risks assessment analysis, date variance	0.02
	analysis may be conducted as a project requires	
Level F	Continuation of Level E. plus progress tracking using 3D imaging and other techniques.	0.77
	C4. Dynamic Site Lavout Plan	
Level A	Site layout plan is not applicable for the project.	0.00
Level B	A site layout plan has not been addressed.	0.16
1.10	The project team examines the project schedule and assesses when Temporary Facilities (TF) will be	0.22
Level C	brought in.	0.33
	Continuation of Level C, plus analyses what sizes will be needed prior to the start of the project. No	
Level D	consideration is given to the addition and removal of TFs at different stages of the project. No analysis is	0.49
	done in regards to the layout of the project to optimize locations of the TFs to limit travel time.	
Level F	Continuation of Level D, plus consideration is given to the addition and removal of TFs at different stages	0.65
	of the project.	0.05
	Continuation of Level E, plus the team analyses the layout of the project including where the different	
Level F	parties will be working and place their TFs in the optimum location in order to limit travel time to and from	0.82
	TFs.	
	C5. Traffic Control Plan	
Level A	Traffic control plans are not applicable for this project.	0.00
Level B	Traffic control plans have not been addressed in this project.	0.17
Level C	The project has some traffic control plans, and it is used on a reactive basis.	0.34
Level D	The project has a traffic control plan, traffic control equipment, and an arrangement for daylight traffic	0.51
	control only and has no trained traffic control persons.	
Level E	The project has a traffic control plan and equipment for all times of the day including trained persons for	0.68
	traffic control.	
r 15	Continuation of level E, plus a trained traffic control supervisor. The project has an approved contingency	0.05
Level F	plan to accommodate unexpected situations and has designed and constructed alternate arrangements for	0.85
-	traffic such as detours and flyovers.	
Laural A	Co. site security Plan	0.00
Level A	She security plan is not applicable for this project.	0.00
Level B	The site does not have security with regard to entry to the site, and the securing equipment is not available.	0.15
	There is control during entry and exit from the site, but does not have any other formal security system	
Level C	throughout the site	0.29
	The site has established security procedures including visitor sign in and sign out procedure and security	
	must be she has established security procedures including visitor sign in and sign out procedure and security	
Level D	assate. Protocole have been identified for searching individuals and their personal property. Searching is	0.44
	conducted randomly	
	Continuation of Level D plus the site has ensured that material is not leaving the job site by instituting	
Level E	"lock-uns" for items that are prone to theft	0.59
	look ups for nome that are prone to them.	
Level F	Continuation of Level E, plus the use of electronic security such as security cameras has been implemented.	0.73

CP. Machinery Positioning Strategy IP/eghts Level A Machinery positioning of machinery at the project site. 0.16 Level C Heavy rigging and illing studies are accomplished on all critical list including evaluation of the machinery. 0.33 Level D Continuation of Level C, plus planning includes the use of 2D layout and studies to aid in locating machinery. 0.65 Level E Continuation of Level C, plus some 3D modelling studies to aid in locating machinery. 0.65 Level E Continuation of Level E, plus therough use of 3D modelling visualization to locate machinery. 0.82 Level A No start-up plan bas been prepared; the plan has not been communicated to the concerned 0.00 A start-up plan bas been prepared; the plan has not been communicated to the concerned 0.17 Level C A basic start-up plan bas been prepared; the plan has not been of continuation of Level C, plus with consideration for itortfraces among sub-contractors or project 0.24 Level F Continuation of Level C, plus with consideration for cost analysis and detailed scheduling components. The plan is suel communicated to all the stakeholders. 0.68 Level F Continuation of Level D, plus with consideration for cost analysis and detailed scheduling components. The reget completion regitterment rhandover procedure is an oppicated price to machinery. 0.68 Level F Continuation of Level D, plus with consideration for cost analysis and detailed scheduling. 0.51		Table 10-2 continued	
Level A Machinery positioning strategy is not applicable. 0.00 Level B There is no strategy for positioning of machinery at the project site. 0.16 Level B There is no strategy for positioning of machinery at the project site. 0.16 Level E Continuation of Level C, plus planning includes the use of 2D layout and studies to aid in locating machinery. 0.65 Level F Continuation of Level F, plus horrough use of 3D modeling studiacts to aid in locating machinery. 0.82 Level A S dart-up plan axists 0.87 bit status and the plan has been prepared; the plan has not been communicated to the concerned 0.01 Level B A status pplan has been developed with input from the project participants, but the plan has not been developed. 0.34 Level C A basis status pplan has been developed with input from the project participants, but the plan has not been developed. 0.34 Continuation of Level C, plus with considerations for interfaces among sub-contractors or project 1.51 Level A The project completion requirement to handware procedure is not analysis and detailed scheduling components. The plan is sub-theorem procedure is not analysis and detailed scheduling components. The project completion requirement or handware procedure is not analysis and detailed scheduling components. The project completion requirement or handware proces that defines the parameters of project completion requirement or handware proces that defines the parameters of project completion and delineates the requirement or handware proces that defines the parameters of project completion		C7. Machinery Positioning Strategy	Weights
Level B There is no stategy for positioning of machinery at the project site. 0.16 Level C Genomy reging and filting studies to accomplished on all critical this includies to aid in locating machinery. 0.33 Level D Continuation of Level C, plus some 3D modelling studies to aid in locating machinery. 0.65 Level E Continuation of Level D, plus some 3D modelling studies to aid in locating machinery. 0.65 Level E Continuation of Level D, plus some 3D modelling studies to aid in locating machinery. 0.62 Level A No stati-up plan has been prepared; the plan has not been communicated to the concerned statcholders. 0.17 Level D A basis stati-up plan has been developed with input from the project participants, but the plan has been developed with input from the project participants, but the plan has been developed. 0.34 Continuation of Level C, plus with considerations for interfaces among sub-contractors or project 0.51 Level D preject and the identifies the ubics and responsibilities of east alkeholders. 0.68 Level R The project completion requirement or handover procedure is no tapplicable. 0.00 Level R The project one plan is approved by project and the factions and been identified. 0.17 Level C The project one plan is approved by project andiffeor requirement or handover procedure is no tapplica	Level A	Machinery positioning strategy is not applicable.	0.00
Level C Havy rigging and lifting studies are accomplished on all critical lifts including evaluation of the machinery. 0.33 Level D Continuation of Level C, plus planning includes the use of 2D layout and studies to aid in locating machinery. 0.65 Level F Continuation of Level D, plus some 3D modeling studies to aid in locating machinery. 0.62 Level F Continuation of Level D, plus some 3D modeling studies to aid in locating machinery. 0.62 Level A A partial start-up plan has been prepared; the plan has not been communicated to the concerned 0.17 Level C Continuation of Level D, plus with considerations for interfaces among sub-contractors or project 0.34 Continuation of Level D, plus with considerations for interfaces among sub-contractors or project 0.34 Continuation of Level D, plus with consideration for cost analysis and detailed scheduling components. The plan has bacheolders on polecy machine plan is used communicated to all the stakeholders. 0.68 Level F Continuation of Level D, plus with consideration for cost analysis and detailed scheduling components. The plan bias bacheolders on polecy anticipants. 0.68 Level A The project completion requirement or handover procedure is not applicable. 0.00 Level A The project completion requirement or handover procedure is not applicable. 0.00 <td>Level B</td> <td>There is no strategy for positioning of machinery at the project site.</td> <td>0.16</td>	Level B	There is no strategy for positioning of machinery at the project site.	0.16
reging selection, and erme location	Level C	Heavy rigging and lifting studies are accomplished on all critical lifts including evaluation of the machinery,	0 33
Level D Continuation of Level C, plus planning includes the use of 2D layout and studies to adi in locating 0.49 Level E Continuation of Level D, plus some 3D modelling studies to adi in locating machinery. 0.82 Level F Continuation of Level E, plus thorough use of 3D modelling studies to adi in locating machinery. 0.82 Level A No start-up plan exists 0.00 Level B A partial start-up plan has been prepared; the plan has not been communicated to the concerned 0.17 Level C A basic start-up plan has been developed with input from the project participants, bat the plan has not been implemented. 0.34 Continuation of Level C, plus with considerations for interfaces among sub-contractors or project 1.51 Level D participants. A start-up plan that identifies the duties and responsibilities of each stak-cholder has been developed. 0.68 Level F participants. A start-up plan that identifies the duties into aphilosible. 0.00 Level F well communicated to all the stak-cholders. 0.68 Level F well communicated to all the stak-cholders. 0.00 Level B The project completion requirement or handover proceed is not aphilosible. 0.00 Level B The project as a formal bandover proceed.	Levere	rigging selection, and crane location.	0.00
Tevel F. Continuation of Level D, plus some 3D modeling studies to aid in locating machinery. 0.65 Level F. Continuation of Level E, plus through use of 3D modeling visualization to locate machinery. 0.82 CR. Project Start-up Plan 0.00 Level A. No stat-up plan has been prepared; the plan has not been communicated to the concerned 0.17 Level A. A basic stat-up plan has been developed with input from the project participants, but the plan has not been 0.34 Continuation of Level D, plus with considerations for interfaces among sub-contractors or project 0.51 Level D participants. A star-up and hut identifies the duites and responsibilities of care stakeholder has been 0.51 Level C continuation of Level D, plus with consideration for cost analysis and detailed scheduling components. The plan is well communicated to all the stakeholders. 0.68 Level A The project completion requirement on handover procedure has not opticable. 0.00 Level B The project completion requirement on handover procedure has not applicable. 0.33 Level B The project completion requirement on handover procedure has not porticable. 0.50 Level E Continuation of Level D, plus the procedure has not been identified. 0.50 Level T The project completion requirement on handover process. 0.50 Level T Continuation of Level D, plus the procedure tases or ovevender task set of the handover.	Level D	Continuation of Level C, plus planning includes the use of 2D layout and studies to aid in locating	0.49
Level F. Continuation of Level D, plus some 3D modeling studies to ad in locating machinery. 0.85 Level F. Continuation of Level D, plus thorough use of 3D modeling visualization to locate machinery. 0.82 Level A. No start-up plan exists. 0.00 Level A. No start-up plan has been prepared; the plan has not been communicated to the concerned startholders. 0.17 Level C. A basic start-up plan has been developed with input from the project participants, but the plan has not been of eveloped. 0.51 Level C. Continuation of Level C, plus with considerations for interfaces among sub-contractors or project 0.68 Level E. Continuation of Level D, plus with consideration for cost analysis and detailed scheduling components. The plan is subclockfors ropicet participants. 0.68 Level F. Continuation of Level E, plus with the plan being implemented on the project by incorporating foedbacks for the bandover procedure is not aphilosible. 0.00 Level T. From the stacholders ropicet participants. 0.72 0.79 0.72 0.79 0.72 0.79 0.72 0.79 0.72 0.79 0.72 0.79 0.72 0.79 0.72 0.79 0.72 0.79 0.72 0.79 0.72 0.79 0.72 0.79 0.72		machinery.	0.65
Level F. Continuation of Level E, plus therough use of 3D modeling visualization to locate machinery. 0.82 Level A. No start-up plan exists. 0.00 Level A. No start-up plan exists. 0.00 Level A. A partial start-up plan has been prepared; the plan has not been communicated to the concerned stakeholders. 0.017 Level C. Continuation of Level C, plus with considerations for interfaces among sub-contractors or project 0.34 Level D. Dericipants. A start-ap plan has been developed with input from the project participants, but the plan has not been of each stakeholder. has been developed. 0.68 Level F. Continuation of Level D, plus with consideration for cost analysis and detailed scheduling components. The plan is well communicated to all the stakeholders. 0.68 Level R. The project completion requirement or handover procedure has not been identified. 0.17 Level B. The project completion requirement or handover procedure has not been identified. 0.17 Level C. The project has a formal handover procesting the shot provew and parced by the stakeholders. 0.68 Level C. Continuation of Level D, plus the procedure has not been identified. 0.17 Level C. The project has a formal handover procesth and defines the parameters of project.	Level E	Continuation of Level D, plus some 3D modelling studies to aid in locating machinery.	0.65
Level A No start-up plan exists. 0.00 Level B A partial start-up plan has been prepared; the plan has not been communicated to the concerned 0.17 Level C A basic start-up plan has been prepared; the plan has not been communicated to the concerned 0.17 Level C A basic start-up plan has been developed with input from the project participants, but the plan has not been 0.34 Continuation of Level C, plus with consideration for cost analysis and detailed scheduling components. The plan that identifies the duices and responsibilities of each stakeholder has been developed. 0.51 Level E Continuation of Level D, plus with consideration for cost analysis and detailed scheduling components. The plan bit will consideration for cost analysis and detailed scheduling components. The project completion requirement or handover procedure is not applicable. 0.00 Level A The project completion requirement or handover procedure is not applicable. 0.00 Level B The project completion requirement or handover procedure is not applicable. 0.50 Level C Continuation of Level D, plus the procedure has not been identified. 0.51 Level A Innovations and handover process that defines the necessary documentation, parameters of applicability during all phases of the handover process. 0.50 Level A Innovations and new technologies 0.51 Leve	Level F	Continuation of Level E, plus thorough use of 3D modeling/visualization to locate machinery.	0.82
Level B Available plan exists. 0.00 Level B Apartial start-up plan has been prepared; the plan has not been communicated to the concerned implemented. 0.17 Level C Assist start-up plan has been developed with input from the project participants, but the plan has not been implemented. 0.34 Continuation of Level C, plus with considerations for interfaces among sub-contractors or project 0.51 Level E Continuation of Level D, plus with consideration for cost analysis and detailed scheduling components. The plan is well communicated to all the stakeholders. 0.68 Level E Continuation of Level D, plus with consideration for cost analysis and detailed scheduling components. The plan is well communicated to all the stakeholders. 0.68 Level T The project completion requirement or handover procedure is not applicable. 0.00 Level T The project completion requirement or handover procedure is not applicable. 0.01 Level T The project has a bandover. 0.50 Level C The project has a bandover. 0.50 Level C The project has a bandover. 0.50 Level C The project has a formal handover process. 0.00 Level C The project has a formal handover process. 0.00 Level B A funovation in new atracib	T	Co. Project Start-up Plan	0.00
Level B A platial started plan has been prepared, the plan has not been forminumed to the contented 0.17 Level C A basic start-up plan has been developed with input from the project participants, but the plan has not been forminumed to the contents or project 0.34 Level D A basic start-up plan has been developed with input from the project participants, but the plan basen of Level C, plus with consideration for interfaces among sub-contractors or project 0.51 Level D Continuation of Level D, plus with consideration for cost analysis and detailed scheduling components. The plan is well communicated to all the stakeholders. 0.68 Level F Continuation of Level D, plus with consideration for cost analysis and detailed scheduling components. The project completion requirement or handover procedure is not applicable. 0.000 Level B The project completion requirement or handover procedure is not applicable. 0.50 Cevel B The project thas a handover procedure that defines the necessary documentation, parameters of applicability during all phases of the handover. 0.50 Level A Innovations and New Technologies 0.68 Level A Innovations and new technologies in sort applicable. 0.000 Level A Innovations and new technologies in sort applicable. 0.60 Level A Innovations and new technologies in sort applicable. 0.00	Level A	No start-up plan exists.	0.00
Level C Absici start-up plan has been developed with input from the project participants, but the plan has not been implemented. 0.34 Level D participants. A start-up plan that identifies the duties and responsibilities of each stakeholder has been developed. 0.51 Level E Continuation of Level D, plus with consideration for cost analysis and detailed scheduling components. The plan is well communicated to all the stakeholders. 0.68 Level F Continuation of Level D, plus with donsideration for cost analysis and detailed scheduling components. The plan is well communicated to all the stakeholders. 0.68 Level A The project completion requirement or handover procedure is not applicable. 0.00 Level B The project completion requirement or handover procedure is not applicable. 0.33 Level C The project completion requirement or handover procedure has not been identified. 0.17 Level C The project as a fundover process that defines the necessary documentation, parameters of completion and other issues to ensure proper handover of a project. 0.56 Level E Continuation of Level D, plus the proceedure has been reviewed and agreed by the stakeholders. 0.66 Level E Continuation and the plan is approved by project management team and is reviewed for applicability during all phases of the handover process. 0.51 Level E Continuation of Level D, plus the proc	Level B	a partial statt-up plan has been prepared, the plan has not been communicated to the concerned	0.17
Level C Project Analysis 0.34 Continuation of Level C, plus with considerations for interfaces among sub-contractors or project 0.51 Level D Evel D participants. A start-up plan that identifies the duties and responsibilities of each stakeholder has been developed. 0.51 Level E Continuation of Level D, plus with consideration for cost analysis and detailed scheduling components. The plan is well communicated to all the stakeholders. 0.68 Level F Continuation of Level D, plus with the plan being implemented on the project by incorporating feedbacks from the stakeholders or project participants. 0.00 Level A The project completion requirement or handover procedure has not been identified. 0.01 Level B The project tas a lomadover procedure has not been identified. 0.50 Level C The project nase a lomadover process that defines the necessary documentation, parameters of control completion and hendover process. 0.66 Level E Continuation of Level D, plus the plan is approved by project management team and is reviewed for applicability during all phases of the handover process. 0.83 Level A The project has an informal program for investigation of innovations and new technologies. 0.31 Level A The project during a longram for investigation of innovations, and the company will investigate the feasibility of the new technologies investigatis on is not addressed. </td <td></td> <td>A basic start-up plan has been developed with input from the project participants, but the plan has not been</td> <td></td>		A basic start-up plan has been developed with input from the project participants, but the plan has not been	
Continuation of Level C, plus with considerations for interfaces among sub-contractors or project 0.51 Level D participants. A start-up plan that identifies the duties and responsibilities of each stakeholders has been 0.51 Level F continuation of Level D, plus with consideration for cost analysis and detailed scheduling components. The 0.68 Level F continuation of Level E, plus with the plan being implemented on the project by incorporating feedbacks 0.85 Level T The project completion requirement or handover procedure has not been identified. 0.00 Level B The project conspletion requirement or handover procedure has not been identified. 0.71 Level C The project has a formal handover process that defines the parameters of project completion and delineates the requirements for the handover procedure has not hean (agreed by the stakeholders. 0.60 Level E Continuation of Level E, plus the plan is approved by project. 0.80 Level A Innovation and new raterials, equipment, information systems is not applicable. 0.00 Level B Innovations and new systems will only occur after the industry-wide implementation. 0.83 Level C The project does not have a formal program for investigation of innovations and new technologies. 0.25 Level C The projec	Level C	implemented	0.34
Level D participants. A start-up plan that identifies the duties and responsibilities of each stakeholder has been developed. 0.51 Level E Continuation of Level D, plus with consideration for cost analysis and detailed scheduling components. The plan is well communicated to all the stakeholders. 0.68 Level F from the stakeholders or project participants. 0.85 Level A The project completion requirement or handover procedure is not applicable. 0.00 Level B The project completion requirement or handover procedure has not been identified. 0.17 Level C The project completion requirement or handover procedure has not been identified. 0.50 Level E Continuation of Level D, plus the procedure has been reviewed and agreed by the stakeholders. 0.66 Level E Continuation of Level D, plus the procedure has been reviewed and agreed by the stakeholders. 0.61 Level A Innovations and new technologies investigation of innovations and new technologies. 0.00 Level A Innovations and new technologies investigation of innovations and new technologies. 0.25 Level C The project does not have a formal program for the investigation of innovations, and the company will investigate the feasibility of the new technologies on a regular basis. 0.25 Level C The oropict does not have a formal program for the i		Continuation of Level C. plus with considerations for interfaces among sub-contractors or project	
Level F Continuation of Level D, plus with consideration for cost analysis and detailed scheduling components. The project completion of Level E, plus with the plan being implemented on the project by incorporating feedbacks 0.68 Level F Continuation of Level E, plus with the plan being implemented on the project by incorporating feedbacks 0.85 Level A The project completion requirement or handover procedure has not been identified. 0.17 Level A The project completion requirement or handover procedure has not been identified. 0.7 Level C The project has a handover procedure that defines the parameters of project completion and delineates the requirements for the handover. 0.50 Level C The project has a formal handover process that defines the necessary documentation, parameters of completion and other issues to ensure proper handover of a project. 0.66 Level E Continuation of Level E, plus the plan is approved by project management team and is reviewed for applicability during all phases of the handover process. 0.61 Level B Innovations and new technologies investigation is not addressed. 0.01 Level C The project does not have a formal program for investigation of innovations, and the company will investigate the feasibility of the new testigation of innovations, and the company will investigate the feasibility of the new testigation is not applicable. 0.00 Level D The organization has an informal pro	Level D	narticinants. A start-up plan that identifies the duties and responsibilities of each stakeholder has been	0.51
Level E Continuation of Level D, plus with consideration for cost analysis and detailed scheduling components. The plan is well communicated to all the stakeholders. 0.68 Level F Continuation of Level D, plus with he plan being implemented on the project by incorporating feedbacks from the stakeholders or project participants. 0.85 C9: Project Completion Plan 0.00 Level A The project completion requirement or handover procedure is not applicable. 0.00 Level B The project has a handover proceed that defines the necessary documentation, parameters of requirements for the handover. 0.33 Level D Continuation of Level D, plus the procedure has been reviewed and agreed by the stakeholders. 0.66 Level C Continuation of Level D, plus the procedure has been reviewed and agreed by the stakeholders. 0.66 Level A Innovations in new materials, equipment, information systems is not applicable. 0.00 Level A Innovations and new technologies 0.25 Level A Innovations and new systems will only occur after the industry-wide implementation. 0.25 Level D The project chas a formal program for investigation of innovations, and new technologies. 0.51 Level C The project does not have a formal program for investigates one new systems. 0.51 <td>Level D</td> <td>developed</td> <td>0.01</td>	Level D	developed	0.01
Level E Jan is well communicated to all the stakeholders. 0.68 Level F Continuation of Level E, plus with the plan being implemented on the project by incorporating feedbacks 0.85 Level A The project completion requirement or handover procedure is not applicable. 0.00 Level A The project completion requirement or handover procedure has not been identified. 0.17 Level C The project has a handover procedure that defines the parameters of project completion and delineates the completion and other issues to ensure proper handover of a project. 0.50 Level E Continuation of Level D, plus the procedure has been reviewed and agreed by the stakeholders. 0.66 Level E Continuation of Level D, plus the procedure has been reviewed and agreed by the stakeholders. 0.68 Level I Innovations and New Technologies 0.83 Level A Innovations and new systems will only occur after the industry-wide implementation. 0.25 Level C The project does not have a formal program for investigation of innovations, and the company will investigation of a regular basis. 0.51 Level E Continuation of Level D, plus there is a formal program to investigate sense. 0.51 Level C The project does not have a formal program to investigate on innovations, and the company will investigate for feasibility of the new technologi		Continuation of Level D plus with consideration for cost analysis and detailed scheduling components. The	
Level F Continuation of Level E, plus with the plan being implemented on the project by incorporating feedbacks 0.85 CP. Project Completion Plan 0.00 Level A The project completion requirement or handover procedure is not applicable. 0.00 Level B The project has a handover procedure that defines the parameters of project completion and delineates the requirements for the handover. 0.33 Level C The project has a formal handover process that defines the necessary documentation, parameters of completion and other issues to ensure proper handover of a project. 0.50 Level E Continuation of Level E, plus the procedure has been reviewed and agreed by the stakeholders. 0.66 Level A Innovations and new materials, equipment, information systems is not applicable. 0.00 Level A Innovations and new technologies investigation of innovations and new technologies. 0.25 Level B Innovations and new systems will only occur after the industry-wide implementation. 0.25 Level E Continuation of Level E, plus the project has a formad program to investigate some new systems. 0.51 Level C The project does not have a formal program for the investigate some new systems. 0.51 Level B Innovations an informal program for the investigate some new systems.	Level E	plan is well communicated to all the stakeholders.	0.68
Level F from the stakeholders or project participants. 0.85 CP. Project Completion Plan 0.00 Level A The project completion requirement or handover procedure has not been identified. 0.01 Level C The project completion requirement or handover procedure has not been identified. 0.33 Level D The project has a formal handover process that defines the parameters of project completion and delineates the requirements for the handover process. 0.50 Level E Continuation of Level E, plus the procedure has been reviewed and agreed by the stakeholders. 0.66 Level A Innovations and New Technologies 0.83 Level A Innovations and New Technologies 0.83 Level B Innovations and new technologies investigation is no tadtressed. 0.13 Level C The project does not have a formal program for investigation of innovations, and the company will investigate the feasibility of the new technologies. 0.25 Level C The project advector of Level D, plus there is a formal program to rivestigate som new systems. 0.51 Level C The project advector of Level D, plus there is a formal program to rivestigate som new systems. 0.51 Level C The project thase a tornal program for the investigates on enew system		Continuation of Level E, plus with the plan being implemented on the project by incorporating feedbacks	
Level A The project completion requirement or handover procedure is not applicable. 0.00 Level B The project completion requirement or handover procedure is not applicable. 0.17 Level C The project the handover process that defines the parameters of project completion and delineates the requirements for the handover process that defines the necessary documentation, parameters of completion and other issues to ensure proper handover of a project. 0.50 Level D Continuation of Level D, plus the procedure has been reviewed and agreed by the stakeholders. 0.66 Level A Innovations and New Technologies 0.83 C10. Innovations and New Technologies 0.83 Level A Innovation in new materials, equipment, information systems is not applicable. 0.00 Level B Innovations and new systems will only occur after the industry-wide implementation. 0.25 Level C The project has a hormal program for the investigation of innovations, and the company will investigate all the available new technologies. 0.64 Level E Continuation of Level B, plus there is a formal program to investigate some new systems. 0.51 Level C The organization has an informal program for investigate all the available new technologies. 0.64 D.Pre-Construction Phase Management Practices Scoring Tools	Level F	from the stakeholders or project participants.	0.85
Level A The project completion requirement or handover procedure is not applicable. 0.00 Level B The project completion requirement or handover procedure has not been identified. 0.17 Level C The project has a fandover procedure that defines the parameters of project completion and delineates the requirements for the handover. 0.33 Level D The project has a fandadover process that defines the necessary documentation, parameters of completion and other issues to ensure proper handover of a project. 0.50 Level E Continuation of Level E, plus the plus the procedure has been reviewed and agreed by the stakeholders. 0.66 Level A Innovations and New Technologies 0.83 Plus Handrig, equipment, information systems is not applicable. 0.00 Level A Innovations and new technologies investigation oi nnovations and new technologies. 0.25 Level C The project does not have a formal program for the investigation of innovations, and new company will investigate to free fassibility of the new technologies on a regular basis. 0.38 Level E Continuation of Level D, plus there is a formal program to investigate some new systems. 0.51 Level E Continuation of Level D, plus there is a formal program to investigate some new systems. 0.51 Level E Continuation of Level D, plus there is a formal program to investigate so	-	C9. Project Completion Plan	
Level B The project completion requirement or handover procedure has not been identified. 0.17 Level C The project has a handover procedure that defines the parameters of project completion and delineates the completion and other issues to ensure proper handover of a project. 0.33 Level D The project has a formal handover process that defines the necessary documentation, parameters of completion and other issues to ensure proper handover of a project. 0.50 Level E Continuation of Level D, plus the procedure has been reviewed and agreed by the stakeholders. 0.66 Level E Continuation of Level D, plus the procedure has been reviewed and agreed by the stakeholders. 0.60 Level B Innovations and new technologies investigation is not applicable. 0.00 Level B Innovations and new technologies investigation of innovations and new technologies. 0.13 Level C The project does not have a formal program for investigation of innovations, and the company will investigation of innovations and new technologies. 0.51 Level D The organization has an informal program for the investigation or innovations, and the company will investigate on a regular basis. 0.51 Level C The necest schedule are not resource loaded and short interval plan 0.51 Level A The use of short interval plan is not applicable to this building project. 0.00	Level A	The project completion requirement or handover procedure is not applicable.	0.00
Level C The project has a handover procedure that defines the parameters of project completion and delineates the requirements for the handover. 0.33 Level D The project has a formal handover process that defines the necessary documentation, parameters of completion and other issues to ensure proper handover of a project. 0.50 Level E Continuation of Level D, plus the procedure has been reviewed and agreed by the stakeholders. 0.66 Level F Continuation of Level E, plus the plan is approved by project management team and is reviewed for applicability during all phases of the handover process. 0.83 Level A Innovation in new materials, equipment, information systems is not applicable. 0.00 Level B Innovation of new technologies investigation of investigation of invostions and new technologies. 0.25 Level C The project has an informal program for investigate some new systems. 0.51 Level B The organization has an informal program for the investigate some new systems. 0.51 Level F Continuation of Level D, plus there is a formal program to investigate some new systems. 0.51 Level A The use of short interval plan is not propard. 0.00 Level B Short interval plan is not program. 0.13 Level C Activities in the project schedule are not resource loaded. 0.18	Level B	The project completion requirement or handover procedure has not been identified.	0.17
Level C requirements for the handover, process that defines the necessary documentation, parameters of completion and other issues to ensure proper handover of a project. 0.50 Level E Continuation of Level D, plus the procedure has been reviewed and agreed by the stakeholders. 0.66 Level F Continuation of Level D, plus the plan is approved by project management team and is reviewed for applicability during all phases of the handover process. 0.83 Level A Innovation in new materials, equipment, information systems is not applicable. 0.00 Level B Innovations and new technologies investigation is not addressed. 0.13 Level C The project does not have a formal program for investigation of innovations, and new technologies. 0.13 Level C The organization has an informal program for the investigation of innovations, and the company will investigate the feasibility of the new technologies on a regular basis. 0.25 Level C Continuation of Level D, plus there is a formal program to investigate some new systems. 0.51 Level A The use of short interval plan is not applicable to this building project. 0.00 Level B Short interval Plan is not applicable to this building project. 0.00 Level A The use of short interval plan is not applicable to this building project. 0.00 Level B Short interval	1 10	The project has a handover procedure that defines the parameters of project completion and delineates the	0.22
Level D The project has a formal handover process that defines the necessary documentation, parameters of completion and other issues to ensure proper handover of a project. 0.50 Level E Continuation of Level D, plus the procedure has been reviewed and agreed by the stakeholders. 0.66 Level F Continuation of Level D, plus the procedure has been reviewed and agreed by the stakeholders. 0.83 <i>Cl0. Innovations and New Technologies</i> 0.00 Level A Innovations and New technologies investigation is not addressed. 0.13 Level C The project does not have a formal program for investigation of innovations, and new technologies. 0.25 Implementation of innovations and new systems will only occur after the industry-wide implementation. 0.26 Level D The organization has an informal program for the investigation of innovations, and the company will investigate the feasibility of the new technologies on a regular basis. 0.38 Level E Continuation of Level E, plus there is a formal program to investigate all the available new technologies. 0.64 D Dre-Construction Phase Management Practices Scoring Tools 0.00 Level B Short interval plan is not applicable to this building project. 0.00 Level C Activities in the available new technologies. 0.51 Level B Short interva	Level C	requirements for the handover.	0.33
Level D completion and other issues to ensure proper handover of a project. 0.30 Level E Continuation of Level D, plus the procedure has been reviewed and agreed by the stakeholders. 0.66 Level F Continuation of Level E, plus the plan is approved by project management team and is reviewed for applicability during all phases of the handover process. 0.83 Level A Innovations and new technologies investigation is not addressed. 0.13 Level B Innovations and new technologies investigation is not addressed. 0.13 Level C The project does not have a formal program for the investigation of innovations, and the company will investigate the feasibility of the new technologies on a regular basis. 0.25 Level D The organization has an informal program for the investigate some new systems. 0.51 Level E Continuation of Level D, plus there is a formal program to investigate some new systems. 0.51 Level F Continuation of Level D, plus there is a formal program to investigate some new systems. 0.51 Level A The use of short interval plan is not applicable to this building project. 0.00 Level C Arbuist the required divities in the project schedule are not resource loaded and short interval plan indicates the required materials, tools and equipment, labour, and project information required to complete each task, and activities in the project schedule are not r	1 10	The project has a formal handover process that defines the necessary documentation, parameters of	0.50
Level E Continuation of Level D, plus the procedure has been reviewed and agreed by the stakeholders. 0.66 Level F Continuation of Level E, plus the plan is approved by project management team and is reviewed for applicability during all phases of the handover process. 0.83 Level A Innovations and New Technologies 0.00 Level B Innovations and new technologies investigation of innovations and new technologies. 0.13 Level C The project does not have a formal program for investigation of innovations, and the company will investigate the feasibility of the new technologies on a regular basis. 0.25 Level D The organization has an informal program for the investigate some new systems. 0.51 Level F Continuation of Level D, plus there is a formal program to investigate some new systems. 0.51 Level F Continuation of Level D, plus there is a formal program to investigate some new systems. 0.64 D1 D. Pre-Construction Phase Management Practices Scoring Tools 0.00 Level A The use of short interval plan is not applicable to this building project. 0.00 Level B Short interval plan is not applicable to this building project. 0.35 Level C Activities in the project schedule are not resource loaded. 0.51 Level B Short interval plan	Level D	completion and other issues to ensure proper handover of a project.	0.50
Level F Continuation of Level E, plus the plan is approved by project management team and is reviewed for applicability during all phases of the handover process. 0.83 CIU. Innovations and New Technologies 0.00 Level A Innovations and new technologies investigation is not addressed. 0.13 Level C The project does not have a formal program for investigation of innovations and new technologies. 0.25 Level D The organization has an informal program for the investigation of innovations, and the company will investigate the feasibility of the new technologies on a regular basis. 0.38 Level E Continuation of Level D, plus there is a formal program to investigate some new systems. 0.51 Level F Continuation of Level D, plus there is a formal program to investigate some new systems. 0.64 D. Pre-Construction Phase Management Practices Scoring Tools 0.10 Level A The use of short interval plan is not applicable to this building project. 0.00 Level B Short interval plan is not applicable to this building project. 0.13 Level B Short interval plan indicates the required materials, tools and equipment, labour, and required project information. 0.35 Level B Short interval plan adtermate plans are prepared such as modified equipment schedules, shortened shifts, or adde	Level E	Continuation of Level D, plus the procedure has been reviewed and agreed by the stakeholders.	0.66
Level 1 applicability during all phases of the handover process. 0.35 C10. Innovations and New Technologies 0.00 Level A Innovations and new technologies investigation is not addressed. 0.13 Level C The project does not have a formal program for investigation of innovations and new technologies. 0.25 Implementation of innovations and new systems will only occur after the industry-wide implementation. 0.38 Level D The organization has an informal program for the investigation of innovations, and the company will investigate the feasibility of the new technologies on a regular basis. 0.38 Level E Continuation of Level E, plus the company investigates all the available new technologies. 0.64 D1. Short Interval Plan 0.00 Level A The use of short interval plan is not applicable to this building project. 0.00 Level A The use of short interval plan indicates the required project information. 0.35 Level C Activities in the project schedule are not resource loaded and short interval plan does not detail the required materials, tools and equipment, labour, and project information. 0.51 Level D Continuation of Level D, and activities in the project schedule are not resource loaded. 0.53 Level A The use of short interval plan indicates the required project information.	Laval F	Continuation of Level E, plus the plan is approved by project management team and is reviewed for	0.83
C10. Innovations and New Technologies Level A Innovation in new materials, equipment, information systems is not applicable. 0.00 Level B Innovations and new technologies investigation is not addressed. 0.13 Level C The project does not have a formal program for investigation of innovations and new technologies. 0.25 Implementation of innovations and new systems will only occur after the industry-wide implementation. 0.38 Level D The organization has an informal program for the investigation of innovations, and the company will investigate the feasibility of the new technologies on a regular basis. 0.51 Level E Continuation of Level D, plus there is a formal program to investigate some new systems. 0.51 Level F Continuation of Level D, plus there is a formal program to investigate some new systems. 0.64 D Dre-Construction Phase Management Practices Scoring Tools 0.00 Level A The use of short interval plan is not applicable to this building project. 0.00 Level B Short interval plan is not applicable to this building project information. 0.53 Level D Continuations of Level D, and activities in the project schedule are not resource loaded. 0.53 Level B Short interval plan adtivities in the project schedule are not resource loaded. 0.53	Level	applicability during all phases of the handover process.	0.85
Level A Innovation in new materials, equipment, information systems is not applicable. 0.00 Level B Innovations and new technologies investigation is not addressed. 0.13 Level C The project does not have a formal program for investigation of innovations and new technologies. Implementation of innovations and new systems will only occur after the industry-wide implementation. 0.25 Level D The organization has an informal program for the investigation of innovations, and the company will investigate the feasibility of the new technologies on a regular basis. 0.38 Level E Continuation of Level D, plus there is a formal program to investigate some new systems. 0.51 Level F Continuation of Level E, plus the company investigates all the available new technologies. 0.64 D Dere-Construction Phase Management Practices Scoring Tools 0.10 Level A The use of short interval plan is not applicable to this building project. 0.00 Level B Short interval plan is not applicable to this building project. 0.35 Level D Continuations of Level D, and activities in the project schedule are not resource loaded. 0.53 Level B Continuations of Level D, and activities in the project schedule are not resource loaded. 0.53 Level E Continuation of Level D, and activities in the project schedule are not resou		C10. Innovations and New Technologies	
Level B Innovations and new technologies investigation is not addressed. 0.13 Level C The project does not have a formal program for investigation of innovations and new technologies. 0.25 Implementation of innovations and new systems will only occur after the industry-wide implementation. 0.38 Level D The organization has an informal program for the investigation of innovations, and the company will investigate the feasibility of the new technologies on a regular basis. 0.38 Level E Continuation of Level D, plus there is a formal program to investigate some new systems. 0.64 Devel F Continuation of Level E, plus the company investigates all the available new technologies. 0.64 Level A The use of short interval plan is not prepared. 0.00 Level B Short interval plan is not prepared. 0.13 Level D Activities in the project schedule are not resource loaded and short interval plan does not detail the required materials, tools and equipment, labour, and required materials, tools and equipment, labour, and project information required to complete each task, and activities in the project schedule are not resource loaded. 0.53 Level E Continuation of Level D, and activities in the project schedule are not resource loaded. 0.50 Level E Continuation of Level D, and activities in the project schedule are not resource loaded. 0.53 <	Level A	Innovation in new materials, equipment, information systems is not applicable.	0.00
Level C The project does not have a formal program for investigation of innovations and new technologies. 0.25 Level D The organization has an informal program for the investigation of innovations, and the company will investigate the feasibility of the new technologies on a regular basis. 0.38 Level E Continuation of Level D, plus there is a formal program to investigate some new systems. 0.51 Level F Continuation of Level D, plus there is a formal program to investigate some new systems. 0.64 D The organization has an informal program to investigate some new systems. 0.51 Level F Continuation of Level D, plus there is a formal program to investigate some new systems. 0.64 D1 Short Interval Plan Level A The use of short interval plan is not applicable to this building project. 0.00 Level C Activities in the project schedule are not resource loaded and short interval plan does not detail the required materials, tools and equipment, labour, and required materials, tools and equipment, labour, and project information 0.53 Level E Continuation of Level D, and activities in the project schedule are resource loaded. 0.70 Level F Continuation of Level D, and activities in the project schedule are resource loaded. 0.70 Level F Continuation of Level D, and activities in the project schedule are resource	Level B	Innovations and new technologies investigation is not addressed.	0.13
Level C Implementation of innovations and new systems will only occur after the industry-wide implementation. 0.25 Level D The organization has an informal program for the investigation of innovations, and the company will investigate the feasibility of the new technologies on a regular basis. 0.38 Level E Continuation of Level D, plus there is a formal program to investigate some new systems. 0.51 Level F Continuation of Level E, plus the company investigates all the available new technologies. 0.64 D Pre-Construction Phase Management Practices Scoring Tools 0.64 D1 Short Interval Plan 0.000 Level B Short interval plan is not applicable to this building project. 0.00 Level C Activities in the project schedule are not resource loaded and short interval plan does not detail the required materials, tools and equipment, labour, and required materials, tools and equipment, labour, and project information 0.53 Level D Continuations of Level D, and activities in the project schedule are resource loaded. 0.70 Level F Continuation of Level E, and alternate plans are prepared such as modified equipment schedules, shortened shifts, lengthened shifts in case the original plans did not work. 0.70 Level F Continuation of Level B, and design is complete but without buildability review; execution is controlled using a master schedule. 0.1		The project does not have a formal program for investigation of innovations and new technologies.	
Level D The organization has an informal program for the investigation of innovations, and the company will investigate the feasibility of the new technologies on a regular basis. 0.38 Level E Continuation of Level D, plus there is a formal program to investigate some new systems. 0.51 Level F Continuation of Level E, plus the company investigates all the available new technologies. 0.64 D. Pre-Construction Phase Management Practices Scoring Tools 0.64 Level A The use of short interval plan is not applicable to this building project. 0.00 Level B Short interval plan is not prepared. 0.18 Level C Activities in the project schedule are not resource loaded and short interval plan does not detail the required materials, tools and equipment, labour, and project information 0.53 Level D Short interval plan indicates the required materials, tools and equipment, labour, and project information required to complete each task, and activities in the project schedule are not resource loaded. 0.53 Level F Continuation of Level D, and attrivities in the project schedule are resource loaded. 0.70 Level F Continuation of Level E, and alternate plans are prepared such as modified equipment schedules, shortened shifts, lengthened shifts in case the original plans did not work. 0.88 Level A A well-defined scope of work is not applicable to this building project.	Level C	Implementation of innovations and new systems will only occur after the industry-wide implementation.	0.25
Level D The organization has an informal program for the investigation of innovations, and the company will 0.38 Level E Continuation of Level D, plus there is a formal program to investigate some new systems. 0.51 Level F Continuation of Level E, plus the company investigates all the available new technologies. 0.64 DI. Short Interval Plan Level A The use of short interval plan is not applicable to this building project. 0.00 Level C Activities in the project schedule are not resource loaded and short interval plan does not detail the required materials, tools and equipment, labour, and required project information. 0.53 Level D Short interval plan indicates the required materials, tools and equipment, labour, and project schedule are not resource loaded. 0.53 Level E Continuations of Level D, and activities in the project schedule are not resource loaded. 0.53 Level E Continuations of Level D, and activities in the project schedule are not resource loaded. 0.53 Level F Continuations of Level E, and alternate plans are prepared such as modified equipment, schedules, shortened shifts, lengthened shifts, or added shifts in case the original plans did not work. 0.70 Level F Continuation of Level B, and design is complete but without buildability review; execution is controlled with a milestone schedule. 0.88 L			
investigate the feasibility of the new technologies on a regular basis. 0.51 Level E Continuation of Level E, plus the company investigates all the available new technologies. 0.64 D. Pre-Construction Phase Management Practices Scoring Tools D1. Short Interval Plan Level A The use of short interval plan is not applicable to this building project. 0.00 Level B Short interval plan is not prepared. 0.18 Level C Activities in the project schedule are not resource loaded and short interval plan does not detail the required materials, tools and equipment, labour, and required project information. 0.35 Level D Short interval plan indicates the required materials, tools and equipment, labour, and required materials, tools and equipment, labour, and project information required to complete each task, and activities in the project schedule are not resource loaded. 0.70 Level E Continuation of Level D, and activities in the project schedule are not resource loaded. 0.88 Level F Continuation of Level E, and alternate plans are prepared such as modified equipment schedules, shortened shifts, or added shifts in case the original plans did not work. 0.88 Level B Working drawings are based on incomplete design, and execution is controlled with a milestone schedule. 0.19 Level C Continuation of Level B, and design is complete but without buil	Level D	The organization has an informal program for the investigation of innovations, and the company will	0.38
Level E Continuation of Level D, plus there is a formal program to investigate some new systems. 0.51 Level F Continuation of Level E, plus the company investigates all the available new technologies. 0.64 D. Pre-Construction Phase Management Practices Scoring Tools D1. Short Interval Plan 0.00 Level B Short interval plan is not prepared. 0.18 Level C Activities in the project schedule are not resource loaded and short interval plan does not detail the required materials, tools and equipment, labour, and required project information required to complete each task, and activities in the project schedule are not resource loaded. 0.35 Level D Continuation of Level D, and activities in the project schedule are not resource loaded. 0.70 Level F Continuation of Level D, and activities in the project schedule are resource loaded. 0.88 Level F Continuation of Level E, and alternate plans are prepared such as modified equipment schedules, shortened shifts, or added shifts in case the original plans did not work. 0.88 Level A A well-defined scope of work is not applicable to this building project. 0.00 Level B Working drawings are based on incomplete design, and execution is controlled with a milestone schedule. 0.19 Level C Continuation of Level C, plus duration for the work package is estimated, material availabilit		investigate the feasibility of the new technologies on a regular basis.	0.51
Level F Continuation of Level E, plus the company investigates all the available new technologies. 0.64 D. Pre-Construction Phase Management Practices Scoring Tools D1. Short Interval Plan Level A The use of short interval plan is not applicable to this building project. 0.00 Level B Short interval plan is not prepared. 0.18 Level C Activities in the project schedule are not resource loaded and short interval plan does not detail the required materials, tools and equipment, labour, and required project information. 0.53 Level D Short interval plan indicates the required materials, tools and equipment, labour, and project information required to complete each task, and activities in the project schedule are not resource loaded. 0.53 Level E Continuation of Level D, and activities in the project schedule are resource loaded. The short interval plan considers the effects of congestion. 0.70 Level F F Continuation of Level E, and alternate plans are prepared such as modified equipment schedules, shortened shifts, or added shifts in case the original plans did not work. 0.88 Level B Working drawings are based on incomplete design, and execution is controlled with a milestone schedule. 0.19 Level C Continuation of Level D, plus the design is complete and buildability review; execution is controlled using a master schedule. 0.58 <tr< td=""><td>Level E</td><td>Continuation of Level D, plus there is a formal program to investigate some new systems.</td><td>0.51</td></tr<>	Level E	Continuation of Level D, plus there is a formal program to investigate some new systems.	0.51
D. Pre-Construction Phase Management Practices Scoring Tools D1. Short Interval Plan Level A The use of short interval plan is not applicable to this building project. 0.00 Level B Short interval plan is not prepared. 0.18 Level C Activities in the project schedule are not resource loaded and short interval plan does not detail the required materials, tools and equipment, labour, and required project information. 0.35 Level D Short interval plan indicates the required materials, tools and equipment, labour, and project information required to complete each task, and activities in the project schedule are not resource loaded. 0.53 Level E Continuations of Level D, and activities in the project schedule are resource loaded. The short interval plan considers the effects of congestion. 0.70 Level F Shifts, lengthened shifts, or added shifts in case the original plans did not work. 0.88 D2. Well-Defined Scope of Work 0.00 Level B Working drawings are based on incomplete design, and execution is controlled with a milestone schedule. 0.19 Level C Continuation of Level B, and design is complete but without buildability review; execution is controlled using a master schedule. 0.58 Level C Continuation of Level B, and design is complete dand buildability review has been performed. 0.58 Level C Continuation of Level B, and design is complete dut without buildability review; execution is controlled using a mas	Level F	Continuation of Level E, plus the company investigates all the available new technologies.	0.64
Level A The use of short interval plan is not applicable to this building project. 0.00 Level B Short interval plan is not prepared. 0.18 Level C Activities in the project schedule are not resource loaded and short interval plan does not detail the required materials, tools and equipment, labour, and required project information. 0.35 Level D Short interval plan indicates the required materials, tools and equipment, labour, and project information required to complete each task, and activities in the project schedule are not resource loaded. 0.53 Level E Continuations of Level D, and activities in the project schedule are resource loaded. The short interval plan considers the effects of congestion. 0.70 Level F Continuation of Level E, and alternate plans are prepared such as modified equipment schedules, shortened shifts, or added shifts in case the original plans did not work. 0.88 D2. Well-Defined Scope of Work 0.00 Level B Working drawings are based on incomplete design, and execution is controlled with a milestone schedule. 0.19 Level C Continuation of Level D, plus duration for the work package is estimated, material availability, testing and inspection requirements are defined, but budget and quantities are not included in the scope of work. 0.70 Level A A well-defined scope IC, plus the design is complete and buildability review has been performed. 0.38 Level		D. Pre-Construction Phase Management Practices Scoring Tools	
Level A The use of short interval plan is not applicable to this building project. 0.00 Level B Short interval plan is not prepared. 0.18 Level C Activities in the project schedule are not resource loaded and short interval plan does not detail the required materials, tools and equipment, labour, and project information. 0.35 Level D Short interval plan indicates the required materials, tools and equipment, labour, and project information. 0.53 Level E Continuations of Level D, and activities in the project schedule are not resource loaded. 0.70 Level F Continuation of Level E, and alternate plans are prepared such as modified equipment schedules, shortened shifts, or added shifts in case the original plans did not work. 0.88 Level B Working drawings are based on incomplete design, and execution is controlled with a milestone schedule. 0.19 Level C Continuation of Level B, and design is complete but without buildability review; execution is controlled using a master schedule. 0.38 Level C Continuation of Level C, plus the design is complete and buildability review has been performed. 0.38 Level B Continuation of Level D, plus duration for the work package is estimated, material availability, testing and inspection requirements are defined, but budget and quantities are not included in the scope of work. 0.77	T	D1. Short Interval Plan	0.00
Level B Short interval plan is not prepared. 0.16 Level C Activities in the project schedule are not resource loaded and short interval plan does not detail the required materials, tools and equipment, labour, and project information 0.35 Level D Short interval plan indicates the required materials, tools and equipment, labour, and project information 0.53 Level D Continuations of Level D, and activities in the project schedule are not resource loaded. 0.70 Level F Continuation of Level E, and alternate plans are prepared such as modified equipment schedules, shortened shifts, or added shifts in case the original plans did not work. 0.88 Level B Working drawings are based on incomplete design, and execution is controlled with a milestone schedule. 0.19 Level C Continuation of Level B, and design is complete but without buildability review; execution is controlled with a milestone schedule. 0.38 Level C Continuation of Level C, plus the design is complete and buildability review has been performed. 0.38 Level C Continuation of Level D, plus duration for the work package is estimated, material availability, testing and inspection requirements are defined, but budget and quantities are not included in the scope of work. 0.70 Level C Continuation of Level D, plus duration for the work package is estimated, material availability, testing and inspection requirements are defined, but budget and quantities ar	Level A	Short interval plan is not prepared	0.00
Level C Activities in the project schedule ale not resource loaded and short interval plan does not detail the required materials, tools and equipment, labour, and project information. 0.35 Level D Short interval plan indicates the required materials, tools and equipment, labour, and project information required to complete each task, and activities in the project schedule are not resource loaded. 0.53 Level E Continuations of Level D, and activities in the project schedule are resource loaded. The short interval plan considers the effects of congestion. 0.70 Level F Continuation of Level E, and alternate plans are prepared such as modified equipment schedules, shortened shifts, lengthened shifts in case the original plans did not work. 0.88 D2. Well-Defined Scope of Work 0.00 Level B Working drawings are based on incomplete design, and execution is controlled with a milestone schedule. 0.19 Level C Continuation of Level B, and design is completed and buildability review has been performed. 0.38 Level C Continuation of Level B, and design is completed and buildability review has been performed. 0.58 Level D Continuation of Level D, plus duration for the work package is estimated, material availability, testing and inspection requirements are defined, but budget and quantities are not included in the scope of work. 0.58	Level D	Activities in the project schedule are not resource loaded and short interval plan does not detail the required	0.16
Indictation of Level B Short interval plan indicates the required materials, tools and equipment, labour, and project information 0.53 Level D Continuations of Level D, and activities in the project schedule are not resource loaded. 0.70 Level F Continuation of Level E, and alternate plans are prepared such as modified equipment schedules, shortened shifts, lengthened shifts, or added shifts in case the original plans did not work. 0.88 Level A A well-defined scope of work is not applicable to this building project. 0.00 Level B Working drawings are based on incomplete design, and execution is controlled with a milestone schedule. 0.19 Level C Continuation of Level B, and design is completed and buildability review has been performed. 0.58 Level C Continuation of Level B, and design is completed and buildability review has been performed. 0.58 Level C Continuation of Level B, and design is completed and buildability review has been performed. 0.58 Level D Continuation of Level B, and design is completed and buildability review has been performed. 0.58 Level E Continuation of Level D, plus duration for the work package is estimated, material availability, testing and inspection requirements are defined, but budget and quantities are not included in the scope of work. 0.77 Level F Continuation of Level E, and budget and quantities are inclu	Level C	materials tools and equipment labour and required project information	0.35
Level D Short interval plan indicates the required matchas, tools and equipment, labour, and project mormation 0.53 Level E Continuations of Level D, and activities in the project schedule are not resource loaded. 0.70 Level F Continuation of Level E, and alternate plans are prepared such as modified equipment schedules, shortened shifts, lengthened shifts, or added shifts in case the original plans did not work. 0.88 D2. Well-Defined Scope of Work 0.00 Level B Working drawings are based on incomplete design, and execution is controlled with a milestone schedule. 0.19 Level C Continuation of Level B, and design is completed and buildability review has been performed. 0.38 Level D Continuation of Level B, and design is completed and buildability review has been performed. 0.53 Level C Continuation of Level C, plus the design is completed and buildability review has been performed. 0.59 Level E Continuation of Level D, plus duration for the work package is estimated, material availability, testing and inspection requirements are defined, but budget and quantities are not included in the scope of work. 0.70 Level F Continuation of Level E, and budget and quantities are included in the scope of work. 0.77		Short interval plan indicates the required materials tools and equipment labour and project information.	
Level E Continuations of Level D, and activities in the project schedule are resource loaded. The short interval plan considers the effects of congestion. 0.70 Level F Continuation of Level E, and alternate plans are prepared such as modified equipment schedules, shortened shifts, lengthened shifts, or added shifts in case the original plans did not work. 0.88 D2. Well-Defined Scope of Work 0.80 Level A A well-defined scope of work is not applicable to this building project. 0.00 Level B Working drawings are based on incomplete design, and execution is controlled with a milestone schedule. 0.19 Level C Continuation of Level B, and design is complete but without buildability review; execution is controlled using a master schedule. 0.38 Level D Continuation of Level C, plus the design is completed and buildability review has been performed. 0.58 Level E Continuation of Level D, plus duration for the work package is estimated, material availability, testing and inspection requirements are defined, but budget and quantities are not included in the scope of work. 0.77 Level F Continuation of Level E, and budget and quantities are included in the scope of work. 0.96	Level D	required to complete each task, and activities in the project schedule are not resource loaded	0.53
Level E Continuation of Level D, and advinces in the project schedule are resource roaded. The short interval phanel 0.70 Level F Continuation of Level E, and alternate plans are prepared such as modified equipment schedules, shortened 0.88		Continuations of Level D and activities in the project schedule are resource loaded. The short interval plan	
continuers the effects of expectation. Level F Continuation of Level E, and alternate plans are prepared such as modified equipment schedules, shortened shifts, lengthened shifts, or added shifts in case the original plans did not work. 0.88 D2. Well-Defined Scope of Work Level A A well-defined scope of work is not applicable to this building project. 0.00 Level B Working drawings are based on incomplete design, and execution is controlled with a milestone schedule. 0.19 Level C Continuation of Level B, and design is complete but without buildability review; execution is controlled using a master schedule. 0.38 Level D Continuation of Level C, plus the design is completed and buildability review has been performed. 0.58 Level E Continuation of Level D, plus duration for the work package is estimated, material availability, testing and inspection requirements are defined, but budget and quantities are not included in the scope of work. 0.77 Level F Continuation of Level E, and budget and quantities are included in the scope of work. 0.96	Level E	considers the effects of congestion	0.70
Level F Sommarian on Level D, and added shifts in case the original plans did not work. 0.88 Shifts, lengthend shifts, or added shifts in case the original plans did not work. 0.00 Level A A well-defined scope of work is not applicable to this building project. 0.00 Level B Working drawings are based on incomplete design, and execution is controlled with a milestone schedule. 0.19 Level C Continuation of Level B, and design is complete but without buildability review; execution is controlled using a master schedule. 0.38 Level D Continuation of Level C, plus the design is completed and buildability review has been performed. 0.58 Level E Continuation of Level D, plus duration for the work package is estimated, material availability, testing and inspection requirements are defined, but budget and quantities are not included in the scope of work. 0.77 Level F Continuation of Level E, and budget and quantities are included in the scope of work. 0.96		Continuation of Level E and alternate plans are prepared such as modified equipment schedules, shortened	
D2. Well-Defined Scope of Work Level A A well-defined scope of work is not applicable to this building project. 0.00 Level B Working drawings are based on incomplete design, and execution is controlled with a milestone schedule. 0.19 Level C Continuation of Level B, and design is complete but without buildability review; execution is controlled using a master schedule. 0.38 Level D Continuation of Level C, plus the design is completed and buildability review has been performed. 0.58 Level E Continuation of Level D, plus duration for the work package is estimated, material availability, testing and inspection requirements are defined, but budget and quantities are not included in the scope of work. 0.77 Level F Continuation of Level E, and budget and quantities are included in the scope of work. 0.96	Level F	shifts, lengthened shifts, or added shifts in case the original plans did not work.	0.88
Level A A well-defined scope of work is not applicable to this building project. 0.00 Level B Working drawings are based on incomplete design, and execution is controlled with a milestone schedule. 0.19 Level C Continuation of Level B, and design is complete but without buildability review; execution is controlled using a master schedule. 0.38 Level D Continuation of Level C, plus the design is completed and buildability review has been performed. 0.58 Level E Continuation of Level D, plus duration for the work package is estimated, material availability, testing and inspection requirements are defined, but budget and quantities are not included in the scope of work. 0.77 Level F Continuation of Level E, and budget and quantities are included in the scope of work. 0.96		D2. Well-Defined Scope of Work	
Level B Working drawings are based on incomplete design, and execution is controlled with a milestone schedule. 0.19 Level C Continuation of Level B, and design is complete but without buildability review; execution is controlled using a master schedule. 0.38 Level D Continuation of Level C, plus the design is completed and buildability review has been performed. 0.58 Level E Continuation of Level D, plus duration for the work package is estimated, material availability, testing and inspection requirements are defined, but budget and quantities are not included in the scope of work. 0.77 Level F Continuation of Level E, and budget and quantities are included in the scope of work. 0.96	Level A	A well-defined scope of work is not applicable to this building project.	0.00
Level B Working drawings are based on incomplete design, and execution is controlled with a milestone schedule. 0.19 Level C Continuation of Level B, and design is complete but without buildability review; execution is controlled using a master schedule. 0.38 Level D Continuation of Level C, plus the design is completed and buildability review has been performed. 0.58 Level E Continuation of Level D, plus duration for the work package is estimated, material availability, testing and inspection requirements are defined, but budget and quantities are not included in the scope of work. 0.77 Level F Continuation of Level E, and budget and quantities are included in the scope of work. 0.96			0.10
Level CContinuation of Level B, and design is complete but without buildability review; execution is controlled using a master schedule.0.38Level DContinuation of Level C, plus the design is completed and buildability review has been performed.0.58Level EContinuation of Level D, plus duration for the work package is estimated, material availability, testing and inspection requirements are defined, but budget and quantities are not included in the scope of work.0.77Level FContinuation of Level E, and budget and quantities are included in the scope of work.0.96	Level B	Working drawings are based on incomplete design, and execution is controlled with a milestone schedule.	0.19
Level C using a master schedule. 0.38 Level D Continuation of Level C, plus the design is completed and buildability review has been performed. 0.58 Level E Continuation of Level D, plus duration for the work package is estimated, material availability, testing and inspection requirements are defined, but budget and quantities are not included in the scope of work. 0.77 Level F Continuation of Level E, and budget and quantities are included in the scope of work. 0.96	I 1 C	Continuation of Level B, and design is complete but without buildability review; execution is controlled	0.20
Level D Continuation of Level C, plus the design is completed and buildability review has been performed. 0.58 Level E Continuation of Level D, plus duration for the work package is estimated, material availability, testing and inspection requirements are defined, but budget and quantities are not included in the scope of work. 0.77 Level F Continuation of Level E, and budget and quantities are included in the scope of work. 0.96	Level C	using a master schedule.	0.38
Level EContinuation of Level D, plus duration for the work package is estimated, material availability, testing and inspection requirements are defined, but budget and quantities are not included in the scope of work.0.77Level FContinuation of Level E, and budget and quantities are included in the scope of work.0.96	Level D	Continuation of Level C, plus the design is completed and buildability review has been performed.	0.58
Level E inspection requirements are defined, but budget and quantities are not included in the scope of work. 0.77 Level F Continuation of Level E, and budget and quantities are included in the scope of work. 0.96	LavalE	Continuation of Level D, plus duration for the work package is estimated, material availability, testing and	0 77
Level F Continuation of Level E, and budget and quantities are included in the scope of work. 0.96	Level E	inspection requirements are defined, but budget and quantities are not included in the scope of work.	0.77
	Level F	Continuation of Level E, and budget and quantities are included in the scope of work.	0.96

Level A Utilization of software in wych packaging: is not applicable. 000 Level B This project uses a software system to generate and track work packages. The system is not integrated; and materials, drawing and other information are arenered manually. 0.0 Level B Continuation of Level D, and the project uses a software system which automatically includes the drawings. 0.3 Level D And the project uses a software system which automatically includes the drawings. 0.4 Level D And the system is updated regularly, and it provides current design drawing. 0.4 Level D Continuation of Level D, and the system is updated electronically. 0.7 Level E Continuation of Level D, and the system is updated electronically. 0.7 Level B Assigning dedicated planner: is not applicable to this building project. 0.0 Level B Hing of the dictated planner is not applicable to this building project. 0.0 Level B The dictate planner in antitation and regularity. 0.1 Level B Continuation of Level D, pubs continued update of schedule and onsite inspection to tack consumption of materials and regularity. 0.4 Level B Continuation of Level D, pubs continued update of schedule and onsite inspection totack consumption of materials and the dict the schedul		Table 10-2 continued	HZ + 1 /
Level B This project uses a software system to generate and track work packages. The system is not integrated; and 0.1 Level B This project uses a software system to generate and track work packages. The system is not integrated; and 0.1 Level C Continuation of Level C, and the project uses a software system thich automatically includes the drawings 0.3 Level D and material delivery status. Schedule, percent complete, test and inspection status, and closure are entered 0.4 Level E formation of Level D, and the system is updated regulatly, and it provides current design drawing 0.6 Level E formation of Level D, bus the work package. 0.7 Decidence D D. Decidence D 0.0 Level E formation of Level D, provides or materials in project. 0.0 Level F Ornination of Level D, provides or depicies the order and project to the schedule and coordinate materials and report their status. 0.2 Level C Single provide the order and project to the schedule as a appropriate. 0.2 Level C Single provide the delivery and inspection of materials and report their status. 0.3 Level C Single provide the delivery and inspection of materials and report their status. 0.3 Level A Or	Laval A	D3. Use of Software in Planning Work Packages	Weights
Level B Instruction for Level C, and percent complete is entered manually by reviewing the status of the work package. 0.3 Level C Continuation of Level C, and the project uses a software system which automatically includes the drawings contraction of Level D. and the system is updated regularly, and it provides current design drawing Level E information actuates of the work package. 0.4 Level D Data of Level C, and the system is updated regularly, and it provides current design drawing Level E information of Level D, and the system is updated electronically. 0.7 Level A Assigning dedicated planner is not applicable to this building project. 0.0 Level B Continuation of Level D, plus the work packages. 0.0 Level B Continuation of Level C, plus the planner initiate and maintain communication regarding the delivery and implement process are assigned. 0.0 Level E Continuation of Level C, plus the planner initiate and maintain communication regarding the delivery and implement projectate are addressed. 0.0 Level F Continuation of Level E, plus continued opdate of schedule and onsite inspection to track consumption of materials and update the schedules as appropriate. 0.0 Level F Continuation of Level E, plus continued communication regarding consumption of materials and report their requirement requirement are addressed. 0.0 Level F Continuation of Level E, plus continued communication regarding consumptio	Level A	Utilization of software in work packaging is not applicable.	0.00
Level C Continuation of Level B, and percent complete is entered manually by reviewing the status of the work package. 0.3 Cevel D and material delivery status. Schedule, percent complete, test and inspection status, and closure are entered of the inspection status, and entered complete is information of Level D, and the system is updated englishy, and it provides current design drawing Level E. Continuation of Level D, and the system is updated electronically. 0.7 Level E Continuation of Level D, plus the work package. 0.7 Level A Assigning dedicated planner is not applicable to this building project. 0.0 Level B Continuation of Level C, plus the work package status is updated electronically. 0.0 Level C Single planner or multiple personnel who prepare initial project schedule and coordinate materials 0.1 Level C Disconnitud qualited of schedule and onsite inspection to track consumption of materials and update the schedules as appropriate. 0.3 Level F Ordination of Level E, plus continued qualited of schedule and onsite inspection to track consumption of status. 0.0 Level B Arginizing information such as the scope of works, safety, quality of work, materials, personnel and equipment trajterments are addressed. 0.0 Level B Some information such as the scope of works, safety, quality of work, materials, personnel and equipiment trajterments are addressed. <t< td=""><td>Level B</td><td>materials, drawing and other information are entered manually.</td><td>0.15</td></t<>	Level B	materials, drawing and other information are entered manually.	0.15
Continuation of Level C, and the project uses a software system which automatically includes the drawings 0.4 Level D or material delivery status. Schedule, percent complete, test and inspection status, and closure are entered inspection status, and percent complete. 0.6 Level E information, updated status of materials, implementation schedule with durations and quantities, test and inspection status, and percent complete. 0.7 Level F continuation of Level E, pus the work package status is updated electronically. 0.7 Level A Assigning dedicated planner has not been addressed. 0.1 Level C Single planner or multiple personent whose persone initial project schedule and coordinate materials or procurement process are assigned. 0.3 Level C Continuation of Level E, pulse continued update of schedule and onsite inspection to track consumption of materials and update the schedules as appropriate. 0.6 Level A Organizing information for each Construction Work Packages (CWP) is not applicable. 0.0 Level A Organizing information sch as materials and equipment requirements are addressed. 0.1 Level A Organizing information is the scope of works, safety, quality of work, materials, personnel and equipment section states. 0.3 Level A Organizing information is the scope of works, safety, quality of work, materials, personnel and equipment of scelector of resectorion of scelector of reach CWP. 0.6 Level C Sonte information sequal expl	Level C	Continuation of Level B, and percent complete is entered manually by reviewing the status of the work package.	0.31
Level D and material delivery status. Schedule, percent complete, test and inspection status, and closure are entered manually by review of the work package. 0.4 Continuation of Level D, and the system is updated regularly, and it provides current design drawing Level F. 0.7 Level F. Continuation of Level E, plus the work package status is updated electronically. 0.7 Level A. Assigning decided planner is not suplicable to this building project. 0.0 Level B. Single planner or multiple personnel who prepare initial project schedule and coordinate materials proment process are sayinged. 0.2 Level E. Continuation of Level C, plus the planner initiate and maintain communication regarding the delivery and insterials as necessary. 0.3 Level E. Continuation of Level E, plus continued opdate of schedule as appropriate. 0.0 Level B. Organizing information for each Construction Work Packages 0.1 Level B. Organizing information of a sametarials an quipoment requirements are addressed. 0.1 Level B. Some information such as the scope of works, safety, quality of work, materials, personnel and equipment status. 0.3 Level B. Continuation of Level D, plus in-depth consideration for meet of the sections are given. 0.6 Level B. Continuation of Level D,		Continuation of Level C, and the project uses a software system which automatically includes the drawings	
Continuation of Level D, and the system is updated regularly, and it provides current design drawing Level E information, updated status of materials, implementation schedule with durations and quantities, test and inspection status, and precent complete D-Dedicated Planner D-D-Dedicated Planner D-D-Dedicated Planner D-D-Dedicated Planner D-D-Dedicated Planner D-D-Dedicated Planner D-D-Dedicated Planner D-D-D-Dedicated Planner D-D-D-D-D-Dedicated Planner D-D-D-D-D-D-D-D-D-D-D-D-D-D-D-D-D-D-D-	Level D	and material delivery status. Schedule, percent complete, test and inspection status, and closure are entered	0.46
Level E information, updated status of materials, implementations of schedule with durations and quantities, test and unspection status, and percent complete. 0.7 Level F Continuation of Level F, bus the work package status is updated electronically. 0.7 Level F Continuation of Level F, bus the work package status is updated electronically. 0.7 Level A Assigning dedicated planner is not splicable to this building project. 0.0 Level B Hring of the dedicated planner has not been addressed. 0.1 Continuation of Level C, plus the planner initiate and maintain communication regarding the delivery and inspection of materials and update the schedules as appropriate. 0.2 Level C Continuation of Level L, plus continued update of schedule and onsite inspection to track consumption of materials as necessary. 0.6 Level F Continuation such as materials and equipment requirements are addressed. 0.1 Level C Single update of the schedules as appropriate. 0.0 Level C Some information such as the scope of works, safety, quality of work, materials, personnel and equipment requirements is compiled, but the information for neat addressed. 0.1 Level C Omaining information addressed in the prove 0.6 Level C Continuation of Level D, plus in-depth considerati		Continuation of Level D and the system is undeted regularly, and it provides surrent design drawing.	
Ever F Continuation of Level E, plus the work package status is updated electronically. 0.7 D Decicated Planner 0.0 Ever A Assigning dolicated planner is not applicable to this building project. 0.0 Ever A Assigning dolicated planner is not applicable to this building project. 0.0 Ever B Finge planner or multiple personel who prepare initial project schedule and coordinate materials or procurement process are assigned. 0.1 Ever B Continuation of Level C, plus the planner initiate and maintain communication regarding the delivery and inspection of materials and update of schedule and onsite inspection to track consumption of materials as necessary. 0.4 Ever B Continuation of Level D, plus continue dupate of schedule and onsite inspection to track consumption of materials and report their status. 0.6 Ever B Desine information such as materials and equipment requirements are addressed. 0.1 Ever C Continuation of Level C, plus indeptic onsideration from ot of the sections are given. 0.6 Ever B Desine information such as the scope of works, safety, quality of work, materials, personnel and equipment requirements are addressed. 0.1 Ever E Continuation of Level E, plus in-depth consideration from most of the sections are given. 0.6 Ever E Continuation of Level E, plus all sections h	Laval E	information undeted status of materials implementation schedule with durations and quantities test and	0.61
Level P Continuation of Level E. plus the work package status is updated electronically. 0.7 Level A Assigning dedicated planner is not applicable to this building project. 0.0 Level A Assigning dedicated planner is not applicable to this building project. 0.0 Level C Single planner or multiple personnel who prepare initial project schedule and coordinate materials or procurement process are assigned. 0.3 Level C Continuation of Level C. plus the planner initiate and maintain communication regarding the delivery and inspection of materials and update the schedules as appropriate. 0.3 Level C Continuation of Level C. plus continued update of schedule and onsite inspection to track consumption of materials and report their status. 0.5 Level A Organizing information such as materials and equipment requirements are addressed. 0.0 Level B Sint information such as the scope of vocks, safety, quality of work, materials, personnel and equipment requirements is compiled, but the information in contopletie. 0.0 Level D Continuation of Level E, plus all educing the reliver 0.6 Level C Continuation of Level E, plus all scheration. 0.6 Level D All information scheration. 0.6 Level C Continuation of Level D, plus indepti considera	Level E	information, updated status of materials, imprementation schedule with durations and quantities, test and inspection status, and percent complete.	0.01
Level A Assigning dedicated planen there posed and project. 00. Level B Assigning dedicated planen than to be na ddressed. 00. Level B Gring of the dedicated planen than to be na ddressed. 00. Level B Gring of the dedicated planen than to be na ddressed. 00. Level B Continuation of Level C, plus the planner initial and maintain communication regarding the delivery and imaterials as necessary. 03. Level E Continuation of Level C, plus the planner initial and maintain communication regarding the delivery and imaterials and update the schedule and onsite inspection to track consumption of materials and report their status. 04. Level F Continuation of Level E, plus continued communication regarding consumption of materials and report their status. 04. Level C Some information sch as materials and equipment requirements are addressed. 00. Level C Some information sch as a materials and equipment requirements are addressed. 03. Level C Some information sch as a materials and each CVP has been addressed. 00. Level C Some information sch as a starefis and need conscheration for most of the sections are given. 0.6 Level C Continuation of Level D, plus in-depth consideration for most of the section are given. 0.6 Level C S	Level F	Continuation of Level F plus the work package status is undated electronically	0.76
Level A Assigning dedicated planner is not applicable to this building project. 0.0 Level B Hring of the dedicated planner is not applicable to this building project. 0.1 Level C Single planner or multiple personnel who prepare initial project schedule and coordinate materials procurement process are assigned. 0.2 Level D Continuation of Level C, plus the planner initiate and maintain communication regarding the delivery and inspection of materials and update the schedules as appropriate. 0.3 Level E Continuation of Level E, plus continued communication regarding consumption of materials and report their status. 0.6 Level R Continuation of Level E, plus continued communication regarding consumption. 0.0 Level B Sacie information such as the scope of works, safety, quality of work, materials, personnel and equipment requirements is complete. 0.0 Level E Continuation of Level D, plus in-depth consideration. 0.6 0.6 Level E Continuation of Level D, plus in-depth consideration. 0.6 0.6 Level B Buildability review is not applicable to this information such as the scope of works, safety, quality of work, materials, personnel and equipment requirements is complete. 0.6 Level E Continuation of Level D, plus in-depth consideration. 0.6 0.6 Level E	Level I	D4 Dedicated Planner	0.70
Level B Hiring of the dedicated planner has not been addressed. 0.1 Level C Single planner or multiple personnel who prepare initial project schedule and coordinate materials 0.2 Level D Continuation of Level C, plus the planner initiate and maintain communication regarding the delivery and instains an encessary. 0.3 Level E Continuation of Level D, plus continued update of schedule and onsite inspection to track consumption of materials an encessary. 0.4 Level B Satis. D5. Construction Work Packages 0.6 Level B Satis. D5. Construction Work Packages 0.1 Level B Satis. D5. Construction Or Norks, Satety, quality of work, materials, personnel and equipment requirements is compiled. 0.3 Level C Continuation of Level D, plus in-depth consideration of most of the sections are given. 0.6 Level B Package and be performed easily base on the information documented for each CWP. 0.0 Level A Baildability review is not adplicable to this building project. 0.0	Level A	Assigning dedicated planner is not applicable to this building project.	0.00
Level C Single planner or multiple personnel who prepare initial project schedule and coordinate materials 0.2 Level D Continuation of Level C. plus the planner initiate and maintain communication regarding the delivery and inspection of materials and update the schedules as appropriate. 0.3 Level E Continuation of Level D. plus continued update of schedule and onsite inspection to track consumption of materials and report their status. 0.4 Level F Continuation of Level E, plus continued communication regarding consumption of materials and report their status. 0.5 Level A Organizing information such as the scope of works, safety, quality of work, materials, personnel and equipment requirements are addressed. 0.0 Level C Some information such as the scope of works, safety, quality of work, materials, personnel and equipment requirements is on organizing information in each as the scope of works, safety, quality of work, materials, personnel and equipment requirements is on organized, but the information is not complete. 0.0 Level C Continuation of Level D, plus in-depth consideration from sost of the sections are given. 0.6 Level A Buildability review is not applicable to this building project. 0.0 Level A Buildability review is not addressed 0.1 Level B Buildability review is not addressed in this project. 0.0	Level B	Hiring of the dedicated planner has not been addressed.	0.12
Level C procurement process are assigned. 0.2 Level D Continuation of Level C, plus the planner initiate and maintain communication regarding the delivery and inspection of materials an update the schedules as appropriate. 0.3 Level F Exercision of materials an update the schedules as appropriate. 0.4 Level F Exercision of Level C, plus continued update of schedule and onsite inspection to track consumption of materials and report their status. 0.6 Level F Section formation sch as materials and equipment requirements are addressed. 0.1 Level B Sing information for each Construction Work Packages (CWP) is not applicable. 0.0 Level C Some information sch as materials and equipment requirements are addressed. 0.1 Level C Some information sch as materials and consideration. 0.5 Level D All information required for execution of each CWP has been addressed. However, the information in each section displays no in-depth consideration for most of the sections are given. 0.6 Level F Continuation of Level E, plus all sections have received in-depth consideration. The construction work package can be performed easily based on the information decumented for each CWP. 0.0 Level B Buildability review is not addressed 0.1 Level B Buildability review is not addressed 0.1		Single planner or multiple personnel who prepare initial project schedule and coordinate materials	
Level D Continuation of Level C, plus the planner initiate and maintain communication regarding the delivery and inspection of materials and update the schedules as appropriate. 0.3 Level E Continuation of Level D, plus continued update of schedule and onsite inspection to track consumption of materials and report their status. 0.4 Level F Continuation of Level D, plus continued communication regarding consumption of materials and report their status. 0.6 Level A Organizing information such as materials and equipment requirements are addressed. 0.0 Level B Basic information such as the scope of works, safetry quality of work, materials, personnel and equipment requirements is compiled, but the information is not compilete. 0.1 Level C Continuation of Level E, plus all sections have received in-depth consideration. The construction work package can be performed easily based on the information of cach CWP. 0.5 Level C Continuation of Level E, plus all sections have received in-depth consideration. The construction work package can be performed easily based on the information documented for each CWP. 0.6 Level B Buildability review is not addressed 0.1 Level C Some scheduling and coordination of the phases of construction have been performed by the contractor part of a detailed buildability review. 0.5 Level C Continuation of Level D, plus the ownere is involvoled in buildability review process.	Level C	procurement process are assigned.	0.24
Level E Continuation of Level D, plus continued update of schedules as appropriate. 0.3 Level F Continuation of Level D, plus continued update of schedule and onsite inspection to track consumption of materials as necessary. 0.4 Level F Continuation of Level E, plus continued communication regarding consumption of materials and report their status. 0.6 Level A Organizing information such as materials and equipment requirements are addressed. 0.1 Level C Some information such as materials and equipment requirements are addressed. 0.1 Level C Some information such as the scope of works, safety, quality of work, materials, personnel and equipment requirements is compliced, but the information in complete. 0.3 Level D Continuation of Level D, plus in-depth consideration. 0.6 Level F Continuation of Level D, plus in-depth consideration for most of the sections are given. 0.6 Level B Buildability review is not adplicable to this building project. 0.0 Level B Buildability review is not adplicable to this building project. 0.0 Level B Continuation of Level D, plus the contractor has prepared a detailed schedule for all phases of construction date seed on the information and compares. 0.7 Level B Some scheduling and coordination of the phases of construction dinthy review.	I 1D	Continuation of Level C, plus the planner initiate and maintain communication regarding the delivery and	0.27
Level E Continuation of Level D, plus continued update of schedule and onsite inspection to track consumption of materials an necessary. 0.4 Level F Continuation of Level D, plus continued communication regarding consumption of materials and report their status. 0.6 Level A Organizing information for each Construction Work Packages 0.0 Level A Organizing information such as the scope of works, safety, quality of work, materials, personnel and equipment requirements is compiled, but the information is not compilet. 0.0 Level C Some information such as the scope of works, safety, quality of work, materials, personnel and equipment requirements is compiled, but the information is not compilet. 0.5 Level C Continuation of Level D, plus in-depth consideration for most of the sections are given. 0.5 Level E Continuation of Level E, plus all sections have received in-depth consideration. The construction work package can be performed easily based on the information odcumented for each CWP. 0.0 Level A Buildability review is not applicable to this building project. 0.0 Level C Some scheduling and coordination of the phases of construction have been performed by the contractor bas for conducting detailed buildability review. 0.7 Level C Continuation of Level D, plus the contractor has prepared a detailed schedule for all phases of construction detailed buildabili	Level D	inspection of materials and update the schedules as appropriate.	0.37
Level E materials as necessary. 0.4 Level F Continuation of Level E, plus continued communication regarding consumption of materials and report their 0.6 Level A Organizing information such as materials and equipment requirements are addressed. 0.0 Level B Basic information such as materials and equipment requirements are addressed. 0.1 Level C Some information such as materials and equipment requirements are addressed. 0.1 Level D Some information such as the scope of works, safety, quality of work, materials, personnel and equipment requirements is compliced. 0.3 Level D Continuation required for execution of each CWP has been addressed. However, the information in each section displays no in-depth consideration for most of the sections are given. 0.6 Level E Continuation of Level E, plus all sections have received in-depth consideration. The construction work package can be performed easily based on the information documented for each CWP. 0 Level B Buildability review is not applicable to this building project. 0.0 Level C Some scheduling and coordination of the phases of construction have been performed by the contractor after conducting detailed buildability review. 0.5 Level E Continuation of Level E, plus the outractor has prepared a detailed schedule for all phases of construction after conducting detailed buildability review. <td>T 15</td> <td>Continuation of Level D, plus continued update of schedule and onsite inspection to track consumption of</td> <td>0.40</td>	T 15	Continuation of Level D, plus continued update of schedule and onsite inspection to track consumption of	0.40
Level F Continuation of Level E, plus continued communication regarding consumption of materials and report their 0.6 Evel A Organizing information for each Construction Work Packages 0.0 Level B Baics information such as the scope of works, safety, quality of work, materials, personnel and equipment requirements is compiled, but the information is not complete. 0.1 Level D Information required for execution of each CWP has been addressed. 0.1 Level D Information required for execution of each CWP. 0.5 Level D Continuation of Level D, plus in-depth consideration. 0.6 Level F Continuation of Level D, plus in-depth consideration for most of the sections are given. 0.6 Level B Baildability review is not applicable to this building project. 0.0 Level B Buildability review is not applicable to this building project. 0.1 Level C Some scheduling and coordination of the phases of construction have been performed by the contractor before conducting detailed buildability review. 0.3 Level E Continuation of Level C, plus the owner is involved in buildability review process. 0.7 Level C Continuation of Level C, plus the owner is involved in buildability review process includes listing the continuatin of Level D, plus the owner is involved in buildability review	Level E	materials as necessary.	0.48
Level A Organizing information such as materials and equipment requirements are addressed. 0.0 Level A Organizing information such as materials and equipment requirements are addressed. 0.1 Level C Some information such as the scope of works, safety, quality of work, materials, personnel and equipment requirements is compiled, but the information is not complete. 0.3 Level D All information required for execution of each CWP has been addressed. However, the information in each scage can be performed easily based on the information documented for each CWP. 0.5 Level E Continuation of Level E, plus all sections have received in-depth consideration. The construction work package can be performed easily based on the information documented for each CWP. 0.6 Level B Buildability review is not adplicable to this building project. 0.0 Level B Some scheduling and coordination of the phases of construction have been performed by the contractor base for construction after conducting detailed buildability review. 0.3 Level C Some scheduling and coordination of the sprepared a detailed schedule for all phases of construction after construction of Level C, plus the contractor has prepared a detailed buildability review. 0.3 Level E Continuation of Level D, plus the owner is involved in buildability review process. 0.7 Level E Utility alignment is not adplicable to thi	Laval E	Continuation of Level E, plus continued communication regarding consumption of materials and report their	0.61
D5. Construction Work Packages 0.0 Level A Organizing information such as materials and equipment requirements are addressed. 0.1 Level B Basic information such as materials and equipment requirements is compiled, but the information is not complete. 0.3 Level C Some information such as the scope of works, safety, quality of work, materials, personnel and equipment requirements is compiled, but the information is not complete. 0.3 Level D All information required for execution of each CWP has been addressed. However, the information in each section displays no in-depth consideration. 0.6 Level F Continuation of Level D, plus in-depth consideration for most of the sections are given. 0.6 Level A Buildability review is not applicable to this building project. 0.0 Level B Buildability review is not applicable to this building project. 0.0 Level C Some scheduling and coordination of the phases of construction have been performed by the contractor before conducting detailed buildability review. 0.5 Level E Continuation of Level D, plus in-depth 0.0 Level A Utility alignment is not applicable to this project. 0.0 Level C Continuation of Level D, plus the orntractor has prepared a detailed schedule for all phases of construction. 0.8	Level F	status.	0.61
Level A Organizing information for each Construction Work Packages (CWP) is not applicable. 000 Level B Basic information such as materials and equipment requirements are addressed. 0.1 Level C Some information such as the scope of works, safety, quality of work, materials, personnel and equipment requirements is compiled, but the information is of complete. 0.3 Level D All information required for execution of each CWP has been addressed. However, the information in each section displays no in-depth consideration. 0.6 Level E Continuation of Level E, plus all sections have received in -depth consideration. 0.6 Level F B initiability review is not addressed 0.0 Level B Buildability review is not addressed 0.1 Level C Some scheduling and coordination of the phases of construction have been performed by the contractor before conducting detailed buildability review. 0.5 Level E Continuation of Level C, plus the owner is involved in buildability review process. 0.7 Level E Continuation of Level D, plus the owner is involved in buildability review process. 0.7 Level A Utility alignment is not applicable to this project. 0.0 Level A Utility alignment is not applicable to this project. 0.0 Level C Continuation of Level C, plus and equipment requirements are surrounding utilities. The process includes listing the caview process for working in areas surrounding utilities. The process includes listing the plain is		D5. Construction Work Packages	
Level B Basic information such as materials and equipment requirements are addressed. 0.1 Level C Some information such as the scope of works, safety, quality of work, materials, personnel and equipment requirements is compiled, but the information is not complete. 0.3 Level D All information required for execution of each CWP has been addressed. However, the information in each section displays no in-depth consideration. 0.5 Level E Continuation of Level D, plus in-depth consideration for most of the sections are given. 0.6 Level A Buildability review is not applicable to this building project. 0.0 Level B Buildability review is not applicable to this buildability review 0.3 Level C Some scheduling and coordination of the phases of construction have been performed by the contractor before conducting detailed buildability review. 0.3 Level D Continuation of Level C, plus the contractor has prepared a detailed schedule for all phases of construction after conducting detailed buildability review. 0.5 Level E Continuation of Level D, plus the owner is involved in buildability review process. 0.7 Level A Utility alignment is not applicable to this project. 0.0 Level C The project has a review process for working in areas surrounding utilities. The process includes listing the existing utilities, identifying their locations, and providing warn	Level A	Organizing information for each Construction Work Packages (CWP) is not applicable.	0.00
Level C Some information such as the scope of works, safety, quality of work, materials, personnel and equipment requirements is compiled, but the information is not complete. 0.3 Level D Information of Level D, plus in-depth consideration. 0.5 Level E Continuation of Level E, plus all sections have received in-depth consideration. The construction work package can be performed easily based on the information documented for each CWP. 0.8 Level F Buildability review is not applicable to this building project. 0.0 Level B Buildability review is not applicable to this building project. 0.0 Level C Some scheduling and coordination of the phases of construction have been performed by the contractor after conducting detailed buildability review. 0.3 Level D Continuation of Level D, plus the contractor has prepared a detailed schedule for all phases of construction after conducting detailed buildability review. 0.5 Level E Continuation of Level D, plus the owner is involved in buildability review process. 0.7 Level E Continuation of Level D, plus the owner is involved in buildability review process. 0.7 Level B Utility alignment is not adpressed in this project. 0.0 Level C Continuation of Level D, plus the construction working in areas surrounding utilities. The process includes listing the plasting utilities, integrated, and providing warning	Level B	Basic information such as materials and equipment requirements are addressed.	0.17
Level C requirements is compiled, but the information is not complete. 0.5 Level D All information required for execution of each CWP has been addressed. However, the information in each section displays no in-depth consideration. 0.5 Level E Continuation of Level D, plus in-depth consideration for most of the sections are given. 0.6 Level F Continuation of Level D, plus in-depth consideration for most of the sections are given. 0.6 Level A Buildability review is not applicable to this building project. 0.0 Level B Buildability review is not addressed 0.1 Level C Continuation of Level C, plus the contractor has prepared a detailed schedule for all phases of construction after conducting detailed buildability review. 0.5 Level C Continuation of Level D, plus the owner is involved in buildability review process. 0.7 Level C Continuation of Level D, plus the owner is involved in buildability review process. 0.7 Level C Continuation of Level D, plus all buildability review and schedules are completed before construction. 0.8 Level C The project has a review process for working in areas surrounding utilities. The process includes listing the existing utilities, identifying their locations, and providing warning signs. 0.3 Level C Continuation of Level D, plus communication with bu	Laval C	Some information such as the scope of works, safety, quality of work, materials, personnel and equipment	0.24
Level D All information required for execution of each CWP has been addressed. However, the information in each section displays no in-depth consideration. 0.5 Level E Continuation of Level D, plus in-depth consideration for most of the sections are given. 0.6 Level F Continuation of Level D, plus all sections have received in-depth consideration. The construction work package can be performed easily based on the information documented for each CWP. 0.0 Level A Buildability review is not applicable to this building project. 0.0 Level D Boindability review is not addressed 0.1 Level D Continuation of Level D, plus the contractor has prepared a detailed schedule for all phases of construction before conducting detailed buildability review. 0.5 Level E Continuation of Level D, plus the owner is involved in buildability review process. 0.7 Level E Continuation of Level D, plus the womer is involved in buildability review process. 0.7 Level B Utility alignment is not applicable to this project. 0.0 Level C The project has a review process for working in areas surrounding utilities. The process includes listing the plane hypicable to this project. 0.3 Level C The project has a review process for working in areas surrounding utilities. The process includes listing the plane hypicable to this project. 0.3	LevelC	requirements is compiled, but the information is not complete.	0.54
Level D section displays no in-depth consideration. 0.5 Level E Continuation of Level E, plus all sections have received in-depth consideration. The construction work 0.6 Level F Continuation of Level E, plus all sections have received in-depth consideration. The construction work 0.8 Level A Buildability review is not applicable to this building project. 0.0 Level A Buildability review is not applicable to this building project. 0.0 Level C Some scheduling and coordination of the phases of construction have been performed by the contractor before conducting detailed buildability review. 0.5 Level C Continuation of Level C, plus the contractor has prepared a detailed schedule for all phases of construction after conducting detailed buildability reviews and schedules are completed before construction. 0.5 Level E Continuation of Level E, plus all buildability reviews and schedules are completed before construction. 0.8 Level A Utility alignment is not applicable to this project. 0.0 Level B Continuation of Level C, plus necessary city council approvals have been obtained and utility adjustment plan is prepared. 0.1 Level C The project has a review process for working in areas surrounding utilities. The process includes listing the existing utilities, identifying their locations, and providing warning signs. 0.4 <td>LavalD</td> <td>All information required for execution of each CWP has been addressed. However, the information in each</td> <td>0.51</td>	LavalD	All information required for execution of each CWP has been addressed. However, the information in each	0.51
Level E Continuation of Level E, plus all sections have received in-depth consideration. The construction work package can be performed easily based on the information documented for each CWP. 0.6 Level F Continuation of Level E, plus all sections have received in-depth consideration. The construction work package can be performed easily based on the information documented for each CWP. 0.8 Level A Buildability review is not applicable to this building project. 0.0 Level B Some scheduling and coordination of the phases of construction have been performed by the contractor before conducting detailed buildability review. 0.3 Level D Continuation of Level C, plus the contract or has prepared a detailed schedule for all phases of construction after conducting detailed buildability review. 0.5 Level E Continuation of Level C, plus the contract or has prepared a detailed schedule for all phases of construction after conducting detailed buildability reviews. 0.7 Level F Continuation of Level C, plus the contract or has prepared a detailed schedule for all phases of construction. 0.8 Level A Utility alignment is not applicable to this project. 0.0 Level B Utility alignment is not applicable to this project. 0.0 Level C Continuation of Level D, plus communication with businesses and local community have been made. 0.4 Level D Continuation of Level	Level D	section displays no in-depth consideration.	0.51
Level F Continuation of Level E, plus all sections have received in-depth consideration. The construction work 0.8 D6. Buildability review D6. Buildability review 0.0 Level A Buildability review is not addressed 0.0 Continuation of Level C, plus the contractor has prepared a detailed schedule for all phases of construction after conducting detailed buildability review. 0.3 Level D Continuation of Level C, plus the contractor has prepared a detailed schedule for all phases of construction after conducting detailed buildability review. 0.5 Level E Continuation of Level D, plus the owner is involved in buildability review process. 0.7 Level E Continuation of Level E, plus all buildability reviews and schedules are completed before construction. 0.8 D7. Utilities Alignment 0.0 0.1 0.1 Level B Utility alignment is not addressed in this project. 0.0 0.0 Level C The project has a review process for working in areas surrounding utilities. The process includes listing the existing utilities, identifying their locations, and providing warning signs. 0.3 Level C Continuation of Level C, plus necessary city council approvals have been obtained and utility adjustment plan is prepared. 0.4 Level E Continuation of	Level E	Continuation of Level D, plus in-depth consideration for most of the sections are given.	0.69
Level 1 package can be performed easily based on the information documented for each CWP. Of D6. Buildability Review 0.6 Level A Buildability review is not addressed 0.1 Level B Buildability review is not addressed 0.3 Level C Some scheduling and coordination of the phases of construction have been performed by the contractor before conducting detailed buildability review. 0.3 Level D Continuation of Level C, plus the contractor has prepared a detailed schedule for all phases of construction after conducting detailed buildability review. 0.5 Level F Continuation of Level D, plus the owner is involved in buildability review process. 0.7 Level F Continuation of Level E, plus all buildability reviews and schedules are completed before construction. 0.8 D7. Utilities Alignment 0.0 0.0 0.0 Level A Utility alignment is not applicable to this project. 0.0 0.0 Level B Continuation of Level C, plus necessary city council approvals have been obtained and utility adjustment plan is prepared. 0.3 Level C Continuation of Level D, plus communication with businesses and local community have been made. 0.6 Continuation of Level D, plus the physical markings descri	Level F	Continuation of Level E, plus all sections have received in-depth consideration. The construction work	0.86
D6. Buildability Review Level A Buildability review is not applicable to this building project. 0.0 Level B Buildability review is not addressed 0.1 Level C Some scheduling and coordination of the phases of construction have been performed by the contractor before conducting detailed buildability review. 0.3 Level D Continuation of Level C, plus the contractor has prepared a detailed schedule for all phases of construction after conducting detailed buildability review. 0.5 Level E Continuation of Level D, plus the owner is involved in buildability review process. 0.7 Level F Continuation of Level D, plus the owner is involved in buildability review process includes listing the project has a review process for working in areas surrounding utilities. The process includes listing the contractors and providing warning signs. 0.0 Level B Continuation of Level C, plus communication with businesses and local community have been made. Continuation of Level C, plus communication with businesses and local community have been made. Continuation of Level E, plus the physical markings described to mark the horizontal route of underground 0.4 Level E Contract types are not applicable 0.1 Level E Continuation of Level C, plus communication with businesses and local community have been made. Continuation of Level C, plus the physical markings described to mark the horizontal route of underground		package can be performed easily based on the information documented for each CWP.	0.00
Level A Buildability review is not applicable to this building project. 0.0 Level B Buildability review is not addressed 0.1 Level C Some scheduling and coordination of the phases of construction have been performed by the contractor after conducting detailed buildability review. 0.3 Level D Continuation of Level C, plus the contractor has prepared a detailed schedule for all phases of construction after conducting detailed buildability review. 0.5 Level F Continuation of Level D, plus the owner is involved in buildability review process. 0.7 Level F Continuation of Level D, plus the owner is involved in buildability review process. 0.7 Level A Utility alignment is not applicable to this project. 0.0 Level B Utility alignment is not applicable to this project. 0.1 Level C The project has a review process for working in areas surrounding utilities. The process includes listing the existing utilities, identifying their locations, and providing warning signs. 0.3 Level D Continuation of Level D, plus communication with businesses and local community have been made. 0.6 Arrangements for temporary shutoffs and interruptions have been made. 0.6 0.7 Level F Continuation of Level D, plus communication with businesses and local community have been made. 0.7 Level F<		D6. Buildability Review	
Level B Buildability review is not addressed 0.1 Level C Some scheduling and coordination of the phases of construction have been performed by the contractor before conducting detailed buildability review. 0.3 Level D Continuation of Level C, plus the contractor has prepared a detailed schedule for all phases of construction after conducting detailed buildability review. 0.5 Level E Continuation of Level D, plus the owner is involved in buildability review process. 0.7 Level F Continuation of Level E, plus all buildability reviews and schedules are completed before construction. 0.8 Devel A Utility alignment is not applicable to this project. 0.0 Level C The project has a review process for working in areas surrounding utilities. The process includes listing the existing utilities, identifying their locations, and providing warning signs. 0.3 Level D Continuation of Level D, plus communication with businesses and local community have been made. 0.6 Arrangements for temporary shutoffs and interruptions have been made. 0.7 0.7 Level A Continuation of Level E, plus the physical markings described to mark the horizontal route of underground 0.4 Level F facilities have been protect, detours have been constructed, and all activities related to utilities alignments 0.7 Level B Conti	Level A	Buildability review is not applicable to this building project.	0.00
Level C Some scheduling and coordination of the phases of construction have been performed by the contractor before conducting detailed buildability review. 0.3 Level D Continuation of Level C, plus the contractor has prepared a detailed schedule for all phases of construction after conducting detailed buildability review. 0.5 Level F Continuation of Level D, plus the owner is involved in buildability review process. 0.7 Level F Continuation of Level D, plus the owner is involved in buildability review process. 0.7 Level A Utility alignment is not applicable to this project. 0.0 Level D Continuation of Level C, plus necessary city council approvals have been obtained and utility adjustment plan is prepared. 0.3 Level B Continuation of Level D, plus communication with businesses and local community have been made. Continuation of Level D, plus communication with businesses and local community have been made. Continuation of Level E, plus the physical markings described to mark the horizontal route of underground have been integrated with other project schedules. 0.7 Level F Accontracting strategies are not applicable 0.0 Level F Continuation of Level C, plus the contract types. It considers fee structure and financing options, but risks associated with other project schedules. 0.4 Level F Accontracting strategies are not applicable 0.0 0.0 <td< td=""><td>Level B</td><td>Buildability review is not addressed</td><td>0.18</td></td<>	Level B	Buildability review is not addressed	0.18
before conducting detailed buildability review. 0.5 Level D Continuation of Level C, plus the contractor has prepared a detailed schedule for all phases of construction after conducting detailed buildability review. 0.5 Level E Continuation of Level D, plus the owner is involved in buildability review process. 0.7 Level F Continuation of Level E, plus all buildability reviews and schedules are completed before construction. 0.8 D D.7. Utilities Alignment 0.0 Level A Utility alignment is not applicable to this project. 0.1 Level C The project has a review process for working in areas surrounding utilities. The process includes listing the existing utilities, identifying their locations, and providing warning signs. 0.4 Level D Continuation of Level C, plus necessary city council approvals have been obtained and utility adjustment plan is prepared. 0.4 Level E Continuation of Level D, plus the physical markings described to mark the horizontal route of underground 0.4 Level F facilities have been protected, detours have been constructed, and all activities related to utilities alignments 0.7 Level F A Contracting strategies are not applicable 0.0 0.6 Level E Continuation of Level E, plus the physical markings described to mark the horizontal route of underground	Level C	Some scheduling and coordination of the phases of construction have been performed by the contractor	0.35
Level D Continuation of Level C, plus the contractor has prepared a detailed schedule for all phases of construction after conducting detailed buildability review. 0.5 Level E Continuation of Level D, plus the owner is involved in buildability review process. 0.7 Level F Continuation of Level D, plus the owner is involved in buildability review process. 0.7 Level F Continuation of Level D, plus the owner is involved in buildability review process. 0.7 Level A Utility alignment is not applicable to this project. 0.1 Level C The project has a review process for working in areas surrounding utilities. The process includes listing the existing utilities, identifying their locations, and providing warning signs. 0.3 Level D Continuation of Level D, plus communication with businesses and local community have been made. 0.6 Continuation of Level E, plus the physical markings described to mark the horizontal route of underground 0.4 Level F facilities have been protected, detours have been constructed, and all activities related to utilities alignments have been integrated with other project schedules. 0.6 Level A Contracting strategies are not applicable 0.0 Level A Contracting strategies are not adplicable 0.0 Level B Contract types are not addressed 0.1		before conducting detailed buildability review.	
after conducting detailed buildability review. Level E Continuation of Level D, plus the owner is involved in buildability review process. 0.7 Level F Continuation of Level D, plus all buildability reviews and schedules are completed before construction. 0.8 D7. Utilities Alignment Level A Utility alignment is not applicable to this project. 0.1 Level B Utility alignment is not addressed in this project. 0.1 Level C The project has a review process for working in areas surrounding utilities. The process includes listing the existing utilities, identifying their locations, and providing warning signs. Level D Continuation of Level C, plus necessary city council approvals have been obtained and utility adjustment plan is prepared. Level E Continuation of Level D, plus communication with businesses and local community have been made. Continuation of Level E, plus the physical markings described to mark the horizontal route of underground Level F facilities have been protected, detours have been constructed, and all activities related to utilities alignments have been integrated with other project schedules. D8. Contract Types Level A Contract types are not applicable Level B Contract types are not applicable Level C The project has a review process for the contract types. It considers fee structure and financing options, but risks associated with different contract types are not considered. Level C Continuation of Level C, plus risks associated with different contract types are considered. Cost-benefit analyses for different types of contracts are performed. Level E Continuation of Level E, plus the loadur considered. Level F regularly.The company has established protocols for selecting designers, material suppliers, and manufacturers.	Level D	Continuation of Level C, plus the contractor has prepared a detailed schedule for all phases of construction	0.53
Level F Continuation of Level D, plus the owner is involved in buildability review process. 0.7 Level F Continuation of Level E, plus all buildability reviews and schedules are completed before construction. 0.8 D7. Utilities Alignment 0.0 Level A Utility alignment is not applicable to this project. 0.0 Level B Utility alignment is not addressed in this project. 0.1 Level C The project has a review process for working in areas surrounding utilities. The process includes listing the existing utilities, identifying their locations, and providing warning signs. 0.3 Level D Continuation of Level C, plus necessary city council approvals have been obtained and utility adjustment plan is prepared. 0.4 Level E Continuation of Level D, plus communication with businesses and local community have been made. 0.6 Continuation of Level E, plus the physical markings described to mark the horizontal route of underground Level F facilities have been protected, detours have been constructed, and all activities related to utilities alignments bave been integrated with other project schedules. 0.0 D8. Contract Types 0.0 0.0 0.0 0.0 0.0 Level C The project has a review process for the contract types. It considers fee structure and financing options, but risks associated with different contract types are considered. Cost-benefit analyses for dif		after conducting detailed buildability review.	
Level F Continuation of Level E, plus all buildability reviews and schedules are completed before construction. 0.8 D7. Utilities Alignment 0.0 Level A Utility alignment is not applicable to this project. 0.1 Level C The project has a review process for working in areas surrounding utilities. The process includes listing the existing utilities, identifying their locations, and providing warning signs. 0.3 Level D Continuation of Level C, plus necessary city council approvals have been obtained and utility adjustment plan is prepared. 0.4 Level E Continuation of Level D, plus communication with businesses and local community have been made. 0.6 Continuation of Level E, plus the physical markings described to mark the horizontal route of underground 0.7 Level F facilities have been protected, detours have been constructed, and all activities related to utilities alignments have been integrated with other project schedules. 0.7 D8 Contract Types 0.0 Level A Contracting strategies are not applicable 0.0 Level C The project has a review process for the contract types. It considers fee structure and financing options, but risks associated with different contract types are not considered. 0.3 Level C Continuation of Level C, plus risks associated with different contract types are considered. 0.3 </td <td>Level E</td> <td>Continuation of Level D, plus the owner is involved in buildability review process.</td> <td>0.70</td>	Level E	Continuation of Level D, plus the owner is involved in buildability review process.	0.70
Level A Utilities and applicable to this project. 0.0 Level B Utility alignment is not addressed in this project. 0.1 Level C The project has a review process for working in areas surrounding utilities. The process includes listing the existing utilities, identifying their locations, and providing warning signs. 0.3 Level D Continuation of Level C, plus necessary city council approvals have been obtained and utility adjustment plan is prepared. 0.4 Level E Continuation of Level D, plus communication with businesses and local community have been made. 0.6 Continuation of Level E, plus the physical markings described to mark the horizontal route of underground 0.7 Level F facilities have been protected, detours have been constructed, and all activities related to utilities alignments is and applicable 0.7 Level B Contracting strategies are not applicable 0.0 Level B Contract types are not addressed 0.1 Level C The project has a review process for the contract types. It considers fee structure and financing options, but analyses for different tornact struct structure and financing options, but analyses for different types of contracts are performed. 0.4 Level D Continuation of Level C, plus the labour market information and reports are studied and considered before awarding contracts. 0.4 Level C	Level F	Continuation of Level E, plus all buildability reviews and schedules are completed before construction.	0.88
Level A Othny alignment is not applicable to this project. 0.0 Level B Utility alignment is not addressed in this project. 0.1 Level C The project has a review process for working in areas surrounding utilities. The process includes listing the existing utilities, identifying their locations, and providing warning signs. 0.3 Level D Continuation of Level C, plus necessary city council approvals have been obtained and utility adjustment plan is prepared. 0.4 Level E Continuation of Level D, plus communication with businesses and local community have been made. Continuation of Level E, plus the physical markings described to mark the horizontal route of underground Level F facilities have been protected, detours have been constructed, and all activities related to utilities alignments is associated with other project schedules. 0.7 Level B Contracting strategies are not applicable 0.0 Level C The project has a review process for the contract types. It considers fee structure and financing options, but analyses for different contract types are not considered. 0.3 Level D Continuation of Level C, plus the labour market information and reports are studied and considered before awarding contracts. 0.4 Level C Continuation of Level D, plus the labour market information and reports are studied and monitored 0.4 Level B Continuation of Level D, plus the local conditions in the project area are studied and moni	T1 A	D/. Utilities Alignment	0.00
Level B Ontify argination is not addressed in this project. 0.1 Level C The project has a review process for working in areas surrounding utilities. The process includes listing the existing utilities, identifying their locations, and providing warning signs. 0.3 Level D Continuation of Level C, plus necessary city council approvals have been obtained and utility adjustment plan is prepared. 0.4 Level E Continuation of Level D, plus communication with businesses and local community have been made. Continuation of Level E, plus the physical markings described to mark the horizontal route of underground Level F facilities have been protected, detours have been constructed, and all activities related to utilities alignments have been integrated with other project schedules. 0.7 Level B Contract Types 0.0 Level B Contract types are not applicable 0.0 Level B Contract types are not applicable 0.3 Level D Continuation of Level C, plus risks associated with different contract types are not considered. 0.3 Level D Continuation of Level C, plus risks associated with different tontract types are studied and considered before analyses for different types of contracts are performed. 0.4 Level D Continuation of Level D, plus the local conditions in the project area are studied and monitored Level F regularly. The company has established protocols for selecting designers, material suppliers, and 0.7	Level A	Utility alignment is not applicable to this project.	0.00
Level C The project has a review process for working in areas surrounding utilities. The process includes listing the existing utilities, identifying their locations, and providing warning signs. 0.3 Level D Continuation of Level C, plus necessary city council approvals have been obtained and utility adjustment plan is prepared. 0.4 Level E Continuation of Level D, plus communication with businesses and local community have been made. 0.6 Continuation of Level E, plus the physical markings described to mark the horizontal route of underground 0.7 Level F facilities have been protected, detours have been constructed, and all activities related to utilities alignments have been integrated with other project schedules. 0.7 Level B Contracting strategies are not applicable 0.0 Level B Contract types are not addressed 0.1 Level C The project has a review process for the contract types. It considers fee structure and financing options, but risks associated with different contract types are considered. 0.3 Level D Continuation of Level C, plus risks associated with different contract types are considered. 0.4 Level D Continuation of Level C, plus risks associated with different. 0.3 Level B Contract types of contracts are performed. 0.4 Level D Continuation of Level D, plus the labour market infor	Level B	The project has a review process for working in group surrounding utilities. The process includes listing the	0.15
Level D Continuation of Level C, plus necessary city council approvals have been obtained and utility adjustment plan is prepared. 0.4 Level E Continuation of Level D, plus communication with businesses and local community have been made. Continuation of Level E, plus the physical markings described to mark the horizontal route of underground Level F facilities have been protected, detours have been constructed, and all activities related to utilities alignments have been integrated with other project schedules. 0.6 Level F Contracting strategies are not applicable 0.0 Level B Contract types are not addressed 0.1 Level C The project has a review process for the contract types. It considers fee structure and financing options, but risks associated with different contract types are not considered. 0.4 Level D Continuation of Level C, plus risks associated with different contract types are considered. 0.3 Level D Continuation of Level D, plus the labour market information and reports are studied and considered before awarding contracts. 0.6 Continuation of Level E, plus the local conditions in the project area are studied and monitored 0.4 Level F regularly.The company has established protocols for selecting designers, material suppliers, and 0.7	Level C	avioting utilities identifying their leastions and providing warning signs	0.30
Level D plan is prepared. 0.4 Level E Continuation of Level D, plus communication with businesses and local community have been made. Arrangements for temporary shutoffs and interruptions have been made. Continuation of Level E, plus the physical markings described to mark the horizontal route of underground 0.6 Level F facilities have been protected, detours have been constructed, and all activities related to utilities alignments have been integrated with other project schedules. 0.7 Level A Contracting strategies are not applicable 0.0 Level B Contract types are not addressed 0.1 Level C The project has a review process for the contract types. It considers fee structure and financing options, but risks associated with different contract types are not considered. 0.3 Level D Continuation of Level C, plus risks associated with different contract types are considered. 0.4 Level E Continuation of Level D, plus the labour market information and reports are studied and considered before awarding contracts. 0.6 Level E Continuation of Level E, plus the local conditions in the project area are studied and monitored 0.6 Level F regularly.The company has established protocols for selecting designers, material suppliers, and manufacturers. 0.7		Continuation of Level C, plus necessary city council approvals have been obtained and utility adjustment	0.46
Level E Continuation of Level D, plus communication with businesses and local community have been made. Arrangements for temporary shutoffs and interruptions have been made. Continuation of Level E, plus the physical markings described to mark the horizontal route of underground 0.6 Level F facilities have been protected, detours have been constructed, and all activities related to utilities alignments have been integrated with other project schedules. 0.7 Level A Contracting strategies are not applicable 0.0 Level B Contract types are not addressed 0.1 Level C The project has a review process for the contract types. It considers fee structure and financing options, but risks associated with different contract types are not considered. 0.3 Level D Continuation of Level D, plus the labour market information and reports are studied and considered before awarding contracts. Continuation of Level D, plus the labour market information and reports are studied and monitored 0.6 Level F regularly.The company has established protocols for selecting designers, material suppliers, and manufacturers. 0.7	Level D	plan is prepared.	0.46
Level E Arrangements for temporary shutoffs and interruptions have been made. Continuation of Level E, plus the physical markings described to mark the horizontal route of underground 0.6 Level F facilities have been protected, detours have been constructed, and all activities related to utilities alignments have been integrated with other project schedules. 0.7 Level A Contracting strategies are not applicable 0.0 Level B Contract types are not addressed 0.1 Level C The project has a review process for the contract types. It considers fee structure and financing options, but risks associated with different contract types are not considered. 0.3 Level D Continuation of Level C, plus risks associated with different contract types are considered. 0.4 Level E Continuation of Level D, plus the labour market information and reports are studied and considered before awarding contracts. 0.6 Level F regularly.The company has established protocols for selecting designers, material suppliers, and manufacturers. 0.7	T 15	Continuation of Level D, plus communication with businesses and local community have been made.	0.61
Continuation of Level E, plus the physical markings described to mark the horizontal route of underground 0.7 Level F facilities have been protected, detours have been constructed, and all activities related to utilities alignments have been integrated with other project schedules. 0.7 Level A Contracting strategies are not applicable 0.0 Level B Contract types are not addressed 0.1 Level C The project has a review process for the contract types. It considers fee structure and financing options, but risks associated with different contract types are not considered. 0.3 Level D Continuation of Level C, plus risks associated with different contract types are considered. 0.4 Level E Continuation of Level D, plus the labour market information and reports are studied and considered before awarding contracts. 0.6 Level F regularly.The company has established protocols for selecting designers, material suppliers, and manufacturers. 0.7	Level E	Arrangements for temporary shutoffs and interruptions have been made.	0.61
Level F facilities have been protected, detours have been constructed, and all activities related to utilities alignments have been integrated with other project schedules. 0.7 D8. Contract Types D8. Contract Types Level A Contracting strategies are not applicable 0.0 Level B Contract types are not addressed 0.1 Level C The project has a review process for the contract types. It considers fee structure and financing options, but risks associated with different contract types are not considered. 0.3 Level D Continuation of Level C, plus risks associated with different contract types are considered. Cost-benefit analyses for different types of contracts are performed. 0.4 Level E Continuation of Level D, plus the labour market information and reports are studied and considered before awarding contracts. Continuation of Level E, plus the local conditions in the project area are studied and monitored 0.6 Level F regularly.The company has established protocols for selecting designers, material suppliers, and manufacturers. 0.7		Continuation of Level E, plus the physical markings described to mark the horizontal route of underground	
have been integrated with other project schedules. D8. Contract Types Level A Contracting strategies are not applicable 0.0 Level B Contract types are not addressed 0.1 Level C The project has a review process for the contract types. It considers fee structure and financing options, but risks associated with different contract types are not considered. 0.3 Level D Continuation of Level C, plus risks associated with different contract types are considered. Cost-benefit analyses for different types of contracts are performed. 0.4 Level E Continuation of Level D, plus the labour market information and reports are studied and considered before awarding contracts. 0.6 Continuation of Level E, plus the local conditions in the project area are studied and monitored 0.6 Level F regularly.The company has established protocols for selecting designers, material suppliers, and manufacturers. 0.7	Level F	facilities have been protected, detours have been constructed, and all activities related to utilities alignments	0.76
D8. Contract Types Level A Contracting strategies are not applicable 0.0 Level B Contract types are not addressed 0.1 Level C The project has a review process for the contract types. It considers fee structure and financing options, but risks associated with different contract types are not considered. 0.3 Level D Continuation of Level C, plus risks associated with different contract types are considered. Cost-benefit analyses for different types of contracts are performed. 0.4 Level E Continuation of Level D, plus the labour market information and reports are studied and considered before awarding contracts. Continuation of Level E, plus the local conditions in the project area are studied and monitored 0.6 Level F regularly.The company has established protocols for selecting designers, material suppliers, and manufacturers. 0.7		have been integrated with other project schedules.	
Level A Contracting strategies are not applicable 0.0 Level B Contract types are not addressed 0.1 Level C The project has a review process for the contract types. It considers fee structure and financing options, but risks associated with different contract types are not considered. 0.3 Level D Continuation of Level C, plus risks associated with different contract types are considered. 0.4 Level E Continuation of Level D, plus the labour market information and reports are studied and considered before awarding contracts. 0.6 Level F regularly.The company has established protocols for selecting designers, material suppliers, and manufacturers. 0.7		D8. Contract Types	
Level B Contract types are not addressed 0.1 Level C The project has a review process for the contract types. It considers fee structure and financing options, but risks associated with different contract types are not considered. 0.3 Level D Continuation of Level C, plus risks associated with different contract types are considered. 0.4 Level E Continuation of Level D, plus the labour market information and reports are studied and considered before awarding contracts. 0.6 Level F regularly.The company has established protocols for selecting designers, material suppliers, and manufacturers. 0.7	Level A	Contracting strategies are not applicable	0.00
Level C The project has a review process for the contract types. It considers fee structure and financing options, but risks associated with different contract types are not considered. 0.3 Level D Continuation of Level C, plus risks associated with different contract types are considered. Cost-benefit analyses for different types of contracts are performed. 0.4 Level E Continuation of Level D, plus the labour market information and reports are studied and considered before awarding contracts. 0.6 Level F regularly.The company has established protocols for selecting designers, material suppliers, and manufacturers. 0.7	Level B	Contract types are not addressed	0.16
Level C risks associated with different contract types are not considered. 0.3 Level D Continuation of Level C, plus risks associated with different contract types are considered. Cost-benefit analyses for different types of contracts are performed. 0.4 Level E Continuation of Level D, plus the labour market information and reports are studied and considered before awarding contracts. 0.6 Level F regularly.The company has established protocols for selecting designers, material suppliers, and manufacturers. 0.7	Level C	The project has a review process for the contract types. It considers fee structure and financing options, but	0.31
Level D Continuation of Level C, plus risks associated with different contract types are considered. Cost-benefit analyses for different types of contracts are performed. 0.4 Level E Continuation of Level D, plus the labour market information and reports are studied and considered before awarding contracts. 0.6 Level F regularly.The company has established protocols for selecting designers, material suppliers, and manufacturers. 0.7	LevelC	risks associated with different contract types are not considered.	0.51
Level D analyses for different types of contracts are performed. 0.4 Level E Continuation of Level D, plus the labour market information and reports are studied and considered before awarding contracts. 0.6 Level F regularly.The company has established protocols for selecting designers, material suppliers, and manufacturers. 0.7	Laval D	Continuation of Level C, plus risks associated with different contract types are considered. Cost-benefit	0.47
Level E Continuation of Level D, plus the labour market information and reports are studied and considered before awarding contracts. 0.6 Level F regularly. The company has established protocols for selecting designers, material suppliers, and manufacturers. 0.7	LEVELD	analyses for different types of contracts are performed.	0.47
awarding contracts. 0.0 Continuation of Level E, plus the local conditions in the project area are studied and monitored Level F regularly. The company has established protocols for selecting designers, material suppliers, and 0.7 manufacturers.	Level F	Continuation of Level D, plus the labour market information and reports are studied and considered before	0.62
Continuation of Level E, plus the local conditions in the project area are studied and monitored Level F regularly.The company has established protocols for selecting designers, material suppliers, and 0.7 manufacturers.	Level E	awarding contracts.	0.02
Level F regularly. The company has established protocols for selecting designers, material suppliers, and 0.7 manufacturers.		Continuation of Level E, plus the local conditions in the project area are studied and monitored	
manufacturers.	Level F	regularly.The company has established protocols for selecting designers, material suppliers, and	0.77
		manufacturers.	

Table 10-2 continued

CHAPETR 10

Summary and Conclusion

	Table 10-2 continued	
	D9. Model Development	Weights
Level A	Models are not required for this project.	0.00
Level B	Integration of the project's 3D and schedule information has not been addressed	0.11
Level C	A 4D Model has been developed for the project.	0.22
Level D	Continuation of Level C and the model is updated manually.	0.32
Level E	Continuation of Level D, plus the model is dynamic and includes material specifications, change order documentation, and other pertinent design and construction information.	0.43
Level F	Continuation of Level E, plus the model is automatically updated based on work progress.	0.54
	D10. Regulatory Requirements	
Level A	Regulatory requirements are not applicable.	0.00
Level B	Required permitting has not been addressed.	0.17
Level C	of the permits; the authority issuing the permits; pre-inspection requirements; and fees have been	0.34
Loval D	conducted.	0.51
Level D	Continuation of Level D plus a system is established to track the acquisition and validity period of each	0.51
Level E	permit.	0.68
Level F	schedule and the permit expire dates	0.85
	E. Human Resource Management Practices Scoring Tools	
	<i>E1. Crew Composition</i>	
Level A	Crew formation/composition is not applicable to this building project.	0.00
Level B	Crew composition is not addressed in this project.	0.15
Level C	Crew formation is addressed after the commencement of the construction of the project.	0.31
Level D	Crew formation is addressed before the commencement of the construction works. Experience and knowledge of employees, job requirements, and location of the project are considered while forming crew.	0.46
Level E	Continuation of Level D, plus the performance of the crew is assessed after completion of each task and	0.62
	Continuation of Level F, plus the crew formation and its performance is assessed regularly on daily weekly	
Level F	and monthly basis and necessary changes are made	0.77
	E2. Skill Assessment and Evaluation	,
Level A	Skills assessment is not applicable to this project.	0.00
Level B	Skills assessment is not addressed.	0.15
Level C	Skills assessment is addressed after the beginning of the construction of the project.	0.30
Loval D	Skills assessment is addressed before the beginning of construction work; and previous experience, skill and	0.44
Level D	knowledge of workers are considered during the assessment.	0.44
Level E	Continuation of Level D, plus the assessment is made after completion of each task and adjustments are made accordingly.	0.59
Lovol F	Continuation of Level E, plus the assessment is performed regularly on daily, weekly, and monthly basis,	0.74
Level	and necessary changes are made and monitored.	0.74
	E3. Employees Training	
Level A	Employees Training is not applicable to this building project.	0.00
Level B	Employees Training is not addressed.	0.15
Level C	Employees I raining is addressed on the jobsite after the beginning of the project.	0.30
Level D	training is provided to an employee when ne/sne begins working for the company and, if needed, extra training will be conducted on the job site.	0.45
Level E	Continuation of Level D, plus worker is certified to work in or supervise specific trade.	0.60
Level F	Continuation of Level E, plus a worker takes part in training regarding the new technologies that are	0.75
	Introduced in that trade on annual of and bi-annual basis.	
Level A	Career development is not applicable to this building project	0.00
Level B	Career development is not addressed in this project.	0.00
	The organization does not have a formal career development plan for employees, but the management	
Level C	discusses future plans with them informally.	0.28
Loval D	The organization has a formal career development plan for employees, but it only addresses short term	0.42
Level D	career developments.	0.43
Level E	Continuation of Level D, plus it addresses long term career developments.	0.57
Level F	Continuation of Level E, plus addresses the expected performance of the employee and how the performance will affect his/her career development.	0.71

Table 10-2 continued E5. Non-Financial Incentive Programs Weights Level A Non-financial incentive/recognition programs are not applicable to this building project. 0.00 Level B Non-financial incentive programs are not addressed in this project. 0.13 The organization has informal recognition programs and employees are recognized occasionally, but not in a Level C 0.26 formal manner Level D The organization has a formal recognition program that provides recognition on long-term basis. 0.39 Continuation of Level D, plus it recognizes workers on a regular basis for performance, positive safety results Level E 0.52 and good safety behaviour. Continuation of Level E, plus the rewards are given on both short and a long-term basis, and they are Level F 0.65 recognized by the upper management of the organization. E6. Financial Incentive Programs Level A Financial incentive program is not applicable to this building project. 0.00 Level B Financial incentive program is not addressed. 0.13 The organization has an informal incentive program that provides incentives occasionally, but not in a formal Level C 0.27 manner. Level D The organization has a formal incentive program that provides incentives on long-term basis. 0.40 Level E Continuation of Level D, plus it provides a monetary bonus for employees based on their performances. 0.54 Continuation of Level E, plus the rewards are given on both a short and long-term basis, and they are Level F 0.67 recognized by the upper management of the organization E7. Social Activities Level A Social activities for employees are not applicable. 0.00 Level B Social activities for the employees are not addressed. 0.13 Level C The organization does not formally plan social activities for the employees, and there is only a yearly 0.27 organization wide social activity. The organization formally plans a social activity once or twice a year in which the project managers will Level D 0.40 attend, along with a yearly organization-wide social activity. Continuation of Level D, plus several times throughout the year which the project managers will attend, along Level E 0.54 with a yearly organization-wide social activity. Continuation of Level E, plus monthly which the project managers will attend and upper management Level F 0.67 including the president will attend on a quarterly basis, along with a yearly organization-wide social activity. E8. Stability of Organisational Structure Level A Maintaining the Stability of the organizational structure is not important. 0.00 Level B No plans to manage change of key people in a contract. 0.16 Level C The key professionals are named or defined in the contract. 0.33 Continuation of Level C, plus the contract states that they cannot be changed without notice and prior Level D 0.49 approval. Level E Continuation of Level D, plus there are pre-approved designated successors. 0.65 Continuation of Level E, plus the contract specifies all professionals, along with possible successors and right Level F 0.82 of approval by the other party E9. Clear Delegation of Responsibility Level A Clear delegation of responsibility is not applicable to this building project. 0.00 Level B There is a simple and centralized delegation of responsibility. 0.17 Level C There is a simple and formal delegation of responsibility which is customized for each project. 0.33 Level D Continuation of Level C, plus there is stable project environment. 0.50 Continuation of Level D, plus the delegation of responsibility vary based on the areas of expertises; for Level E 0.67 instance, preparation of separate documents for technical and administrative staff. Continuation of Level E, plus there is a formal delegation of authority that is clearly defined for all parties 0.84 Level F involved in the construction. The plan is reviewed periodically and evolves when necessary. E10. Retention Plan for Experienced Personnel 0.00 Level A Retention plan for experienced personnel is not applicable to this building project. 0.15 Level B A retention plan is not addressed. Level C Each project manager is responsible for retention of his workers. 0.31 Incentives such as employee training and others may be available, but it is not a requirement. Junior staffs Level D 0.46 have higher pay and preferred hiring status on the next project for the same employer. Continuation of Level D, plus employee training is required for junior staff. The employer makes available a Level E 0.61 list of opportunities for the next project. Continuation of Level E, plus testing and certification on the site lead to pay increases. Employer meets with Level F 0.76 individual worker and offers job at the new project site as required. E11. Exit Interviews Level A Exit interview is not applicable. 0.00 Level B Exit interview is not conducted. 0.12 Level C There is exit interview for key personnel only. 0.23 Level D Random exit interviews when there is time. 035 Level E Formal exit interview for all personnel. 0.46 There are formal exit interviews for all personnel: feedback is obtained from the employees: lessons are 0.57 Level F learnt; and the management is committed to take corrective actions.

Table 10-2 continued

	F. Safety and Health Practices Scoring Tools	
	F1. Housekeeping	Weights
Level A	Housekeeping is not applicable to this building project	0.00
Level B	Regular housekeeping has not been addressed on the project.	0.18
Level C	Housekeeping occurs only after accidents occur.	0.36
Level D	Housekeeping occurs on a bi-weekly basis.	0.53
Level E	Major travel paths are organized and clean. Housekeeping occurs weekly.	0.71
Level F	All work areas are well organized and designated crews are regularly cleaning the site.	0.89
	F2. Health and Safety Policy	
Level A	Health and safety policy is not applicable to this project.	0.00
Level B	The company does not have a formal health and safety policy.	0.19
Level C	There is a health and safety policy at company and project levels. The policy is documented and publicized.	0.38
Levere	It deals with the health and safety of workers.	0.56
Level D	Continuation of Level C, plus it is periodically updated based on the organizational and industry feedback.	0.57
	Continuations of Level D, plus the organization screens project participants based on their health and safety	
Level E	programs and chooses those with records of good performance	0.76
	Continuation of Level E. plus health and safety policy is integrated with procurement processes. Money is	
Level F	budgeted for the project to address various health and safety issues	0.95
	F3 Health and Safety Plan	0.75
Level A	Health and safety plans are not applicable to this building project	0.00
Level R	No health and safety targets have been examined and considered for the project.	0.00
Lever D	Some techniques to reduce accidents on the project are utilized. The project has a reactive approach	0.17
Level C	towards safety	0.37
Lavel D	Most but not all accident reduction techniques are utilized on the project	0.55
Level D	All potential agaident reduction techniques are utilized on the project.	0.33
LevelE	An potential accident reduction rechniques are fully utilized on the project.	0.74
Level F	zero accident techniques are funy utilized on the project. The project has a very proactive approach towards	0.92
	F4 Safe Work Method Statement	
Level A	Task Safety Analysis or Safe Work Method Statement (SWMS) is not applicable to this building project.	0.00
Level B	No Task Safety Analysis or SWMS is utilized.	0.18
1.10	Task Safety Analysis is utilized only in high-risk areas of the project, and SWMS is prepared after the	0.27
LevelC	occurrence of the accidents.	0.37
Level D	SWMS is prepared and utilized for most tasks in this project.	0.55
Level E	Safe Work Method Statement (SWMS) is utilized daily on the project.	0.73
. IF	SWMS's are utilized daily on the project for all tasks and the project team prepares additional SWMS's as	0.01
Level F	the task changes.	0.91
	F5. Hazards Analysis	
Level A	Hazard identification process is not applicable to this building project.	0.00
Level B	No hazard identification process is in place on this project.	0.19
Level C	Hazards are identified for a few tasks that are not covered in SWMS.	0.37
Level D	Hazards are identified for most tasks that are not covered in SWMS.	0.55
Level E	Hazards are identified for all tasks or for both high-risk and low-risk tasks.	0.74
L aval E	Hazards are identified for all tasks (high-risk and low-risk) and incorporated into the project's task specific	0.02
Levelr	safety planning process.	0.92
	F6. Health and Safety Training	
Level A	Health and Safety training is not applicable to this building project.	0.00
Level B	The project does not have a project specific training program for new employee.	0.18
	There is project specific induction regarding personal protective equipment, housekeeping and access to	
Level C	site, ladders and safe access to elevated platforms, fall protection, excavations and trenching, tools and	0.36
Levere	equipment, electrical hazards and fire prevention. Supervisors receive additional induction about safety	0.50
	meetings, first aid and medical treatment processes, and alcohol and drugs in the workplace.	
	Continuation of Level C, plus some employees should pass fitness for duty testing prior to starting	
Lavel D	construction of some tasks. The induction addresses management commitment, general project safety rules,	0.52
LeverD	emergency procedures, hazard communication, housekeeping, injury/illness reporting, confined spaces,	0.55
	compressed gas cylinders, back injury prevention, hand/power tool safety and others.	
Level E	Continuation of Level D, plus orientation addresses zero accidents philosophy.	0.71
Level F	Continuation of Level E, plus there is behavioural based training.	0.89
	F7. Toolbox Safety Meetings	
Level A	The project does not conduct safety meetings.	0.00
Level B	The project issues toolbox topics via handouts to employees on a periodic basis.	0.18
Level C	The project conducts a monthly meeting at or near breaks. Meetings reiterate job site safety rules.	0.36
Level D	The project conducts a weekly meeting at or near breaks. Meetings reiterate job site safety rules.	0.55
	The project conducts weekly meetings at a prearranged time, generally at the start of the day. Meetings	
Laurin	address current job status, hazards posed by upcoming project activities, and corrective actions taken. It	0.72
Level E	reviews the recorded injuries and near misses, and reiterates job site safety rules and expectations. Time is	0.73
	set aside during the meeting for interactive discussion which allows worker feedback.	
Level F	Continuation of Level E, plus the date of the meeting varies or the meeting can be conducted daily.	0.91

Multi-storey building projects with construction materials management practices scores less than 2.4 (baseline construction materials management practices score) are considered as low-score projects, and they could have an associated less productivity. On the other hand, projects having construction materials management practices scores greater than 2.4 are high-score projects, and the associated productivity could also be high. Projects with construction equipment management practices scores less than 0.93 (baseline construction materials equipment practices score) are considered as low-score projects, and their productivity could be low. Projects with a construction equipment management practices related to construction methods scores greater than 4.84 (baseline management practices related to construction methods score) can be considered as high-score projects, and the associated productivity could be high; and those with a score less than 4.84 are low-score projects, and their productivity could be low.

Projects having pre-construction phase management practice scores less than 4.52 (baseline pre-construction phase management practices score) are considered as low-score and less-productive projects. However, projects with a pre-construction phase management practices score exceeding 4.52 can be regarded as high-score projects and could have higher productivity. A human resource management practice score of 4.39 is used a baseline to divide projects into high-score and low-score. Accordingly, projects with HRM practices scores less than 4.39 are considered to have low productivity, and the implementation levels of the HRM practices should be increased to enhance productivity. On the other hand, multi-storey building projects with HRM scores exceeding 4.39 can be considered as high-score projects, and they could have higher productivity. Likewise, multi-storey building projects with safety and health practices scores less than 5.23 (baseline safety and health practices scores exceeding 5.23 are high-score projects, and their productivity could be high.

10.2.3 The Probability-Based Regression Models, the Sigmoid Graphs, and the Linear Regression Model

Seven logistic regression models that can be used for predicting the probability of exceeding baseline productivity factors are built and validated (Equation 10-1 to

Equation 10-7). The reliability tests of the probability-based regression models were conducted, and acceptable results were found. By using the seven models, seven sigmoid graphs have been developed to determine the corresponding probabilities when the scores of the construction management practices are known. The first logistic regression model (Equation 10-1) predicts the probability of exceeding productivity factor of 0.90 when the overall score of the construction management practices is known. The overall score is determined by using the scoring tools presented in Table 10-2. The sigmoid graph of the overall construction management practices is shown in Figure 10-1.

Logistic Regression Model of the Overall Construction Management Practices and Productivity

$$\mathbf{P}_{i} = \frac{e^{0.41 \text{CMP}_{i} \cdot 7.65}}{1 + e^{0.41 \text{CMP}_{i} \cdot 7.65}} \qquad \text{Equation 10-1}$$

 P_i is the probability of exceeding productivity factor (PF) = 0.90 for an ith project, CMP_i denotes the overall score of the construction management practices for an ith project, and e is the mathematical constant (approximately = 2.72).



Logistic Regression Model of the Construction Materials Management Practices and Productivity

The second logistic regression model is used to predict the probability of exceeding productivity factor of 1.05 when a score of construction materials management practices for a certain multi-storey building project is determined. The model is shown in Equation 10-2, and the associated sigmoid graph is indicated in Figure 10-2.

 $\mathbf{P}_{i} = \frac{e^{1.81MMP_{i}-6.46}}{1+e^{1.81MMP_{i}-6.46}}$ Equation 10-2

In Equation 10-2, P_i is the probability of exceeding productivity factor (PF) of 1.05 for an ith project, MMP_i is the score of the construction materials management practices for an ith project, and e denotes Euler's number which is approximately 2.72.



Figure 10-2 Probability Graph Plotted Using the Logistic Regression Model of the Construction Materials Management Practices and Productivity

Logistic Regression Model of the Construction Equipment Management Practices and Productivity

The third model predicts the probability of exceeding productivity factor of 0.90 given a score of construction equipment management practices. The model is presented in Equation 10-3, and the probability graph is shown in Figure 10-3.

$$\mathbf{P_{i}} = \frac{e^{2.29 \text{CEMP}_{i} - 0.98}}{1 + e^{2.29 \text{CEMP}_{i} - 0.98}} \qquad \text{Equation 10-3}$$

Where P_i =probability of exceeding PF= 0.90 for an ith project, CEMP_i = score of construction equipment management practice for an ith project, and e= mathematical constant.



Figure 10-3 Probability Graph Plotted Using the Logistic Regression Model of the Construction Equipment Management Practices and Productivity

Logistic Regression Model of the Management Practices Related to Construction Methods and Productivity

The fourth model is used to predict the probability of exceeding productivity factor of 0.90 when a score of management practices related to construction methods is determined. The model is presented in Equation 10-4, and the sigmoid graph plotted by using the model's equation is shown in Figure 10-4.

$$\mathbf{P_{i}} = \frac{e^{1.51MPRCM_{i}-5.79}}{1 + e^{1.51MPRCM_{i}-5.79}} \qquad \text{Equation 10-4}$$

In the equation, P_i is the probability of exceeding PF=0.90 for an ith project, MPRCM_i represents a score of management practices related to construction methods for an ith project, and e is Euler's number.



Figure 10-4 Probability Graph Plotted Using the Logistic Regression Model of the Management Practices Related to Construction Methods and Productivity

Logistic Regression Model of the Pre-Construction Phase Management Practices and Productivity

The fifth model (Equation 10-5) is developed to predict the probability of exceeding the productivity factor of 1.0 when a score of pre-construction phase management practices is known. The probability graph which is plotted by using the model is shown in Figure 10-5.

$$\mathbf{P_{i}} = \frac{e^{1.10PCPMP_{i}-5.26}}{1+e^{1.10PCPMP_{i}-5.26}} \qquad \text{Equation 10-5}$$

Where P_i = probability of exceeding PF=1.00 for an ith project, PCPMP_i = a score of pre-construction phase management practices of an ith project, and e = mathematical constant which is approximately 2.72.





Figure 10-5 Probability Graph Plotted Using the Logistic Regression Model of the Pre-Construction Phase Management Practices and Productivity

Logistic Regression Model of the Human Resource Management Practices and Productivity

The six model predicts the probability of exceeding productivity factor of 0.90 given a score of human resource management practices. The model is represented by Equation 10-6, and the associated sigmoid graph is presented in Figure 10-6.

 $\mathbf{P}_{i} = \frac{e^{1.5HRMP_{i}-5.68}}{1+e^{1.5HRMP_{i}-5.68}} \qquad \text{Equation 10-6}$ $P_{i} = \text{probability of exceeding PF=0.9 for an } i^{\text{th}} \text{ project, HRMP}_{i} = a \text{ score of HRM}$ practices for an i^{th} project, and e= mathematical constant.





320

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

Logistic Regression Model of the Safety and Health Practices and Productivity

The seventh model (Equation 10-7) is used to predict the probability of exceeding productivity factor of 0.90 when a score of safety and health practices is determined. The probability graph plotted using Equation 10-7 is shown in Figure 10-7.

$$\mathbf{P}_{i} = \frac{e^{1.81SHP_{i}8.79}}{1 + e^{1.81SHP_{i}-8.79}} \qquad \text{Equation 10-7}$$

In the equation, P_i is the probability of exceeding productivity factor of 0.9 for an ith project, SHP_i refers to a score of safety and health practices for an ith project, and e = Euler's number.



Figure 10-7 Probability Graph Plotted Using the Logistic Regression Model of the Safety and Health Practices and Productivity

Linear Regression Models of the Construction Management Practices and Productivity The linear regression model which is used to predict the productivity factor of a multistorey building project based on the overall score of the overall construction management practices is shown in Equation 10-8.

PF_i = 0.02CMP_i + 0.58 Equation 10-8

In the equation, PF_i refers to the productivity factor of an i^{th} project and CMP_i is the overall score of the construction management practices for an i^{th} project.

The linear regression models of the six categories of the construction management practices and productivity are presented in Equation 10-9 to Equation 10-14.

$PF_i = 0.09CMMP_i + 0.74$	Equation 10-9: Linear Re	Linear Regression Model of Construct			
	Materials	Management	Practices	and	
	Productiv	ity.			

- $PF_i = 0.13CEMP_i + 0.84$ Equation 10-10: Linear Regression Model of Construction Equipment Management Practices and Productivity.
- $PF_i = 0.08MPRCM_i + 0.61$ Equation 10-11: Linear Regression Model of Management Practices Related to Construction Methods and Productivity.
- $PF_i = 0.06PCPM_i + 0.71$ Equation 10-12: Linear Regression Model of Pre-Construction Phase Management Practices and Productivity.
- $PF_i = 0.07HRMP_i + 0.64$ Equation 10-13: Linear Regression Model of Human Resource Management Practices and Productivity.
- $PF_i = 0.09SHP_i + 0.49$ Equation 10-14: Linear Regression Model of Safety and Health Practices and Productivity.

Where PF_i = the productivity factor of an ith project, $CMMP_i$ = a score of the construction materials management practices for an ith project, $CEMP_i$ = a score of the construction equipment management practices for an ith project, MPRCM_i represents a score of the management practices related to construction methods for an ith project, PCPM_i is a score of the pre-construction phase management practices for an ith project, HRMP_i = a score of the human resource management practices for an ith project and SHP_i is a score of the safety and health practices for an ith project.

10.3 Implications of the Study

Improving the productivity of construction projects is essential for the economic growth of a country, for increasing the profit margins of contractors, and for reducing the project delays and the associated penalty such as liquidated damages. Over the last three decades, numerous researchers conducted studies which can help contractors to increase the productivity of their construction projects. Despite the presence of many studies on construction productivity, the construction industry is being blamed as the low productive industry as compared to other sectors. Thus, investigation of the techniques

which could enhance construction productivity is essential. Most problems identified by previous researchers both in Australia and international context are related to the management of the construction projects. However, the adverse effect of these problems can be minimised by adopting management practices that are suitable to enhance productivity in specific project types such as building, industrial and infrastructure construction projects in a particular location or country. Furthermore, the practices should be measured, and their adequacy needs to be assessed so that corrective actions can be taken by the project managers. Moreover, the management practices should be planned, controlled, and evaluated; and the associated productivity level should be estimated based on the planned practices.

The finding of this study confirmed that not all construction management practices that are identified for enhancing productivity in infrastructure and industrial projects in North America are suitable for multi-storey building projects in Australia. Moreover, the priorities of the practices are different. For instance, site tools management strategy, tools tracking systems and on-site tools maintenance are among the three practices that are common to infrastructure and industrial projects (CII, 2013a, CII, 2013b). However, it is found that the tools management systems are not suitable for multi-storey building projects. Similarly, the scoring tools developed for engineering projects are not appropriate to measure construction management practices in multi-storey building projects as the type and the weights of the practices vary based on the types of projects. Furthermore, previous researchers did not integrate the management practices with productivity. The users of the existing tools are unable to predict the productivity of their project based on the scores of their construction management practices. Thus, this research is conducted to identify the construction management practices which are suitable to improve productivity in multi-storey building projects in the context of Victoria State Australia, to refine and validate the management practices scoring tools for multi-storey building projects, and to develop and validate the productivity prediction tools based on the measured values of the management practices. This section addresses the practical as well as the theoretical contributions of this study.

10.3.1 Practical Implications

This study has practical implications for contractors in Australia and other countries. Contractors involved in the delivery of multi-storey building projects in Victoria State,

Australia can implement the identified construction management practices to improve productivity in their projects. Contractors in other countries can also implement the identified practices to enhance the productivity of their multi-storey building projects. However, as the practices could vary from country to country, validation is required to adapt to any local context. Contractors can focus on the implementation of the most important management practices identified in the study.

Contractors can use the scoring tools to assess the adequacy of their management practices. They can measure their practices using the scoring tools and compare with the baseline score developed in this research and take corrective actions. If the management practices score of a certain multi-storey building project is less than the baseline score, then the management practices of the particular project can be considered as inadequate, and the productivity of the project would be low. Thus, the project manager of a contractor should implement appropriate practices to potentially improve the productivity of the project.

The scoring tools can also be used as management practices' planning, monitoring, and evaluation tools during planning, construction and completion phases of a multi-storey building project respectively. During the project planning phase, the scoring tools and the logistic regression model or the linear regression model can be used to plan the management practices that could be implemented on a certain multi-storey building project. The logistic regression model can be used when the project manager wants to assess the expected productivity of a certain multi-storey building project with respect to a baseline productivity factor when a score of the management practices is known. Based on the predicted probability of exceeding the baseline productivity factor, the contractor's project manager can implement corrective actions to increase the chance of achieving a higher level of productivity as compared to the baseline. The linear regression model can be used when the project manager is interested in estimating the productivity factor of a certain multi-storey building construction project using the measured value of the construction management practices. Based on the predicted productivity factor, the project manager can make necessary decisions which could assist him/her to increase the productivity of a certain multi-storey building project. Similarly, by using the predicted productivity factor, the contractor's project manager can quantify the expected project delays in days or months.

During the construction phase, the scoring tools can be used to control the implementation of the management practices. The project manager of a certain multistorey building project can use the tools to monitor if the implemented practices comply with the planned practices and take corrective actions if there is a deviation. During the project closure phase, the scoring tools can be used to evaluate if the practices which are implemented on a certain multi-storey building project are associated with the estimated productivity factor of the project. If the implemented practices are not associated with the desired level of productivity, lessons can be learnt from the completed project, and appropriate practices will be planned for future similar projects.

Finally, a user-friendly tool is developed by using Visual Basic Applications (VBA) in Microsoft Excel 2016 version so that the management practices scoring tools can be used easily. The tool comprises of four sheets: (1) User's Guide, (2) Scoring Sheet, (3) Productivity Reports and (4) Planned Practices. The 'User's Guide' sheet explains the procedures to be followed by the users. Table 10-3 shows the screenshot of the User's Guide sheet.

	5 •∂-	 		y x	ĸ
File	Home	e Insert PageLayout Formulas Data Review View Developer 🖓 Tell me what you want to do		R, sh	are
E4	×				~
	А	B		с	-
1			_		
2		User's Guide			Н
з		Step 1: Provide project cost, project start date and planned project completion date on page 1 in the 'scoring sheet.'			Н
4		Step 2: Choose the levels of planning of the management practices on page 2 in the 'scoring sheet' using 'x' sign. The ver description of the levels of each practice is obtained by clicking the button in the column 'verbal description.'	rbal		
		Step 3: Click the 'Productivity Reports' sheet to get the actual duration, project delay, probability of exceeding the productivity			
5		factor of 0.90 and productivity outputs.			
6		Note: Productivity Factor = (actual productivity) / (planned productivity)			
7		= (project cost / actual duration) / (project cost / planned duration)			
		Step 4: Compare the actual duration with planned duration, decide if the project delay is not acceptable, and check if the estimate	ated		
8		probability is less than 50% and the productivity of the project is low.			Н
9		Step 5: Decide if the planned practices are adequate. If not, repeat Step 2 to Step 4 by choosing other levels of the practices.			
5		Note: the planned practices are displayed in the sheet 'planned practices,' and they change as the levels of planning in the 'Scor	ring		
10		Sheet' vary.	Ĩ		
		Step 6: After the desired level of productivity is obtained, use the practices indicated in 'planned practices' sheet	for		
11		implementation purpose.			
					v
•	-	User's Guide Scoring Sheet Productivity Reports Planned Practices 🕣		Þ	
Ready			3:18 AM	-+ 120	0%
	О Тур	pe here to search U 🖏 🔁 🔁 🔜 📰 🗱 🏘 🔨 🔨 🗛 🖓	G 7/07/201	7 🖓	

Table 10-3	The	User's	Guide	Sheet
------------	-----	--------	-------	-------

In Page 1 of the 'Scoring Sheet,' users are requested to provide project cost, project start date and planned project completion date. In page 2 of the 'Scoring Sheet,' users choose the level of planning of each management practice. Table 10-4 presents the two pages of the 'Scoring Sheet' which do not contain the project details. Users can also view the

detailed descriptions of each level of planning by clicking the command button under the column 'verbal descriptions.' For instance, the contractors' project managers can click the command button MMP 1 to read the message box which displays the description of the levels of the practice 'procurement plans for materials.'

□ 5 * ♂ * ∓ User Friendly Tool May 208_2017 - Excel					
File Home Insert Page Layout Formulas Data Review	v View Developer ${\mathbb Q}$ Tell me what you want to do				
T31 \mathbf{v} : $\mathbf{X} \mathbf{v} \mathbf{f}_{\mathbf{X}}$					
A B C D E	F G H I J K L M N O P				
2	Management Practices Rating Sheet				
3	Levels of Planning Verbal Description				
4	No. Management Practices A B C D E F				
5	1 Procurement Plans for Materials MMP 1				
	2 Long lead Materials Identification MMP 2				
	3 Materials Status Database MMP 3				
8	4 Materials Derivery Schedule MMP 4				
9	5 Material Inspection Process				
10	2 Decomposite Dispection Learn				
12 No Device Information	Producement Plan for construction Equipment ENP 1 ENP 1 ENP 1 ENP 1				
12 IV. Project Information	S Construction Equipment Productivity Analysis				
14 2) Devicest Start Data (dd/mm/m)	10 Let end d C. h. d.d. MPRCM 1				
14 2 Project Start Date (domini yy)	10 Integrated Schedule				
15 S Flaimed Project Completion Date (dd/min/yy)	11 work Schedule Strategies MPRCM3				
17	12 Schedule Control				
17	14 Traffic Control Plan MPRCM 5				
10	15 Site Security Plan MPRCM 6				
20	16 Machinery Positioning Strategy MPRCM 7				
21	17 Project Start-up Plan MPRCM 8				
22	18 Project Completion Plan MPRCM 9				
23	19 Innovations and New Technologies MPRCM 10				
24	20 Short Interval Plan PCPMP 1				
25	21 Well-Defined Scope of Work PCPMP 2				
	22 Use of Software in Planning Work Packages PCPMP 3				
27 Page 1	23 Dedicated Planner PCPMP 4				
28	24 Construction Work Packages PCPMP 5				
29	25 Buildability Review PCPMP 6				
30	26 Utilities Alignment PCPMP 7				
31	27 Contract Types PCPMP 8				
User's Guide Scoring Sheet Productivity Reports	Planned Practices +				
Ready 🛅	· · · · · · · · · · · · · · · · · · ·				
Type here to search	🙉 🦲 🔼 🗿 🔚 📲 🔊				

Table 10-4 The Scoring Sheet of the User-Friendly Tool

The project information and the management practices of an example multi-storey building project are provided in Table 10-5. After filling the forms on page 1 and page 2 of the 'Scoring Sheet,' the project managers of multi-storey building projects can open the sheet 'Productivity Reports' to get productivity related information which will help them to make necessary decision. The outputs in the 'Productivity Reports' sheet include the probability of exceeding the baseline productivity factor of 0.90, project delay, the expected productivity of the project and the actual completion date of the project which is predicted based on the levels of planning of the practices. Table 10-6 presents the outputs of the example project whose information is provided in the 'Scoring Sheet.'

CHAPETR 10

Summary and Conclusion

H	5-⊜-∓						User	Friendly Tool_	June 5 _2011	- Excel						Arga	w Gurmu	Ŧ	—	ð	\times
File	Home	Insert Page Layout	Formulas	Data	Review	View	Developer	♀ Tell me	e what you v	ant to do										Я, s	hare
R14	*	$\times \checkmark f_x$																			~
A	в	с) E	F	G	н		1113	KL	MIN	0	Р	Q	R	s	т	U	v	w	
2	1 1						Ма	nagement Pra	actices Ratin	g Sheet									1		
3					,		1/	Dennetises	Lev	els of Plan	ming	Verbal Descript	ion								
5					-	1 Procurem	nent Plans for Ma	aterials		x		MMP 1	_								
6						2 Long lead	1 Materials Ident	ification		x		MIMP 2	Microsoft Ex	vcel							
7						3 Materials	s Status Database			x		MIMP 3	WIICIOSOTE E	(CCI							
8						4 Materials	s Delivery Sched	ule		x	_	MIMIP 4									
9						5 Material	Inspection Proce	155 m		x	_	MMP 5	Level A = A	procure	ement plan f	or materia	als is not ap	pplicable;l	Level B = T	here is no)
11						7 Procurem	ent Plan for Con	" struction Equir	oment	x	-	EMP 1	documente	ed procu	rement plar	for mate	rials; Level	C = A pro	curement	plan and	
12	No.	Project Information				8 Construc	tion Equipment 1	Productivity An	nalysis	x		EMP 2	of Level C.	plus the	plan includ	ateriais ar es all mate	erials and c	onsumab	les. Also, t	here is an	
13	1 Project Cos	t (millions AUD)		30		9 Construc	tion Equipment 1	Maintenance		x		EMP 3	established	protoco	ol for identif	ing reput	ation of po	otential ve	ndors; Lev	el E =	
14	2 Project Star	t Date (dd/mm/yy)	1/01/	2018		10 Integrated	d Schedule			x		MPRCM 1	Continuatio	on of Lev	/el D, plus ti	ne plan ide	entifies neo	essary eq	uipment a	nd onsite	
15	3 Planned Pro	ject Completion Date (dd/m	m/yy) 1/01/	2019		11 Work Sch	hedule Strategies			x	_	MPRCM 2	resources to	o suppoi nt sched	rt delivery; L Iule is integr	evel F = C ated with	ontinuatio	n of Level	IE, plus the	e hat	
16					⊢	12 Schedule	Control Site Larrout Play			x	_	MPRCM 4	automatica	ally upda	tes the proc	urement s	schedule as	s the cons	truction so	hedule	
18						14 Traffic C	ontrol Plan	1		x	+	MPRCM 5	changes	· ·							
19						15 Site Secu	rity Plan			x	-	MPRCM 6									
20						16 Machiner	ry Positioning St	rategy		x		MPRCM 7							_		
21						17 Project S	tart-up Plan			x		MPRCM 8								OK	
22						18 Project C	Completion Plan	hantaning		x	_	MPRCM 9 MPRCM 10									
23						20 Short Int	erval Plan	nnologies		x	_	PCPMP 1	_								
25		-				21 Well-Def	ined Scope of W	ork		x	+	PCPMP 2									
26		Page 1				22 Use of Sc	oftware in Planni	ng Work Packa	zes 🖉	x		PCPMP 3									
27		I age I				23 Dedicated	d Planner		90	x		PCPMP 4									
28						24 Construc	tion Work Packa	ges	-	x	_	PCPMP 5 PCPMD 6	_								
30						26 Utilities	Alignment			x	-	PCPMP 7									
31						27 Contract	Types			x	+	PCPMP 8									
32						28 Model D	evelopment			x		PCPMP 9									
33						29 Regulator	ry Requirements			x		PCPMP 10									
34						30 Crew Co	mposition	tunting.		x	_	HRM1									
- -	Use	's Guide Scoring She	et Produ	uctivity Rep	oorts	Planned	Practices	(+)				THAN 2		4							
Ready	1							<u> </u>												+ 8	0%
	O Type he	re to search		Ū (]) (2, 6			x					-			~ 🗈	G. EN	G 5/06/2	M 017 -	

⊟ ⁵	o• c≥ - =		14. User Friendly Tool - Excel	Argaw Gurmu 🖪							
File	Home Insert	Page Layout Formulas Dat	a Review View Developer Q Tell me what you want to a	do	A. Share						
F15	• : ×	√ <i>f</i> e			v						
	А		В	С	D						
1											
2											
3			Outputs								
4		Probabilitiy of Exceeding Productivity Factor of 0.90 17.88%									
5		Expected Productivity of the Project Low									
6		Planned Manager	ment Practices	Inadequate							
7		Planned Duration	(cal. days)	365							
8		Actual Duration ((cal. days)	415							
9		Actual Completio	on Date (dd/mm/yy)	20-February-2019							
10		Project Delay (ca	l. days)	50							
11											
12											
	User's Guid	Scoring Sheet Productivi	ty Reports Planned Practices (+)	: 4							
Ready 🖁	ta			III III	+ 100%						

Table 10-6 Productivity Reports for the Example Project

The 'Planned Practices' sheet shows the management practices scores and the verbal descriptions of the levels of planning which the user has chosen in the 'Scoring Sheet.' All the verbal descriptions of the 47 management practices can be viewed by clicking the command button of each practice under the 'verbal description' column (Table 10-7).

H	• • ∂ •	÷							14. User Friendly Tool - Excel	Argaw Gurr	mu 🖽 — 🗇	ı ×		
File	Home	Insert	Page Layout	Formulas	Data	Review	View	Developer	$\ensuremath{\mathbb{Q}}$ Tell me what you want to do			∕⊊ Sha		
8	Ŧ	: ×	√ f _x											
	А	В			С			D		E	F			
		Sr. No.	Managem	ent Practio	ces			Score	Levels	of Planning	Verbal Description	ms		
1			1 Procureme	ent Plans fo	or Mater	rials		0.34	Level C = A procurement plan an materials and costly items.	d schedule exists only for large	MMP 1			
			2 Long lead	Materials I	dentific	ation		0.37	Level C = A procurement plan an materials.	d schedule exists for long-lead	MMP 2			
			3 Materials S	Status Data	base			0.26	Level C = There is a formal pape	Microsoft Excel		×		
			4 Materials I	Delivery Sc	hedule			0.30	Level C = Materials delivery is pla integrated with a project schedule	Level A = A procurement plan for materials is not appl documented procurement plan for materials. Level C	icable;Level B = There is no = A procurement plan and			
	5 Material Inspection Process							0.28	Level C =Materials inspection is a only.	schedule exists only for large materials and costly items; Level D = Continuation of Level C, plus the plan includes all materials and consumables. Also, there is an established protocol for identifying reputation of potential vendors; Level E =				
	6 Materials Inspection Team					0.23	Level C = There is a designated p inspection but no training and qua	Continuation of Level D, plus the plan identifies necessary equipment and onsite resources to support delivery, Level F = Continuation of Level E, plus the procurement schedule is integrated with a project information system that						
D			7 Procureme	ent Plan for	Const	ruction E	quipmen	t 0.28	Level C = Construction equipmen or leasing are prepared for this bu	automatically updates the procurement schedule as th changes	e construction schedule			
1			8 Constructi	ion Equipm	ient Pro	ductivity	Analysis	s 0.24	Level C =Construction equipmen but they are not tied to a project s budget only.		ОК	긕		
2			9 Constructi	ion Equipm	ient Ma	intenance	•	0.32	Level C =Construction equipmen maintenance is carried out by ope	t is logged in simplified spreadsheet and erator request.	EMP 3			
4	•	User's Gu	ide Scoring S	Sheet Pro	ductivity R	leports	Planned P	ractices	Level C =Developing a schedule	with no resources present and managing		Þ		
ady											· · · · · · · · · · · · · · · · · · ·	+ 120		
I	О Тур					(D) 🧕	9 2	Ŷ	📻 🛛 🖬 🚳	· · · · · · · · · · · · · · · · · · ·	(4) ENG 3:46 AM	, 🖵		

Table 10-7 The Management Practices Scores of the Example Project

10.3.2 Theoretical Implications

This research contributes to the body of knowledge in construction management by identifying the management practices which are suitable to enhance labour productivity in multi-storey building construction projects. Researchers in other countries can use the independent variables (the construction management practices) identified in this study to investigate the relationship between productivity of multi-storey building projects and the construction management practices in other contexts. Moreover, they can also analyse the relative importance of the identified management practices in improving the productivity of multi-storey building construction projects in other countries and compare their findings with this study's findings. It is also theoretically useful to understand which management practices that have the potential to enhance productivity in multi-storey building projects and engineering projects are similar and which are different and why. The findings should facilitate international comparisons as well as sectoral comparison and provide useful information to companies seeking construction works in Australia. After validation, researchers in other countries can use the scoring tools to measure the management practices and investigate the relationship between productivity and the practices. They can also develop productivity prediction models based on a score of the practices.

Besides the contribution to the body of knowledge in construction management, this research contributes to the growth accounting framework, which is based on neoclassical economic growth theory, by identifying independent variables or the construction management practices that can drive labour productivity in the context of multi-storey building construction projects. The neoclassical economic growth theory states that the two main sources of increasing the output per worker are the technological advancement and capital accumulation. The growth accounting framework suggested that the management practices could also be the source of growth in the labour productivity. However, the framework does not explain which management practices, in which sector, are suitable to enhance labour productivity. Thus, this research helps to increase the understanding of the construction management practices that have potential to improve productivity in the multi-storey building projects. The study identified 47 practices, prepared the measuring tools, and developed prediction models. It is found that the construction management practices can explain

about 48% of the variance in the productivity factor, and the relationship between productivity and the practices is statistically significant. The remaining variance could be explained by other factors such as advance in technology and government regulations among others.

10.4 Study Limitations and Future Research Directions

This research developed models which can be used to predict the probability of exceeding baseline productivity factor when a score of the management practices is known. However, it did not identify the costs associated with each level of implementation of the construction management practices. Thus, future researchers can investigate the costs of implementation of the construction management practices and develop a model which integrates cost and the practices. Consequently, the contractors' project managers can make decisions regarding the implementation of the management practices on certain multi-storey building projects by considering the associated costs besides the estimated productivity levels.

This study focuses on the development and validation of the prediction models at the multi-storey building construction project level. Thus, activity level productivity could not be estimated using the models. Hence, future researchers can collect activity level productivity data, measure the management practices using the scoring tools prepared in this research, and develop as well as validate the productivity prediction models at activity levels. For instance, a model which can be used to predict the productivity of the concrete activity based on a score of the construction management practices can be built.

Due to the anonymity of the information related to construction productivity, the data such as floor areas, the number of storeys, and the layout of the project sites could not be obtained. Consequently, the associations among the floors areas, the number of storeys of the building projects and the construction management practices were not analysed. Thus, future studies can investigate these relationships by collecting the corresponding data. Furthermore, since some of the management practices identified in this research might not be suitable to enhance productivity in building projects with less than three storeys, it is recommended that future researchers need to identify which of the management practices are appropriate and which are not.

The other research direction is regarding the use of the management practices identified in this study in other countries. Since the practices might vary from country to country, validation is required prior to the implementation of the management practices that have the potential to improve productivity in multi-storey building construction projects in other contexts. Thus, future researchers can investigate context-specific management practices by using the findings of this study as a starting point. Moreover, as the weights of the practices could vary based on the local regulations, working culture, climate and other factors, future researchers can refine and validate the scoring tools and test the tools' reliability by using similar methodology adopted in this study.

Since the relative importance and the type of the management practices which could enhance productivity in multi-storey building projects and engineering projects might be different, future researchers in Australia should identify and prioritise the practices that are appropriate for increasing productivity in infrastructure and industrial construction projects in the context of Australia. The finding of this study can be used as a starting point for future investigations. It can also be used for comparison purpose.

REFERENCES

- Abdul Kadir, M., Lee, W., Jaafar, M., Sapuan, S. and Ali, A. (2005),"Factors affecting construction labour productivity for Malaysian residential projects", *Structural Survey*, Vol. 23 No. 1, pp. 42-54.
- Abou-Assaleh, T., Cercone, N., Keselj, V. and Sweidan, R. (2004), "N-gram-based detection of new malicious code", *in Computer Software and Applications Conference, 2004. COMPSAC 2004. Proceedings of the 28th Annual International, 2004*, IEEE, pp. 41-42.
- Akintoye, A. (2000),"Analysis of factors influencing project cost estimating practice", Construction Management & Economics, Vol. 18 No. 1, pp. 77-89.
- Alby, V. (1994),"Productivity: Measurement and management", AACE International Transactions, Vol. 38 No. 4, pp. 1-7.
- Alinaitwe, H. M., Mwakali, J. A. and Hansson, B. (2007),"Factors Affecting the Productivity of Building Craftsmen Studies of Uganda", *Journal of Civil Engineering and Management*, Vol. 13 No. 3, pp. 169-176.
- Allmon, E., Haas, C. T., Borcherding, J. D. and Goodrum, P. M. (2000),"U.S. construction labor productivity trends, 1970-1998", *Journal of Construction Engineering and Management*, Vol. 126 No. 2, pp. 97-104.
- Amaratunga, D., Baldry, D., Sarshar, M. and Newton, R. (2002),"Quantitative and qualitative research in the built environment: application of "mixed" research approach", *Work study*, Vol. 51 No. 1, pp. 17-31.
- Arditi, D. (1985),"Construction productivity improvement", *Journal of construction engineering and management,* Vol. 111 No. 1, pp. 1-14.
- Arditi, D. and Mochtar, K. (1996),"Productivity improvement in the Indonesian construction industry", *Construction Management and Economics*, Vol. 14 No. 1, pp. 13-24.
- Attalla, M. M. (1997), Project Control Techniques: Reconstruction of Occupied Buildings, MSc Degree Thesis, University of Waterloo, Canada.
- Australian Bureau of Statistics (2015a), Australian System of National Accounts, Concepts, Sources and Methods, Cat. no. 5216.0, Canberra.
- Australian Bureau of Statistics (2015b), *Estimates of Industry Multifactor Productivity,* 2014-15, Cat. no.5260, Canberra.

- Australian Bureau of Statistics (2016a), *Australian National Accounts: National Income, Expenditure and Product,* Cat. no.5206.0, Canberra.
- Australian Bureau of Statistics (2016b), Construction Work Done, Australia, Preliminary, June 2016, Cat. no.8755.0, Canberra.
- Australian Bureau of Statistics (2016c), Counts of Australian Businesses, including Entries and Exits, Jun 2011 to Jun 2015, Cat. no.8165.0, Canberra.
- Australian Bureau of Statistics (2017), Producer Price Indexes, Australia, March 201, Cat. no.6427.0, Canberra
- Austroads (2014), Building and Construction Procurement Guide: Principles and Options, Sydney, Australia.
- BCA (2012), Pipeline or Pipe Dream? Securing Australia's Investment Future, Business Council of Australia (BCA), Melbourne, Australia.
- Bell, L. C. and Stukhart, G. (1987),"Costs and benefits of materials management systems", *Journal of Construction Engineering and Management*, Vol. 113 No. 2, pp. 222-234.
- Best, R. (2012),"International comparisons of cost and productivity in construction: a bad example", *Australasian Journal of Construction Economics and Building*, Vol. 12 No. 3, pp. 82.
- Building Code (Fitness for Work/Alcohol and Other Drugs in the Workplace) Amendment Instrument (2015), Canberra, Australia.

Building Regulations (2006), Canberra, Australia.

- Caldas, C. H., Kim, J.-Y., Haas, C. T., Goodrum, P. M. and Zhang, D. (2014),"Method to Assess the Level of Implementation of Productivity Practices on Industrial Projects", *Journal of Construction Engineering and Management*, Vol. 141 No. 1, pp. 04014061.
- CFMEU (2016a), "2016 Interim RDO calendar", available at: https://vic.cfmeu.org.au [accessed 9 May 2016].
- CFMEU (2016b), "OHS Reps Information", available at: https://vic.cfmeu.org.au/ohs/reps-info [accessed 5 October 2016].
- Chalker, M. and Loosemore, M. (2016),"Trust and productivity in Australian construction projects: A subcontractor perspective", *Engineering, Construction and Architectural Management*, Vol. 23 No. 2, pp. 192-210.

- Chan, P. W. and Kaka, A. (2007),"Productivity improvements: understand the workforce perceptions of productivity first", *Personnel Review*, Vol. 36 No. 4, pp. 564-584.
- Chancellor, W. (2015),"Drivers of Productivity: a Case Study of the Australian Construction Industry", *Construction Economics & Building*, Vol. 15 No. 3, pp. 85-97.
- Choi, C. and Harris, F. (1991), "A Model for Determining Optimum Crane Position", *in Proceedings of the institution of civil engineers, 1991*, pp. 627-634.
- CII (2013a), Best Productivity Practices Implementation Index for Industrial Projects, Construction industry institute (CII), Austin, Texas.
- CII (2013b), Best Productivity Practices Implementation Index for Infrastructure Projects, Construction industry institute (CII), Austin, Texas.
- City of Melbourne (2005), "Construction Management Plan Guideline", available at: http://www.melbourne.vic.gov.au/SiteCollectionDocuments/constructionmanagement-guidelines.pdf [accessed 19 September 2016].
- COAA (2013a), "Construction Work Packages Best Practice ", available at: http://www.coaa.ab.ca/Portals/ConstructionPerformance/documents/reportspapers/WFP-RPT%20CWP%20Best%20Practice%20Report.pdf [accessed 23 November 2016].
- COAA (2013b), "WorkFace Planning Rules", available at: http://www.coaa.ab.ca/Portals/ConstructionPerformance/documents/wfpprocess/WFP-PRC%20WFP%20Rules.pdf [accessed 23 November 2016].
- Cohen, J. (2013), "Statistical Power Analysis for the Behavioral Sciences", Elsevier Science, Saint Louis, Missouri.
- Crespo Márquez, A. and Sánchez Herguedas, A. (2004),"Learning about failure root causes through maintenance records analysis", *Journal of Quality in Maintenance Engineering*, Vol. 10 No. 4, pp. 254-262.
- Creswell, J. W. (2013), Research design: Qualitative, quantitative, and mixed methods approaches, Sage publications, California.
- Dai, J., Goodrum, P. M. and Maloney, W. F. (2009a),"Construction craft workers' perceptions of the factors affecting their productivity", *Journal of Construction Engineering and Management*, Vol. 135 No. 3, pp. 217-226.

- Dai, J., Goodrum, P. M., Maloney, W. F. and Srinivasan, C. (2009b),"Latent Structures of the Factors Affecting Construction Labor Productivity", *Journal of Construction Engineering & Management*, Vol. 135 No. 5, pp. 397-406.
- Department of Treasury and Finance (2015), "Construction Supplier Register", available at: http://www.dtf.vic.gov.au [accessed 30 September 2015].
- Department of Treasury and Finance (2016), available at: http://www.dtf.vic.gov.au/Victorias-Economy/Economic-policy-andguidelines/Indexation-of-fees-and-penalties [accessed 30 September 2016].
- Doloi, H. (2008),"Analysing the novated design and construct contract from the client's, design team's and contractor's perspectives", *Construction Management and Economics*, Vol. 26 No. 11, pp. 1181-1196.
- Doloi, H. (2008),"Application of AHP in improving construction productivity from a management perspective", *Construction Management and Economics*, Vol. 26 No. 8, pp. 839-852.
- Doloi, H. (2012),"Cost overruns and failure in project management: understanding the roles of key stakeholders in construction projects", *Journal of Construction Engineering and Management*, Vol. 139 No. 3, pp. 267-279.
- Doloi, H., Sawhney, A., Iyer, K. and Rentala, S. (2012),"Analysing factors affecting delays in Indian construction projects", *International Journal of Project Management*, Vol. 30 No. 4, pp. 479-489.
- Durdyev, S. and Mbachu, J. (2011),"On-site labour productivity of New Zealand construction industry: Key constraints and improvement measures", *Construction Economics and Building*, Vol. 11 No. 3, pp. 18-33.
- Elbeltagi, E., Hegazy, T. and Eldosouky, A. (2004),"Dynamic layout of construction temporary facilities considering safety", *Journal of Construction Engineering and Management*, Vol. 130 No. 4, pp. 534-541.
- El-Gohary, K. M. and Aziz, R. F. (2014),"Factors Influencing Construction Labor Productivity in Egypt", *Journal of Management in Engineering*, Vol. 30 No. 1, pp. 1-9.
- Enshassi, A., Mohamed, S. and Abushaban, S. (2009),"Factors affecting the performance of construction projects in the Gaza strip", *Journal of Civil Engineering and Management*, Vol. 15 No. 3, pp. 269-280.

Enshassi, A., Mohamed, S., Mustafa, Z. A. and Mayer, P. E. (2007),"Factors affecting labour productivity in building projects in the Gaza Strip", *Journal of Civil Engineering and Management*, Vol. 13 No. 4, pp. 245-254.

Environment Protection Act (2010), Victoria, Australia.

- EPA (2016), "Noise in commercial construction sites and large residential and mixeduse developments", available at: http://www.epa.vic.gov.au/business-andindustry/guidelines/noise-guidance/commercial-developments [accessed 9 May 2016].
- Fagbenle, O. I., Adeyemi, A. Y. and Adesanya, D. A. (2004),"The impact of non financial incentives on bricklayers' productivity in Nigeria", *Construction Management and Economics*, Vol. 22 No. 9, pp. 899-911.
- Fair Work Commission (2016), "Building and Construction General On-site Award 2010", available at: https://www.fwc.gov.au/documents/documents/modern_awards/pdf/ma000020.p df [accessed 27 September 2016].
- Fangel, M. (1984),"Planning project start-up", International Journal of Project Management, Vol. 2 No. 4, pp. 242-245.
- Fowler Jr, F. J. and Cosenza, C. (2009),"Design and evaluation of survey questions", *The SAGE handbook of applied social research methods*, SAGE Publications: US, 375-412.
- Ghoddousi, P. and Hosseini, M. R. (2012),"A survey of the factors affecting the productivity of construction projects in Iran", *Technological & Economic Development of Economy*, Vol. 18 No. 1, pp. 99-116.
- Goodrum, P. M., McLaren, M. A. and Durfee, A. (2006),"The application of active radio frequency identification technology for tool tracking on construction job sites", *Automation in Construction*, Vol. 15 No. 3, pp. 292-302.
- Gordon, C. M. (1994),"Choosing appropriate construction contracting method", *Journal* of Construction Engineering and Management, Vol. 120 No. 1, pp. 196-210.
- Graham, G. H. and Unruh, J. (1990),"The motivational impact of nonfinancial employee appreciation practices on medical technologists", *The Health Care Manager*, Vol. 8 No. 3, pp. 9-18.

- Grau, D., Caldas, C. H., Haas, C. T., Goodrum, P. M. and Gong, J. (2009),"Assessing the Impact of Materials Tracking Technologies on Construction Craft Productivity", *Automation in construction*, Vol. 18 No. 7, pp. 903-911.
- Hadavi, A. and Krizek, R. J. (1993),"Short-term goal setting for construction", *Journal* of construction engineering and management, Vol. 119 No. 3, pp. 622-630.
- Hampson, K. D. and Brandon, P. (2004), Construction 2020-A vision for Australia's property and construction industry, CRC Construction Innovation, Brisbane, Australia.
- Hamzeh, F. R., Ballard, G. and Tommelein, I. D. (2008), "Improving construction workflow-the connective role of lookahead planning", *in Proceedings for the 16th annual conference of the International Group for Lean Construction, 2008*, pp. 635-646.
- Hanley, J. A. and McNeil, B. J. (1982),"The meaning and use of the area under a receiver operating characteristic (ROC) curve", *Radiology*, Vol. 143 No. 1, pp. 29-36.
- Hanna, A. S., Chang, C.-K., Sullivan, K. T. and Lackney, J. A. (2008),"Impact of shift work on labor productivity for labor intensive contractor", *Journal of Construction Engineering and Management*, Vol. 134 No. 3, pp. 197-204.
- Hewage, K. N., Gannoruwa, A. and Ruwanpura, J. Y. (2011),"Current Status of Factors Leading to Team Performance of On-Site Construction Professionals in Alberta Building Construction Projects", *Canadian Journal of Civil Engineering*, Vol. 38 No. 6, pp. 679-689.
- HIA (2012), "Building Products: A Compliance Free Zone?", available at: http://tas.hia.com.au/documents/HIA%20Tas/2012/August%2020/Building%20
 Products%20A%20Compliance%20Free%20Zone.pdf [accessed 14 November 2016].
- Hinze, J. and Wilson, G. (2000),"Moving toward a zero injury objective", *Journal of Construction Engineering and Management*, Vol. 126 No. 5, pp. 399-403.
- Holt, G. (1997),"Construction Research Questionnaires and Attitude Measurement: Relative Index or Mean?", *Journal of Construction Procurement*, Vol. 3 No.2, pp. 88-96.
- Hong, E. N. C., Hao, L. Z., Kumar, R., Ramendran, C. and Kadiresan, V. (2012),"An effectiveness of human resource management practices on employee retention in 337

institute of higher learning: A regression analysis", *International journal of business research and management*, Vol. 3 No. 2, pp. 60-79.

- Huang, A. L., Chapman, R. E. and Butry, D. T. (2009), Metrics and tools for measuring construction productivity: Technical and empirical considerations, US Department of Commerce, National Institute of Standards and Technology.
- Hughes, R. and Thorpe, D. (2014),"A review of enabling factors in construction industry productivity in an Australian environment", *Construction Innovation: Information, Process, Management*, Vol. 14 No. 2, pp. 210-228.
- Industry Commission (1997), Assessing Australia's Productivity Performance, Canberra.
- Jamieson, S. (2004),"Likert scales: how to (ab) use them", *Medical education*, Vol. 38 No. 12, pp. 1217-1218.
- Jarkas, A. and Bitar, C. (2012),"Factors Affecting Construction Labor Productivity in Kuwait", *Journal of Construction Engineering and Management*, Vol. 138 No. 7, pp. 811-820.
- Jergeas, G. (2009), *Improving construction productivity on Alberta oil and gas capital projects*, Alberta Finance and Enterprise, Alberta.
- Kakar, P., Raziq, A. and Khan, F. (2015),"Impact of Human Resource Management Practices on Employee Retention: A Case of Banking Sector in Quetta Baluchistan", *Journal of Management Info*, Vol. 5 No. 1, pp. 97-119.
- Kaming, P. F., Olomolaiye, P. O., Holt, G. D. and Harris, F. C. (1997), "Factors influencing craftsmen's productivity in Indonesia", *International journal of project management*, Vol. 15 No. 1, pp. 21-30.
- Kang, Y., O'Brien, W. J., Thomas, S. and Chapman, R. E. (2008),"Impact of information technologies on performance: cross study comparison", *Journal of Construction Engineering and Management*, Vol. 134 No.11, pp. 852-863.
- Kazaz, A. and Ulubeyli, S. (2007),"Drivers of productivity among construction workers: A study in a developing country", *Building and Environment*, Vol. 42 No. 5, pp. 2132-2140.
- Kerzner, H. R. (2010), *Project Management-Best Practices: Achieving Global Excellence*, John Wiley and Sons, New Jersey.

- Kim, H.-Y. (2013),"Statistical notes for clinical researchers: assessing normal distribution (2) using skewness and kurtosis", *Restorative dentistry & endodontics*, Vol. 38 No. 1, pp. 52-54.
- Krishnaswami, O. R. and Satyaprasad, B. G. (2010), *Business Research Methods,* Himalaya Publishing House, Mumbai.
- Lam, P. T., Wong, F. K. and Wong, F. W. (2007),"Building features and site-specific factors affecting buildability in Hong Kong", *Journal of Engineering, Design* and Technology, Vol. 5 No. 2, pp. 129-147.
- Lavender, S. (2014), Management for the construction industry, Routledge, New York.
- Li, Y. and Liu, C. (2010),"Malmquist indices of total factor productivity changes in the Australian construction industry", *Construction Management and Economics*, Vol. 28 No. 9, pp. 933-945.
- Liao, P.-C. (2008), *Influence factors of engineering productivity and their impact on project performance*, Doctoral Dissertation, The University of Texas at Austin.
- Liberda, M., Ruwanpura, J. and Jergeas, G. (2003), "Construction productivity improvement: A study of human, management and external issues", *in Proceedings of the Construction Research Congress, 2003*, ASCE pp. 1-8.
- Lim, E. C. and Alum, J. (1995),"Construction productivity: Issues encountered by contractors in Singapore", *International Journal of Project Management*, Vol. 13 No. 1, pp. 51-58.
- Loosemore, M. (2014),"Improving construction productivity: a subcontractor's perspective", *Engineering, Construction and Architectural Management*, Vol. 21 No. 3, pp. 245-260.
- Loosemore, M. and McGeorge, D. (2002), "Workplace Regulation, Reform and Productivity in the International Building and Construction Industry", available at: www. royalcombci. gov. au/docs/Complete% 20Discussion% 20Papper% 2015.
- Loosemore, M., Dainty, A. and Lingard, H. (2003), *Human resource management in construction projects: strategic and operational approaches*, Taylor and Francis, London.
- Love, P. E. (2001), *Determinants of rework in Australian construction projects*, Doctoral Dissertation, Monash University.

- Lozano, L. M., García-Cueto, E. and Muñiz, J. (2008), "Effect of the number of response categories on the reliability and validity of rating scales", *Methodology*, Vol. 4 No. 2, pp. 73-79.
- Lurd, A. and Lurd, M. (2013), "Measures of Central Tendency", available at: https://statistics.laerd.com [accessed 13 June 2016].
- Luthans, K. (2000), "Recognition: a powerful, but often overlooked, leadership tool to improve employee performance", *Journal of Leadership & Organizational Studies*, Vol. 7 No. 1, pp. 31-39.
- Makulsawatudom, A., Emsley, M. and Sinthawanarong, K. (2004),"Critical factors influencing construction productivity in Thailand", *The journal of KMITNB*, Vol. 14 No. 3, pp. 1-6.
- Maloney, W. F. (1983),"Productivity improvement: The influence of labor", *Journal of construction Engineering and Management*, Vol. 109 No. 3, pp. 321-334.
- Manavazhi, M. R. and Adhikari, D. K. (2002),"Material and equipment procurement delays in highway projects in Nepal", *International Journal of Project Management*, Vol. 20 No. 8, pp. 627-632.
- McDonald, D. F. and Zack, J. G. (2004), *Estimating lost labor productivity in construction claims*, AACE International Inc., Virginia.
- Mertens, D. M. (2010),"Transformative mixed methods research", *Qualitative inquiry*, Vol. 16 No. 6, pp. 469-474.
- Miles, M. B. and Huberman, A. M. (1994), *Qualitative data analysis: an expanded sourcebook*, SAGE Publications, California.
- Morse, C. L. (1990), Construction Tool Management, MSc. Thesis, Purdue University.
- Moselhi, O., Assem, I. and El-Rayes, K. (2005),"Change orders impact on labor productivity", *Journal of Construction Engineering and Management*, Vol. 131 No. 3, pp. 354-359.
- Motwani, J., Kumar, A. and Novakoski, M. (1995),"Measuring construction productivity: a practical approach", *Work study*, Vol. 44 No. 8, pp. 18-20.
- Naoum, S. G. (2015), "Productivity in Construction Projects", *In:* ROBINSON, H. W.
 (ed.) *Design economics for the built environment: impact of sustainability on project evaluation.* Wiley: Chichester, West Sussex, UK.

- Naoum, S. G. (2016),"Factors influencing labor productivity on construction sites: A state-of-the-art literature review and a survey", *International Journal of Productivity and Performance Management*, Vol. 65 No. 3, pp. 401-421.
- Narang, U. (2013),"HRM Practices–its impact on Employee retention", International journal of multidisciplinary research in social and management sciences, Vol. 1 No. 4, pp. 48-51.
- Nasir, H. (2013), Best Productivity Practices Implementation Index (BPPII) for Infrastructure Projects, Doctoral Dissertation, University of Waterloo.
- Nasir, H., Haas, C. T., Caldas, C. H. and Goodrum, P. M. (2015),"An Integrated Productivity-Practices Implementation Index for Planning the Execution of Infrastructure Projects", *Journal of Infrastructure Systems*, Vol. 22 No. 2, pp. 04015022.
- Nasir, H., Haas, C. T., Young, D. A., Razavi, S. N., Caldas, C. and Goodrum, P. (2010),"An implementation model for automated construction materials tracking and locating", *Canadian Journal of Civil Engineering*, Vol. 37 No. 4, pp. 588-599.
- Neil, J. and Knudsen, M. (1990), *Project Control for Construction*, Construction Industry Institute (CII), Austin, Texas.
- Nitithamyong, P. and Tan, Z. (2007),"Determinants for effective performance of external project management consultants in Malaysia", *Engineering, Construction and Architectural Management*, Vol. 14 No. 5, pp. 463-478.
- O'Connor, J. T. and Yang, L.-R. (2004),"Project performance versus use of technologies at project and phase levels", *Journal of construction engineering and management*, Vol. 130 No. 3, pp. 322-329.
- Occupational Health and Safety Regulations (2007), Melbourne, Australia.
- Olomolaiye, P., Wahab, K. and Price, A. (1987),"Problems influencing craftsmen's productivity in Nigeria", *Building and Environment*, Vol. 22 No. 4, pp. 317-323.
- Organisation for Economic Co-operation and Development (2001), Measuring Productivity: Measurement of Aggregate and Industry Level Productivity Growth, Paris.
- Peduzzi, P., Concato, J., Kemper, E., Holford, T. R. and Feinstein, A. R. (1996),"A simulation study of the number of events per variable in logistic regression analysis", *Journal of clinical epidemiology*, Vol. 49 No. 12, pp. 1373-1379.
- Pereira, D. G., Afonso, A. and Medeiros, F. M. (2015),"Overview of Friedman's Test and Post-hoc Analysis", *Communications in Statistics-Simulation and Computation*, Vol. 44 No. 10, pp. 2636-2653.
- Pheng Low, S. (2001),"Quantifying the relationships between buildability, structural quality and productivity in construction", *Structural Survey*, Vol. 19 No. 2, pp. 106-112.
- PMBoK (2000), *Guide to the project Management body of knowledge*, Project Management Institute, Pennsylvania, USA.
- PMI (2013), A guide to the project management body of knowledge (PMBOK® guide),
 Project Management Institute (PMI), Pennsylvania, USA.
- Poh, P. S. and Chen, J. (1998),"The Singapore buildable design appraisal system: a preliminary review of the relationship between buildability, site productivity and cost", *Construction Management & Economics*, Vol. 16 No. 6, pp. 681-692.
- Prasertrungruang, T. and Hadikusumo, B. (2007),"Heavy equipment management practices and problems in Thai highway contractors", *Engineering, Construction* and Architectural Management, Vol. 14 No. 3, pp. 228-241.
- PrefabAUS (2014), "Prefabricating Australia: Growing Our Off-site Construction Sector", *PrefabAUS 2014 Inaugural Conference* Melbourne, Australia.
- Productivity Commission (2013), Productivity Update, May.
- Productivity Commission (2015), Productivity Update, July.
- Productivity Commission (2016), Productivity Update, April.
- PWC (2013), Productivity Scorecard; Reconstructing Productivity in the Construction Industry, Price Waterhouse Coopers (PWC), Canberra.
- Rivas, R., Borcherding, J., González, V. and Alarcón, L. (2011),"Analysis of Factors Influencing Productivity Using Craftsmen Questionnaires: Case Study in a Chilean Construction Company", *Journal of Construction Engineering and Management*, Vol. 137 No. 4, pp. 312-320.
- Road Safety (Traffic Management) Regulations (2009), Melbourne, Australia.
- Rojas, E. M. and Aramvareekul, P. (2003),"Labor productivity drivers and opportunities in the construction industry", *Journal of Management in Engineering*, Vol. 19 No. 2, pp. 78-82.
- Rose, S., Spinks, N. and Canhoto, A. I. (2014), *Management Research. [electronic resource]: Applying the Principles,* Taylor and Francis, New Jersey.

- Rose, S., Spinks, N. and Canhoto, A. I. (2015), *Management research: Applying the principles*, Routledge, New York.
- Safe Work Australia (2011), "How to Manage Work Health and Safety Risks: Code of Practice ", available at: http://www.safeworkaustralia.gov.au/sites/SWA/about/Publications/Documents/

633/How_to_Manage_WHS_Risks.pdf [accessed 14 November 2016].

Safe Work Australia (2013), Construction Work Code of Practice, Canberra, Australia.

Safe Work Australia (2014a), "Construction Information Sheets ", available at: http://www.safeworkaustralia.gov.au/sites/swa/about/publications/pages/constru ction-information-sheets [accessed 3 October 2016].

Safe Work Australia (2014b), "Safe Work Method Statement for High Risk Construction Work", available at: http://www.safeworkaustralia.gov.au/sites/SWA/about/Publications/Documents/ 898/information-sheet-safe-work-method-statement.pdf [accessed 14 November 2016].

- Safe Work Australia (2014c), "Traffic Management: Guide for Construction Work", available at: http://www.safeworkaustralia.gov.au [accessed 2 May 2016].
- Santos, J. R. A. (1999), "Cronbach's alpha: A tool for assessing the reliability of scales", *Journal of extension*, Vol. 37 No. 2, pp. 1-5.
- Sawacha, E., Naoum, S. and Fong, D. (1999),"Factors affecting safety performance on construction sites", *International journal of project management*, Vol. 17 No. 5, pp. 309-315.
- Solow, R. M. (1956),"A contribution to the theory of economic growth", *The quarterly journal of economics*, Vol. 70 No. 1, pp. 65-94.
- Solow, R. M. (1957),"Technical change and the aggregate production function", *The review of Economics and Statistics*, Vol. 39 No. 3, pp. 312-320.
- Standards Australia (1990), *Workplace Injury and Disease Recording Standard*, National Occupational Health and Safety Commission, Sydney.
- Standards Australia (2000), General Conditions of Contract AS 2124-1992, NSW, Australia.
- Stewart, L. (2002a),"Leasing liberates cash to power growth", *Construction Equipment*, Vol. 105 No. 12, pp. 19-20.

- Stewart, L. (2002b),"Why rent? For low-use machines", *Construction Equipment*, Vol. 105 No. 7, pp. 50.
- Steyerberg, E. W., Harrell, F. E., Borsboom, G. J., Eijkemans, M., Vergouwe, Y. and Habbema, J. D. F. (2001),"Internal validation of predictive models: efficiency of some procedures for logistic regression analysis", *Journal of clinical epidemiology*, Vol. 54 No. 8, pp. 774-781.
- Tabassi, A. A. and Bakar, A. A. (2009),"Training, motivation, and performance: The case of human resource management in construction projects in Mashhad, Iran", *International Journal of Project Management*, Vol. 27 No. 5, pp. 471-480.
- Tabassi, A. A., Ramli, M. and Bakar, A. H. A. (2012),"Effects of training and motivation practices on teamwork improvement and task efficiency: The case of construction firms", *International Journal of Project Management*, Vol. 30 No. 2, pp. 213-224.
- Talisse, R. B. and Aikin, S. F. (2008), *Pragmatism. [electronic resource] : a guide for the perplexed,* Continuum, New York.
- Thomas, A. V. and Sudhakumar, J. (2014), "Modelling masonry labour productivity using multiple regression", in Raiden, A. B.and Aboagye-Nimo, E(eds.) Procs 30th Annual ARCOM Conference 2014 Portsmouth, UK, Association of Researchers in Construction Management, pp. 1346-1354.
- Thomas, H. R. and Sanvido, V. E. (2000),"Role of the fabricator in labor productivity", *Journal of construction engineering and management*, Vol. 126 No. 5, pp. 358-365.
- Thomas, H. R., Sanvido, V. E. and Sanders, S. R. (1989),"Impact of material management on productivity-A case study", *Journal of Construction Engineering and Management*, Vol. 115 No. 3, pp. 370-384.
- Thomas, S. R., Macken, C. L. and Lee, S.-H. (2001), Impacts of design/information technology on building and industrial projects, U.S. Department of Commerce, Gaithersburg, Maryland.
- Tommelein, I. and Zouein, P. (1993),"Interactive dynamic layout planning", *Journal of Construction Engineering and Management*, Vol. 119 No. 2, pp. 266-287.
- Tommelein, I., Levitt, R. and Hayes Roth, B. (1992),"Site Layout Modeling: How Can Artificial Intelligence Help?", *Journal of Construction Engineering and Management*, Vol. 118 No. 3, pp. 594-611.

- Toner, P., Green, R., Croce, N. and Mills, B. (2001),"No case to answer: productivity performance of the Australian construction industry", *The Economic and Labour Relations Review*, Vol. 12 No. 1, pp. 104-125.
- Tucker, R. L. (1986),"Management of Construction Productivity", *Journal of Management in Engineering*, Vol. 2 No. 3, pp. 148-156.
- Turbit, N. (2005), "Defining the Scope of a Project ", available at: http://www.projectperfect.com.au/downloads/Info/info_define_the_scope.pdf [accessed 23 November 2016].
- VBA (2016), "Occupancy permits", available at: http://www.vba.vic.gov.au/consumers/permits/occupancy-permits [accessed 11 November 2016].
- Vittinghoff, E. and McCulloch, C. E. (2007),"Relaxing the rule of ten events per variable in logistic and Cox regression", *American journal of epidemiology*, Vol. 165 No. 6, pp. 710-718.
- Walker, D. H. (1995),"An investigation into construction time performance", Construction Management and Economics, Vol. 13 No. 3, pp. 263-274.
- Waly, A., Thabet, W. and Wakefield, R. (1999), "An automated model for generating a short-interval schedule", *in Proceedings of 8DBMC Conference and CIB W78 Workshop*, 1999, pp. 2386-2392.
- Wang, Y., Goodrum, P. M., Haas, C., Glover, R. and Vazari, S. (2010),"Analysis of the benefits and costs of construction craft training in the United States based on expert perceptions and industry data", *Construction Management and Economics*, Vol. 28 No. 12, pp. 1269-1285.
- Wheeldon, D. (2012), "Why building materials imports are on the rise in Australia?", available at: http://www.architectureanddesign.com.au/news/why-buildingmaterials-imports-are-on-the-rise-in#comments [accessed 15 September 2016].
- Willems, G. and Van Aelst, S. (2005),"Fast and robust bootstrap for LTS", *Computational statistics & data analysis,* Vol. 48 No. 4, pp. 703-715.
- Wireman, T. (2005), *Developing performance indicators for managing maintenance,* Industrial Press Inc., New York.
- Work Health and Safety Regulations (2011), Canberra, Australia.
- Worksafe Tasmania (2016), "Work Health and Safety Policy Samples", available at: http://www.worksafe.tas.gov.au/ [accessed 28 November 2016].

345

- Worksafe Victoria (2006), *A handbook for workplaces: Forklift safety reducing the risk*, Melbourne, Australia.
- Worksafe Victoria (2008a), "Provisional Improvement Notices", available at: https://www.worksafe.vic.gov.au/__data/assets/pdf_file/0016/10438/FOR607web-version.pdf [accessed 10 November 2016].
- Worksafe Victoria (2008b), A handbook for the construction regulations: Working safely in the general construction industry, Melbourne, Australia.
- Worksafe Victoria (2016), "Induction Training ", available at: http://www.worksafe.vic.gov.au/safety-and-prevention/yourindustry/construction/how-to-comply/induction-training [accessed 29 November 2016].
- Yan, X. and Su, X. (2009), *Linear regression analysis. [electronic resource]: theory and computing*, World Scientific, Singapore.
- Yi, W. and Chan, A. P. C. (2014),"Critical review of labor productivity research in construction journals", *Journal of Management in Engineering*, Vol. 30 No. 2, pp. 214-225.
- Young, A., Wilkie, J., Ewing, R. and Rahman, J. (2008),"International comparison of industry productivity", *Economic Round-up*, Vol. No. 3, pp. 45-61.
- Zakeri, M., Olomolaiye, P. O., Holt, G. D. and Harris, F. C. (1996),"A survey of constraints on Iranian construction operatives' productivity", *Construction Management & Economics*, Vol. 14 No. 5, pp. 417-426.
- Zhang, P., Harris, F. C., Olomolaiye, P. and Holt, G. D. (1999),"Location optimization for a group of tower cranes", *Journal of construction engineering and management*, Vol. 125 No. 2, pp. 115-122.

APPENDIX 1: QUESTIONNAIRE

PART -I

A. Relative Importance of the Construction Management Practices That Could Enhance **Productivity in Multi-Storey Building Projects**

A1: Please rate the following construction materials management practices based on their degree of importance in improving productivity in multi-storey building construction projects on the scale of 1 to 5.

		Level of Importance				
No.	Construction Materials Management Practices	Not important (1)	Slightly important (2)	Somewhat important (3)	Very important (4)	Extremely important (5)
1	Procurement Plans for Materials					
2	Long-Lead Materials Identification					
3	Materials Status Database					
4	Materials Delivery Schedule					
5	Material Inspection Process					
6	Materials Inspection Team					

A2: Please rate the following construction equipment management practices based on their degree of importance in improving productivity in multi-storey building construction projects on the scale of 1 to 5.

		Level of Importance				
		Not	Slightly	Somewhat	Very	Extremely
No.	Construction Equipment Management	important	important	important	important	important
	Practices	(1)	(2)	(3)	(4)	(5)
1	Procurement Plan for Construction					
	Equipment					
2	Construction Equipment Productivity					
	Analysis					
3	Construction Equipment Maintenance					

A3: Please rate the following management practices related to construction methods based on their degree of importance in improving productivity in multi-storey building construction projects on the scale of 1 to 5

			Lev	el of Impor	tance	
No.	Management Practices Related to Construction Methods	Not important (1)	Slightly important (2)	Somewhat important (3)	Very important (4)	Extremely important (5)
1	Integrated Schedule					
2	Work Schedule Strategies					
3	Schedule Control					
4	Dynamic Site Lay Out Plan					
5	Traffic Control Plan					
6	Site Security Plan					
7	Machinery Positioning Strategy					
8	Project Start-Up Plan					
9	Project Completion Plan					
10	Innovations and New Technologies					

A4: Please rate the following pre-construction phase management practices based on their degree of importance in improving productivity in multi-storey building construction projects on the scale of 1 to 5.

		Level of Importance				
No.	Pre-construction Phase Management Practices	Not important (1)	Slightly important (2)	Somewhat important (3)	Very important (4)	Extremely important (5)
1	Well-defined Scope of Work					
2	Short Interval Plan					
3	Buildability Review					
						3,

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

4	Construction Work Packages			
5	Regulatory requirement			
6	Contract Types			
7	Use of Software in planning Work Packages			
8	Utilities Alignment			
9	Dedicated Planner			
10	Model Development			

A5: Please rate the following human resource management practices based on their degree of importance in improving productivity in multi-storey building construction projects on the scale of 1 to 5.

			Lev	el of Impor	tance	
		Not	Slightly	Somewhat	Very	Extremely
No.	Human Resource Management Practices	important	important	important	important	important
		(1)	(2)	(3)	(4)	(5)
1	Crew Composition					
2	Skill Assessment and Evaluation					
3	3 Employees Training					
4	Career Development					
5	Non-Financial Incentive Programs					
6	Financial Incentive Programs					
7	Social Activities					
8	Stability of Organizational Structure					
9	Clear Delegation of Responsibility					
10	Retention Plan for Experienced Personnel					
11	Exit Interviews					

A6: Please rate the following safety and health practices based on their degree of importance in improving productivity in multi-storey building construction projects on the scale of 1 to 5.

		Level of Importance				
		Not	Slightly	Somewhat	Very	Extremely
No.	Safety and Health Practices	important (1)	important (2)	important (3)	important (4)	important (5)
1	Housekeeping					
2	Safety and Health Policy					
3	Safety and Health Plan					
4	Safe Work Method Statement					
5	Hazards Analysis					
6	Safety and Health Training					
7	Toolbox Safety Meetings					

PART-II

B. Project Characteristics

Project type (residential, commercial etc.)	
Project value	
Floor area	
Project start date:	
Planned completion date:	
Actual completion date:	

C. Validated Survey Instruments (adapted from CII (2013a) and CII (2013b))

C1: Validated Survey Instrument for Construction Materials Management Practices

Please check ($\sqrt{}$) one box per element in space provided in front of the level of implementation of the management practice. Please do not leave any elements blank.

Level A A procurement plan for materials and equipment is not applicable	
Dever M procurement plan for materials and equipment is not appreciate	
Level B There is no documented procurement plan for materials.	

348

	Level C	A procurement plan and schedule exists only for large materials and costly items.
	Level D	Continuation of Level C, plus plan includes all materials and consumables. Also, there is an established protocol for identifying reputation of potential vendors.
	Level E	Continuation of Level D, plus plan identifies necessary equipment and onsite resources to support delivery.
	Level F	Continuation of Level E, plus the procurement schedule is integrated with a project information system that automatically updates the procurement schedule as the construction schedule
(√)		2. Long lead Materials Identification
	Level A	A separate procurement plan for long-lead materials is not applicable.
	Level B	There is no documented procurement plan for long-lead materials.
	Level C	A procurement plan and schedule exists for long-lead materials.
	Level D	Continuation of Level C, plus there is an established protocol for identifying reputation of potential vendors.
	Level E	Continuation of Level D, plus plan identifies necessary equipment and onsite resources to support delivery.
	Level F	Continuation of Level E, plus the procurement schedule is integrated with a project information system that automatically updates the procurement schedule as the construction schedule changes.
(√)		3. Materials Status Database
	Level A	Material status database is not applicable.
	Level B	There is no formal paper based system used to track materials status.
	Level C	There is a formal paper based system to track materials status.
	Level D	A proprietary internal materials status software tool is used, but it is not integrated with your company's project control systems or used by other contractors.
	Level E	An available software tool is used but it is only integrated internally with your company's project control systems.
	Level F	An available software tool is used by all stakeholders that are integrated with your supply chain and other project control systems.
(√)		4. Materials Delivery Schedule
	Level A	Materials delivery schedule is not applicable.
	Level B	There is no documented materials delivery schedule.
	Level C	Materials delivery is planned early in the project and is integrated with a project schedule.
	Level D	Continuation of Level C plus the schedule is automatically updated on receipt of new information as procurement proceeds.
	Level E	Continuation of Level D plus the schedule is automatically linked with procurement, materials management, and overall project scheduling systems.
	Level F	Continuation of Level E plus materials delivery planning and management is completely integrated with other automated project processes including automated materials tracking throughout the supply chain.
(√)		5. Material Inspection Process
	Level A	A material inspection process is not applicable.
	Level B	There is no materials inspection process.
	Level C	A materials inspection process is only utilized for large items or costly items on a project.
	Level D	A materials inspection process is utilized that includes all items delivered to the site. There is a lack of organization of the process and materials are not separated into stages of the receipt process nor does it record the location of the materials and mark the materials for tracking
	Level E	A materials inspection process is used at the supplier and onsite, and organizes materials receipt inspections immediately upon delivery of materials, verifies that materials conform to standards, and organizes materials for tracking.
	Level F	Continuation of Level E, plus the process includes separation of material into categorical stages of the receipt process (e.g. awaiting inspection storage area restocking scrap, and/or awaiting for
		shipment, verification if the materials conform to specifications, standards, drawings, etc., record of the location of materials and marked materials for tracking, and prioritization for quality).
(√)		shipment, verification if the materials conform to specifications, standards, drawings, etc., record of the location of materials and marked materials for tracking, and prioritization for quality). 6. Materials Inspection Team
(√)	Level A	shipment, verification if the materials conform to specifications, standards, drawings, etc., record of the location of materials and marked materials for tracking, and prioritization for quality). 6. Materials Inspection Team Materials inspection team is not applicable.

Level C	There is a designated materials inspection team but no training and qualifications of the
	individual's skill level is specified.
Level D	Continuations of Level C, plus inspections are performed by project managers or workers rather
	than the team.
Level E	Continuation of Level D, plus the inspection team can adequately inspect materials and
	understand the material specifications.
Level F	Continuation of Level E, plus the members of the inspection team are experts at inspection
	processes and procedures, and knows how to inspect materials and understands the material
	specifications.

C2: Validated Survey Instrument for Construction Equipment Management Practices

Please check ($\sqrt{}$) one box per element in space provided in front of the level of implementation of the management practice. Please do not leave any elements blank.

(√		1. Procurement Plan for Construction Equipment
	Level A	A procurement plan for construction equipment is not applicable for this building project.
	Level B	Construction equipment procurement is not prepared for this building project.
	Level C	Construction equipment procurement strategies such as hiring or leasing are prepared for this building project.
	Level D	Continuation of Level C and there is a procedure for identifying reputation of potential equipment suppliers
	Level E	Continuation of Level D, plus plan identifies necessary equipment and onsite resources to support delivery.
	Level F	Continuation of Level E and construction equipment schedule is integrated with other project schedules.
(√)		2. Construction Equipment Productivity Analysis
	Level A	Construction equipment is not used on this building project.
	Level B	Construction equipment is utilized but requirements and usage are not planned and tracked.
	Level C	Construction equipment requirements are planned and tracked but they are not tied to a schedule. The usage is tracked against budget only.
	Level D	Continuations of Level C and regular meeting is conducted to assess the requirement of construction equipment.
	Level E	Continuation of Level D, and construction equipment resource curves are drawn and resources are leveled.
	Level F	Continuation of Level E and equipment schedule is adjusted based on the audit report that shows equipment downtime and other information.
(√)		3. Construction Equipment Maintenance
	Level A	Construction equipment maintenance is not applicable for this building project.
	Level B	Construction equipment maintenance is not planned for or this building project.
	Level C	Construction equipment is logged in simplified spreadsheet and maintenance is carried out by operator request.
	Level D	Continuation of Level C and equipment maintenance record is linked to individual construction equipment, and maintenance is centrally scheduled and administered.
	Level E	Continuation of Level D and a computer program is used to record and administer equipment maintenance information such as required and accomplished maintenance logs and usage logs.
	Level F	Continuation of Level E and maintenance schedule is electronically updated and maintenance due notices are automatically issued to concerned parties.

C3: Validated Survey Instrument for Management Practices Related to Construction Methods

Pleas	lease check ($$) one box per element in space provided in front of the level of implementation of the					
mana	anagement practice. Please do not leave any elements blank.					
(√)	1.Integrated Schedule					
	Level A The use of an integrated schedule using CPM is not applicable.					
	Level B The use of an integrated schedule using CPM has not been addressed					
	Level C	Developing a schedule with no resources present and managing schedule status via duration or remaining duration but no link to earned percent complete progress from associated deliverables per activity.				

	Level D	Developing a schedule with resources present but no link to earned percent complete progress from associated deliverables per activity			
	Level E	Developing a schedule with resources present but no link to earned percent complete progress			
	Lever	from associated deliverables per activity. Resources are updated to reflect current work content			
		or quantity adjustments.			
	Level F	Continuation of Level E and updated to include quantity adjustments. Earned progress for the			
		activity is based on measured or assessed work completed per deliverables per activity. Progress			
		measurement performed in application adapted specifically for each deliverable.			
()	2. Work	Schedule Strategies			
()	Level A	The development of a work schedule strategy is not applicable			
	Level B	evel B The development of a work schedule strategy has not been addressed			
	Level C	The strategy is based on a single work schedule be it either a straight time such as 36 hours per			
		week schedule, overtime, or other work schedule strategies.			
	Level D	Strategy considers multiple work schedules considering critical and near critical activity sequences.			
	Level E	Continuation of Level D, plus strategies consider the potential impact on worker fatigue,			
		supervision, safety, and absenteeism.			
	Level F	Continuation of Level E, plus each potential strategy's impact is analysed for manpower density			
		and congestion at an area or sub-area level.			
(√)	3. Sched	lule Control			
	Level A	The development of a schedule compliance plan is not applicable			
	Level B	The development of a schedule compliance plan has not been addressed			
	Level C	Consistent follow-up to monitor the following tasks: schedule updated periodically, critical path			
		analysis, and progress narrative prepared as required and effective team participation in schedule			
		updates.			
	Level D	Continuation of Level C, plus quantity reports are regularly performed. Upon request, or as the			
		project requires, may include any of the following: change management analysis, risks			
		assessment scenarios/analysis, date variance analysis to approved baseline or previous update			
		period, start / finish percent achieved ratio analysis, communication with material suppliers to			
		ensure material will arrive on site when planned.			
	Level E	Continuation of Level D, plus monitor the following: schedule rigorously updated based on			
		manual input of quantity reports, critical and near critical path analysis, progress narrative			
		prepared and effective team participation in schedule updates. Quantity reports rigorously done			
		by trained individual(s). Material suppliers routinely contacted to track the status of material			
	1 1 1 1	delivery dates.			
	Level F	Continuation of Level E, plus will consistently include all of the following, based on project			
		requirements and observed schedule status conditions: change management analysis, risks			
		assessment scenarios/analysis, date variance analysis to approved baseline or previous update			
		imaging and other techniques			
<u>.</u>					
(\mathbf{v})	<u>4. Dyna</u>	mic Sue Layout Plan			
	Level A	Site layout plan is not applicable for the project.			
	Level B	A site layout plan has not been addressed.			
	Level C	i ne project team examines the project schedule and assesses when Temporary Facilities (TF) will be brought in			
	Level D	Continuation of Level C plus what sizes will be needed prior to the start of the project. No			
	LeverD	consideration is given to the addition and removal of TFs at different stages of the project. No			
		analysis is done in regards to the layout of the project to optimize locations of the TFs to limit			
		travel time.			
	Level E	Continuation of Level D, plus consideration is given to the addition and removal of TFs at			
		different stages of the project.			
	Level F	Continuation of Level E, plus the team analyses the layout of the project including where the			
		different parties will be working and place their TFs in the optimum location in order to limit			
		travel time to and from TFs.			
$(\sqrt{)}$	5. Traffi	c Control Plan			
	Level A	Traffic control plans are not applicable for the project.			
<u> </u>	Level B	Traffic control plans have not been addressed for the project.			
	Level C	The project has some traffic control plans and is used on a reactive basis			
	Level D	The project has a traffic control plan equipment and an arrangement for daylight traffic control			
		only and has no trained traffic control persons.			
	Level F	The project has a traffic control plan and equipment for all times of the day including trained			
1		persons for traffic control			

r		
	Level F	Continuation of level E, plus a trained traffic control supervisor. It has an approved contingency
		plan in place to accommodate unexpected situations and has designed and constructed alternate
		arrangements for traffic such as detours, flyovers, etc.
(√)	6. Site S	ecurity Plan
	Level A	Site security plan is not applicable for the project.
	Level B	The site does not institute security in regards to entry to the site, securing commodities, or tools
		and equipment.
	Level C	The site controls entry and exit from the site, but does not have any other formal security
		throughout the site.
	Level D	The site has established security procedures including visitor sign in and sign out procedure and
		security guards at every gate. The site has implemented security measures to ensure the
		preservation of company assets. Protocols have been identified for searches of individuals and
		their personal property. Searches are conducted randomly.
	Level E	Continuation of Level D, plus the site has ensured that material is not leaving the job site by
		instituting "lock-ups" for items that are prone to theft.
	Level F	Continuation of Level E, plus the use of electronic security such as security cameras has been
		implemented.
()	7.Mach	inerv Positioning Strategy
	Level	Machinery positioning strategy is not applicable.
	А	
	Level B	There is no strategy for positioning of machinery at the project site.
	Level C	Heavy rigging and lifting studies are accomplished on all critical lifts including evaluation of the
		machinery, rigging selection, and crane location.
	Level D	Continuation of Level C, plus planning includes the use of 2D layout and studies to aid in
		constructability for locating and utilizing machinery.
	Level E	Continuation of Level D, plus some 3D modelling studies to aid in constructability for locating
		and utilizing machinery.
	Level F	Continuation of Level E, plus planning includes the use of 3D layout studies and 3D
		modelling/visualization to aid in constructability for locating and utilizing machinery.
()	8.Projec	et Start-up Plan
	Level A	No start-up plan exists.
	Level B	A partial start-up plan has been prepared; the plan has not been communicated to the concerned
		stakeholders.
	Level C	A basic start-up plan has been developed with input from the project participants, but the plan
		has not been implemented.
		Continuation of Level C, plus with considerations for interfaces among sub-contractors or
	Level D	project participants. A start-up plan has been developed that identifies the duties and
		responsibilities of each stakeholder.
		Continuation of Level D, plus with consideration for cost analysis and detailed scheduling
	Level E	components. The plan is well communicated to all the stakeholders.
		Continuation of Level E, plus with the plan being implemented on the project by incorporating
	Level F	feedbacks from the stakeholders or project participants.
(√)	9.Projec	ct Completion Plan
	Level A	The project completion requirement or handover procedure is not applicable.
	Level B	The project completion requirement or handover procedure has not been identified.
	Level C	The project has a handover procedure that defines the parameters of project completion and
		delineates the requirements for the handover.
	Level D	The project has a formal handover process that defines the necessary documentation, parameters
		of completion and other issues to assure proper handover of a project.
	Level E	Continuation of Level D, plus the procedure has been reviewed and agreed by the stakeholders.
	Level F	Continuation of Level E plus the plan is approved by project management team and is reviewed
<i>(</i>	1.0 -	tor applicability during all phases of the handover process.
(1)	10.Inno	vations and New Technologies
	Level A	Innovation in new materials, equipment, information systems is not applicable.
	Level B	Innovations and new technologies investigation is not addressed.
	Level C	The project does not have a formal program for the investigation of innovations and new
		technologies. Implementation of innovations and new systems will only occur after the industry-
		wide implementation.
	Level D	The organization has an informal program for the investigation of innovations, and they will
	r 1	investigate the feasibility of the new technologies on a regular basis.
	Level E	Continuation of Level D, plus the program is formal to investigate new systems and they will
	1	investigate the reasibility of the new technologies on a regular basis.

Level F Continuation of Level E, plus they investigate all new technologies using a formal system of rating the new technology.

C4: Validated Survey Instrument for Pre-Construction Phase Management Practices

Please check ($\sqrt{}$) one box per element in space provided in front of the level of implementation of the management practice. Please do not leave any elements blank. $\sqrt{}$ 1.Short Interval Plan Level A The use of short interval plan is not applicable to this building project. Level B Short interval plan is not prepared. Level C Activities in the project schedule are not resource loaded and short interval plan does not detail the required materials, tools and equipment, labour, and required project information. Short interval plan indicates the required materials, tools and equipment, labour, and project Level D information required to complete each task, and activities in the project schedule are not resource loaded. Level E Continuations of Level D, and activities in the project schedule are resource loaded. The short interval plan considers the effects of congestion. Continuation of Level E, and alternate plans are prepared such as modified equipment Level F schedules, shortened shifts, lengthened shifts, or added shifts in case the original plans did not work $\sqrt{}$ 2. Well-Defined Scope of Work Level A A well-defined scope of work is not applicable to this building project. Level B Working drawings are based on incomplete design, and execution is controlled with a milestone schedule. Level C Continuation of Level B, and design is complete but without buildability review; execution is controlled with a master schedule. Level D Continuation of Level C, plus design is completed and buildability review has been performed. Continuation of Level D, plus duration for the work package is defined, material availability, Level E testing and inspection requirements are defined, but budget and quantities are not included in the scope of work. Level F Continuation of Level E, and budget and quantities are included in the scope of work. $\sqrt{}$ 3. Use of Software in Planning Work Packages Level A Utilization of software in work packaging is not applicable. Level B This project uses a software system to generate and track work packages. The system is not integrated. Materials, drawing and other information are entered manually. Level C Continuation of Level B, and percent complete is entered by reviewing the package. Work package status is updated manually. Continuation of Level C, and the project uses a software system which automatically includes Level D the drawings and material delivery status. Schedule, percent complete, test and inspection status, and closure are entered manually by review of the work package. Level E Continuation of Level D, and the system is updated regularly and automatically, and provides current design drawing information, updated status of materials, implementation schedule with durations and quantities, test and inspection status, percent complete and closure. However work package status is updated to the master schedule manually. Level F Continuation of Level E, and work package status is updated electronically. √) 4.Dedicated Planner Level A Assigning dedicated planner is not applicable to this building project. Level B Hiring of dedicated planner has not been addressed. Level C Dedication of single planner or multiple personnel who prepare initial plan and coordinate with procurement are assigned. Level D Continuation of Level C, and the planner initiate and maintain communication regarding the delivery and inspection of materials and update the schedules as appropriate. Continuation of Level D, plus continued update of schedule, and onsite inspection to track Level E consumption of materials as necessary. Level F Continuation of Level E, plus continued communication regarding consumption of materials, and report their status. 353

$(\sqrt{)}$	5 Constr	uction Work Packages
	Level A	Organizing information for each Construction Work Packages (CWP) is not applicable.
	Level B	Information such as scope of works; safety, quality, materials, personnel and equipment requirements; and others that are required for the execution of each work package are compiled.
	Level C	Some information required for execution of CWP is complete.
	Level D	All information required for execution of each CWP have been addressed. The information in each section displays no in-depth consideration.
	Level E	Continuation of Level D, plus in-depth consideration for most of the sections are given.
	Level F	Continuation of Level E, plus all sections have received in-depth consideration. The construction work package can be performed easily based on the information documented for each CWP.
(√)	6.Builda	bility Review
	Level A	Buildability review is not applicable to this building project.
	Level B	Buildability review is not addressed
	Level C	Some scheduling and coordination of the phases of construction has been performed by principal contractor before conducting detailed buildability review.
	Level D	Continuation of Level C, plus the principal contractor has prepared a detailed schedule for all phases of construction after conducting detailed buildability review.
	Level E	Continuation of Level D, plus the owner is involved in buildability review process.
	Level F	Continuation of Level E, plus all buildability reviews and schedules are completed before construction
(√)	7.Utilitie	s Alignment
	Level A	Utility alignment is not applicable to this project.
	Level B	Utility alignment is not addressed in this project.
	Level C	The project has a review process for working in areas surrounding utilities. The process includes list of utilities, locations, and warning signs.
	Level D	Continuation of Level C, plus necessary city council approvals have been obtained and utility adjustment plan is prepared.
	Level E	Continuation of Level D, plus communication with businesses and local community have been made. Arrangements for temporary shutoffs and interruptions have been made.
	Level F	Continuation of Level E, plus the physical markings described to mark the horizontal route of underground facilities have been protected, detours have been constructed, and the all activities related to utilities alignments have been integrated with other project schedules.
(√)	8.Contra	ict Types
	Level A	Contracting strategies are not applicable.
	Level B	Contract types are not addressed.
	Level C	The project has a review process for the contract types. It considers fee structure and financing options, but risks associated with different contract types are not considered.
	Level D	Continuation of Level C, plus risks associated with different contract types are also considered. Cost benefit analysis for different types of contracts are performed.
	Level E	Continuation of Level D, plus the labour market information and reports are also studied and considered before awarding contracts.
	Level F	Continuation of Level E, plus the local conditions in the project area are studied and monitored regularly. The company has established protocols for selecting sub-contractors, designers, material suppliers, and manufacturers.
(√)	9.Model	Development
	Level A	Models are not required for this project.
	Level B	Integration of the projects 3D and schedule information has not been addressed.
	Level C	A 4D Model has been established for the project.
	Level D	Continuation of Level C and the model is updated manually.
	Level E	Continuation of Level D, plus the model is dynamic and includes material specifications, change order documentation, and other pertinent design and construction information.
	Level F	Continuation of Level E, plus the model is automatically updated based on work progress.
(√)	10.Regul	atory Requirements

Level A	Regulatory requirements are not applicable.
Level B	Required permitting has not been addressed.
Level C	Initial investigations based on the type of permits; timeline to get the permit approved; the validity period of the permit; the authority issuing the permit; pre-inspection requirements; and fees have been conducted.
Level D	Continuation of level C, plus permit requirements are tied to schedules and milestones.
Level E	Continuation of Level D, plus a system is established to track the acquisition and validity period of each permit.
Level F	Continuation of Level E, plus the system is automatically updated based on continued updates to schedule and permit expiry dates.

C5: Validated Survey Instrument for Human Resource Management Practices

Please check ($\sqrt{}$) one box per element in space provided in front of the level of implementation of the management practice. Please do not leave any elements blank.

	•	•	
(√)	1. Crew	Composition	
	Level A	Crew formation/composition is not applicable to this building project.	
	Level B	Crew composition is not addressed in this project.	
	Level C	Crew formation is addressed after the commencement of the construction of the building project.	
	Level D	Crew formation is addressed before the commencement of construction works; experience and knowledge of employees, job requirements, and location are considered.	
	Level E	Continuation of Level D, plus the performance of the crew is assessed after completion of each task and corrective measures are taken accordingly to improve productivity.	
	Level F	Continuation of Level E, plus the crew formation and its performance is assessed regularly on daily, weekly, and monthly basis and necessary changes are made.	
(√)	2.Skill A	ssessment and Evaluation	
	Level A	Skills assessment is not applicable to this project.	
	Level B	Skills assessment is not addressed.	
	Level C	Skills assessment is addressed after the beginning of the construction this project.	
	Level D	Skills assessment is addressed before the beginning of construction work; previous experience, skill and knowledge of workers are considered.	
	Level E	Continuation of Level D, plus the assessment is made again after completion of each task and adjustments are made accordingly to improve productivity.	
	Level F	Continuation of Level E, plus the assessment is performed regularly on daily, weekly, and monthly basis, and necessary changes are made and monitored.	
(√)	3. Emplo	oyees Training	
	Level A	Employees Training is not applicable to this building project.	
	Level B	Employees Training is not addressed.	
	Level C Employees Training is addressed on the jobsite after the beginning of the project.		
	Level D	Employees Training is provided to an employee when he/she begins working for the company, and if needed extra training will occur on the job site.	
	Level E	Continuation of Level D, plus worker is certified to work in or supervise specific trade. Before each project, new training in the trade will take place if necessary.	
	Level F	Continuation of Level E, plus a worker takes part in training for new technologies that are introduced in that trade annually and bi-annually.	
(√)	4.Caree	r Development	
	Level A	Career development is not applicable to this building project.	
	Level B	Career development is not addressed in this project.	
	Level C	The organization does not have a formal career development plan for employees, but the management will discuss future plans with them.	
	Level D	The organization has a formal career development plan for employees, but it only addresses short term career developments.	
	Level E	Continuation of Level D, plus it addresses long term career developments and options.	
	Level F	Continuation of Level E, plus addresses the expected performance of the employee and how the performance will affect his/her career development.	
(√)	5. Non-l	Financial Incentive Programs	

	Level A	Non-financial incentive/recognition programs are not applicable to this building project.		
	Level B	Non-financial incentive programs are not addressed in this project.		
	Level C	The organization has an informal recognition programs that will recognize employees occasionally, but not in a formal manner.		
	Level D	The organization has a formal recognition program that provides recognition on long term basis.		
	Level E	Continuation of Level D, plus it recognizes workers on a regular basis for both positive safety results and good safety behaviour.		
	Level F	Continuation of Level E, plus the rewards are given on both short and long term basis, and they are recognized by the upper management of the organization.		
(√	6. Financ	cial Incentive Programs		
	Level A	Financial incentive program is not applicable to this building project.		
	Level B	Financial incentive program is not addressed.		
	Level C The organization has an informal incentive program that will recognize employ			
	Level D	The organization has a formal incentive program that provides incentives on long term basis.		
	Level E	Continuation of Level D, plus it provides a monetary bonus for employees based on their performances.		
	Level F	Continuation of Level E, plus the rewards are given on both a short and long term basis, and they are recognized by the upper management of the organization.		
(√)	7.Social	Activites		
	Level A	Social activities for the employees are not applicable.		
	Level B	Social activities for the employees are not addressed.		
	Level C	The organization does not formally plan social activities for the employees, and there is only a yearly organization wide social activity.		
	Level D	The organization formally plans a social activity once or twice a year in which the project managers will attend, along with a yearly organization wide social activity.		
	Level E	Continuation of Level D, plus several times throughout the year which the project managers		
	Level F	Continuation of Level F, plus monthly which the project managers will attend and upper management including the president will attend on a quarterly basis, along with a yearly organization wide social activity.		
(√)	8. Stabili	ty of Organisational Structure		
	Level A	Maintaining the Stability of the Organizational Structure is not important.		
	Level B	No plans to manage change of key people in contract.		
	Level C	The key professionals are named or define in the contract.		
	Level D	Continuation of Level C, plus state that they cannot be changed without notice and prior approval.		
	Level E	Continuation of Level D, plus there are pre-approved designated successors.		
	Level F	Continuation of Level E, plus the contract specifies all professionals of sub-contractor and principal contractor teams, along with possible successors and right of approval by the other party.		
(√)	9. Clear	Delegation of Responsibility		
	Level A	Clear Delegation of Responsibility is not applicable to this building project.		
	Level B	There is simple and centralized delegation of responsibility.		
	Level C	There is simple and very formal delegation of responsibility.		
	Level D	There is stable project environment and more formal delegation of responsibility.		
	Level E	The delegation of responsibility is formal, but differing between technical, administrative, etc.		
	Level F	There is a formal delegation of authority that is clearly defined for all involved parties. The plan is reviewed periodically and evolves when necessary.		
(√)	10. Reter	tion Plan for Experienced Personnel		
	Level A	Retention Plan for Experienced Personnel is not applicable to this building project.		
	Level B	A retention plan is not addressed.		
	Level C	Each project manager is responsible for retention of his workers.		

	Level D	Incentives such as employee training and others are available but it is not requirement. Junior staffs have higher pay and preferred hiring status on the next project for the same employer.					
	Level E	Continuation of Level D, plus employee training is required for junior staff. The employer makes available a list of opportunities for the next project.					
	Level F	Continuation of Level E, plus testing and certification on the site lead to pay increases. Employer meets with individual worker and offers job(s) at new project site(s) as per requirements.					
(√)	11. Exit Interviews						
	Level A Exit Interview is not applicable.						
	Level B No exit interview.						
	Level C	There is exit interview for key personnel only.					
	Level D	Random exit interviews when there is time.					
	Level E	Formal exit interview for all personnel.					
	Level F	Formal exit interview for all personnel and feedback to management about lessons learned and how to improve retention when applicable.					

C6: Validated Survey Instrument for Safety and Health Practices

Plea	ase check (() one box per element in space provided in front of the level of implementation of the				
		management practice. Please do not leave any elements blank.				
(√)		1. Housekeeping				
	Level A	Housekeeping is not applicable to this building project				
	Level B	Regular housekeeping has not been addressed on the project.				
	Level C	Housekeeping occurs only after incidents occur.				
Level D		Housekeeping occurs on a bi-weekly scheduled basis.				
	Level E	Major travel paths are organized and clean. "Roll backs" are held weekly.				
	Level F	All work areas are well organized and designated crews are regularly cleaning				
(√)		2.Safety and Health Policy				
	Level A	Health and safety policy for the organization is not applicable				
	Level B	The company does not have a formal health and safety policy.				
	Level C	There is health and safety policy at company level and at particular project . The policy is				
	Level D	Continuation of Level C, plus it is periodically updated based on the organizational and industry feedback.				
	Continuations of Level D, plus the organization screens sub-contractors for their health and safety programs and chooses those with records of good performance.					
	Level F	Continuation of Level E, plus health and safety policy is integrated with procurement processes. Money is budgeted on the construction projects to address various health and safety issues.				
(√)		3. Safety and Health Plan				
	Level A	Health and safety plans are not applicable to this building project.				
	Level B	No health and safety targets have been examined and considered for the project.				
	Level C	Some techniques to reduce accidents on the project are utilized. The project has a reactive approach towards safety.				
	Level D	Most but not accident reduction techniques are utilized on the project.				
	Level E	All potential accident reduction techniques are utilized on the project.				
	Level F	Zero Accident Techniques are fully utilized on the project. The project has a very proactive approach towards safety.				
(√)		4.Safe Work Method Statement				
	Level A	Task Safety Analysis or Safe Work Method Statement (SWMS) is not applicable to this building project.				
	Level B	No Task Safety Analysis or SWMS is utilized.				
	Level C	Limited Task Safety Analysis is utilized only on high risk areas of the project. The project has a reactive approach towards safety.				
	Level D	SWMS is prepared and utilized for most tasks on this project.				

	Level E	Safe Work Method Statement (SWMS) is utilized daily on the project.					
	Level F	SWMS's are utilized daily on the projects on all tasks and some crews perform additional					
	Leveri	SWMS's as task changes.					
(√)		5. Hazards Analysis					
	Level A	The process for hazard identification process is not applicable on this building project.					
	Level B	No hazard identification process is in place on this project.					
	Level C	Hazards are identified for a few tasks that not covered in SWMS.					
	Level D	Hazards are identified for most tasks that are not covered in SWMS.					
	Level E Hazards are identified for all proposed scope of works.						
	Level F Hazards are identified for all proposed scope of work and incorporated into the project' specific safety planning process.						
(√)		6. Health and Safety Training					
	Level A	Health and Safety training is not applicable to this building project.					
	Level B The project does not have a project specific training program for new employee.						
	Level C	There is project specific induction regarding personal protective equipment, housekeeping and access to site, ladders and safe access to elevated platforms, fall protection, excavations and trenching, tools and equipment, electrical hazards and fire prevention. Supervisors receive additional induction about conducting of safety meetings, first aid and medical treatment processes, consequences for violation of job site work rules and violence, alcohol and drugs in the workplace.					
	Level D	Continuation of Level C, plus some employees should pass fitness for duty testing prior to starting construction of some tasks. The induction addresses management commitment, general project safety rules, emergency procedures, hazard communication, housekeeping, barricades, injury/illness reporting, lock-out and tag-out processes, confined spaces, compressed gas cylinders, back injury prevention, and hand/power tool safety.					
	Level E	Continuation of Level D, plus orientation addresses zero accidents philosophy and general project safety rules.					
	Level F	Continuation of Level E, plus workers trained on behavioural based training.					
(√)	1	7. Toolbox Safety Meetings					
	Level A	The Project does not conduct safety meetings.					
	Level B	The Project issues toolbox topics via handouts to employees on a periodic basis.					
	Level C	The project conducts a monthly meeting at or near breaks. Meetings reiterate job site safety rules.					
	Level D	The project conducts a weekly meeting at or near breaks. Meetings reiterate job site safety rules.					
	Level E The project conducts weekly meetings at a prearranged time, generally at the st Meetings address current job status and hazards presented by upcoming pro corrective actions, review recorded injuries and near misses, or reiterate job si and expectations. Time is set aside during the meeting for interactive discussive worker feedback.						
	Level F	Continuation of Level E, plus the day of the meeting vary on which they occur, or conduct them daily. Meetings address current job status and hazards presented by upcoming project activities, corrective actions, review recorded injuries and near misses, or reiterate job site safety rules and expectations.					

Thank you very much for your time.

Argaw Gurmu PhD Candidate at The University of Melbourne Faculty of Architecture, Building and Planning Parkville, Vic 3010 M-0424662009 <u>E-agurmu@student.unimelb.edu.au</u>

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

APPENDIX 2: RAW	DATA FOR DESCRI	IPTIVE DATA ANALYSIS
------------------------	------------------------	-----------------------------

Deamour Jarre ?		Long Long Long	Mat1-	Material-	Mati-1	Materia 1
Respondents	Procurement	Long-Lead	Materials	Materials Delivery	Material	Materials
Code	Plans jor Materials	Identification	Databasa	Schedule	Process	Team
R1	4 00	5 00	<u>3 00</u>	3 00	3 00	3 00
R2	5.00	5.00	4.00	5.00	4.00	4.00
R3	4.00	5.00	4.00	4.00	3.00	3.00
R4	4.00	5.00	4.00	4.00	5.00	4.00
R5	5.00	5.00	1.00	3.00	3.00	3.00
R6	4 00	4 00	3.00	3.00	2.00	2.00
R7	4 00	5.00	4 00	4 00	4 00	4 00
R8	4 00	5.00	4 00	4 00	3 00	2.00
R9	4 00	5.00	3 00	4 00	5.00	4 00
R10	5.00	4 00	4 00	5.00	4 00	4 00
R11	4 00	5.00	2.00	2.00	3 00	3.00
R12	3.00	4 00	3.00	3.00	3.00	3.00
R13	5.00	4 00	3.00	3.00	3.00	3.00
R14	5.00	5.00	5.00	5.00	3.00	2.00
R15	4 00	5.00	3.00	3.00	3.00	2.00
R16	5.00	5.00	4 00	4 00	3.00	2.00
R17	5.00	5.00	4 00	3.00	3.00	3.00
R18	4 00	4 00	3.00	3.00	3.00	3.00
R19	4 00	3.00	3.00	4 00	3.00	3.00
R20	3.00	4 00	2.00	3.00	4 00	1.00
R21	4 00	5.00	4 00	5.00	5.00	3.00
R22	4 00	4 00	2.00	4 00	3.00	2.00
R23	4 00	5.00	4 00	4 00	3.00	3.00
R24	4 00	5.00	4 00	4 00	5.00	2.00
R25	1.00	2.00	1.00	2.00	3.00	1.00
R26	3.00	4 00	2.00	3.00	2.00	1.00
R27	4 00	5.00	2.00	2.00	1.00	1.00
R28	5.00	5.00	5.00	5.00	5.00	4 00
R29	5.00	5.00	4 00	5.00	4 00	2.00
R30	5.00	5.00	4 00	5.00	4 00	4 00
R31	5.00	5.00	4 00	5.00	4 00	3.00
R32	5.00	4 00	3.00	4 00	4 00	3.00
R32	3.00	4 00	2.00	2.00	2.00	4 00
R34	5.00	5.00	3.00	4 00	4 00	4 00
R35	5.00	5.00	3.00	4 00	4 00	2.00
R36	5.00	5.00	4 00	4 00	3.00	3.00
R37	5.00	4 00	4 00	5.00	4 00	4 00
R38	4 00	5.00	4 00	3.00	4 00	3 00
R30	4.00	5.00	4.00	4 00	5.00	4.00

Respondents' Code	Procurement Plans for Construction Machinery	Construction Machinery Productivity Analyses	Construction Machinery Maintenance
R1	2.00	3.00	4.00
R2	4.00	5.00	4.00
R3	3.00	3.00	3.00
R4	5.00	5.00	5.00
R5	3.00	1.00	5.00
R6	3.00	4.00	3.00
R7	4.00	3.00	3.00
R8	4.00	3.00	4.00
R9	4.00	3.00	5.00
R10	3.00	4.00	5.00
R11	3.00	3.00	5.00
R12	3.00	3.00	3.00
R13	4.00	3.00	2.00
R14	2.00	2.00	2.00
R15	4.00	3.00	5.00
R16	3.00	4.00	5.00
R17	5.00	3.00	5.00
R18	2.00	2.00	2.00
R19	2.00	2.00	5.00
R20	3.00	3.00	4.00
R21	4.00	3.00	3.00
R22	3.00	3.00	4.00
R23	4.00	3.00	4.00
R24	4.00	5.00	5.00
R25	1.00	1.00	4.00
R26	3.00	2.00	4.00
R27	4.00	2.00	5.00
R28	4.00	4.00	5.00
R29	4.00	4.00	2.00
R30	5.00	4.00	4.00
R31	3.00	3.00	3.00
R32	4.00	2.00	5.00
R33	2.00	1.00	1.00
R34	5.00	3.00	3.00
R35	3.00	3.00	5.00
R36	4.00	3.00	5.00
R37	5.00	3.00	4.00
R38	3.00	3.00	5.00
R39	4.00	4.00	4.00

A2: Raw Data for Descriptive Analysis of Construction Equipment Management Practices

Respond	d Integrated	Works	Schedule	Dynamic Site	Traffic	Site	Machinerv	Project	Project	Innovations
ents'	schedule	Schedule	control	Layout Plan	Control	Security	Positioning	Start-up	Completion	and New
Code		Strategies			Plan	Plan	Strategy	Plan	Plan	Technologies
R1	4.0	4.0	4.0	4.0	5.0	3.0	4.0	4.0	4.0	4.0
R2	4.0	5.0	5.0	5.0	5.0	5.0	4.0	4.0	4.0	4.0
R3	3.0	4.0	5.0	4.0	4.0	4.0	4.0	5.0	5.0	3.0
R4	4.0	5.0	5.0	3.0	4.0	4.0	5.0	4.0	5.0	4.0
R5	1.0	4.0	3.0	4.0	4.0	3.0	2.0	5.0	5.0	1.0
R6	3.0	4.0	5.0	3.0	3.0	3.0	3.0	4.0	4.0	3.0
R7	4.0	4.0	4.0	5.0	5.0	4.0	5.0	4.0	3.0	3.0
R8	4.0	3.0	4.0	5.0	4.0	4.0	4.0	5.0	5.0	3.0
R9	4.0	4.0	4.0	3.0	5.0	4.0	5.0	2.0	5.0	3.0
R10	4.0	5.0	4.0	4.0	5.0	3.0	4.0	4.0	4.0	4.0
R11	4.0	4.0	4.0	5.0	5.0	3.0	5.0	5.0	3.0	2.0
R12	3.0	3.0	4.0	4.0	4.0	4.0	4.0	5.0	5.0	4.0
R13	2.0	3.0	4.0	4.0	4.0	4.0	5.0	5.0	5.0	2.0
R14	1.0	3.0	3.0	3.0	3.0	3.0	5.0	5.0	1.0	3.0
R15	3.0	4.0	4.0	4.0	5.0	5.0	5.0	4.0	4.0	3.0
R16	2.0	4.0	5.0	5.0	4.0	5.0	5.0	4.0	5.0	4.0
R17	4.0	4.0	4.0	5.0	5.0	3.0	5.0	5.0	5.0	3.0
R18	4.0	3.0	3.0	4.0	5.0	3.0	4.0	5.0	4.0	2.0
R19	3.0	4.0	5.0	3.0	5.0	3.0	5.0	3.0	4.0	3.0
R20	4.0	3.0	3.0	4.0	5.0	3.0	3.0	3.0	3.0	4.0
R21	4.0	4.0	4.0	5.0	4.0	4.0	5.0	5.0	4.0	5.0
R22	3.0	3.0	3.0	3.0	4.0	3.0	4.0	4.0	4.0	3.0
R23	4.0	4.0	4.0	4.0	3.0	4.0	4.0	4.0	5.0	3.0
R24	3.0	3.0	5.0	5.0	5.0	4.0	4.0	4.0	4.0	4.0
R25	2.0	3.0	3.0	4.0	4.0	3.0	2.0	5.0	5.0	2.0
R26	3.0	4.0	4.0	4.0	4.0	4.0	4.0	3.0	2.0	3.0
R27	3.0	3.0	3.0	4.0	5.0	2.0	4.0	4.0	4.0	2.0
R28	4.0	4.0	4.0	4.0	5.0	5.0	4.0	5.0	5.0	4.0
R29	3.0	3.0	3.0	4.0	5.0	3.0	4.0	4.0	5.0	3.0
R30	4.0	4.0	5.0	4.0	4.0	4.0	5.0	5.0	4.0	3.0
R31	2.0	4.0	3.0	3.0	3.0	4.0	2.0	4.0	4.0	5.0
R32	3.0	2.0	2.0	5.0	4.0	4.0	5.0	4.0	3.0	3.0
R33	3.0	3.0	3.0	3.0	3.0	3.0	4.0	3.0	3.0	4.0
R34	3.0	3.0	4.0	4.0	3.0	2.0	2.0	4.0	4.0	3.0
R35	3.0	4.0	4.0	5.0	5.0	5.0	5.0	5.0	3.0	3.0
R36	4.0	4.0	4.0	4.0	4.0	5.0	3.0	5.0	5.0	4.0
R37	5.0	4.0	4.0	5.0	4.0	3.0	4.0	5.0	5.0	2.0
R38	3.0	3.0	3.0	4.0	4.0	4.0	4.0	4.0	4.0	2.0
R39	4.0	4.0	4.0	4.0	4.0	4.0	4.0	4.0	5.0	4.0

A3: Raw Data for Descriptive Analysis of the Management Practices Related to Construction Methods

Respond ents' Code	Short Interval Planning	Well Defined Scope of Work	Use of Software in Planning	Dedicated Planner	Construction Work Packages	Buildability Review	Utilities Alignme nt	Contract types	: Model Development	Regulatory Requirements
R1	5	5	3	3	4	4	3	4	3	5
R2	5	5	4	4	5	4	4	5	4	5
R3	4	4	4	4	4	5	3	4	3	4
R4	5	5	5	3	5	4	5	5	3	5
R5	5	5	5	1	5	4	2	3	1	3
R6	5	5	3	2	3	5	3	3	2	2
R7	4	4	3	3	4	4	4	5	3	4
R8	4	5	3	3	4	4	4	4	2	4
R9	5	5	4	4	5	5	4	4	2	5
R10	5	5	4	4	5	4	4	4	3	5
R11	4	5	5	1	4	4	3	3	1	5
R12	4	5	3	3	4	4	4	4	2	4
R13	5	5	3	3	5	5	4	5	2	5
R14	5	5	3	3	3	5	5	3	3	5
R15	4	5	4	3	4	5	3	4	4	4
R16	5	5	5	3	3	4	5	5	4	3
R17	5	5	5	5	5	5	5	4	3	5
R18	2	5	5	1	4	4	5	4	1	5
R19	5	5	4	3	4	4	3	4	3	3
R20	4	5	3	2	3	4	4	2	3	5
R21	5	5	5	3	5	5	4	5	2	5
R22	4	4	3	3	4	3	3	4	3	4
R23	4	5	4	5	4	4	4	4	4	4
R24	5	4	3	5	5	5	5	3	5	5
R25	3	5	5	3	4	4	3	3	3	4
R26	4	2	2	4	5	5	3	3	4	4
R27	5	5	3	4	4	4	4	4	2	5
R28	5	5	4	5	5	5	5	5	3	5
R29	5	4	2	1	5	4	3	4	1	4
R30	5	5	5	4	4	5	4	4	3	5
R31	3	5	4	2	5	4	3	4	2	3
R32	4	5	3	2	4	4	4	4	2	4
R33	4	5	2	3	4	3	3	3	1	2
R34	4	5	4	4	5	5	4	3	3	4
R35	4	5	4	2	4	5	4	4	2	4
R36	5	5	4	3	4	4	4	4	5	5
R37	4	5	5	3	5	5	4	5	3	5
R38	4	5	4	2	3	5	3	2	2	4
R39	4	5	5	3	5	5	4	4	3	4

A4: Raw Data for Descriptive Analysis of the Pre-Construction Phase Management Practices

362

Respond	Crew	Skill	Employee	Career	Non-	Financial	Social	Stability of	Clear	Retention	Exit
ents' Code	<i>Composition</i>	i Assessme nt	es Training	Developm ent	Financial Incentive Program	Incentive Program	Activitie s	al Structure	Delegation of Responsibility	Plan for Experience d Personnel	Interview !
R1	4	3	3	4	4	4	3	4	4	3	2
R2	5	5	5	5	4	5	4	5	5	5	4
R3	4	4	4	4	3	3	3	4	3	3	3
R4	5	4	4	4	3	3	4	3	5	5	5
R5	4	1	2	1	1	1	3	4	5	3	1
R6	2	2	1	2	3	4	3	5	5	4	2
R7	4	4	4	5	4	4	3	4	4	4	3
R8	4	4	4	4	4	1	3	4	4	3	3
R9	4	4	5	4	3	1	2	5	4	5	3
R10	4	4	3	3	4	5	4	4	5	4	5
R11	4	4	4	3	2	2	4	4	4	4	2
R12	5	4	4	4	4	4	3	4	5	4	4
R13	4	4	3	4	5	5	4	4	4	3	2
R14	4	4	3	1	2	4	2	5	4	3	1
R15	4	4	4	3	3	3	3	4	5	4	3
R16	5	5	4	4	4	2	2	3	4	5	4
R17	5	4	5	4	4	4	5	5	4	5	4
R18	3	3	4	4	1	4	3	5	4	4	4
R19	4	4	4	3	3	3	2	3	5	3	3
R20	3	4	4	4	3	3	4	4	4	3	3
R21	5	5	4	4	5	4	3	3	5	3	2
R22	4	4	4	4	3	3	4	4	3	4	1
R23	3	4	4	3	3	3	3	4	4	3	4
R24	5	5	5	4	3	4	5	5	5	4	4
R25	4	4	4	4	2	2	4	4	4	4	2
R26	3	4	4	4	4	4	3	3	2	2	3
R27	5	3	3	3	4	4	3	4	4	4	2
R28	5	5	5	4	4	4	3	4	5	4	4
R29	3	1	2	2	2	2	3	4	5	4	2
R30	4	4	4	4	5	3	4	5	5	5	2
R31	3	4	4	3	4	4	2	4	3	5	3
R32	2	3	3	3	3	4	2	3	3	2	2
R33	1	1	1	3	3	4	4	4	4	3	3
R34	3	3	3	3	3	3	3	2	3	4	3
R35	2	4	5	4	1	3	4	5	4	5	2
R36	4	4	5	5	3	4	5	5	5	4	4
R37	4	4	4	4	4	3	4	5	4	4	2
R38	5	3	4	3	3	3	3	4	3	4	3
R39	4	4	4	4	4	5	5	4	5	4	3

A5: Raw Data for Descriptive Analysis of Human Resource Management Practices

363

Respon dents' Code	House Keeping	Health and Safety Policy	Health and Safety Plans	Safe Work Method Statement	Hazards Analysis	Health and Safety Training	Toolbox Safety Meetings
R1	4	5	5	5	5	4	5
R2	5	5	5	5	5	4	5
R3	3	5	5	5	5	5	4
R4	5	5	5	5	5	5	5
R5	5	5	5	5	4	3	3
R6	4	5	3	5	5	3	5
R7	4	5	5	5	5	5	5
R8	4	5	5	5	5	5	5
R9	4	4	5	5	5	5	4
R10	5	5	5	4	5	4	5
R11	5	5	4	5	5	5	5
R12	5	5	5	5	5	5	5
R13	4	5	5	4	3	3	5
R14	4	4	4	3	4	3	3
R15	5	5	5	4	5	5	4
R16	5	5	5	5	5	5	5
R17	5	4	4	5	5	4	5
R18	4	5	5	5	5	5	4
R19	3	5	5	5	5	5	5
R20	4	5	5	4	5	5	5
R21	4	5	5	5	5	5	4
R22	4	4	3	4	4	4	4
R23	4	4	4	4	4	4	4
R24	5	5	5	5	5	5	5
R25	5	5	5	5	5	5	5
R26	5	4	5	5	5	5	5
R27	4	5	5	5	5	5	5
R28	5	5	5	5	5	5	5
R29	5	5	5	4	4	4	4
R30	5	5	5	5	5	5	5
R31	5	5	5	5	5	5	5
R32	4	5	4	4	4	4	4
R33	4	3	3	3	3	3	3
R34	4	4	4	3	3	3	3
R35	5	5	4	5	5	5	5
R36	5	5	5	5	5	5	5
R37	4	4	4	3	3	5	5
R38	4	5	4	4	4	3	4
R39	5	5	5	5	5	5	5

A6: Raw Data for Descriptive Analysis of the Safety and Health Practices

APPENDIA 3: SCORES OF THE BUILDING PROJE
--

A1:	Construction	Materials	Management	Practices Sco	res
-----	--------------	------------------	------------	---------------	-----

R	PF	PPM	LLM	MSD	MDS	MIP	MIT	MMTS
R1	1.00	0.51	0.55	0.13	0.30	0.28	0.11	1.88
R2	0.87	0.85	0.92	0.40	0.60	0.56	0.46	3.77
R3	1.02	0.68	0.73	0.53	0.30	0.56	0.11	2.91
R4	1.02	0.68	0.73	0.53	0.30	0.56	0.34	3.14
R5	1.01	0.34	0.73	0.40	0.30	0.28	0.23	2.28
R6	0.82	0.51	0.55	0.00	0.30	0.00	0.00	1.36
R7	0.78	0.85	0.92	0.26	0.45	0.56	0.46	3.49
R8	1.06	0.68	0.73	0.26	0.30	0.56	0.46	2.99
R9	1.14	0.51	0.73	0.53	0.45	0.42	0.34	2.98
R10	1.13	0.85	0.73	0.66	0.60	0.56	0.57	3.97
R11	0.59	0.34	0.37	0.13	0.30	0.28	0.11	1.53
R12	0.79	0.51	0.37	0.26	0.45	0.56	0.34	2.49
R13	1.07	0.51	0.37	0.13	0.30	0.42	0.11	1.84
R14	0.82	0.68	0.73	0.13	0.15	0.28	0.00	1.97
R15	0.88	0.34	0.37	0.26	0.30	0.56	0.34	2.17
R16	1.00	0.34	0.37	0.26	0.15	0.56	0.11	1.79
R17	0.67	0.85	0.18	0.13	0.45	0.42	0.34	2.37
R18	1.00	0.51	0.73	0.00	0.60	0.28	0.11	2.23
R19	1.08	0.51	0.37	0.00	0.30	0.28	0.00	1.45
R20	0.88	0.34	0.37	0.26	0.30	0.14	0.00	1.41
R21	0.96	0.68	0.73	0.26	0.45	0.42	0.11	2.66
R22	1.00	0.51	0.73	0.00	0.30	0.56	0.34	2.44
R23	0.89	0.68	0.37	0.53	0.30	0.56	0.23	2.66
R24	0.92	0.17	0.18	0.00	0.30	0.42	0.11	1.18
R25	0.96	0.17	0.18	0.00	0.45	0.28	0.00	1.08
R26	0.76	0.17	0.18	0.13	0.30	0.42	0.11	1.32
R27	1.08	0.51	0.55	0.13	0.45	0.28	0.00	1.92
R28	0.60	0.34	0.37	0.13	0.15	0.56	0.11	1.66
R29	0.92	0.51	0.37	0.26	0.30	0.56	0.34	2.34
R30	1.12	0.85	0.92	0.66	0.75	0.28	0.46	3.91
R31	0.91	0.34	0.92	0.26	0.60	0.42	0.23	2.77
R32	1.14	0.68	0.73	0.53	0.60	0.70	0.57	3.81
R33	0.90	0.17	0.55	0.26	0.45	0.14	0.00	1.57
R34	1.08	0.85	0.92	0.26	0.30	0.42	0.23	2.97
R35	1.06	0.34	0.37	0.00	0.30	0.56	0.11	1.68
R36	1.00	0.51	0.73	0.26	0.45	0.56	0.23	2.74
R37	1.08	0.85	0.92	0.66	0.60	0.56	0.34	3.92
R38	1.00	0.34	0.37	0.13	0.30	0.28	0.11	1.53
R39	1.10	0.85	0.92	0.53	0.30	0.56	0.46	3.61

Legend: R= Respondents, PF= Productivity Factor, MMTS= Materials Management Total Score, PPM=Procurement Plan for Materials, LLM= Long-lead Materials Identification, MSD=Materials Status Database, MDS= Materials Delivery Schedule, MIP=Materials Inspection Process MIT = Materials Inspection Team

R	PF	PPCE	CEPA	СЕМ	СЕМР
R1	1.00	0.14	0.12	0.32	0.58
R2	0.87	0.55	0.48	0.32	1.35
R3	1.02	0.28	0.24	0.32	0.83
R4	1.02	0.28	0.48	0.32	1.08
R5	1.01	0.14	0.12	0.32	0.58
R6	0.82	0.14	0.12	0.63	0.89
R7	0.78	0.55	0.48	0.63	1.67
R8	1.06	0.28	0.36	0.47	1.11
R9	1.14	0.55	0.36	0.32	1.23
R10	1.13	0.55	0.36	0.32	1.23
R11	0.59	0.14	0.12	0.32	0.58
R12	0.79	0.42	0.36	0.63	1.41
R13	1.07	0.42	0.24	0.32	0.97
R14	0.82	0.00	0.12	0.32	0.44
R15	0.88	0.28	0.12	0.32	0.71
R16	1.00	0.00	0.12	0.47	0.59
R17	0.67	0.00	0.24	0.47	0.72
R18	1.00	0.00	0.00	0.00	0.00
R19	1.08	0.00	0.24	0.32	0.56
R20	0.88	0.14	0.12	0.32	0.58
R21	0.96	0.14	0.12	0.32	0.58
R22	1.00	0.55	0.36	0.47	1.39
R23	0.89	0.42	0.48	0.47	1.37
R24	0.92	0.14	0.12	0.32	0.58
R25	0.96	0.28	0.24	0.16	0.68
R26	0.76	0.00	0.00	0.00	0.00
R27	1.08	0.55	0.36	0.63	1.55
R28	0.60	0.00	0.00	0.16	0.16
R29	0.92	0.55	0.61	0.32	1.47
R30	1.12	0.69	0.48	0.32	1.49
R31	0.91	0.28	0.36	0.32	0.96
R32	1.14	0.69	0.36	0.63	1.69
R33	0.90	0.14	0.12	0.16	0.42
R34	1.08	0.28	0.48	0.32	1.08
R35	1.06	0.00	0.12	0.47	0.59
R36	1.00	0.55	0.36	0.47	1.39
R37	1.08	0.69	0.61	0.79	2.09
R38	1.00	0.14	0.24	0.32	0.70
R39	1.10	0.55	0.12	0.47	1.15

A2: Construction Equipment Management Practices Scores

Legend: R= Respondents, PF= Productivity Factor, R= respondents, CEMP= Total score of the Construction Equipment Management Practices, PPCE=Procurement Plans for Construction Equipment, CEPA=Construction Equipment Productivity Analysis, CEM=Construction Equipment Maintenance

AS: PI	roject S	cores of	Manage	ment Pr	actices R	elated to	o Consi	ructio	n Me	tnoas		
R	PF	IS	WSS	SC	DSLP	ТСР	SSP	MPS	PSP	PCP	INT	CMTS
R1	1.00	0.16	0.17	0.15	0.11	0.26	0.15	0.31	0.65	0.68	0.44	3.09
R2	0.87	0.49	0.84	0.61	0.57	0.66	0.59	0.62	0.82	0.85	0.73	6.77
R3	1.02	0.33	0.17	0.31	0.57	0.39	0.44	0.62	0.65	0.68	0.73	4.89
R4	1.02	0.49	0.84	0.76	0.57	0.39	0.44	0.46	0.33	0.68	0.59	5.55
R5	1.01	0.33	0.33	0.61	0.57	0.53	0.29	0.46	0.82	0.68	0.73	5.36
R6	0.82	0.16	0.50	0.61	0.23	0.66	0.73	0.31	0.65	0.85	0.73	5.44
R7	0.78	0.33	0.84	0.31	0.57	0.66	0.73	0.31	0.82	0.85	0.73	6.14
R8	1.06	0.65	0.84	0.31	0.57	0.66	0.73	0.46	0.82	0.68	0.44	6.16
R9	1.14	0.49	0.84	0.31	0.23	0.39	0.44	0.31	0.82	0.68	0.59	5.09
R10	1.13	0.33	0.67	0.61	0.57	0.53	0.44	0.62	0.82	0.85	0.29	5.72
R11	0.59	0.33	0.17	0.61	0.11	0.26	0.44	0.31	0.65	0.68	0.29	3.86
R12	0.79	0.49	0.50	0.61	0.34	0.39	0.29	0.46	0.65	0.51	0.59	4.85
R13	1.07	0.82	0.50	0.31	0.46	0.26	0.29	0.31	0.82	0.51	0.29	4.57
R14	0.82	0.33	0.84	0.76	0.57	0.26	0.44	0.31	0.16	0.34	0.29	4.31
R15	0.88	0.33	0.84	0.00	0.46	0.53	0.44	0.46	0.49	0.68	0.59	4.81
R16	1.00	0.16	0.67	0.00	0.00	0.66	0.73	0.31	0.65	0.85	0.29	4.33
R17	0.67	0.33	0.17	0.15	0.11	0.26	0.44	0.31	0.33	0.51	0.29	2.90
R18	1.00	0.33	0.67	0.00	0.00	0.53	0.44	0.31	0.82	0.68	0.29	4.06
R19	1.08	0.33	0.50	0.15	0.46	0.26	0.29	0.31	0.33	0.68	0.00	3.31
R20	0.88	0.33	0.17	0.15	0.46	0.66	0.44	0.31	0.82	0.34	0.29	3.96
R21	0.96	0.49	0.50	0.61	0.46	0.53	0.44	0.46	0.49	0.68	0.44	5.10
R22	1.00	0.82	0.67	0.61	0.34	0.26	0.44	0.46	0.82	0.85	0.59	5.86
R23	0.89	0.33	0.17	0.31	0.46	0.66	0.44	0.31	0.65	0.00	0.29	3.61
R24	0.92	0.16	0.17	0.15	0.11	0.13	0.15	0.31	0.65	0.51	0.29	2.64
R25	0.96	0.16	0.17	0.00	0.11	0.00	0.00	0.15	0.65	0.34	0.29	1.89
R26	0.76	0.16	0.33	0.15	0.11	0.00	0.29	0.31	0.49	0.34	0.59	2.78
R27	1.08	0.49	0.84	0.76	0.57	0.66	0.44	0.62	0.82	0.68	0.29	6.17
R28	0.60	0.00	0.17	0.00	0.00	0.26	0.44	0.31	0.33	0.34	0.29	2.14
R29	0.92	0.65	0.84	0.31	0.46	0.26	0.29	0.31	0.33	0.85	0.73	5.03
R30	1.12	0.33	0.84	0.61	0.00	0.39	0.44	0.31	0.82	0.85	0.73	5.31
R31	0.91	0.33	0.33	0.61	0.00	0.53	0.29	0.31	0.49	0.51	0.29	3.69
R32	1.14	0.49	0.50	0.15	0.46	0.53	0.44	0.62	0.82	0.85	0.73	5.58
R33	0.90	0.00	0.33	0.00	0.23	0.00	0.00	0.00	0.65	0.00	0.73	1.95
R34	1.08	0.49	0.33	0.15	0.46	0.66	0.44	0.31	0.82	0.85	0.73	5.24
R35	1.06	0.65	0.67	0.76	0.34	0.26	0.29	0.00	0.82	0.68	0.00	4.48
R36	1.00	0.82	0.84	0.76	0.46	0.39	0.44	0.31	0.82	0.68	0.44	5.95
R37	1.08	0.82	0.84	0.76	0.57	0.53	0.44	0.62	0.82	0.85	0.73	6.97
R38	1.00	0.49	0.67	0.31	0.46	0.53	0.44	0.46	0.82	0.68	0.73	5.58
R39	1.10	0.65	0.17	0.61	0.34	0.66	0.73	0.46	0.82	0.68	0.73	5.86

п . t C e 1 / 3.6 (1 . **D** 1 . . $\mathbf{\alpha}$.

Legend: R=Respondents, PF= Productivity Factor, R=respondents, CMTS= Construction Methods Total Score, IS=Integrated Schedule, WSS= Work Schedule Strategies, SC=Schedule Control, DSLP= Dynamic Site Layout Plan, TCP=Traffic Control Plan, SSP=Site Security Plan, MPS=Machinery Positioning Strategy, PSP=Project Start-up Plan, PCP= Project Completion Plan, INT=Innovation and new Technologies

A4: Pre-Construction Phase Management Practices Scores

				8							
PF	SIP	WSW	US	DP	CWP	BR	UA	СТ	MD	RR	PPMP
1.00	0.70	0.58	0.31	0.24	0.69	0.35	0.76	0.00	0.22	0.34	4.18
0.87	0.70	0.77	0.46	0.37	0.86	0.88	0.76	0.46	0.00	0.85	6.10
1.02	0.35	0.58	0.31	0.24	0.69	0.53	0.76	0.62	0.32	0.51	4.90
1.02	0.88	0.58	0.61	0.00	0.86	0.70	0.76	0.46	0.00	0.51	5.36
1.01	0.35	0.58	0.15	0.00	0.86	0.88	0.46	0.46	0.00	0.34	4.07
0.82	0.88	0.77	0.31	0.37	0.86	0.70	0.76	0.15	0.22	0.68	5.68
0.78	0.53	0.58	0.46	0.37	0.86	0.53	0.76	0.77	0.00	0.85	5.69
1.06	0.53	0.77	0.31	0.24	0.69	0.35	0.46	0.77	0.32	0.51	4.94
1.14	0.88	0.58	0.31	0.37	0.86	0.53	0.76	0.46	0.22	0.51	5.46
1.13	0.70	0.58	0.61	0.37	0.51	0.70	0.76	0.46	0.11	0.51	5.31
0.59	0.35	0.38	0.31	0.12	0.51	0.70	0.61	0.15	0.00	0.51	3.65
0.79	0.53	0.58	0.31	0.37	0.69	0.53	0.61	0.46	0.11	0.34	4.50
1.07	0.35	0.77	0.31	0.24	0.69	0.53	0.76	0.31	0.11	0.34	4.40
0.82	0.35	0.38	0.15	0.24	0.00	0.35	0.30	0.31	0.00	0.51	2.61
0.88	0.35	0.96	0.31	0.00	0.69	0.88	0.30	0.77	0.43	0.34	5.03
1.00	0.35	0.96	0.46	0.00	0.86	0.53	0.76	0.62	0.11	0.51	5.15
0.67	0.35	0.19	0.61	0.37	0.34	0.53	0.30	0.15	0.11	0.34	3.30
1.00	0.35	0.77	0.46	0.24	0.69	0.53	0.76	0.00	0.22	0.51	4.52
1.08	0.35	0.38	0.31	0.00	0.69	0.53	0.30	0.00	0.11	0.34	3.00
0.88	0.35	0.38	0.15	0.00	0.00	0.35	0.76	0.15	0.11	0.51	2.77
0.96	0.35	0.58	0.15	0.12	0.00	0.35	0.76	0.46	0.00	0.68	3.46
1.00	0.70	0.58	0.31	0.00	0.86	0.88	0.76	0.46	0.11	0.68	5.33
0.89	0.70	0.19	0.46	0.37	0.69	0.35	0.76	0.31	0.11	0.68	4.61
0.92	0.35	0.58	0.31	0.00	0.34	0.53	0.76	0.31	0.00	0.68	3.85
0.96	0.35	0.58	0.15	0.00	0.86	0.00	0.15	0.15	0.22	0.34	2.80
0.76	0.53	0.77	0.31	0.12	0.86	0.35	0.30	0.31	0.11	0.34	3.99
1.08	0.70	0.77	0.31	0.37	0.69	0.70	0.76	0.77	0.11	0.68	5.85
0.60	0.18	0.19	0.15	0.00	0.34	0.18	0.00	0.15	0.00	0.00	1.19
0.92	0.88	0.58	0.31	0.37	0.86	0.70	0.30	0.31	0.11	0.85	5.25
1.12	0.53	0.58	0.76	0.61	0.69	0.35	0.61	0.31	0.22	0.85	5.50
0.91	0.35	0.58	0.31	0.37	0.51	0.53	0.61	0.31	0.32	0.51	4.39
1.14	0.70	0.96	0.61	0.61	0.86	0.88	0.76	0.77	0.11	0.85	7.11
0.90	0.35	0.58	0.15	0.37	0.00	0.00	0.00	0.00	0.00	0.34	1.79
1.08	0.70	0.96	0.61	0.37	0.69	0.70	0.76	0.77	0.11	0.34	6.01
1.06	0.35	0.77	0.31	0.00	0.00	0.18	0.00	0.00	0.11	0.34	2.05
1.00	0.53	0.77	0.15	0.00	0.34	0.53	0.76	0.46	0.22	0.34	4.09
1.08	0.88	0.96	0.76	0.00	0.86	0.88	0.76	0.77	0.00	0.85	6.72
1.00	0.53	0.58	0.31	0.24	0.69	0.88	0.76	0.46	0.32	0.51	5.27
1.10	0.70	0.96	0.76	0.37	0.86	0.70	0.61	0.46	0.43	0.68	6.53
	PF 1.00 0.87 1.02 1.02 1.01 0.82 0.78 1.06 1.14 1.13 0.59 0.79 1.07 0.82 0.88 1.00 0.67 1.00 1.08 0.88 0.96 1.00 0.89 0.92 0.96 1.08 0.60 0.92 0.96 1.08 0.60 0.92 0.96 1.08 0.60 0.92 1.12 0.91 1.14 0.90 1.08 1.00 1.08 1.00 1.08 1.00 1.08 1.00 1.00 </td <td>PF SIP 1.00 0.70 0.87 0.70 1.02 0.35 1.02 0.88 1.01 0.35 0.82 0.88 0.78 0.53 1.06 0.53 1.14 0.88 1.13 0.70 0.59 0.35 0.79 0.53 1.07 0.35 0.82 0.35 0.79 0.53 1.07 0.35 0.82 0.35 0.79 0.53 1.00 0.35 0.67 0.35 1.00 0.35 1.00 0.35 0.96 0.35 0.96 0.35 0.96 0.35 0.96 0.35 0.96 0.35 0.96 0.35 0.96 0.35 0.96 0.35 0.96 0.35 0.96</td> <td>PF SIP WSW 1.00 0.70 0.58 0.87 0.70 0.77 1.02 0.35 0.58 1.02 0.88 0.58 1.01 0.35 0.58 0.82 0.88 0.77 0.78 0.53 0.58 1.06 0.53 0.77 1.14 0.88 0.58 1.07 0.53 0.58 0.79 0.53 0.58 1.07 0.35 0.77 0.82 0.35 0.38 0.79 0.53 0.58 1.07 0.35 0.77 0.82 0.35 0.38 0.79 0.53 0.78 1.00 0.35 0.96 1.00 0.35 0.77 1.08 0.35 0.38 0.80 0.35 0.58 1.00 0.35 0.58 0.96 0.35 0.58</td> <td>PF SIP WSW US 1.00 0.70 0.58 0.31 0.87 0.70 0.77 0.46 1.02 0.35 0.58 0.31 1.02 0.88 0.58 0.61 1.01 0.35 0.58 0.15 0.82 0.88 0.77 0.31 0.78 0.53 0.58 0.46 1.06 0.53 0.77 0.31 1.14 0.88 0.58 0.31 1.13 0.70 0.58 0.61 0.59 0.35 0.38 0.31 1.07 0.35 0.77 0.31 0.82 0.35 0.38 0.31 1.07 0.35 0.77 0.31 0.82 0.35 0.77 0.31 1.08 0.35 0.77 0.31 0.82 0.35 0.77 0.46 0.67 0.35 0.77 0.46</td> <td>PF SIP WSW US DP 1.00 0.70 0.58 0.31 0.24 0.87 0.70 0.77 0.46 0.37 1.02 0.35 0.58 0.31 0.24 1.02 0.88 0.58 0.61 0.00 1.01 0.35 0.58 0.15 0.00 0.82 0.88 0.77 0.31 0.37 0.78 0.53 0.58 0.46 0.37 1.06 0.53 0.77 0.31 0.24 1.14 0.88 0.58 0.31 0.37 0.59 0.35 0.38 0.31 0.37 0.79 0.53 0.58 0.31 0.00 1.07 0.35 0.77 0.31 0.24 0.82 0.35 0.38 0.15 0.24 0.82 0.35 0.38 0.15 0.24 0.82 0.35 0.38 0.31 0.00<td>PF SIP WSW US DP CWP 1.00 0.70 0.58 0.31 0.24 0.69 0.87 0.70 0.77 0.46 0.37 0.86 1.02 0.35 0.58 0.31 0.24 0.69 1.02 0.88 0.58 0.61 0.00 0.86 1.01 0.35 0.58 0.15 0.00 0.86 0.82 0.88 0.77 0.31 0.37 0.86 0.78 0.53 0.58 0.46 0.37 0.86 1.06 0.53 0.77 0.31 0.24 0.69 1.14 0.88 0.58 0.31 0.37 0.86 1.07 0.35 0.77 0.31 0.24 0.69 1.07 0.35 0.77 0.31 0.24 0.69 0.82 0.35 0.96 0.46 0.00 0.86 0.67 0.35 0.96 0.4</td><td>PF SIP WSW US DP CWP BR 1.00 0.70 0.58 0.31 0.24 0.69 0.35 0.87 0.70 0.77 0.46 0.37 0.86 0.88 1.02 0.35 0.58 0.31 0.24 0.69 0.53 1.02 0.88 0.58 0.61 0.00 0.86 0.70 1.01 0.35 0.58 0.15 0.00 0.86 0.70 0.11 0.35 0.58 0.46 0.37 0.86 0.53 0.78 0.53 0.57 0.31 0.24 0.69 0.35 1.14 0.88 0.58 0.31 0.37 0.86 0.53 1.13 0.70 0.58 0.61 0.37 0.51 0.70 0.59 0.55 0.38 0.31 0.46 0.00 0.35 0.79 0.53 0.58 0.31 0.00 0.53</td><td>PF SIP WSW US DP CWP BR UA 1.00 0.70 0.58 0.31 0.24 0.69 0.35 0.76 0.87 0.70 0.77 0.46 0.37 0.86 0.88 0.76 1.02 0.35 0.58 0.51 0.00 0.86 0.70 0.76 1.01 0.35 0.58 0.15 0.00 0.86 0.88 0.46 0.82 0.88 0.77 0.31 0.37 0.86 0.53 0.76 1.06 0.53 0.77 0.31 0.24 0.69 0.35 0.46 1.14 0.88 0.58 0.31 0.37 0.51 0.70 0.76 0.59 0.35 0.38 0.31 0.37 0.51 0.70 0.61 1.13 0.70 0.58 0.31 0.00 0.53 0.76 0.59 0.35 0.36 0.31 0.00 <t< td=""><td>PF SIP WSW US DP CWP BR UA CT 1.00 0.70 0.58 0.31 0.24 0.69 0.35 0.76 0.00 0.87 0.70 0.77 0.46 0.37 0.86 0.88 0.76 0.46 1.02 0.35 0.58 0.51 0.00 0.86 0.70 0.76 0.46 1.01 0.35 0.58 0.61 0.00 0.86 0.70 0.76 0.46 0.82 0.88 0.77 0.31 0.37 0.86 0.53 0.76 0.46 0.78 0.53 0.77 0.31 0.24 0.69 0.35 0.46 0.77 1.14 0.88 0.58 0.31 0.37 0.86 0.53 0.76 0.46 0.59 0.35 0.38 0.31 0.37 0.69 0.53 0.61 0.37 0.51 0.70 0.61 0.31 <</td><td>PF SIP WSW US DP CWP BR UA CT MD 1.00 0.70 0.58 0.31 0.24 0.69 0.35 0.76 0.00 0.22 0.87 0.70 0.77 0.46 0.37 0.86 0.88 0.76 0.46 0.00 1.02 0.35 0.58 0.51 0.00 0.86 0.70 0.76 0.46 0.00 0.82 0.88 0.77 0.31 0.37 0.86 0.53 0.76 0.46 0.22 0.78 0.53 0.58 0.46 0.37 0.86 0.53 0.76 0.77 0.00 1.06 0.53 0.77 0.31 0.24 0.69 0.35 0.46 0.77 0.32 1.14 0.88 0.58 0.31 0.37 0.51 0.70 0.76 0.44 0.11 0.52 0.38 0.31 0.24 0.69 0.53</td><td>PF SIP WSW US DP CWP BR UA CT MD RR 1.00 0.70 0.58 0.31 0.24 0.69 0.35 0.76 0.00 0.22 0.34 0.87 0.70 0.77 0.46 0.37 0.86 0.88 0.76 0.46 0.00 0.85 1.02 0.88 0.58 0.61 0.00 0.86 0.70 0.76 0.46 0.00 0.53 1.01 0.35 0.58 0.15 0.00 0.86 0.70 0.76 0.15 0.22 0.68 0.82 0.88 0.77 0.31 0.37 0.86 0.53 0.76 0.46 0.00 0.85 1.13 0.70 0.58 0.61 0.37 0.86 0.53 0.76 0.46 0.11 0.51 0.59 0.35 0.38 0.31 0.37 0.56 0.53 0.76 0.46 0.11</td></t<></td></td>	PF SIP 1.00 0.70 0.87 0.70 1.02 0.35 1.02 0.88 1.01 0.35 0.82 0.88 0.78 0.53 1.06 0.53 1.14 0.88 1.13 0.70 0.59 0.35 0.79 0.53 1.07 0.35 0.82 0.35 0.79 0.53 1.07 0.35 0.82 0.35 0.79 0.53 1.00 0.35 0.67 0.35 1.00 0.35 1.00 0.35 0.96 0.35 0.96 0.35 0.96 0.35 0.96 0.35 0.96 0.35 0.96 0.35 0.96 0.35 0.96 0.35 0.96 0.35 0.96	PF SIP WSW 1.00 0.70 0.58 0.87 0.70 0.77 1.02 0.35 0.58 1.02 0.88 0.58 1.01 0.35 0.58 0.82 0.88 0.77 0.78 0.53 0.58 1.06 0.53 0.77 1.14 0.88 0.58 1.07 0.53 0.58 0.79 0.53 0.58 1.07 0.35 0.77 0.82 0.35 0.38 0.79 0.53 0.58 1.07 0.35 0.77 0.82 0.35 0.38 0.79 0.53 0.78 1.00 0.35 0.96 1.00 0.35 0.77 1.08 0.35 0.38 0.80 0.35 0.58 1.00 0.35 0.58 0.96 0.35 0.58	PF SIP WSW US 1.00 0.70 0.58 0.31 0.87 0.70 0.77 0.46 1.02 0.35 0.58 0.31 1.02 0.88 0.58 0.61 1.01 0.35 0.58 0.15 0.82 0.88 0.77 0.31 0.78 0.53 0.58 0.46 1.06 0.53 0.77 0.31 1.14 0.88 0.58 0.31 1.13 0.70 0.58 0.61 0.59 0.35 0.38 0.31 1.07 0.35 0.77 0.31 0.82 0.35 0.38 0.31 1.07 0.35 0.77 0.31 0.82 0.35 0.77 0.31 1.08 0.35 0.77 0.31 0.82 0.35 0.77 0.46 0.67 0.35 0.77 0.46	PF SIP WSW US DP 1.00 0.70 0.58 0.31 0.24 0.87 0.70 0.77 0.46 0.37 1.02 0.35 0.58 0.31 0.24 1.02 0.88 0.58 0.61 0.00 1.01 0.35 0.58 0.15 0.00 0.82 0.88 0.77 0.31 0.37 0.78 0.53 0.58 0.46 0.37 1.06 0.53 0.77 0.31 0.24 1.14 0.88 0.58 0.31 0.37 0.59 0.35 0.38 0.31 0.37 0.79 0.53 0.58 0.31 0.00 1.07 0.35 0.77 0.31 0.24 0.82 0.35 0.38 0.15 0.24 0.82 0.35 0.38 0.15 0.24 0.82 0.35 0.38 0.31 0.00 <td>PF SIP WSW US DP CWP 1.00 0.70 0.58 0.31 0.24 0.69 0.87 0.70 0.77 0.46 0.37 0.86 1.02 0.35 0.58 0.31 0.24 0.69 1.02 0.88 0.58 0.61 0.00 0.86 1.01 0.35 0.58 0.15 0.00 0.86 0.82 0.88 0.77 0.31 0.37 0.86 0.78 0.53 0.58 0.46 0.37 0.86 1.06 0.53 0.77 0.31 0.24 0.69 1.14 0.88 0.58 0.31 0.37 0.86 1.07 0.35 0.77 0.31 0.24 0.69 1.07 0.35 0.77 0.31 0.24 0.69 0.82 0.35 0.96 0.46 0.00 0.86 0.67 0.35 0.96 0.4</td> <td>PF SIP WSW US DP CWP BR 1.00 0.70 0.58 0.31 0.24 0.69 0.35 0.87 0.70 0.77 0.46 0.37 0.86 0.88 1.02 0.35 0.58 0.31 0.24 0.69 0.53 1.02 0.88 0.58 0.61 0.00 0.86 0.70 1.01 0.35 0.58 0.15 0.00 0.86 0.70 0.11 0.35 0.58 0.46 0.37 0.86 0.53 0.78 0.53 0.57 0.31 0.24 0.69 0.35 1.14 0.88 0.58 0.31 0.37 0.86 0.53 1.13 0.70 0.58 0.61 0.37 0.51 0.70 0.59 0.55 0.38 0.31 0.46 0.00 0.35 0.79 0.53 0.58 0.31 0.00 0.53</td> <td>PF SIP WSW US DP CWP BR UA 1.00 0.70 0.58 0.31 0.24 0.69 0.35 0.76 0.87 0.70 0.77 0.46 0.37 0.86 0.88 0.76 1.02 0.35 0.58 0.51 0.00 0.86 0.70 0.76 1.01 0.35 0.58 0.15 0.00 0.86 0.88 0.46 0.82 0.88 0.77 0.31 0.37 0.86 0.53 0.76 1.06 0.53 0.77 0.31 0.24 0.69 0.35 0.46 1.14 0.88 0.58 0.31 0.37 0.51 0.70 0.76 0.59 0.35 0.38 0.31 0.37 0.51 0.70 0.61 1.13 0.70 0.58 0.31 0.00 0.53 0.76 0.59 0.35 0.36 0.31 0.00 <t< td=""><td>PF SIP WSW US DP CWP BR UA CT 1.00 0.70 0.58 0.31 0.24 0.69 0.35 0.76 0.00 0.87 0.70 0.77 0.46 0.37 0.86 0.88 0.76 0.46 1.02 0.35 0.58 0.51 0.00 0.86 0.70 0.76 0.46 1.01 0.35 0.58 0.61 0.00 0.86 0.70 0.76 0.46 0.82 0.88 0.77 0.31 0.37 0.86 0.53 0.76 0.46 0.78 0.53 0.77 0.31 0.24 0.69 0.35 0.46 0.77 1.14 0.88 0.58 0.31 0.37 0.86 0.53 0.76 0.46 0.59 0.35 0.38 0.31 0.37 0.69 0.53 0.61 0.37 0.51 0.70 0.61 0.31 <</td><td>PF SIP WSW US DP CWP BR UA CT MD 1.00 0.70 0.58 0.31 0.24 0.69 0.35 0.76 0.00 0.22 0.87 0.70 0.77 0.46 0.37 0.86 0.88 0.76 0.46 0.00 1.02 0.35 0.58 0.51 0.00 0.86 0.70 0.76 0.46 0.00 0.82 0.88 0.77 0.31 0.37 0.86 0.53 0.76 0.46 0.22 0.78 0.53 0.58 0.46 0.37 0.86 0.53 0.76 0.77 0.00 1.06 0.53 0.77 0.31 0.24 0.69 0.35 0.46 0.77 0.32 1.14 0.88 0.58 0.31 0.37 0.51 0.70 0.76 0.44 0.11 0.52 0.38 0.31 0.24 0.69 0.53</td><td>PF SIP WSW US DP CWP BR UA CT MD RR 1.00 0.70 0.58 0.31 0.24 0.69 0.35 0.76 0.00 0.22 0.34 0.87 0.70 0.77 0.46 0.37 0.86 0.88 0.76 0.46 0.00 0.85 1.02 0.88 0.58 0.61 0.00 0.86 0.70 0.76 0.46 0.00 0.53 1.01 0.35 0.58 0.15 0.00 0.86 0.70 0.76 0.15 0.22 0.68 0.82 0.88 0.77 0.31 0.37 0.86 0.53 0.76 0.46 0.00 0.85 1.13 0.70 0.58 0.61 0.37 0.86 0.53 0.76 0.46 0.11 0.51 0.59 0.35 0.38 0.31 0.37 0.56 0.53 0.76 0.46 0.11</td></t<></td>	PF SIP WSW US DP CWP 1.00 0.70 0.58 0.31 0.24 0.69 0.87 0.70 0.77 0.46 0.37 0.86 1.02 0.35 0.58 0.31 0.24 0.69 1.02 0.88 0.58 0.61 0.00 0.86 1.01 0.35 0.58 0.15 0.00 0.86 0.82 0.88 0.77 0.31 0.37 0.86 0.78 0.53 0.58 0.46 0.37 0.86 1.06 0.53 0.77 0.31 0.24 0.69 1.14 0.88 0.58 0.31 0.37 0.86 1.07 0.35 0.77 0.31 0.24 0.69 1.07 0.35 0.77 0.31 0.24 0.69 0.82 0.35 0.96 0.46 0.00 0.86 0.67 0.35 0.96 0.4	PF SIP WSW US DP CWP BR 1.00 0.70 0.58 0.31 0.24 0.69 0.35 0.87 0.70 0.77 0.46 0.37 0.86 0.88 1.02 0.35 0.58 0.31 0.24 0.69 0.53 1.02 0.88 0.58 0.61 0.00 0.86 0.70 1.01 0.35 0.58 0.15 0.00 0.86 0.70 0.11 0.35 0.58 0.46 0.37 0.86 0.53 0.78 0.53 0.57 0.31 0.24 0.69 0.35 1.14 0.88 0.58 0.31 0.37 0.86 0.53 1.13 0.70 0.58 0.61 0.37 0.51 0.70 0.59 0.55 0.38 0.31 0.46 0.00 0.35 0.79 0.53 0.58 0.31 0.00 0.53	PF SIP WSW US DP CWP BR UA 1.00 0.70 0.58 0.31 0.24 0.69 0.35 0.76 0.87 0.70 0.77 0.46 0.37 0.86 0.88 0.76 1.02 0.35 0.58 0.51 0.00 0.86 0.70 0.76 1.01 0.35 0.58 0.15 0.00 0.86 0.88 0.46 0.82 0.88 0.77 0.31 0.37 0.86 0.53 0.76 1.06 0.53 0.77 0.31 0.24 0.69 0.35 0.46 1.14 0.88 0.58 0.31 0.37 0.51 0.70 0.76 0.59 0.35 0.38 0.31 0.37 0.51 0.70 0.61 1.13 0.70 0.58 0.31 0.00 0.53 0.76 0.59 0.35 0.36 0.31 0.00 <t< td=""><td>PF SIP WSW US DP CWP BR UA CT 1.00 0.70 0.58 0.31 0.24 0.69 0.35 0.76 0.00 0.87 0.70 0.77 0.46 0.37 0.86 0.88 0.76 0.46 1.02 0.35 0.58 0.51 0.00 0.86 0.70 0.76 0.46 1.01 0.35 0.58 0.61 0.00 0.86 0.70 0.76 0.46 0.82 0.88 0.77 0.31 0.37 0.86 0.53 0.76 0.46 0.78 0.53 0.77 0.31 0.24 0.69 0.35 0.46 0.77 1.14 0.88 0.58 0.31 0.37 0.86 0.53 0.76 0.46 0.59 0.35 0.38 0.31 0.37 0.69 0.53 0.61 0.37 0.51 0.70 0.61 0.31 <</td><td>PF SIP WSW US DP CWP BR UA CT MD 1.00 0.70 0.58 0.31 0.24 0.69 0.35 0.76 0.00 0.22 0.87 0.70 0.77 0.46 0.37 0.86 0.88 0.76 0.46 0.00 1.02 0.35 0.58 0.51 0.00 0.86 0.70 0.76 0.46 0.00 0.82 0.88 0.77 0.31 0.37 0.86 0.53 0.76 0.46 0.22 0.78 0.53 0.58 0.46 0.37 0.86 0.53 0.76 0.77 0.00 1.06 0.53 0.77 0.31 0.24 0.69 0.35 0.46 0.77 0.32 1.14 0.88 0.58 0.31 0.37 0.51 0.70 0.76 0.44 0.11 0.52 0.38 0.31 0.24 0.69 0.53</td><td>PF SIP WSW US DP CWP BR UA CT MD RR 1.00 0.70 0.58 0.31 0.24 0.69 0.35 0.76 0.00 0.22 0.34 0.87 0.70 0.77 0.46 0.37 0.86 0.88 0.76 0.46 0.00 0.85 1.02 0.88 0.58 0.61 0.00 0.86 0.70 0.76 0.46 0.00 0.53 1.01 0.35 0.58 0.15 0.00 0.86 0.70 0.76 0.15 0.22 0.68 0.82 0.88 0.77 0.31 0.37 0.86 0.53 0.76 0.46 0.00 0.85 1.13 0.70 0.58 0.61 0.37 0.86 0.53 0.76 0.46 0.11 0.51 0.59 0.35 0.38 0.31 0.37 0.56 0.53 0.76 0.46 0.11</td></t<>	PF SIP WSW US DP CWP BR UA CT 1.00 0.70 0.58 0.31 0.24 0.69 0.35 0.76 0.00 0.87 0.70 0.77 0.46 0.37 0.86 0.88 0.76 0.46 1.02 0.35 0.58 0.51 0.00 0.86 0.70 0.76 0.46 1.01 0.35 0.58 0.61 0.00 0.86 0.70 0.76 0.46 0.82 0.88 0.77 0.31 0.37 0.86 0.53 0.76 0.46 0.78 0.53 0.77 0.31 0.24 0.69 0.35 0.46 0.77 1.14 0.88 0.58 0.31 0.37 0.86 0.53 0.76 0.46 0.59 0.35 0.38 0.31 0.37 0.69 0.53 0.61 0.37 0.51 0.70 0.61 0.31 <	PF SIP WSW US DP CWP BR UA CT MD 1.00 0.70 0.58 0.31 0.24 0.69 0.35 0.76 0.00 0.22 0.87 0.70 0.77 0.46 0.37 0.86 0.88 0.76 0.46 0.00 1.02 0.35 0.58 0.51 0.00 0.86 0.70 0.76 0.46 0.00 0.82 0.88 0.77 0.31 0.37 0.86 0.53 0.76 0.46 0.22 0.78 0.53 0.58 0.46 0.37 0.86 0.53 0.76 0.77 0.00 1.06 0.53 0.77 0.31 0.24 0.69 0.35 0.46 0.77 0.32 1.14 0.88 0.58 0.31 0.37 0.51 0.70 0.76 0.44 0.11 0.52 0.38 0.31 0.24 0.69 0.53	PF SIP WSW US DP CWP BR UA CT MD RR 1.00 0.70 0.58 0.31 0.24 0.69 0.35 0.76 0.00 0.22 0.34 0.87 0.70 0.77 0.46 0.37 0.86 0.88 0.76 0.46 0.00 0.85 1.02 0.88 0.58 0.61 0.00 0.86 0.70 0.76 0.46 0.00 0.53 1.01 0.35 0.58 0.15 0.00 0.86 0.70 0.76 0.15 0.22 0.68 0.82 0.88 0.77 0.31 0.37 0.86 0.53 0.76 0.46 0.00 0.85 1.13 0.70 0.58 0.61 0.37 0.86 0.53 0.76 0.46 0.11 0.51 0.59 0.35 0.38 0.31 0.37 0.56 0.53 0.76 0.46 0.11

Legend: Respondents, PF= Productivity Factor, R=Respondents PPMP= Preconstruction Phase Management Practices Total Score, SIP= Short Interval Plan, WSW=Well-defined Scope of Work, US= Use of Software in Planning Work Packages, DP= Dedicated Planner, CWP= Construction Work Package, BR= Buildability Review, UA= Utilities Alignment, CT= Contract Types, MD= Model Development, RR=Regulatory Requirement.

A5: Human Resource Management Practices Scores

R	PF	CC	SA	ET	CD	NFIP	FIP	SoA	SOS	CDR	RPEP	EI	HRM
R1	1.00	0.46	0.44	0.45	0.28	0.39	0.40	0.27	0.16	0.17	0.15	0.11	3.30
R2	0.87	0.77	0.59	0.75	0.57	0.26	0.67	0.40	0.49	0.84	0.61	0.57	6.52
R3	1.02	0.31	0.44	0.45	0.57	0.26	0.54	0.54	0.33	0.17	0.31	0.57	4.47
R4	1.02	0.77	0.74	0.75	0.71	0.26	0.54	0.54	0.49	0.84	0.76	0.57	6.96
R5	1.01	0.46	0.00	0.45	0.71	0.26	0.27	0.67	0.33	0.33	0.61	0.57	4.67
R6	0.82	0.77	0.74	0.15	0.71	0.65	0.67	0.54	0.16	0.50	0.61	0.23	5.73
R7	0.78	0.77	0.74	0.45	0.71	0.65	0.67	0.54	0.33	0.84	0.31	0.57	6.57
R8	1.06	0.46	0.74	0.45	0.42	0.52	0.13	0.40	0.65	0.84	0.31	0.57	5.50
R9	1.14	0.77	0.74	0.45	0.57	0.26	0.40	0.40	0.49	0.84	0.31	0.23	5.45
R10	1.13	0.77	0.44	0.45	0.42	0.65	0.67	0.54	0.33	0.67	0.61	0.57	6.13
R11	0.59	0.46	0.44	0.45	0.28	0.13	0.13	0.40	0.33	0.17	0.61	0.11	3.52
R12	0.79	0.46	0.44	0.45	0.42	0.39	0.40	0.54	0.49	0.50	0.61	0.34	5.06
R13	1.07	0.62	0.59	0.45	0.42	0.26	0.40	0.54	0.82	0.50	0.31	0.46	5.36
R14	0.82	0.46	0.30	0.30	0.28	0.26	0.00	0.13	0.33	0.84	0.76	0.57	4.23
R15	0.88	0.77	0.44	0.45	0.28	0.26	0.27	0.40	0.33	0.84	0.00	0.46	4.50
R16	1.00	0.46	0.44	0.45	0.00	0.00	0.00	0.40	0.16	0.67	0.00	0.00	2.59
R17	0.67	0.15	0.15	0.45	0.14	0.13	0.13	0.13	0.33	0.17	0.15	0.11	2.05
R18	1.00	0.62	0.74	0.45	0.57	0.00	0.54	0.40	0.33	0.67	0.00	0.00	4.30
R19	1.08	0.62	0.44	0.75	0.57	0.00	0.27	0.40	0.33	0.50	0.15	0.46	4.49
R20	0.88	0.46	0.30	0.45	0.28	0.39	0.00	0.40	0.33	0.17	0.15	0.46	3.39
R21	0.96	0.46	0.44	0.45	0.71	0.26	0.54	0.40	0.49	0.50	0.61	0.46	5.32
R22	1.00	0.77	0.74	0.75	0.71	0.39	0.40	0.54	0.82	0.67	0.61	0.34	6.74
R23	0.89	0.77	0.74	0.75	0.42	0.39	0.40	0.54	0.33	0.17	0.31	0.46	5.27
R24	0.92	0.31	0.44	0.45	0.28	0.00	0.00	0.40	0.16	0.17	0.15	0.11	2.48
R25	0.96	0.46	0.15	0.30	0.28	0.00	0.00	0.40	0.16	0.17	0.00	0.11	2.04
R26	0.76	0.46	0.44	0.45	0.57	0.39	0.27	0.54	0.16	0.33	0.15	0.11	3.88
R27	1.08	0.77	0.44	0.75	0.71	0.65	0.67	0.54	0.49	0.84	0.76	0.57	7.19
R28	0.60	0.31	0.44	0.00	0.00	0.00	0.00	0.00	0.00	0.17	0.00	0.00	0.92
R29	0.92	0.77	0.44	0.75	0.28	0.13	0.00	0.40	0.65	0.84	0.31	0.46	5.03
R30	1.12	0.46	0.44	0.45	0.28	0.00	0.00	0.40	0.33	0.84	0.61	0.00	3.81
R31	0.91	0.46	0.44	0.45	0.28	0.39	0.27	0.40	0.33	0.33	0.61	0.00	3.97
R32	1.14	0.46	0.59	0.45	0.57	0.39	0.54	0.54	0.49	0.50	0.15	0.46	5.14
R33	0.90	0.00	0.30	0.30	0.28	0.39	0.40	0.40	0.00	0.33	0.00	0.23	2.64
R34	1.08	0.46	0.44	0.30	0.57	0.39	0.27	0.67	0.49	0.33	0.15	0.46	4.54
R35	1.06	0.31	0.30	0.30	0.00	0.00	0.27	0.54	0.65	0.67	0.76	0.34	4.14
R36	1.00	0.46	0.44	0.45	0.42	0.13	0.40	0.40	0.82	0.84	0.76	0.46	5.59
R37	1.08	0.46	0.59	0.45	0.71	0.26	0.27	0.54	0.82	0.84	0.76	0.57	6.27
R38	1.00	0.46	0.44	0.45	0.28	0.39	0.40	0.54	0.49	0.67	0.31	0.46	4.89
R39	1.10	0.77	0.59	0.75	0.00	0.65	0.67	0.54	0.65	0.17	0.61	0.34	5.74

Legend: R= Respondents, PF= Productivity Factor, R=Respondents, HRM= Human Resource Management Practices Total Score, CC= Crew Composition, SA=Skill Assessment and Evaluation, ET= Employee Training, CD = Career Development, NFIP= Non-Financial Incentive Programs, FIP= Financial Incentive Programs, SoA= Social Activities, SOS= Stability of Organizational Structure, CDR= Clear Delegation of Responsibility, RPEP=Retention Plan for Experienced Personnel, EI= Exit Interviews

A6: Safety and Health Practices Scores

R	PF	HK	SHPo	SHP	SWMS	HA	SHT	TSM	SHP
R1	1.00	0.18	0.76	0.55	0.91	0.92	0.71	0.91	4.94
R2	0.87	0.89	0.95	0.55	0.91	0.92	0.35	0.73	5.31
R3	1.02	0.71	0.95	0.92	0.73	0.92	0.89	0.91	6.03
R4	1.02	0.89	0.76	0.55	0.55	0.92	0.71	0.91	5.29
R5	1.01	0.89	0.38	0.37	0.91	0.92	0.35	0.73	4.55
R6	0.82	0.89	0.95	0.92	0.91	0.92	0.89	0.91	6.39
R7	0.78	0.89	0.95	0.92	0.91	0.92	0.89	0.91	6.39
R8	1.06	0.89	0.76	0.92	0.91	0.92	0.35	0.73	5.49
R9	1.14	0.89	0.95	0.92	0.91	0.92	0.71	0.73	6.03
R10	1.13	0.71	0.57	0.74	0.91	0.92	0.71	0.73	5.29
R11	0.59	0.71	0.57	0.55	0.91	0.37	0.35	0.54	4.01
R12	0.79	0.89	0.95	0.74	0.73	0.92	0.89	0.91	6.02
R13	1.07	0.89	0.95	0.92	0.91	0.74	0.71	0.91	6.03
R14	0.82	0.35	0.19	0.37	0.55	0.37	0.18	0.73	2.73
R15	0.88	0.89	0.76	0.74	0.91	0.92	0.71	0.73	5.66
R16	1.00	0.89	0.76	0.92	0.91	0.92	0.35	0.73	5.49
R17	0.67	0.89	0.76	0.55	0.91	0.92	0.18	0.91	5.12
R18	1.00	0.89	0.57	0.92	0.91	0.92	0.71	0.73	5.65
R19	1.08	0.89	0.95	0.92	0.91	0.92	0.35	0.73	5.68
R20	0.88	0.53	0.38	0.74	0.91	0.92	0.35	0.54	4.39
R21	0.96	0.89	0.95	0.92	0.91	0.92	0.71	0.73	6.03
R22	1.00	0.89	0.95	0.00	0.91	0.92	0.89	0.73	5.29
R23	0.89	0.89	0.95	0.92	0.73	0.92	0.89	0.91	6.21
R24	0.92	0.53	0.76	0.55	0.73	0.92	0.35	0.18	4.03
R25	0.96	0.53	0.76	0.92	0.37	0.74	0.53	0.91	4.76
R26	0.76	0.35	0.57	0.74	0.73	0.74	0.35	0.91	4.39
R27	1.08	0.89	0.95	0.92	0.91	0.92	0.71	0.73	6.03
R28	0.60	0.89	0.38	0.74	0.73	0.37	0.35	0.54	4.00
R29	0.92	0.89	0.95	0.92	0.91	0.92	0.89	0.73	6.21
R30	1.12	0.89	0.95	0.92	0.91	0.92	0.71	0.54	5.85
R31	0.91	0.53	0.95	0.92	0.91	0.92	0.71	0.91	5.86
R32	1.14	0.89	0.95	0.92	0.91	0.92	0.71	0.73	6.03
R33	0.90	0.89	0.76	0.92	0.91	0.55	0.53	0.91	5.48
R34	1.08	0.89	0.95	0.92	0.91	0.92	0.53	0.73	5.85
R35	1.06	0.89	0.76	0.55	0.91	0.92	0.53	0.73	5.29
R36	1.00	0.89	0.95	0.92	0.91	0.92	0.89	0.73	6.21
R37	1.08	0.89	0.95	0.92	0.91	0.92	0.71	0.91	6.21
R38	1.00	0.53	0.76	0.74	0.91	0.92	0.53	0.73	5.12
R39	1.10	0.89	0.95	0.92	0.91	0.92	0.89	0.91	6.39

Legend: R= Respondents, PF= Productivity Factor, R=Respondents SHP= Safety and Health Practices Total Score, HK= Housekeeping, SHPo = Safety and Health Policy, SHP= Safety and Health Plan, SWMS = Safe Work Method Statement, HA= Hazard Analysis, SHT= Safety and Health Training, and TSM= Toolbox Safety Meeting.

APPENDIX 4: CORRELATION ANALYSIS RESULTS

Part -I

1 uni	-																				
		1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17	18	19	20
1	r	1.00																		<u> </u>	
	р																			<u> </u>	
2	r	0.73	1.00																	L	
	р	0.00																		L	
3	r	0.45	0.39	1.00																L	
	р	0.00	0.01																	L	
4	r	0.45	0.29	0.26	1.00															L	
	р	0.00	0.08	0.10																L	
5	r	0.43	0.49	0.42	0.42	1.00															
	р	0.01	0.00	0.01	0.01															L	
6	r	0.26	0.25	0.38	0.27	0.47	1.00													L	
	р	0.12	0.13	0.02	0.10	0.00														L	
7	r	0.23	0.28	0.33	0.32	0.31	0.33	1.00													
	р	0.17	0.08	0.04	0.05	0.06	0.04														
8	r	0.46	0.36	0.62	0.39	0.56	0.46	0.47	1.00												
	р	0.00	0.02	0.00	0.02	0.00	0.00	0.00													
9	r	0.42	0.29	0.64	0.31	0.61	0.47	0.33	0.76	1.00											
	р	0.01	0.07	0.00	0.06	0.00	0.00	0.04	0.00												
10	r	0.36	0.22	0.35	0.39	0.23	0.27	0.54	0.48	0.44	1.00										
	р	0.02	0.17	0.03	0.01	0.17	0.09	0.00	0.00	0.00											
11	r	0.45	0.56	0.46	0.24	0.50	0.32	0.40	0.38	0.46	0.41	1.00									
	р	0.00	0.00	0.00	0.15	0.00	0.05	0.01	0.02	0.00	0.01									L	
12	r	0.50	0.45	0.40	0.44	0.53	0.50	0.23	0.42	0.54	0.24	0.64	1.00								
	р	0.00	0.00	0.01	0.00	0.00	0.00	0.16	0.01	0.00	0.14	0.00									
13	r	0.43	0.31	0.49	0.24	0.28	0.50	0.22	0.30	0.39	0.31	0.53	0.56	1.00							
	р	0.01	0.05	0.00	0.14	0.08	0.00	0.18	0.06	0.01	0.06	0.00	0.00							L	
14	r	0.13	-0.03	0.21	0.36	0.24	0.19	0.23	0.04	0.19	-0.04	0.34	0.42	0.34	1.00						
15	r	0.19	0.31	0.43	0.08	0.32	0.15	0.21	0.47	0.42	0.40	0.37	0.20	0.44	0.03	1.00				L	
	р	0.26	0.06	0.01	0.63	0.05	0.35	0.20	0.00	0.01	0.01	0.02	0.22	0.01	0.87					L	
16	r	0.31	0.37	0.48	0.32	0.51	0.43	0.31	0.47	0.54	0.32	0.61	0.50	0.64	0.40	0.69	1.00			L	
	р	0.06	0.02	0.00	0.04	0.00	0.01	0.06	0.00	0.00	0.05	0.00	0.00	0.00	0.01	0.00					
17	r	0.31	0.22	0.47	0.51	0.47	0.49	0.29	0.31	0.53	0.54	0.50	0.53	0.61	0.53	0.44	0.66	1.00			
	р	0.05	0.17	0.00	0.00	0.00	0.00	0.07	0.05	0.00	0.00	0.00	0.00	0.00	0.00	0.01	0.00				
18	r	0.28	0.17	0.32	0.26	0.45	0.34	0.28	0.37	0.48	0.53	0.59	0.44	0.52	0.43	0.41	0.59	0.77	1.00	<u> </u>	
	р	0.08	0.30	0.05	0.11	0.00	0.03	0.08	0.02	0.00	0.00	0.00	0.00	0.00	0.01	0.01	0.00	0.00		<u> </u>	
19	r	0.47	0.45	0.34	0.31	0.41	0.15	0.24	0.46	0.33	0.41	0.43	0.21	0.18	0.11	0.41	0.35	0.42	0.42	1.00	
	р	0.00	0.00	0.03	0.05	0.01	0.37	0.13	0.00	0.04	0.01	0.01	0.21	0.26	0.51	0.01	0.03	0.01	0.01	 	
20	r	0.34	0.29	0.38	0.30	0.54	0.52	0.40	0.44	0.57	0.51	0.49	0.39	0.40	0.23	0.27	0.48	0.65	0.63	0.43	1.00
	р	0.04	0.07	0.02	0.06	0.00	0.00	0.01	0.01	0.00	0.00	0.00	0.01	0.01	0.16	0.10	0.00	0.00	0.00	0.01	

Part -II

		1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17	18	19	20
21	r	0.38	0.27	0.29	0.46	0.48	0.35	0.27	0.48	0.34	0.26	0.11	0.31	0.15	0.16	0.31	0.25	0.30	0.07	0.29	0.32
-	р	0.02	0.10	0.08	0.00	0.00	0.03	0.09	0.00	0.04	0.10	0.49	0.05	0.37	0.34	0.06	0.13	0.06	0.68	0.07	0.05
22	r	0.38	0.42	0.32	0.27	0.54	0.25	0.49	0.49	0.45	0.28	0.66	0.37	0.43	0.32	0.40	0.63	0.40	0.48	0.36	0.51
	р	0.02	0.01	0.05	0.10	0.00	0.12	0.00	0.00	0.00	0.08	0.00	0.02	0.01	0.04	0.01	0.00	0.01	0.00	0.02	0.00
23	r	0.29	0.20	0.11	0.17	0.20	0.38	0.47	0.08	0.14	0.25	0.50	0.35	0.36	0.47	0.00	0.38	0.44	0.40	0.21	0.42
	р	0.07	0.22	0.51	0.29	0.21	0.02	0.00	0.64	0.40	0.12	0.00	0.03	0.02	0.00	0.98	0.02	0.01	0.01	0.21	0.01
24	r	0.24	0.25	0.31	0.21	0.50	0.53	0.29	0.51	0.55	0.40	0.18	0.22	0.23	0.05	0.27	0.38	0.42	0.35	0.22	0.59
	р	0.14	0.13	0.06	0.21	0.00	0.00	0.08	0.00	0.00	0.01	0.28	0.18	0.15	0.74	0.10	0.02	0.01	0.03	0.18	0.00
25	r	0.38	0.32	0.67	0.29	0.62	0.45	0.25	0.60	0.67	0.22	0.26	0.31	0.24	0.12	0.25	0.43	0.43	0.34	0.34	0.47
	р	0.02	0.05	0.00	0.07	0.00	0.00	0.13	0.00	0.00	0.18	0.10	0.06	0.14	0.46	0.12	0.01	0.01	0.03	0.03	0.00
26	r	0.52	0.40	0.42	0.47	0.43	0.18	0.51	0.37	0.46	0.48	0.36	0.36	0.26	0.20	0.11	0.16	0.39	0.32	0.32	0.46
	р	0.00	0.01	0.01	0.00	0.01	0.26	0.00	0.02	0.00	0.00	0.02	0.02	0.12	0.22	0.49	0.34	0.01	0.05	0.05	0.00
27	r	0.56	0.50	0.71	0.36	0.49	0.47	0.28	0.68	0.61	0.51	0.42	0.47	0.54	0.13	0.55	0.54	0.57	0.49	0.47	0.38
	р	0.00	0.00	0.00	0.02	0.00	0.00	0.09	0.00	0.00	0.00	0.01	0.00	0.00	0.44	0.00	0.00	0.00	0.00	0.00	0.02
28	r	0.19	0.11	0.02	0.18	0.05	0.06	0.18	0.05	-0.03	0.19	-0.03	0.03	0.00	0.06	-0.01	0.09	0.14	-0.02	0.04	0.11
	р	0.25	0.50	0.90	0.27	0.75	0.70	0.27	0.77	0.87	0.24	0.87	0.85	0.99	0.71	0.94	0.58	0.41	0.92	0.83	0.52
29	r	0.44	0.47	0.40	0.27	0.53	0.37	0.25	0.30	0.52	0.23	0.47	0.45	0.36	0.32	0.21	0.38	0.57	0.47	0.41	0.52
	р	0.00	0.00	0.01	0.10	0.00	0.02	0.13	0.06	0.00	0.15	0.00	0.00	0.03	0.04	0.20	0.02	0.00	0.00	0.01	0.00
30	r	0.43	0.38	0.31	0.19	0.41	0.16	0.18	0.20	0.51	0.40	0.33	0.27	0.18	0.09	0.17	0.33	0.49	0.38	0.21	0.61
	р	0.01	0.02	0.05	0.25	0.01	0.34	0.26	0.22	0.00	0.01	0.04	0.09	0.26	0.58	0.30	0.04	0.00	0.02	0.21	0.00
31	r	0.41	0.52	0.32	0.29	0.40	0.26	0.46	0.34	0.49	0.47	0.41	0.40	0.21	0.12	0.39	0.43	0.45	0.37	0.32	0.58
	р	0.01	0.00	0.05	0.07	0.01	0.11	0.00	0.03	0.00	0.00	0.01	0.01	0.21	0.46	0.02	0.01	0.00	0.02	0.04	0.00
32	r	0.22	0.15	0.47	0.07	0.23	0.06	0.35	0.35	0.54	0.41	0.29	0.13	0.27	0.11	0.30	0.37	0.42	0.43	0.23	0.41
	р	0.17	0.37	0.00	0.66	0.16	0.69	0.03	0.03	0.00	0.01	0.07	0.44	0.09	0.52	0.06	0.02	0.01	0.01	0.17	0.01
33	r	0.29	0.18	0.47	0.25	0.37	0.36	0.31	0.47	0.55	0.38	0.33	0.42	0.16	0.29	0.02	0.12	0.33	0.36	0.22	0.45
	р	0.08	0.28	0.00	0.13	0.02	0.03	0.06	0.00	0.00	0.02	0.04	0.01	0.33	0.08	0.88	0.46	0.04	0.03	0.19	0.00
34	r	0.50	0.29	0.39	0.30	0.11	0.37	0.31	0.44	0.35	0.40	0.27	0.33	0.25	0.09	0.04	0.22	0.35	0.20	0.37	0.50
	р	0.00	0.07	0.01	0.06	0.52	0.02	0.06	0.01	0.03	0.01	0.09	0.04	0.12	0.57	0.81	0.18	0.03	0.22	0.02	0.00
35	r	0.41	0.32	0.49	0.35	0.32	0.34	0.57	0.39	0.48	0.38	0.41	0.44	0.23	0.37	0.18	0.24	0.38	0.18	0.28	0.46
	р	0.01	0.05	0.00	0.03	0.05	0.04	0.00	0.01	0.00	0.02	0.01	0.00	0.15	0.02	0.27	0.15	0.02	0.27	0.08	0.00
36	r	0.34	0.09	0.39	0.47	0.31	0.40	0.29	0.64	0.48	0.48	0.19	0.23	0.25	0.03	0.22	0.21	0.36	0.30	0.36	0.42
	р	0.03	0.60	0.01	0.00	0.05	0.01	0.08	0.00	0.00	0.00	0.26	0.15	0.12	0.84	0.17	0.19	0.03	0.07	0.02	0.01
37	r	0.23	0.21	0.42	0.47	0.37	0.20	0.12	0.34	0.39	0.47	0.43	0.37	0.24	0.17	0.45	0.45	0.56	0.53	0.46	0.43
	р	0.16	0.20	0.01	0.00	0.02	0.22	0.48	0.04	0.01	0.00	0.01	0.02	0.13	0.29	0.00	0.00	0.00	0.00	0.00	0.01
38	r	0.24	0.34	0.28	0.24	0.51	0.23	0.10	0.23	0.32	0.21	0.31	0.41	0.22	0.25	0.26	0.36	0.41	0.47	0.27	0.40
	р	0.14	0.03	0.09	0.14	0.00	0.17	0.54	0.17	0.04	0.20	0.05	0.01	0.17	0.12	0.11	0.02	0.01	0.00	0.10	0.01
39	r	0.17	0.19	0.36	0.21	0.26	0.11	-0.05	0.19	0.26	0.16	0.37	0.45	0.25	0.17	0.14	0.21	0.38	0.38	0.46	0.39
	р	0.29	0.24	0.02	0.19	0.11	0.52	0.77	0.26	0.11	0.33	0.02	0.00	0.13	0.30	0.40	0.19	0.02	0.02	0.00	0.02
40	r	0.38	0.26	0.60	0.24	0.25	0.27	0.15	0.50	0.51	0.51	0.47	0.36	0.40	0.01	0.28	0.23	0.37	0.45	0.31	0.32

Part -III

_																					
		1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17	18	19	20
41	r	0.32	2 0.35	0.14	0.22	0.38	0.24	0.18	0.17	0.26	0.19	0.36	0.26	0.23	0.17	0.37	0.37	0.30	0.32	0.42	0.23
	р	0.0	5 0.03	0.38	0.18	0.02	0.15	0.28	0.29	0.11	0.25	0.02	0.11	0.16	0.31	0.02	0.02	0.06	0.05	0.01	0.17
42	r	0.2	7 0.16	0.23	0.21	0.41	0.38	0.41	0.23	0.34	0.45	0.45	0.39	0.27	0.37	0.30	0.32	0.66	0.64	0.50	0.49
	р	0.09	9 0.33	0.16	0.19	0.01	0.02	0.01	0.16	0.04	0.00	0.00	0.01	0.10	0.02	0.06	0.05	0.00	0.00	0.00	0.00
43	r	0.33	3 0.18	-0.08	0.17	0.16	0.17	0.17	-0.01	0.09	0.20	0.19	0.24	0.25	0.32	0.04	0.00	0.34	0.28	0.14	0.15
	р	0.04	4 0.28	0.62	0.30	0.34	0.31	0.31	0.97	0.57	0.22	0.25	0.14	0.12	0.05	0.82	1.00	0.04	0.09	0.39	0.35
44	r	0.38	8 0.26	0.09	0.50	0.55	0.18	0.10	0.18	0.17	-0.10	0.17	0.33	0.08	0.30	-0.10	0.23	0.21	0.10	0.27	0.15
	р	0.02	2 0.11	0.57	0.00	0.00	0.29	0.55	0.26	0.30	0.55	0.29	0.04	0.61	0.07	0.56	0.16	0.20	0.53	0.10	0.35
45	r	0.5	7 0.30	0.42	0.34	0.54	0.23	0.35	0.52	0.57	0.36	0.40	0.34	0.32	0.28	0.29	0.45	0.34	0.42	0.50	0.43
	р	0.00	0 0.07	0.01	0.03	0.00	0.16	0.03	0.00	0.00	0.02	0.01	0.04	0.05	0.09	0.07	0.00	0.03	0.01	0.00	0.01
46	r	-0.0	2 -0.12	-0.01	-0.17	-0.18	0.22	0.20	0.19	0.11	0.22	0.17	0.06	0.11	0.13	0.09	0.03	0.10	0.16	0.19	0.24
	р	0.89	9 0.48	0.95	0.29	0.28	0.18	0.22	0.25	0.49	0.18	0.29	0.73	0.51	0.42	0.57	0.84	0.55	0.34	0.25	0.14
47	r	0.23	3 0.14	0.23	0.21	0.32	0.38	0.24	0.19	0.43	0.47	0.31	0.34	0.29	0.27	0.35	0.32	0.63	0.41	0.39	0.52
Dart	IV	7																			
1 411	-1 V	21	22	22	24	25	26	27	20	20	20	21	20	22	24	25	26	27	20	20	40
41		21	1 0.20	23	24	25	26	2/	28	29	30	31	32	33	34	35	36	3/	38	39	40
41	Г	0.3	0.29	0.16	0.25	0.24	0.03	0.28	-0.11	0.14	0.29	0.39	0.21	0.28	-0.07	0.20	0.21	0.42	0.39	0.11	0.15
40	р	0.03	$\frac{5}{0.07}$	0.34	0.12	0.15	0.87	0.08	0.50	0.38	0.07	0.01	0.19	0.08	0.68	0.11	0.20	0.01	0.01	0.51	0.37
42	Г	0.25	9 0.33	0.44	0.34	0.27	0.37	0.37	0.21	0.45	0.30	0.47	0.38	0.38	0.21	0.43	0.31	0.50	0.23	0.28	0.10
42	р	0.0	/ 0.04	0.01	0.03	0.10	0.02	0.02	0.19	0.00	0.02	0.00	0.02	0.02	0.20	0.01	0.05	0.00	0.15	0.08	0.34
43	r	0.34	4 0.16	0.36	0.18	-0.14	0.17	0.21	0.37	0.23	0.15	0.31	0.00	0.15	0.11	0.19	0.10	0.10	0.11	-0.17	-
		0.02	0.22	0.02	0.27	0.41	0.21	0.20	0.02	0.16	0.26	0.00	0.00	0.27	0.51	0.25	0.55	0.52	0.40	0.20	0.03
4.4	р	0.03	5 0.32	0.02	0.27	0.41	0.31	0.20	0.02	0.10	0.30	0.06	0.99	0.37	0.51	0.25	0.55	0.55	0.49	0.29	0.85
44	r	0.3	<u>5 0.20</u>	0.22	0.02	0.37	0.20	0.12	0.19	0.19	0.20	0.11	0.14	0.06	0.18	0.21	0.04	0.33	0.27	0.06	0.02
4.5	р	0.03	3 0.22	0.17	0.88	0.02	0.23	0.48	0.25	0.25	0.23	0.50	0.40	0.70	0.27	0.19	0.82	0.04	0.10	0.72	0.89
45	Г	0.2	/ 0.44	0.16	0.23	0.41	0.49	0.40	0.33	0.39	0.31	0.30	0.40	0.28	0.19	0.33	0.28	0.30	0.27	0.20	0.24
16	р	0.10	0.00	0.34	0.15	0.01	0.00	0.01	0.04	0.01	0.00	0.00	0.00	0.08	0.25	0.04	0.08	0.02	0.09	0.23	0.14
40	Г	0.12	2 0.18	0.23	0.32	-0.07	0.00	0.01	0.17	-0.11	0.11	0.17	0.04	0.19	0.38	0.40	0.30	-0.01	-0.15	0.05	0.08
47	р	0.4	7 0.26	0.15	0.05	0.09	1.00	0.96	0.30	0.52	0.52	0.29	0.83	0.25	0.02	0.01	0.00	0.95	0.35	0.78	0.05
4/	I	0.22	2 0.22	0.33	0.32	0.23	0.40	0.27	0.35	0.30	0.49	0.51	0.29	0.33	0.30	0.58	0.41	0.37	0.19	0.29	0.13
Part	- V																				
_			41	42		43	44	45	5	46	47										
4	1	r	1.00																		
		р																			
42	2	r	0.46	1.00																	
		р	0.00																		
43	3	r	0.33	0.57	1	.00															
		р	0.04	0.00																	
44	4	r	0.36	0.25	0	.22	1.00														
		р	0.02	0.12	0	.18															
45	5	r	0.31	0.45	0	.14	0.40	1.0	00		1										
		р	0.05	0.00	0	.38	0.01														
46	5	r	0.00	0.30	0	.24	-0.20	-0.0	02	1.00											

		р	0.98	0.06	0.	15	0.22	0.92													
47		r	0.30	0.66	0.	42	0.09	0.40)	0.40	1.00										
Part	-VI	÷				·		·		·											
	-	21	22	23	24	25	26	27	28	29	30	31	32	33	34	35	36	37	38	39	40
21	r	1.00		-		-			-	-					-						
	р																				
22	r	0.73	1.00																		
	р	0.00																			
23	r	0.45	0.39	1.00																	
	р	0.00	0.01																		
24	r	0.45	0.29	0.26	1.00																
	р	0.00	0.08	0.10																	
25	r	0.43	0.49	0.42	0.42	1.00															
	р	0.01	0.00	0.01	0.01																
26	r	0.26	0.25	0.38	0.27	0.47	1.00														
	р	0.12	0.13	0.02	0.10	0.00															
27	r	0.23	0.28	0.33	0.32	0.31	0.33	1.00													
	р	0.17	0.08	0.04	0.05	0.06	0.04	o 1 -	4.00												
28	r	0.46	0.36	0.62	0.39	0.56	0.46	0.47	1.00												
•	р	0.00	0.02	0.00	0.02	0.00	0.00	0.00	0.54	1.00	-	-				-					
29	r	0.42	0.29	0.64	0.31	0.61	0.47	0.33	0.76	1.00											
20	р	0.01	0.07	0.00	0.06	0.00	0.00	0.04	0.00	0.44	1.00									┟────┦	
30	T	0.30	0.22	0.35	0.39	0.23	0.27	0.54	0.48	0.44	1.00										
21	p	0.02	0.17	0.03	0.01	0.17	0.09	0.00	0.00	0.00	0.41	1.00									
51	I n	0.43	0.30	0.40	0.24	0.30	0.52	0.40	0.58	0.40	0.41	1.00								┟────┦	
32	p r	0.00	0.00	0.00	0.13	0.00	0.03	0.01	0.02	0.00	0.01	0.64	1.00								
52	n	0.00	0.45	0.40	0.44	0.00	0.00	0.25	0.42	0.04	0.24	0.04	1.00								
33	r r	0.00	0.00	0.01	0.00	0.00	0.00	0.10	0.01	0.00	0.14	0.53	0.56	1.00						┢───┤	
55	n	0.43	0.05	0.49	0.14	0.28	0.00	0.22	0.06	0.01	0.06	0.00	0.00	1.00							
34	r	0.13	-0.03	0.21	0.36	0.24	0.19	0.23	0.04	0.19	-0.04	0.34	0.42	0.34	1.00						
	r	0.19	0.31	0.43	0.08	0.32	0.15	0.21	0.47	0.42	0.40	0.37	0.20	0.44	0.03	1.00					
35	р	0.26	0.06	0.01	0.63	0.05	0.35	0.20	0.00	0.01	0.01	0.02	0.22	0.01	0.87						
	r	0.31	0.37	0.48	0.32	0.51	0.43	0.31	0.47	0.54	0.32	0.61	0.50	0.64	0.40	0.69	1.00				
36	р	0.06	0.02	0.00	0.04	0.00	0.01	0.06	0.00	0.00	0.05	0.00	0.00	0.00	0.01	0.00					
	r	0.31	0.22	0.47	0.51	0.47	0.49	0.29	0.31	0.53	0.54	0.50	0.53	0.61	0.53	0.44	0.66	1.00			
37	р	0.05	0.17	0.00	0.00	0.00	0.00	0.07	0.05	0.00	0.00	0.00	0.00	0.00	0.00	0.01	0.00				
	r	0.28	0.17	0.32	0.26	0.45	0.34	0.28	0.37	0.48	0.53	0.59	0.44	0.52	0.43	0.41	0.59	0.77	1.00		
38	р	0.08	0.30	0.05	0.11	0.00	0.03	0.08	0.02	0.00	0.00	0.00	0.00	0.00	0.01	0.01	0.00	0.00			
	r	0.47	0.45	0.34	0.31	0.41	0.15	0.24	0.46	0.33	0.41	0.43	0.21	0.18	0.11	0.41	0.35	0.42	0.42	1.00	
39	р	0.00	0.00	0.03	0.05	0.01	0.37	0.13	0.00	0.04	0.01	0.01	0.21	0.26	0.51	0.01	0.03	0.01	0.01		
	r	0.34	0.29	0.38	0.30	0.54	0.52	0.40	0.44	0.57	0.51	0.49	0.39	0.40	0.23	0.27	0.48	0.65	0.63	0.43	1.00
40	р	0.04	0.07	0.02	0.06	0.00	0.00	0.01	0.01	0.00	0.00	0.00	0.01	0.01	0.16	0.10	0.00	0.00	0.00	0.01	

Legend 1= Integrated Schedule; 2=Work Schedule Strategies; 3=Schedule Control; 4= Dynamic Site Layout Plan; 5=Traffic Control Plan; 6=Site Security Plan; 7=Machinery Positioning Strategy; 8=Project start-up plan; 9=Project Completion Plan;10=Innovations and New Technologies; 11= Procurement Plans for Materials;12=Long -Lead Materials Identification; 13=Materials Status Database; 14=Materials Delivery Schedule; 15= Material Inspection Process; 16=Materials Inspection Team; 17=Procurement Plans for Construction Equipment; 18= Construction Equipment Productivity Analysis; 19=Construction Equipment Maintenance; 20=Short Interval Plan; 21=Well Defined Scope of Work; 22=Use of Software in Planning Work Packages;23=Dedicated Planner; 24=Construction Work Packages;25=Buildability Review; 26=Utilities Alignment; 27=Contract Types; 28=Model Development; 29=Regulatory Requirement; 30= Crew Composition; 31=Skill Assessmen;32=Employees Training;33=Career Development;34=Non-Financial Incentive Programs; 35=Financial Incentive Programs;36=Social Activities; 37= Stability of Organization Structure; 38=Clear Delegation of Responsibility;39=Retention Plan for Experienced Personnel; 40=Exit interviews; 41=Housekeeping;42=Health and Safety Policy;43=Health and Safety Plan;44=Safe work Method Statement;45=Hazards Analysis;46= Health and Safety Training; 47=Toolbox Safety Meetings.

APPENDIX 5: RAW DATA FOR LOGISTIC REGRESSION ANALYSES

A1: Raw Data for Logistic Regression Analysis of Construction Materials Management Practices and Productivity

Codes	No.	Respondents	Random	Productivity	Construction Materials
		_	Numbers	Factor (PF)	Management Practices
	1	R7	0.0083	0.78	3.49
	2	R34	0.0160	1.08	2.97
	3	R11	0.0363	0.59	1.53
	4	R32	0.0561	1.14	3.81
	5	R25	0.1277	0.96	1.08
Validation Dataset ¹¹	6	R24	0.1442	0.92	1.18
Model Building Dataset ¹²	7	R2	0.1600	0.88	3.77
Model Building Dataset ¹³	8	R16	0.2287	1.00	1.79
Model Building Builset	9	R4	0.2297	1.02	3.14
	10	R13	0.2753	1.07	1.84
	11	R3	0.2984	1.02	2.91
	12	R23	0.3212	0.89	2.66
	13	R6	0.3592	0.82	1.36
	1	R33	0.4091	0.90	1.57
	2	R18	0.4116	1.00	2.23
	3	R14	0.4137	0.82	1.97
	4	R12	0.4265	0.79	2.49
	5	R39	0.4361	1.10	3.61
M_{2} 1 1 D_{1} 1 1 D_{2} D_{2} (21)	6	R26	0.4496	0.76	1.32
Model Building Dataset ²²	7	R38	0.4546	1.00	1.53
Model Building Dataset ²³	8	R30	0.4590	1.12	3.91
Woder Building Dataset	9	R9	0.6022	1.14	2.98
	10	R27	0.6119	1.08	1.92
	11	R37	0.6255	1.08	3.92
	12	R29	0.6372	0.92	2.34
	13	R35	0.7068	1.06	1.68
	1	R8	0.7331	1.06	2.99
	2	R31	0.7343	0.91	2.77
21	3	R19	0.7451	1.08	1.45
Model Building Dataset ³¹	4	R5	0.7493	1.01	2.28
Model Building Dataset ³²	5	R10	0.8059	1.13	3.97
Validation Dataset ³³	6	R28	0.8183	0.60	1.66
	7	R20	0.8200	0.88	1.41
	8	R1	0.8350	1.00	1.88
	9	R15	0.8588	0.88	2.17
	10	R36	0.8672	1.00	2.74
	11	R17	0.9209	0.67	2.37
	12	R22	0.9288	1.00	2.44
	13	R21	0.9911	0.96	2.66

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

Codes	No.	Respondents	Random Numbers	Productivity Factor (PF)	Construction Materials Management Practices Scores
VD^{14}	1	R7	0.0083	0.78	3.49
MBD^{15}	2	R34	0.0160	1.08	2.97
MBD^{10}	3	R11	0.0363	0.59	1.53
MBD	4	R32	0.0561	1.14	3.81
	5	R25	0.1277	0.96	1.08
	6	R24	0.1442	0.92	1.18
	7	R2	0.1600	0.88	3.77
	8	R16	0.2287	1.00	1.79
	9	R4	0.2297	1.02	3.14
	10	R13	0.2753	1.07	1.84
	1	R3	0.2984	1.02	2.91
MBD^{24}	2	R23	0.3212	0.89	2.66
VD^{23} MDD ²⁶	3	R6	0.3592	0.82	1.36
MBD ²⁷	4	R33	0.4091	0.90	1.57
MIDD	5	R18	0.4116	1.00	2.23
	6	R14	0.4137	0.82	1.97
	7	R12	0.4265	0.79	2.49
	8	R39	0.4361	1.10	3.61
	9	R26	0.4496	0.76	1.32
	10	R38	0.4546	1.00	1.53
	1	R30	0.4590	1.12	3.91
MBD^{34}	2	R9	0.6022	1.14	2.98
MBD^{35} VD^{36}	3	R27	0.6119	1.08	1.92
MBD ³⁷	4	R37	0.6255	1.08	3.92
	5	R29	0.6372	0.92	2.34
	6	R35	0.7068	1.06	1.68
	7	R8	0.7331	1.06	2.99
	8	R31	0.7343	0.91	2.77
	9	R19	0.7451	1.08	1.45
	10	R5	0.7493	1.01	2.28
MBD44	1	R10	0.8059	1.13	3.97
MBD45	2	R28	0.8183	0.60	1.66
VD46 MBD47	3	R20	0.8200	0.88	1.41
WIDD4/	4	R1	0.8350	1.00	1.88
	5	R15	0.8588	0.88	2.17
	6	R36	0.8672	1.00	2.74
	7	R17	0.9209	0.67	2.37
	8	R22	0.9288	1.00	2.44
	9	R21	0 9911	0.96	2 66

b. 4-fold Cro	oss Validation
---------------	----------------

Legend: MBD= Model Building Dataset; VD=Validation Dataset

A2: Ra	aw Data for Logistic Reg	ression Analysis of Construction Equipment Management
Pı	ractices and Productivity	
я	3-fold Cross Validation	

Codes	No.	Respondents	Random Numbers	Productivity Factor (PF)	Construction Equipment Management Practices
	1	R7	0.0083	0.78	1.67
	2	R34	0.0160	1.08	1.08
	3	R11	0.0363	0.59	0.58
	4	R32	0.0561	1.14	1.69
	5	R25	0.1277	0.96	0.68
Validation Dataset ¹¹	6	R24	0.1442	0.92	0.58
Model Building Dataset ¹² Model Building Dataset ¹³	7	R2	0.1600	0.88	1.35
Model Dunung Dataset	8	R16	0.2287	1.00	0.59
	9	R4	0.2297	1.02	1.08
	10	R13	0.2753	1.07	0.97
	11	R3	0.2984	1.02	0.83
	12	R23	0.3212	0.89	1 37
	13	R6	0.3592	0.82	0.89
	1	R33	0.4091	0.90	0.42
	2	R18	0.4116	1.00	0.00
	3	R14	0.4137	0.82	0.44
	4	R12	0.4265	0.79	1.41
	5	R39	0.4361	1.10	1.15
Model Building Dataset ²¹	6	R26	0.4496	0.76	0.00
Validation Dataset ²²	7	R38	0.4546	1.00	0.70
Model Building Dataset	8	R30	0.4590	1.12	1 49
	9	R9	0.6022	1.14	1.23
	10	R27	0.6119	1.08	1.55
	11	R37	0.6255	1.08	2.09
	12	R29	0.6372	0.92	1 47
	13	R35	0.7068	1.06	0.59
	1	R8	0.7331	1.06	1.11
	2	R31	0.7343	0.91	0.96
Madal Duilding Dataset ³¹	3	R19	0.7451	1.08	0.56
Model Building Dataset ³²	4	R5	0.7493	1.01	0.58
Validation Dataset ³³	5	R10	0.8059	1.13	1 23
	6	R28	0.8183	0.60	0.16
	7	R20	0.8200	0.88	0.58
	8	R1	0.8350	1.00	0.58
	9	R15	0.8588	0.88	0.71
	10	R36	0.8672	1.00	1 39
	11	R17	0.9209	0.67	0.72
	12	R22	0.9288	1.00	1 20
	13	R21	0 9911	0.96	1.37
Codes	No.	Respondents	Random Numbers	Productivity Factor (PF)	Construction Equipment Management Practices Scores
---------------------------------	-----	-------------	-------------------	-----------------------------	---
	1	R7	0.0083	0.78	1.67
	2	R34	0.0160	1.08	1.08
VD^{14}	3	R11	0.0363	0.59	0.58
MBD^{13} MDD^{16}	4	R32	0.0561	1.14	1.69
MBD ¹⁷	5	R25	0.1277	0.96	0.68
	6	R24	0.1442	0.92	0.58
	7	R2	0.1600	0.88	1.35
	8	R16	0.2287	1.00	0.59
	9	R4	0.2297	1.02	1.08
	10	R13	0.2753	1.07	0.97
	1	R3	0.2984	1.02	0.83
	2	R23	0.3212	0.89	1.37
24	3	R6	0.3592	0.82	0.89
MBD^{24} VD^{25}	4	R33	0.4091	0.90	0.42
MBD^{26}	5	R18	0.4116	1.00	0.00
MBD ²⁷	6	R14	0.4137	0.82	0.44
	7	R12	0.4265	0.79	1.41
	8	R39	0.4361	1.10	1.15
	9	R26	0.4496	0.76	0.00
	10	R38	0.4546	1.00	0.70
	1	R30	0.4590	1.12	1.49
	2	R9	0.6022	1.14	1.23
34	3	R27	0.6119	1.08	1.55
MBD^{37} MBD ³⁵	4	R37	0.6255	1.08	2.09
VD^{36}	5	R29	0.6372	0.92	1.47
MBD ³⁷	6	R35	0.7068	1.06	0.59
	7	R8	0.7331	1.06	1.11
	8	R31	0.7343	0.91	0.96
	9	R19	0.7451	1.08	0.56
	10	R5	0.7493	1.01	0.58
	1	R10	0.8059	1.13	1.23
	2	R28	0.8183	0.60	0.16
	3	R20	0.8200	0.88	0.58
	4	R1	0.8350	1.00	0.58
	5	R15	0.8588	0.88	0.71
MBD44	6	R36	0.8672	1.00	1.39
MBD45 VD46	7	R17	0.9209	0.67	0.72
MBD47	8	R22	0.9288	1.00	1.39
	9	R21	0.9911	0.96	0.58

D. 4-1010 C1085 Valuation	b.	4-fold	Cross	Validation
---------------------------	----	--------	-------	------------

Legend: MBD= Model Building Dataset; VD=Validation Dataset

a. 3-fold Cross Validat	tion		-		
Codes	No.	Respondents	Random Numbers	Productivity Factor (PF)	Management Practices Related to Construction Methods Score
	1	R7	0.0083	0.78	6.78
	2	R34	0.0160	1.08	6.32
	3	R11	0.0363	0.59	3.76
	4	R32	0.0561	1.14	6.82
••••••••••••••••••••••••••••••••••••••	5	R25	0.1277	0.96	2.19
Validation Dataset ¹¹ Model Building Dataset ¹²	6	R24	0.1442	0.92	3.96
Model Building Dataset ¹³	7	R2	0.1600	0.88	6.80
C	8	R16	0.2287	1.00	5.42
	9	R4	0.2297	1.02	5.40
	10	R13	0.2753	1.07	4.15
	11	R3	0.2984	1.02	6.32
	12	R23	0.3212	0.89	4.36
	13	R6	0.3592	0.82	5.69
	1	R33	0.4091	0.90	2.55
	2	R18	0.4116	1.00	4.98
	3	R14	0.4137	0.82	3.06
	4	R12	0.4265	0.79	4 91
	5	R39	0.4361	1.10	6.46
Model Building Dataset ²¹	6	R26	0.4496	0.76	3.78
Model Building Dataset ²³	7	R38	0.4546	1.00	5.58
inouer Dunung Dunuoer	8	R30	0.4590	1.12	4 96
	9	R9	0.6022	1.14	4 93
	10	R27	0.6119	1.08	5.84
	11	R37	0.6255	1.08	6.24
	12	R29	0.6372	0.92	4.82
	13	R35	0.7068	1.06	3.15
	1	R8	0.7331	1.06	6 30
	2	R31	0.7343	0.91	3.88
Model Duilding Detect ³¹	3	R19	0.7451	1.08	3.62
Model Building Dataset ³²	4	R5	0.7493	1.01	5.02
Validation Dataset ³³	5	R10	0.8059	1.13	5.70
	6	R28	0.8183	0.60	J./9 2 1 A
	7	R20	0.8200	0.88	2.14
	8	R1	0.8350	1.00	4.07
	9	R15	0.8588	0.88	4.5/
	10	R36	0.8672	1.00	4.//
	11	R17	0.9209	0.67	4.90
	12	R22	0.9288	1.00	3.60
	13	R21	0.9200	0.96	5.97
	13	1121	0.7711	0.90	4.20

A3: Raw Data for Logistic Regression Analysis of Management Practices Related to Construction Methods and Productivity

Codes	No.	Respondents	Random Numbers	Productivity Factor (PF)	Management Practices Related to Construction Methods Score
	1	R7	0.0083	0.78	6.78
	2	R34	0.0160	1.08	6.32
VD^{14}	3	R11	0.0363	0.59	3.76
MBD^{13} MPD^{16}	4	R32	0.0561	1.14	6.82
MBD ¹⁷	5	R25	0.1277	0.96	2.19
	6	R24	0.1442	0.92	3.96
	7	R2	0.1600	0.88	6.80
	8	R16	0.2287	1.00	5.42
	9	R4	0.2297	1.02	5.40
	10	R13	0.2753	1.07	4.15
	1	R3	0.2984	1.02	6.32
	2	R23	0.3212	0.89	4.36
24	3	R6	0.3592	0.82	5.69
MBD^{24} VD^{25}	4	R33	0.4091	0.90	2.55
MBD^{26}	5	R18	0.4116	1.00	4.98
MBD ²⁷	6	R14	0.4137	0.82	3.06
	7	R12	0.4265	0.79	4.91
	8	R39	0.4361	1.10	6.46
	9	R26	0.4496	0.76	3.78
	10	R38	0.4546	1.00	5.58
	1	R30	0.4590	1.12	4.96
	2	R9	0.6022	1.14	4.93
24	3	R27	0.6119	1.08	5.84
MBD^{34} MDD^{35}	4	R37	0.6255	1.08	6.24
VD^{36}	5	R29	0.6372	0.92	4.82
MBD ³⁷	6	R35	0.7068	1.06	3.15
	7	R8	0.7331	1.06	6.30
	8	R31	0.7343	0.91	3.88
	9	R19	0.7451	1.08	3.62
	10	R5	0.7493	1.01	5.77
	1	R10	0.8059	1.13	5.79
	2	R28	0.8183	0.60	2.14
	3	R20	0.8200	0.88	4.07
	4	R1	0.8350	1.00	4.37
	5	R15	0.8588	0.88	4.77
MBD44	6	R36	0.8672	1.00	4.96
MBD45	7	R17	0.9209	0.67	3.60
VD46 MBD47	8	R22	0.9288	1.00	5.97
דענויו/	0	D 21	0 9911	0.96	4 20

b.	4-fold	Cross	Validation
<u> </u>	1 1010	C1000	, anaanon

Legend: MBD= Model Building Dataset; VD=Validation Dataset

Codes No. Respondents Random Numbers Productivity Factor (PF) Preconstruction Phase Management Practice Scores validation Dataset ¹¹ 1 R7 0.0083 0.78 5.69 validation Dataset ¹¹ 3 R11 0.0363 0.59 3.65 validation Dataset ¹¹ 6 R24 0.1442 0.92 3.85 Model Building Dataset ¹¹³ 7 R2 0.1600 0.88 6.10 9 R4 0.2287 1.00 5.15 9 R4 0.2287 1.02 5.36 10 R13 0.2753 1.07 4.40 11 R3 0.2984 1.02 4.90 12 R23 0.3212 0.89 4.61 13 R6 0.3592 0.82 5.68 Validation Dataset ²² 6 R3 0.4116 1.00 4.52 5.69 1 R33 0.4091 0.90 1.79 2.61 4.812 0.4546 1.00 5.57	a. 3-told Cross Valida	tion	D	ע ג		ים אין אין
Validation Dataset ¹¹ 1 R7 0.0083 0.78 5.69 Validation Dataset ¹¹ 3 R11 0.0363 0.59 3.651 Validation Dataset ¹¹ 5 R22 0.1561 1.14 7.11 5 R25 0.1277 0.96 2.80 Model Building Dataset ¹² 7 R2 0.1600 0.88 6.10 9 R4 0.2297 1.00 5.15 9 $R4$ 0.2297 1.02 5.36 10 R13 0.2753 1.07 4.40 4.90 12 R23 0.3212 0.89 4.61 13 R6 0.3592 0.82 5.68 1 R33 0.4091 0.90 1.79 2 R18 0.4116 1.00 4.50 3 R14 0.4137 0.82 2.61 4 R12 0.4265 0.79 4.50	Codes	No.	Respondents	Random Numbers	Productivity Factor (PF)	Preconstruction Phase Management Practices Scores
Validation Dataset ¹¹ 2 R34 0.0160 1.08 6.01 Validation Dataset ¹² 3 R11 0.0363 0.59 3.65 Model Building Dataset ¹² 6 R24 0.1277 0.96 2.80 Model Building Dataset ¹² 7 R2 0.1600 0.88 6.10 9 R4 0.2287 1.00 5.15 9 R4 0.2297 1.02 5.36 10 R13 0.2753 1.07 4.40 11 R33 0.2984 1.02 4.90 12 R23 0.3212 0.89 4.61 13 R6 0.3592 0.82 5.68 12 R26 0.4496 0.76 3.99 Validation Dataset ¹² 7 R38 0.4561 1.10 6.53 Validation Dataset ¹² 7 R38 0.4546 1.00 5.27 Model Building Dataset ²¹ 6 R26 0.4496 1.14		1	R7	0.0083	0.78	5.69
		2	R34	0.0160	1.08	6.01
4 R52 0.0561 1.14 7.11 Validation Dataset ¹¹ 5 R25 0.1277 0.96 2.80 Model Building Dataset ¹² 7 R2 0.1600 0.88 6.10 Model Building Dataset ¹³ 7 R2 0.1600 0.88 6.10 9 R4 0.2287 1.00 5.15 9 R4 0.2297 1.02 5.36 10 R13 0.2753 1.07 4.40 11 R3 0.2984 1.02 4.90 12 R23 0.3212 0.89 4.61 13 R6 0.3592 0.82 5.68 1 R33 0.4091 0.90 1.79 2 R18 0.4116 1.00 4.52 3 R14 0.4137 0.82 2.61 4 R12 0.4265 0.79 4.50 5 R39 0.4361 1.10 6.53 4 R12 0.4265 0.79 5.50 9 R9 <td></td> <td>3</td> <td>R11</td> <td>0.0363</td> <td>0.59</td> <td>3.65</td>		3	R11	0.0363	0.59	3.65
Validation Dataset ¹¹ 5 R25 0.1277 0.96 2.80 Model Building Dataset ¹² 7 R2 0.1600 0.88 6.10 Model Building Dataset ¹² 7 R2 0.1600 0.88 6.10 9 R4 0.2287 1.00 5.15 9 R4 0.2297 1.02 5.36 10 R13 0.2753 1.07 4.40 4.90 12 R23 0.3212 0.89 4.61 13 R6 0.3592 0.82 5.68 5.68 5.68 1 R33 0.4091 0.90 1.77 2.818 0.4116 1.00 4.52 3 R14 0.4137 0.82 2.61 4.812 4.50 Validation Dataset ²¹ 6 R26 0.4361 1.10 6.53 3.99 Validation Dataset ²¹ 7 R38 0.4546 1.00 5.27 Model Building Dataset ²¹ 7 R38 0.6022 1.14		4	R32	0.0561	1.14	7.11
Validation Dataset ¹¹ 6 R24 0.1442 0.92 3.85 Model Building Dataset ¹² 7 R2 0.1600 0.88 6.10 Model Building Dataset ¹² 7 R2 0.1600 0.88 6.10 Model Building Dataset ¹³ 8 R16 0.2287 1.00 5.15 9 R4 0.2297 1.02 5.36 10 R13 0.2753 1.07 4.40 11 R3 0.2984 1.02 4.90 12 R23 0.3212 0.89 4.61 12 R23 0.3212 0.89 4.61 13 R6 0.3592 0.82 5.68 13 R6 0.3592 0.82 2.61 4 R12 0.4265 0.79 4.50 5 R39 0.4361 1.10 6.53 6.72 7 R38 0.4546 1.00 5.27 Model Building Dataset ²¹ 7 R38 0.4546 1.00 5.50	11	5	R25	0.1277	0.96	2.80
Model Building Dataset 7 R2 0.1600 0.88 6.10 Model Building Dataset 8 R16 0.2287 1.00 5.15 9 R4 0.2297 1.02 5.36 10 R13 0.2753 1.07 4.40 11 R3 0.2984 1.02 4.90 12 R23 0.3212 0.89 4.61 13 R6 0.3592 0.82 5.68 1 R33 0.4091 0.90 1.79 2 R18 0.4116 1.00 4.52 3 R14 0.4137 0.82 2.61 4 R12 0.4265 0.79 4.50 5 R39 0.4361 1.10 6.53 4 R12 0.4265 0.79 4.50 Validation Dataset ²¹ 6 R26 0.4496 0.76 3.99 Validation Dataset ²² 7 R38 0.4546 1.00 <td< td=""><td>Validation Dataset¹¹</td><td>6</td><td>R24</td><td>0.1442</td><td>0.92</td><td>3.85</td></td<>	Validation Dataset ¹¹	6	R24	0.1442	0.92	3.85
Model Building Dataset 8 R16 0.2287 1.00 5.15 9 R4 0.2297 1.02 5.36 10 R13 0.2753 1.07 4.40 11 R3 0.2984 1.02 4.90 12 R23 0.3212 0.89 4.61 13 R6 0.3592 0.82 5.68 12 R18 0.4116 1.00 4.52 3 R14 0.4137 0.82 2.61 4 R12 0.4265 0.79 4.50 5 R39 0.4361 1.10 6.53 6 R26 0.4496 0.76 3.99 Vaidation Dataset ²¹ 6 R26 0.4496 1.00 5.27 8 R30 0.4590 1.12 5.50 9 9 R9 0.6022 1.14 5.46 10 R27 0.6119 1.08 5.85 11 R37 0.6255 1.08	Model Building Dataset ¹³	7	R2	0.1600	0.88	6.10
9 R4 0.2297 1.02 5.36 10 R13 0.2753 1.07 4.40 11 R3 0.2984 1.02 4.90 12 R23 0.3212 0.89 4.61 13 R6 0.3592 0.82 5.68 1 R33 0.4091 0.90 1.79 2 R18 0.4116 1.00 4.52 3 R14 0.4137 0.82 2.61 4 R12 0.4265 0.79 4.50 5 R39 0.4361 1.10 6.53 Validation Dataset ²³ 6 R26 0.4496 0.76 3.99 Model Building Dataset ²³ 6 R26 0.4496 0.76 3.99 Model Building Dataset ²³ 7 R38 0.4546 1.00 5.27 9 R9 0.6022 1.14 5.46 10 R27 0.6119 1.08 5.85	woder Dundning Dataset	8	R16	0.2287	1.00	5.15
10 R13 0.2753 1.07 4.40 11 R3 0.2984 1.02 4.90 12 R23 0.3212 0.89 4.61 13 R6 0.3592 0.82 5.68 12 R18 0.4091 0.90 1.79 2 R18 0.4116 1.00 4.52 3 R14 0.4137 0.82 2.61 4 R12 0.4265 0.79 4.50 5 R39 0.4361 1.10 6.53 Model Building Dataset ²¹ 6 R26 0.4496 0.76 3.99 Validation Dataset ²² 7 R38 0.4546 1.00 5.27 Model Building Dataset ²³ 8 R30 0.4590 1.12 5.50 9 R9 0.6022 1.14 5.46 10 R27 0.6119 1.08 5.85 11 R37 0.6255 1.08 6.72 <tr< td=""><td></td><td>9</td><td>R4</td><td>0.2297</td><td>1.02</td><td>5.36</td></tr<>		9	R4	0.2297	1.02	5.36
Model Building Dataset ²¹ 11 R3 0.2984 1.02 4.90 12 R23 0.3212 0.89 4.61 13 R6 0.3592 0.82 5.68 1 R33 0.4091 0.90 1.79 2 R18 0.4116 1.00 4.52 3 R14 0.4137 0.82 2.61 4 R12 0.4265 0.79 4.50 5 R39 0.4361 1.10 6.53 9 R9 0.4361 1.00 5.27 Model Building Dataset ²¹ 6 R26 0.4496 0.76 3.99 Validation Dataset ²² 7 R38 0.4546 1.00 5.27 9 R9 0.6022 1.14 5.46 10 R27 0.6119 1.08 5.85 11 R37 0.6255 1.08 6.72 12 R29 0.6372 0.92 5.25		10	R13	0.2753	1.07	4.40
12 R23 0.3212 0.89 4.61 13 R6 0.3592 0.82 5.68 1 R33 0.4091 0.90 1.79 2 R18 0.4116 1.00 4.52 3 R14 0.4137 0.82 2.61 4 R12 0.4265 0.79 4.50 5 R39 0.4361 1.10 6.53 4 R12 0.4265 0.79 4.50 5 R39 0.4361 1.10 6.53 6 R26 0.4496 0.76 3.99 Validation Dataset ²² 7 R38 0.4546 1.00 5.27 8 R30 0.4590 1.12 5.50 9 9 R9 0.6022 1.14 5.46 10 R27 0.6119 1.08 5.85 11 R37 0.6255 1.08 6.72 12 R29 0.6372 0.92 5.25 <td></td> <td>11</td> <td>R3</td> <td>0.2984</td> <td>1.02</td> <td>4.90</td>		11	R3	0.2984	1.02	4.90
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		12	R23	0.3212	0.89	4.61
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		13	R6	0.3592	0.82	5.68
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		1	R33	0.4091	0.90	1.79
		2	R18	0.4116	1.00	4.52
		3	R14	0.4137	0.82	2.61
		4	R12	0.4265	0.79	4.50
Model Building Dataset216R26 0.4496 0.76 3.99 Validation Dataset227R38 0.4546 1.00 5.27 Model Building Dataset238R30 0.4590 1.12 5.50 9R9 0.6022 1.14 5.46 10R27 0.6119 1.08 5.85 11R37 0.6255 1.08 6.72 12R29 0.6372 0.92 5.25 13R35 0.7068 1.06 2.05 13R35 0.7068 1.06 4.94 2R31 0.7343 0.91 4.39 Model Building Dataset313R19 0.7451 1.08 3.00 Model Building Dataset316R28 0.8183 0.60 1.19 7R20 0.8059 1.13 5.31 6R28 0.8183 0.60 1.19 7R20 0.8200 0.88 2.77 8R1 0.8350 1.00 4.18 9R15 0.8588 0.88 5.03 10R36 0.8672 1.00 4.09 11R17 0.9209 0.67 3.30 12R22 0.9288 1.00 5.33 13R21 0.9911 0.96 3.46		5	R39	0.4361	1.10	6.53
Validation Dataset 23 7 R38 0.4546 1.00 5.27 Model Building Dataset 23 8 R30 0.4590 1.12 5.50 9 R9 0.6022 1.14 5.46 10 R27 0.6119 1.08 5.85 11 R37 0.6255 1.08 6.72 12 R29 0.6372 0.92 5.25 13 R35 0.7068 1.06 2.05 13 R35 0.7068 1.06 4.94 2 R31 0.7343 0.91 4.39 Model Building Dataset 31 3 R19 0.7451 1.08 3.00 Model Building Dataset 32 4 R5 0.7493 1.01 4.07 Validation Dataset 33 5 R10 0.8059 1.13 5.31 6 R28 0.8183 0.60 1.19 7 R20 0.8200 0.88 2.77 8 R1 0.8350 1.00 4.08 9 R15 0.858	Model Building Dataset ²¹	6	R26	0.4496	0.76	3.99
Model Building Dataset8R30 0.4590 1.12 5.50 9R9 0.6022 1.14 5.46 10R27 0.6119 1.08 5.85 11R37 0.6255 1.08 6.72 12R29 0.6372 0.92 5.25 13R35 0.7068 1.06 2.05 1R8 0.7331 1.06 4.94 2R31 0.7343 0.91 4.39 3R19 0.7451 1.08 3.00 Model Building Dataset ³¹ 4 R5 0.7493 1.01 4R5 0.7493 1.01 4.07 Validation Dataset ³³ 6 R28 0.8183 0.60 1.19 7R20 0.8200 0.88 2.77 8R1 0.8350 1.00 4.18 9R15 0.8588 0.88 5.03 10R36 0.8672 1.00 4.09 11R17 0.9209 0.67 3.30 12R22 0.9288 1.00 5.33 13R21 0.9911 0.96 3.46	Validation Dataset ²²	7	R38	0.4546	1.00	5.27
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Model Building Dataset ²³	8	R30	0.4590	1.12	5 50
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		9	R9	0.6022	1.14	5 46
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		10	R27	0.6119	1.08	5.85
12 $R29$ 0.6372 0.92 5.25 13 $R35$ 0.7068 1.06 2.05 1 $R8$ 0.7331 1.06 4.94 2 $R31$ 0.7343 0.91 4.39 2 $R31$ 0.7451 1.08 3.00 Model Building Dataset ³¹ 4 $R5$ 0.7493 1.01 4.07 5 $R10$ 0.8059 1.13 5.31 6 $R28$ 0.8183 0.60 1.19 7 $R20$ 0.8200 0.88 2.77 8 $R1$ 0.8350 1.00 4.18 9 $R15$ 0.8588 0.88 5.03 10 $R36$ 0.8672 1.00 4.09 11 $R17$ 0.9209 0.67 3.30 12 $R22$ 0.9288 1.00 5.33 13 $R21$ 0.9911 0.96 2.46		11	R37	0.6255	1.08	6.72
13 $R35$ 0.7068 1.06 2.05 13 $R35$ 0.7068 1.06 2.05 14 $R8$ 0.7331 1.06 4.94 2 $R31$ 0.7343 0.91 4.39 3 $R19$ 0.7451 1.08 3.00 Model Building Dataset ³² 4 $R5$ 0.7493 1.01 4 $R5$ 0.7493 1.01 4.07 Validation Dataset ³³ 5 $R10$ 0.8059 1.13 6 $R28$ 0.8183 0.60 1.19 7 $R20$ 0.8200 0.88 2.77 8 $R1$ 0.8350 1.00 4.18 9 $R15$ 0.8588 0.88 5.03 10 $R36$ 0.8672 1.00 4.09 11 $R17$ 0.9209 0.67 3.30 12 $R22$ 0.9288 1.00 5.33 13 $R21$ 0.9911 0.96 2.46		12	R29	0.6372	0.92	5 25
Model Building Dataset1R8 0.7331 1.06 4.94 2R31 0.7343 0.91 4.39 Model Building Dataset3R19 0.7451 1.08 3.00 Model Building Dataset4R5 0.7493 1.01 4.07 Validation Dataset5R10 0.8059 1.13 5.31 6R28 0.8183 0.60 1.19 7R20 0.8200 0.88 2.77 8R1 0.8350 1.00 4.18 9R15 0.8588 0.88 5.03 10R36 0.8672 1.00 4.09 11R17 0.9209 0.67 3.30 12R22 0.9288 1.00 5.33 13R21 0.9911 0.96 2.46		13	R35	0.7068	1.06	2.05
Model Building Dataset $ 2 $ R31 $ 0.7343 $ $ 0.91 $ $ 4.39 $ Model Building Dataset $ 32 $ $ 4 $ R5 $ 0.7451 $ $ 1.08 $ $ 3.00 $ Validation Dataset $ 32 $ $ 4 $ R5 $ 0.7493 $ $ 1.01 $ $ 4.07 $ Validation Dataset $ 33 $ $ 5 $ R10 $ 0.8059 $ $ 1.13 $ $ 5.31 $ $ 6 $ R28 $ 0.8183 $ $ 0.60 $ $ 1.19 $ $ 7 $ R20 $ 0.8200 $ $ 0.88 $ $ 2.77 $ $ 8 $ R1 $ 0.8350 $ $ 1.00 $ $ 4.18 $ $ 9 $ R15 $ 0.8588 $ $ 0.88 $ $ 5.03 $ $ 10 $ R36 $ 0.8672 $ $ 1.00 $ $ 4.09 $ $ 11 $ R17 $ 0.9209 $ $ 6.67 $ $ 3.30 $ $ 12 $ R22		1	R8	0.7331	1.06	4.94
Model Building Dataset 31 3R190.74511.083.00Model Building Dataset 32 4R50.74931.014.07Validation Dataset 33 5R100.80591.135.316R280.81830.601.197R200.82000.882.778R10.83501.004.189R150.85880.885.0310R360.86721.004.0911R170.92090.673.3012R220.92881.005.3313R210.99110.962.46		2	R31	0.7343	0.91	4.39
Woder Building Dataset4R5 0.7493 1.01 4.07 Model Building Dataset5R10 0.8059 1.13 5.31 Validation Dataset5R10 0.8059 1.13 5.31 6R28 0.8183 0.60 1.19 7R20 0.8200 0.88 2.77 8R1 0.8350 1.00 4.18 9R15 0.8588 0.88 5.03 10R36 0.8672 1.00 4.09 11R17 0.9209 0.67 3.30 12R22 0.9288 1.00 5.33 13R21 0.9911 0.96 2.46	Madal Duilding Dataget ³¹	3	R19	0.7451	1.08	3.00
Validation Dataset 33 5R100.80591.135.316R280.81830.601.197R200.82000.882.778R10.83501.004.189R150.85880.885.0310R360.86721.004.0911R170.92090.673.3012R220.92881.005.3313R210.99110.962.46	Model Building Dataset ³²	4	R5	0.7493	1.01	4.07
6 R28 0.8183 0.60 1.19 7 R20 0.8200 0.88 2.77 8 R1 0.8350 1.00 4.18 9 R15 0.8588 0.88 5.03 10 R36 0.8672 1.00 4.09 11 R17 0.9209 0.67 3.30 12 R22 0.9288 1.00 5.33 13 R21 0.9911 0.96 2.46	Validation Dataset ³³	5	R10	0.8059	1.13	5.31
7 R20 0.8200 0.88 2.77 8 R1 0.8350 1.00 4.18 9 R15 0.8588 0.88 5.03 10 R36 0.8672 1.00 4.09 11 R17 0.9209 0.67 3.30 12 R22 0.9288 1.00 5.33 13 R21 0.9911 0.96 2.46		6	R28	0.8183	0.60	1.19
8 R1 0.8350 1.00 4.18 9 R15 0.8588 0.88 5.03 10 R36 0.8672 1.00 4.09 11 R17 0.9209 0.67 3.30 12 R22 0.9288 1.00 5.33 13 R21 0.9911 0.96 2.46		7	R20	0.8200	0.88	2.77
9 R15 0.8588 0.88 5.03 10 R36 0.8672 1.00 4.09 11 R17 0.9209 0.67 3.30 12 R22 0.9288 1.00 5.33 13 R21 0.9911 0.96 3.46		8	R1	0.8350	1.00	4 18
10 R36 0.8672 1.00 4.09 11 R17 0.9209 0.67 3.30 12 R22 0.9288 1.00 5.33 13 R21 0.9911 0.96 3.46		9	R15	0.8588	0.88	5.03
$\begin{array}{cccccccccccccccccccccccccccccccccccc$		10	R36	0.8672	1.00	4 09
12 R22 0.9288 1.00 5.33 13 R21 0.9911 0.96 2.46		11	R17	0.9209	0.67	3 30
13 R21 0.9911 0.96 2.46		12	R22	0.9288	1.00	5.20
		13	R21	0.9911	0.96	2.55

A4: Raw Data for Logistic Regression Models for Pre-Construction Phase Management Practices and Productivity

<u>b.</u> 4-fc	old Cr	oss Validation			
Codes	No.	Respondents	Random Numbers	Productivity Factor (PF)	Preconstruction Phase Management Practices Scores
	1	R7	0.0083	0.78	6.57
	2	R34	0.0160	1.08	4 54
VD^{14}	3	R11	0.0363	0.59	3 52
MBD ¹⁵	4	R32	0.0561	1 14	5.14
MBD^{16}	5	R25	0.1277	0.96	2 04
MBD	6	R24	0.1442	0.92	2.01
	7	R2	0.1600	0.88	6.52
	8	R16	0.2287	1.00	2 59
	9	R10 R4	0.2297	1.00	6.96
	10	R13	0.2257	1.02	5.36
	10	R15 R3	0.2735	1.07	4 47
	2	R23	0.3212	0.89	5.27
	2	R25	0.3592	0.82	5.27
MBD ²⁴	4	R33	0.3372	0.82	2 64
VD^{25}	5	R18	0.4091	1.00	4 30
MBD ²⁰ MBD ²⁷	6	R10 R14	0.4137	0.82	4.30
WIDD	7	R17	0.4265	0.32	5.06
	8	R12 R39	0.4265	1.10	5.00
	0	R35	0.4301	0.76	3.88
	10	R20 R38	0.4546	1.00	4 89
	10	R30	0.4590	1.00	3.81
	2	R9	0.6022	1.12	5.61
	3	R27	0.6119	1.08	7 19
MBD ³⁴	4	R37	0.6255	1.00	6.27
MBD^{35}	5	R29	0.6255	0.92	5.03
VD ³⁰ MBD ³⁷	6	R35	0.7068	1.06	4 14
WIDD	7	R35 R8	0.7331	1.00	5.50
	8	R31	0.7343	0.91	3.07
	9	R 19	0.7343	1.08	4 49
	10	R19 R5	0.7491	1.00	4.67
	10	R10	0.8059	1.01	613
	2	R10 R28	0.8183	0.60	0.92
	3	R20	0.8200	0.88	3 39
	4	R1	0.8350	1.00	3 30
	- - -	R15	0.8588	0.88	4 50
MBD44	6	R36	0.8672	1 00	5 59
MBD45	7	R17	0.9209	0.67	2.05
VD46	, 8	R22	0.9289	1.00	6.74
MBD47	9	R21	0.9911	0.96	5 32

Legend: MBD= Model Building Dataset; VD=Validation Dataset

382

Codes	No.	Respondents	Random Numbers	Productivity Factor (PF)	Human Resource Management Practices
	1	D7	0.0002	0.70	Scores
		R7	0.0083	0.78	6.57
	2	R34	0.0160	1.08	4.54
	3	RII	0.0363	0.59	3.52
	4	R32	0.0561	1.14	5.14
Validation Dataset ¹¹	5	R25	0.1277	0.96	2.04
Model Building Dataset ¹²	6	R24	0.1442	0.92	2.48
Model Building Dataset ¹³	7	R2	0.1600	0.88	6.52
	8	R16	0.2287	1.00	2.59
	9	R4	0.2297	1.02	6.96
	10	R13	0.2753	1.07	5.36
	11	R3	0.2984	1.02	4.47
	12	R23	0.3212	0.89	5.27
	13	R6	0.3592	0.82	5.73
	1	R33	0.4091	0.90	2.64
	2	R18	0.4116	1.00	4.30
	3	R14	0.4137	0.82	4.23
	4	R12	0.4265	0.79	5.06
	5	R39	0.4361	1.10	5.74
Model Building Dataset ²¹	6	R26	0.4496	0.76	3.88
Validation Dataset ²² Model Building Dataset ²³	7	R38	0.4546	1.00	4.89
Woder Dunding Dataset	8	R30	0.4590	1.12	3.81
	9	R9	0.6022	1.14	5 45
	10	R27	0.6119	1.08	7 19
	11	R37	0.6255	1.08	6.27
	12	R29	0.6372	0.92	5.03
	13	R35	0.7068	1.06	4 14
	1	R8	0.7331	1.06	5 50
	2	R31	0.7343	0.91	3.97
31	3	R19	0.7451	1.08	4 49
Model Building Dataset ³²	4	R5	0.7493	1.01	4.67
Validation Dataset ³³	5	R10	0.8059	1.13	6.13
	6	R28	0.8183	0.60	0.15
	7	R20	0.8200	0.88	2 20
	8	R1	0.8350	1.00	2.27
	9	R15	0.8588	0.88	5.50 4.50
	10	R36	0.8672	1.00	4.30
	11	R17	0.9209	0.67	5.59
	12	R22	0.9288	1.00	2.05
	12	R21	0.9200	0.96	6.74
	13	1121	0.7711	0.70	5 32

A5: Raw Data for Logistic Regression Analysis of Human Resource Management Practices and Productivity

Codes	No.	Respondents	Random Numbers	Productivity Factor (PF)	Human Resource Management Practices Scores
	1	R7	0.0083	0.78	6.57
	2	R34	0.0160	1.08	4.54
VD^{14}	3	R11	0.0363	0.59	3.52
MBD^{15}	4	R32	0.0561	1.14	5.14
MBD ¹⁷	5	R25	0.1277	0.96	2.04
MBB	6	R24	0.1442	0.92	2.48
	7	R2	0.1600	0.88	6.52
	8	R16	0.2287	1.00	2.59
	9	R4	0.2297	1.02	6.96
	10	R13	0.2753	1.07	5.36
	1	R3	0.2984	1.02	4.47
	2	R23	0.3212	0.89	5.27
24	3	R6	0.3592	0.82	5.73
MBD^{24} VD^{25}	4	R33	0.4091	0.90	2.64
MBD^{26}	5	R18	0.4116	1.00	4.30
MBD ²⁷	6	R14	0.4137	0.82	4.23
	7	R12	0.4265	0.79	5.06
	8	R39	0.4361	1.10	5.74
	9	R26	0.4496	0.76	3.88
	10	R38	0.4546	1.00	4.89
	1	R30	0.4590	1.12	3.81
	2	R9	0.6022	1.14	5.45
N (D D ³⁴	3	R27	0.6119	1.08	7.19
MBD^{34} MPD ³⁵	4	R37	0.6255	1.08	6.27
VD^{36}	5	R29	0.6372	0.92	5.03
MBD ³⁷	6	R35	0.7068	1.06	4.14
	7	R8	0.7331	1.06	5.50
	8	R31	0.7343	0.91	3.97
	9	R19	0.7451	1.08	4.49
	10	R5	0.7493	1.01	4.67
	1	R10	0.8059	1.13	6.57
	2	R28	0.8183	0.60	4.54
	3	R20	0.8200	0.88	3.52
	4	R1	0.8350	1.00	5.14
	5	R15	0.8588	0.88	2.04
MBD44	6	R36	0.8672	1.00	2.48
MBD45 VD46	7	R17	0.9209	0.67	6.52
MBD47	8	R22	0.9288	1.00	2.59
	9	R21	0.9911	0.96	6.96

b.	4-fold	Cross	Validation
----	--------	-------	------------

Legend: MBD= Model Building Dataset; VD=Validation Dataset

A6: Raw Data for Logistic Regression Analysis of Safety and Health Practices and Productivity

Codes	No.	Respondents	Random Numbers	Productivity Factor (PF)	Safety and Health Practices Scores
	1	R7	0.0083	0.78	6.39
	2	R34	0.0160	1.08	5.85
	3	R11	0.0363	0.59	4.01
	4	R32	0.0561	1.14	6.03
	5	R25	0.1277	0.96	4.76
Validation Dataset ¹¹	6	R24	0.1442	0.92	4.03
Model Building Dataset ¹² Model Building Dataset ¹³	7	R2	0.1600	0.88	5.31
Model Building Dataset	8	R16	0.2287	1.00	5.49
	9	R4	0.2297	1.02	5.29
	10	R13	0.2753	1.07	6.03
	11	R3	0.2984	1.02	6.03
	12	R23	0.3212	0.89	6.21
	13	R6	0.3592	0.82	6.39
	1	R33	0.4091	0.90	5.48
	2	R18	0.4116	1.00	5.65
	3	R14	0.4137	0.82	2.73
	4	R12	0.4265	0.79	6.02
	5	R39	0.4361	1.10	6.39
Model Building Dataset ²¹	6	R26	0.4496	0.76	4.39
Validation Dataset ²² Model Building Dataset ²³	7	R38	0.4546	1.00	5.12
Woder Dunding Dataset	8	R30	0.4590	1.12	5.85
	9	R9	0.6022	1.14	6.03
	10	R27	0.6119	1.08	6.03
	11	R37	0.6255	1.08	6.21
	12	R29	0.6372	0.92	6.21
	13	R35	0.7068	1.06	5.29
	1	R8	0.7331	1.06	5.49
	2	R31	0.7343	0.91	5.86
Model Building Dataset ³¹	3	R19	0.7451	1.08	5.68
Model Building Dataset ³²	4	R5	0.7493	1.01	4.55
Validation Dataset ³³	5	R10	0.8059	1.13	5.29
	6	R28	0.8183	0.60	4.00
	7	R20	0.8200	0.88	4.39
	8	R1	0.8350	1.00	4.94
	9	R15	0.8588	0.88	5.66
	10	R36	0.8672	1.00	6.21
	11	R17	0.9209	0.67	5.12
	12	R22	0.9288	1.00	5.29
	13	R21	0.9911	0.96	6.03

385

Codes	No.	Respondents	Random Numbers	Productivity Factor (PF)	Safety and Health Practices Scores
	1	R7	0.0083	0.78	6.39
	2	R34	0.0160	1.08	5.85
VD^{14}	3	R11	0.0363	0.59	4.01
MBD ¹⁵	4	R32	0.0561	1.14	6.03
MBD ¹⁰ MBD ¹⁷	5	R25	0.1277	0.96	4.76
MBD	6	R24	0.1442	0.92	4.03
	7	R2	0.1600	0.88	5.31
	8	R16	0.2287	1.00	5.49
	9	R4	0.2297	1.02	5.29
	10	R13	0.2753	1.07	6.03
	1	R3	0.2984	1.02	6.03
	2	R23	0.3212	0.89	6.21
24	3	R6	0.3592	0.82	6.39
MBD^{24} VD^{25}	4	R33	0.4091	0.90	5.48
MBD^{26}	5	R18	0.4116	1.00	5.65
MBD ²⁷	6	R14	0.4137	0.82	2.73
	7	R12	0.4265	0.79	6.02
	8	R39	0.4361	1.10	6.39
	9	R26	0.4496	0.76	4.39
	10	R38	0.4546	1.00	5.12
	1	R30	0.4590	1.12	5.85
	2	R9	0.6022	1.14	6.03
24	3	R27	0.6119	1.08	6.03
MBD^{34} MBD ³⁵	4	R37	0.6255	1.08	6.21
VD^{36}	5	R29	0.6372	0.92	6.21
MBD ³⁷	6	R35	0.7068	1.06	5.29
	7	R8	0.7331	1.06	5.49
	8	R31	0.7343	0.91	5.86
	9	R19	0.7451	1.08	5.68
	10	R5	0.7493	1.01	4.55
	1	R10	0.8059	1.13	5.29
	2	R28	0.8183	0.60	4.00
	3	R20	0.8200	0.88	4.39
	4	R1	0.8350	1.00	4.94
	5	R15	0.8588	0.88	5.66
	6	R36	0.8672	1.00	6.21
	7	R17	0.9209	0.67	5.12
	8	R22	0.9288	1.00	5.29
	9	R21	0.9911	0.96	6.03

b. 4-fold Cross Validation

Codes	No.	Respondents	Random Numbers	Productivity Factor (PF)	Aggregated Construction Management Practices
	1	D7	0.0083	0.78	20.50
	2	R34	0.0083	1.08	30.39 26.77
	3	R11	0.0100	0.59	20.77
	4	R11 R32	0.0561	1 14	17.00
	5	R25	0.0301	0.96	50.59 12.55
Validation Dataset ¹¹	6	R23	0.1277	0.90	15.55
Model Building Dataset ¹² Model Building Dataset ¹³	7	R24	0.1442	0.92	10.09
	8	R16	0.1000	1.00	29.80
	9	R10 R4	0.2207	1.00	21.03
	10	R13	0.2257	1.02	27.22
	10	R15 P3	0.2755	1.07	22.75
	11	R3 P22	0.2964	0.89	25.47
	12	R23	0.3212	0.89	24.48
	15	RO	0.5392	0.82	25.74
	1	R33	0.4091	0.90	14.44
	2	R10 D14	0.4110	1.00	21.69
	3	K14 D12	0.4157	0.82	15.05
	4	R12	0.4205	0.79	24.39
Model Building Dataset ²¹	5	R39	0.4301	1.10	29.88
Validation Dataset ²²	6 7	R26	0.4496	0.76	17.36
Model Building Dataset ²³	/	K38	0.4546	1.00	23.09
Woder Bundling Dataset	8	R30	0.4590	1.12	25.52
	9	R9	0.6022	1.14	26.09
	10	R27	0.6119	1.08	28.38
	11	R37	0.6255	1.08	31.45
	12	R29	0.6372	0.92	25.13
	13	R35	0.7068	1.06	16.90
	1	R8	0.7331	1.06	26.33
	2	R31	0.7343	0.91	21.82
Model Building Dataset ³¹	3	R19	0.7451	1.08	18.80
Model Building Dataset 32	4	R5	0.7493	1.01	21.91
Validation Dataset 33	5	R10	0.8059	1.13	27.72
	6	R28	0.8183	0.60	10.07
	7	R20	0.8200	0.88	16.60
	8	R1	0.8350	1.00	19.25
	9	R15	0.8588	0.88	22.84
	10	R36	0.8672	1.00	24.98
	11	R17	0.9209	0.67	17.16
	12	R22	0.9288	1.00	27.15
	13	R21	0.9911	0.96	22.25

A7: Raw Data for Logistic Regression Analysis of Aggregated Construction Management Practices and Productivity a. 3-fold Cross Validation

387

Codes	No.	Respondents	Random Numbers	Productivity Factor (PF)	Aggregated Construction Management Practices Scores
	1	R7	0.0083	0.78	30.59
	2	R34	0.0160	1.08	26.77
VD^{14}	3	R11	0.0363	0.59	17.06
MBD^{15}	4	R32	0.0561	1.14	30.59
MBD ¹⁰ MBD ¹⁷	5	R25	0.1277	0.96	13.55
MDD	6	R24	0.1442	0.92	16.09
	7	R2	0.1600	0.88	29.86
	8	R16	0.2287	1.00	21.03
	9	R4	0.2297	1.02	27.22
	10	R13	0.2753	1.07	22.75
	1	R3	0.2984	1.02	25.47
	2	R23	0.3212	0.89	24.48
24	3	R6	0.3592	0.82	25.74
MBD^{24} VD^{25}	4	R33	0.4091	0.90	14.44
MBD^{26}	5	R18	0.4116	1.00	21.69
MBD ²⁷	6	R14	0.4137	0.82	15.05
	7	R12	0.4265	0.79	24.39
	8	R39	0.4361	1.10	29.88
	9	R26	0.4496	0.76	17.36
	10	R38	0.4546	1.00	23.09
	1	R30	0.4590	1.12	25.52
	2	R9	0.6022	1.14	26.09
24	3	R27	0.6119	1.08	28.38
MBD^{34} MPD^{35}	4	R37	0.6255	1.08	31.45
VD^{36}	5	R29	0.6372	0.92	25.13
MBD ³⁷	6	R35	0.7068	1.06	16.90
	7	R8	0.7331	1.06	26.33
	8	R31	0.7343	0.91	21.82
	9	R19	0.7451	1.08	18.80
	10	R5	0.7493	1.01	21.91
	1	R10	0.8059	1.13	27.72
	2	R28	0.8183	0.60	10.07
	3	R20	0.8200	0.88	16.60
	4	R1	0.8350	1.00	19.25
	5	R15	0.8588	0.88	22.84
	6	R36	0.8672	1.00	24.98
	7	R17	0.9209	0.67	17.16
	8	R22	0.9288	1.00	27.15
	9	R21	0.9911	0.96	22.25

b. 4-fold Cross Validation

Company Code	Number of years in Industry	Company Size (No. of Employees)	Annual Turnover
R1	44.0	440.0	529.0
R2	76.0	-	-
R3	25.0	550.0	652.0
R4	19.0	200.0	612.0
R5	16.0	31.0	-
R6	68.0	210-500	449.0
R7	54.0	3750.0	2505.0
R8	54.0	3750.0	2505.0
R9	54.0	3750.0	2505.0
R10	21.0	51-200	199.0
R11	21.0	51-200	199.0
R12	18.0	501-1000	702.0
R13	58.0	>10000	2762.0
R14	98.0	501-1000	904.0
R15	12.0	51-200	-
R16	12.0	-	-
R17	41.0	5001-10000	541.0
R18	20.0	65.0	-
R19	20.0	65.0	-
R20	37.0	40.0	-
R21	44.0	440.0	529.0
R22	43.0	>250	470.0
R23	68.0	210-500	449.0
R24	-	-	-
R25	11.0	11-50	-
R26	10.0	-	-
R27	29.0	1001-5000	1098.0
R28	10.0	11-50	-
R29	29.0	1001-5000	1098.0
R30	66.0	201-500	246.0
R31	14.0	51-200	-
R32	29.0	1001-5000	1098.0
R33	104.0	1300.0	722.0
R34	58.0	>10000	2762.0
R35	43.0	>250	470.0
R36	32.0	1001-5000	-
R37	45.0	11-50	-
R38	43.0	>250	470.0
R39	61.0	51-200	196.0

APPENDIX 6: RAW DATA FOR COMPANY PROFILES

Notes: Codes for company size during analysis are: 1 for 0-19 employees, 2 for between 20-199 employees and 3 for greater than 200 employees.

APPENDIX 7: INTERVIEW RESULTS

Respondent 1	Respondent 2	Respondent 3	Respondent 4 and 5
Respondent 1 "The main contractor sends the monthly report to us (client representative). The program is prepared using MS project and percent complete reports for each floor is submitted by the main contractor. For example, Structural progress report is submitted as FRPL17P1 meaning 'Formwork', 'Reo,' 'Pour' for level seventeen pour number 1; we make monthly site visit. Company XXX is the best structural sub-contractor and it is leading the local construction industry in using different technologies such as jump formwork, slip formwork and similar technologies. The company uses kibbles having different capacities to pour concrete. It takes between 5-6 days to complete casting a floor including columns. Wind and rain are the main reasons for variation of the pouring days."	Respondent 2 "The most common building project procurement methods practiced in the context of Australia are: - i. Traditional Procurement • Lump sum • Measurement • Cost reimbursement ii. Design and Construct Procurement iii. Management Procurement"	 Respondent 3 "After the reinforcement has been placed and it is ready for casting, the structural engineer shall inspect and give his comments prior to pouring the concrete. The comments are provided to the main contractor and then forwarded with cover letter to the concreting subcontractor. Crane is used to cast concrete for columns using kibbles and pump is used for slab. The slab is post tensioned in both ways. The precast cast concrete subcontractor delivers the materials using just-in-time approach. Crane is owned by structural sub- contractor, and other subcontractors also use the crane when it is not used for concreting and related works. As concrete is on the critical path, priority is given for this item. For instance, the day after casting concrete, the crane is entirely used by the structural subcontractor as they will transport their panels and make ready for another pour. There is meeting every day regarding the usage of crane. More than 80% of the structure of our projects is subcontracted to the same contractor due to its capacity and good 	 Respondent 4 and 5 "1. Organization Structure of a Project There are more than twenty sub-contractors which execute various works like concrete, electrical, mechanical, glass, tiles etc., and the main contractor has 6 supervisors to oversee the activities of the subbies. The main contractor has also consultancy agreement with one structural consultant. 2. Organizational Structure of the Main Builder The construction manager usually sits at the head office and is also responsible for the management of other projects too. The project manager sits on site and he may or may not manage other sites. There is one site manager who is responsible for coordination of the works of the six supervisors under the project manager. 3. Organization for Concrete Work The concrete sub-contractor has its own site supervisor which is me and I am responsible for overseeing the tasks of different crews (bar benders, Concrete floater, Concrete Formwork Stripper others). I supervise all the concreting operations from formwork laying to casting concrete to stripping the formwork. I report to construction manager who is sitting at the head office. There is also supervisor from main contractor who is also responsible for crane scheduling, structure works, drop zone that is giving permission to sub-contractors working underneath and making sure that their working zone is free. He also manages main contractor's laborers that execute general works which are not sub-contracted such as housekeeping."
		contractor due to its capacity and good working relationship."	which are not sub-contracted such as housekeeping.

			Respondent 6		
Construction Management Practices	Summary	Conclusion	Construction Management Practices	Summary	Conclusion
Materials Delivery Schedule	"Material delivery schedule has an effect on productivity and we really manage it so much."	Suitable	Retention Plan for Experienced Personnel	"It is through promotion. They promote their personnel. That does not necessarily have to people who have been there for long time. It is people who they believe fit for the role. If you are not efficient at your job, there will be less likely to promote	Suitable
Short Interval Planning	"we have contract program, works program, target programs and two weekly programs."	Suitable	Exit interviews	"There is a process for them. But I don't think there are very well used"	Suitable
Well Defined Scope of Work	"In a contract, there is a section called scope of works which we line items of all the works they have to do. And then also within the contract, we can even nominate the site team that we want."	Suitable	Integrated Schedule	"Program and finance is integrated. Labour from subcontractor is integrated back into reporting financing and programming."	Suitable
Use of Software in Planning workpackages	"I am familiar with MS project, pretty standard."	Suitable	Work Schedule Strategies	"We just use the union calendar. It is 36 hours per week. The enterprise bargaining agreement stipulates how long we can work for. If you want to	Suitable
Dedicated Planner	"Absolutely yeah, they have a planning team. There is someone who is responsible for the works program and for example the two-year program for this project is produced by an in- house planner. But once it is given to the site, the site team plans the short term activities."	Suitable	Dynamic Site Layout Plan	"It always moves. The office will move at least once. The car park is usually used as an area for storage. That is planned for."	Suitable
Design Readiness for Construction	"There is a logistic planner, someone who reviews the design from a constructability point of view."	Suitable	Site Security Plan	"There is security system on weekends in particular. We employ security staff. It is not as such. I think, crime is not as big on building sites."	Suitable
Employees Technical Training	"Skilled trades like electricians and plumbers are the ones that have regulatory obligations that they have to do in the apprenticeship training."	Suitable	Machinery Positioning Strategy	"A good example is we looked quite closely at how big the precast panels were on our site project. So our panels were too big for our crane so the crane was not strong enough to up-lift the panels so a lot of panels got broken into two smaller ones so the crane can lift. A lot of work is done early on."	Suitable
Non-Financial Incentive Programs	"At a head contractor, they use that very heavily, non-financial benefits. There is a lot of variety in the work that I do. I find really interesting. I really enjoy the work that I do. It is not necessarily due to financial benefit."	Suitable	Innovations and New Technologies	"At the moment we are trying to use a new drawing review system where drawings are reviewed electronically rather than by a pen and paper."	Suitable
Financial Incentive Programs	"They provide financial incentives to motivate them not necessarily non-financial incentives."	Suitable	Formal Health and Safety Policy	"safe site is a happy site'. It is also far more productive site if people respect the work site"	Suitable
Social Activities	"My company has fitness days; they pay for people doing marathon things like that. They definitely use non-financial incentives"	Suitable	Clear Delegation of Responsibility	"People are given their roles; update the roles where there is lot of project. There is a clear structure that is in place. In my company, there is strong management processes."	Suitable
Maintain Stability of Organization Structure	"Preparing good organizational structures are considered good practices. Absolutely, yeah".	Suitable			

		Res	spondent 7		
Construction Management Practices	Summary	Conclusion	Construction Management Practices	Summary	Conclusion
Short Interval Planning	'Depending on which tier and on the frequency of statusing short interval planning is conducted. The very lowest denominator tier 3 may be statused each week or each fort night.'	Suitable	Contract types	'Most of the contracts are fixed so that no extra costs that will be incurred. There is scope of works that may have a set of drawings that they have attended to.'	Suitable
Well Defined Scope of Work	'That is mainly done by the contract administrators and package managers. Broadly speaking, facade have its own scope of work, structure has its own scope of works, finishing and fit out has several scope of works'	Suitable	Crew Composition	'When it comes to any sub-contract work, they are responsible for their own formation of their crews and composition; for us we organize our crew.'	Suitable
Use of Software in Planning	'We use software in planning normally. MS project, p6 and other programs that we need. MS Project is pretty standard everyone uses it.'	Suitable	Integrated Schedule	'With schedules/programs, there is a couple of tiers. You high light critical areas, detail out these critical areas and sign up the subcontractors to those programs.'	Suitable
Dedicated Planner	'I am dedicated planner. Yes, dedicated planner on site is needed because people sometimes do not look ahead as far as the project planner needs to look. So strategic direction needs to be applied and it is good to have someone there.'	Suitable	Work Schedule Strategies	'If you are in delay, you request extra time to work over time. We work six days a week from CFMEU RDO calendar, so that is normal construction hours for us.'	Suitable
Construction Work Packages	'From scope of works; each one will have a program or work packages.'	Suitable	Schedule Control	'You get subcontractors feedback for the works they are in charge of; depending on what phase of the job you are you give their percentage complete, remaining duration and you have to forecast out whether they are going to be finished.'	Suitable
Buildability Review	'We are going to review the design before construction. We have design manager/ a team of the people depending on the scale of the job you might have one person or several people. Obviously we have to design for safety as well like the way they do a balustrade detail and how we actually install has to be considered. apprenticeship training.'	Suitable	Dynamic Site Layout Plan	'These are implemented at stages; there is always a construction plan. When you get to certain level and depending on which job; from the start of the job you generally write up a construction plan and have a certain set off where you think your site shade is going to be. Yes, we do maneuver those elements; and that is all part of your project program.'	Suitable

		Re	spondent 8		
Construction Management Practices	Summary	Conclusion	Construction Management Practices	Summary	Conclusion
Procurement Plans for Materials	'For example, you want to procure a façade; you are going to have concept, you are going to have an indication of what you need to do, what is the performance criteria, what is the façade for, what does it looks like / what is architecture which is the design; then you look into the detail requirements of that façade/performance of that facade; then you need to understand how to deliver it or manage it, you break up the building in to packages.'	Suitable	Well- Defined Scope of Work	'You must define your scope. That is the biggest problem today. To me it is the biggest risk the industry face. It is our ability to communicate in detail and in plan the scopes; and the inputs of those scope to each trade; and dependency of those scope to each other.'	Suitable
Long-Lead Materials Identification	'Usually the long lead times are façade, services such as lifts, generators, boilers all these key primary elements have long lead times.'	Suitable	Dedicated Planner	'Oh yes. Planner is critical that he gets by his own experience, records, and referencing durations. Most companies will find that they have the durations to build. I can build a floor and finish a floor in X days. So he is plugging in that, the guys on site try to deliver as per that.'	Suitable
Materials Delivery Schedule	'It is about managing, identifying and procuring the critical path materials. And to ensure that is kept. Well, there are. The tendency here is the same people building the job are doing inspection. If you get your architect doing defect, you got sign off that the quality has been achieved.'	Suitable	Buildability Review	'Design reviews and checking even in here you need to do regularly. You always reviewing and checking and testing whether design is right. That is why people go for shop drawings, RFI's etc.'	Suitable
Material Inspection Process	'Well, there are. The tendency here is the same people building the job are doing inspection. If you get your architect doing defect, you got sign off that the quality has been achieved.'	Suitable	Contract types	'What people doing are giving them the drawings and allow for everything. So the contractor really tries to get the subcontractor take risk.'	Suitable
Construction Machinery Maintenance	'Under the contract, you provide notices for rectification order in subcontract. Occupational health and safety obligations usually address these issues, equipment must be maintained, serviced, signed off on regular basis. It is through subcontracting; most of all these issues are done by subcontractors.'	Suitable	Model Development	'It is fantastic the more you can get that because it helps to communicate. The more 3D models and the more cuttings and drawings coming in the better off we are because they help to explain.'	Suitable
Management Strategy	It is through subcontracting; most of all these issues are done by subcontractors.'	Not Suitable	Machinery Positioning Strategy		
Short Interval Planning	'It is absolutely important. The project should have a minimum of a week ahead to four weeks ahead to six weeks ahead to a year ahead. Your planning should be within a g week; you should always know what you will try to do next week. By addressing and checking back against the program, it will allow you to plan mitigation strategies. It will help you to identify what makes you sleep'	Suitable		'The crane is subject to each size. We need to have the five principles again (design, detail, sequence, program and cost). Design and detail for example relate to weight of crane. Weight and reach become fundamental not only in building but also in removing the crane.'	Suitable

		Res	spondent 9 and 10		
Construction Management Practices	Summary	Conclusion	Construction Management Practices	Summary	Conclusion
Procurement Plans for Materials	'We hire subcontractors to procure materials.'	Suitable	Non-Financial Incentive Programs	'Not really.'	Not Suitable
Long-Lead Materials Identification	'Lift and other lead time items are identified earlier.20 weeks for lift. Lift is always a long lead item. The items vary for site to site.'	Suitable	Financial Incentive Programs	'You are given a job and you will do. No incentive'	Suitable
Procurement Team	'We usually use supply and install contracting method'	Suitable	Social Activities	Soccer, dinner, beer on Friday after noon'	Suitable
On-Site Materials Tracking Technology	'We do not let materials on site for robbery. We use just in time technique.'	Not Suitable	Stability of Organization Structure	'The project team vary from project to project but generally anyone above PM will stay the same.'	Suitable
Materials Delivery Schedule	'We give the delivery dates for subcontractors. The Forman on site sit down with them and fix the schedule.'	Suitable	Clear Delegation of Responsibility	We do have all clear delegation of responsibility that also comes with our job descriptions.	Suitable
Material Inspection Process	'Forman on site makes sure that materials come to site as per the specification.'	Suitable	Retention Plan for Experienced Personnel Exit interviews	'No retention plan.'	Not Suitable
Procurement Plan for Construction Equipment	'Most of our subcontractors use their own lifting mechanisms.'	Suitable	Exit interviews	'I had one from my last company.'	Suitable
Construction Machinery Productivity Analysis	'Project manager, project coordinator and site Forman sit down and fix the position of crane. The subcontractor is responsible for its machinery unless tower crane is required for the project. We get specialist contractor for tower crane and make use of its knowledge in choosing appropriate crane.'	Suitable	Integrated Schedule	'We have master programs which are integrated to procurement program'	Suitable
Site Tools Management	'We have only basic tools.'	Not Suitable	Work Schedule Strategies	'6 days a week for our contractors.'	Suitable
Tools Tracking Technologies	'Each tool has its number kit and it is registered. Site Forman controls. However, it is not a big concern as our laborers do prelims.'	Suitable	Schedule Control	'Forty nightly subby meetings. We identify who is ahead and who is not. There is constant monitor.'	Suitable
Short Interval Planning	'Yeah, we do forty nightly programs, monthly programs and overall programs.'	Suitable	Dynamic Site Layout Plan	'It depends on the site. Every time we move shade, water and power depending on the stage of work. We might use building as our site office.'	Suitable
Well- Defined Scope of Work	'Generally, as a rule of thumb subcontractors know their scope of works. We try to tie things to contract.'	Suitable	Traffic Control Plan	Yeah we employ someone to do for us by providing a copy of our plan and tell them our activities. It is done by certified traffic	Suitable
Use of Software in Planning Dedicated Planner	'MS project, job pack, team binder, Aconex is good for variation and RFI.'	Suitable	Site Security Plan	'It varies from job to job.'	Suitable
Dedicated Planner	'No, the project manager can do that.	Not Suitable	Machinery Positioning Strategy	'We have spoken about that.'	Suitable

		Respondent	9 and 10 Continued		
Construction Management Practices	Summary	Conclusion	Construction Management Practices	Summary	Conclusion
Construction Work Packages	'It is all combined in to one. We tie up the scope of works. The contractor agreed to bring in more labour if he is behind the schedule and agreed to time framework. All these things are included in contract together with the scope of works.'	Suitable	Project Completion Plan	'We bring building surveyors and request for certificate of occupancy. In case of novation we novate building surveyors.'	Suitable
Buildability Review	'During tender phase architects ask us for cost saving. It incurs us lot of cost.'	Suitable	Innovations and New Technologies	'Not really.'	Not Suitable
Contract types	'Lump sum and design plus construct.'	Suitable	Housekeeping	'It is important for productivity. The site needs to be clean otherwise it is subjected to accident.'	Suitable
Model Development	'It is fantastic; especially structural steel works need 3D models.'	Suitable	Health and Safety Policy	'Massive safety plans.'	Suitable
Regulatory Requirements	'Traffic management permits, service permit; it all depends on project types.'	Suitable	Safe work Method Statement	'SWMS/task safety analysis is the responsibility of subcontractor.'	Suitable
Crews Composition	'We do that.'	Suitable	Drugs and Alcohol Testing Program	'No it is not conducted on site.'	Not Suitable
Skills Assessment and Evaluation	'We know by word of mouth who best subcontractor is. It depends on the size of the work package too.'	Suitable	Health and Safety Training Programs	'Yes, OHS'	Suitable
Employees Training	'The company provide training for us. Administrative, first aid and OSH training.'	Suitable	Toolbox Safety Meetings	'It is conducted on weekly basis.'	Suitable
Career Development	'Yes, again it depends on the company you are working with.'	Suitable			

			Respondent 11		
Construction Management Practices	Summary	Conclusion	Construction Management Practices	Summary	Conclusion
Procurement Plans for Materials	⁴ There are two persons responsible for controlling the procurement procedures. Based on our budget we tender to make sure of competitiveness. Sub-contractors are tendered to purchase their own materials and all their own access such as scaffolding. We analysed some gains from purchasing reinforcement bar for this project and we procure it ourselves. Majority of other	Suitable	Regulatory Requirements	'We have safety system and quality assurance system in our company. Building surveyor, fire engineer, CMP requires permit.'	Suitable
Materials Delivery Schedule	"We track during manufacturing, when it is on site, after installation of façade. We write on white board the delivery of materials and track every day."	Suitable	Crew Composition	'Two engineers, Formen, general Forman, senior project manager, construction manager, labourers.'	Suitable
Material Inspection Process	'For plaster boards there is no really that much quality assurance; for concrete we get concrete mix dockets, concrete strength test; check the painting; Glass before it comes to site at factory(China).'	Suitable	Skills Assessment and Evaluation	'They need to have proper training to do specific task. Scaffolding by licensed scaffolder, we see the cards for dogging, rigging, crane works before workers start job. This is done during induction stage.'	Suitable
Material Inspection team	'We get independent building surveyor for checking fire ceiling and other items.'	Suitable	Career Development	'Our company is fairly good in this aspect.'	Suitable
Post Receipt Preservation & Maintenance	'We do not have storage on site.'	Suitable	Non-Financial Incentive Programs	'They do some advertising if you do good job.'	Suitable
Construction Machinery Productivity Analysis	'For the structure, for instance, we do analyse by tracking how many tracks are poured per day with the help of cranes.'	Suitable	Financial Incentive Programs	'Pay for performance scheme is practiced at our company.'	Suitable
Construction Equipment Maintenance	'For the cranes, there are periodic maintenance that we have to meet.	Suitable	Social Activities	'There is an allocated budget for every project.'	Suitable
Site Tools Management Strategy	'We have few, up to 5 to 6 construction labourers on site. Generally, they do safety related works.'	Not Suitable	Stability of Organization Structure	'Regarding our team site setup, we have PM, CM, general Forman (site manager). It is fairly good at the moment.'	Suitable
Tools Tracking Technologies	'There is an asset register.'	Not Suitable	Clear Delegation of Responsibility	'We have clear roles and responsibilities that you sign in to it.'	Suitable
Short Interval Planning	"We have two weekly plans and relate it with main contract program."	Suitable	Retention Plan for Experienced Personnel	'The more experience the more you get paid. The company look after those persons but there is no	Suitable
Well- Defined Scope of Work	'We have templates from previous projects which have been built over a number of years. We normally adapt to specific site. For example, for tiles and other items and put in some inclusions.'	Suitable	Exit interviews	'When you quitting there is an interview to understand why you quit? Where are you going? Please come back and see us when you need.'	Suitable
Use of Software in Planning	'We use primavera P6; two weekly plan on excel spread sheet.'	Suitable	Integrated Schedule	'We do have procurement schedule for long lead items. Specially for this project FFE (fittings, fixture and equipment) schedules are integrated to work schedule.'	Suitable

		Resp	oondent 11 Continued		
Construction Management Practices	Summary	Conclusion	Construction Management Practices	Summary	Conclusion
Dedicated Planner	'We have dedicated planner at head office. But depending on the size of project we have also on site dedicated planner. For example, for Royal Melbourne Hospital we used to have dedicated planner.'	Suitable	Work Schedule Strategies	'We work one every two Saturdays which is scheduled. There is 6 RDO's per year and you are encouraged not to come on these days.'	Suitable
Construction Work Packages	'The project is broken down in to approximately to 20-25 packages. Tiling, façade, structure, landscape, civil works etc. Lift as separate package.'	Suitable	Schedule Control	'We have practical completion date in the contract and monitor it. We mark on the plan the areas which is available for subcontractor to start its work.'	Suitable
Buildability Review	'The design is mostly not ready enough for construction. We get input from subcontractors as a design input.'	Suitable	Dynamic Site Layout Plan	'Our site lay out is dynamic, at the middle of the project it is the same.	Suitable
Contract types	⁶ Over \$50,000 we go for major contracts which is a more detailed contract and minor works contract for less than \$50,000. We have also PSA (professional service agreement) contract. For subcontractors we prefer lump sum which is broken down to the details so that we check how they come up	Suitable	Traffic Control Plan	'Being on community road, we have traffic controller and put different signages.'	Suitable
Model Development	'It is much more important for advertising than for construction. We have some drop ceiling and services clashing which can be identified using 3D.'	Suitable	Site Security Plan	'It is tied to site lay out and all gates are pad locked etc.'	l Suitable
Machinery Positioning Strategy	'As crane is costly item we need to fix the number of crane at the beginning. There is crane study during tendering phase.'	Suitable	Innovations and New Technologies	'ACONEX is great software. OMTRAK is defecting software. Supervisor go on site and send to subcontractor to rectify the defects.'	Suitable
Project start-up plan	'There is project start up plan which is done by PM.'	Suitable	Health and Safety Policy	'safe site is a happy site'. It is also far more productive site if people respect the work site.'	Suitable
Project Completion Plan	'Every six weeks there is project meeting with operation managers on how we go concerning time, safety, cost and other issues.'	Suitable			

		Respondent 12			
Construction Management Practices	Summary	Conclusion	Construction Management Practices	Summary	Conclusion
Procurement Plans for Materials	'We track our procurement from contract award to delivery on site.'	Suitable	Employees Training	'There are trainings for our staff; crane courses, scaffolding courses and others which are related to their works.'	Suitable
Long-Lead Materials Identification	'It come through experience. We talk with contractors about the lead time. Lifts and facades are the two biggest items with long lead time.'	Suitable	Career Development	'There are discussions where people want to go say site management, contract administrator etc.'	Suitable
Procurement Team for Materials	'We do not have one person who is responsible for procurement. We have a person having a combined role.'	Not Suitable	Non-Financial Incentive Programs	'It is mainly for subcontractor.'	Suitable
Materials Status Database	'Façade is tracked using spread sheet. In general, the big picture of materials is tracked by coordination teams.'	Suitable	Financial Incentive Programs	'We sign up to the contract and it is not common.'	Suitable
On-Site Materials Tracking Technology	'We do not track materials on site. Not as I am aware of.'	Not Suitable	Social Activities	'We do team exercises with our project team.'	Suitable
Materials Delivery Schedule	'We work back form construction program and prepare material delivery schedule. We normally provide short buffer in case something go wrong.'	Suitable	Stability of Organization Structure	'Depending on the life cycle of the project the organization changes, we try to keep the stability of the structure because that can be disruptive.'	Suitable
Material Inspection Process	"We inspect upfront; we engage consultants on occasion to conduct factory inspection. We try to conduct at each phase."	Suitable	Clear Delegation of Responsibility	'We have role and responsibility for all of our staff. That is important for efficiency sit down with line manager and do that.'	Suitable
Material Inspection team	'Project coordinator together with industry experts. For example, for façade we have façade expert.'	Suitable	Retention Plan for Experienced Personnel	'Communication with line manager is important and we strive not to loose the staff. We try to promote internally rather than recruiting externally.'	Suitable
Post Receipt Preservation and Maintenance	'Our site is pretty limited in space. Building projects have limited storage area and we try to avoid risk by storing materials on site.'	Not Suitable	Exit interviews	'The organization does it.'	Suitable
Procurement Plan for Construction equipment	"We subcontract everything. We have separate company that supply crane or we might package it up with formwork contractor."	Suitable	Integrated Schedule	'We do. It comes to our program. When we tracking the trade.'	Suitable
Construction Machinery Productivity Analysis	'If the building is tall we need high speed hoist as opposed to mid speed hoist. If the precast is heavy, we need certain size crane. It comes to the knowledge of site management.'	Suitable	Work Schedule Strategies	'We do on occasions on RDO, on Sundays. It depends on the project restrictions by authority. Sometimes you are forced to do if you are behind the schedule.'	Suitable
Construction Equipment Maintenance	'It is separate agreement and if they provide crane, they have to maintain. Our supervisor monitors it.'	Suitable	Schedule Control	'We have site supervisors who work closely with subcontractor and monitor their progress. For example, we have structural supervisor, façade supervisor etc.'	Suitable

	Respo	ondent 12 Conti	nued		
Construction	I		Construction		
Management Practices	Summary	Conclusion	Management Practices	Summary	Conclusion
Site Tools Management Strategy	'We have tool registers on site. But there are only few labourers say if the project has got 200 workers only 20 of them are our labourers.'	Not Suitable	Dynamic Site Layout Plan	'We try to keep consistency. We try to set up our layout with site logistics and consistently with delivery.'	Suitable
Tools Tracking Technologies	'We use tool register for the limited tools that are purchased for our labourers.'	Not Suitable	Traffic Control Plan	'We do that and get approval from the council.'	Suitable
On-site Tools Maintenance	'The site manager order maintenance or discard them.'	Not Suitable	Site Security Plan	'It is normally done by site management team at the start of the project.'	Suitable
Short Interval Planning	'we have contract program, works program, target programs and two weekly programs.'	Suitable	Project start-up plan	'we prepare project start up.'	Suitable
Well- Defined Scope of Work	'It is normally prepared at the tender phase and keep the copy on site for the site team to refer to it.'	Suitable	Project Completion Plan	'we do also prepare project completion.'	Suitable
Use of Software in Planning	'We use MS project, excel spread sheet. There is no special software.'	Suitable	Housekeeping	'It is managed by our site team.'	Suitable
Construction Work Packages	"We tend to package work which suits to the capacity of a contractor. For example, one contractor does formwork, reinforcement and concreting."	Suitable	Health and Safety Policy	'There is a company policy and we have fully accredited.'	Suitable
Buildability Review	'We normal pick it up during design phase. We have design manager, a project coordinator looks at it and refine it. We sometimes bring defect staff right at the very beginning.'	Suitable	Safe work Method Statement	'We do detail safe work method statement.'	Suitable
Utilities Alignment	'We talk to authorities. We do that power, gas whatever it is. The subcontractors also talk to the respective authority. We normally do a survey prior to starting the jobs.'	Suitable	Hazards Analysis	'The same thing as above, yes we do.'	Suitable
Contract types	'We preferred to work on scope of works and fixed contract. If we are not sure on scope of works, we will do by our labour as we cannot fix the price.'	Suitable	Drugs and Alcohol Testing Program	'We do not do any testing.'	Not Suitable
Model Development	'It is important for consultants. It depends on the type of project. For structural steel we use 3-D models.'	Suitable	Health and Safety Training Programs	'Industry red card training is provided.'	Suitable
Regulatory Requirements	'Traffic management plan and construction management plan the response time from council is inconsistent and needs management.'	Suitable	Toolbox Safety Meetings	'On weekly and fortnight basis we conduct meeting.'	Suitable
Crew Composition	'The subcontract organizes their own labour. We do half more than half a dozen of supervisors, project coordinators, Project Manager.'	Suitable			
Skills Assessment and Evaluation	'It is part of the induction process OH and S to demonstrate their competency.'	Suitable			

]	Respondent 13			
Construction Management Practices	Summary	Conclusion	Construction Management Practices	Summary	Conclusion
Procurement Plans for Materials	*Estimate the required materials up front. Successful procurement systems look at the lead times. You need to know what you are ordering, when to order, and how much you need. There is procurement schedules for various materials.'	Suitable	Career Development	'Company has its own culture for career paths.'	Suitable
Long-Lead Materials Identification	'Tiles are from Italy or overseas. You get quotations for these items during tendering. You identify materials up front including their lead times for example tiles from Italy, 8 weeks.'	Suitable	Non-Financial Incentive Programs	'There are incentives like health care, and other which are not paid for us directly'	Suitable
Procurement Team for Materials	 Generally contract Administrator and project engineers are responsible for checking when these materials arrive on site.' 	Not Suitable	Financial Incentive Programs	'There are bonus if you complete the project less than the targeted budget.'	Suitable
Materials Status Database	'We match estimated time during tender with actual quantity; double check quantities. We bring materials one day before installation.'	Suitable	Social Activities	'To reward the team when the team meet the target, there are such activities.'	Suitable
On-Site Materials Tracking Technology	'The person on site ticks the delivery docket. Purchase order, delivery docket and invoice are checked. If there is surplus it is returned back by the supplier. Materials are not stored on site as it is costly to handle them'	Not Suitable	Stability of Organization Structure	'Change is constant otherwise the company will die. '	Suitable
Materials Delivery Schedule	'There are both delivery and procurement schedules.'	Suitable	Clear Delegation of Responsibility	'We have meetings, most people know their roles, if it is not clear they ask during the meeting.'	Suitable
Material Inspection Process	'We conduct various inspections, reinforcement, formwork, concrete test/crack test. We take copy of every delivery docket of the concrete.'	Suitable	Retention Plan for Experienced Personnel	'It depends on person's performance.'	Suitable
Material Inspection team	'Structural engineer inspects reinforcement bars. Other consultants do depending on the type of work.'	Suitable	Exit interviews	'Depending on the company structure, there is an exit interview.'	Suitable
Procurement Plan for Construction equipment	⁶ Based on SWMS we select appropriate equipment and machinery. ⁷	Suitable	Work Schedule Strategies	'I work on RDO's and finish early. It depends on the budget of project.'	Suitable
Construction Machinery Productivity Analysis	we see what is the maximum distance that is covered by the crane. We are checking weather conditions; we look at forecasts. For example, if it is windy there is no point to bring crane to site.'	Suitable	Schedule Control	'We look at the target program and prepare percentage complete'	Suitable
Construction Equipment Maintenance	'There is log book and plant risk register. If there are no maintenance checks done recently, it cannot be used for the project.'	Suitable	Traffic Control Plan	'You need to be mindful of how the traffic is coming. You have people with radis, check the capacity of the road etc.'	Suitable
Site Tools Management Strategy	'Most main builders do not have labour; generally, we want our subcontractors to do the main activities.'	Suitable	Site Security Plan	'As the project progresses, you have different levels of security.'	Suitable

	Respo	ondent 13 Conti	nued		
Construction Management Practices	Summary	Conclusion	Construction Management Practices	Summary	Conclusion
Short Interval Planning	'Yes, we do through project team meeting. Project manager, contract administrator, project engineer and Forman sit together to plan. Construction manager might involve based on the size of the project.'	Suitable	Machinery Positioning Strategy	You look at what you reach safely. You need to assess the worst case scenario.	Suitable
Well- Defined Scope of Work	'Review all the documentation to understand what to build and how to build. And prepare who is doing what and when.'	Suitable	Project start-up plan	'It is how you are going to build.'	Suitable
Use of Software in Planning	'ACONEX is good for picking up items such as defects and for communication. MS project is generally used for time planning.'	Suitable	Project Completion Plan	'It is how we handover to the owner. How you dismantle temporary works.'	Suitable
Dedicated Planner	'Generally, project manager is a planner.'	Suitable	Innovations & New Technologies	'Always new software, printers, new forms etc.'	Suitable
Construction Work Packages	'We generally do not want to break the structure into packages. We conduct interview with subcontractor. Before engaging we need to gage subcontractor.'	Suitable	Housekeeping	'Documentation is done by contract administrator; our laborers do general cleaning; subcontractors' trades clean their own working areas.'	Suitable
Buildability Review	'You should get recommendations from minimum of three recommendations from other builders before engaging a contractor.'	Suitable	Health and Safety Policy	'Yes, we have OH& S manager.'	Suitable
Utilities Alignment	'Dial before you Dig will locate for you before starting construction.'	Suitable	Health and Safety Plan	We try to prepare Safe work method statement	Suitable
Contract types	'You always try to minimize your risk by subcontracting. Your conditions in the head contract will be reflected in subcontracts too.'	Suitable	Safe work Method Statement	'We do detail safe work method statement.'	Suitable
Model Development	'It is starting now by different architects. It is very good for big projects; it will help you fix the ceiling height.'	Suitable	Hazards Analysis	'It is risk assessment.'	Suitable
Regulatory Requirements	'Planning approval, development approval and building approval. Demolitions permit, footpath permits etc.'	Suitable	Drugs and Alcohol Testing Program	'We cover during induction but there is no formal testing program.'	Not Suitable
Crew Composition	'It depends on the project.'	Suitable	Health and Safety Training Programs	'Everybody has to take a minimum of construction safety induction training (white card).'	Suitable
Skills Assessment and Evaluation	'We ask recommendations from previous employer, photos of what they have done previously'	Suitable	Toolbox Safety Meetings	We do it every Friday,	Suitable

		Respondent 14			
Construction Management Practices	Summary	Conclusion	Construction Management Practices	Summary	Conclusion
Procurement Plans for Materials	 'We sign contract with subcontractors that will procure materials with the help of our contract administrator. We will research how much each supplier charges when we decide to procure materials by ourselves. We develop shop drawings before ordering the materials. Generally, we prefer to package work on supply and install basis. Procurement plan is right on the front end. We identify the potential suppliers which might be in China or Italy.' 	Suitable	Employees Training	'There is training on usage of new machineries for our site managers, labourers. Legislations and regulations changes and training is important.'	Suitable
Long-Lead Materials Identification	'During tendering, we look at job and identify long lead items like tiles (from Italy), lift, façade etc tabulate them and put in the program. Power or sub-station are also identified as long lead item so that we commission the building in time.'	Suitable	Career Development	'There is a yearly review to discuss about the career of an individual.'	Suitable
Procurement Team for Materials	'Contracts administrator selects subcontractors, pays them and manage the variations. Our company has got PM, contract manager, project engineer and site manager.'	Not Suitable	Non-Financial Incentive Programs	'No.'	Not Suitable
Materials Status Database	'It is the responsibility of subcontractor to quantify the materials which is sufficient for the job. We provide the documents and ask subcontractors whether everything that we need from us is satisfactory or not. If they say yes, it is then their own risk.'	Suitable	Financial Incentive Programs	'Project managers get bonus. But you are not working for your bonus.'	Suitable
On-Site Materials Tracking Technology	'We do not have direct labour and this eliminates the need to track materials on site and usage of such technologies. Most building companies do not have carpenters, plumbers etc.'	Not Suitable	Social Activities	'Every two months there is an event.'	Suitable

	Respo	ondent 14 Conti	nued		
Construction Management Practices	Summary	Conclusion	Construction Management Practices	Summary	Conclusion
Materials Delivery Schedule	'On certain jobs, you cannot deliver during the day time, due to school days and the fact that the road is in CBD etc. We have big white board at our project site office and subcontractors book in the time slot.'	Suitable	Stability of Organization Structure	'Towards the end, there is less manpower on site. The company may say only contract administrator and labourers should be sufficient to run the project.'	Suitable
Material Inspection Process	"We can also go to China or other country if needed. We generally inspect before installation, take photo graph and email	Suitable	Clear Delegation of Responsibility	'Yeah. You have got roles and responsibility written in your contract.'	Suitable
Material Inspection team	'We create an ITP (inspection test plan). We prepare the checklist for each item and our site manager ticks it.'	Suitable	Retention Plan for Experienced Personnel	'There is not actually a plan. But if you do good job they will pay more money.'	Not Suitable
Post Receipt Preservation & Maintenance	'It depends on project site's size. Sometimes we store materials based on the availability of space on site. We provide 10% for wastage.'	Not Suitable	Exit interviews	'Yeah. The person leaving the company go through this process.'	Suitable
Procurement Plan for Construction equipment	'You have to book your crane early when the industry is busy if you get the job. You have to lock in the price with the supplier too before commencement of the work.'	Suitable	Integrated Schedule	'We run a cost against time. It is a gage where you are. 50% of the job you should be 50% of money spend.'	Suitable
Construction Machinery Productivity Analysis	'Longest reach (radius) is analysed to position the crane. The heaviest lift is also considered. We approach crane company for advice.'	Suitable	Work Schedule Strategies	'We provide a standard 10hrs for four days per week for subcontractors in a contract. But they prefer to work five days including Saturday. The individual get more money on Saturday than company. People are not productive on Saturdays, Sundays and RDO.'	Suitable
Construction Equipment Maintenance	'There is standard maintenance. You got inspections that you need to do based on regulations say on weekly, monthly and yearly basis.'	Suitable	Schedule Control	'There is always changes and management case by case is preferred. You manage design form the front end and resolve the issues with a client early.'	Suitable
Site Tools Management Strategy	'We buy and keep in the container.'	Not Suitable	Dynamic Site Layout Plan	'We have system in place and changes over time. We set up the areas.'	Suitable
Tools Tracking Technologies	'You will be better off including the cost of tools in the cost job than purchasing by yourselves.'	Not Suitable	Traffic Control Plan	'We have ticketed workers/controllers. Sometimes subcontractors do their own controlling.'	Suitable
On-site Tools Maintenance	'The site manager controls the tools. For smaller items we do not worry about it.'	Not Suitable	Site Security Plan	'Depending the type of work we might have plan for working with children, police checks; swap cards etc. Controlling accesses.'	Suitable
Short Interval Planning	'Generally two weekly look ahead. The project manager sits with site manager and look what we do for next two week and track against main program.'	Suitable	Machinery Positioning Strategy	'Similar to what we have discussed above (machinery productivity analysis)'	Suitable

	Resp	ondent 14 Conti	nued		
Well- Defined Scope of Work	'We get subcontractors price based on our tender price. We have generic scope and specific scope for a project. We dissect the work because we want to vary the scope.'	Suitable	Project start-up plan	'We have a list of everything before mobilizing to site; we have done all permits, procurement etc. We fine tune what we are doing.'	Suitable
Use of Software in Planning	'Primavera, MS project.'	Suitable	Project Completion Plan	'Yes for sure.'	Suitable
Dedicated Planner	'No, generally the project manager do the program.'	Not Suitable	Innovations & New Technologies	'There is no specific department at our company which is dedicated for technology research. It is project specific. For example, we seek for technology which helps to cure concrete earlier in one of our projects due to inclement weather.'	Suitable
Construction Work Packages	'It is a bit risky splitting works. For example, we do not want to spilt concrete package twice.'	Suitable	Housekeeping	'We make sure that our scope includes cleaning in subcontractor's contract.'	Suitable
Buildability Review	'The front end work is to review the document. You cannot catch everything but we understand the design and review from constructability point of view.'	Suitable	Health and Safety Policy	'We have got our own policy and it is displayed on site. It is part of the induction process.'	Suitable
Utilities Alignment	'The client does but on occasion we get involved in to reduce delay on our job.'	Suitable	Health and Safety Plan	'Lost time injury, medical injuries. Construction industry is a long way from it.	Suitable
Contract types	'D and C provides opportunity to mold and helps to save money.'	Suitable	Safe work Method Statement	'We do risk assessment at the beginning of the job. Accordingly, subcontractors prepare their own SWMS.'	Suitable
Model Development	'For coordination of services, for steel work we use 3D models.	Suitable	Hazards Analysis	'The same as above(SWMS).'	Suitable
Regulatory Requirements	'We involve in building permit which is the biggest one. Asset protection permit, permit for connection of services that you need to go for.	Suitable	Drugs and Alcohol Testing Program	'We do not do as a requirement at the moment.'	Not Suitable
Crew Composition	'We let sub-contractors to start the job on this date and at an agreed price. We organise the management crew.'	s Suitable	Health and Safety Training Programs	'We do training on use of equipment, use of Asbestos.'	Suitable
Skills Assessment and Evaluation	'We do not necessarily assess the skill except for special works like crane driver. We look at CV, competencies, tickets, and references."	Suitable	Toolbox Safety Meetings	'Every week/ Thursday and it is minuted.'	Suitable

404

]	Respondent 15			
Construction Management	Summary	Conclusion	Construction Management	Summarv	Conclusion
Practices	~~~~~		Practices	2	
Procurement Plans for Materials	'We develop procurement programs. There is sample intake and data process. We prepare tracking sheets and charts. We work back from the date we request to date of manufacture. Different stages of the procurement are represented using different colours on the chart.'	Suitable	Employees Training	'There are various trainings; precast errection, supervision, cranes, health and safety etc.'	Suitable
Long-Lead Materials Identification	'Anything that we import from overseas are categorized as long lead item. Eg. Façade, tiles, light fittings are some of the long lead item materials. We identify these materials early and let the work packages.'	Suitable	Career Development	'We have development plans in our company. We meet three times a year with our employees.'	Suitable
Procurement Team for Materials	'No, a project coordinator oversees the procurement of materials for specific trade.'	Not Suitable	Non-Financial Incentive Programs	'If you meet your target time and budget you keep your work.'	Suitable
Materials Status Database	'We do off-site inspection, materials installed on site. We do weekly report and monthly report to track the status of materials.'	Suitable	Financial Incentive Programs	'There can be. There is certain agreement with the company depending on your position.'	Suitable
On-Site Materials Tracking Technology	'We rely on subcontractors for tracking materials. We generally inspect off-site.'	Not Suitable	Social Activities	'At certain interval/ month we have social activities.'	Suitable
Materials Delivery Schedule	'There is weekly materials schedule.'	Suitable	Stability of Organization Structure	'It is the director who allocates the employees for a project. Our structure comprises of PM, Contract administrator, project coordinator, site manager, supervisors (structure, façade, joinery, carpentry etc.). We try to maintain the stability of our structure.'	Suitable
Material Inspection Process	'We have quality management plan at the start of the job. In that plan we identify the quality management process: supply, manufacture and install. We get sample; we hold that sample until the end of the jo so that we check against that. We also get the technical data or product data signed off by the architect/client/building surveyor. We have also defecting process: our supervisor inspects the installation; our coordinator reports that it is inspected; we make monthly visit with consultants. So we have got various people who constantly inspects.'	Suitable	Clear Delegation of Responsibility	' 'Yes I do. There is clear delegation of responsibilities.'	Suitable

	Re	spondent 15 Co	ntinued		
Construction	Summary	Conclusion	Construction	Summary	Conclusion
Management Practices	Summary	conclusion	Management Practices	Summary	Conclusion
Material Inspection	'Site supervisor, consultant, building	Suitable	Retention Plan for	'It is our target to retain our key personnel (site	
team	surveyor, coordinator is involved in the inspection of materials.'		Experienced Personnel	managers, contract managers)'	Suitable
Post Receipt	'We do not store materials on site and the	Not Suitable	Exit interviews	'I have not done any exit interview but the	
Preservation &	practice is not common.'			company does.'	Suitable
Maintenance					
	'We procure construction plants and				
	equipment under hire agreement. There is				
Procurement Plan for	maintenance obligations provided in the			'The work schedule, material schedule and	
Construction	agreement. Alimak provides maintenance,	Suitable	Integrated Schedule	finance schedule are all integrated in one	Suitable
equipment	maintenance records that the machineries are			program.	
	working properly.'				
	'Yes we do. For example, for the building				
	with 20 to 30 storeys, we provide the				
	standard hoists. So the height of the building			'Yes, we schedule the works based on where	
Construction	determines the type of hoist. For more than	0.411		we are in terms of program, safety/ risk and	0.411
Machinery Draduativity Analysis	30 storeys we have high speed hoists. The	Suitable	Work Schedule Strategies	what the work requires. For instance, we are	Suitable
Floductivity Analysis	foot print or the size of the building also			work hours '	
	determines the number of hoists. The whole			work nours.	
	logistic regarding crane is planned early.				
Construction	'There is regular interval inspections. We				
Equipment	have to get calibration records; electrical				
Maintenance	items are tested and tagged; there is			'We have weekly meetings with	
	regulatory requirement for inspecting	Suitable	Schedule Control	subcontractors. We get status program and we	Suitable
	every new equipment comes to site there is			keep meeting every we.	
	induction.'				
Site Tools	'We have limited site tools and in terms of	Not Suitable	Dynamic Site Layout	'We tender based on our site logistics plan	Suitable
Tools Tracking	'No, not really.'	Not Suitable	Traffic Control Plan	'Yes we have traffic management plan.'	Suitable
Technologies					Suitable
On-site Tools	'Our direct labour is minimal. For example,	Not Suitable	Site Security Plan	'We do not have project specific plan but the	Suitable
	We prepare short term works program. First				
	there is a contract program with the client;				
Short Interval Planning	and that we prepare works program that	Suitable	Machinery Positioning	'We have discussed earlier, there is strategy in	Suitable
Short interval i faminig	program develop into completion program	Suitable	Strategy	place.'	Suitable
	Besides the short work programs there are				
	procurement programs.'				
	'When we are tendering each trade package,				
Well- Defined Scope	each scope of works outlines specifics &				
of Work	general items, site items, quality, safety and	Suitable		'We have a project management plan and	
	environment. The well-defined scope of		Project start-up plan	construction management plan which everyone	Suitable
	works will reference the drawings and			has to sign on to.'	
	specification this will minimize the external				
Use of Software in	'MS project and there is head office support	Suitable		'There is project completion program There	
Planning	in statusing the project every month '	Suitable	Project Completion Plan	are things that need to be completed for	Suitable
	in statusing the project every month.		roject completion rian	closure of the project '	Surable
			1	· · · · · · · · · · · · · · · · · · ·	

Respondent 15 Continued							
Construction Management Practices	Summary	Conclusion	Construction Management Practices	Summary	Conclusion		
Dedicated Planner	'There is dedicated planner at head office. We have programmer.'	Suitable	Innovations & New Technologies	'We are always looking for new technologies and innovations on how we do things/construct, safety, new technologies for our subcontractors.'	Suitable		
Construction Work Packages	*Each trade has its own package that we put for tender which includes program, drawings, specifications, scope of works, invitation to tender, site logistics. Building is split in two 20-30 packages. The capacity of local subcontractors is taken in to account when the project is packaged.'	Suitable	Housekeeping	'We do a safety walk each week. Housekeeping is the responsibility of subcontractors. There are bins on each floor. We have safety committee checking every day.'	Suitable		
Buildability Review	"We have a RED STICK process. When a consultant issues a drawing the document controller receives the drawing and put on a red drawing stick. Coordinators, design manger, project manager, services managers review the drawings and send back for revision to the consultant. So any drawing is reviewed by us for compliance and for any scope changes which has a major cost impact."	Suitable	Health and Safety Policy	'There is companywide policy and project specific.'	Suitable		
Utilities Alignment	'We engage DIAL BEFORE DIG during early phase. We engage the underground detection company for checking. Generally, for utilities we do not accept that risk it is the client's risk. All of the external works are excluded from our contract.'	Suitable	Health and Safety Plan	'We target our lost time and injuries. We have target criteria those.'	Suitable		
Contract types	[•] It is driven by dollar value for us. D& C subcontract agreement are over \$50,000. Anything over \$25,000 minor works agreement. Anything less than \$25,000 we can do under purchase order for minor works. This one is fixed lump sum D & C. We have also guaranteed maximum price, construct only, supply agreement, consultancy agreement?	Suitable	Safe work Method Statement	'Every task has a Safe work method statement (SWMS).'	Suitable		
Model Development	"We do use it to review based on the complexity of the project. But we rely on our consultants to provide us/ for visualization."	Suitable	Hazards Analysis	⁶ Check list to be filled every two months. There are risk workshops to identify other hazards which are not identified during task safety analysis. ²	Suitable		
Regulatory Requirements	'There is demolition permit, building permit for each phase of the building, occupancy permit, other permits from authorities to do certain works during construction phase.'	Suitable	Drugs and Alcohol Testing Program	'No, it is not started.'	Suitable		
Crew Composition	'Our subcontractors have their own crews. We do management crews.'	Suitable	Health and Safety Training Programs	⁴ All employees should go through certificate 4 in health and safety, and first aid. Graduates and cadets should take about six months training on safety during their rotation on	Suitable		
Skills Assessment and Evaluation	'When we are letting the job we make sure that subcontractor has competency to do specific task.'	Suitable	Toolbox Safety Meetings	'We conduct a minimum of two weekly safety talks with our subcontractors'	Suitable		

		Respondent 1	6		
Construction Management Practices	Summary	Conclusion	Construction Management Practices	Summary	Conclusion
Procurement Plans for Materials	'As a lot of our products or more than half of them are imported from China / overseas, we identify critical materials and schedule their procurement early. We prepare good documentation and track what is changing during the procurement.'	Suitable	Employees Training	'OHS and sort of training is provided for our staff / supervisors and site managers.'	Suitable
Long-Lead Materials Identification	'Before starting the work, we start ordering long lead time by considering, manufacturing, transportation and installation times.'	Suitable	Career Development	'It is up to the employee to perform well and get promotion. There is no formal development plan.'	Suitable
Procurement Team for Materials	'We work with subcontractors that can procure materials.'	Not Suitable	Non-Financial Incentive Programs	'Not really.'	Suitable
Materials Status Database	'We quantify major items such as doors, windows and follow up; but for other items like tiles subcontractors quantify by themselves and take the risk. As it is difficult to work wastage for some trades than others.'	Suitable	Financial Incentive Programs	'There is no bonus scheme.'	Suitable
On-Site Materials Tracking Technology	'Our labours do only prelims such as removing rubbish. Major jobs are subcontracted.'	Not Suitable	Social Activities	'There is Christmas party, media and catch up for dinner.'	Suitable
Materials Delivery Schedule	'There is delivery board on site. We do not want to store materials on site.'	Suitable	Stability of Organization Structure	'It is pretty clear. Project coordinator, junior contract administrator, coordinators, site manager. I think it is good set up to have project manager on site than offsite as the decision will be quick.'	Suitable
Material Inspection Process	'Our suppliers ensure the quality of the materials.'	Suitable	Clear Delegation of Responsibility	'The roles for our staff is clearly provided to them. What supervisors roles are, what site manager's role are etc. exists.'	Suitable
Material Inspection team	'Project coordinators and supervisors inspect materials.'	Suitable	Retention Plan for Experienced Personnel	'There have been a lot of changes over last six months.'	Suitable
Post Receipt Preservation & Maintenance	'No there is no storage area for materials on site, there is limited area for sub-contractors.'	Not Suitable	Integrated Schedule	'We are adjusting our labour based on the intensity of job. I suppose there is but no formal analysis.'	Suitable
Procurement Plan for Construction equipment	'Alimak and crane are the critical machineries for building construction.'	Suitable	Work Schedule Strategies	'We work every Saturday, everyday to 5pm; The best way to is adjust your labour hours between Monday to Friday. Working on Sunday is expensive. If you get labour during week days, it is a bit cheaper.'	Suitable
Construction Machinery Productivity Analysis	'We do time and motion study. How long does it take? and how many panels/windows does it take?'	Suitable	Schedule Control	'We have contract program and target program, we adjust our target program and monitor subcontractor how much labour they have got on site and how long does it take to execute certain activity.'	Suitable

Respondent 16 Continued							
Construction Management Practices	Summary	Conclusion	Construction Management Practices	Summary	Conclusion		
Construction Equipment Maintenance	'There is weekly services.'	Suitable	Dynamic Site Layout Plan	'It is always changing. We have set up /plan before we actually change our site layout. For example, we have moved our office twice.ch changes based on the stage of construction.'	Suitable		
Site Tools Management Strategy	'There are very small tools on site. We do not have major tools.'	Not Suitable	Traffic Control Plan	'There is traffic management plan; when we transport heavy precast we have one TMP (traffic management plan) and there are 3-4 TMP's for our project.'	Suitable		
On-site Tools Maintenance	'When tool is brocken we buy new ones.'	Not Suitable	Site Security Plan	'Yes, it changes with site layout plan.'	Suitable		
Short Interval Planning	'We provide more simple program which is two weeks ahead for our subcontractors.'	Suitable	Machinery Positioning Strategy	'The biggest factor is the weight of precast panels. We position cranes near our loading bays. Sometimes we break precast panels to reduce cost of heavy weight lifting cranes.'	Suitable		
Well- Defined Scope of Work	"We try to capture everything in the contact before awarding to the subcontractor. There are not so many variations as the contract type is design and construct."	Suitable	Project start-up plan	'There is a check list to make sure that the given elements are in place.'	Suitable		
Use of Software in	'MS project and excel is utilized.'	Suitable	Project Completion Plan	'There is also project completion plan.'	Suitable		
Dedicated Planner	'No, we do not have dedicated planner. Project manager, contract administrator,	Suitable	Innovations & New Technologies	'I keep abreast with technologies like materials etc'	Suitable		
Construction Work Packages	'Structure, façade (internal and external), external works are some of our packages. Sometimes we separate concrete packages as PT but to reduce risk we provide to one contractor.'	Suitable	Housekeeping	'We work to make sure that subcontractors clean the site.'	Suitable		
Buildability Review	'We do that but it is minor. We are pretty limited in what we are changing. But buildability issues are on our hand.'	Suitable	Health and Safety Policy	'There is ISO certified OH and S policy.'	Suitable		
Utilities Alignment	'Underground power is the biggest thing. It actually varies from project to project.'	Suitable	Health and Safety Plan	'No matter what you do there might be accident.'	Not Suitable		
Contract types	'Always lump sum is preferred'	Suitable	Safe work Method Statement	'Sub-contractor prepare SWMS and we review that)'	Suitable		
Model Development	'We check for different services: mechanical, fire and electrical and see whether they work together or not.'	Suitable	Hazards Analysis	'The biggest thing is external window installation and fall protection system.'	Suitable		
Regulatory Requirements	'There is noise restrictions and we comply with that.'	Suitable	Drugs and Alcohol Testing Program	'I have never heard of alcohol testing but it is up to safety representative to figure out drunken worker.'	Suitable		
Crew Composition	'Our team is structured in a way that works for us. Sub-contractors do for themselves.'	Suitable	Health and Safety Training Programs	'Everyone on site should have certificate and there is pretty good procedure.'	Suitable		
Skills Assessment and Evaluation	'We do not assess the subcontractors' skill.'	Suitable					

		Respondent 1	17		
Construction Management Practices	Summary	Conclusion	Construction Management Practices	Summary	Conclusion
Procurement Plans for Materials	'We do not physically procure materials. We prepare brief and tender out to subcontractors. We sit down with consultant and specify the materials; sometimes we do not pay for materials until they are delivered to site. We also provide subcontractors an advance payment to facilitate the delivery of materials. The best practice in my opinion is to pay materials when it is supplied on site. Any horizontal movement of materials are the responsibility of the sub-contractor.'	Suitable	Employees Training	'Safety training, negotiation, mental resilience training as the work is stressful.'	Suitable
Long-Lead Materials Identification	'Providing buffer time in the schedule of these critical materials is recommended. However, long buffer time may sometimes hold the cash flow of a contractor and reasonable buffer time should be provided.'	Suitable	Career Development	'Definitely.'	Suitable
Procurement Team for Materials	'Cost planner do the estimate, I prepare the brief and we tender.'	Not Suitable	Non-Financial Incentive Programs	'It is available. You can speak to PM/CM to get the incentives.'	Suitable
Materials Status Database	'Break the installation areas into little chunks, produce shop drawings for each section, and prioritize the manufacturing of the materials accordingly.'	Suitable	Financial Incentive Programs	'I do not think there is for this project. It does exist for other projects.'	Suitable
On-Site Materials Tracking Technology	'We use ACONEX to track defects and report it. It shows the location of defects, who is responsible to rectify etc. Subcontractors can log in and see the details which are sent to them. They can also send RFI. However, ACONEX is not an on-site materials technology.'	Not Suitable	Social Activities	'Company definitely encourages that.'	Suitable
Materials Delivery Schedule	'There is delivery schedule for the supply of materials on site. There is even an hourly delivery schedule for precast panels.'	Suitable	Stability of Organization Structure	'We have project matrix which shows the responsibility of the employees. The senior managers constantly update it. The team comprises of Forman, site engineer, senior project engineer, area manager and construction manager.'	Suitable
Material Inspection Process	'We check reinforcement bars, ligatures and other issues before the precast concrete is poured by going to manufacturing plant.'	Suitable	Clear Delegation of Responsibility	'Internally, within the company the work is delegated to me.'	Suitable
Material Inspection team	'The engineer appointed for specific trade is responsible from design to installation and all other bits and pieces for that trade.'	Suitable	Retention Plan for Experienced Personnel	'I am sure there is.'	Suitable
Post Receipt Preservation and Maintenance	'Cost and availability of space on site determines. We follow just in time approach.'	Not Suitable	Exit interviews	'I heard of a few people had an exit interview.'	Suitable
Procurement Plan for Construction equipment	'We do hire cranes. We got the crane inside and remove the crane after completing installing precast panels. Depending on the lifting capacity of the crane, we hire it. We use tower crane.'	Suitable	Integrated Schedule	'Absolutely work schedule and material delivery schedules are integrated.'	Suitable

	Res	pondent 17 Co	ntinued		
Construction Management Practices	Summary	Conclusion	Construction Management Practices	Summary	Conclusion
Construction Machinery Productivity Analysis	'It is measured as the number of lifts per hour. If there is clash in the usage of crane on site, we schedule its usage in a way that is suitable for different subcontractors.'	Suitable	Work Schedule Strategies	'We work every Saturday and RDO's. They do trade off work life balance.'	Suitable
Construction Equipment Maintenance	'Maintenance is done for the sake of safety. We ask the latest maintenance record and details of the product. We carry out inspection at times.'	Suitable	Schedule Control	'Yeah definitely. I track based on the two weekly plan.'	Suitable
Site Tools Management Strategy	'Typically, our laborers bring their own tools and we pay them.'	Not Suitable	Dynamic Site Layout Plan	'We change several times as the team working on the site increases/changes.'	Suitable
Short Interval Planning	"We do that. We get overall program done by senior construction manager, area managers do their plan with some buffer and site engineers do more detailed one. Forman and site engineer do the day to day planning usually in the form of two weekly plan."	Not Suitable	Traffic Control Plan	'We do have one.'	Suitable
Well- Defined Scope of Work	'I produced scope of work based on the previous scopes developed for other projects.'	Not Suitable	Site Security Plan	'We do have that too.'	Suitable
Use of Software in Planning	'P6 is used a lot, excel spread sheet.'	Suitable	Machinery Positioning Strategy	'We discussed above. (machinery productivity analysis)'	Suitable
Dedicated Planner	'Depending on the scale of a project we have regional planner and project specific planner.'	Suitable	Project Completion Plan	'It is captured in construction program as key mile stones.'	Suitable
Construction Work Packages	'We do not want to break in to very small packages. We breakdown in a way that it worths subcontractors to tender.'	Suitable	Innovations & New Technologies	'Generally, construction industry is naturally slow in picking up the technology. Our company give the opportunity for its employee to keep abreast with technology.'	Suitable
Buildability Review	'Yes, we definitely do that.'	Suitable	Housekeeping	'We have team of labourers who are responsible for housekeeping.'	Suitable
Utilities Alignment	'It depends on the project type. Sometimes we can use existing services.'	Suitable	Health and Safety Policy	⁶ We call it GMR (Global Minimum Requirement). Literally it is a bible for safety. We have thinner version for each project which we give to each subcontractor. ²	Suitable
Contract types	'Always lump sum. You take risk out of you. Daily labour is very difficult to manage.'	Suitable	Health and Safety Plan	'For example we look at detail design of precast panel cast in plate and modify/redesign it so that it is safe to work with it.'	Suitable
Model Development	'If the project is big, we use 3D software. For small project we do not as it is expensive.'	Suitable	Safe work Method Statement	'We do SWMS.'	Suitable
Crew Composition	'There are set up team (to measure distance and height) and installing team for precast panels.'	Suitable	Hazards Analysis	'Definitely.'	Suitable
Skills Assessment and Evaluation	'We do see the previous experience to assess the competency.'	Suitable	Drugs and Alcohol Testing Program	'Not in the past but we are planning to start its implementation.'	Not Suitable
Health and Safety Training Programs	'Training is usually conducted online.'	Suitable	Toolbox Safety Meetings	'Once every week and we request our subcontractors to do every morning'	Suitable

Respondent 18							
Construction Management Practices	Summary	Conclusion	Construction Management Practices	Summary	Conclusion		
Procurement Plans for Materials	"We go through at the start of all our programs and report down on package leading. Package leading is a program to suit the long lead time items. We have fair understanding of long lead items '	Suitable	Employees Training	'Yes, absolutely. I have got three lead hands here at the moment; two of them have finished carpentry apprenticeship, one will finish in October.'	Suitable		
Long-Lead Materials Identification	'We identify all of the long lead time items and we set the program up to suit.'	Suitable	Career Development	"We have performance reviews basically every six months which is a formal sit down to discuss how you are going whether you are progressing well, whether you are enjoying etc."	Suitable		
Procurement Team for Materials	'We do not manage it but we engage people to manage it for us. We keep track of them; and we work through with them how we provide that information and how we go about doing it. And we monitor them by getting statuses of where they are up to with their works. But we do not actually do it ourselves.'	Not Suitable	Non-Financial Incentive Programs	'Some companies do.'	Suitable		
Materials Status Database	'We get report back from the contractors on where we are. We go through and do that statusing alone which comprises of percentage complete on all of these items that gives us the date; we go through and monitor that on a regular basis; we also identify remaining durations, where we were up to (percentage complete); we also do with the procurement. If it is procurement of lift items and similar items, we will more likely set up spread sheet '	Suitable	Financial Incentive Programs	'Non-financial programs are probably easy ones that I can think of and financial that is minimal.'	Suitable		
On-Site Materials Tracking Technology	"We try to bring materials on site when it is needed on site. Because if you bring it on site beforehand, then you have storage issues and that become problems, and we do not want to do that.'	Not Suitable	Social Activities	'We try to catch up together with the team every time.'	Suitable		
Materials Delivery Schedule	'We will generally use spread sheet for tracking items: whether materials are manufactured, they are on the boat and coming to Australia, they got through customs; they are on site. Then we start putting in the dates when it starts getting installed. We will be able to monitor that very thoroughly through'	Suitable	Stability of Organization Structure	'Absolutely, that is an absolute fundamental. You do not want the people to leave the job if you want the stability going because knowledge of where the job and how the job is priced running through right to the end to what the defects that knowledge is just gold. '	Suitable		
Material Inspection Process	"The subcontractors go through quality assurance and they will provide us the quality assurance documents for all the items that they are bringing on site; and we will go through and do quality assurance checks on them to make sure that has been according to the plan that has been put forward."	applicable	Clear Delegation of Responsibility	'Yeah we do. It is really hard to get clear delegation of responsibility set out in writing to cover everything there is always be gap'	Suitable		

	Re	spondent 18 Co	ntinued		
Construction Management Practices	Summary	Conclusion	Construction Management Practices	Summary	Conclusion
Short Interval Planning	'We do two weekly look a heads. Generally the site team including foremen pulled out of that information and they will have a duration for activities and works.'	Suitable	Machinery Positioning Strategy	'We sit down, attend time and work through all of these elements; it mixes in with lot of things such as traffic control plan, what are our access points; how we are able to get materials in; how big is the foot print of the job do we need one cranes, two/three cranes, where do we put those cranes, we put them in for the maximum flexibility, we put them in to make sure we get as much coverage as you can;'	Suitable
Well- Defined Scope of Work	"That is always the aim. It depends on how clear the job is; how good the documentation is; and how good the administrator is whether or not that is possible. I like to get a set of drawings and put a highlight around the outside of the drawings and say everything inside this study for example concrete is yours'	Suitable	Project start-up plan	⁶ Project start up plan should be done during the tender phase. If you do when the job you often running, then you are ready to start straight away. ²	Suitable
Use of Software in Planning	'It certainly improves productivity for programmer. We use P6/primavera.'	Suitable	Project Completion Plan	'Project completion plan should really start on day one. All this should go through spread sheet and listing out all of the things that need to be completed that you need to hand over for closure of the project.'	Suitable
Dedicated Planner	'Absolutely. The more complex the job, the more time you need for your planners; the less complex the job the less time you need for your planners. It is fundamental to have good planner on job and have them from the start and have them follow through.'	Suitable	Innovations & New Technologies	'I try to keep abreast of them but it can sometimes be awkward to find the right ones.'	Suitable
Construction Work Packages	'We try to do that by size chunks that suits different contractors. So depending on who the contractor is and what their particular skill set is and depending on what budget constraints are that we are trying to work to, we sit down and try to shape the package to suit contractors to get best price and the best productivity.'	Suitable	Housekeeping	'Housekeeping is big one; we try to do that with all of the subcontractors. They are responsible for keeping their areas clean and cleaning up after themselves on a daily basis.'	Suitable
Buildability Review	'We do or we like to. A lot of our design managers are architects who understand detailing but do not necessarily understand the building. Sometimes site based people are better to fulfil that role because it is how much time is it going to take to do can be as important as the actual cost of the item itself.'	Suitable	Health and Safety Policy	'Formal health and safety policy is very important. It is something which is managed on a daily basis.'	Suitable
Respondent 18 Continued					
--------------------------------------	--	------------	--	---	------------
Construction Management Practices	Summary	Conclusion	Construction Management Practices	Summary	Conclusion
Utilities Alignment	'That is the massive issue now because of the restrictions with connecting into the old networks. Some of the old sewers lines and the storm water do not necessarily have capability for hooking up the new buildings that has been put in place. So there is a lot of works needs to go into those things to make sure that you understand early where you are; and what is the best way to go about delivering those things. That becomes an item that will sit in long lead time items.'	Suitable	Health and Safety Plan	'We have construction safety essential in our kit for working above the line. We put engineering control rather than relying on administrative controls'	Suitable
Contract types	'It is best to back to back with the head contract. If you have got design and contact contract I would like to put my subcontractor as design and contract. If I have got fixed lump sum head contract, I would like fixed lump sum for all of the subcontracts.'	Suitable	Safe work Method Statement	'Absolutely we do these for everything'	Suitable
Model Development	'It makes a massive difference if it is done correctly. I think the investment is worth by client because it takes a lot of risk out of tendering especially for service contractors in complex projects'	Suitable	Hazards Analysis	'Absolutely we do these for everything'	Suitable
Regulatory Requirements	'It is not too much of a problem with building permit; it can be a problem with different councils to get them to sign off your construction management plans and all of those elements.'	Suitable	Drugs and Alcohol Testing Program	'Generally on our jobs we do not have that policy because it is outside the employment arrangement for our employees.'	Not Suitab
Crew Composition	'Generally it comes through sub-contractor. We have crew of management staff in here'	Suitable	Health and Safety Training Programs	'Everyone who starts with us has to go through health and safety training giving as much information as we can and guidance on how to be safe on site.'	Suitable
Skills Assessment and Evaluation	'We go through pretty much with all of our staff. The subcontractor is responsible for training of their tradesmen.'	Suitable	Toolbox Safety Meetings	'We do those when we do specific task that requires people come together and work on specific task'	Suitable

0:		Respondent			
Construction Management Practices	Summary	Conclusion	Construction Management Practices	Summary	Conclusion
Procurement Plans for Materials	'- There is a practice for exploring substitute materials by subcontractor before ordering materials. The cost saving which is obtained by substitution, if accepted, will be shared among main contractor and client Buffer time is provided in ordering materials Checks and balances are in place to make sure that the material is actually made and it is on the way. There mechanisms for confirmations: from manufacturers whether the material has been made via receipt dockets; from transporters via shipping dockets and bill of loading Sometimes the staff of the main builder will fly over to the country where material is manufactured to verify that it is actually been made and it is in progress Monitoring of materials is made by splitting the procurement in to levels by their dates There is weekly meeting to monitor to the progress of procurement. The progress of procurement of materials is reviewed against on site productivity.'	Suitable	Employees Training	'The company provides first aid training; put the employees on courses to become better leaders, negotiators and communicators; crane and scaffolding courses. The respondent described that the industry in construction is always changing, trying to become getting better and safer etc. so people need to be taught that and it needs to be communicated within the industry. Their company is doing for the same purpose.'	Suitable
Long-Lead Materials Identification	'The project team goes through each trade and understands what is typical and got longest lead time and firm up on that trade and its procurement first.'	Suitable	Career Development	'The company will promote you to do a job. They want you to become better as opposed to go and hire someone else.'	Suitable
Procurement Team for Materials	'There is no separate procurement team. A project coordinator that is looking after the specific trade with the assistance of project manager, who makes sure that the coordinator is procuring in the right time frame and right order, is responsible for follow up of the procurement.'	Not Suitable	Non-Financial Incentive Programs	'There is recognition program in a company. Sometimes they announce at team meeting, at medium function, your names are put on the board etc.'	Suitable
Materials Status Database	'The project team use an excel spreadsheet with a series of dates and subheading such as shop drawing, in production, ready for shipping, shipped, ETA (estimated time of arrival) and arrival date.'	Suitable	Financial Incentive Programs	'There is a promotion scheme in the firm; and if you are promoted with that promotion typically comes a pay rise depending on the level of responsibility and work now you have to do.'	Suitable
On-Site Materials Tracking Technology	'-It is not like all materials come to site and it is stacked somewhere and someone is going to find it. When material is brought to site it is straight away put in a location that is very close to where it is get installed. In terms of identifying where it goes on site, some materials are labelled.'	Not Suitable	Social Activities	'To keep the team fresh and exited, the team go for bowling, dinner and others refreshments say to celebrate the completion of a building.'	Suitable

Respondent 19 Continued					
Construction Management Practices	Summary	Conclusion	Construction Management Practices	Summary	Conclusion
Materials Delivery Schedule	'It is scheduled on a weekly basis and it is done with one of our supervisors on site. Delivery schedule is monitored every day and can change. Material delivery schedule is displayed on a big white board in the meeting room, and everyone can see and know when things are coming and at what time.'	Suitable	Stability of Organization Structure	'There is one dedicated person at head office that does look at all employees within the company and what they best suited to and whether they could go for next.'	Suitable
Material Inspection Process	'Inspection is made through fabrication; when it is overseas; when it arrived, when it comes to site, and during installation.'	Suitable	Clear Delegation of Responsibility	⁴ Responsibility of an employee is always made very clear at the beginning of the project. Throughout the project, as the project changes, people's role might also change and that is done through team meeting. Everyone is delegated what they meant to be doing and also minuted in an email and send to everybody so they can always refer to what each person is responsible for. ⁴	Suitable
Material Inspection	'The relevant coordinator in charge of that	Suitable	Retention Plan for	'With experienced personnel when they	Suitable
Post Receipt Preservation & Maintenance	"There are means for protections." 'There are means for protection offsite by subcontractors. They will have an offsite storage facility in a warehouse under cover and will not bring the material until it ready for installation.'	Not Suitable	Experienced Personnel Exit interviews	"The company is very interested in understanding: Why the employees leave? How? What can the company do to change employees mind?"	Suitable
Procurement Plan for Construction equipment	'- Hiring (leasing) techniques is used; crane and hoist are usually the common equipment to be hired. In scheduling machinery, the project team dissect the main program and produce trade specific schedule Analysis such as space availability on site, the capacity of machinery is conducted before deciding (hiring) the number of machineries to be deployed on site. The machineries are identified prior to commencement of construction and included in the sub-contract agreement Crane is the potential equipment that has issues of availability in commercial projects. Sometimes there are too many construction projects in Melbourne and hold all cranes and there are no cranes left for a particular job. This issue is solved by contacting crane companies at the very beginning of the job to secure crane for the project.'	Suitable	Integrated Schedule	'There are programs like, We have got plaster being delivered on this date and plaster being installed on this date. So if plaster is not delivered on this date then we cannot install it on these dates". These are done and updated weekly. Thus, material schedule and work schedule are integrated on the project site.'	Suitable

Respondent 19 Continued					
Construction Management Practices	Summary	Conclusion	Construction Management Practices	Summary	Conclusion
Construction Machinery Productivity Analysis	'-With a crane it is analysed as how fast it can move in the air. Crane's output is also determined by the type of power input (electric/diesel) is analysed. Electric crane does not make noise so it allows to start early in the morning without disturbing the neighbors whereas diesel crane that has got an engine that makes noise and does not allow to start early in the morning The types of hoists (high vs low speed) are analysed. The slower speed cost less money but it takes longer time; a higher speed goes faster and allows the transportation of materials quicker The choice of hoist/Alimak is also based on the number of stories of the building. For a very high rise building, say greater than thirty stories, it is recommended to use two high speed hoists side by side.'	Suitable	Work Schedule Strategie:	'There is an agreement which dictates the minimum working hours of fifty-six per week1; and beyond that the main-builder s negotiates with sub-contractor on the ability to work or not on a Sunday, after hours, on a Saturday, on RDO's etc. Sometimes it has to be because of safety and other issues.'	Suitable
Construction Equipment Maintenance	'The sub-contractor from which the pieces of plant are hired will come to the site and service them. Crane, for example, is serviced once every two weeks; the hoist is once a week. The sub-contractor provides receipt to the main builder describing that it has been maintained.'	Suitable	Schedule Control	^c -When work program for a particular trade is prepared, the project manager/ project team and subcontractors in charge of that task discuss and fix the program. There is involvement of the subcontractor in programmingThe agreed program is then given to supervisor on site who is in charge of that trade and he monitors itIf the sub- contractor is not progressing as per the agreement then the project manager rings and hold a meeting; and tell the subcontractor take corrective actionThe subcontractor agrees to work on Sunday and RDO's; to bring extra crew and get back on program.'	Suitable
Site Tools Management Strategy	'Every tool which is purchased on site is placed in a register and it is then deemed to be leaving somewhere. However, there are a few labourers about five on site and it is not big issue.'	Not Suitable	Dynamic Site Layout Plan	'-The site plan is constantly changing for the sake of productivity. The gates in and out, for instance, might be changed throughout construction period. The gates might be changed after completion of excavation, during superstructure and landscaping worksGates to the site should be planned to allow easy movement of tracks and workersThe respondent recommended that it is nice to have two gates (gate in and gate out). If you have only one gate, it takes longer time for trucks.'	Suitable

Respondent 19 Continued					
Construction Management Practices	Summary	Conclusion	Construction Management Practices	Summary	Conclusion
Tools Tracking Technologies	*A tool register is used to track tools on site. The register shows who took the tool, where it is on site and when it purchased etc. There not many tools on site as our laborers do cleaning and safety related task'	Not Suitable	Traffic Control Plan	'Traffic control plan is done at different stages during construction. There is one done during digging/excavation, there are others when pouring concrete when we finishing the project.'	Suitable
On-site Tools Maintenance	'The life span of the tool is monitored and thrown away if it is beyond the life span, or gathered and given to service men to rectify or recondition.'	Not Suitable	Site Security Plan	'There is strict regime on the project site in terms of how they build their hoardings, the lights going on the hoardings, after hour security guards, cameras, and localized standard lamps.'	Suitable
Short Interval Planning	'-The project is broken down into trades and the project teams focus on a trade which is under execution at that point in time. They re- write a program that is faster than the original programThere is a program for the whole building and there is also a program for smaller part of the building. For instance, if the main program dictates nine days for excavation, then the specific programs might be dig for three days, compact for two days, fill for two days etc.'	Suitable	Machinery Positioning Strategy	'The critical machine or crane's location on site is planned by considering the weight to be loaded, the street, distance of placement and the nearby buildings. The crane needs to reach the street; it needs to reach heaviest lift; it needs to reach entire site. There might be another building which is taller than the building under construction and the crane should not hit the building. All these factors are taken into account when positioning a crane.'	Suitable
Well- Defined Scope of Work	'No one ever gets right in defining scope of works because there are so many things which are not known in construction. It is something that is done through experience. All the things that make the contract to build specific trade is included in the scope of works. Scope of works is basically clarifying in words what that sub-contractor needs to do to deliver their part of a project '	Suitable	Project start-up plan	'Prior to the start of the project, the project team conduct start up meeting which basically allows them to go through the list of items that trigger our knowledge and memory and say oh yes we have done that, we have done that, we have still to do this etc.'	Suitable
Use of Software in Planning	MS project and excel are commonly used as a planning tool.	Suitable	Project Completion Plan	'This is more or less done towards the end of the job where the project team comes up with a regime on how they are going to finish the project on time and within the budget.'	Suitable
Dedicated Planner	At head office, there is dedicated planner/programmer. The planner has the most knowledge because he has written all the programs for other projects; he knows how long it is going to take to pour 100m2 of concrete etc. based on his previous experience.	Suitable	Innovations & New Technologies	'All supervisors at the project site have got an IPADS and they use it for communication purpose such as email, inspections etc. Since the last five years the project team reduced the use of pen and paper for most of its communications. There are also projector, white board, automatic photocopy, and Wi-Fi on site. Supervisors got Wi-Fi having good signal from site office and that eases the communication.'	Suitable
Construction Work Packages	'The main builder packages up a portion of work with one sub-contractor to improve productivity. When the main builder awards packages to concreter, for instance, the sub- contractor does concrete; reo; formwork; pump concrete; supply concrete and provide all the machines that come with it. The respondent prefers the subcontractor to be multi-skilled.'	Suitable	Housekeeping	'There are two dedicated workers that look after the housekeeping. They make sure that the site is clean.'	Suitable

Respondent 19 Continued					
Construction Management Practices	Summary	Conclusion	Construction Management Practices	Summary	Conclusion
Buildability Review	'In most circumstances the design is 80% completed prior to award of the contract and the main builder will complete the reminder of 20% of the design with the consultants/designers. They are novated to the main builder.'	Suitable	Health and Safety Policy	'There is one main policy at head office and there is another policy which is prepared for specific project site.'	Suitable
Utilities Alignment	The builder does not create the facilities. They are on street and will be supplied them.	Suitable	Health and Safety Plan	'The respondent believes that there is always accident on project sites. However, the level of accident varies, for example someone's finger might be hit with hammer and other minor accidents might occur during construction process.	Suitable
Contract types	'Predominantly lump sum or one figure contract is used. Sometimes when the scope of work is not clearly defined, main builder employs daily laborers to execute the activity.'	Suitable	Safe work Method Statement	*-There is a practice in preparing a safe work method statement by conducting safety analysis for a particular trade. The whole step to carry out that task is identified; a hierarchy how risk will be minimized or eliminated is prepared; responsible person is assigned; and finally monitored to make sure that it is executedThere is safety representative at head office that is in charge of Occupational Health and Safety analysis.'	Suitable
Model Development	'Sometimes, depending on the difficulty of what the builder is constructing, there is the usage of models. With the structural steel, for instance, a structural steel contractor will model it up to come up with shop drawings and understand how to build.'	Suitable	Hazards Analysis	'Hazard analysis, which is a global one, is conducted to identify the potential hazards during the construction process. Moreover, safe work method statement is also conducted to identify the potential hazard for specific task.'	Suitable
Regulatory Requirements	'There are various bits and pieces that you need to get approval for prior to building in city or elsewhere. Sometimes there are known lead times, for example, construction management plan requires from 6-12 weeks to get approved and this should be considered during planning. Working on Saturday, past hours of 5 o'clock at night need a permit. These need to be planned ahead in order to make sure that you get approval from the council.'	Suitable	Drugs and Alcohol Testing Program	[•] Drug and Alcohol testing is not conducted at the construction project. But the respondent believes that it is something coming to industry. Since there is work on Saturdays, workers come to site with hangover which has an impact on productivity and safety. [•]	Suitable
Crew Composition	'The main builder requires each subcontractor to have one leading Forman or one leading manager within their crews. The subcontractors are required to make sure that they have experienced labour on site to achieve the desired outcome. We do have a management crew.'	Suitable	Health and Safety Training Programs	'To adapt to lots of changes in the construction industry, there are training programs for the staff with the latest bit of information.'	Suitable
Skills Assessment and Evaluation	'The main builder does not analyze the skill of tradesmen, they will defect their works and inform to the manager of the subcontractor to take corrective action. The main builder is constantly pressuring the subcontractor by making strict supervision of the trades. For instance, what is this? Where is silicon? What is missing? Thus, the subcontractors identify their weaknesses and take corrective action by either providing training to their employees or taking other measures.'	Suitable	Toolbox Safety Meetings	'The project team meets every two weeks to talk about safety, access, weather, productivity, safety and other new things.'	Suitable

APPENDIX 8: LIST OF JOURNAL ARTICLES CONTAINING THE KEYWORDS 'PRODUCTIVITY' AND 'CONSTRUCTION'

1. Journal of Construction Engineering and Management (JCEM)

No.	Title	Year	Author (s)
1	A Regional and Provincial Productivity Analysis	2016	Chancellor, Will; Lu, Wilson
	of the Chinese Construction Industry: 1995 to 2012		WS
2	Estimating Cumulative Damages due to Disruptions in Repetitive Construction	2016	Lee, Jaeseob
3	Productivity Growth in Construction	2016	Sveikauskas, Leo;Rowe, Samuel;Mildenberger, James;Price, Jennifer;Young, Arthur
4	Deconstructing the Construction Industry: A Spatiotemporal Clustering Approach to Profitability Modeling	2016	Choi, Kunhee;Lee, Hyunwoo
5	Oops Simulation: Cost-Benefits Trade-Off Analysis of Reliable Planning for Construction Activities	2016	Hajifathalian, Kiarash;Howell, Gregory A.;Wambeke, Brad W.;Hsiang, Simon ORCID;Liu, Min
6	Development and Comparative Analysis of Construction Industry Labor Productivity Metrics	2016	Vereen, Stephanie C.;Rasdorf, William J.;Hummer, Joseph E.
7	Developing and Optimizing Context-Specific Fuzzy Inference System- Based Construction Labor Productivity Models	2016	Tsehayae, Abraham Assefa;Fayek, Aminah Robinson
8	Incorporating Multiskilling and Learning in the Optimization of Crew Composition	2016	Ahmadian Fard Fini, Alireza R.;Rashidi, Taha Hossein;Akbarnezhad, Ali;Travis Waller, S.
9	Effect of buildability on labor productivity: A practical quantification approach	2016	Jarkas, Abdulaziz M.
10	Applying Artificial Neural Networks for Measuring and Predicting Construction- Labor Productivity	2015	Heravi, Gholamreza;Eslamdoo st, Ehsan
11	Influence Factors of Learning-Curve Effect in High-Rise Building Constructions	2015	Lee, Bogyeong;Lee, Hyunsoo;Park, Moonseo;Kim, Hyunsoo
12	Setting baseline rates for on-site work categories in the construction industry	2015	Shahtaheri, Maryam;Nasir, Hassan;Haas, Carl T M
13	Measuring the construction industry's productivity performan ce: Critique of international productivity comparisons at industry level	2015	Vogl, Bernard;Abdel-Wahab, Mohamed S.
14	Quality change and implications for productivity development: Housing construction in Sweden 1990-2010	2015	Borg, Lena;Song, Hansuck
15	Productivity forecasting of newly added workers based on time-series analysis and site learning	2015	Kim, Hyunsoo,Lee;Hyunsoo,P ark; Moonseo;Ahn, Changbum;Hwang, Sungjoo
16	Method to assess the level of implementation of productivity practices on industrial projects	2015	Caldas, Carlos H.;Kim, Jungyeol;Haas, Carl T M;Goodrum, Paul M.; Zhang, Di

17	Implementing lean production in copper mining development projects: Case study	2015	Castillo, G., Alarcón, L.F., González, V.A.
18	Production system classification matrix: Matching product standardization and production-system design	2015	Jonsson, Henric;Rudberg, Martin
19	Formalized approach to discretize a continuous plant in construction simulations	2014	Lau, SC., Lu, M., Poon, CS.
20	Productivity in daytime and nighttime construction of urban sewer systems	2014	Nguyen, L.D., Nguyen, T.K.N., Tran, D.Q., Villiers, C.
21	Critical success factors and enablers for optimum and maximum industrial modularization	2014	O'Connor, J.T., O'Brien, W.J., Choi, J.O.
22	Cognitive workload demands using 2D and 3D spatial engineering information formats	2014	Dadi, G.B., Goodrum, P.M., Taylor, T.R.B., Carswell, C.M.
23	Automated productivity-based schedule animation: Simulation-based approach to time- cost trade-off analysis	2014	Gelisen, G., Griffis, F.H.
24	Improved baseline method to calculate lost construction productivity	2014	Zhao, T., Dungan, J.M.
25	Analysis of disruptions caused by construction field rework on productivity in residential projects	2014	Arashpour, M., Wakefield, R., Blismas, N., Lee, E.W.M.
26	Interpersonal conflict in construction: Cost, cause, and consequence	2014	Brockman, J.L.
27	Factors affecting construction labor productivity in Kuwait	2014	Jarkas, Abdulaziz M.;Bitar, Camille G.
28	Validation methodologies and their impact in construction productivity research	2014	Liu, J., Shahi, A., Haas, C.T., Goodrum, P., Caldas, C.H.
29	Automated lifting system integrated with construction hoists for table formwork in tall buildings	2014	Kim, T., Lim, H., Cho, H., Kang, KI.
30	State of practice of building information modeling in the electrical construction industry	2014	Hanna, A.S., Yeutter, M., Aoun, D.G.
31	Simulation-based and statistical analysis of the learning effect in floating caisson construction operations	2014	Panas, A., Pantouvakis, J.P.
32	Tower crane cycle times: Case study of remote- control versus cab-control operation	2014	Shapira, A., Elbaz, A.
33	Trade-level productivity measurement: Critical challenges and solutions	2013	Hwang, Bongang;Soh, Chinkiat
34	Organizational barriers to productivity and innovation in large-scale, U.Sbased photovoltaic system construction projects	2013	Yuventi, J., Levitt, R., Robertson, H.
35	Using contractor bid amounts to estimate the impact of night construction on cost for transportation construction	2013	R.E., Thurn, S.B., Ellis, R.D., Lewis, D.W.
36	Algorithmic method for scraper load-time optimization	2013	Marinelli, M., Lambropoulos, S.
37	Task demands in masonry work: Sources, performance implications, and management strategies	2013	Mitropoulos, P., Memarian, B.
38	Assessment of the productivity of nighttime asphalt paving operations	2012	Mostafavi, A., Valentin, V., Abraham, D.M., Louis, J.
39	Greenhouse gas emissions from onsite equipment usage in road construction	2012	Kim, B., Lee, H., Park, H., Kim, H.

40	Buildability factors influencing concreting	2012	Jarkas, Abdulaziz M.
41	Identification and quantification of non-value- adding effort from errors and changes in design and construction projects	2012	Han, S., Lee, S., Peña-Mora, F.
42	Ergonomic analysis and the need for its integration for planning and assessing construction tasks	2012	Inyang, N., Al-Hussein, M., El-Rich, M., Al-Jibouri, S.
43	Dynamics of working hours in construction	2012	Alvanchi, A., Lee, S., Abourizk, S.
44	Construction small-projects rework reduction for capital facilities	2012	Zhang, D., Haas, C.T., Goodrum, P.M., Caldas, C.H., Granger, R.
45	Effects of production control strategy and duration variance on productivity and work in process: Simulation-based investigation	2012	Hajifathalian, K., Wambeke, B.W., Liu, M., Hsiang, S.M.
46	Streamlining the construction productivity improvement process with the proposed role of a construction productivity improvement officer	2012	Ranasinghe, Upul,Ruwanpura, Janaka Y.;Liu, Xin
47	Construction productivity measures for innovation projects	2012	Brchner, J., Olofsson, T.
48	Effects of omitted variable bias on construction real output and its implications on productivity trends in the United States	2012	Dyer, B., Goodrum, P.M., Viele, K.
49	Using last planner and a risk assessment matrix to reduce variation in mechanical related construction tasks	2012	Wambeke, B.W., Liu, M., Hsiang, S.M.
50	Influence of buildability factors on rebar installation labor productivity of columns	2012	Jarkas, Abdulaziz M.
51	Activity analysis for direct-work rate improvement in construction	2011	Gouett, M.C., Haas, C.T., Goodrum, P.M., Caldas, C.H.
52	Model to predict the impact of a technology on construction productivity	2011	Goodrum, P.M., Haas, C.T., Caldas, C., (), Yeiser, J., Homm, D.
53	Causes of variation in construction project task starting times and duration	2011	Wambeke, B.W., Hsiang, S.M., Liu, M
54	Differences in perspectives regarding labor productivity between spanish-and english- speaking craft workers	2011	Dai, Jiukun;Goodrum, Paul M.
55	Site management of work-in-process buffers to enhance project performance using the reliable commitment model: Case study	2011	Gonzlez, V., Alarcn, L.F., Maturana, S., Bustamante, J.A.
56	Linear scheduling model with varying production rates	2011	Duffy, G.A., Oberlender, G.D., Seok Jeong, D.H.
57	Use of safety and lean integrated kaizen to improve performance in modular homebuilding	2011	Ikuma, L.H., Nahmens, I., James, J.
58	Improving productivity on a troubled bridge project	2011	Minchin, Robert Edward;Lewis, Don W.;McLeod, Luke
59	Analysis of factors influencing productivity using craftsmen questionnaires: Case study in a Chilean construction company	2011	Rivas, Rodrigo A.;Borcherding, John D;Alarcn, Luis Fernando;Gonzlez, Vicente
60	Critical investigation into the applicability of the learning curve theory to rebar fixing labor productivity	2010	Jarkas, A.M.

61	Analysis of adverse weather for excusable delays	2010	Nguyen, L.D., Kneppers, J., García De Soto, B., Ibbs, W.
62	Sleep deprivation and its consequences in construction workers	2010	Powell, R., Copping, A.
63	Cross-validation of short-term productivity forecasting methodologies	2010	Hwang, S.
64	Contemporaneous time series and forecasting methodologies for predicting short-term productivity	2010	Hwang, S., Liu, L.Y.
65	Benefits and barriers of construction project monitoring using high-resolution automated cameras	2010	Bohn, J.S., Teizer, J.
66	Improved baseline productivity analysis technique	2010	Lin, CL., Huang, HM.
67	Probabilistic duration estimation model for high- rise structural work	2009	Lee, HS., Shin, JW., Park, M., Ryu, HG
68	Analysis of observed skill affinity patterns and motivation for multiskilling among craft workers in the u.s. industrial construction sector	2009	Wang, Y., Goodrum, P.M., Haas, C.T., Glover, R.W.
69	Simulating learning dynamics in project networks	2009	Taylor, J.E., Levitt, R., Villarroel, J.A.
70	Relationship between automation and integration of construction information systems and labor productivity	2009	Zhai, D., Goodrum, P.M., Haas, C.T., Caldas, C.H.
71	Using agent-based modeling to study construction labor productivity as an emergent property of individual and crew interactions	2009	Watkins, M., Mukherjee, A., Onder, N., Mattila, K.
72	Latent structures of the factors affecting construction labor productivity	2009	Dai, Jiukun;Goodrum, Paul M.;Maloney, William F. ;Srinivasan, Cidambi
73	Safety as an emergent property: Investigation into the work practices of high-reliability framing crews	2009	Mitropoulos, P., Cupido, G.
74	Labor performance analysis for microtunneling projects	2009	Hegab, M., Smith, G.R.
75	Relationship between changes in material technology and construction productivity	2009	Goodrum, Paul M.;Zhai, Dong ;Yasin, Mohammed Fadzil
76	Impact of inspected buffers on production parameters of construction processes	2009	Sawhney, A., Walsh, K.D., Bashford, H.H., Palaniappan, S.
77	Construction craft workers' perceptions of the factors affecting their productivity	2009	Dai, Jiukun;Goodrum, Paul M. ;Maloney, William F.
78	Analysis of techniques leading to radical reduction in project cycle time	2008	Hastak, M., Gokhale, S., Govani K. Hong T. Safi B.
79	Measuring and modeling labor productivity using historical data	2008	Song, L., Abourizk, S.M.
80	Relative productivity in the AEC industries in the United States for on-site and off-site activities	2008	Eastman, C.M., Sacks, R.
81	Construction process reengineering by integrating lean principles and computer simulation techniques	2008	Mao, X., Zhang, X.
82	Vision system for tower cranes	2008	Shapira, A., Rosenfeld, Y., Mizrahi, I.
83	Analysis methods in time-based claims	2008	Arditi, D., Pattanakitchamroon, T.
84	Impact of shift work on labor productivity for labor intensive contractor	2008	Hanna, Awad;Chang, Chulki;Sullivan, Kenneth Timothy:Lackney Jeffery A

86 Measuring the productivity of the construction industry in China by using DEA-based malmquist productivity indices 2007 Xue, X., Shen, Q., Wang, Y., Lu, J. 87 Productivity aspects of urban freeway rehabilitation with accelerated construction 2007 Lee, EB., Lee, H., Ibbs, C.W. 88 Modeling the effect of subjective factors on productivity of trenchless technology application to burine infrastructure systems 2007 Ali, S., Zayed, T., Hegab, M. 90 Contractor prebid planning principles 2007 Bradley, L., Bailey, C., Townsend, K. 91 Technology development decision economics for real-time rolling resistance monitoring of haut roads 2007 Chang, Chulki ;Hanna, Awad;Lackney, Jeffery A., Sullivan, Kenneth Timothy 93 Production equations for unsteady-state construction processes 2007 Hanna, Awad;Chang, Chulki;Lackney, Jeffery A., Sullivan, Kenneth Timothy 94 Delay time analysis in microtunneling projects 2007 Hanna, Awad;Chang, Chulki;Lackney, Jeffery A., Sullivan, Kenneth Timothy 95 Impact of overmanning on mechanical and sheet metal labor productivity 2006 Zhang, H., Tam, C.M., Li, H., Si, J.J. 96 Sorigenetitial construction process 2007 Hagab, M.Y., Smith, G.R., Sallivan, Kenneth Timothy 97 Structural insulated	85	Document Slip-form application to concrete structures	2008	Zayed, T., Sharifi, M.R., Baciu S. Amer M
87 Productivity aspects of urban freeway rehabilitation with accelerated construction 2007 Lee, EB., Lee, H., Ibbs, C.W. 88 Modeling the effect of subjective factors on poductivity of trenchless technology application to buried infrastructure systems 2007 Ali, S., Zayed, T., Hegab, M. 89 Improving employees' work-life balance in the construction industry: Project alliance case study 2007 Lingard, H., Brown, K., Bradley, L., Bailey, C., Townsend, K. 90 Contractor prebid planning principles 2007 Thomas, H.R., Ellis, R.D. 91 Technology development decision economics for real-time rolling resistance monitoring of haul roads 2007 Dunston, P.S., Sinfield, J.V., Lee, TY. 92 Quantifying the impact of schedule compression on labor productivity for mechanical and sheet metal contractor 2007 Hegab, M.Y., Smith, G.R. 93 Production equations for unsteady-state metal labor productivity 2007 Hegab, M.Y., Smith, G.R. 94 Delay time analysis in microhunneling projects 2007 Huma, Awad; Chang, Chulki; Lackney, Jeffery A.,Sullivan, Kenneth Timothy 97 Structural insulated panels: Impact on the residential construction operations 2006 Mullens, M.A., Arif, M. 98 Construction delays in Hong Kong civil engineering projects <td>86</td> <td>Measuring the productivity of the construction industry in China by using DEA-based malmquist productivity indices</td> <td>2008</td> <td>Xue, X., Shen, Q., Wang, Y., Lu, J.</td>	86	Measuring the productivity of the construction industry in China by using DEA-based malmquist productivity indices	2008	Xue, X., Shen, Q., Wang, Y., Lu, J.
88 Modeling the effect of subjective factors on productivity of trenchless technology application to buried infrastructure systems 2007 Ali, S., Zayed, T., Hegab, M. 89 Improving employees' work-life balance in the construction industry: Project alliance case study 2007 Lingard, H., Brown, K., Bradley, L., Bailey, C., Townsend, K. 90 Contractor prebid planning principles 2007 Thomas, H.R., Ellis, R.D. 91 Technology development decision economics for real-time rolling resistance monitoring of haul on labor productivity for mechanical and sheet metal contractor 2007 Chang, Chulki ;Hanna, Awad.Lackney, Jeffery A.;Sullivan, Kenneth Timothy 92 Quantifying the impact of schedule compression on labor productivity for mechanical and sheet metal aloor processes 2007 Hegab, M.Y., Smith, G.R. 94 Delay time analysis in microtunneling projects 2007 Hegab, M.Y., Smith, G.R. 95 Impact of overmanning on mechanical and sheet metal labor productivity 2006 Mullens, M.A., Arif, M. 96 Particle swarm optimization-supported simulation for construction operations 2006 Mullens, M.A., Arif, M. 97 Structural insulated panels: Impact on the residential construction modeling in microtunneling projects 2006 Lo, T.Y., Fung, I.W.H., Tung, K.C.F 98	87	Productivity aspects of urban freeway rehabilitation with accelerated construction	2007	Lee, EB., Lee, H., Ibbs, C.W.
89 Improving employees' work-life balance in the construction industry: Project alliance case study 2007 Lingard, H., Brown, K., Bradley, L., Bailey, C., Townsend, K. 90 Contractor prebid planning principles 2007 Thomas, H.R., Ellis, R.D. 91 Technology development decision economics for real-time rolling resistance monitoring of haul roads 2007 Dunston, P.S., Sinfield, J.V., Lee, TY. 92 Quantifying the impact of schedule compression on labor productivity for mechanical and sheet metal contractor 2007 Chang, Chulki ;Hanna, Awad;Lackney, Jeffery A, Sullivan, Kenneth Timothy 93 Production equations for unsteady-state construction processes 2007 Hagab, M.Y., Smith, G.R. 94 Delay time analysis in microtunneling projects 2007 Hanna, Awad;Lackney, Jeffery A,Sullivan, Kenneth Timothy 95 Impact of overmaning on mechanical and sheet metal labor productivity 2007 Hanna, Awad;Chang, Chulki;Lackney, Jeffery A,Sullivan, Kenneth Timothy 96 Particle swarm optimization-supported simulation for construction operations 2006 Mullens, M.A., Arif, M. 97 Structural insulated panels: Impact on the residential construction projects 2006 Lo, T.Y., Fung, I.W.H., Tung, K.C.F 98 Construction modeling in microtunneling projects <td>88</td> <td>Modeling the effect of subjective factors on productivity of trenchless technology application to buried infrastructure systems</td> <td>2007</td> <td>Ali, S., Zayed, T., Hegab, M.</td>	88	Modeling the effect of subjective factors on productivity of trenchless technology application to buried infrastructure systems	2007	Ali, S., Zayed, T., Hegab, M.
90 Contractor prebid planning principles 2007 Thomas, H.R., Ellis, R.D. 91 Technology development decision economics for real-time rolling resistance monitoring of haul roads 2007 Dunston, P.S., Sinfield, J.V., Lee, TY. 92 Quantifying the impact of schedule compression on labor productivity for mechanical and sheet metal contractor 2007 Walsh, K.D., Sawhney, A., Sullivan, Kenneth Timothy 93 Production equations for unsteady-state construction processes 2007 Hegab, M.Y., Smith, G.R. 94 Delay time analysis in microtunneling projects 2007 Hegab, M.Y., Smith, G.R. 95 Impact of overmanning on mechanical and sheet metal labor productivity 2007 Hanna, Awad/Chang, Chulki; Lackney, Jeffery A.;Sullivan, Kenneth Timothy 96 Particle swarm optimization-supported simulation for construction operations 2006 Mullens, M.A., Arif, M. 97 Structural insulated panels: Impact on the residential construction process 2006 Lo, T.Y., Fung, I.W.H., Tung, K.C.F 98 Construction delays in Hong Kong civil engineering projects 2006 Hegab, M., Smith, G.R., Salem, O.M. 100 Fast-track urban freeway rehabilitation with 55-H weekend closures: I-710 long beach case study 2006 Lee, EB., Lee, H., Harvey, J	89	Improving employees' work-life balance in the construction industry: Project alliance case study	2007	Lingard, H., Brown, K., Bradley, L., Bailey, C., Townsend, K.
91 Technology development decision economics for real-time rolling resistance monitoring of haul roads 2007 Dunston, P.S., Sinfield, J.V., Lee, TY. 92 Quantifying the impact of schedule compression on labor productivity for mechanical and sheet metal contractor 2007 Chang, Chulki ;Hanna, Awad;Lackney, Jeffery A;Sullivan, Kenneth Timothy 93 Production equations for unsteady-state construction processes 2007 Walsh, K.D., Sawhney, A., Bashford, H.H. 94 Delay time analysis in microtunneling projects 2007 Hanna, Awad;Chang, Chulki;Lackney, Jeffery A;Sullivan, Kenneth Timothy 95 Impact of overmanning on mechanical and sheet metal labor productivity 2007 Hanna, Awad;Chang, Chulki;Lackney, Jeffery A;Sullivan, Kenneth Timothy 96 Particle swarm optimization-supported simulation for construction operations 2006 Mullens, M.A., Arif, M. 97 Structural insulated panels: Impact on the residential construction projects 2006 Lo, T.Y., Fung, I.W.H., Tung, K.C.F 99 Soil penetration modeling in microtunneling projects 2006 Hegab, M., Smith, G.R., Salem, O.M. 100 Fast-track urban freeway rehabilitation with 55-H weekend closures: I-710 long beach case study 2006 Hinze, J. Devenport, J.N., Giang, G 104 Contractor process imp	90	Contractor prebid planning principles	2007	Thomas, H.R., Ellis, R.D.
92 Quantifying the impact of schedule compression on labor productivity for mechanical and sheet metal contractor 2007 Chang, Chulki ;Hanna, Awad;Lackney, Jeffery A.;Sullivan, Kenneth Timothy 93 Production equations for unsteady-state construction processes 2007 Walsh, K.D., Sawhney, A., Bashford, H.H. 94 Delay time analysis in microtunneling projects 2007 Hegab, M.Y., Smith, G.R. 95 Impact of overmanning on mechanical and sheet metal labor productivity 2007 Hanna, Awad;Chang, Chulki;Lackney, Jeffery A;Sullivan, Kenneth Timothy 96 Particle swarm optimization-supported simulation for construction operations 2006 Zhang, H., Tam, C.M., Li, H., Shi, J.J. 97 Structural insulated panels: Impact on the residential construction process 2006 Mullens, M.A., Arif, M. 98 Construction delays in Hong Kong civil projects 2006 Lee, T.Y., Fung, I.W.H., Tung, K.C.F 99 Soil penetration modeling in microtunneling projects 2006 Hegab, M., Smith, G.R., Salem, O.M. 100 Fast-track urban freeway rehabilitation with 55-H weekend closures: I-710 long beach case study J.T. 101 Faustoria projects 2006 Thomas, Herve Randolph;Horman, Michael J. 102 Measuring project	91	Technology development decision economics for real-time rolling resistance monitoring of haul roads	2007	Dunston, P.S., Sinfield, J.V., Lee, TY.
93 Production equations for unsteady-state construction processes 2007 Walsh, K.D., Sawhney, A., Bashford, H.H. 94 Delay time analysis in microtunneling projects 2007 Hegab, M.Y., Smith, G.R. 95 Impact of overmanning on mechanical and sheet metal labor productivity 2007 Hanna, Awad;Chang, Chulki;Lackney, Jeffery A;Sullivan, Kenneth Timothy 96 Particle swarm optimization-supported simulation for construction operations 2006 Zhang, H., Tam, C.M., Li, H., Sin, J.J. 97 Structural insulated panels: Impact on the residential construction process 2006 Lo, T.Y., Fung, I.W.H., Tung, K.C.F 98 Construction delays in Hong Kong civil engineering projects 2006 Hegab, M., Smith, G.R., Salem, O.M. 100 Fast-track urban freeway rehabilitation with 55-H weekend closures: I-710 long beach case study J.T. 101 Fundamental principles of workforce management 2006 Hinze, J., Devenport, J.N., Giang, G 102 Measuring project level productivity on transportation projects 2006 Hinze, J., Devenport, J.N., Giang, G 103 Analysis of construction worker injuries that do not result in lost time 2006 Leilis Jr., R.D., Lee, SH. 105 Effects of schedule pressure on constru	92	Quantifying the impact of schedule compression on labor productivity for mechanical and sheet metal contractor	2007	Chang, Chulki ;Hanna, Awad;Lackney, Jeffery A.;Sullivan, Kenneth Timothy
94 Delay time analysis in microtunneling projects 2007 Hegab, M.Y., Smith, G.R. 95 Impact of overmanning on mechanical and sheet metal labor productivity 2007 Hanna, Awad;Chang, Chulki,Lackney, Jeffery A;Sullivan, Kenneth Timothy 96 Particle swarm optimization-supported simulation for construction operations 2006 Zhang, H., Tam, C.M., Li, H., Shi, J.J. 97 Structural insulated panels: Impact on the residential construction process 2006 Mullens, M.A., Arif, M. 98 Construction delays in Hong Kong civil engineering projects 2006 Lo, T.Y., Fung, I.W.H., Tung, K.C.F 99 Soil penetration modeling in microtunneling projects 2006 Hegab, M., Smith, G.R., Salem, O.M. 100 Fast-track urban freeway rehabilitation with 55-H weekend closures: I-710 long beach case study 2006 Thomas, Herve Randolph;Horman, Michael J. 101 Fundamental principles of workforce management 2006 Ellis Jr., R.D., Lee, SH. 102 Measuring project level productivity on transportation projects 2006 Hinze, J., Devenport, J.N., Giang, G 104 Contractor process improvement for enhancing construction productivity 2006 Lies, F., S., Ryu, HG., Yu, JG., Yu, JG., Sur, JG., Sur, JG., Sur, JG., Sur, JJ.	93	Production equations for unsteady-state construction processes	2007	Walsh, K.D., Sawhney, A., Bashford, H.H.
95 Impact of overmanning on mechanical and sheet metal labor productivity 2007 Hanna, Awad;Chang, Chulki;Lackney, Jeffery A;Sullivan, Kenneth Timothy 96 Particle swarm optimization-supported simulation for construction operations 2006 Zhang, H., Tam, C.M., Li, H., Shi, J.J. 97 Structural insulated panels: Impact on the residential construction process 2006 Mullens, M.A., Arif, M. 98 Construction delays in Hong Kong civil engineering projects 2006 Lo, T.Y., Fung, I.W.H., Tung, K.C.F 99 Soil penetration modeling in microtunneling projects 2006 Hegab, M., Smith, G.R., Salem, O.M. 100 Fast-track urban freeway rehabilitation with 55-H weekend closures: 1-710 long beach case study 2006 Lee, EB., Lee, H., Harvey, J.T. 101 Fundamental principles of workforce management 2006 Thomas, Herve Randolph;Horman, Michael J. 102 Measuring project level productivity on transportation projects 2006 Hinze, J., Devenport, J.N., Giang, G 104 Contractor process improvement for enhancing construction productivity 2006 Nepal, M.P., Park, M., Son, B. 105 Effects of schedule pressure on construction performance 2005 Ibbs, W. 106 Impact of chang	94	Delay time analysis in microtunneling projects	2007	Hegab, M.Y., Smith, G.R.
96 Particle swarm optimization-supported simulation for construction operations 2006 Zhang, H., Tam, C.M., Li, H., Shi, J.J. 97 Structural insulated panels: Impact on the residential construction process 2006 Mullens, M.A., Arif, M. 98 Construction delays in Hong Kong civil engineering projects 2006 Lo, T.Y., Fung, I.W.H., Tung, K.C.F 99 Soil penetration modeling in microtunneling projects 2006 Hegab, M., Smith, G.R., Salem, O.M. 100 Fast-track urban freeway rehabilitation with 55-H weekend closures: I-710 long beach case study 2006 Lee, EB., Lee, H., Harvey, J.T. 101 Fundamental principles of workforce management 2006 Thomas, Herve Randolph;Horman, Michael J. 102 Measuring project level productivity on transportation projects 2006 Ellis Jr., R.D., Lee, SH. 103 Analysis of construction worker injuries that do not result in lost time 2006 Cottrell, David S. 104 Contractor process improvement for enhancing construction productivity 2005 Ibbs, W. 105 Effects of schedule pressure on construction performance 2005 Ibbs, W. 106 Impact of change's timing on labor productivity 2005 Lee, HS., Ryu, HG., Yu, J H., Kim, JJ. 108 </td <td>95</td> <td>Impact of overmanning on mechanical and sheet metal labor productivity</td> <td>2007</td> <td>Hanna, Awad;Chang, Chulki;Lackney, Jeffery A;Sullivan, Kenneth Timothy</td>	95	Impact of overmanning on mechanical and sheet metal labor productivity	2007	Hanna, Awad;Chang, Chulki;Lackney, Jeffery A;Sullivan, Kenneth Timothy
97 Structural insulated panels: Impact on the residential construction process 2006 Mullens, M.A., Arif, M. 98 Construction delays in Hong Kong civil engineering projects 2006 Lo, T.Y., Fung, I.W.H., Tung, K.C.F 99 Soil penetration modeling in microtunneling projects 2006 Hegab, M., Smith, G.R., Salem, O.M. 100 Fast-track urban freeway rehabilitation with 55-H weekend closures: 1-710 long beach case study 2006 Lee, EB., Lee, H., Harvey, J.T. 101 Fundamental principles of workforce management 2006 Thomas, Herve Randolph;Horman, Michael J. 102 Measuring project level productivity on transportation projects 2006 Hinze, J., Devenport, J.N., Giang, G 103 Analysis of construction worker injuries that do not result in lost time 2006 Cottractor process improvement for enhancing construction productivity 105 Effects of schedule pressure on construction productivity 2006 Nepal, M.P., Park, M., Son, B. 106 Impact of change's timing on labor productivity 2005 Lee, HS., Ryu, HG., Yu, JH., Sulfvan, Kenneth Timothy, Sargent, Joseph R. 109 Determination of most economical scrapers fleet 2005 Eldin, N.N., Mayfield, J. 108 Factors Affecting Absenteeism in Electrical construction 2005 <td>96</td> <td>Particle swarm optimization-supported simulation for construction operations</td> <td>2006</td> <td>Zhang, H., Tam, C.M., Li, H., Shi, J.J.</td>	96	Particle swarm optimization-supported simulation for construction operations	2006	Zhang, H., Tam, C.M., Li, H., Shi, J.J.
98 Construction delays in Hong Kong civil engineering projects 2006 Lo, T.Y., Fung, I.W.H., Tung, K.C.F 99 Soil penetration modeling in microtunneling projects 2006 Hegab, M., Smith, G.R., Salem, O.M. 100 Fast-track urban freeway rehabilitation with 55-H weekend closures: I-710 long beach case study 2006 Lee, EB., Lee, H., Harvey, J.T. 101 Fundamental principles of workforce 2006 Thomas, Herve Randolph;Horman, Michael J. 102 Measuring project level productivity on transportation projects 2006 Ellis Jr., R.D., Lee, SH. 103 Analysis of construction worker injuries that do not result in lost time 2006 Hinze, J., Devenport, J.N., Giang, G 104 Contractor process improvement for enhancing construction productivity 2006 Nepal, M.P., Park, M., Son, B. 105 Effects of schedule pressure on construction performance 2005 Lee, HS., Ryu, HG., Yu, J H., Kim, JJ. 106 Impact of change's timing on labor productivity 2005 Lee, HS., Ryu, HG., Yu, J H., Kim, JJ. 108 Factors Affecting Absenteeism in Electrical Construction 2005 Eldin, N.N., Mayfield, J. 109 Determination of most economical scrapers fleet 2005 Eldin, N.N., Mayfield, J.	97	Structural insulated panels: Impact on the residential construction process	2006	Mullens, M.A., Arif, M.
99 Soil penetration modeling in microtunneling projects 2006 Hegab, M., Smith, G.R., Salem, O.M. 100 Fast-track urban freeway rehabilitation with 55-H weekend closures: 1-710 long beach case study 2006 Lee, EB., Lee, H., Harvey, J.T. 101 Fundamental principles of workforce management 2006 Thomas, Herve Randolph;Horman, Michael J. 102 Measuring project level productivity on transportation projects 2006 Hinze, J., Devenport, J.N., Giang, G 103 Analysis of construction worker injuries that do not result in lost time 2006 Hinze, J., Devenport, J.N., Giang, G 104 Contractor process improvement for enhancing construction productivity 2006 Nepal, M.P., Park, M., Son, B. 105 Effects of schedule pressure on construction performance 2005 Ibbs, W. 107 Method for calculating schedule delay considering lost productivity 2005 Lee, HS., Ryu, HG., Yu, J H., Kim, JJ. 108 Factors Affecting Absenteeism in Electrical Construction 2005 Hanna, Awad;Menches, Cindy L;Sullivan, Kenneth Timothy;Sargent, Joseph R. 109 Determination of most economical scrapers fleet 2005 Eldin, N.N., Mayfield, J. 110 Effects of delivery systems on change order size and frequency in mechanical construction 2005<	98	Construction delays in Hong Kong civil engineering projects	2006	Lo, T.Y., Fung, I.W.H., Tung, K.C.F
100Fast-track urban freeway rehabilitation with 55-H weekend closures: I-710 long beach case study2006Lee, EB., Lee, H., Harvey, J.T.101Fundamental principles of workforce management2006Thomas, Herve Randolph;Horman, Michael J.102Measuring project level productivity on transportation projects2006Ellis Jr., R.D., Lee, SH.103Analysis of construction worker injuries that do not result in lost time2006Hinze, J., Devenport, J.N., Giang, G104Contractor process improvement for enhancing construction productivity2006Cottrell, David S.105Effects of schedule pressure on construction performance2006Nepal, M.P., Park, M., Son, B.106Impact of change's timing on labor productivity considering lost productivity2005Lee, HS., Ryu, HG., Yu, J H., Kim, JJ.108Factors Affecting Absenteeism in Electrical construction2005Hanna, Awad;Menches, Cindy L;Sullivan, Kenneth Timothy;Sargent, Joseph R.109Determination of most economical scrapers fleet2005Eldin, N.N., Mayfield, J.110Effects of delivery systems on change order size and frequency in mechanical construction construction2005Riley, D.R., Diller, B.E., Kerr, D.	99	Soil penetration modeling in microtunneling projects	2006	Hegab, M., Smith, G.R., Salem, O.M.
101Fundamental principles of workforce management2006Thomas, Herve Randolph;Horman, Michael J.102Measuring project level productivity on transportation projects2006Ellis Jr., R.D., Lee, SH.103Analysis of construction worker injuries that do not result in lost time2006Hinze, J., Devenport, J.N., Giang, G104Contractor process improvement for enhancing construction productivity2006Cottrell, David S.105Effects of schedule pressure on construction performance2006Nepal, M.P., Park, M., Son, B.106Impact of change's timing on labor productivity2005Ibbs, W.107Method for calculating schedule delay considering lost productivity2005Lee, HS., Ryu, HG., Yu, J H., Kim, JJ.108Factors Affecting Absenteeism in Electrical Construction2005Hanna, Awad;Menches, Cindy L;Sullivan, Kenneth Timothy;Sargent, Joseph R.109Determination of most economical scrapers fleet2005Eldin, N.N., Mayfield, J.110Effects of delivery systems on change order size and frequency in mechanical construction D.2005Riley, D.R., Diller, B.E., Kerr, D.	100	Fast-track urban freeway rehabilitation with 55-H weekend closures: I-710 long beach case study	2006	Lee, EB., Lee, H., Harvey, J.T.
102Measuring project level productivity on transportation projects2006Ellis Jr., R.D., Lee, SH.103Analysis of construction worker injuries that do not result in lost time2006Hinze, J., Devenport, J.N., Giang, G104Contractor process improvement for enhancing construction productivity2006Cottrell, David S.105Effects of schedule pressure on construction performance2006Nepal, M.P., Park, M., Son, B.106Impact of change's timing on labor productivity2005Ibbs, W.107Method for calculating schedule delay considering lost productivity2005Lee, HS., Ryu, HG., Yu, J108Factors Affecting Absenteeism in Electrical Construction2005Hanna, Awad;Menches, Cindy L;Sullivan, Kenneth Timothy;Sargent, Joseph R.109Determination of most economical scrapers fleet2005Eldin, N.N., Mayfield, J.110Effects of delivery systems on change order size and frequency in mechanical construction2005Riley, D.R., Diller, B.E., Kerr, D.111Crew production rates for contract time2005O'Conpor, I.T. Huh, Y	101	Fundamental principles of workforce management	2006	Thomas, Herve Randolph;Horman, Michael J.
103Analysis of construction worker injuries that do not result in lost time2006Hinze, J., Devenport, J.N., Giang, G104Contractor process improvement for enhancing construction productivity2006Cottrell, David S.105Effects of schedule pressure on construction performance2006Nepal, M.P., Park, M., Son, B.106Impact of change's timing on labor productivity2005Ibbs, W.107Method for calculating schedule delay considering lost productivity2005Lee, HS., Ryu, HG., Yu, J108Factors Affecting Absenteeism in Electrical Construction2005Hanna, Awad;Menches, Cindy L;Sullivan, Kenneth Timothy;Sargent, Joseph R.109Determination of most economical scrapers fleet2005Eldin, N.N., Mayfield, J.110Effects of delivery systems on change order size and frequency in mechanical construction2005Riley, D.R., Diller, B.E., Kerr, D.111Crew production rates for contract time2005O'Connor, LT, Huh Y	102	Measuring project level productivity on transportation projects	2006	Ellis Jr., R.D., Lee, SH.
104Contractor process improvement for enhancing construction productivity2006Cottrell, David S.105Effects of schedule pressure on construction performance2006Nepal, M.P., Park, M., Son, B.106Impact of change's timing on labor productivity2005Ibbs, W.107Method for calculating schedule delay considering lost productivity2005Lee, HS., Ryu, HG., Yu, J108Factors Affecting Absenteeism in Electrical Construction2005Hanna, Awad;Menches, Cindy L;Sullivan, Kenneth Timothy;Sargent, Joseph R.109Determination of most economical scrapers fleet2005Eldin, N.N., Mayfield, J.110Effects of delivery systems on change order size and frequency in mechanical construction2005Riley, D.R., Diller, B.E., Kerr, D.111Crew production rates for contract time2005O'Connor, I.T. Hub, Y	103	Analysis of construction worker injuries that do not result in lost time	2006	Hinze, J., Devenport, J.N., Giang, G
105Effects of schedule pressure on construction performance2006Nepal, M.P., Park, M., Son, B.106Impact of change's timing on labor productivity2005Ibbs, W.107Method for calculating schedule delay considering lost productivity2005Lee, HS., Ryu, HG., Yu, J108Factors Affecting Absenteeism in Electrical Construction2005Hanna, Awad;Menches, Cindy L;Sullivan, Kenneth Timothy;Sargent, Joseph R.109Determination of most economical scrapers fleet2005Eldin, N.N., Mayfield, J.110Effects of delivery systems on change order size and frequency in mechanical construction2005Riley, D.R., Diller, B.E., Kerr, D.111Crew production rates for contract time2005O'Conpor, I.T. Hub, Y.	104	Contractor process improvement for enhancing construction productivity	2006	Cottrell, David S.
106Impact of change's timing on labor productivity2005Ibbs, W.107Method for calculating schedule delay considering lost productivity2005Lee, HS., Ryu, HG., Yu, J H., Kim, JJ.108Factors Affecting Absenteeism in Electrical Construction2005Hanna, Awad;Menches, Cindy L;Sullivan, Kenneth Timothy;Sargent, Joseph R.109Determination of most economical scrapers fleet2005Eldin, N.N., Mayfield, J.110Effects of delivery systems on change order size and frequency in mechanical construction2005Riley, D.R., Diller, B.E., Kerr, D.111Crew production rates for contract time2005O'Conpor, I.T. Hub, Y.	105	Effects of schedule pressure on construction performance	2006	Nepal, M.P., Park, M., Son, B.
107 Method for calculating schedule delay considering lost productivity 2005 Lee, HS., Ryu, HG., Yu, J H., Kim, JJ. 108 Factors Affecting Absenteeism in Electrical Construction 2005 Hanna, Awad;Menches, Cindy L;Sullivan, Kenneth Timothy;Sargent, Joseph R. 109 Determination of most economical scrapers fleet 2005 Eldin, N.N., Mayfield, J. 110 Effects of delivery systems on change order size and frequency in mechanical construction 2005 Riley, D.R., Diller, B.E., Kerr, D. 111 Crew production rates for contract time 2005 O'Conport, I.T., Hub, Y.	106	Impact of change's timing on labor productivity	2005	Ibbs, W.
108 Factors Affecting Absenteeism in Electrical Construction 2005 Hanna, Awad;Menches, Cindy L;Sullivan, Kenneth Timothy;Sargent, Joseph R. 109 Determination of most economical scrapers fleet 2005 Eldin, N.N., Mayfield, J. 110 Effects of delivery systems on change order size and frequency in mechanical construction 2005 Riley, D.R., Diller, B.E., Kerr, D. 111 Crew production rates for contract time 2005 O'Connor, I.T. Hub, Y.	107	Method for calculating schedule delay considering lost productivity	2005	Lee, HS., Ryu, HG., Yu, J H., Kim, JJ.
109 Determination of most economical scrapers fleet 2005 Eldin, N.N., Mayfield, J. 110 Effects of delivery systems on change order size and frequency in mechanical construction 2005 Riley, D.R., Diller, B.E., Kerr, D. 111 Crew production rates for contract time 2005 O'Connor, I.T. Hub, Y.	108	Factors Affecting Absenteeism in Electrical Construction	2005	Hanna, Awad;Menches, Cindy L;Sullivan, Kenneth Timothy;Sargent, Joseph R.
110 Effects of delivery systems on change order size and frequency in mechanical construction 2005 Riley, D.R., Diller, B.E., Kerr, D. 111 Crew production rates for contract time 2005 O'Connor, I.T. Hub, Y.	109	Determination of most economical scrapers fleet	2005	Eldin, N.N., Mayfield, J.
111 Crew production rates for contract time 2005 O'Connor LT Hub Y	110	Effects of delivery systems on change order size and frequency in mechanical construction	2005	Riley, D.R., Diller, B.E., Kerr, D.
	111	Crew production rates for contract time	2005	O'Connor, J.T., Huh, Y.

	estimation: Bent footing, column, and cap of highway bridges		
112	Predicting industrial construction labor	2005	Favek Aminah
	productivity using fuzzy expert systems		Robinson:Oduba, Avodele
113	Issues in subcontracting practice	2005	Arditi, D., Chotibhongs, R.
114		2005	
114	Benchmarking of Construction Productivity	2005	Park, HS., Thomas, S.R., Tucker, R.L
115	Productivity and cost regression models for pile	2005	Zayed, T.M., Halpin, D.W.
116	Role of inventory huffers in construction labor	2005	Horman Michael L'Thomas
110	performance	2003	Herve Randolph
117	Web-based benchmarking system for the construction industry	2005	Lee, SH., Thomas, S.R., Tucker, R.L.
118	Development model for competitive construction	2005	Xu, T., Tiong, R.L.K., Chew,
	industry in the people's republic of China		D.A.S., Smith, N.J.
119	Fundamental principles of site material	2005	Thomas, Herve
_	management		Randolph; Riley, David R.
	5		;Messner, John I
120	Pile construction productivity assessment	2005	Zayed, T.M., Halpin, D.W.
121	Impact of extended overtime on construction	2005	Hanna, A.S., Taylor, C.S., Sullivan K T
122	Productivity and cost assessment for continuous	2005	Zaved, T.M.
	flight auger piles		
123	Utility-function model for engineering	2005	Georgy, M.E., Chang, LM.,
	performance assessment		Zhang, L.
124	Waste-based management in residential	2005	Zhang, J., Eastham, D.L.,
	construction		Bernold, L.E.
125	Change orders impact on labor productivity	2005	Moselhi, Osama El
			Sayed; Assem, Ihab; El-Rayes,
			Khaled Rayes
126	Describing a beta probability distribution function	2005	Schexnayder, C., Knutson, K.,
	for construction simulation		Fente, J.
127	Stochastic time-cost optimization model	2005	Zheng, D.X.M., Ng, S.T.
	incorporating fuzzy sets theory and		
	nonreplaceable front		
128	Quantifying levels of wasted time in construction	2005	Horman, M.J., Kenley, R.
	with meta-analysis		
129	Cumulative Effect of Project Changes for	2004	Hanna, A.S., Camlic, R.,
	Electrical		Peterson, P.A., Lee, MJ.
	and Mechanical Construction		
130	Predicting construction company decline	2004	Koksal, A., Arditi, D.
131	Selection of concrete pump using the superiority	2004	Tam, C.M., Tong, T.K.L.,
	and inferiority ranking method		Wong, Y.W.
132	Impact of change orders on small labor-intensive	2004	Hanna, A.S., Gunduz, M.
1.6.5	projects	• • • •	
133	Document Process versus data oriented	2004	Zayed, T.M., Halpin, D.W.
	techniques in pile construction productivity		
	assessment		
134	Implementing and applying six sigma in	2004	Pheng, L.S., Hui, M.S.
105	construction	2001	
135	Quantitative assessment for piles productivity	2004	Zayed, T.M., Halpin, D.W.
126	Iactors	2004	7 J T.M. H 1 ' D.W.
136	Simulation as a tool for pile Productivity assessment	2004	Zayed, I.M., Halpin, D.W.
137	Analysis of trends in construction research: 1985-	2004	Abudayyeh, O., Dibert-
	2002		DeYoung, A., Jaselskis, E.

138	Calculating roller requirements for chin seal	2004	Gransberg D.D. Karaca I
130	projects	2004	Senadheera, S.
139	Constructability concepts in Kuala Selangor Cable-Stayed Bridge in Malaysia	2004	Nima, M.A., Abdul-Kadir, M.R., Jaafar, M.S., Alghulami,
140	Establish concrete placing rates using quality	2004	Lu M Anson M
140	control records from Hong Kong building	2004	
	construction projects		
141	Long-term impact of equipment technology on	2004	Goodrum, Paul M.:Haas, Carl
	labor productivity in the U.S. construction		ТМ
	industry at the activity level		
142	Analytical model for analyzing construction	2004	Ho, S.P., Liu, L.Y.
	claims and opportunistic bidding		
143	Integrated electronic commerce model for the	2003	Zhang, N., Tiong, R.
	construction industry		
144	HKCONSIM: A practical simulation solution to	2003	Lu, M., Anson, M., Tang, S.L.,
	planning concrete plant operations in Hong Kong		Ying, Y.C
145	Project delivery systems and project change:	2003	Ibbs, C.W., Kwak, Y.H., Ng,
	Quantitative analysis		T., Murat Odabasi, A.
146	Improving labor flow reliability for better	2003	Thomas, H.R., Horman, M.J.,
	productivity as lean construction principle		Minchin Jr., R.E., Chen, D.
147	Bridge falsework productivity - Measurement and	2003	Tischer, T.E., Kuprenas, J.A.
1.40		2002	
148	Laboratory-based productivity study on	2003	Anand, K.B., Ramamurthy, K.
140	Alternative masonry systems	2002	Con DE Isso DDA
149	indicators for construction	2003	COX, R.F., ISSa, K.K.A.,
150	Nevel method of execution	2002	Antens, D. Brown, D.C.
150	Novel method of excavation	2003	Blown, D.C.
151	Is construction labor productivity really declining?	2003	Rojas, E.M., Aramvareekul, P.
152	Partial factor productivity and equipment	2002	Goodrum, Paul M.:Haas, Carl
-	technology change at activity level in U.S.		ТМ
	construction industry		
153	Physiological demands during construction work	2002	Abdelhamid, T.S., Everett, J.G.
154	Factors in Productivity and Unit Cost for	2002	Jonasson, S., Dunston, P.S.,
	Advanced Machine Guidance		Ahmed, K., Hamilton, J.
155	Identification and resolution of work space	2002	Guo, SJ.
	conflicts in building construction		
156	Benchmarking productivity indicators for	2002	Hanna, A.S., Peterson, P.,
	electrical/mechanical projects		Lee, MJ.
157	Case study of obsolescence and equipment productivity	2002	Bhurisith, I., Touran, A.
158	Multiple simulation analysis for probabilistic cost	2002	Isidore, L.J., Back, W.E.
	and schedule integration		
159	Site layout planning using nonstructural fuzzy	2002	Tam, C.M., Tong, T.K.L.,
	decision support system		Leung, A.W.T., Chiu, G.W.C.
160	Reducing variability to improve performance as a	2002	Thomas, H.R., Horman, M.J.,
	lean construction principle		De Souza, U.E.L., Zavřski, I.
161	2000 Peurifoy lecture: Construction practices in	2002	Thomas, H.R.
1(2	aeveloping countries	2001	
162	Estimating labor production rates for industrial	2001	Abourizk, S., Knowles, P.,
162	Construction activities	2001	Hermelink D.I
103	Linear scheduring model: Float characteristics	2001	панненик, D.J.
164	Role of the fabricator in labor productivity	2000	Thomas, H.R., Sanvido, V.E.
165	Importance of capacity constraints to construction	2000	O'Brien, W.J., Fischer, M.A.

	cost and schedule		
166	Schedule acceleration, work flow, and labor productivity	2000	Thomas, H.R.
167	U.S. construction labor productivity trends, 1970- 1998	2000	Allmon, E., Haas, C.T., Borcherding, J.D., Goodrum, P.M.
168	Adaptive control for safe and quality rebar fabrication	2000	Dunston, P.S., Bernold, L.E.
169	Loss of labor productivity due to delivery methods and weather	1999	Thomas, H.R., Riley, D.R., Sanvido, V.E.
170	Effect of truck payload weight on production	1999	Schexnayder, C., Weber, S.L., Brooks, B.T.
171	Physiological demands of concrete slab placing and finishing work	1999	Abdelhamid, T.S., Everett, J.G.
172	Construction baseline productivity: Theory and practice	1999	Thomas, H.R., Završki, I.
173	Impact of change orders on labor efficiency for mechanical construction	1999	Hanna, A.S., Russell, J.S., Gotzion, T.W., Nordheim, E.V.
174	Earthmoving productivity estimation using linear regression techniques	1999	Smith, S.D.
175	Time series analysis for construction productivity experiments	1999	Abdelhamid, T.S., Everett, J.G.
176	Construction labor productivity modeling with neural networks	1998	Sonmez, R., Rowings, J.E.
177	Multiskilled labor utilization strategies in construction	1998	Burleson, R.C., Haas, C.T., Tucker, R.L., Stanley, A.
178	Quantitative effects of construction changes on labor productivity	1995	Thomas, H.R., Napolitan, C.L
179	Action-response model and loss of productivity in construction	1994	Halligan, D.W., Demsetz, L.A., Brown, J.D., Pace, C.B.
180	Effects of design-integrated process planning on productivity in rebar placement	1994	Salim, M.D., Bernold, L.E.
181	Forecasting labor productivity using factor model	1994	Thomas, H.R., Sakarcan, A.S.
182	Automation and robotics opportunities: Construction versus manufacturing	1994	Everett, J.G., Slocum, A.H.
183	Source evaluation of solid waste in building construction	1994	Gavilan, R.M., Bernold, L.E.
184	Introducing new process technologies into construction companies	1994	Laborde, M., Sanvido, V.
185	Modified roof erection system	1994	Stevens, J.D., Murray, A.L.
186	Human factors in introducing on-site construction automation	1993	Navon, R., Kelly, P.W., Johnston, D.W.
187	Masonry productivity forecasting model	1993	Sanders, S.R., Thomas, H.R.
188	CRANIUM: Device for improving crane productivity and safety	1993	Everett, J.G., Slocum, A.H.
189	Interaction between subcycles: One key to improved methods	1993	Howell, G., Laufer, A., Ballard, G.
190	Span of control of construction foreman: Situational analysis	1991	Laufer, A., Shohet, I.M.
191	Applying computer-integrated manufacturing concepts to construction	1990	Sanvido, V.E., Medeiros, D.J.
192	Bar codes: Prescription for precision, performance, and productivity	1990	
			Blakey L H

No.	Title	Year	Author (s)
1	Creating a baseline for labour productivity of reinforced concrete building construction in Kuwait	2015	Jarkas, A.M., Horner, R.M.W.
2	The Australian construction industry: is the shadow economy distorting productivity?	2015	Chancellor, W., Abbott, M.
3	The construction productivity debate and the measurement of service qualities	2014	Sezer, A.A., Bröchner, J.
4	An analysis of construction productivity differences between Canada and the United States	2014	Nasir, H., Ahmed, H., Haas, C., Goodrum, P.M.
5	Economic development and construction productivity in Malaysia	2014	Chia, F.C., Skitmore, M., Runeson, G., Bridge, A.
6	An exploratory study of the relationship between construction workforce physical strain and task level productivity	2014	Gatti, U.C., Migliaccio, G.C., Bogus, S.M., Schneider, S.
7	Searching for value: Construction strategy exploration and linear planning	2014	Russell, A.D., Tran, N., Staub- French, S.
8	Effects of the location-based management system on production rates and productivity	2014	Seppänen, O., Evinger, J., Mouflard, C.
9	Managing short-term efficiency and long-term development through industrialized construction	2014	Eriksson, P.E., Olander, S., Szentes, H., Widén, K.
10	Industrialized construction in the Swedish infrastructure sector: Core elements and barriers	2014	Larsson, J., Eriksson, P.E., Olofsson, T., Simonsson, P.
11	Challenges and opportunities for productivity improvement studies in linear, repetitive, and location-based scheduling	2014	Lucko, G., Alves, T.D.C.L., Angelim, V.L.
12	The role of the supply chain in the elimination and reduction of construction rework and defects: an action research approach	2014	Taggart, M., Koskela, L., Rooke, J.
13	Remuneration and absenteeism on a large construction site	2014	Kim, J., Philips, P.
14	Can project monitoring and control be fully automated?	2014	Isaac, S., Navon, R.
15	A total factor productivity measure for the construction industry and analysis of its spatial difference: A case study in China	2013	Wang, X., Chen, Y., Liu, B., Shen, Y., Sun, H.
16	Market structure of China's construction industry based on the Panzar-Rosse model	2013	Liu, B., Wang, X., Chen, Y., Shen, Y.
17	An analysis of construction productivity in Malaysia	2012	Chia, F.C., Skitmore, M., Runeson, G., Bridge, A.
18	The efficient scheduling of resources in engineering construction projects: Reflections on a case study from Iran	2012	Greenwood, D., Gledson, B.J.
19	Construction of multi-storey concrete structures in Italy: Patterns of productivity and learning curves	2012	Pellegrino, R., Costantino, N., Pietroforte, R., Sancilio, S.
20	Evaluation of trends in the UK construction industry using growth and productivity accounts	2011	Ruddock, L., Ruddock, S.
21	Effectiveness of craft time utilization in construction	2011	Gong, J., Borcherding, J.D., Caldas, C.H.
22	A value chain statistical definition of construction and the performance of the sector	2011	Squicciarini, M., Asikainen, AL.

2. Construction Management and Economics (CME)

23	Trends of productivity growth in the construction industry across Europe, US and Japan	2011	Abdel-Wahab, M., Vogl, B.
24	An innovative approach for generation of a time location plan in road construction projects	2011	Shah, R.K., Dawood, N.
25	Revisiting the applicability of learning curve theory to formwork labour productivity	2011	Jarkas, A., Horner, M.
26	The impact of management practices on mechanical construction productivity	2011	Shan, Yongwei ;Goodrum, Paul M. ;Zhai, Dong;Haas, Carl T M ;Caldas, Carlos H.
27	Malmquist indices of total factor productivity changes in the Australian construction industry	2010	Li, Y., Liu, C.
28	The influence of buildability factors on rebar fixing labour productivity of beams	2010	Jarkas, A.M.
29	Reassessing productivity in the construction sector to reflect hidden innovation and the knowledge economy	2009	Ruddock, L., Ruddock, S.
30	Using the IKEA model and virtual prototyping technology to improve construction process management	2008	Li, H., Guo, H., Skibniewski, M.J., Skitmore, M.
31	Application of AHP in improving construction productivity from a management perspective	2008	Doloi, H.
32	MBNQA-oriented self-assessment quality management system for contractors: Fuzzy AHP approach	2008	Lam, KC., Lam, M.CK., Wang, D
33	Analysis of craft workers' and foremen's perceptions of the factors affecting construction labour productivity	2007	Dai, Jiukun ;Goodrum, Paul M.; Maloney, William F.
34	Perspective of UK housebuilders on the use of offsite modern methods of construction	2007	Pan, W., Gibb, A.F., Dainty, A.R.J.
35	Document Process planning methodology: Dynamic short-term planning for off-site construction in Slovenia	2007	Radosavljevic, M., Horner, M.
36	Construction equipment productivity estimation using artificial neural network model	2006	Ok, S.C., Sinha, S.K.
37	Technology transfer: International collaboration in Sri Lanka	2006	Ganesan, S., Kelsey, J.
38	Growth, employment and the construction industry in Trinidad and Tobago	2006	Ramsaran, R., Hosein, R.
39	Productivity and delays assessment for concrete batch plant-truck mixer operations	2005	Zayed, T.M., Halpin, D.W., Basha, I.M
40	Benefits and disadvantages of ERP in industrialised timber frame housing in Sweden	2005	Bergström, M., Stehn, L.
41	The JIT materials management system in developing countries	2005	Polat, G., Arditi, D.
42	Deterministic models for assessing productivity and cost of bored piles	2005	Zayed, T.M., Halpin, D.W.
43	Using linear model for learning curve effect on highrise floor construction	2005	Couto, J.P., Teixeira, J.C.
44	Cost vs. production: Labour deployment and productivity in social housing construction in England, Scotland, Denmark and Germany	2004	Clarke, L., Herrmann, G.
45	The use of construction specifications in Singapore	2004	Lam, P.T.I., Kumaraswamy, M.M., Ng, S.T.
46	The impact of non- financial incentives on bricklavers' productivity in Nigeria	2004	Fagbenle, O.I., Adeyemi, A.Y., Adesanya, D.A.

47	Document A longitudinal analysis on the relationship between construction output and GDP in Hong Kong	2004	Yiu, C.Y., Lu, X.H., Leung, M.Y., Jin, W.X.
48	Total factor productivity growth accounting in the construction industry of Singapore	2003	Zhi, M., Hua, G.B., Wang, S.Q., Ofori, G.
49	The construction industry as a loosely coupled system: Implications for productivity and innovation	2002	Dubois, A., Gadde, LE.
50	Predicting downtime costs of tracked hydraulic excavators operating in the UK opencast mining industry	2002	Edwards, D.J., Holt, G.D., Harris, F.C.
51	The divergence in aggregate and activity estimates of US construction productivity	2002	Goodrum, P.M., Haas, C.T., Glover, R.W.
52	The evidence of complex variability in construction labour productivity	2002	Radosavljević M., Malcom W.
53	Schedule-dependent evolution of site layout planning	2001	Elbeltagi, E., Hegazy, T., Hosny, A.H., Eldosouky, A.
54	Benchmarking on-site productivity in France and the UK: A CALIBRE approach	2001	Winch, G., Carr, B.
55	Implementing a multiskilled workforce	2001	Haas, T.C., Rodriguez, A.M., Glover, R., Goodrum, P.M.
56	Design for manufacture: A strategy for successful application to buildings	2001	Fox, S., Marsh, L., Cockerham, G.
57	Productivity oriented analysis of design revisions	2001	Manavazhi, M.R., Xunshi, Z.
58	The predictive ability of Bromilow's time-cost model	2001	Ng, S.T., Mak, M.M.Y., Skitmore, R.M., Lam, K.C., Varnam, M.
59	The European construction industry and its competitiveness: A construct of the European commission	2000	Janssen, J.
60	Forecasting construction industry demand, price and productivity in Singapore: The Box-Jenkins approach	2000	Hua, G.B., Pin, T.H.O.
61	Trends in productivity improvement in the US construction industry	2000	Arditi, D., Mochtar, K.
62	The application of JIT philosophy to construction: A case study in site layout	1999	Pheng, L.S., Hui, M.S.
63	The impact of change orders on mechanical construction labour efficiency	1999	Hanna, A.S., Russell, J.S., Vandenberg, P.J.
64	Construction resource/method factors influencing productivity for high rise concrete construction	1999	Proverbs, D.G., Holt, G.D., Olomolaiye, P.O.
65	How 'just-in-time' wastages can be quantified: Case study of a private condominium project	1998	Pheng, L.S., Tan, S.K.L.
66	The growth of self-employment in British construction	1998	Winch, G.
67	Holistic enhancement of the production analysis of bituminous paving operations	1998	Gowda, R.K., Singh, A., Connolly, M
68	Contributors to construction delays	1998	Kumaraswamy, M.M., Chan, D.W.M.
69	Factors influencing construction time and cost overruns on high-rise projects in Indonesia	1997	Kaming, P.F., Olomolaiye, P.O., Holt, G.D., Harris, F.C.
70	A survey of constraints on Iranian construction operatives' productivity	1996	Zakeri, M., Olomolaiye, P.O., Holt, G.D., Harris, F.C.

430

71	The elasticity of capital-labour substitution in Singapore construction	1996	Tan, W.
72	Productivity improvement in the Indonesian construction industry	1996	Arditi, D., Mochtar, K.
73	Determinants of construction duration	1995	Kumaraswamy, M.M., Chan, D.W.M.
74	An investigation into construction time performance	1995	Walker, D.H.T.
75	Estimating industry-level productivity trends in the building industry from building cost and price data	1993	Chau, K.W.
76	What does the construction foreman do?	1991	Shohet, I.M., Laufer, A.
77	Research of factors influencing construction productivity	1990	Herbsman, Z., Ellis, R.
78	An evaluation of the relationships between bricklayers' motivation and productivity	1990	Olomolaiye, P.O.
79	A system for monitoring and improving construction operative productivity in Nigeria	1989	Olomolaiye, P.O., Ogunlana, S.O.
80	Forecasting Productivity By Work Sampling	1989	Handa, V.K., Abdalla, O.
81	The measurement of productivity in the construction industry	1987	Lowe, J.G.
82	An improved systematic activity sampling technique for work study	1986	Peer, S.
83	Incentives and motivation in the construction industry: A critique	1985	Hague, D.J.
84	US productivity and fast tracking starts on the drawing board	1983	Gray, C., Flanagan, R.

3. Engineering, Construction and Architectural Management (ECAM)

No.	Title	Year	Author (s)
1	Construction labor productivity convergence: A conditional frontier approach	2016	Ma, L., Liu, C., Mills, A.
2	Trust and productivity in Australian construction projects: A subcontractor perspective	2016	Chalker, M., Loosemore, M.
3	Linking employee empowerment with productivity in off-site construction	2015	Alazzaz, F., Whyte, A.
4	Climate and construction delays: Case study in Chile	2015	Ballesteros-Pérez, P., Del Campo-Hitschfeld, M.L., González-Naranjo, M.A., González-Cruz, M.C
5	Valuing innovation in construction and infrastructure: Getting clients past a lowest price mentality	2015	Loosemore, M., Richard, J.
6	A case productivity model for automatic climbing system	2014	Zayed, T., Mohamed, E.
7	Improving construction productivity: A subcontractor's perspective	2014	Loosemore, M.
8	Relationship between building floor and construction labor productivity A case of structural work	2013	Dunlop, P., Smith, S.D.
9	Why do work sampling studies in construction? the case of plumbing work in Scandinavia	2013	Josephson, PE., Bjorkman, L.
10	Contractors perspective toward factors affecting labor productivity in building construction	2013	Mahamid, I.

11	Productivity losses in smoking breaks on construction sites: A case study	2012	Yung, P., Agyekum-Mensah,
12	Simulation modeling and sensitivity analysis of a tunneling construction project's supply chain	2011	Ebrahimy, Y., AbouRizk, S.M., Fernando, S.,
			Mohamed, Y.
13	Comparative analysis of operational coefficients' impact on excavation operations	2010	Panas, A., Pantouvakis, J.P.
14	Trends of skills and productivity in the UK construction industry	2008	Abdel-Wahab, M.S., Dainty, A.R.J., Ison, S.G., Bowen, P., Hazlehurst, G.
15	Nonlinear optimisation and rational cash flow	2007	Beran, V., Dlask, P.
16	Worker safety issues in night-time highway construction	2005	Arditi, D., Ayrancioglu, M., Shi, J.J.
17	Downtime model development for construction equipment management	2004	Nepal, M.P., Park, M.
18	Planning, estimation and productivity in the lean concrete pour	2004	Dunlop, P., Smith, S.D.
19	Rule induction in productivity analysis: A rough set approach	2003	Attoh-Okine, N.O.
20	Assessment of risks in high rise building construction in Jakarta	2003	Santoso, D.S., Ogunlana, S.O., Minato, T.
21	Testing Herzberg's two-factor theory in the Thai construction industry	2003	Ruthankoon, R., Ogunlana, S.O.
22	An example of developing a business model for information and communication technologies (ICT) adoption on construction projects - The National Museum of Australia project	2003	Duyshart, B., Walker, D., Mohamed, S., Hampson, K.
23	Experiences with multiskilling among non-union craft workers in US industrial construction projects	2003	Carley, L.A., Haas, C.T., Borcherding, J.D., Goodrum, P.M.
24	Photo-net: An integrated system for controlling construction progress	2003	Authors of Document Abeid, J., Arditi, D.
25	Building competitive advantage: Construction education in Japan	2003	Tan, W.
26	How building design imperatives constrain construction productivity and quality	2002	Fox, S., Marsh, L., Cockerham, G.
27	ESTIVATE: A model for calculating excavator productivity and output costs	2000	Edwards, D.J., Holt, G.D.
28	Breakwater construction: An effective method for industrial waste utilization	1999	Abraham, D.M., Joanne Yeh, M.H.
29	Promoting the health of construction workers	1998	Uher, T.E., Ritchie, J.
30	Planned construction times and labour utilization — a comparison of UK and French contractors	1996	Proverbs, D.G., Olomolaiye, P.O., Harris, F.C.
31	Do site managers and the head office perceive productivity factors differently?	1996	Naoum, S., Hackman, J.

4. Journal of Management in Engineering (JME)

No.	Title	Year	Author (s)
1	Statistical Analysis of the Effectiveness of Management Programs in Improving Construction Labor Productivity on Large Industrial Projects	2016	Shan, Y., Zhai, D., Goodrum, P.M., Haas, C.T., Caldas, C.H.
2	Analysis of workflow variability and its impacts on productivity and performance in construction of multistory buildings	2015	Arashpour, M., Arashpour, M.

3	Utilization of 3D visualization of mobile crane operations for modular construction on-site assembly	2015	Han, S.H., Hasan, S., Bouferguène, A., Al-Hussein, M., Kosa, J
4	Factors influencing construction labor productivity in Egypt	2014	El-Gohary, K.M., Aziz, R.F.
5	Motivational factors impacting the productivity of construction master craftsmen in Kuwait	2013	Jarkas, A.M., Radosavljevic, M.
6	Identifying, recruiting, and retaining quality field supervisors and project managers in the electrical construction industry	2013	Rojas, E.M.
7	Lean transformation in a modular building company: A case for implementation	2013	Yu, H., Al-Hussein, M., Al- Jibouri, S., Telyas, A.
8	Work flow variation and labor productivity: Case study	2011	Liu, M., Ballard, G., Ibbs, W.
9	Factors affecting engineering productivity	2011	Liao, PC., O'Brien, W.J., Thomas, S.R., Dai, J., Mulva, S.P.
10	Comparative study of activity-based construction labor productivity in the united states and china	2011	Shen, Z., Jensen, W., Berryman, C., Zhu, Y.
11	Critical issues and possible solutions for motivating foreign construction workers	2008	Han, S.H., Park, S.H., Jin, E.J., Kim, H., Seong, Y.K.
12	Six sigma-based approach to improve performance in construction operations	2008	Han, S.H., Chae, M.J., Im, K.S., Ryu, H.D.
13	Case for drug testing of construction workers	2006	Minchin Jr., R.E., Glagola, C.R., Guo, K., Languell, J.L.
14	Strategic management of human resources in construction	2006	Brandenburg, S.G., Haas, C.T., Byrom, K.
15	Baseline determination in construction labor productivity-loss claims	2003	Gulezian, R., Samelian, F.
16	Benchmarking, benchaction, and benchlearning: Rework mitigation in projects	2003	Love, P.E.D., Smith, J.
17	Labor productivity drivers and opportunities in the construction industry	2003	Rojas, E.M., Aramvareekul, P.
18	Role of workforce management in bridge superstructure labor productivity	2003	Thomas, H.R., Minchin Jr., R.E., Chen, D.
19	Motivation parameters for engineering managers using Maslow's theory	1999	Shoura, M.M., Singh, A.
20	Summary of Construction Congress V	1998	Russell, Jeffrey S., Jaselskis, Edward J., Anderson, Stuart D., Hendrickson, Mark L.
21	Partnering: Agreeing to agree	1998	Slater, Thomas S.
22	Construction quality auditing	1997	Calder, Doug A.J.
23	Subcontractor partnering: I'll believe it when I see it	1997	Love, Sally
24	Regional comparison of Indonesian construction productivity	1997	Kaming, P.F., Olomolaiye, P.O., Holt, G.D., Harris, F.C.
25	Difficulties with implementation of goal setting for construction	1994	Hadavi, A., Krizek, R.J.
26	TQM: A view from the playing field	1993	Strange, P.S., Vaughan, G.D.
27	Empowering high-performing people to promote project quality	1993	Tener, R.K.
28	Design productivity: A quality problem	1988	McGeorge, J.F.

No.	Title	Year	Author (s)
1	Cultural differences in motivation factors influencing the management of foreign laborers in the Korean construction industry	2015	Kim, S., Kim, JD., Shin, Y., Kim, GH.
2	Construction industry productivity and the potential for collaborative practice	2014	Fulford, R., Standing, C.
3	Dynamic modeling of labor productivity in construction projects	2013	Nasirzadeh, F., Nojedehi, P.
4	The view of freedom and standardisation among managers in Swedish construction contractor projects	2013	Polesie, P.
5	An improved methodology for selecting similar working days for measured mile analysis	2011	Ibbs, W., Liu, M.
6	Construction quality in China during transition: A review of literature and empirical examination	2010	Yung, P., Yip, B.
7	Major factors influencing productivity of water and wastewater treatment plant construction: Evidence from the deep south USA	2008	Mojahed, S., Aghazadeh, F.
8	Labor productivity: Benchmarking and variability in Egyptian projects	2007	Abdel-Razek, R.H., Abd Elshakour M, H., Abdel-Hamid, M.
9	Large construction projects in developing countries: A case study from Vietnam	2004	Long, N.D., Ogunlana, S., Quang, T., Lam, K.C.
10	Demotivating factors influencing the productivity of civil engineering projects	2004	Ng, S.T., Skitmore, R.M., Lam, K.C., Poon, A.W.C.
11	An overview into the concept of partnering	2001	Naoum, S.
12	A satisfying leadership behaviour model for design consultants	2001	Cheung, S.O., Thomas Ng, S., Lam, K.C., Yue, W.M.
13	A GA-based fuzzy optimal model for construction time-cost trade-off	2001	Leu, SS., Chen, AT., Yang, CH.
14	The management of labour on high rise construction projects: An international investigation	1999	Proverbs, D.G., Holt, G.D., Olomolaiye, P.O.
15	Back to the basics: Biblical wisdom for effective construction project management	1998	Low, S.P.
16	The role of system dynamics in understanding the impact of changes to key project personnel on design production within construction projects	1998	Chapman, R.J.
17	Severity diagnosis of productivity problems - A reliability analysis	1998	Kaming, P.F., Holt, G.D., Kometa, S.T., Olomolaiye, P.O.
18	Total quality management in the construction process	1997	Arditi, D., Gunaydin, H.M.
19	Site organization and supervision in housing projects in the Gaza Strip	1997	Enshassi, A.
20	Construction crew design processes	1997	Hassanein, A.A.G., Melin, J.
21	Factors influencing craftsmen's productivity in Indonesia	1997	Kaming, P.F., Olomolaiye, P.O., Holt, G.D., Harris, F.C.
22	Issues encountered by contractors in Singapore	1995	Alum, J.
23	Conceptual phase of construction projects	1995	Abdul-Kadir, M.R., Price, A.D.F.

5. International Journal of Project Management (IJPM)

24 Quality circles in temporary organizations: lessons from construction projects	1991 Rosenfeld, Y., Warszawski, Laufer, A.	А.,
--	---	-----

6. Automation in Construction (AIC)

No. Title Year	Author (s)
1 Micro-motion level simulation for efficiency 2016 Golabch	i, A., Han, S.,
analysis and duration estimation of manual AbouRis	zk, S., Kanerva, J.
2 Linear scheduling with multiple crews based 2016 Su, Y., I	Lucko, G.
on line-of-balance and productivity	
scheduling method with singularity functions	
³ Systems information modeling: From file 2016 Love, P.	.E.D., Zhou, J.,
instrumentation and control systems	vS, J.
4 Measuring the impact of BIM on labor 2015 Poirier,	E.A., Staub-French,
productivity in a small specialty contracting S., Forg	ues, D.
enterprise through action-research	
5 Optimization-based excavator pose 2015 Vahdati	khakı, F., Hammad,
estimation using real-time location systems A., Sidd	
6 Process reengineering and improvement for 2015 Chen, J.	-H., Yang, LR.,
/ Simulation-based construction productivity 2015 Mirahad	II, F., Zayed, T.
Reasoning	
8 Productivity improvement of precast shop 2015 Nath, T.	, Attarzadeh, M.,
drawings generation through BIM-based Tiong, F	R.L.K.,
process re-engineering Chidam	baram, C., Yu, Z.
9 Dynamic planning of construction activities 2015 Alzraiee	e, H., Zayed, T.,
using hybrid simulation Moselhi	, 0.
10Smart scanning and near real-time 3D2015Wang, C	C., Cho, Y.K.
surface modeling of dynamic construction	
11 Physiological condition monitoring of 2014 Gatti II	C Schneider S
construction workers	cio, G.C.
12 A new approach for automation of location- 2014 Shah, R	.K.
based earthwork scheduling in road	
construction projects	
¹³ Parallel vs. Sequential cascading MEP 2014 Lee, G.,	Kim, J.W.
building case study	
14 Documentation errors in instrumentation and 2013 Love, P.	.E.D., Zhou, J., Sing,
electrical systems: Toward productivity CP., K	im, J.T.
improvement using System Information	
Modeling	1. 1. D. D
¹³ Computer simulation and analysis framework [2013 Pantouv	alzia I D Danag A

16	Location tracking and data visualization	2013	Teizer, J., Cheng, T., Fang,
	technology to advance construction		Υ.
	ironworkers' education and training in safety		
1.7	and productivity		
17	Lifting demand-based zoning for minimizing	2013	Park, M., Ha, S., Lee, HS.,
	worker vertical transportation time in high-		(), Kim, H., Han, S.
10	rise building construction	2012	Kunar S. Las M. Las M.
18	Development of optimized point cloud	2013	Kwon, S., Lee, M., Lee, M.,
	to create earthwork site models		Lee, S., Lee, J.
19	Productivity and CO2 emission analysis for	2013	Hasan S. Boulfarguana A
17	tower crane utilization on high-rise building	2015	Al-Hussein M Gillis P
	projects		Telvas A
20	Automated task-level activity analysis	2013	Cheng, T., Teizer, J.,
	through fusion of real time location sensors		Migliaccio, G.C., Gatti, U.C.
21	and worker's thoracic posture data	2012	Drodhananaa N. Taizan I
21	Automatic spatio-temporal analysis of	2015	Plauliananga, N., Teizer, J.
	GPS data		
22	Supervised vs unsupervised learning for	2012	Oral M Oral E L Avdin
	construction crew productivity prediction		A.
22	Evolution of the i Booth@ angite information	2012	Duvernure IV Hewere
23	management kicsk	2012	Kuwanpura, J. I., newage, K.N. Silva I.P.
			K.N., Silva, L.F.
24	Analysis of field applicability of the rotation-	2012	Lee, C., Lee, G., Park, S.,
	controllable tower-crane hook block		Cho, J.
25	Document An object recognition, tracking,	2011	Gong, J., Caldas, C.H.
	and contextual reasoning-based video		
	interpretation method for rapid productivity		
26	analysis of construction operations	2011	
20	Evaluation of image-based modeling and	2011	Golparvar-Fard, M., Bonn,
	laser scanning accuracy for emerging		J., Teizer, J., Savarese, S., Doño Moro, E
	techniques		
27	Static and dynamic performance evaluation	2011	Saidi K.S. Teizer I
_ /	of a commercially-available ultra wideband	2011	Franaszek M Lytle A M
	tracking system		
28	Formalisms for query capture and data	2011	Pradhan, A., Akinci, B.,
	source identification to support data fusion		Haas, C.T.
	for construction productivity monitoring		
29	Document Predicting construction crew	2010	Oral, E.L., Oral, M.
	productivity by using Self Organizing Maps		
30	Document Estimation of earthworks	2010	Authors of DocumentHola,
	execution time cost by means of artificial		B., Schabowicz, K.
	neural networks		
31	Optimizing the utilization of multiple labor	2010	Jun, D.H., El-Rayes, K.
	shifts in construction projects		
32	Assessing the impact of materials tracking	2009	Grau, D., Caldas, C.H.,
	technologies on construction craft		Haas, C.T., Goodrum, P.M.
	productivity		Gong, J.
		1	-

33	Development of automated change order impact detection and quantification system	2009	Lee, MJ., Ryoo, B.Y., Sullivan, K.T., Hanna, A.S.
34	A performance evaluation of a Stewart platform based Hume concrete pipe manipulator	2009	Kim, Y.S., Lee, J.H., Yoo, H.S., Lee, J.B., Jung, U.S.
35	IT usage in Alberta's building construction projects: Current status and challenges	2008	Hewage, K.N., Ruwanpura, J.Y., Jergeas, G.F.
36	Integrating discrete event and lighting simulation for analyzing construction work zones lighting plans	2008	Nassar, K.
37	Automated retrieval of 3D CAD model objects in construction range images	2008	Bosche, F., Haas, C.T.
38	Simulation study on construction process of FRP bridge deck panels	2007	Hong, T., Hastak, M.
39	Cell-based representation and analysis of spatial resources in construction simulation	2007	Zhang, C., Hammad, A., Zayed, T.M., Wainer, G., Pang, H.
40	Document Synergising R&D initiatives for e- enhancing management support systems	2006	Kumaraswamy, M.M., Palaneeswaran, E., Rahman, M.M., Ugwu, O.O., Ng, S.T.
41	Modeling the ready mixed concrete delivery system with neural networks	2006	Graham, L.D., Forbes, D.R., Smith, S.D.
42	Automated project performance control of construction projects	2005	Navon, R.
43	Planning gang formwork operations for building construction using simulations	2004	Huang, RY., Chen, JJ., Sun, KS.
44	A fuzzy based multi-objective path planning of construction sites	2004	Soltani, A.R., Fernando, T.
45	A CAD-based model for site planning	2004	Sadeghpour, F., Moselhi, O., Alkass, S.
46	Strategic implementation of IT/IS projects in construction: A case study	2002	Stewart, R.A., Mohamed, S., Daet, R.
47	Development of virtual and real-field construction management systems in innovative, intelligent field factory	2000	Authors of DocumentMaruyama, Y., Iwase, Y., Koga, K., (), Ito, T., Tamaki, K.
48	Document Automated construction system for high-rise reinforced concrete buildings	2000	Wakisaka, T., Furuya, N., Inoue, Y., Shiokawa, T.
49	The impact of information technology in design and construction: The owner's perspective	1998	Johnson, R.E., Clayton, M.J.
50	Automation of existing tower cranes: Economic and technological feasibility	1998	Rosenfeld, Y., Shapira, A.
51	Planning and estimating in practice and the use of integrated computer models	1997	Laptali, E., Bouchlaghem, N., Wild, S.

52	Development of automated weather-	1997	Tanijiri, Hideo, Ishiguro,
	unaffected building construction system		Bunya, Arai, Takashi, (),
			Morishima, Yasuto,
			Takasaki, Nobuhiro
53	What can real-time positioning do for	1996	Beliveau, Y.J.
	construction?		
54	Conceptual design of a flooring robot:	1995	Navon, R.
	development methodology and results		
55	Document Construction process simulation	1994	Stouffs, R., Krishnamurti,
	with rule-based robot path planning		R., Lee, S., Oppenheim, I.J.
56	An enhanced construction specific SQL	1994	Kibert, C.J., Hollister, K.C.
57			
57	Site layout of construction temporary	1994	Cheng, MY., O'Connor,
5/	Site layout of construction temporary facilities using an enhanced-geographic	1994	Cheng, MY., O'Connor, J.T.
57	Site layout of construction temporary facilities using an enhanced-geographic information system (GIS)	1994	Cheng, MY., O'Connor, J.T.
57	Site layout of construction temporary facilities using an enhanced-geographic information system (GIS) Determining functional requirements for	1994 1994	Cheng, MY., O'Connor, J.T. Hsieh, Ty., Haas, C.

7. Building Research and Information (BRI)

No.	Title	Year	Author (s)
1	Mapping rework causes and effects using artificial neural networks	2008	Palaneeswaran, E., Love, P.E.D., Kumaraswamy, M.M., Ng, T.S.T.
2	Competitiveness assessment system for China's construction industry	2008	Sha, K., Yang, J., Song, R.
3	Measuring productivity in the construction industry	2006	Crawford, P., Vogl, B.
4	How useful and reliable are construction statistics?	2006	Briscoe, G.
5	Good thinking and poor value: On the socialization of knowledge in construction	2006	Tombesi, P.
6	Performance-based building: Lessons from implementation in New Zealand	2005	Duncan, J.
7	Design quality in buildings	2004	Eley, J.
8	POE in England's Local Government context: A client perspective	2002	Cordy, P.
9	Walking the tightrope: The Probe team's response to BRI commentaries	2002	Bordass, B., Leaman, A., Cohen, R.
10	Predicting construction plant maintenance expenditure	2001	Edwards, D.J., Holt, G.D.
11	Reforming China's construction state-owned enterprises	2001	Sha, K., Lin, S.
12	Assessing building performance in use 5: Conclusions and implications	2001	Bordass, B., Leaman, A., Ruyssevelt, P.
13	Construction process performance variability: Focus on labour productivity	1996	Lema, N.M., Price, A.D.F.

14	Project managers' perception of production problems - An Indonesian case study:	1996	Kaming, P.F., Olomolaiye, P.O., Holt, G.D., Kometa, S.T., Harris, F.C.
15	An expert-simulation system for construction productivity forecasting	1996	Boussabaine, A.H., Duff, A.R.
16	Factors affecting the safety of the construction industry	1996	Jannadi, M.O.

8. Building and Environment (BAE)

No.	Title	Year	Author (s)
1	The air emission assessment of a South Korean traditional building during its life cycle	2016	Sim, J., Sim, J.
2	Evaluating the impacts of high-temperature outdoor working environments on construction labor productivity in China: A case study of rebar workers	2016	Li, X., Chow, K.H., Zhu, Y., Lin, Y.
3	Optimizing work-rest schedule for construction rebar workers in hot and humid environment	2013	Yi, W., Chan, A.P.C.
4	Drivers of productivity among construction workers: A study in a developing country	2007	Kazaz, A., Ulubeyli, S.
5	Improved productivity using a modified table formwork system for high-rise building in Korea	2005	Kim, GH., An, SH., Cho, HH., Seo, DS., Kang, K I.
6	Document Benchmarking change order impacts on productivity for electrical and mechanical projects	2005	Gunduz, M., Hanna, A.S.
7	A quantitative approach for evaluation of negative impact of overmanning on electrical and mechanical projects	2004	Gunduz, M.
8	A different approach to construction labour in Turkey: Comparative productivity analysis	2004	Kazaz, A.Y., Ulubeyli, S.
9	JIT in developing countries-a case study of the Turkish prefabrication sector	2003	Oral, E.L., Mistikoglu, G., Erdis, E.
10	Thermal environment and construction workers' productivity: Some evidence from Thailand	2003	Srinavin, K., Mohamed, S.
11	Recourse to earth for low-cost housing in Nigeria	2002	Olotuah, A.O.
12	Integrating buildability in ISO 9000 quality management systems: Case study of a condominium project	2001	Pheng, L.S., Abeyegoonasekera, B.
13	A method for estimating labour requirements and costs for international construction projects at inception	1999	Proverbs, D.G., Holt, G.D., Olomolaiye, P.O.
14	A comparative evaluation of planning engineers' formwork productivity rates in European construction	1998	Proverbs, D.G., Holt, G.D., Olomolaiye, P.O.

9. Canadian Journal of Civil Engineering (CJCE)

No.	Title	Year	Author (s)
1	Calculating Cumulative Inefficiency using Earned Value Management in Construction Projects	2015	Lee, JS.
2	Identification and comparative analysis of key parameters influencing construction labour productivity in building and industrial projects	2014	Tsehayae, A.A., Fayek, A.R.
3	Framework for improving workflow stability: Deployment of optimized capacity buffers in a synchronized construction production	2014	Arashpour, M., Wakefield, R., Blismas, N., Lee, E.W.M.
4	Development and implementation of a benchmarking and metrics program for construction performance and productivity improvement	2012	Nasir, H., Haas, C.T., Rankin, J.H., (), Forgues, D., Ruwanpura, J. Year the Document was Publish
5	Generational differences on craft workers' perceptions of the factors affecting labour productivity	2012	Dai, J., Goodrum, P.M.
6	Estimating productivity of earthmoving operations using spatial technologies	2012	Montaser, A., Bakry, I., Alshibani, A., Moselhi, O.
7	Assessing key factors impacting the performance and productivity of oil and gas projects in Alberta	2012	Chanmeka, A., Thomas, S.R., Caldas, C.H., Mulva, S.P.
8	Understanding construction workforce absenteeism in industrial construction	2011	Sichani, M.S., Lee, S., Fayek, A.R.
9	Schematic cost estimating model for super tall buildings using a high-rise premium ratio	2011	Authors of DocumentLee, J S., Lee, HS., Park, MS.
10	Delay analysis considering production rate	2011	Lee, JS., Diekmann, J.E.
11	Scheduling model for repetitive construction processes for high-rise buildings	2011	Cho, K., Hong, T., Hyun, C.
12	Buildability factors affecting formwork labour productivity of building floors	2010	Jarkas, A.M.
13	Enhancing construction project supply chains and performance evaluation methods: A case study of a bridge construction project	2010	Pan, NH., Lin, YY., Pan, NF.
14	A novel solution for construction on-site communication - The information booth	2009	Hewage, K.N., Ruwanpura, J.Y.
15	Soft computing approach to construction performance prediction and diagnosis	2008	Dissanayake, M., Fayek, A.R.

16	Production prediction of conventional and global positioning system-based earthmoving systems using simulation and multiple regression analysis	2008	Han, S., Hong, T., Lee, S.
17	Document Application of lean thinking to improve the productivity of water and sewer service installations	2007	Kung, D., Alex, D.P., Al- Hussein, M., Fernando, S.
18	Impact of occasional overtime on construction labor productivity: Quantitative analysis	2007	Sonmez, R.
19	Predicting construction productivity using situation-based simulation models	2006	Choy, E., Ruwanpura, J.Y.
20	Workforce training initiatives for megaproject success	2006	Fayek, A.R., Yorke, M., Cherlet, R.
21	Simulation analysis of productivity variation by global positioning system (GPS) implementation in earthmoving operations	2006	Han, S., Lee, S., Hong, T., Chang, H.
22	Carpentry workers issues and efficiencies related to construction productivity in commercial construction projects in Alberta	2006	Hewage, K.N., Ruwanpura, J.Y.
23	Workflow management and productivity control for asphalt pavement operations	2006	Choi, J., Minchin Jr., R.E.
24	Cost management for concrete batch plant using stochastic mathematical models	2006	Zayed, T.M., Nosair, I.A.
25	Current situation with the production and use of supplementary cementitious materials (SCMs) in concrete construction in Canada	2005	Bouzoubaâ, N., Fournier, B.
26	Quantified benefit of implementing enterprise resource planning through process simulation	2004	Lee, S., Arif, A.U., Jang, H.
27	Results of a pilot study to examine the effective integration of apprentices into the industrial construction sector	2003	Robinson Fayek, A., Shaheen, A., Oduba, A.
28	An optimal construction resource leveling scheduling simulation model	2002	Leu, SS., Hung, TH.
29	Special purpose simulation templates for tunnel construction operations	2001	Ruwanpura, J.Y., AbouRizk, S.M., Er, K.C., Fernando, S.
30	Analysis of cycle excavation and productivity of large-scale rock tunnel projects - Lesson learned in Taiwan	2001	Guo, SJ.
31	A fuzzy expert system for design performance prediction and evaluation	2001	Fayek, A.R., Sun, Z.
32	Simulation of the construction of cable- stayed bridges	1998	Abraham, D.M., Halpin, D.W.

33	Representing a project's physical view in support of project management functions	1998	Russell, A., Chevallier, N.
34	Factors affecting construction productivity: Newfoundland versus rest of Canada	1994	Hanna, A.S., Heale, Donald G.
35	Factors influencing formwork productivity	1993	Smith, Gary R., Hanna, Awad S.
36	Document Impact of change orders on construction productivity	1991	Moselhi, Osama, Leonard, Charles, Fazio, Paul
37	Robotics in construction. Implementation and economic evaluation	1989	Moselhi, Osama, Hason, Stanley

10. Journal of Computing in Civil Engineering (JCCE)

No.	Title	Year	Author (s)
1	Framework of the virtual construction simulator 3 for construction planning and management education	2015	Lee, S., Nikolic, D., Messner, J.I.
2	Automated real-time monitoring system to measure shift production of tunnel construction projects	2013	Ranaweera, K., Ruwanpura, J., Fernando, S.
3	Automated visual recognition of dump trucks in construction videos	2012	Rezazadeh Azar, E., McCabe, B.
4	Planning-based approach for fusing data from multiple sources for construction productivity monitoring	2012	Pradhan, A., Akinci, B.
5	Near real-time motion planning and simulation of cranes in construction: Framework and system architecture	2012	Albahnassi, H., Hammad, A.
6	Reliability-based hybrid data fusion method for adaptive location estimation in construction	2012	Razavi, S.N., Haas, C.T.
7	Epistemic model to monitor the position of mobile sensing nodes on construction sites with rough location data	2012	Grau, D.
8	Integrated simulation system for construction operation and project scheduling	2010	Lee, DE., Yi, CY., Lim, TK., Arditi, D.
9	Computer vision-based video interpretation model for automated productivity analysis of construction operations	2010	Gong, J., Caldas, C.H.
10	Simulation-based identification of possible locations for mobile cranes on construction sites	2008	Tantisevi, K., Akinci, B.
11	Using hue, saturation, and value color space for hydraulic excavator idle time analysis	2007	Zou, J., Kim, H.

12	Decision tree approach to classify and quantify cumulative impact of change orders on productivity	2004	Lee, MJ., Hanna, A.S., Loh, WY.
13	An ontology for relating features with activities to calculate costs	2003	Authors of DocumentStaub- French, S., Fischer, M., Kunz, J., Paulson, B.
14	Document Building project model support for automated labor monitoring	2003	Sacks, R., Navon, R., Goldschmidt, E.
15	Statistical-fuzzy approach to quantify cumulative impact of change orders	2002	Hanna, A.S., Lotfallah, W.B., Lee, MJ.
16	Production scheduling for precast plants using a flow shop sequencing model	2002	Chan, W.T., Hu, H.
17	Estimating labor productivity using probability inference neural network	2000	Lu, M., Abourizk, S.M., Hermann, U.H.
18	Design-integrated process planner for rebar placement	1995	Salim, M., Bernold, L.E.
19	Estimating construction productivity: Neural-network-based approach	1994	Chao, LC., Skibniewski, M.J.
20	Speech-based data-entry systems for construction	1993	Bernold, L.E.

11. International Journal of Productivity and Performance Management

No.	Title	Year	Author (s)
1	Factors influencing labor productivity on construction sites: A state-of-the-art literature review and a survey	2016	Naoum, S.G.
2	Influence of occupational heat stress on labour productivity – a case study from Chennai, India	2016	Chinnadurai, J., Venugopal, V., Kumaravel, P., Paramesh, R.
3	Converging early contractor involvement (ECI) and lean construction practices for productivity enhancement some preliminary findings from Singapore	2015	Pheng, L.S., Gao, S., Lin, J.L.
4	Labour productivity in Iranian construction projects perceptions of chief executive officers	2015	Ghoddousi, P., Poorafshar, O., Chileshe, N., Hosseini, M.R.
5	Prominent demotivational factors influencing the productivity of construction project managers in Qatar	2014	Jarkas, A.M., Radosavljevic, M., Wuyi, L.

12. Construction Innovation (CI)

No.	Title	Year	Author
1	Labour productivity model for reinforced concrete construction projects	2011	Jang, H., Kim, K., Kim, J., Kim, J.
2	Analysis of labour productivity of formwork operations in building construction	2010	Moselhi, O., Khan, Z.
			44

3	Estimating productivity using simulation: A case study of Gaza beach embankment protection project	2007	Rustom, R.N., Yahia, A.
---	--	------	----------------------------

13. Australasian Journal of Construction Economics and Building (AJCEB) or CEB

No.	Title	Year	Author (s)
	Drivers of productivity: A case study of the		
1	Australian construction industry	2015	Chancellor, W.
	Factors promoting innovation and efficiency in		
	the construction industry: A comparative study		Chancellor, W., Abbott,
2	of New Zealand and Australia	2015	M., Carson, C.
	Document Strategies for effective management		
	of health and safety in confined site		
3	construction	2013	Spillane, J., Oyedele, L.
	International comparisons of cost and		
4	productivity in construction: A bad example	2012	Best, R.

14. Journal of Civil Engineering and Management (JCiEM)

No.	Title	Year	Author (s)
1	Evaluating motivation of construction workers: a comparison of fuzzy rule-based model with the traditional expectancy theory	2016	Yeheyis, M., Reza, B., Hewage, K., Ruwanpura, J.Y., Sadiq, R.
2	An it model of a knowledge map which supports management in small and medium- sized companies using selected polish construction enterprises as an example	2015	Hoła, B., Sawicki, M., Skibniewski, M.
3	Predicting productivity loss caused by change orders using the evolutionary fuzzy support vector machine inference model	2015	Cheng, MY., Wibowo, D.K., Prayogo, D., Roy, A.F.V.
4	Modelling masonry crew productivity using two artificial neural network techniques	2015	Gerek, I.H., Erdis, E., Mistikoglu, G., Usmen, M.
5	Measuring information dependency for construction engineering projects	2013	Liao, PC., Thomas, S.R., O'brien, W.J.
6	Benchmarking project level engineering productivity	2012	Liao, PC., Thomas, S.R., O'Brien, W.J., (), Mulva, S.P., Kim, I.
7	Document Technical comparisons of simulation-based productivity prediction methodologies by means of estimation tools focusing on conventional earthmovings	2011	Han, S., Hong, T., Kim, G., Lee, S.
8	The impacts of buildability factors on formwork labour Productivity of columns	2010	Jarkas, A.M.
9	Organizational effectiveness of ugandan building firms as viewed by craftsmen	2009	Alinaitwe, H., Mwakali, J.A., Hansson, B.

444

	Effect of basic motivational factors on construction workforce productivity in		Kazaz, A., Manisali, E.,
10	Turkey	2008	Ulubeyli, S.
	An assessment of clients' performance in having an efficient building process in		
11	Liganda	2000	Alinaitwa II M
11	Uganda	2008	Allhaltwe, H.M.
			Enshassi, A., Mohamed,
	Factors affecting labour productivity in		S., Mustafa, Z.A., Mayer,
12	building projects in the Gaza strip	2007	P.E.
			Alinaitwe, H.M.,
	Factors affecting the productivity of building		Mwakali, J.A., Hansson,
13	craftsmen - Studies of Uganda	2007	B.
	Mathematical-neural model for assessing		
14	productivity of earthmoving machinery	2007	Schabowicz, K., Hola, B.
	Assessing the degree of industrialisation in		Alinaitwe, H.M.,
15	construction - A case of Uganda	2006	Mwakali, J., Hansson, B.

15. Structural Survey (SS)

No.	Title	Year	Author (s)
1	On-site construction productivity in Malaysian infrastructure projects	2016	Durdyev, S., Ismail, S.
2	Factors affecting construction labour productivity for Malaysian residential projects	2005	Abdul Kadir, M.R., Lee, W.P., Jaafar, M.S., Sapuan, S.M., Ali, A.A.A.
3	Quantifying the relationships between buildability, structural quality and productivity in construction	2001	Pheng Low, S.
4	A linear programming decision tool for selecting the optimum excavator	2001	Edwards, D.J., Malekzadeh, H., Yisa, S.B.

16. Journal of Infrastructure Systems (JIS)

No.	Title	Year	Author (s)
1	An integrated productivity-practices implementation index for planning the execution of infrastructure projects	2016	Nasir, H., Haas, C.T., Caldas, C.H., Goodrum, P.M.
2	Framework for modeling on-site productivity of preventive maintenance activities for wastewater collection systems	2015	Zaman, H., Bouferguene, A., Al-Hussein, M., Lorentz, C.
3	Productivity impact of infrastructure in Turkey, 1987-2006	2014	Fedderke, J.W., Kaya, T. E.
4	Evaluation of On-Site Fuel Use and Emissions over the Duration of a Commercial Building Project	2012	Rasdorf, W., Lewis, P., Marshall, S.K., Arocho, I., Frey, H.C.

5	Critical Assessment Indicators for Measuring Benefits of Rural Infrastructure Investment in China	2012	Shen, L., Lu, W., Peng, Y., Jiang, S.
6	Cost structure and efficiency of korea's road and rail in the manufacturing industries	2011	Min, S.
7	Dynamic life-cycle modeling of pavement overlay systems: Capturing the impacts of users, construction, and roadway deterioration	2010	Zhang, H., Lepech, M.D., Keoleian, G.A., Qian, S., Li, V.C.
8	Efficient use of regional transport infrastructure, communication networks, and human capital	2009	Schaffer, A., Siegele, J.
9	Subway station diagnosis index condition assessment model	2009	Semaan, N., Zayed, T.
10	Infrastructure management: Integrated AHP/ANN model to evaluate municipal water mains' performance	2008	Al-Barqawi, H., Zayed, T.
11	Method for comparative performance assessment and evaluation of consolidating community water systems as a regional water system	2007	Rogers, R.J., Louis, G.E.
12	Evaluation of trenchless technology methods for municipal infrastructure system	2007	Jung, Y.J., Sinha, S.K.
13	Minimizing total cost for urban freeway reconstruction with integrated construction/traffic analysis	2005	Lee, EB., Ibbs, C.W., Thomas, D.
14	Estimation on regional benefit and optimal level of road capital stock	2002	Kim, E., Shin, M.
15	Cost/benefits of robotics in infrastructure and environmental renewal	2000	Gregory, R.A., Kangari, R.
16	Infrastructure and economic development: Airport capacity in Chicago region, 2001-18	1997	Hewings, G.J.D., Schindler, G.R., Israilevich, P.R.
17	Vehicle size and weight regulations and highway infrastructure management	1997	Fekpe, E.S.K.
18	Infrastructure's impact on development: Lessons from WDR 1994	1995	Kessides, C., Ingram, G.K.

17. Additional Articles

No.	Title	Year	Author (s)
1	Productivity improvements: understand the workforce perceptions of productivity first	2007	Chan, P. W. & Kaka, A
2	On-site labour productivity of New Zealand construction industry: Key constraints and improvement measures.	2011	Durdyev, S. & Mbachu, J.
3	A survey of the factors affecting the productivity of construction projects in Iran	2012	Ghoddousi, P. & Hosseini, M. R.
4	A review of enabling factors in construction industry productivity in an Australian environment	2014	Hughes, R. & Thorpe, D.
5	Construction Productivity Improvement: A Study of Human, Management and External Issues	2003	Liberda, M., Ruwanpura, J. & Jergeas, G.
6	Critical factors influencing construction productivity in Thailand	2004	Makulsawatudom, A., Emsley, M. & Sinthawanarong, K.

18. Final List of Articles

Journal	No.	Title	Year	Author(s)
	1	Effect of buildability on labor productivity: A practical quantification approach	2016	Jarkas, Abdulaziz M.
	2	Applying Artificial Neural Networks for Measuring and Predicting Construction- Labor Productivity	2015	Heravi, Gholamreza;Eslamd oost, Ehsan
	3	Method to assess the level of implementation of productivity practices on industrial projects	2015	Caldas, Carlos H.;Kim, Jungyeol;Haas, Carl T M;Goodrum, Paul M.; Zhang, Di
	4	Factors affecting construction labor productivity in Kuwait	2014	Jarkas, Abdulaziz M.;Bitar, Camille G.
CEM	5	Assessment of the productivity of night time asphalt paving operations	2012	Mostafavi, A., Valentin, V., Abraham, D.M., Louis, J.
ſ	6	Buildability factors influencing concreting labor productivity	2012	Jarkas, Abdulaziz M.
	7	Streamlining the construction productivity improvement process with the proposed role of a construction productivity improvement officer	2012	Ranasinghe, Upul,Ruwanpura, Janaka Y.;Liu, Xin
	8	Influence of buildability factors on rebar installation labor productivity of columns	2012	Jarkas, Abdulaziz M.
	9	Differences in perspectives regarding labor productivity between spanish-and english- speaking craft workers	2011	Dai, Jiukun;Goodrum, Paul M.
	10	Improving productivity on a troubled	2011	Minchin, Robert

		bridge project		Edward;Lewis, Don W.;McLeod, Luke
-	11	Analysis of factors influencing productivity using craftsmen questionnaires: Case study in a Chilean construction company	2011	Rivas, Rodrigo A.;Borcherding, John D;Alarcn, Luis Fernando:Gonzlez, Vicente
-	12	Latent structures of the factors affecting construction labor productivity	2009	Dai, Jiukun;Goodrum, Paul M.;Maloney, William F. ;Srinivasan, Cidambi
-	13	Relationship between changes in material technology and construction productivity	2009	Goodrum, Paul M.;Zhai, Dong ;Yasin, Mohammed Fadzil
	14	Construction craft workers' perceptions of the factors affecting their productivity	2009	Dai, Jiukun;Goodrum, Paul M. ;Maloney, William F.
	15	Impact of shift work on labor productivity for labor intensive contractor	2008	Hanna, Awad;Chang, Chulki;Sullivan, Kenneth Timothy;Lackney, Jeffery A.
-	16	Quantifying the impact of schedule compression on labor productivity for mechanical and sheet metal contractor	2007	Chang, Chulki ;Hanna, Awad;Lackney, Jeffery A.;Sullivan, Kenneth Timothy
	17	Impact of overmanning on mechanical and sheet metal labor productivity	2007	Hanna, Awad;Chang, Chulki;Lackney, Jeffery A;Sullivan, Kenneth Timothy
	18	Fundamental principles of workforce management	2006	Thomas, Herve Randolph;Horman, Michael J.
	19	Contractor process improvement for enhancing construction productivity	2006	Cottrell, David S.
	20	Impact of change's timing on labor productivity	2005	Ibbs, W.
	21	Factors Affecting Absenteeism in Electrical Construction	2005	Hanna, Awad;Menches, Cindy L;Sullivan, Kenneth Timothy;Sargent, Joseph R.
	22	Fundamental principles of site material management	2005	Thomas, Herve Randolph;Riley, David R. ;Messner, John I
	23	Impact of extended overtime on construction labor productivity	2005	Hanna, A.S., Taylor, C.S., Sullivan, K.T.
	24	Change orders impact on labor productivity	2005	Moselhi, Osama El Sayed;Assem, Ihab;El- Rayes, Khaled Rayes
	25	Cumulative Effect of Project Changes for Electrical and Mechanical Construction	2004	Hanna, A.S., Camlic, R., Peterson, P.A., Lee, MJ.
	26	Impact of change orders on small labor- intensive projects	2004	Hanna, A.S., Gunduz, M.
	27	Long-term impact of equipment technology on labor productivity in the U.S. construction industry at the activity level	2004	Goodrum, Paul M.;Haas, Carl T M
	28	Partial factor productivity and equipment technology change at activity level in U.S. construction industry	2002	Goodrum, Paul M.;Haas, Carl T M
	29	Role of the fabricator in labor productivity	2000	Thomas, H.R., Sanvido, V.E.

	30	Schedule acceleration, work flow, and labor productivity	2000	Thomas, H.R.
	31	Loss of labor productivity due to delivery methods and weather	1999	Thomas, H.R., Riley, D.R., Sanvido V E
	32	Impact of change orders on labor efficiency for mechanical construction	1999	Hanna, A.S., Russell, J.S., Gotzion, T.W., Nordheim,
	33	Construction labor productivity modeling with neural networks	1998	E.v. Sonmez, R., Rowings, J.E.
	34	Quantitative effects of construction changes on labor productivity	1995	Thomas, H.R., Napolitan, C.L
	35	Action-response model and loss of productivity in construction	1994	Halligan, D.W., Demsetz, L.A., Brown, J.D., Pace, C.B.
	36	Forecasting labor productivity using factor model	1994	Thomas, H.R., Sakarcan, A.S.
CME	37	An exploratory study of the relationship between construction workforce physical strain and task level productivity	2014	Gatti, U.C., Migliaccio, G.C., Bogus, S.M., Schneider, S.
	38	The impact of management practices on mechanical construction productivity	2011	Shan, Yongwei ;Goodrum, Paul M. ;Zhai, Dong;Haas, Carl T M ;Caldas, Carlos H.
	39	The influence of buildability factors on rebar fixing labour productivity of beams	2010	Jarkas, A.M.
	40	Application of AHP in improving construction productivity from a management perspective	2008	Doloi, H.
	41	Analysis of craft workers' and foremen's perceptions of the factors affecting construction labour productivity	2007	Dai, Jiukun ;Goodrum, Paul M.; Maloney, William F.
	42	The JIT materials management system in developing countries	2005	Polat, G., Arditi, D.
	43	The impact of non- financial incentives on bricklayers' productivity in Nigeria	2004	Fagbenle, O.I., Adeyemi, A.Y., Adesanya, D.A.
	44	Implementing a multiskilled workforce	2001	Haas, T.C., Rodriguez, A.M., Glover, R., Goodrum, P.M.
	45	Trends in productivity improvement in the US construction industry	2000	Arditi, D., Mochtar, K.
	46	The impact of change orders on mechanical construction labour efficiency	1999	Hanna, A.S., Russell, J.S., Vandenberg, P.J.
	47	Construction resource/method factors influencing productivity for high rise concrete construction	1999	Proverbs, D.G., Holt, G.D., Olomolaiye, P.O.
	48	A survey of constraints on Iranian construction operatives' productivity	1996	Zakeri, M., Olomolaiye, P.O., Holt, G.D., Harris, F.C.
	49	Productivity improvement in the Indonesian construction industry	1996	Arditi, D., Mochtar, K.
	50	Research of factors influencing construction productivity	1990	Herbsman, Z., Ellis, R.
	51	An evaluation of the relationships between bricklayers' motivation and productivity	1990	Olomolaiye, P.O.
	52	A system for monitoring and improving construction operative productivity in Nigeria	1989	Olomolaiye, P.O., Ogunlana, S.O.

449
·				
	53	Trust and productivity in Australian construction projects: A subcontractor	2016	Chalker, M., Loosemore, M.
M	54	Improving construction productivity: A subcontractor's perspective	2014	Loosemore, M.
BCA	55	Contractors perspective toward factors	2013	Mahamid, I.
н		affecting labor productivity in building		
	50	construction	1007	
	56	Do site managers and the head office	1996	Naoum, S., Hackman, J.
	57	Statistical Analysis of the Effectiveness of	2016	Shan, Y., Zhai, D.,
		Management Programs in Improving		Goodrum, P.M., Haas, C.T.,
		Construction Labor Productivity on Large		Caldas, C.H.
	59	Industrial Projects	2014	El Cohora V.M. Aziz D.E.
ЧЕ	20	productivity in Egypt	2014	EI-Gollary, K.M., AZIZ, K.F.
F,	59	Motivational factors impacting the	2013	Jarkas, A.M., Radosavljevic,
		productivity of construction master		M.
	60	craftsmen in Kuwait	• • • •	
	60	Labor productivity drivers and	2003	Rojas, E.M., Aramvareekul,
	61	Construction industry productivity and the	2014	F. Fulford R Standing C
	01	potential for collaborative practice	2011	i unora, ic., Standing, C.
	62	Dynamic modeling of labor productivity in	2013	Nasirzadeh, F., Nojedehi, P.
		construction projects		
	63	Major factors influencing productivity of	2008	Mojahed, S., Aghazadeh, F.
		construction: Evidence from the deep south		
		USA		
Md	64	Demotivating factors influencing the	2004	Ng, S.T., Skitmore, R.M.,
II I	(5	productivity of civil engineering projects	1000	Lam, K.C., Poon, A.W.C.
	65	severity diagnosis of productivity	1998	Kaming, P.F., Holt, G.D., Kometa, S.T., Olomolaive
		problems - A renability analysis		P.O.
	66	Factors influencing craftsmen's	1997	Kaming, P.F., Olomolaiye,
		productivity in Indonesia		P.O., Holt, G.D., Harris,
	67	Issues encountered by contractors in	1005	F.C.
	07	Singapore	1995	Alulli, J.
	68	Measuring the impact of BIM on labor	2015	Poirier, E.A., Staub-French,
		productivity in a small specialty		S., Forgues, D.
C		contracting enterprise through action-		
AI	60	research Assessing the impact of materials tracking	2000	Grau D. Caldas C.H.
	09	technologies on construction craft	2009	Haas, C.T., Goodrum, P.M.,
		productivity		Gong, J.
П	70	Project managers' perception of production	1996	Kaming, P.F., Olomolaiye,
BR		problems - An Indonesian case study:		P.O., Holt, G.D., Kometa,
	71	Evaluating the impacts of high-temperature	2016	5.1., панія, г.С. Li X Chow K H Zhu V
	/ 1	outdoor working environments on	2010	Lin, Y.
		construction labor productivity in China: A		
Щ		case study of rebar workers	• • • =	TT 1 TT 1 1 ~
BA	72	Drivers of productivity among construction	2007	Kazaz, A., Ulubeyli, S.
	73	A quantitative approach for evaluation of	2004	Gunduz M
	15	negative impact of overmanning on	2004	Sunuuz, 191.
		electrical and mechanical projects		

	74	JIT in developing countries-a case study of the Turkish prefabrication sector	2003	Oral, E.L., Mistikoglu, G., Erdis, E.
	75	Thermal environment and construction workers' productivity: Some evidence from Thailand	2003	Srinavin, K., Mohamed, S.
	76	Identification and comparative analysis of key parameters influencing construction labour productivity in building and industrial projects	2014	Tsehayae, A.A., Fayek, A.R.
	77	Generational differences on craft workers' perceptions of the factors affecting labour productivity	2012	Dai, J., Goodrum, P.M.
	78	Buildability factors affecting formwork labour productivity of building floors	2010	Jarkas, A.M.
CJCE	79	Impact of occasional overtime on construction labor productivity: Quantitative analysis	2007	Sonmez, R.
	80	Carpentry workers issues and efficiencies related to construction productivity in commercial construction projects in Alberta	2006	Hewage, K.N., Ruwanpura, J.Y.
	81	Factors affecting construction productivity: Newfoundland versus rest of Canada	1994	Hanna, A.S., Heale, Donald G.
	82	Document Impact of change orders on construction productivity	1991	Moselhi, Osama; Leonard, Charles; Fazio, Paul
CE	83	Decision tree approach to classify and quantify cumulative impact of change orders on productivity	2004	Lee, MJ., Hanna, A.S., Loh, WY.
JC	84	Statistical-fuzzy approach to quantify cumulative impact of change orders	2002	Hanna, A.S., Lotfallah, W.B., Lee, MJ.
	85	Factors influencing labor productivity on construction sites: A state-of-the-art literature review and a survey	2016	Naoum, S.G.
Md	86	Influence of occupational heat stress on labour productivity – a case study from Chennai, India	2016	Chinnadurai, J., Venugopal, V., Kumaravel, P., Paramesh, R.
Idli	87	Labour productivity in Iranian construction projects perceptions of chief executive officers	2015	Ghoddousi, P., Poorafshar, O., Chileshe, N., Hosseini, M.R.
	88	Prominent demotivational factors influencing the productivity of construction project managers in Qatar	2014	Jarkas, A.M., Radosavljevic, M., Wuyi, L.
	89	Predicting productivity loss caused by change orders using the evolutionary fuzzy support vector machine inference model	2015	Cheng, MY., Wibowo, D.K., Prayogo, D., Roy, A.F.V.
	90	The impacts of buildability factors on formwork labour Productivity of columns	2010	Jarkas, A.M.
JCiEM	91	Effect of basic motivational factors on construction workforce productivity in Turkey	2008	Kazaz, A., Manisali, E., Ulubeyli, S.
	92	Factors affecting labour productivity in building projects in the Gaza strip	2007	Enshassi, A., Mohamed, S., Mustafa, Z.A., Mayer, P.E.
	93	Factors affecting the productivity of building craftsmen - Studies of Uganda	2007	Alinaitwe, H.M., Mwakali, J.A., Hansson, B.
SS	94	On-site construction productivity in Malaysian infrastructure projects	2016	Durdyev, S., Ismail, S.

	95	Factors affecting construction labour productivity for Malaysian residential projects	2005	Abdul Kadir, M.R., Lee, W.P., Jaafar, M.S., Sapuan, S.M., Ali, A.A.A.
JiS	96	An integrated productivity-practices implementation index for planning the execution of infrastructure projects	2016	Nasir, H., Haas, C.T., Caldas, C.H., Goodrum, P.M.
	97	Productivity improvements: understand the workforce perceptions of productivity first	2007	Chan, P. W. & Kaka, A
	98	On-site labour productivity of New Zealand construction industry: Key constraints and improvement measures.	2011	Durdyev, S. & Mbachu, J.
al papers	99	A survey of the factors affecting the productivity of construction projects in Iran	2012	Ghoddousi, P. & Hosseini, M. R.
Addition	100	A review of enabling factors in construction industry productivity in an Australian environment	2014	Hughes, R. & Thorpe, D.
	101	Construction Productivity Improvement: A Study of Human, Management and External Issues	2003	Liberda, M., Ruwanpura, J. & Jergeas, G.
	102	Critical factors influencing construction productivity in Thailand	2004	Makulsawatudom, A., Emsley, M. & Sinthawanarong, K.

APPENDIX 9: ACCEPTANCE LETTERS AND PUBLISHED PAPER

A. Journal of Construction Engineering and Management

Date: To: cc: From: Subject:	05/03/2017 "Argaw Tarekegn Gurmu" argtar1982@gmail.com aaibinu@unimelb.edu.au "Journal of Construction Engr. & Management" journal-submissions5@asce.org Decision on Manuscript MS COENG-5825R2						
ACCEPT FI Ref.: Ms. Investigati Storey Bui	NAL No. COENG-5825R2 ion of Construction Equipment Management Practices for Improving Labour Productivity in Multi- lding Construction Projects						
Dear Mr. G Your Techr	Surmu, M.S.C.; Alibade Ayodeji Albinu, Ph.D Surmu, nical Paper, listed above, has been accepted for publication in ASCE's Journal of Construction						
Engineerin Thank you will now be within you	Engineering and Management. Thank you for your contribution to the journal of Journal of Construction Engineering and Management. You will now be listed as a reviewer in the journals database. We ask that you review two papers within a year within your field of studies.						
Your manu notified of	Your manuscript will now be forwarded to a Production Editor who will prepare it for publication. You will be notified of a publication date once your paper has been schedule for an issue.						
Thank you	Thank you for submitting your work to ASCE's Journal of Construction Engineering and Management.						
Sincerely,							
Taryn Dolli Editoria l C	ings oordinator						

B. International Journal of Productivity and Performance Management

Decision Letter (IJPPM-02-2017-0032,R1)

From: johnheap@yahoo.com

- To: argtar1982@gmail.com, aaibinu@unimelb.edu.au
- cc:

Subject: International Journal of Productivity and Performance Management - Decision on Manuscript ID IJPPM-02-2017-0032.R1

Body: 29-May-2017

Dear Mr. GURMU:

It is a pleasure to accept your manuscript entitled "Survey of Management Practices Enhancing Labour Productivity in Multi-Storey Building Construction Projects" in its current form for publication in International Journal of Productivity and Performance Management,

By publishing in this journal, your work will benefit from Emerald EarlyCite. This is a pre-publication service which allows your paper to be published online earlier, and so read by users and, potentially, cited earlier,

Please go to your Author Centre at https://mc.manuscriptcentral.com/ijppm (Manuscripts with Decisions for the submitting author or Manuscripts I have coauthored for all listed co-authors) to complete the copyright assignment form. We cannot publish your paper without this. All authors are requested to complete the form and to input their full contact details. If any of the contact information is incorrect you can update it by clicking on your name at the top right of the screen. Please note that this must be done prior to you submitting your copyright form. If you would like more information about Emerald's copyright policy, please visit the Information & Forms section in your Author Centre. If you have an ORCID please check your account details to ensure that your ORCID is validated,

FOR OPEN ACCESS AUTHORS: Please note if you have indicated that you would like to publish your article as Open Access via Emerald's Gold Open Access route, you are required to complete a Creative Commons Attribution Licence - CCBY 4.0 (in place of the standard copyright assignment form referenced above), You will receive a follow up email within the next 30 days with a link to the CCBY licence and information regarding payment of the Article Processing Charge. If you have indicated that you might be eligible for a prepaid APC voucher, you will also be informed at this point if a voucher is available to you (for more information on APC vouchers please see http://www.emeraldpublishing.com/oapartnerships

Thank you for your contribution. On behalf of the Editors of International Journal of Productivity and Performance Management, we look forward to your continued contributions to the Journal.

Yours sincerely, Mr. John Heap Editor, International Journal of Productivity and Performance Management johnheap@yahoo.com

C. Construction Economics and Building (Next Page)



A study of best management practices for enhancing productivity in building projects: construction methods perspectives

Argaw Tarekegn Gurmu¹, Ajibade Ayodeji Aibinu and Toong Khuan Chan

Melbourne School of Design, The University of Melbourne, Australia

Abstract

This research investigates management practices that have the potential to enhance productivity in building projects by focusing on construction methods. In phase 1 of the study, face-to-face interviews with nineteen experts were conducted to identify the best management practices for construction methods. The qualitative data analysis reached saturation and resulted in a list of best practices for construction methods that are relevant to the local industry. The second phase used an industry-wide survey to prioritize the best practices. Accordingly, project start-up plan, traffic control plan, machinery positioning strategy, project completion plan, and dynamic site layout plan were shown to be the top five best practices for construction methods. The study also revealed that high levels of implementation of best practices are associated with low levels of project delays. The use of best practices also varied according to the project costs. There were no discernible differences between the top five best practices. The authors suggest that they should be implemented jointly to improve productivity in building projects. Contractors could use the logistic regression model developed, to predict the probability of exceeding a baseline productivity factor and, on that basis, implement corrective actions to achieve the desired level of productivity.

Keywords: Best management practices, construction methods, construction productivity, Australia.

Paper type: Research article

Introduction

Productivity improvement in the construction industry is important for the growth of a country's economy at large and profitability of contractors in particular. Productivity is described as the ratio of output to input (Jarkas and Bitar, 2011). In this study, output represents the value of a completed building project in dollars and input refers to project

Copyright: Construction Economics and Building 2016. © 2016 Argaw Gurmu, Ajibade Aibinu and Toong Chan. This is an Open Access article distributed under the terms of the Creative Commons Attribution 4.0 Unported (CC BY 4.0) License (https://creativecommons.org/licenses/by/4.0/), allowing third parties to copy and redistribute the material in any medium or format and to remix, transform, and build upon the material for any purpose, even commercially, provided the original work is properly cited and states its license.

Citation: Gurmu, A.T., Aibinu, A.A. and Chan, T.K. 2016. A study of best management practices for enhancing productivity in building projects: construction methods perspectives, *Construction Economics and Building*, 16(3), 1-19. DOI: http://dx.doi.org/10.5130/AJCEB.v16i3.4882

Corresponding author: Argaw Gurmu; Email - agurmu@student.unimelb.edu.au



duration in days. Previous studies indicate that management related problems such as lack of integration of works of sub-contractors, out of sequence work assignments, work overload, and poor site layouts are among the major factors impacting project productivity (Dai et al., 2009). However, these problems can be reduced by adopting best management practices that are suitable for particular project types such as building, infrastructure, and industrial. Best management practices are processes that, when executed effectively, lead to enhanced project performance (CII, 2016). Construction management practices that have the potential to improve productivity could be categorized into: construction methods (Caldas et al., 2014); construction materials management (long lead materials identification, materials procurement and delivery plans, and inspection and test plans) (Bell and Stukhart, 1987); construction equipment and tools management (procurement plans for machinery, maintenance of equipment and tools and productivity analysis of construction equipment) (Stewart, 2002); execution approaches (buildability review, short interval plans, work package, and the scope of works) (Lam, Wong and Chan, 2006); human resource management (crew composition, skill assessment, training and career development plans) (Hewage, Gannoruwa and Ruwanpura, 2011); health and safety practices (housekeeping, task safety analysis, tool box meetings, and safety training) (Hinze and Wilson, 2000). In this research, construction methods best practices were investigated while pre-construction activities such as buildability reviews were not considered. Although previous studies identified best practices for infrastructure and industrial projects, they cannot be directly adopted for building projects. For instance, while traffic control plan, project completion plan, communications, coordination, and agreements are identified as best practice for infrastructure projects, they are not included in the list of best practices for industrial projects (Nasir, 2013). Moreover, the priority given to some of the common elements of both project types are different. Thus, building projects could have different best practices. However, a little research has been done on what these best practices might be and which of them should be given priority in the context of Victoria, Australia. The objectives of this study are:

- To identify best practices for construction methods which have the potential to improve productivity in building projects
- To prioritize the best practices for construction methods and develop a tool to measure them
- To develop a logistic regression model that predicts the probability of exceeding a specific productivity value based on a score of the best practices for construction methods

Literature review

Construction methods influencing productivity in construction projects

Project management methods are defined as a system of practices, techniques, procedures, and rules used by those who work in the discipline (PMI, 2013). In construction projects, the techniques of integration of different schedules; schedule controlling methods; mechanisms used in the preparation of site layout, project start-up and completion

procedures; and investigation of suitable technologies are some of the methods considered as best practices.

Arditi and Mochtar (1996) suggested integration of management functions is one of the areas where productivity gain could be obtained. Caldas et al. (2014) confirmed that integrated schedule is one of the best practices for improving productivity in industrial projects. In the context of the Victorian construction industry, since many sub-contractors are involved in the construction of a particular building project, principal contractors might require techniques for integrating the schedules of these sub-contractors. According to ABS, 78% of building construction works are executed by small and medium firms (Australian Bureau of Statistics, 2013).

Scheduled overtime has a positive impact on productivity, as it builds the morale of employees (CII, 2013a). Hanna et al. (2008) found that shift schedule is effective as compared to overtime and overmanning in reducing the project duration. The authors opined that the use of shift schedule reduces physical fatigue and site congestion that could occur if overtime and overmanning practices are implemented. Given the working hours restrictions set by various authorities in Victoria, such as Environmental Protection Authority, adopting different working hours strategies such as scheduled overtime could be important practices for delivery of building projects within the contract time.

Schedule execution and management refers to schedule controlling techniques such as methods for measuring work progress, analysing data, reporting results and taking corrective actions. Some of the techniques include units completed, incremental milestone, start/finish, supervisor judgment, S-Curve, time variance and forecast, analysis tree, percent complete report, activity crashing and activity overlapping (Attalla, 1997). Delay in the progress of a particular sub-contractor could affect the progress of others and lead to overall project delay. Thus, implementation of the schedule controlling techniques might be an important practice for principal contractors that manage numerous sub-contractors in the context of Victoria.

Dynamic site layout refers to a sequence of layouts each of which is used for a discrete time interval or for a certain project phase, and together covering the entire duration of a construction project (Tommelein and Zouein, 1993). As most building projects in the central business district of Melbourne, Australia, have restricted working spaces, dynamic site layout plans could help principal contractors in reducing congestion on sites. By using dynamic layout plan, some parts of a building under construction can be used as store, office and other temporary facility allowing early start of the external works.

Choi and Harris (1992) proposed a mathematical model for determining the most suitable tower crane location for building projects. Safe Work Australia implemented a code of practice to assist contractors in preparing good traffic management plans by providing information about traffic signs, distance between pedestrians and vehicles, and vehicles movement (Safe Work Australia, 2014). In Victoria, as most construction materials are imported from overseas, any loss or damage to them could incur loss of productivity due to unavailability of the materials locally. Thus, site security plans might be an important practice to reduce theft and loss of materials. Investigation of the most appropriate location for a crane could also be a significant practice in Victoria. If a crane is positioned



wrongly, its relocation cost could be high. Thus, developing a strategy to position the crane using different models might be important for principal contractors in Victoria, to reduce cost and increase productivity. Traffic control plans could also be a significant practice as there are various local regulations about traffic such as Road Safety (Traffic Management) Regulations 2009 that could influence building projects' performance.

Fangel (1984) recommended preparation of a project management manual that contains project start-up details. The author also suggested project start-up meetings prior to commencing a construction project. Kerzner (2010) proposed project kick-off meetings as one of the best practices that should be included in the project start-up process. Nasir (2013) confirmed that project start-up and completion plans, and new technologies are the best practices to enhance productivity in infrastructure projects. In the context of building projects in Victoria, various sub-contractors finish their works at different times and a principal contractor. Thus, completion plans could be a significant practice during handover of a building project to the client. New technologies might also be important. For instance, to detect clashes between different services in a building, different software could be used.

Research methodology

The research context

The construction sector in Victoria, Australia is dominated by a few large contractors that engage numerous small companies. In 2015, the proportion of companies involved in building construction were 0.06%,0.78% and 99.15% for firms employing over 200 workers, 20-199 workers, and 0-19 workers respectively (Australian Bureau of Statistics, 2016). The local construction industry is characterized by the presence of a strong Construction, Forestry, Mining and Energy Union that prepares a calendar for construction sites (CFMEU, 2016). The Union signs an enterprise bargaining agreement in which minimum payments and other working conditions are agreed between contractor and Union. Regulatory bodies such as the Fair Work Commission stipulate minimum wages, working hours, overtime payments, penalty rates and other employment conditions in building construction projects. Accordingly, the ordinary number of working hours is 38 per week between 7:00am and 6:00pm (Fair Work Commission, 2016). In addition, Environment Protection Authority Victoria (EPA) has a guideline to control noise from building projects. Accordingly, normal working hours are restricted to 7:00 am to 6:00 pm during weekdays and 7:00 am to 1:00 pm on Saturdays (EPA, 2016). Most construction materials are imported from overseas due to increasing investment costs in the manufacturing sector (Wheeldon, 2012). Prefabricated construction systems are increasing at about 5% per year (PrefabAUS, 2014).

Measurement of management practices for construction methods and construction projects' productivity

Management practices are measured by using validated standard questionnaires. Bloom and Van Reenen (2007) developed a technique to measure management practices in the 458



manufacturing, education, and health sectors. The authors validated their survey tool by collecting data from 732 firms operating in Germany, France, UK and US. In the construction industry, management practices questionnaires have been developed by the Construction Industry Institute (CII) through the collaboration of researchers and industry practitioners. The tool has been validated by collecting data from various projects in North America. This research uses CII Survey Tool (Appendix- 1B2) after conducting validation by discussing with local experts, and collecting productivity data from various building projects.

Construction projects' productivity can be expressed in either absolute or relative terms. In absolute term, the units of measure of both output and input are shown in the productivity value. The relative measure or Productivity Factor (PF) refers to the ratio of actual to planned productivity. Actual productivity is computed using project value as output and actual completion time as input. Planned productivity is calculated using project value as output and planned time as input. Productivity factor (PF) is a more useful measure to compare the productivity of different construction projects than the absolute measure (Nasir, 2013).Thus; PF is used in this research.

Data collection

This study was conducted in two phases. In phase 1, qualitative data was collected using indepth interviews that took an average of one and half hours. The objective of this stage was to identify the best practices for construction methods using knowledge and experience of local construction experts. Nineteen professionals who have been working with the prequalified principal contractors in Victoria, Australia and having experience ranging from five to forty years were interviewed (Table 1). The experts have been working as general manager, construction manager, project manager, project coordinator, project engineer, site engineer, contract administrator, supervisor and cost manager. Snowballing technique was used to select the participants for interviews. Moreover, the experts were selected based on their experience in working with exemplar contractors. The exemplars were principal contractors that had the capacity to deliver numerous projects within a fiscal year. They were chosen based on the companies information provided in May 2015, IBIS World Report. Semi-structured questions were prepared for the interview. The questions included: Does this practice exist? How is it practiced by local contractors? Is this practice best for improving the productivity of building projects? What other practices enhance the productivity of building projects? The interviews were conducted until the data analysis reached saturation point. Saturation refers to the point where similar reasons for accepting or rejecting a particular best practice were given by the participants.

Table 1: Interviewees' years of experience in construction industry

Experience	Number
0-5	0
5-10	6
10-15	5
15-20	2
20+	6
Total	19

CUTS

In phase 2, an industry-wide questionnaire survey was conducted (Appendix-1). The objective of this stage was to prioritize best practices for construction methods; develop a scoring tool and a logistic regression model. Thirty-nine pre-qualified principal contractors, based on their managerial capability, financial capacity, expertise and experience, were selected as the units of analysis (Department of Treasury and Finance, 2015). Victorian construction industry is comprised of a few principal contractors, all of whom were considered in this study. After identification of the unit of analysis, professionals who have been involved in the construction of building projects were contacted to respond to the questionnaires. General Managers, construction directors, construction managers and project managers, project coordinators and site managers with an average of fifteen years of experience participated in the survey that was administered using face-to-face interviews in which the researchers clarified the questions and took notes. The experts were asked to rate the relative importance of each element in improving productivity of building construction projects based on their experience in the industry using a 1 to 5 response scale(Appendix-1A). Moreover, they were requested to indicate the level of implementation of the best practices for construction methods on a particular building project that had been completed within the previous five years (Appendix-1B).

Mixed methods were used in this study. Since management practices could vary from country to country and from project to project, interviews were used to investigate the context-specific best practices during Phase I. A constructivist paradigm was adopted during stage I as there is no single best management practice. Different construction industries have their own best practices and no single best practice exists. Thus, contextspecific best practices for construction methods should be investigated prior to conducting industry-wide questionnaire survey and analysing data objectively. This aspect of the research makes it different from previous similar studies that entirely used a positivist paradigm.

Data analysis

Qualitative data analysis consists of three concurrent flows of activity: data reduction, data display, and conclusion drawing (Rose, Spinks and Canhoto, 2015). Data reduction is a form of analysis that sharpens sorts, focuses, discards, and organizes data in a way that final conclusions can be drawn (Miles and Huberman, 1994). Writing summaries, coding, and making clusters are common methods used in data reduction process. The latter two methods are more suitable when the research is entirely inductive in nature. In this study, writing summaries was used as a technique to reduce the transcribed interviews. The reduced data is displayed using matrices, graphs, charts and networks. In this research, matrix technique is used as it is suitable to display the summaries of the responses in matrix boxes. Once the data is displayed, the conclusion is drawn by either noticing the patterns of similarities and differences between categories and/or processes, clustering, making contrasts and comparisons and noting relations between concepts (Rose, Spinks and Canhoto, 2015). The latter three strategies for conclusion drawing are more appropriate if the study is entirely qualitative in nature. Qualitative data was analysed and used as input to the quantitative phase in this research context. Identifying the pattern of similarity between the responses was used to draw a conclusion.

To analyse the data collected during phase I, the audiotaped interviews were first transcribed, and a matrix was prepared in Excel spreadsheet to match the responses of an expert and management practices. A summary of each interview result was written in a matrix box and conclusion was drawn for each practice. Similar iterative procedures were used for all the interview results. The similarity between the successive summaries was observed to find saturation point. After analysing the outcome of the fifteenth interview, similar explanations for management practices was observed. Although the saturation point was reached at the fifteenth interviewee, more interviews were conducted until the nineteenth participant, for the sake of confirmation. Finally, the best practices that were described as applicable to building projects by all participants were included in the list for the industry-wide survey.

To prioritize the best practices, Relative Importance Index (RII) was computed by using the equation below, and weights were assigned to the best practices (Enshassi et al., 2007). Furthermore, Friedman's and Wilcoxon's tests were conducted to check whether there were significant differences among the best practices.

$$RII = \frac{5(n_5) + 4(n_4) + 3(n_3) + 2(n_2) + n_1}{5(n_1 + n_2 + n_3 + n_4 + n_5)} *100$$

The number of respondents who selected 1 for not important, 2 for slightly important, 3 for somewhat important, 4 for very important, and 5 for extremely important practice to improve productivity in building projects are shown as n_1 , n_2 , n_3 , n_4 , and n_5 respectively.

To develop a scoring tool for the best practices, the weight of each practice was distributed proportionally from Level B to Level F, and Level A was assigned 0. The results are shown in the Findings and Discussion section, and Appendix-2. One-way ANOVA test was conducted to check if there was a significant difference between projects having higher score and lower score, and to validate the tool used to measure the best practices. Correlation analysis was conducted to check the relationship among best practices for construction methods, project delays, project costs, company size, experience of construction firms and annual turnover.

Logistic regression analysis was conducted to develop a model, to predict probability with respect to baseline productivity factor, based on a project's construction methods score. The model was validated by dividing the data obtained from 39 companies randomly into two (Zayed and Halpin, 2005). Thirty-one data points were used to build a model and the remaining data sets were used for validation purpose. The rule of thumb for sample size for logistic regression analysis states that the number of "Events Per Variable (EPV)" should be greater than 10 (Peduzzi et al., 1996). Some authors argue that the minimum of 10 events per predictor is conservative (Vittinghoff and McCulloch, 2007). In this research context, there were 18 projects with productivity factors greater than baseline PF = 0.97 (positive events) and there is one predictor (construction methods). EPV = 18>10 and logistic regression analysis can be conducted. Moreover, to increase the reliability of the model, bootstrapping was conducted. Before developing the final model, six alternative models were developed by varying the baseline PF. Using validation datasets, probabilities were predicted, Receiver Operating Characteristic (ROC) curves were drawn and Area under Curve (AUC) was computed.



Findings and discussion

Identification of best practices for construction methods in building projects

Ten best practices for construction methods were identified. These are integrated schedule, work schedule strategies, schedule execution and management, dynamic site layout plan, traffic control plan, site security plan, machinery positioning strategy, project start-up plan, project completion plan, and innovations and new technologies. For the sake of brevity, a summary of the findings of only two best practices is indicated in Table 2.

Construction schedule that integrates work, materials procurement and delivery, machinery, sub-contractors, financial and other schedules have positive impact on the productivity of building projects. If work schedule is integrated with material and machinery schedules then the project teams are aware of when a particular activity needs to be done, what type of machinery should be hired, how much material and manpower should be deployed, and thus track the delivery of materials. The research participants described that the best-performing principal contractors link all the schedules together.

Developing a suitable working-hours strategy is also found to be another important management practice that increases productivity. There are working hour restrictions imposed by city councils, and contractors are required to develop a strategy to reduce project delays. Most commercial contractors in Victoria follow the calendar prepared by Construction, Forestry, Mining and Energy Union, which is typically 36 working hours per week. They schedule to work either four, five or six days a week and sign contracts with their subcontractors accordingly. Some respondents described that on Saturdays they reduce the working hours to 50%. They described that most workers are not productive on Saturday, Sunday, and rostered days off. Moreover, the participants indicated that it is expensive to work on Sundays and after hours, as the rates provided in the Enterprise Bargaining Agreement are high.

Construction Methods	Summary	Conclusion
Integrated schedule	Linking various programs together is essential. Most principal contractors try to link materials procurement, lead time, work status, main work program and other issues such as the schedule for FFE (Fittings, Fixture, and Equipment) together.	applicable
Dynamic site layout plan	The site plan is constantly changing for the sake of productivity. The gates in and out, for instance, might be changed throughout the construction period. They could be changed after completion of excavation, during superstructure and landscaping works. The site layout is also planned by considering materials delivery and the requirement of temporary facilities at various stages of the project.	applicable

Table 2: Summary of interview results

Changing a site plan based on phases of a construction project also improves productivity. The interviewees described that although the practice of adopting dynamic site layout is important, it should be planned ahead to be effective. They indicated that experienced contractors include their site logistics plan in a tender document. The changes in a plan that is based on the stages of construction are clearly shown in the tender document so that sub-contractors are aware of the future changes in the site layout. Moreover, the

respondents described that integration of traffic control and site layout plans is of paramount importance in reducing the loss of productivity.

Construction machinery positioning strategy is also an important practice that influences productivity in the construction of building projects. The critical machine for these projects is a tower crane and its location on a site is planned by considering the weight to be loaded, distance from the street from which materials are lifted, the area of a building under construction and distance from existing buildings. The respondents described that careful analysis of the position of a crane using either 2D drawings or 3D models is an important practice.

Relative importance of the best practices for construction methods

Based on the results of RII analysis, the five most critical practices that have the potential to improve the productivity in building projects (Table 3) are traffic control plan, project start-up plan, machinery positioning strategy, project completion plan, and dynamic site layout plan. The finding of this study confirmed that best practices that are given top priority in other project types cannot directly be used for building projects. Traffic control plan was found to be the most significant practice that could enhance the productivity of building projects. However, for infrastructure projects the practice is not given top priority, and for the industrial projects traffic control plan is not included in the list of best practices (CII, 2013a; b). In Victoria, regulations oblige contractors that use public roads during construction periods, to prepare traffic management plans. Failure to prepare such plans will result in financial penalty and could lead to suspension of works if accidents occur. According to Road Safety (Traffic Management) Regulations (2009), any person conducting an activity on a road or road related areas must maintain a copy of the traffic management plan on the worksite at all times when the work is undertaken, and it should be available for inspection on request by an authorized person. The absence of the plan is an offense against the Regulation. Therefore, to reduce project delay and related penalty, principal contractors considered traffic control planning as the most important practice.

Project start-up planning is also considered among crucial practices that should be given priority in building projects. The finding of this study suggests that principal contractors in Victoria, Australia attempt to reduce the initial project delay by using a project start-up plan which could be used as a checklist to remind the project team regarding the information that is required to commence a project. In the plan, the pre-commencement meeting dates, resource requirements, and any other information which could be forgotten during commencement date are included. As per the plan, the principal contractor conducts meetings with its sub-contractors and resolves matters such as interference among subcontractors which could be one of the causes of the delays during project start-up.

Elements	Weight (%)	Rank
Traffic Control Plan	86	1
Project start-up plan	86	1
Machinery Positioning Strategy	83	3
Project Completion Plan	82	4
Dynamic Site Layout Plan	82	5
Schedule Execution and	77	6
Work Schedule Strategies	73	7
Site Security Plan	73	7

Table 3: Relative importance of the best practices



Integrated Schedule	66	9
Innovations and New Technologies	64	10

Similarly, machinery positioning strategy is considered to be the most significant practice for building projects. However, the practice is rated as having less importance for infrastructure projects and as an intermediate significance for industrial projects. Since the critical machinery for high-rise building projects is a tower crane, its location should be carefully analysed for the sake of productivity. The construction systems in Victoria, such as the use of heavy precast concrete panels in building projects, could influence the location of cranes. If the machine is not positioned in a suitable location, these concrete elements could not be placed in accordance with the scheduled time. The contractor might change the design of the panels to reduce their weights or look for other alternatives to place them. Moreover, since most multi-story building projects are in the city of Melbourne where there are restricted spaces, machinery positioning strategy is the crucial practice to improve productivity.

In Table 4, the results of Friedman test are presented. As the p-value (<0.001) is much less than 0.05, the null hypothesis which states the mean of all the best practices is equal should be rejected. Therefore, there is a statistically significant difference between the best practices. To identify where such difference lies, Wilcoxon Test was conducted.

Elements	Mean Rank
Integrated Schedule	3.72
Work Schedule Strategies	4.76
Schedule Execution and Management	5.49
Dynamic Site Layout Plan	6.24
Traffic Control Plan	6.76
Site Security Plan	4.86
Machinery and Equipment Positioning	6.29
Project start-up plan	6.82
Project Completion Plan	6.59
Innovations and New Technologies	3.47
df	9
χ^2 approximation	74.73
p-value	< 0.001

Table 4: Outputs of the Friedman test

The results of Wilcoxon test are shown in Table 5. Accordingly, from forty-five possible combinations, statistically significant differences exist among twenty-nine of them (Table 5). Ten possible combinations among the top five best practices were found to be statistically insignificant at the 5% significance level. This shows that there is no major difference among the top five best practices. Therefore, traffic control plan, project start-up plan, machinery positioning strategy, project completion plan and dynamic site layout plan are equally important to improve productivity in building projects.

Table 5: P-values obtained using Wilcoxon analysis

	IS	WSS	SEM	DSLP	ТСР	SSP	MPS	PSP	РСР	INT
IS		.013	.001	<.001	<.001	.040	<.001	<.001	<.001	.557
WSS			.046	.014	<.001	1.00	.028	.002	.013	.003
SEM				.242	.023	.158	.252	.036	.140	<.001
DSLP					.159	.009	.950	.162	.947	<.001
ТСР						.001	.233	.917	.414	<.001
SSP							.023	.002	.025	.004



MPS				.339	.836	<.001
PSP					.446	<.001
PCP						<.001
INT						

Legend: IS=Integrated Schedule, WSS= Work Schedule Strategies, SEM=Schedule Execution and Management, DSLP= Dynamic Site Layout Plan, TCP=Traffic Control Plan, SSP=Site Security Plan, MPS=Machinery Positioning Strategy, PSP=Project Start-up Plan, PCP= Project Completion Plan, INT=Innovation and new Technologies.

Best practices for construction methods scoring tool and validation

To develop the Scoring Tool (Appendix-2) for best practices, the weight obtained using RII analysis was proportionally distributed. For instance, the weight for the practice integrated schedule is 0.66, and the proportions are Level A=0; Level B=1/5*(0.66)=0.13; Level C=2/5*(0.66)=0.26; Level D=3/5*(0.66)=0.40; Level E=4/5*(0.66)=0.53; and Level F=0.66 (Table 6). Similarly, the scoring tools for other best practices were developed. In Table 6, the score of the practice "integrated schedule" for one of the projects is shown in the last column. Since the respondent of that particular project ticked Level C, the equivalent score for "integrated schedule" is 0.26. The total construction methods score for a project is computed by adding the scores of the ten best practices. All the thirty-nine survey data was converted to scores of the best practices for construction methods. Overall there are 390 (39*10) best practices scores.

Survey Da	ta Collected using Standard Tool (CII, 2013b)	Distributed Weights	Score
Level A	The use of an integrated schedule using CPM is not applicable.	0	
Level B	The use of an integrated schedule using CPM has not been addressed	0.13	
Level C	Developing a schedule with no resources present and managing schedule status via duration or remaining duration but no link to earned percent complete progress from associated deliverables per activity.	0.26	0 26
Level D	Developing a schedule with resources present but no link to earned percent complete progress from associated deliverables per activity.	0.40	0.20
Level E	Developing a schedule with resources present but no link to earned percent complete progress from associated deliverables per activity. Resources are updated to reflect current work content or quantity adjustments.	0.53	
Level F	Continuation of Level E and updated to include quantity adjustments. Earned progress for the activity is based on measured or assessed work completed per deliverables per activity. Progress measurement performed in application adapted specifically for each deliverable.	0.66	

Table 6: Integrated schedule score

One-way ANOVA was conducted, to test the hypothesis that building projects with higher scores of the best practices for construction methods have also higher productivity. The projects were grouped based on the baseline mean score of 4.84. Projects with scores less than the mean value were classified under Group 1(low score) and those having scores greater than the baseline were classified under Group 2 (high score). In Table 7, the descriptive statistics and ANOVA results are presented. The mean productivity factor of Group 1 and Group 2 are 0.88 and 1.01 respectively. The p-value (0.004) is less than 0.05 indicating that there is statistically significant difference between the two groups. The finding implies that building projects with a higher level of implementation of the best practices for construction methods also have higher productivity. Therefore, the tool developed to measure the best practices is valid. The tool is shown in Appendix-2.

Descriptive					ANOVA		_			
			95% CI for	Mean						
	Ν	Mean	Lower Bound	Upper Bound		Sum of Squares	df.	Mean Square	F	Sig.
Group	18	0.882	0.808	0.955	Between	0.161	1	0.161	9.643	0.004
Group	21	1.012	0.960	1.061	Within Groups	0.617	37	0.017		
Total	39	0.951	0.905	0.998	Total	0.777	38			

Table 7: One-way ANOVA for PF and construction methods score

Relationship between project delay and best practices

Correlation analysis was conducted to investigate the relation between project delay and the best practices and the results are indicated in Table 8.

Table 8: Spearman correlation coefficients for project delay

		PF	IS	WSS	SEM	DSLP	ТСР	SSP	MPS	PSP	РСР	INT	СМ
Delay	Coeff.	940	470	204	195	262	177	011	349	524	366	079	425
	Sig.	<.001	.003	.213	.234	.107	.280	.949	.030	.001	.022	.635	.007

Legend: PF= Productivity Factor, IS=Integrated Schedule, WSS= Work Schedule Strategies, SEM=Schedule Execution and Management, DSLP= Dynamic Site Layout Plan, TCP=Traffic Control Plan, SSP=Site Security Plan, MPS=Machinery Positioning Strategy, PSP=Project Start-up Plan, PCP= Project Completion Plan, INT=Innovation and new Technologies, CM= Construction Methods.

There is statistically significant negative correlation between integrated schedule, machinery positioning strategy, project start-up plan, project completion plan, and project delay. The results show that high levels of implementation of the best practices are associated with low project delay.

Correlation analysis was also conducted to check if implementation levels of the best practices vary based on project cost, annual turnover of companies, experience of companies, and company sizes. The results of the analysis are shown in Table 9. Statistically significant correlation between project cost and construction methods (p=0.001<0.05) was found. Thus, one of the main reasons for variations in the level of implementation of the best practices could be the change in project costs. As the project costs increase due to increment in the scope of works, the levels of implementation also increase. Although there is a positive correlation between construction methods and annual turnover, experience, and company size, the p-values are not statistically significant. This shows that larger and experienced companies may not fully implement the best practices if the scope of a project is small.

		Construction	Annual	Company	Company	Project
		Methods	Turnover	Experience	Size	Cost
A naval Tumoron	Coeff.	0.119				
Annual Turnover	Sig.	0.570				
Company	Coeff.	0.224	0.062			
Experience	Sig.	0.176	0.769			
C C.	Coeff.	0.261		0.520		
Company Size	Sig.	0.109		0.001		
D · · · C · ·	Coeff.	0.538	0.356	0.547	0.488	1.00
Project Cost	Sig.	0.001	0.081	0.001	0.002	
	•	•	•	•	•	40

Table 9: Spearman correlation coefficients of company profile and project cost

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects



Logistic regression model building and validation

Binary logistic regression model was developed to predict the probability of exceeding the baseline productivity factor of 0.97 which is the mean PF value. Some of the results are shown in Table 10. For the sake of brevity all the outputs are not presented in this paper. The model's Omnibus Test indicates significance value of 0.016 < 0.05; Hosmer and Lemeshow Test shows a p-value of 0.172 > 0.05 indicating the strength of the model. Furthermore, the model's overall p-value is 0.029 < 0.050 and it is acceptable. The model in Block-0 during the analysis indicates the predictive capacity of 58.1% whereas the final model has a predictive capacity of 74.2%. Thus, the selected model is good. Moreover, the coefficient of the variable after bootstrapping using 1000 samples is statistically significant (p=0.01 < 0.05).

Using the coefficients indicated in Table 10, the final model is Log (p/1-p) = 0.749*CM-3.366. To compute probabilities, the model's equation is simplified as follows. Let $L=B_0+B_1$ *CM, then $P=e^L/(1+e^L)$. By using this formula, probabilities are predicted and the sigmoid graph is drawn as shown in Figure 1. Users can compute their projects' CM score using the scoring technique (Appendix-2) developed in this research and read the probability from the graph. For instance, if a project manager computes the construction methods score of 4.0 then the probability that a project's PF exceeding 0.97 is 40%. Thus, the actual productivity could be less than planned productivity and the possibility of occurrence of project delay is high. However, the project manager can increase the chance of completing a project on or before the contract date by increasing the level of implementation of the best practices. The tools developed in this study have paramount significance for assisting building project teams to carry out such predictions.

The model was validated by predicting the probabilities of the validation datasets. Receiver Operating Characteristic (ROC) Curve was drawn (Figure 2) and Area under the Curve (AUC) was computed to be 0.833. Thus, the model is valid as AUC is greater than 0.5 and close to 1.0.

Variables in the Equation										
								95% C.I.f	or EXP(B)	
	В	S.E.	Wald	df		Sig. Exp(B		Lower	Upper	
Construction Methods (CM)	0.749	0.343	4.785	1		0.029 2.116		1.081	4.141	
Constant	-3.366	1.724	3.809	1		0.050 0.035				
Bootstrap for Variables in t	he Equat	ion								
						Bootstra	р			
	D					95% Cor	nfidence Int	e Interval		
	В	Bias	S.E	Sig.		Lower		Upper		
Construction Methods (CM)	0.749	0.120	0.465	0.01	13	0.156		1.995		
Constant	-3.366	-0.548	2.243	0.03	31	-9.286		-0.448		
Classification Table										
Observed		Pr	redicted							
		PI	7		Perc	centage Correct				
PF	0	0	1		61.5	5				
	1	8	5		83.3	5				
		3	1.	5	74.2	2				
The cut value is 0.5	500									

Table 10: Summary of the outputs from logistic regression analysis



CUTSESS

Figure 1: Probability plot of construction methods



Figure 2: ROC curve

Conclusion

Ten best practices for construction methods were identified and prioritized for building projects in Victoria, Australia. Among the ten best practices, traffic control plan, project start-up plan, machinery positioning strategy, project completion plan, and dynamic site layout plan were found to be the top five most important best practices. The study revealed that high levels of implementation of best practices are associated with low project delays. The levels of implementation of best practices increase as project costs increase. Positive correlation between company's annual turnover, company size, company experience, and best practices were found. However, these relationships are not statistically significant. Logistic regression model was developed to predict probability based on the best practices score.

This study has implications for both contractors and future researchers. The principal contractors involved in the construction of building projects in Victoria, Australia can implement the identified best practices to improve the productivity of their projects. They can also score the construction methods and use the logistic regression model to predict the probability of exceeding a baseline productivity factor. Based on the predicted productivity, contractors can implement corrective actions to achieve the desired level of productivity. Contractors in other countries can also use these practices to enhance the productivity of their building projects, but the best practices should be adapted to local 468



context as they could vary from country to country. Furthermore, sub-contractors both in Australia and other countries might not enhance their building projects' productivity by implementing the identified best practices as this study focuses on the management practices of principal contractors only. Future researchers can use this study as a background to investigate best practices for construction methods that could enhance construction productivity from sub-contractors' perspectives. Furthermore, researchers in other countries can also use the best practices identified in this research to validate their suitability in a particular environment and to identify the most critical practices.

References

- Arditi, D. and Mochtar, K., 1996. Productivity improvement in the Indonesian construction industry. *Construction Management and Economics*, 14(1), pp.13-24.
- Attalla, M.M., 1997. Project Control Techniques: Reconstruction of Occupied Buildings. MSc Thesis, University of Waterloo, Canada.
- Australian Bureau of Statistics, 2013. Private Sector Construction Industry, Australia, 2011-12. *Cat. no.8772.0.* Canberra.
- Australian Bureau of Statistics, 2016. Counts of Australian Businesses, including Entries and Exits, Jun 2011 to Jun 2015. *Cat. no.8165.0.* Canberra.
- Bell, L.C. and Stukhart, G., 1987. Costs and benefits of materials management systems. *Journal of Construction Engineering and Management*, 113(2), pp.222-34.
- Bloom, N. and Van Reenen, J., 2007. Measuring and explaining management practices across firms and countries. *Quarterly Journal of Economics*, 122(4), pp.1351-408.
- Caldas, C.H., Kim, J.-Y., Haas, C.T., Goodrum, P.M. and Zhang, D., 2014. Method to Assess the Level of Implementation of Productivity Practices on Industrial Projects. *Journal of Construction Engineering and Management*, 141(1), p.04014061.
- CFMEU, 2016. 2016 Interim RDO calendar [online]. Available at: https://vic.cfmeu.org.au.
- Choi, C. and Harris, F., 1992. A model for Determining Optimum Crane Position: Thomas Telford Services Ltd Thomas Telford House, 1 Heron Quay, London, England 503-4.
- CII, 2013a. Best Productivity Practices Implementation Index for Industrial Projects. Austin, Texas.
- CII, 2013b. Best Productivity Practices Implementation Index for Infrastructure Projects. Austin, Texas.
- CII, 2016. Construction Industry Best Practices [online]. Available at: https://www.constructioninstitute.org.
- Dai, J., Goodrum, P.M., Maloney, W.F. and Srinivasan, C., 2009. Latent Structures of the Factors Affecting Construction Labor Productivity. *Journal of Construction Engineering & Management*, 135(5), pp.397-406.
- Department of Treasury and Finance, 2015. *Construction Supplier Register* [online] Victoria, Australia. Available at: http://www.dtf.vic.gov.au.
- Enshassi, A., Mohamed, S., Mustafa, Z.A. and Mayer, P.E., 2007. Factors affecting labour productivity in building projects in the Gaza Strip. *Journal of Civil Engineering and Management*, 13(4), pp.245-54.
- EPA, 2016. Noise in commercial construction sites and large residential and mixed-use developments [online]. Available at: http://www.epa.vic.gov.au.
- Fair Work Commission, 2016. Building and Construction General On-site Award 2010 [online]. Available at: https://www.fwc.gov.au/.
- Fangel, M., 1984. Planning project start-up. International Journal of Project Management, 2(4), pp.242-5.

An Investigation into Construction Management Practices Influencing Labour Productivity in Multi-Storey Building Projects

CUTS

- Hanna, A.S., Chang, C.-K., Sullivan, K.T. and Lackney, J.A., 2008. Impact of shift work on labor productivity for labor intensive contractor. *Journal of construction engineering and management*, 134(3), pp.197-204.
- Hewage, K.N., Gannoruwa, A. and Ruwanpura, J.Y., 2011. Current Status of Factors Leading to Team Performance of On-Site Construction Professionals in Alberta Building Construction Projects. *Canadian Journal of Civil Engineering*, 38(6), pp.679-89.
- Hinze, J. and Wilson, G., 2000. Moving toward a zero injury objective. *Journal of Construction Engineering and Management*, 126(5), pp.399-403.
- Jarkas, A.M. and Bitar, C.G., 2011. Factors affecting construction labor productivity in Kuwait. *Journal of Construction Engineering and Management*.
- Kerzner, H., 2010. Project Management-Best Practices: Achieving Global Excellence: John Wiley and Sons.
- Lam, P.T., Wong, F.W. and Chan, A.P., 2006. Contributions of designers to improving buildability and constructability. *Design Studies*, 27(4), pp.457-79.
- Miles, M.B. and Huberman, A.M., 1994. *Qualitative data analysis: An expanded sourcebook*: SAGE Publications.
- Nasir, H., 2013. Best Productivity Practices Implementation Index (BPPII) for Infrastructure Projects. Doctoral Dissertation, University of Waterloo.
- Peduzzi, P., Concato, J., Kemper, E., Holford, T.R. and Feinstein, A.R., 1996. A simulation study of the number of events per variable in logistic regression analysis. *Journal of clinical epidemiology*, 49(12), pp.1373-9.
- PMI, 2013. A Guide to the Project Management Body of Knowledge (PMBOK Guide)—Fifth Edition. Pennsylvania, USA.
- PrefabAUS, 2014. Prefabricating Australia: Growing Our Off-site Construction Sector. In: *PrefabAUS 2014 Inaugural Conference* Melbourne, Australia, 11-13 August.
- Rose, S., Spinks, N. and Canhoto, A.I., 2015. *Management research : applying the principles*: Abingdon, Oxon ; New York : Routledge, 2015.
- Safe Work Australia, 2014. Traffic Management: Guide for Construction Work [online] Canberra, Australia. Available at: http://www.safeworkaustralia.gov.au.
- Stewart, L., 2002. Why rent? For low-use machines. Construction Equipment, 105(7), p.50.
- Tommelein, I. and Zouein, P., 1993. Interactive dynamic layout planning. *Journal of construction engineering and management*, 119(2), pp.266-87.
- Vittinghoff, E. and McCulloch, C.E., 2007. Relaxing the rule of ten events per variable in logistic and Cox regression. *American journal of epidemiology*, 165(6), pp.710-18.
- Wheeldon, D., 2012. Why building materials imports are on the rise in Australia? *Infolink Architecture and Design*. Melbourne, Australia.
- Zayed, T.M. and Halpin, D.W., 2005. Productivity and cost regression models for pile construction. *Journal of Construction Engineering and Management*, 131(7), pp.779-89.

Appendix-1

Questionnaire

A. Relative Importance of the Management Practices for Construction Methods

Please rate the following elements based on their degree of importance in improving productivity in building construction projects on the scale of 1 to 5.

	Level of	f Importa	nce		
Best Practices for Construction Methods	Not	Slightly	Somewhat	Very	Extremely
	important	important	important	important	important
	(1)	(2)	(3)	(4)	(5)

CUTS

1. Integrated Schedule			
2. Work Schedule Strategies			
3. Schedule Execution and Management			
4. Dynamic Site Layout Plan			
5. Traffic Control Plan			
6. Site Security Plan			
7. Machinery & Equipment Positioning Strategy			
8. Project start-up plan			
9. Project Completion Plan			
10. Innovations & New Technologies			

B. Data Collection Tool from Building Projects B1. Project Characteristics

,	
Project type (residential, commercial etc.)	
Project value	
Floor area	
Project start date:	
Planned completion date:	
Actual completion date:	

B2. Best Practices for Construction Methods Survey Tool(unweighted) (adapted from (CII, 2013b)) Please check ($\sqrt{}$) one box per element in space provided in front of the level of implementation of the management practice. Please do not leave any elements blank.

1.Integrat	ed Schedule
Level A	The use of an integrated schedule using CPM is not applicable.
Level B	The use of an integrated schedule using CPM has not been addressed.
Level C	Developing a schedule with no resources present and managing schedule status via duration or remaining duration but not linked to earned percent complete progress from associated deliverables per activity
Level D	animation but not interface to cannot be progress from associated performance per activity.
Leverb	associated deliverables per activity.
Level E	Developing a schedule with resources present but not linked to earned percent complete progress from
	associated deliverables per activity. Resources are updated to reflect current work content or quantity
	adjustments.
Level F	Continuation of Level E and updated to include quantity adjustments. Earned progress for the activity is
	based on measured or assessed work completed per deliverables per activity. Progress measurement
	performed in application adapted specifically for each deliverable.
2. Work S	chedule Strategies
Level A	The development of a work schedule strategy is not applicable
Level B	The development of a work schedule strategy has not been addressed
Level C	The strategy is based on a single work schedule be it either a straight time such as 36 hours per week
	schedule, overtime, or other work schedule strategies.
Level D	Strategy considers multiple work schedules considering critical and near critical activity sequences.
Level E	Continuation of Level D, plus strategies considers the potential impact on worker fatigue, supervision,
	safety, and absenteeism.
Level F	Continuation of Level E, plus each potential strategy's impact is analysed for manpower density and
	congestion at an area or sub-area level.
3. Schedu	le Execution and Management
Level A	The development of a schedule compliance plan is not applicable
Level B	The development of a schedule compliance plan has not been addressed
Level C	Consistent follow-up to monitor the following tasks: schedule updated periodically, critical path analysis,
	and progress narrative prepared as required and effective team participation in schedule updates.
Level D	Continuation of Level C, plus quantity reports are regularly performed. Upon request, or as the project
	requires, may include any of the following: change management analysis, risks assessment
	scenarios/analysis, date variance analysis to approved baseline or previous update period, start / finish
	percent achieved ratio analysis, communication with material suppliers to ensure material will arrive on
	site when planned.
Level E	Continuation of Level D, plus monitor the following: schedule rigorously updated based on manual input
	of quantity reports, critical and near critical path analysis, progress narrative prepared and effective team
	participation in schedule updates. Quantity reports rigorously done by trained individual(s). Material
	suppliers routinely contacted to track the status of material delivery dates.



Level F	Continuation of Level E, plus will consistently include all of the following, based on project requirements
	and observed schedule status conditions: change management analysis, risks assessment
	scenarios/analysis, date variance analysis to approved baseline or previous update period, start / finish
	percent achieved ratio analysis; also included progress tracking using 3D imaging and other techniques.
4. Dynam	ic Site Layout Plan
Level A	Site layout plan is not applicable for the project.
Level B	A site layout plan has not been addressed.
Level C	The project team examines the project schedule and assesses when Temporary Facilities (TF) will be
	brought in.
Level D	Continuation of Level C, plus what sizes will be needed prior to the start of the project. No consideration
	is given to the addition and removal of TFs at different stages of the project. No analysis is done in
	regards to the layout of the project to optimize locations of the TFs to limit travel time to and from.
Level E	Continuation of Level D, plus consideration is given to the addition and removal of TFs at different
	stages of the project.
Level F	Continuation of Level E, plus the team analyses the layout of the project including where the different
	parties will be working and place their TFs in the optimum location in order to limit travel time to and
	from TFs.
5. Traffic	Control Plan
Level A	Traffic control plans are not applicable for the project.
Level B	Traffic control plans have not been addressed for the project.
Level C	The project has some traffic control plans and is used on a reactive basis.
Level D	The project has a traffic control plan, equipment, and an arrangement for daylight traffic control only and
	has no trained traffic control persons.
Level E	The project has a traffic control plan and equipment for all times of the day including trained persons for
	traffic control.
Level F	Continuation of level E, plus a trained traffic control supervisor. It has an approved contingency plan in
	place to accommodate unexpected situations, and has designed and constructed alternate arrangements
	for traffic such as detours, flyovers, etc.
6. Site See	curity Plan
Level A	Site security plan is not applicable for the project.
Level B	The site does not institute security in regards to entry to the site, securing commodities, or tools and
	equipment.
Level C	The site controls entry and exit from the site, but does not have any other formal security throughout the
	site.
Level D	The site has established security procedures including visitor sign in and sign out procedure and security
	guards at every gate. The site has implemented security measures to ensure the preservation of company
	assets. Protocols have been identified for searches of individuals and their personal property. Searches are
T 1 D	conducted randomly.
Level E	Continuation of Level D, plus the site has ensured that material is not leaving the job site by instituting
L1 IZ	lock-ups for items that are prone to their.
Level F	Continuation of Level E, plus the use of electronic security such as security cameras has been
7 Maalaia	mpleneted.
7. Machin	Machinery activity is not explicible
Level A	There is no strategy for positioning of machinery at the project site
Level D	Heavy rigging and lifting studies are accomplished on all critical lifts including evaluation of the
Lever	machinery rigging selection and crane location
Level D	Continuation of Level C plus planning includes the use of 2D layout and studies to aid in constructability
LINCID	for locating and utilizing machinery.
Level F	Continuation of Level D plus some 3D modelling studies to aid in constructability for locating and
Lever	utilizing machinery
Level F	Continuation of Level E plus planning includes the use of 3D layout studies and 3D
1.00011	modelling/visualization to aid in constructability for locating and utilizing machinery.
8. Project	Start-up Plan
Level A	No start-up plan exists.
Level B	A partial start-up plan has been prepared; the plan has not been communicated to the concerned stakeholders.
Level C	A basic start-up plan has been developed with input from the project participants, but the plan has not been implemented
Level D	Continuation of Level C plus with considerations for interfaces among sub-contractors or project
LAVELD	participants. A start-up plan has been developed that identifies the duties and responsibilities of each
	stakeholder.
Level E	Continuation of Level D plus with consideration for cost analysis and detailed scheduling components
	The plan is well communicated to all the stakeholders.

Level F	Continuation of Level E, plus with the plan being implemented on the project by incorporating feedbacks
	from the stakeholders or project participants.
9. Project	Completion Plan
Level A	The project completion requirement or handover procedure is not applicable.
Level B	The project completion requirement or handover procedure has not been identified.
Level C	The project has a handover procedure that defines the parameters of project completion and delineates
	the requirements for the handover.
Level D	The project has a formal handover process that defines the necessary documentation, parameters of
	completion and other issues to assure proper handover of a project.
Level E	Continuation of Level D, plus the procedure has been reviewed and agreed by the stakeholders.
Level F	Continuation of Level E plus the plan is approved by project management team and is reviewed for
	applicability during all phases of the handover process.
10. Innova	ations and New Technologies
Level A	Innovation in new materials, equipment, information systems is not applicable.
Level B	Innovations and new technologies investigation is not addressed.
Level C	The project does not have a formal program for the investigation of innovations and new technologies.
	Implementation of innovations and new systems will only occur after the industry-wide implementation.
Level D	The organization has an informal program for the investigation of innovations, and they will investigate
	the feasibility of the new technologies on a regular basis.
Level E	Continuation of Level D, plus the program is formal to investigate new systems and they will investigate
	the feasibility of the new technologies on a regular basis.
Level F	Continuation of Level E, plus they investigate all new technologies using a formal system of rating the
	new technology.

Appendix-2

Best Practices for Construction Methods Scoring Tool (Weighted) for Building Projects

1.Integrate	d Schedule	Weights
Level A	The use of an integrated schedule using CPM is not applicable.	0.00
Level B	The use of an integrated schedule using CPM has not been addressed	0.13
Level C	Developing a schedule with no resources present and managing schedule status via duration	
	or remaining duration but no link to earned percent complete progress from associated deliverables per activity.	0.26
Level D	Developing a schedule with resources present but no link to earned percent complete progress from associated deliverables per activity.	0.40
Level E	Developing a schedule with resources present but no link to earned percent complete progress from associated deliverables per activity. Resources are updated to reflect current work content or quantity adjustments.	0.53
Level F	Continuation of Level E and updated to include quantity adjustments. Earned progress for the activity is based on measured or assessed work completed per deliverables per activity. Progress measurement performed in application adapted specifically for each deliverable.	0.66
2. Work Sc	hedule Strategies	
Level A	The development of a work schedule strategy is not applicable	0.00
Level B	The development of a work schedule strategy has not been addressed	0.15
Level C	The strategy is based on a single work schedule be it either a straight time such as 36 hours per week schedule, overtime, or other work schedule strategies.	0.29
Level D	Strategy considers multiple work schedules considering critical and near critical activity sequences.	0.44
Level E	Continuation of Level D, plus strategies considers the potential impact on worker fatigue, supervision, safety, and absenteeism.	0.58
Level F	Continuation of Level E, plus each potential strategy's impact is analysed for manpower density and congestion at an area or sub-area level.	0.73
3. Schedul	e Execution and Management	
Level A	The development of a schedule compliance plan is not applicable	0.00
Level B	The development of a schedule compliance plan has not been addressed	0.15
Level C	Consistent follow-up to monitor the following tasks: schedule updated periodically, critical path analysis, and progress narrative prepared as required and effective team participation in schedule updates.	0.31

CUTS

L aval D		
	Continuation of Level C, plus quantity reports are regularly performed. Upon request, or as	0.46
	the project requires may include any of the following change measurement analysis ricks	
	the project requires, may include any of the following: change management analysis, fisks	
	assessment scenarios/analysis, date variance analysis to approved baseline or previous update	
	period, start / finish percent achieved ratio analysis, communication with material suppliers to	
	ensure material will arrive on site when planned.	
Lorral E	Continuation of Lovel D, also monitor the following schedule signs only undeted based on	0.62
Level E	Continuation of Level D, plus monitor the following: schedule rigorously updated based on	0.62
	manual input of quantity reports, critical and near critical path analysis, progress narrative	
	prepared and effective team participation in schedule updates. Quantity reports rigorously	
	done by trained individual(s) Material suppliers routinely contacted to track the status of	
	material delivery dates	
	material delivery dates.	
Level F	Continuation of Level E, plus will consistently include all of the following, based on project	0.77
	requirements and observed schedule status conditions; change management analysis, risks	
	assessment scenarios analysis date variance analysis to approved baseline or previous update	
	assessment section analysis, date variance analysis to approved basemic of previous update	
	period, start / finish percent achieved ratio analysis; also included progress tracking using 5D	
	imaging and other techniques.	
4. Dynami	c Site Layout Plan	
Level A	Site layout plan is not applicable for the project	0.00
LEVELT	Site layout plan is not applicable for the project.	0.00
Level B	A site layout plan has not been addressed.	0.16
Level C	The project team examines the project schedule and assesses when Temporary Facilities (TF)	0.33
	will be brought in	
T 1D		0.40
Level D	Continuation of Level C, plus what sizes will be needed prior to the start of the project. No	0.49
	consideration is given to the addition and removal of TFs at different stages of the project. No	
	analysis is done in regards to the layout of the project to optimize locations of the TFs to limit	
	travel time	
T 15		0.45
Level E	Continuation of Level D, plus consideration is given to the addition and removal of TFs at	0.65
	different stages of the project.	
Level F	Continuation of Level E, plus the team analyses the layout of the project including where the	0.82
Leveri	different post in the most interval and a set TVE is the reference interval and a to the	0.02
	different parties will be working and place their TFs in the optimum location in order to limit	
	travel time to and from TFs.	
5. Traffic O	Control Plan	
Level A	Traffic control plans are not applicable for the project	0.00
Level II	The formation is a construction of the project.	0.00
Level D	I rame control plans have not been addressed for the project.	0.17
Level C	The project has some traffic control plans and is used on a reactive basis.	0.34
Lorrol D		
	The project has a traffic control plan, equipment, and an arrangement for daylight traffic	0.511
Level D	The project has a traffic control plan, equipment, and an arrangement for daylight traffic control particle and has no trained traffic control persons.	0.511
Level D	The project has a traffic control plan, equipment, and an arrangement for daylight traffic control only and has no trained traffic control persons.	0.511
Level E	The project has a traffic control plan, equipment, and an arrangement for daylight traffic control only and has no trained traffic control persons. The project has a traffic control plan and equipment for all times of the day including trained	0.511 0.68
Level E	The project has a traffic control plan, equipment, and an arrangement for daylight traffic control only and has no trained traffic control persons. The project has a traffic control plan and equipment for all times of the day including trained persons for traffic control.	0.511 0.68
Level E	The project has a traffic control plan, equipment, and an arrangement for daylight traffic control only and has no trained traffic control persons. The project has a traffic control plan and equipment for all times of the day including trained persons for traffic control.	0.511 0.68 0.85
Level E Level F	The project has a traffic control plan, equipment, and an arrangement for daylight traffic control only and has no trained traffic control persons. The project has a traffic control plan and equipment for all times of the day including trained persons for traffic control. Continuation of level E, plus a trained traffic control supervisor. It has an approved sentiment of a supervisor of the day including and persons and has designed and	0.511 0.68 0.85
Level E Level F	The project has a traffic control plan, equipment, and an arrangement for daylight traffic control only and has no trained traffic control persons. The project has a traffic control plan and equipment for all times of the day including trained persons for traffic control. Continuation of level E, plus a trained traffic control supervisor. It has an approved contingency plan in place to accommodate unexpected situations and has designed and	0.511 0.68 0.85
Level E Level F	The project has a traffic control plan, equipment, and an arrangement for daylight traffic control only and has no trained traffic control persons. The project has a traffic control plan and equipment for all times of the day including trained persons for traffic control. Continuation of level E, plus a trained traffic control supervisor. It has an approved contingency plan in place to accommodate unexpected situations and has designed and constructed alternate arrangements for traffic such as detours, flyovers, etc.	0.511 0.68 0.85
Level E Level F 6. Site Seco	The project has a traffic control plan, equipment, and an arrangement for daylight traffic control only and has no trained traffic control persons. The project has a traffic control plan and equipment for all times of the day including trained persons for traffic control. Continuation of level E, plus a trained traffic control supervisor. It has an approved contingency plan in place to accommodate unexpected situations and has designed and constructed alternate arrangements for traffic such as detours, flyovers, etc. urity Plan	0.511 0.68 0.85
Level E Level F 6. Site Sect	The project has a traffic control plan, equipment, and an arrangement for daylight traffic control only and has no trained traffic control persons. The project has a traffic control plan and equipment for all times of the day including trained persons for traffic control. Continuation of level E, plus a trained traffic control supervisor. It has an approved contingency plan in place to accommodate unexpected situations and has designed and constructed alternate arrangements for traffic such as detours, flyovers, etc. urity Plan	0.511 0.68 0.85 0.00
Level E Level F 6. Site Sect Level A	The project has a traffic control plan, equipment, and an arrangement for daylight traffic control only and has no trained traffic control persons. The project has a traffic control plan and equipment for all times of the day including trained persons for traffic control. Continuation of level E, plus a trained traffic control supervisor. It has an approved contingency plan in place to accommodate unexpected situations and has designed and constructed alternate arrangements for traffic such as detours, flyovers, etc. urity Plan Site security plan is not applicable for the project.	0.511 0.68 0.85 0.00 0.00
Level D Level E Level F 6. Site Sect Level A Level B	The project has a traffic control plan, equipment, and an arrangement for daylight traffic control only and has no trained traffic control persons. The project has a traffic control plan and equipment for all times of the day including trained persons for traffic control. Continuation of level E, plus a trained traffic control supervisor. It has an approved contingency plan in place to accommodate unexpected situations and has designed and constructed alternate arrangements for traffic such as detours, flyovers, etc. urity Plan Site security plan is not applicable for the project. The site does not institute security in regards to entry to the site, securing commodities, or	0.511 0.68 0.85 0.00 0.15
Level D Level F 6. Site Sect Level A Level B	The project has a traffic control plan, equipment, and an arrangement for daylight traffic control only and has no trained traffic control persons. The project has a traffic control plan and equipment for all times of the day including trained persons for traffic control. Continuation of level E, plus a trained traffic control supervisor. It has an approved contingency plan in place to accommodate unexpected situations and has designed and constructed alternate arrangements for traffic such as detours, flyovers, etc. urity Plan Site security plan is not applicable for the project. The site does not institute security in regards to entry to the site, securing commodities, or tools and equipment.	0.511 0.68 0.85 0.00 0.15
Level D Level E 6. Site Sect Level A Level B Level C	The project has a traffic control plan, equipment, and an arrangement for daylight traffic control only and has no trained traffic control persons. The project has a traffic control plan and equipment for all times of the day including trained persons for traffic control. Continuation of level E, plus a trained traffic control supervisor. It has an approved contingency plan in place to accommodate unexpected situations and has designed and constructed alternate arrangements for traffic such as detours, flyovers, etc. urity Plan Site security plan is not applicable for the project. The site does not institute security in regards to entry to the site, securing commodities, or tools and equipment. The site controls entry and exit from the site, but does not have any other formal security.	0.511 0.68 0.85 0.00 0.15 0.29
Level D Level E 6. Site Seco Level A Level B Level C	The project has a traffic control plan, equipment, and an arrangement for daylight traffic control only and has no trained traffic control persons. The project has a traffic control plan and equipment for all times of the day including trained persons for traffic control. Continuation of level E, plus a trained traffic control supervisor. It has an approved contingency plan in place to accommodate unexpected situations and has designed and constructed alternate arrangements for traffic such as detours, flyovers, etc. urity Plan Site security plan is not applicable for the project. The site does not institute security in regards to entry to the site, securing commodities, or tools and equipment. The site controls entry and exit from the site, but does not have any other formal security throughout the site.	0.511 0.68 0.85 0.00 0.15 0.29
Level D Level E 6. Site Sect Level A Level B Level C	The project has a traffic control plan, equipment, and an arrangement for daylight traffic control only and has no trained traffic control persons. The project has a traffic control plan and equipment for all times of the day including trained persons for traffic control. Continuation of level E, plus a trained traffic control supervisor. It has an approved contingency plan in place to accommodate unexpected situations and has designed and constructed alternate arrangements for traffic such as detours, flyovers, etc. urity Plan Site security plan is not applicable for the project. The site does not institute security in regards to entry to the site, securing commodities, or tools and equipment. The site controls entry and exit from the site, but does not have any other formal security throughout the site.	0.511 0.68 0.85 0.00 0.15 0.29
Level D Level F 6. Site Sect Level A Level B Level C Level D	The project has a traffic control plan, equipment, and an arrangement for daylight traffic control only and has no trained traffic control persons. The project has a traffic control plan and equipment for all times of the day including trained persons for traffic control. Continuation of level E, plus a trained traffic control supervisor. It has an approved contingency plan in place to accommodate unexpected situations and has designed and constructed alternate arrangements for traffic such as detours, flyovers, etc. urity Plan Site security plan is not applicable for the project. The site does not institute security in regards to entry to the site, securing commodities, or tools and equipment. The site controls entry and exit from the site, but does not have any other formal security throughout the site. The site has established security procedures including visitor sign in and sign out procedure	0.511 0.68 0.85 0.00 0.15 0.29 0.44
Level D Level E Level F 6. Site Sect Level A Level B Level C Level D	The project has a traffic control plan, equipment, and an arrangement for daylight traffic control only and has no trained traffic control persons. The project has a traffic control plan and equipment for all times of the day including trained persons for traffic control. Continuation of level E, plus a trained traffic control supervisor. It has an approved contingency plan in place to accommodate unexpected situations and has designed and constructed alternate arrangements for traffic such as detours, flyovers, etc. urity Plan Site security plan is not applicable for the project. The site does not institute security in regards to entry to the site, securing commodities, or tools and equipment. The site controls entry and exit from the site, but does not have any other formal security throughout the site. The site has established security procedures including visitor sign in and sign out procedure and security guards at every gate. The site has implemented security measures to ensure the	0.511 0.68 0.85 0.00 0.15 0.29 0.44
Level D Level F 6. Site Sect Level A Level B Level C Level D	The project has a traffic control plan, equipment, and an arrangement for daylight traffic control only and has no trained traffic control persons. The project has a traffic control plan and equipment for all times of the day including trained persons for traffic control. Continuation of level E, plus a trained traffic control supervisor. It has an approved contingency plan in place to accommodate unexpected situations and has designed and constructed alternate arrangements for traffic such as detours, flyovers, etc. urity Plan Site security plan is not applicable for the project. The site does not institute security in regards to entry to the site, securing commodities, or tools and equipment. The site controls entry and exit from the site, but does not have any other formal security throughout the site. The site has established security procedures including visitor sign in and sign out procedure and security guards at every gate. The site has implemented security measures to ensure the preservation of company assets. Protocols have been identified for searches of individuals and	0.511 0.68 0.85 0.00 0.15 0.29 0.44
Level D Level F 6. Site Sect Level A Level B Level C Level D	The project has a traffic control plan, equipment, and an arrangement for daylight traffic control only and has no trained traffic control persons. The project has a traffic control plan and equipment for all times of the day including trained persons for traffic control. Continuation of level E, plus a trained traffic control supervisor. It has an approved contingency plan in place to accommodate unexpected situations and has designed and constructed alternate arrangements for traffic such as detours, flyovers, etc. urity Plan Site security plan is not applicable for the project. The site does not institute security in regards to entry to the site, securing commodities, or tools and equipment. The site controls entry and exit from the site, but does not have any other formal security throughout the site. The site has established security procedures including visitor sign in and sign out procedure and security guards at every gate. The site has implemented security measures to ensure the preservation of company assets. Protocols have been identified for searches of individuals and their personal property. Searches are conducted trandomly.	0.511 0.68 0.85 0.00 0.15 0.29 0.44
Level D Level E Level F 6. Site Sect Level A Level B Level C Level D	The project has a traffic control plan, equipment, and an arrangement for daylight traffic control only and has no trained traffic control persons. The project has a traffic control plan and equipment for all times of the day including trained persons for traffic control. Continuation of level E, plus a trained traffic control supervisor. It has an approved contingency plan in place to accommodate unexpected situations and has designed and constructed alternate arrangements for traffic such as detours, flyovers, etc. urity Plan Site security plan is not applicable for the project. The site does not institute security in regards to entry to the site, securing commodities, or tools and equipment. The site controls entry and exit from the site, but does not have any other formal security throughout the site. The site has established security procedures including visitor sign in and sign out procedure and security guards at every gate. The site has implemented security measures to ensure the preservation of company assets. Protocols have been identified for searches of individuals and their personal property. Searches are conducted randomly.	0.511 0.68 0.85 0.00 0.15 0.29 0.44
Level D Level E 6. Site Sect Level A Level B Level C Level D Level E	The project has a traffic control plan, equipment, and an arrangement for daylight traffic control only and has no trained traffic control persons. The project has a traffic control plan and equipment for all times of the day including trained persons for traffic control. Continuation of level E, plus a trained traffic control supervisor. It has an approved contingency plan in place to accommodate unexpected situations and has designed and constructed alternate arrangements for traffic such as detours, flyovers, etc. urity Plan Site security plan is not applicable for the project. The site does not institute security in regards to entry to the site, securing commodities, or tools and equipment. The site controls entry and exit from the site, but does not have any other formal security throughout the site. The site has established security procedures including visitor sign in and sign out procedure and security guards at every gate. The site has implemented security measures to ensure the preservation of company assets. Protocols have been identified for searches of individuals and their personal property. Searches are conducted randomly. Continuation of Level D, plus the site has ensured that material is not leaving the job site by	0.511 0.68 0.85 0.00 0.15 0.29 0.44
Level D Level E Level F 6. Site Sect Level A Level B Level C Level D Level D	The project has a traffic control plan, equipment, and an arrangement for daylight traffic control only and has no trained traffic control persons. The project has a traffic control plan and equipment for all times of the day including trained persons for traffic control. Continuation of level E, plus a trained traffic control supervisor. It has an approved contingency plan in place to accommodate unexpected situations and has designed and constructed alternate arrangements for traffic such as detours, flyovers, etc. urity Plan Site security plan is not applicable for the project. The site does not institute security in regards to entry to the site, securing commodities, or tools and equipment. The site controls entry and exit from the site, but does not have any other formal security throughout the site. The site has established security procedures including visitor sign in and sign out procedure and security guards at every gate. The site has implemented security measures to ensure the preservation of company assets. Protocols have been identified for searches of individuals and their personal property. Searches are conducted randomly. Continuation of Level D, plus the site has ensured that material is not leaving the job site by instituting "lock-ups" for items that are prone to theft.	0.511 0.68 0.85 0.00 0.15 0.29 0.44 0.59
Level D Level F 6. Site Sect Level A Level B Level C Level D Level E Level F	The project has a traffic control plan, equipment, and an arrangement for daylight traffic control only and has no trained traffic control persons. The project has a traffic control plan and equipment for all times of the day including trained persons for traffic control. Continuation of level E, plus a trained traffic control supervisor. It has an approved contingency plan in place to accommodate unexpected situations and has designed and constructed alternate arrangements for traffic such as detours, flyovers, etc. urity Plan Site security plan is not applicable for the project. The site does not institute security in regards to entry to the site, securing commodities, or tools and equipment. The site controls entry and exit from the site, but does not have any other formal security throughout the site. The site has established security procedures including visitor sign in and sign out procedure and security guards at every gate. The site has implemented security measures to ensure the preservation of company assets. Protocols have been identified for searches of individuals and their personal property. Searches are conducted randomly. Continuation of Level D, plus the site has ensured that material is not leaving the job site by instituting "lock-ups" for items that are prone to theft. Continuation of Level E, plus the use of electronic security such as security cameras has been	0.511 0.68 0.85 0.00 0.15 0.29 0.44 0.59 0.73
Level D Level F 6. Site Sect Level A Level A Level B Level C Level D Level E Level F	The project has a traffic control plan, equipment, and an arrangement for daylight traffic control only and has no trained traffic control persons. The project has a traffic control plan and equipment for all times of the day including trained persons for traffic control. Continuation of level E, plus a trained traffic control supervisor. It has an approved contingency plan in place to accommodate unexpected situations and has designed and constructed alternate arrangements for traffic such as detours, flyovers, etc. urity Plan Site security plan is not applicable for the project. The site does not institute security in regards to entry to the site, securing commodities, or tools and equipment. The site controls entry and exit from the site, but does not have any other formal security throughout the site. The site has established security procedures including visitor sign in and sign out procedure and security guards at every gate. The site has implemented security measures to ensure the preservation of company assets. Protocols have been identified for searches of individuals and their personal property. Searches are conducted randomly. Continuation of Level D, plus the site has ensured that material is not leaving the job site by instituting "lock-ups" for items that are prone to theft. Continuation of Level E, plus the use of electronic security such as security cameras has been implemented	0.511 0.68 0.85 0.00 0.15 0.29 0.44 0.59 0.73
Level D Level E Level F 6. Site Sect Level A Level B Level C Level D Level D	The project has a traffic control plan, equipment, and an arrangement for daylight traffic control only and has no trained traffic control persons. The project has a traffic control plan and equipment for all times of the day including trained persons for traffic control. Continuation of level E, plus a trained traffic control supervisor. It has an approved contingency plan in place to accommodate unexpected situations and has designed and constructed alternate arrangements for traffic such as detours, flyovers, etc. urity Plan Site security plan is not applicable for the project. The site does not institute security in regards to entry to the site, securing commodities, or tools and equipment. The site controls entry and exit from the site, but does not have any other formal security throughout the site. The site has established security procedures including visitor sign in and sign out procedure and security guards at every gate. The site has implemented security measures to ensure the preservation of company assets. Protocols have been identified for searches of individuals and their personal property. Searches are conducted randomly. Continuation of Level D, plus the site has ensured that material is not leaving the job site by instituting "lock-ups" for items that are prone to theft. Continuation of Level E, plus the use of electronic security such as security cameras has been implemented.	0.511 0.68 0.85 0.00 0.15 0.29 0.44 0.59 0.73
Level D Level E Level F 6. Site Sect Level A Level B Level C Level D Level D Level F 7.Machin	The project has a traffic control plan, equipment, and an arrangement for daylight traffic control only and has no trained traffic control persons. The project has a traffic control plan and equipment for all times of the day including trained persons for traffic control. Continuation of level E, plus a trained traffic control supervisor. It has an approved contingency plan in place to accommodate unexpected situations and has designed and constructed alternate arrangements for traffic such as detours, flyovers, etc. urity Plan Site security plan is not applicable for the project. The site does not institute security in regards to entry to the site, securing commodities, or tools and equipment. The site controls entry and exit from the site, but does not have any other formal security throughout the site. The site has established security procedures including visitor sign in and sign out procedure and security guards at every gate. The site has implemented security measures to ensure the preservation of company assets. Protocols have been identified for searches of individuals and their personal property. Searches are conducted randomly. Continuation of Level D, plus the site has ensured that material is not leaving the job site by instituting "lock-ups" for items that are prone to theft. Continuation of Level E, plus the use of electronic security such as security cameras has been implemented. ery Positioning Strategy	0.511 0.68 0.85 0.00 0.15 0.29 0.44 0.59 0.73
Level D Level E Level F 6. Site Sect Level A Level B Level C Level D Level E Level F 7.Machin Level A	The project has a traffic control plan, equipment, and an arrangement for daylight traffic control only and has no trained traffic control persons. The project has a traffic control plan and equipment for all times of the day including trained persons for traffic control. Continuation of level E, plus a trained traffic control supervisor. It has an approved contingency plan in place to accommodate unexpected situations and has designed and constructed alternate arrangements for traffic such as detours, flyovers, etc. urity Plan Site security plan is not applicable for the project. The site does not institute security in regards to entry to the site, securing commodities, or tools and equipment. The site controls entry and exit from the site, but does not have any other formal security throughout the site. The site security guards at every gate. The site has implemented security measures to ensure the preservation of company assets. Protocols have been identified for searches of individuals and their personal property. Searches are conducted randomly. Continuation of Level D, plus the site has ensured that material is not leaving the job site by instituting "lock-ups" for items that are prone to theft. Continuation of Level E, plus the use of electronic security such as security cameras has been implemented. ery Positioning Strategy Machinery positioning strategy is not applicable.	0.511 0.68 0.85 0.00 0.15 0.29 0.44 0.59 0.73 0.00
Level D Level F 6. Site Sect Level A Level B Level C Level D Level E Level F 7.Machin Level A Level B	The project has a traffic control plan, equipment, and an arrangement for daylight traffic control only and has no trained traffic control persons. The project has a traffic control plan and equipment for all times of the day including trained persons for traffic control. Continuation of level E, plus a trained traffic control supervisor. It has an approved contingency plan in place to accommodate unexpected situations and has designed and constructed alternate arrangements for traffic such as detours, flyovers, etc. urity Plan Site security plan is not applicable for the project. The site does not institute security in regards to entry to the site, securing commodities, or tools and equipment. The site controls entry and exit from the site, but does not have any other formal security throughout the site. The site has established security procedures including visitor sign in and sign out procedure and security guards at every gate. The site has implemented security measures to ensure the preservation of company assets. Protocols have been identified for searches of individuals and their personal property. Searches are conducted randomly. Continuation of Level D, plus the site has ensured that material is not leaving the job site by instituting "lock-ups" for items that are prone to theft. Continuation of Level E, plus the use of electronic security such as security cameras has been implemented. ery Positioning Strategy Machinery positioning strategy is not applicable.	0.511 0.68 0.85 0.00 0.15 0.29 0.44 0.59 0.73 0.00 0.16
Level D Level F 6. Site Sect Level A Level A Level B Level C Level D Level F 7.Machin Level A Level B Level A	The project has a traffic control plan, equipment, and an arrangement for daylight traffic control only and has no trained traffic control persons. The project has a traffic control plan and equipment for all times of the day including trained persons for traffic control. Continuation of level E, plus a trained traffic control supervisor. It has an approved contingency plan in place to accommodate unexpected situations and has designed and constructed alternate arrangements for traffic such as detours, flyovers, etc. urity Plan Site security plan is not applicable for the project. The site does not institute security in regards to entry to the site, securing commodities, or tools and equipment. The site controls entry and exit from the site, but does not have any other formal security throughout the site. The site has established security procedures including visitor sign in and sign out procedure and security guards at every gate. The site has implemented security measures to ensure the preservation of company assets. Protocols have been identified for searches of individuals and their personal property. Searches are conducted randomly. Continuation of Level D, plus the site has ensured that material is not leaving the job site by instituting "lock-ups" for items that are prone to theft. Continuation of Level E, plus the use of electronic security such as security cameras has been implemented. Every Positioning Strategy is not applicable. There is no strategy for positioning of machinery at the project site.	0.511 0.68 0.85 0.00 0.15 0.29 0.44 0.59 0.73 0.00 0.16 0.23
Level D Level E Level F 6. Site Sect Level A Level B Level C Level D Level E Level F 7.Machin Level A Level B Level C	 The project has a traffic control plan, equipment, and an arrangement for daylight traffic control only and has no trained traffic control persons. The project has a traffic control plan and equipment for all times of the day including trained persons for traffic control. Continuation of level E, plus a trained traffic control supervisor. It has an approved contingency plan in place to accommodate unexpected situations and has designed and constructed alternate arrangements for traffic such as detours, flyovers, etc. urity Plan Site security plan is not applicable for the project. The site does not institute security in regards to entry to the site, securing commodities, or tools and equipment. The site controls entry and exit from the site, but does not have any other formal security throughout the site. The site has established security procedures including visitor sign in and sign out procedure and security guards at every gate. The site has implemented for searches of individuals and their personal property. Searches are conducted randomly. Continuation of Level D, plus the site has ensured that material is not leaving the job site by instituting "lock-ups" for items that are prone to theft. Continuation of Level E, plus the use of electronic security such as security cameras has been implemented. ery Positioning Strategy Machinery positioning strategy is not applicable. There is no strategy for positioning of machinery at the project site. 	0.511 0.68 0.85 0.00 0.15 0.29 0.44 0.59 0.73 0.00 0.16 0.33
Level D Level E Level F 6. Site Sect Level A Level B Level C Level D Level E Level F 7.Machin Level A Level B Level C	The project has a traffic control plan, equipment, and an arrangement for daylight traffic control only and has no trained traffic control persons. The project has a traffic control plan and equipment for all times of the day including trained persons for traffic control. Continuation of level E, plus a trained traffic control supervisor. It has an approved contingency plan in place to accommodate unexpected situations and has designed and constructed alternate arrangements for traffic such as detours, flyovers, etc. urity Plan Site security plan is not applicable for the project. The site does not institute security in regards to entry to the site, securing commodities, or tools and equipment. The site has established security procedures including visitor sign in and sign out procedure and security guards at every gate. The site has implemented security measures to ensure the preservation of company assets. Protocols have been identified for searches of individuals and their personal property. Searches are conducted randomly. Continuation of Level E, plus the site has ensured that material is not leaving the job site by instituting "lock-ups" for items that are prone to theft. Continuation of Level E, plus the use of electronic security such as security cameras has been implemented. Every Positioning Strategy Machinery positioning strategy is not applicable. There is no strategy for positioning of machinery at the project site. Heavy rigging and lifting studies are accomplished on all critical lifts including evaluation of the machinery, rigging selection, and crane location.	0.511 0.68 0.85 0.00 0.15 0.29 0.44 0.59 0.73 0.00 0.16 0.33
Level D Level E Level F 6. Site Sect Level A Level B Level C Level D Level F 7.Machin Level A Level B Level C Level D	 The project has a traffic control plan, equipment, and an arrangement for daylight traffic control only and has no trained traffic control persons. The project has a traffic control plan and equipment for all times of the day including trained persons for traffic control. Continuation of level E, plus a trained traffic control supervisor. It has an approved contingency plan in place to accommodate unexpected situations and has designed and constructed alternate arrangements for traffic such as detours, flyovers, etc. urity Plan Site security plan is not applicable for the project. The site does not institute security in regards to entry to the site, securing commodities, or tools and equipment. The site controls entry and exit from the site, but does not have any other formal security throughout the site. The site stablished security procedures including visitor sign in and sign out procedure and security guards at every gate. The site has implemented security measures to ensure the preservation of Level D, plus the site has ensured that material is not leaving the job site by instituting "lock-ups" for items that are prone to theft. Continuation of Level E, plus the use of electronic security such as security cameras has been implemented. ery Positioning Strategy Machinery positioning strategy is not applicable. There is no strategy for positioning of machinery at the project site. Heavy rigging and lifting studies are accomplished on all critical lifts including evaluation of the machinery, rigging selection, and crane location. 	0.511 0.68 0.85 0.00 0.15 0.29 0.44 0.59 0.73 0.00 0.16 0.33 0.49
Level D Level F 6. Site Sect Level A Level B Level C Level D Level F 7.Machin Level A Level B Level C Level D	I he project has a traffic control plan, equipment, and an arrangement for daylight traffic control only and has no trained traffic control persons. The project has a traffic control plan and equipment for all times of the day including trained persons for traffic control. Continuation of level E, plus a trained traffic control supervisor. It has an approved contingency plan in place to accommodate unexpected situations and has designed and constructed alternate arrangements for traffic such as detours, flyovers, etc. urity Plan Site security plan is not applicable for the project. The site does not institute security in regards to entry to the site, securing commodities, or tools and equipment. The site does not institute security procedures including visitor sign in and sign out procedure and security guards at every gate. The site has implemented security measures to ensure the preservation of company assets. Protocols have been identified for searches of individuals and their personal property. Searches are conducted randomly. Continuation of Level D, plus the site has ensured that material is not leaving the job site by instituting "lock-ups" for items that are prone to theft. Continuation of Level E, plus the use of electronic security such as security cameras has been implemented. Hery Positioning Strategy Machinery positioning strategy is not applicable. There is no strategy for positioning of machinery at the project site. Heavy rigging and lifting studies are accomplished on all critical lifts including evaluation of the machinery, rigging selection, and crane location. Continuation of Level C, plus planning includes the use of 2D layout and studies to aid in constructability for locating and utilizing machinery	0.511 0.68 0.85 0.00 0.15 0.29 0.44 0.59 0.73 0.00 0.16 0.33 0.49
Level D Level E Level F 6. Site Sect Level A Level B Level C Level D Level E Level F 7.Machin Level A Level B Level C Level D	I he project has a traffic control plan, equipment, and an arrangement for daylight traffic control only and has no trained traffic control persons. The project has a traffic control plan and equipment for all times of the day including trained persons for traffic control. Continuation of level E, plus a trained traffic control supervisor. It has an approved contingency plan in place to accommodate unexpected situations and has designed and constructed alternate arrangements for traffic such as detours, flyovers, etc. urity Plan Site security plan is not applicable for the project. The site does not institute security in regards to entry to the site, securing commodities, or tools and equipment. The site controls entry and exit from the site, but does not have any other formal security throughout the site. The site has established security procedures including visitor sign in and sign out procedure and security guards at every gate. The site has implemented security measures to ensure the preservation of company assets. Protocols have been identified for searches of individuals and their personal property. Searches are conducted randomly. Continuation of Level D, plus the site has ensured that material is not leaving the job site by instituting "lock-ups" for items that are prone to theft. Continuation of Level E, plus the use of electronic security such as security cameras has been implemented. tery Positioning Strategy Machinery positioning strategy is not applicable. There is no strategy for positioning of machinery at the project site. Heavy rigging and lifting studies are accomplished on all critical lifts including evaluation of the machinery, rigging selection, and crane location. Continuation of Level C, plus planning includes the use of 2D layout and studies to aid in constructability for locating and utilizing machinery.	0.511 0.68 0.85 0.00 0.15 0.29 0.44 0.59 0.73 0.00 0.16 0.33 0.49 0.65
Level D Level E Level F 6. Site Sect Level A Level B Level C Level D Level E Level F 7.Machin Level A Level B Level C Level D Level C	I he project has a traffic control plan, equipment, and an arrangement for daylight traffic control only and has no trained traffic control persons. The project has a traffic control plan and equipment for all times of the day including trained persons for traffic control. Continuation of level E, plus a trained traffic control supervisor. It has an approved contigency plan in place to accommodate unexpected situations and has designed and constructed alternate arrangements for traffic such as detours, flyovers, etc. urity Plan Site security plan is not applicable for the project. The site does not institute security in regards to entry to the site, securing commodities, or tools and equipment. The site controls entry and exit from the site, but does not have any other formal security throughout the site. The site has established security procedures including visitor sign in and sign out procedure and security guards at every gate. The site has implemented security measures to ensure the preservation of company assets. Protocols have been identified for searches of individuals and their personal property. Searches are conducted randomly. Continuation of Level D, plus the site has ensured that material is not leaving the job site by instituting "lock-ups" for items that are prone to theft. Continuation of Level E, plus the use of electronic security such as security cameras has been implemented. Ery Positioning Strategy is not applicable. There is no strategy for positioning of machinery at the project site. Heavy rigging and lifting studies are accomplished on all critical lifts including evaluation of the machinery, rigging selection, and crane location. Continuation of Level C, plus planning includes the use of 2D layout and studies to aid in constructability for locating and utilizing machinery.	0.511 0.68 0.85 0.00 0.15 0.29 0.44 0.59 0.73 0.00 0.16 0.33 0.49 0.65

474

CUTS



Level F	Continuation of Level E, plus planning includes the use of 3D layout studies and 3D modelling/visualization to aid in constructability for location and utilizing machinery.	0.82
8 Project	Start-un Plan	
Level A	No start-up plan exists	0.00
Level B	A partial start-up plan has been prepared; the plan has not been communicated to the concerned stakeholders.	0.17
Level C	A basic start-up plan has been developed with input from the project participants, but the plan has not been implemented.	0.34
Level D	Continuation of Level C, plus with considerations for interfaces among sub-contractors or project participants. A start-up plan has been developed that identifies the duties and responsibilities of each stakeholder.	0.51
Level E	Continuation of Level D, plus with consideration for cost analysis and detailed scheduling components. The plan is well communicated to all the stakeholders.	0.68
Level F	Continuation of Level E, plus with the plan being implemented on the project by incorporating feedbacks from the stakeholders or project participants.	0.85
9.Project	Completion Plan	
Level A	The project completion requirement or handover procedure is not applicable.	0.00
Level B	The project completion requirement or handover procedure has not been identified.	0.17
Level C	The project has a handover procedure that defines the parameters of project completion and delineates the requirements for the handover.	0.33
Level D	The project has a formal handover process that defines the necessary documentation, parameters of completion and other issues to assure proper handover of a project.	0.50
Level E	Continuation of Level D, plus the procedure has been reviewed and agreed by the stakeholders.	0.66
Level F	Continuation of Level E plus the plan is approved by project management team and is reviewed for applicability during all phases of the handover process.	0.83
10.Innov	ations and New Technologies	
Level A	Innovation in new materials, equipment, information systems is not applicable.	0.00
Level B	Innovations and new technologies investigation is not addressed.	0.13
Level C	The project does not have a formal program for the investigation of innovations and new technologies. Implementation of innovations and new systems will only occur after the industry-wide implementation.	0.25
Level D	The organization has an informal program for the investigation of innovations, and they will investigate the feasibility of the new technologies on a regular basis.	0.38
Level E	Continuation of Level D, plus the program is formal to investigate new systems and they will investigate the feasibility of the new technologies on a regular basis.	0.51
Level F	Continuation of Level E, plus they investigate all new technologies using a formal system of rating the new technology.	0.64

University Library



A gateway to Melbourne's research publications

Minerva Access is the Institutional Repository of The University of Melbourne

Author/s: Gurmu, Argaw

Title:

An investigation into construction management practices influencing labour productivity in multi-storey building projects

Date:

2017

Persistent Link: http://hdl.handle.net/11343/194881

File Description: Thesis

Terms and Conditions:

Terms and Conditions: Copyright in works deposited in Minerva Access is retained by the copyright owner. The work may not be altered without permission from the copyright owner. Readers may only download, print and save electronic copies of whole works for their own personal non-commercial use. Any use that exceeds these limits requires permission from the copyright owner. Attribution is essential when quoting or paraphrasing from these works.